ADVISORY COMMITTEE ON BANKRUPTCY RULES

Austin, TX April 22-23, 2014

APPENDICES

TABLE OF CONTENTS

TAB A	APPENDIX A:	FORMS FOR FINAL APPROVAL	5
	A.1 Effectiv	e Date 2014	7
	A.2 Effectiv	e Date 2015	59
TAB B	APPENDIX B:	FORMS FOR PUBLICATION	. 217
тав с	APPENDIX C:	SUMMARY OF COMMENTS ON CHAPTER 13 PLAN	. 419
TAB D	Appendix D:	SUMMARY OF COMMENTS ON PUBLISHED FORMS	. 521

April 22-23, 2014 Page 3 of 596

April 22-23, 2014 Page 4 of 596

APPENDIX A

April 22-23, 2014 Page 5 of 596

April 22-23, 2014 Page 6 of 596

Appendix A Forms for Final Approval Effective Date 2014

April 22-23, 2014 Page 7 of 596

April 22-23, 2014 Page 8 of 596

Draft: May 10, 2013

[Caption as in Form 16A, 16B, or 16D, as appropriate]

NOTICE OF APPEAL AND STATEMENT OF ELECTION

Part 1: Identify the appellant(s)

1.	Name(s) of appellant(s):			
2.	Position of appellant(s) in the adversary proceeding or bankruptcy case that is the subject of this appeal:			
	For appeals in an adversary proceeding.	For appeals in a bankruptcy case and not in an adversary proceeding.		
	Defendant	☐ Debtor		
	Other (describe)	Creditor		
		☐ Trustee		
		Other (describe)		
Part 2	2: Identify the subject of this appe	<u>al</u>		
1.	Describe the judgment, order, or decree ap	opealed from:		
2.	State the date on which the judgment, order	er, or decree was entered:		
Part 3	3: Identify the other parties to the a	appeal		
	e names of all parties to the judgment, order lephone numbers of their attorneys (attach a	, or decree appealed from and the names, addresses dditional pages if necessary):		
1.	Party: Attorney: _			
	- -			
	_			
2.	Party: Attorney: _			
				

April 22-23, 2014 Page 9 of 596

Part 4: Optional election to have appeal heard by District Court (applicable only in certain districts)

If a Bankruptcy Appellate Panel is available in this judicial district, the Bankruptcy Appellate Panel will hear this appeal unless, pursuant to U.S.C. § 158(c)(1), a party elects to have the appeal heard by the United States District Court. If an appellant filing this notice wishes to have the appeal heard by the United States District Court, check below. Do not check the box if the appellant wishes the Bankruptcy Appellate Panel to hear the appeal.

☐ Appellant(s) elect to have the appeal heard by the United States District Court rather than by

the Bankruptcy Appellate Panel.	
Part 5: Sign below	
	Date:
Signature of attorney for appellant(s) (or appellant(s) if not represented by an attorney)	
Name, address, and telephone number of attorney	
(or appellant(s) if not represented by an attorney):	

Fee waiver notice: If appellant is a child support creditor or its representative and appellant has filed the form specified in § 304(g) of the Bankruptcy Reform Act of 1994, no fee is required.

April 22-23, 2014 Page 10 of 596

COMMITTEE NOTE

The form is amended and renumbered. It is amended to add to the Notice of Appeal an optional Statement of Election to have the appeal heard by the district court rather than by the bankruptcy appellate panel. Current Rule 8005(a) eliminates the requirement, imposed by former Rule 8001(e), that a separate document be used in making an election to have an appeal heard by the district court rather than the bankruptcy appellate panel. It instead requires a statement that conforms substantially to the Official Form for such an election. Form 17A effectuates Rule 8005(a)'s requirement for election by an appellant by combining the notice of appeal and statement of election. It thereby facilitates compliance with the statutory requirement that an appellant wishing to make an election do so at the time of filing the appeal. 28 U.S.C. § 158(c)(1)(A).

The statement of election in Part 4 is applicable only in districts for which appeals to a bankruptcy appellate panel have been authorized. If an appeal is being taken from a bankruptcy court located in a circuit that does not have a bankruptcy appellate panel or in a district that has not authorized appeals to be heard by the circuit's bankruptcy appellate panel, the appellant should not complete Part 4.

When a bankruptcy appellate panel is available to hear an appeal, completion of Part 4 is optional. An appellant that wants its appeal heard by the bankruptcy appellate panel should not complete this part.

The form is renumbered as Official Form 17A because a new companion form—Optional Appellee Statement of Election to Proceed in the District Court—is designated as Official Form 17B, and another bankruptcy appellate form—Certificate of Compliance with Rule 8015(a)(7)(B) or 8016(d)(2)—is designated as Official Form 17C.

The fixed caption has been deleted because the short title caption on the current form is not appropriate if the debtor is the appellant or if the appeal is in an adversary proceeding. *See* 11 U.S.C. § 342(c); Rule 7008; Rule 9004(b). The form should be captioned as in Official Form 16A, Caption (Full); Official Form 16B, Caption (Short Title); or Official Form 16D, Caption for Use in Adversary proceeding, as appropriate.

April 22-23, 2014 Page 11 of 596

April 22-23, 2014 Page 12 of 596

Draft: May 10, 2013

[Caption as in Form 16A, 16B, or 16D, as appropriate]

OPTIONAL APPELLEE STATEMENT OF ELECTION TO PROCEED IN DISTRICT COURT

This form should be filed only if all of the following are true:

- this appeal is pending in a district served by a Bankruptcy Appellate Panel,
- the appellant(s) did not elect in the Notice of Appeal to proceed in the District Court rather than in the Bankruptcy Appellate Panel,
- no other appellee has filed a statement of election to proceed in the district court, and
- you elect to proceed in the District Court.

Part 1: Identify the appellee(s) electing to proceed in the District Court

1.	Name(s) of appellee(s):				
2.	Position of appellee(s) in the adversary proceeding or bankruptcy case that is the subject of this appeal:				
	For appeals in an adversary proceeding.	For appeals in a bankruptcy case and not in an adversary proceeding.			
	☐ Defendant	☐ Debtor			
	Other (describe)	☐ Creditor			
		☐ Trustee			
		Other (describe)			
Apı	ve) elect to have the appeal heard by the Unit pellate Panel. 3: Sign below	ted States District Court rather than by the Bankruptc			
		Date:			
	ure of attorney for appellee(s) (or appellee(s) epresented by an attorney)				
Name,	address, and telephone number of attorney				
(or app	pellee(s) if not represented by an attorney):				

April 22-23, 2014 Page 13 of 596

April 22-23, 2014 Page 14 of 596

COMMITTEE NOTE

This form is new. It is the Official Form for an appellee to state its election to have an appeal heard by the district court rather than by the bankruptcy appellate panel. If an appellee desires to make that election and the appellant or another appellee has not already done so, the appellee must file a statement that conforms substantially to this form within 30 days of service of the Notice of Appeal. 28 U.S.C. § 158(c)(1)(B).

The form is applicable only in districts for which appeals to a bankruptcy appellate panel have been authorized. If an appeal is being taken from a bankruptcy court located in a circuit that does not have a bankruptcy appellate panel or in a district that has not authorized appeals to be heard by the circuit's bankruptcy appellate panel, the appellee should not complete this form.

When a bankruptcy appellate panel is available to hear an appeal, completion of the form is optional. An appellee that wants its appeal heard by the bankruptcy appellate panel should not complete this form.

The form should be captioned as in Official Form 16A, Caption (Full); Official Form 16B, Caption (Short Title); or Official Form 16D, Caption for Use in Adversary proceeding, as appropriate. *See* 11 U.S.C. § 342(c); Rule 7008; Rule 9004(b).

April 22-23, 2014 Page 15 of 596

April 22-23, 2014 Page 16 of 596

Draft: May 10, 2013

[This certification must be appended to your brief if the length of your brief is calculated by maximum number of words or lines of text rather than number of pages.]

Certificate of Compliance With Rule 8015(a)(7)(B) or 8016(d)(2)

This br	ief complies with the type-volume limitation of Rule 8015(a)(7)(B) or 8016(d)(2) because:			
	this brief contains [state the number of] words, excluding the parts of the brief exempted by Rule 8015(a)(7)(B)(iii) or 8016(d)(2)(D), or			
	this brief uses a monospaced typeface having no more than 10½ characters per inch and contains [state the number of] lines of text, excluding the parts of the brief exempted by Rule 8015(a)(7)(B)(iii) or 8016(d)(2)(D).			
Signati	Date:ure			
Print na	ame of person signing certificate of compliance:			

April 22-23, 2014 Page 17 of 596

April 22-23, 2014 Page 18 of 596

COMMITTEE NOTE

This form is new. When the length of a brief is calculated by the maximum number of words or lines of text rather than by number of pages, Rules 8015(a)(7)(C) and 8016(d)(3) require an attorney or unrepresented party to certify that the brief complies with the applicable type-volume limitation. Completion of this form satisfies that certification requirement. This form is not needed if the brief meets the applicable page limitation under Rule 8015(a)(7)(A) or 8016(d)(1).

The form does not include a caption because it is included in the brief.

April 22-23, 2014 Page 19 of 596

April 22-23, 2014 Page 20 of 596

Instructions for the Chapter 7 Statement of Your Current Monthly Income and Means Test Calculation

United States Bankruptcy Court 12/01/14

If you are filing under chapter 11, 12, or 13, do not fill out this form.

How to fill out these forms

Official Forms 22A–1 and 22A –2 determine whether your income and expenses create a presumption of abuse that may prevent you from obtaining relief from your debts under chapter 7 of the Bankruptcy Code. Chapter 7 relief can be denied to a person who has primarily consumer debts if the court finds that the person has enough income to repay creditors an amount that, under the Bankruptcy Code, would be a sufficient portion of their claims.

You must file 22A –1, the *Chapter 7 Statement of Your Current Monthly Income* (Official Form 22A –1) if you are an individual filing for bankruptcy under chapter 7. This form will determine your current monthly income and compare whether your income is more than the median income for households of the same size in your state. If your income is not above the median, there is no presumption of abuse and you will not have to fill out the second form.

Similarly, Statement of Exemption from Presumption of Abuse Under § 707(b)(2) (Official Form 22A-1Supp) determines whether you may be exempted from the presumption of abuse because you do not have primarily consumer debts or because you have provided certain military or homeland defense services. If one of these exemptions applies, you should file a supplement, Form 22A-1Supp, and verify the supplement by completing Part 3 of Form 22A-1. If you qualify for an exemption, you are not required to fill out any part of Form 22A-1 other than the verification. If the exemptions do not apply, you should complete all of the parts of Form 22A-1 and file it without the supplemental form.

If you and your spouse are filing together, you and your spouse may file a single Form 22A-1. However, if an exemption on Form 22A-1Supp applies to only one of you, separate forms may be required.

11 U.S.C. § 707(b)(2)(C).

If your completed Form 22A-1 shows income above the median, you must file the second form, *Chapter 7 Means Test Calculation* (Official Form 22A –2). The calculations on this form—sometimes called the *Means Test*—reduce your income by living expenses and payment of certain debts, resulting in an amount available to pay other debts. If this amount is high enough, it will give rise to a *presumption of abuse*. A presumption of abuse does not mean you are actually trying to abuse the bankruptcy system. Rather, the presumption simply means that you are presumed to have enough income that you should not be granted relief under chapter 7. You may overcome the presumption by showing special circumstances that reduce your income or increase your expenses.

If you cannot obtain relief under chapter 7, you may be eligible to continue under another chapter of the Bankruptcy Code and pay creditors over a period of time.

Read each question carefully. You may not be required to answer every question on this form. For example, your military status may determine whether you must fill out the entire form. The instructions will alert you if you may skip questions.

If you have nothing to report for a line, write \$0.

Information for completing the forms

To fill out several lines of the forms, you must look up information provided on websites or from other sources. For information to complete line 13 of Form 22A-1 and lines 6-15, 30, and 36 of Form 22A-2, go

to: www.justice.gov/ust/eo/bapcpa/meanstesting.htm.

For the *Bankruptcy Basics* information referred to on line 36 of Form 22A-2, go

to: www.uscourts.gov/FederalCourts/Bankruptcy/BankruptcyBasics.aspx.

If you do not have a computer with internet access, you may be able to use a public computer at the bankruptcy clerk's office or at a public library.

Instructions for Official Form 22A-1 Chapter 7 Statement of Your Current Monthly Income and Means Test Calculation

Understand the terms used in the form

This form uses *you* and *Debtor 1* to refer to a debtor filing alone. A married couple may file a bankruptcy case together—called a *joint case*—and in joint cases, this form uses *you* to ask for information from both debtors. When information is needed about the spouses separately, the form uses *Debtor 1* and *Debtor 2* to distinguish between them. In joint cases, one of the spouses must report information as *Debtor 1* and the other as *Debtor 2*. The same person must be *Debtor 1* in all of the forms.

Things to remember when filling out these forms

- Be as complete and accurate as possible.
- If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known).
- If two married people are filing together, both are equally responsible for supplying correct information.

Do not include these instructions when you submit your bankruptcy forms to the court. Keep them for your records.

Instructions for Official Form 22A-1 Chapter 7 Statement of Your Current Monthly Income and Means Test Calculation

page 2

Fill in this information to identify your case:	Check one box only as directed in this form and in
	Form 22A-1Supp:
Debtor 1 First Name Middle Name Last Name	
Debtor 2	1. There is no presumption of abuse.
(Spouse, if filing) First Name Middle Name Last Name	2. The presumption of abuse will be calculated under
United States Bankruptcy Court for the: District of	Chapter 7 Means Test Calculation (Official Form 22A–2).
(State) Case number	3. The Means Test does not apply now because of
(If known)	qualified military service but it could apply later.
	☐ Check if this is an amended filing
Official Form 22A—1	
Chapter 7 Statement of Your Current M	ionthly income 12/14
Be as complete and accurate as possible. If two married people are filing toget	her both are equally responsible for being accurate. If more space
is needed, attach a separate sheet to this form. Include the line number to which	
pages, write your name and case number (if known). If you believe that you are	exempted from a presumption of abuse because you do not have
primarily consumer debts or because of qualifying military service, complete a	nd file Statement of Exemption from Presumption of Abuse Under
§ 707(b)(2) (Official Form 22A-1Supp) with this form.	
Part 1: Calculate Your Current Monthly Income	
What is your marital and filing status? Check one only.	
Not married. Fill out Column A, lines 2-11.	
☐ Married and your spouse is filing with you. Fill out both Columns A and	3, lines 2-11.
Married and your spouse is NOT filing with you. You and your spouse	are:
Living in the same household and are not legally separated. Fill or	ut both Columns A and B. lines 2-11.
Living separately or are legally separated. Fill out Column A, lines 2	·
under penalty of perjury that you and your spouse are legally separate	
are living apart for reasons that do not include evading the Means Tes	
Fill in the average monthly income that you received from all sources, de	ived during the 6 full months before you file this bankruptcy
case. 11 U.S.C. § 101(10A). For example, if you are filing on September 15, th	
amount of your monthly income varied during the 6 months, add the income for include any income amount more than once. For example, if both spouses own	
one column only. If you have nothing to report for any line, write \$0 in the space	
	Column A Column B
	For you Debtor 2 or
	non-filing spouse
2. Your gross wages, salary, tips, bonuses, overtime, and commissions (bef	ore all
payroll deductions).	·
Alimony and maintenance payments. Do not include payments from a spous Column B is filled in.	e if \$
	· · · · · · · · · · · · · · · · · · ·
 All amounts from any source which are regularly paid for household expe of you or your dependents, including child support. Include regular contribution 	
from an unmarried partner, members of your household, your dependents, pare	
and roommates. Include regular contributions from a spouse only if Column B is	s not \$
filled in. Do not include payments you listed on line 3.	-
5. Net income from operating a business, profession, or farm	
Gross receipts (before all deductions) \$	
Ordinary and necessary operating expenses - \$	
Net monthly income from a business, profession, or farm \$ Cop	y here → \$
6. Net income from rental and other real property Gross receipts (before all deductions)	
Gross receipts (before all deductions) \$ Ordinary and necessary operating expenses -\$	
	•
Net monthly income from rental or other real property \$ Co	· · · · · · · · · · · · · · · · · · ·
7. Interest, dividends, and royalties	\$

tor 1 First Name Middle Name Last Name		Case number (if known)_		
		Column A For you	Column B Debtor 2 or non-filing spouse	
Unemployment compensation		\$	\$	
Do not enter the amount if you contend that the amount received under the Social Security Act. Instead, list it here:	_			
For you\$				
For your spouse\$	<u></u>			
Pension or retirement income. Do not include any amount benefit under the Social Security Act.	t received that was a	\$	\$	
Income from all other sources not listed above. Specify to Do not include any benefits received under the Social Secur as a victim of a war crime, a crime against humanity, or interterrorism. If necessary, list other sources on a separate page	rity Act or payments received rnational or domestic			
10a		\$	\$	
10b		\$	\$	
10c. Total amounts from separate pages, if any.		+\$	+ \$	
Calculate your total current monthly income. Add lines 2 column. Then add the total for Column A to the total for Column		\$	+ \$	= \$Total current
Calculate your current monthly income for the year. Follo 12a. Copy your total current monthly income from line 11		Сор	y line 11 here → 12a.	\$
Multiply by 12 (the number of months in a year).			Г	x 12
12b. The result is your annual income for this part of the fo	rm.		12b.	\$
Calculate the median family income that applies to you.	Follow these steps:			
Fill in the state in which you live.				
Fill in the number of people in your household.			-	
Fill in the median family income for your state and size of hor To find a list of applicable median income amounts, go onling instructions for this form. This list may also be available at the	ne using the link specified in t		13.	\$
How do the lines compare?	.,.,			
14a. ☐ Line 12b is less than or equal to line 13. On the top Go to Part 3.	of page 1, check box 1, The	ere is no presumptio	on of abuse.	
14b. Line 12b is more than line 13. On the top of page 1 Go to Part 3 and fill out Form 22A–2.	, check box 2, The presump	tion of abuse is det	ermined by Form 22A	2.
art 3: Sign Below				
By signing here, I declare under penalty of perjury the	nat the information on this sta	atement and in any	attachments is true an	d correct.
×	*			
Signature of Debtor 1	Sign	nature of Debtor 2		
Date	Dat	e	, _	
If you shocked line 14s, do NOT fill out or file Form	204 0			

If you checked line 14b, fill out Form 22A-2 and file it with this form.

Fill in this information to identify your case:				
Debtor 1				
_	First Name	Middle Name	Last Name	
Debtor 2				
(Spouse, if filing)	First Name	Middle Name	Last Name	
United States Bankruptcy Court for the:			District of(State)	
Case number (If known)				

Check the appropriate box as directed in lines 40 or 42:	
According to the calculations required by this Statement:	
☐ 1. There is no presumption of abuse.	
☐ 2. There is a presumption of abuse.	
☐ Check if this is an amended filing	

Official Form 22A-2

Chapter 7 Means Test Calculation

12/14

To fill out this form, you will need your completed copy of Chapter 7 Statement of Your Current Monthly Income (Official Form 22A-1).

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for being accurate. If more space is needed, attach a separate sheet to this form. Include the line number to which the additional information applies. On the top of any additional pages, write your name and case number (if known).

Part 1:	Determine Your Adjusted Income			
1. Copy	your total current monthly income.	Copy line 11 from Offic	ial Form 22A-1 here→1	· \$
2. Did y	ou fill out Column B in Part 1 of Form 22A–1?			
□ N	o. Fill in \$0 on line 3d.			
□ Y	es. Is your spouse filing with you?			
	No. Go to line 3.			
	Yes. Fill in \$0 on line 3d.			
On lir used	st your current monthly income by subtracting any part of your sehold expenses of you or your dependents. Follow these steps: ne 11, Column B of Form 22A–1, was any amount of the income your for the household expenses of you or your dependents? o. Fill in 0 on line 3d. es. Fill in the information below:			
	State each purpose for which the income was used For example, the income is used to pay your spouse's tax debt or to support people other than you or your dependents	Fill in the amount you are subtracting from your spouse's income		
	3a	\$		
	3b	\$		
	3c	+ \$		
	3d. Total. Add lines 3a, 3b, and 3c	\$	Copy total here →3d.	- \$
4. Adjus	st your current monthly income. Subtract line 3d from line 1.			\$

First Name

Part 2: Calculate Your Deductions from Your Income

Middle Name

The Internal Revenue Service (IRS) issues National and Local Standards for certain expense amounts. Use these amounts to answer the questions in lines 6-15. To find the IRS standards, go online using the link specified in the separate instructions for this form. This information may also be available at the bankruptcy clerk's office.

Deduct the expense amounts set out in lines 6-15 regardless of your actual expense. In later parts of the form, you will use some of your actual expenses if they are higher than the standards. Do not deduct any amounts that you subtracted from your spouse's income in line 3 and do not deduct any operating expenses that you subtracted from income in lines 5 and 6 of Form 22A–1.

If your expenses differ from month to month, enter the average expense.

Whenever this part of the form refers to you, it means both you and your spouse if Column B of Form 22A-1 is filled in.

5. The number of people used in determining your deductions from income

Last Name

Fill in the number of people who could be claimed as exemptions on your federal income tax return, plus the number of any additional dependents whom you support. This number may be different from the number of people in your household.

National Standards

You must use the IRS National Standards to answer the guestions in lines 6-7.

6. **Food, clothing, and other items:** Using the number of people you entered in line 5 and the IRS National Standards, fill in the dollar amount for food, clothing, and other items.

\$_____

7. **Out-of-pocket health care allowance:** Using the number of people you entered in line 5 and the IRS National Standards, fill in the dollar amount for out-of-pocket health care. The number of people is split into two categories—people who are under 65 and people who are 65 or older—because older people have a higher IRS allowance for health care costs. If your actual expenses are higher than this IRS amount, you may deduct the additional amount on line 22.

People who are under 65 years of age

7a. Out-of-pocket health care allowance per person

7b. Number of people who are under 65

X

7c. Subtotal. Multiply line 7a by line 7b.

Copy line 7c

\$_____

People who are 65 years of age or older

7d. Out-of-pocket health care allowance per person

7e. Number of people who are 65 or older

7f. **Subtotal.** Multiply line 7d by line 7e.

Copy line 7f
here → + \$

g. Total. Add lines 7c and 7f.....

Copy total here	
7g.	\$

Official Form 22A-2

Local Standards

You must use the IRS Local Standards to answer the questions in lines 8-15.

Based on information from the IRS, the U.S. Trustee Program has divided the IRS Local Standard for housing for bankruptcy purposes into two parts:

- Housing and utilities Insurance and operating expenses
- Housing and utilities Mortgage or rent expenses

To answer the questions in lines 8-9, use the U.S. Trustee Program chart.

To find the chart, go online using the link specified in the separate instructions for this form. This chart may also be available at the bankruptcy clerk's office.

8. Housing and utilities – Insurance and operating expenses: Using the number of people you entered in line 5, fill in the dollar amount listed for your county for insurance and operating expenses.

P

- 9. Housing and utilities Mortgage or rent expenses:
 - 9a. Using the number of people you entered in line 5, fill in the dollar amount listed for your county for mortgage or rent expenses.

9a. \$_____

9b. Total average monthly payment for all mortgages and other debts secured by your home.

To calculate the total average monthly payment, add all amounts that are contractually due to each secured creditor in the 60 months after you file for bankruptcy. Then divide by 60.

Name of the creditor	Average monthly payment	
	\$	
	\$	
	+ \$	
9b. Total average monthly payment	\$ C	opy li ere 🖜

Repeat this amount on line 33a.

9c. Net mortgage or rent expense.

Subtract line 9b (total average monthly payment) from line 9a (mortgage or rent expense). If this amount is less than \$0, enter \$0.

9c. \$_____ | Copy | line 9c | \$_____

10. If you claim that the U.S. Trustee Program's division of the IRS Local Standard for housing is incorrect and affects the calculation of your monthly expenses, fill in any additional amount you claim.

\$_____

Explain why:

- 11. Local transportation expenses: Check the number of vehicles for which you claim an ownership or operating expense.
 - 0. Go to line 14.
 - 1. Go to line 12.
 - 2 or more. Go to line 12.
- 12. **Vehicle operation expense:** Using the IRS Local Standards and the number of vehicles for which you claim the operating expenses, fill in the *Operating Costs* that apply for your Census region or metropolitan statistical area.

13. **Vehicle ownership or lease expense:** Using the IRS Local Standards, calculate the net ownership or lease expense for each vehicle below. You may not claim the expense if you do not make any loan or lease payments on the vehicle. In addition, you may not claim the expense for more than two vehicles.

Vehicle 1

Describe Vehicle 1:

13a. Ownership or leasing costs using IRS Local Standard

13a. **\$**

13b. Average monthly payment for all debts secured by Vehicle 1.

Do not include costs for leased vehicles.

Middle Name

To calculate the average monthly payment here and on line 13e, add all amounts that are contractually due to each secured creditor in the 60 months after you filed for bankruptcy. Then divide by 60.

Name of each creditor for Vehicle 1

Average monthly payment

Repeat this amount on line 33b.

13c. Net Vehicle 1 ownership or lease expense Subtract line 13b from line 13a. If this amount is less than \$0, enter \$0. Copy net Vehicle 1 expense here

\$

Vehicle 2

Describe Vehicle 2:

13d. Ownership or leasing costs using IRS Local Standard

13d. ¢

Copy 13b

here -

 Average monthly payment for all debts secured by Vehicle 2. Do not include costs for leased vehicles.

Name of each creditor for Vehicle 2 Average monthly payment

Copy 13e here \$ Repeat this amount on line 33c.

13f. Net Vehicle 2 ownership or lease expense

Subtract line 13e from 13d. If this amount is less than \$0, enter \$0.

Copy net Vehicle 2 expense here....

\$_____

14. **Public transportation expense**: If you claimed 0 vehicles in line 11, using the IRS Local Standards, fill in the *Public Transportation* expense allowance regardless of whether you use public transportation.

\$

15. Additional public transportation expense: If you claimed 1 or more vehicles in line 11 and if you claim that you may also deduct a public transportation expense, you may fill in what you believe is the appropriate expense, but you may not claim more than the IRS Local Standard for *Public Transportation*.

\$_____

Additional Expense Deductions These are additional deductions allowed by the Means Test. Note: Do not include any expense allowances listed in lines 6-24.					
25. Health insurance, disability insurance, and health savings account expenses. The monthly expenses for health insurance, disability insurance, and health savings accounts that are reasonably necessary for yourself, your spouse, or your dependents.					
Health insurance \$					
Disability insurance \$					
Health savings account + \$					
Total \$ Copy total here→	\$				
Do you actually spend this total amount?					
□ No. How much do you actually spend? □ Yes					
26. Continued contributions to the care of household or family members. The actual monthly expenses that you will continue to pay for the reasonable and necessary care and support of an elderly, chronically ill, or disabled member of your household or member of your immediate family who is unable to pay for such expenses.	\$				
27. Protection against family violence. The reasonably necessary monthly expenses that you incur to maintain the safety of you and your family under the Family Violence Prevention and Services Act or other federal laws that apply.	\$				
By law, the court must keep the nature of these expenses confidential.					
28. Additional home energy costs. Your home energy costs are included in your non-mortgage housing and utilities allowance on line 8.					
If you believe that you have home energy costs that are more than the home energy costs included in the non-mortgage housing and utilities allowance, then fill in the excess amount of home energy costs.	\$				
You must give your case trustee documentation of your actual expenses, and you must show that the additional amount claimed is reasonable and necessary.					
29. Education expenses for dependent children who are younger than 18. The monthly expenses (not more than \$156.25* per child) that you pay for your dependent children who are younger than 18 years old to attend a private or public elementary or secondary school.	\$				
You must give your case trustee documentation of your actual expenses, and you must explain why the amount claimed is reasonable and necessary and not already accounted for in lines 6-23.					
* Subject to adjustment on 4/01/16, and every 3 years after that for cases begun on or after the date of adjustment.					
30. Additional food and clothing expense. The monthly amount by which your actual food and clothing expenses are higher than the combined food and clothing allowances in the IRS National Standards. That amount cannot be more than 5% of the food and clothing allowances in the IRS National Standards.	\$				
To find a chart showing the maximum additional allowance, go online using the link specified in the separate instructions for this form. This chart may also be available at the bankruptcy clerk's office.					
You must show that the additional amount claimed is reasonable and necessary.					
31. Continuing charitable contributions. The amount that you will continue to contribute in the form of cash or financial instruments to a religious or charitable organization. 26 U.S.C. § 170(c)(1)-(2).	\$				
32. Add all of the additional expense deductions. Add lines 25 through 31.	\$				

Deductions for Debt Payment

33. For debts that are secured by an interest in property that you own, including home mortgages, vehicle loans, and other secured debt, fill in lines 33a through 33g.

To calculate the total average monthly payment, add all amounts that are contractually due to each secured creditor in the 60 months after you file for bankruptcy. Then divide by 60.

Loans on your first two vehicles: 33b. Copy line 13b here.			Mortgages on your home:			Average monthly payment				
33b. Copy line 13b here. 33c. Copy line 13e here. Name of each creditor for other secured debt ldentify property that secures the debt include taxes or insurance? 33d. 33e. 33e. 33f. 37e. 3	3	За.	Copy line 9b here			\$				
Name of each creditor for other secured debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures the debt Identify property that secures Identify property Identify proper			Loans on your first two vehicles:							
Name of each creditor for other secured debt Identify property that secures include taxes or insurance? 33d	3	3b.	Copy line 13b here			\$				
the debt include taxes or insurance? No	3	Зс.	Copy line 13e here		>	\$				
33d	N	am	e of each creditor for other secured debt		include taxes or					
33e	33	8d			_ 110	\$				
33f Yes + \$ 23g. Total average monthly payment. Add lines 23g through 23f.	33	8e				\$				
22a Lotal avorago monthly naymont Add linoc 22a through 22t	33	ßf			_ 110	+ \$				
	33g	. T	otal average monthly payment. Add lines	33a through 33f		\$	1 11 -		Ç	\$

- 34. Are any debts that you listed in line 33 secured by your primary residence, a vehicle, or other property necessary for your support or the support of your dependents?
 - ☐ No. Go to line 35.
 - Yes. State any amount that you must pay to a creditor, in addition to the payments listed in line 34, to keep possession of your property (called the *cure amount*). Next, divide by 60 and fill in the information below.

Name of the creditor	Identify property that secures the debt	Total cure amount		Monthly cure amount		
		\$	÷ 60 =	\$		
		\$	÷ 60 =	\$		
		\$	÷ 60 =	+ \$		
			Total	\$	Copy total here	\$

- 35. Do you owe any priority claims such as a priority tax, child support, or alimony—that are past due as of the filing date of your bankruptcy case? 11 U.S.C. \S 507.
 - ☐ No. Go to line 36.
 - Yes. Fill in the total amount of all of these priority claims. Do not include current or ongoing priority claims, such as those you listed in line 19.

Total amount of all past-due priority claims.....

÷ 60 =	

\$_____

Official Form 22A-2

* Subject to adjustment on 4/01/16, and every 3 years after that for cases filed on or after the date of adjustment.

☐ The line 39d is at least \$7,475*, but not more than \$12,475*. Go to line 41.

Middle Name

41.	41a.	Fill in the amount of your total nonpriority unsecured debt. If you filled out A
		Summary of Your Assets and Liabilities and Certain Statistical Information Schedules
		(Official Form 6), you may refer to line 3b on that form.

38a. \$_____

41b. **25% of your total nonpriority unsecured debt.** 11 U.S.C. § 707(b)(2)(A)(i)(l) Multiply line 41a by 0.25.

Х	.25		
\$_		Copy here	\$

42. Determine whether the income you have left over after subtracting all allowed deductions is enough to pay 25% of your unsecured, nonpriority debt.

Check the box that applies:

- ☐ Line 39d is less than line 41b. On the top of page 1 of this form, check box 1, *There is no presumption of abuse.* Go to Part 5.
- Line 39d is equal to or more than line 41b. On the top of page 1 of this form, check box 2, *There is a presumption of abuse*. You may fill out Part 4 if you claim special circumstances. Then go to Part 5.

Part 4: Give Details About Special Circumstances

- 43. Do you have any special circumstances that justify additional expenses or adjustments of current monthly income for which there is no reasonable alternative? 11 U.S.C. § 707(b)(2)(B).
 - No. Go to Part 5.
 - ☐ Yes. Fill in the following information. All figures should reflect your average monthly expense or income adjustment for each item. You may include expenses you listed in line 25.

You must give a detailed explanation of the special circumstances that make the expenses or income adjustments necessary and reasonable. You must also give your case trustee documentation of your actual expenses or income adjustments.

Give a detailed explanation of the special circumstances	Average monthly expense or income adjustment
	\$
	\$
	\$
	\$

Part 5: Sign Below

By signing here, I declare under penalty of perjury that the information on this statement and in any attachments is true and correct.

Signature of Debtor 1 Signature of Debtor 2

 Date ______MM / DD / YYYY

April 22-23, 2014 Page 34 of 596

Fill in this information to identify your case:				
Debtor 1 _	First Name	Middle Name	Last Name	
Debtor 2				
(Spouse, if filing)	First Name	Middle Name	Last Name	
United States B	ankruptcy Court for	the:	District of	
			(State)	
Case number (If known)				

Draft March 14, 2014

☐ Check if this is an amended filing

Official Form 22B

Chapter 11 Statement of Your Current Monthly Income

12/14

You must file this form if you are an individual and are filing for bankruptcy under Chapter 11. If more space is needed, attach a separate sheet to this form. Include the line number to which the additional information applies. On the top of any additional pages, write your name and case number (if known).

Pá	rt 1: Calculate Your Average Monthly Income								
1.	What is your marital and filing status? Check one only.								
	 Not married. Fill out Column A, lines 2-11. ■ Married and your spouse is filing with you. Fill out both Columns A and B, lines 2-11. 								
	lacksquare Married and your spouse is NOT filing with you. Fill $lacksquare$	out Column A, lin	es 2-11.						
	Fill in the average monthly income that you received from all sources, derived during the 6 full months before you file this bankruptcy case. 11 U.S.C. § 101(10A). For example, if you are filing on September 15, the 6-month period would be March 1 through August 31. If the amount of your monthly income varied during the 6 months, add the income for all 6 months and divide the total by 6. Fill in the result. Do not include any income amount more than once. For example, if both spouses own the same rental property, put the income from that property in one column only. If you have nothing to report for any line, write \$0 in the space.								
				Column A For Debtor 1	Column B Debtor 2 or non-filing spouse				
2.	Your gross wages, salary, tips, bonuses, overtime, and opayroll deductions).	commissions (b	pefore all	\$	\$				
3.	Alimony and maintenance payments. Do not include payments. Do not include payments.	ments from a spo	ouse if	\$	\$				
4.	All amounts from any source which are regularly paid for household expenses of you or your dependents, including child support. Include regular contributions from an unmarried partner, members of your household, your dependents, parents, and roommates. Include regular contributions from a spouse only if Column B is not filled in. Do not include payments you listed on line 3.								
5.	Net income from operating a business, profession, or fa	ırm							
	Gross receipts (before all deductions)	\$							
	Ordinary and necessary operating expenses	- \$							
	Net monthly income from a business, profession, or farm	\$	Copy here →	\$	\$				
6.	Net income from rental and other real property								
	Gross receipts (before all deductions)	\$							
	Ordinary and necessary operating expenses	- \$							
	Net monthly income from rental or other real property	\$	Copy here→	\$	\$				

Official Form 22B

14. Your current monthly income. Subtract line 13d from line 12.

14. \$ _____

Chapter 11 Statement of Your Current Monthly Income

page 2

Official Form 22B

signing here, under penalty of perjury I declare that the information on this statement and in any attachments is true and correct. Signature of Debtor 1 Signature of Debtor 2	2: Sign Below	
×		
Signature of Debtor 1 Signature of Debtor 2		
Signature of Debtor 1 Signature of Debtor 2	signing here, under penalty of perjury I dec	clare that the information on this statement and in any attachments is true and correct.
	signing here, under penalty of perjury I dec	clare that the information on this statement and in any attachments is true and correct.

April 22-23, 2014 Page 38 of 596

Official Form 22B

Instructions for the Chapter 11 Statement of Your Current Monthly Income

United States Bankruptcy Court 12/01/14

If you are filing under chapter 7, 12, or 13, do not fill out this form.

How to Fill Out this Form

You must file the *Chapter 11 Statement of Your Current Monthly Income* (Official Form 22B) if you are an individual filing for bankruptcy under Chapter 11.

If you have nothing to report for a line, write \$0.

Understand the terms used in the form

This form uses *you* and *Debtor 1* to refer to a debtor filing alone. A married couple may file a bankruptcy case together—called a *joint case*—and in joint cases, this form

uses *you* to ask for information from both debtors. When information is needed about the spouses separately, the form uses *Debtor 1* and *Debtor 2* to distinguish between them. In joint cases, one of the spouses must report information as *Debtor 1* and the other as *Debtor 2*. The same person must be *Debtor 1* in all of the forms.

Things to remember when filling out this form

- Be as complete and accurate as possible.
- If more space is needed, attach a separate sheet to this form. Include the line number to which the additional information applies On the top of any additional pages, write your name and case number (if known).
- If two married people are filing together, both are equally responsible for supplying correct information.

Do not include these instructions when you submit your bankruptcy forms to the court. Keep them for your records.

April 22-23, 2014 Page 39 of 596

April 22-23, 2014 Page 40 of 596

Official Forms 22C-1 and 22C-2

Instructions for the Chapter 13 Statement of Your Current Monthly Income, Calculation of Commitment Period and Chapter 13 Calculation of Your Disposable Income

United States Bankruptcy Court

12/01/14

If you are filing under chapter 7, 11, or 12, do not fill out this form.

How to Fill Out these Forms

Official Forms 22C—1 and 22C—2 determine the commitment period for your payments to creditors, how the amount you may be required to pay to creditors is established, and, in some situations, how much you must pay.

You must file 22C -1, the Chapter 13 Statement of Your Current Monthly Income and Calculation of Commitment Period (Official Form 22C -1) if you are an individual and you are filing under chapter 13. This form will report your current monthly income and determine whether your income is at or below the median income for households of the same size in your state. If your income is equal to or less than the median, you will not have to fill out the second form. Form 22C -1 also will determine your applicable commitment period—the time period for making payments to your creditors, unless the court orders otherwise.

If your income is above the median, you must file the second form, 22C—2, *Chapter13 Calculation of Your Disposable Income*. The calculations on this form—sometimes called the *Means Test*—reduce your income by living expenses and payment of certain debts, resulting in an amount available to pay unsecured debts. Your chapter 13 plan may be required to provide for payment of this amount toward unsecured debts.

Read each question carefully. You may not be required to answer every question on this form. The instructions will alert you if you may skip questions.

Some of the questions require you to go to other sources for information. In those cases, the form has instructions for where to find the information you need.

Generally, if you and your spouse are filing together, you should file one statement together.

Understand the terms used in these form

These forms use *you* and *Debtor 1* to refer to a debtor filing alone. A married couple may file a bankruptcy case together—called a *joint case*—and in joint cases, these forms use *you* to ask for information from both debtors. When information is needed about the spouses separately, the forms use *Debtor 1* and *Debtor 2* to distinguish between them. In joint cases, one of the spouses must report information as *Debtor 1* and the other as *Debtor 2*. The same person must be *Debtor 1* in all of the forms.

Information for completing the forms

To fill out several lines of the forms, you must look up information provided on websites or from other sources. For information to complete line 16c of Form 22C-1 and lines 6-15, 30, and 36 of Form 22C-2, go to: www.justice.gov/ust/eo/bapcpa/meanstesting.htm.

If you do not have a computer with internet access, you may be able to use a public computer at the bankruptcy clerk's office or at a public library.

Things to remember when filling out this form

- Be as complete and accurate as possible.
- If more space is needed, attach a separate sheet to this form. Include the line number to which the additional information applies. On the top of any additional pages, write your name and case number (if known).
- If two married people are filing together, both are equally responsible for supplying correct information.

Do not include these instructions when you submit your bankruptcy forms to the court. Keep them for your records.

April 22-23, 2014 Page 41 of 596

April 22-23, 2014 Page 42 of 596

Fill in this information to identify your case:					
Debtor 1	First Name	Middle Name	Last Name		
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name		
United States Bankruptcy Court for the:			District of(State)		
Case number (If known)					

Check as directed in lines 17 and 21:
According to the calculations required by this Statement:
1. Disposable income is not determined under 11 U.S.C. § 1325(b)(3).
2. Disposable income is determined under 11 U.S.C. § 1325(b)(3).
☐ 3. The commitment period is 3 years.
4. The commitment period is 5 years.

☐ Check if this is an amended filing

Official Form 22C-1

Chapter 13 Statement of Your Current Monthly Income and Calculation of Commitment Period

12/14

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for being accurate. If more space is needed, attach a separate sheet to this form. Include the line number to which the additional information applies. On the top of any additional pages, write your name and case number (if known).

Pa	Trt 1: Calculate Your Average Monthly Incom	ie				
1.	What is your marital and filing status? Check one only. Not married. Fill out Column A, lines 2-11. Married. Fill out both Columns A and B, lines 2-11.					
	Fill in the average monthly income that you from all so 11 U.S.C. § 101(10A). For example, if you are filing on Se of your monthly income varied during the 6 months, add the any income amount more than once. For example, if both column only. If you have nothing to report for any line, write	eptember 15, the 6-mor ne income for all 6 mor spouses own the sam	nth period nths and d	would be March 1 thi	rough August 31. If the a	mount clude
				Column A For Debtor 1	Column B Debtor 2 or non-filing spouse	
2.	Your gross wages, salary, tips, bonuses, overtime, and payroll deductions).	d commissions (before	re all	\$	\$	
3.	. Alimony and maintenance payments. Do not include payments from a spouse if Column B is filled in.			\$	\$	
4.	All amounts from any source which are regularly paid you or your dependents, including child support. Include an unmarried partner, members of your household, your croommates. Include regular contributions from a spouse of in. Do not include payments you listed on line 3.	\$	\$			
5.	Net income from operating a business, profession, or	farm				
	Gross receipts (before all deductions)	\$				
	Ordinary and necessary operating expenses	- \$	1			
	Net monthly income from a business, profession, or farm	\$	Copy here	\$	\$	
6.	Net income from rental and other real property					
	Gross receipts (before all deductions)	\$				
	Ordinary and necessary operating expenses	- \$	_			
	Net monthly income from rental or other real property	\$	Copy here	\$	\$	

Debtor 1	Case number (if kno	wn)	
First Name Middle Name Last Name			
	Column A For Debtor 1	Column B Debtor 2 or non-filing spouse	
7. Interest, dividends, and royalties	\$	\$	
8. Unemployment compensation	\$	\$	
Do not enter the amount if you contend that the amount received was a benefit under the Social Security Act. Instead, list it here:			
For you\$			
For your spouse \$			
 Pension or retirement income. Do not include any amount received that was a benefit under the Social Security Act. 	\$	\$	
10. Income from all other sources not listed above. Specify the source and amount. Do not include any benefits received under the Social Security Act or payments received as a victim of a war crime, a crime against humanity, or international or domestic terrorism. If necessary, list other sources on a separate page and put the total on line 10c.			
10a	\$	\$	
10b.	\$	\$	
10c. Total amounts from separate pages, if any.	+\$	+ \$	
	- 4		
 Calculate your total average monthly income. Add lines 2 through 10 for each column. Then add the total for Column A to the total for Column B. 	\$	+ \$	= \
12. Copy your total average monthly income from line 11.			\$
13. Calculate the marital adjustment. Check one:			
☐ You are not married. Fill in 0 in line 13d.			
 ☐ You are married and your spouse is filing with you. Fill in 0 in line 13d. ☐ You are married and your spouse is not filing with you. Fill in the amount of the income listed in line 11, Column B, that was NOT regularly or your dependents, such as payment of the spouse's tax liability or the spouse's syour dependents. In lines 13a-c, specify the basis for excluding this income and the amount of incomnecessary, list additional adjustments on a separate page. 	support of someone of	other than you or	
If this adjustment does not apply, enter 0 on line 13d.			
13a	\$	-	
13b		-	
13c	+\$	_	
13d. Total	\$	Copy here. → 13d.	
14. Your current monthly income. Subtract line 13d from line 12.		14.	
15. Calculate your current monthly income for the year. Follow these steps:			\$
			\$
15a. Copy line 14 here 🛨		15a.	\$
15a. Copy line 14 here →		15a.	\$ \$ x 12
••		15a.	\$

De	ebtor 1					Case number (if known)		
		First Name	Middle Name	Last Name				
16.	Calcı	ulate the median f	amily income t	hat applies to you	J. Follow these st	eps:		
	16a.	Fill in the state in v	which you live.					
	16b.	Fill in the number	of people in your	household.				
			olicable median	income amounts, g	go online using th	e link specified in the separate tcy clerk's office.	16c.	\$
17.	How	do the lines comp	pare?					
	17a.					his form, check box 1, <i>Disposable in ble Income</i> (Official Form 22C–2).	ncome is not dete	rmined under 11 U.S.C.
	17b. [§ 1325(b)(3). (Go to Part 3 and			heck box 2, <i>Disposable income is a</i> e Income (Official Form 22C-2). O		
Pa	rt 3:	Calculate \	Your Commit	ment Period Un	nder 11 U.S.C.	§1325(b)(4)		
18.	Сору	your total averag	je monthly inco	me from line 11.			18.	\$
19.	that c		mitment period i			se is not filing with you, and you con you to deduct part of your spouse's		·
		marital adjustment		fill in 0 on line 19a	а.		19a	_ \$
	Subt	ract line 19a from	line 18.				19b	. \$
20.	Calc	ulate your current	monthly incon	ne for the year. Fo	ollow these steps:			
	20a.	Copy line 19b					20a.	\$
		Multiply by 12 (the	number of mon	ths in a year).				x 12
	20b.	The result is your	current monthly	income for the yea	r for this part of th	ne form.	20b.	\$
	20c. C	Copy the median fa	mily income for	your state and size	e of household fro	m line 16c		\$
21.	How	do the lines comp	pare?					
	_ 3	years. Go to Part	4.			the top of page 1 of this form, chec		nmitment period is
		ine 20b is more that heck box 4, <i>The co</i>				the court, on the top of page 1 of th	is form,	
P	art 4:	Sign Below	,					ı
	D.	aigning hara unde	or namalty of nami	umul do aloro that t	ha information on	this statement and in any attachma	nto io truo and a	rea at
	Ву		er penaity of perj	ury i declare that t	ne information on	this statement and in any attachme	ents is true and co	rrect.
		Signature of Debtor	1			Signature of Debtor 2		
		Date	000/			Date		
		MM / DD / Y	1 1 1 1			MM / DD / YYYY		
	If y	ou checked 17a, d	lo NOT fill out or	file Form 22C–2.				
	If y	ou checked 17b, fi	ll out Form 22C-	-2 and file it with th	nis form. On line 3	35 of that form, copy your current mo	onthly income from	n line 14 above.

April 22-23, 2014 Page 46 of 596

Fill in this information to identify your case:					
Debtor 1	First Name	Middle Name	Last Name		
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name		
United States E	Bankruptcy Court fo	or the:	District of(State)		
Case number (If known)					

Draft March 18, 2014

☐ Check if this is an amended filing

Official Form 22C-2

Chapter 13 Calculation of Your Disposable Income

12/14

To fill out this form, you will need your completed copy of Chapter 13 Statement of Your Current Monthly Income and Calculation of Commitment Period (Official Form 22C-1).

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for being accurate. If more space is needed, attach a separate sheet to this form. Include the line number to which the additional information applies. On the top of any additional pages, write your name and case number (if known).

Part 1:

Calculate Your Deductions from Your Income

The Internal Revenue Service (IRS) issues National and Local Standards for certain expense amounts. Use these amounts to answer the questions in lines 6-15. To find the IRS standards, go online using the link specified in the separate instructions for this form. This information may also be available at the bankruptcy clerk's office.

Deduct the expense amounts set out in lines 6-15 regardless of your actual expense. In later parts of the form, you will use some of your actual expenses if they are higher than the standards. Do not include any operating expenses that you subtracted from income in lines 5 and 6 of Form 22C–1, and do not deduct any amounts that you subtracted from your spouse's income in line 13 of Form 22C–1.

If your expenses differ from month to month, enter the average expense.

Note: Line numbers 1-4 are not used in this form. These numbers apply to information required by a similar form used in chapter 7 cases.

5. The number of people used in determining your deductions from income

Fill in the number of people who could be claimed as exemptions on your federal income tax return, plus the number of any additional dependents whom you support. This number may be different from the number of people in your household.

National Standards You must use the IRS National Standards to answer the questions in lines 6-7.

6. **Food, clothing, and other items:** Using the number of people you entered in line 5 and the IRS National Standards, fill in the dollar amount for food, clothing, and other items.

\$		
D.		

7. Out-of-pocket health care allowance: Using the number of people you entered in line 5 and the IRS National Standards, fill in the dollar amount for out-of-pocket health care. The number of people is split into two categories—people who are under 65 and people who are 65 or older—because older people have a higher IRS allowance for health care costs. If your actual expenses are higher than this IRS amount, you may deduct the additional amount on line 22.

April 22-23, 2014 Page 47 of 596

Local **Standards**

You must use the IRS Local Standards to answer the questions in lines 8-15.

7g. **Total**. Add lines 7c and 7f.

Based on information from the IRS, the U.S. Trustee Program has divided the IRS Local Standard for housing for bankruptcy purposes into two parts:

- Housing and utilities Insurance and operating expenses
- Housing and utilities Mortgage or rent expenses

To answer the questions in lines 8-9, use the U.S. Trustee Program chart. To find the chart, go online using the link specified in the separate instructions for this form. This chart may also be available at the bankruptcy clerk's office.

Housing and utilities - Insurance and operating expenses: Using the number of people you entered in line 5, fill in

- Housing and utilities Mortgage or rent expenses:
 - 9a. Using the number of people you entered in line 5, fill in the dollar amount listed for your county for mortgage or rent expenses.

the dollar amount listed for your county for insurance and operating expenses.

9b. Total average monthly payment for all mortgages and other debts secured by your home.

To calculate the total average monthly payment, add all amounts that are contractually due to each secured creditor in the 60 months after you file for bankruptcy. Next divide by 60.

Name of the creditor	Average monthly payment			
	\$ \$			
9b.Total average monthly payment	+ \$	Copy line 9b here→	- \$	Repeat this amount on line 33a.
Net mortgage or rent expense. Subtract line 9b (total average monthly payment) fro expense). If this number is less than \$0, enter \$0.	om line 9a (<i>mortgage</i>	or rent	\$	Copy 9c here→

9c.

Explain why:

\$	Copy 9c here→	\$
----	---------------	----

Copy total

here → 7g.

10. If you claim that the U.S. Trustee Program's division of the IRS Local Standard for housing is incorrect and affects the calculation of your monthly expenses, fill in any additional amount you claim

Official Form 22C-2

14. **Public transportation expense**: If you claimed 0 vehicles in line 11, using the IRS Local Standards, fill in the *Public Transportation* expense allowance regardless of whether you use public transportation.

\$_____

15. Additional public transportation expense: If you claimed 1 or more vehicles in line 11 and if you claim that you may also deduct a public transportation expense, you may fill in what you believe is the appropriate expense, but you may not claim more than the IRS Local Standard for *Public Transportation*.

\$____

First Name

Middle Name

Last Name

Other Necessary Expenses	In addition to the expense deductions listed above, you are allowed your monthly expenses for the following IRS categories.	
employment taxes, s your pay for these ta	onthly amount that you actually pay for federal, state and local taxes, such as income taxes, self- social security taxes, and Medicare taxes. You may include the monthly amount withheld from axes. However, if you expect to receive a tax refund, you must divide the expected refund by 12 mber from the total monthly amount that is withheld to pay for taxes.	\$
Do not include real e	estate, sales, or use taxes.	
Involuntary deduct union dues, and unif	ions: The total monthly payroll deductions that your job requires, such as retirement contributions, orm costs.	•
Do not include amou	ints that are not required by your job, such as voluntary 401(k) contributions or payroll savings.	\$
together, include pay	total monthly premiums that you pay for your own term life insurance. If two married people are filing ments that you make for your spouse's term life insurance. iums for life insurance on your dependents, for a non-filing spouse's life insurance, or for any form of life term.	\$
9. Court-ordered payr	nents: The total monthly amount that you pay as required by the order of a court or administrative busal or child support payments.	\$
Do not include paym	ents on past due obligations for spousal or child support. You will list these obligations in line 35.	
as a condition for	Il monthly amount that you pay for education that is either required: your job, or or mentally challenged dependent child if no public education is available for similar services.	\$
	monthly amount that you pay for childcare, such as babysitting, daycare, nursery, and preschool. ents for any elementary or secondary school education.	\$
required for the heal savings account. Inc	are expenses, excluding insurance costs: The monthly amount that you pay for health care that is the and welfare of you or your dependents and that is not reimbursed by insurance or paid by a health clude only the amount that is more than the total entered in line 7. insurance or health savings accounts should be listed only in line 25.	\$
you and your depend service, to the extendis not reimbursed by Do not include paym	s and telephone services: The total monthly amount that you pay for telecommunication services for dents, such as pagers, call waiting, caller identification, special long distance, or business cell phone to necessary for your health and welfare or that of your dependents or for the production of income, if it your employer. The total monthly amount that you pay for telecommunication services for business cell phone to production of income, if it your employer. The total monthly amount that you pay for telecommunication services for business cell phone to product on the production of income, if it your employer. The total monthly amount that you pay for telecommunication services for dents, such as pagers, call waiting, caller identification, special long distance, or business cell phone to product on the production of income, if it your employer.	+ \$
4. Add all of the experience Add lines 6 through 2	nses allowed under the IRS expense allowances. 23.	\$
Additional Expense Deductions	These are additional deductions allowed by the Means Test. Note: Do not include any expense allowances listed in lines 6-24.	
	lisability insurance, and health savings account expenses. The monthly expenses for health insurance, and health savings accounts that are reasonably necessary for yourself, your spouse, or your	
Health insurance	\$	
Disability insurar	nce \$	
Health savings a	recount + \$	
Total	\$ Copy total here →	. \$
Do you actually	spend this total amount?	
	o you actually spend?	
continue to pay for the	utions to the care of household or family members. The actual monthly expenses that you will ne reasonable and necessary care and support of an elderly, chronically ill, or disabled member of your er of your immediate family who is unable to pay for such expenses.	\$
	family violence. The reasonably necessary monthly expenses that you incur to maintain the safety of under the Family Violence Prevention and Services Act or other federal laws that apply.	\$
By law, the court mu	st keep the nature of these expenses confidential.	

Official Form 22C-2

28.	Additional home energy costs.	Your home energy	costs are included i	n your no	on-mortgage housing ar	nd utilities allowance
	on line 4.					

If you believe that you have home energy costs that are more than the home energy costs included in the non-mortgage housing and utilities allowance, then fill in the excess amount of home energy costs.

You must give your case trustee documentation of your actual expenses, and you must show that the additional amount claimed is reasonable and necessary.

29. Education expenses for dependent children who are younger than 18. The monthly expenses (not more than \$156.25* per child) that you pay for your dependent children who are younger than 18 years old to attend a private or public elementary or secondary school.

You must give your case trustee documentation of your actual expenses, and you must explain why the amount claimed is reasonable and necessary and not already accounted for in lines 6-23.

- Subject to adjustment on 4/01/16, and every 3 years after that for cases begun on or after the date of adjustment.
- 30. Additional food and clothing expense. The monthly amount by which your actual food and clothing expenses are higher than the combined food and clothing allowances in the IRS National Standards. That amount cannot be more than 5% of the food and clothing allowances in the IRS National Standards.

To find a chart showing the maximum additional allowance, go online using the link specified in the separate instructions for this form. This chart may also be available at the bankruptcy clerk's office.

You must show that the additional amount claimed is reasonable and necessary.

31. Continuing charitable contributions. The amount that you will continue to contribute in the form of cash or financial instruments to a religious or charitable organization. 11 U.S.C. § 548(d)3 and (4).

+			
т			

Do not include any amount more than 15% of your gross monthly income.

32. Add all of the additional expense deductions.

Add lines 25 through 31.



Deductions for Debt Payment

33. For debts that are secured by an interest in property that you own, including home mortgages, vehicle loans, and other secured debt, fill in lines 33a through 33g.

To calculate the total average monthly payment, add all amounts that are contractually due to each secured creditor in the 60 months after you file for bankruptcy. Then divide by 60.

			Average monthly payment
Nortgages on your home			
3a. Copy line 9b here			\$
ans on your first two vehicles			
c. Copy line 13b here			\$
c. Copy line 13e here		→	\$
ame of each creditor for other ecured debt	Identify property that secures the debt	Does payment include taxes or insurance?	
		No Yes	\$
e		No Yes	\$
3f		No Yes	+ \$
g. Total average monthly payment	. Add lines 33a through 33f		\$

Official Form 22C-2

34. Are any debts that you listed in line 33 secured by your primary residence, a vehicle, or other property necessary for	or
your support or the support of your dependents?	

No. Go to line 35.

☐ Yes. State any amount that you must pay to a creditor, in addition to the payments listed in line 34, to keep possession of your property (called the cure amount). Next, divide by 60 and fill in the information below.

Name of the creditor	Identify property that secures the debt	Total cure amount		Monthly cure amount
		\$	÷ 60 =	\$
		\$	÷ 60 =	\$
		\$	÷ 60 = +	+ \$

Copy total Total

35. Do you owe any priority claims—such as a priority tax, child support, or alimony—that are past due as of the filing date of your bankruptcy case? 11 U.S.C. § 507.

☐ No. Go to line 36.

☐ Yes. Fill in the total amount of all of these priority claims. Do not include current or ongoing priority claims, such as those you listed in line 19.

÷ 60 Total amount of all past-due priority claims.

36. Projected monthly Chapter 13 plan payment

Current multiplier for your district as stated on the list issued by the Administrative Office of the United States Courts (for districts in Alabama and North Carolina) or by the Executive Office for United States Trustees (for all other districts).

To find a list of district multipliers that includes your district, go online using the link specified in the separate instructions for this form. This list may also be available at the bankruptcy clerk's office.

Copy total Average monthly administrative expense here -

37. Add all of the deductions for debt payment. Add lines 33g through 36.



Total Deductions from Income

38. Add all of the allowed deductions.

Copy line 24, All of the expenses allowed under IRS expense allowances.....

Copy line 32, All of the additional expense deductions.....

Copy line 37, All of the deductions for debt payment.....

Copy here

Total deductions

2-4-2-	D = 4 =	Varm Diamasah				422F/L\/2\
art 2:	Determine	Your Disposab	ie income	Unaer 11	U.S.C. 9	1325(D)(Z)

39.			monthly income from lin							\$
40.	Fill in any reas The monthly av payments for a	sonably ne verage of a dependen h applicabl	ent Monthly Income and of ecessary income you recony child support payments to child, reported in Part I of the nonbankruptcy law to the	eive for support , foster care payn Form 22C-1, tha	for dependent c nents, or disability it you received in	hildren.	\$			
41.	employer withh	eld from w 541(b)(7) p	ment deductions. The mo ages as contributions for q lus all required repayment 62(b)(19).	ualified retiremen	nt plans, as specif		\$			
42.	Total of all dec	ductions a	llowed under 11 U.S.C. §	707(b)(2)(A) . Co	ppy line 38 here	→	\$			
43.	and you have n expenses. You	no reasona must give	rcumstances. If special ci ble alternative, describe th your case trustee a detaile tentation for the expenses.	e special circums ed explanation of	tances and their	enses				
	Describe the s	pecial circu	mstances	Aı	mount of expense					
	43a				\$					
	43b				\$					
	43c			<u>+</u>	\$,				
	43d. Total . Add	d lines 43a	through 43c		\$	Copy 43d	+\$			
44.	Total adjustme	ents. Add l	ines 40 and 43d			→	\$		Copy total here	- \$
		-	disposable income under	r § 1325(b)(2). St	ubtract line 44 fror	m line 39.				\$
46	have changed the time your after you filed	d or are virt case will b your petiti	xpenses. If the income in rually certain to change afte e open, fill in the information, check 22C-1 in the first in when the increase occur	er the date you fil on below. For exa t column, enter lir	ed your bankruptoumple, if the wage ne 2 in the second	cy petition a s reported d column, e	and during increased			
	Form	Line	Reason for change		Date of change		ease or rease?	Amount	of change	
	22C-1 22C-2						ncrease Decrease	\$		
	22C-1 22C-2						ncrease Decrease	\$		
	22C-1 22C-2						ncrease Decrease	\$		
	☐ 22C-1					☐ Ir	ncrease	¢		

Official Form 22C-2

☐ 22C-2

Decrease

Part 4:

Sign Below

By signing here, under penalty of perjury you declare that the information on this statement and in any attachments is true and correct.

X

Signature of Debtor 1

Date _____

X

Signature of Debtor 2

Date _____

COMMITTEE NOTE

Official Forms 22A-1, 22A-1Supp, 22A-2, 22C-1, and 22C-2 are new versions of the "means test" forms used by individuals in chapter 7 and 13, formerly Official Forms 22A and 22C. The original forms were substantially revised as part of the Forms Modernization Project, which was designed so that the individuals completing the forms would do so more accurately and completely. Official Form 22B, used by individuals in chapter 11, has also been revised as part of the project.

The revised versions of the means test forms present the relevant information in a format different from the original versions of the forms. For chapter 7, former Official Form 22A has been split into two forms: 22A-1 and 22A-2. The first form, Official Form 22A-1, Chapter 7 Statement of Your Current Monthly Income, is to be completed by all chapter 7 debtors. It calculates a debtor's current monthly income and compares that calculation to the median income for households of the same size in the debtor's state. The second form, Official Form 22A-2, Chapter 7 Means Test Calculation, is to be completed only by those chapter 7 debtors whose income is above the applicable state median. The prior version of Official Form 22A was introduced by several questions bearing on the applicability of the means test. Debtors who do not have primarily consumer debts, as well as certain members of the armed forces, are exempt from a presumption of abuse under the means test, and so are excused from completing the form. However, the great majority of individual debtors in chapter 7 do not fall within the exemptions. Accordingly, the exemptions from means testing have been placed in a separate supplement, Official Form 22A-1Supp, that will be filed only where applicable, making Form 22A present the relevant information more directly and in a manner consistent with the parallel chapter 13 form.

For the chapter 13 means test, there is a similar split of income and expense calculations. All chapter 13 debtors must complete Official Form 22C-1, *Chapter 13 Statement of Your Current Monthly Income and Calculation of Commitment Period*, which calculates current monthly income and the plan commitment period. Debtors only need to complete the second form, Official Form 22C-2, *Chapter13 Calculation of Your Disposable Income*, if their current monthly income exceeds the applicable median. Form 22C-2 calculates disposable income under 11 U.S.C. § 1325(b)(3), through a report of allowed expense deductions.

April 22-23, 2014 Page 55 of 596

Line 60 of former Official Form 22C has not been repeated in Official Form 22C-2. This line allowed debtors to list, but not deduct from income, "Other Necessary Expense" items that are not included within the categories specified by the Internal Revenue Service. Because debtors are separately allowed to list—and deduct—any expenses arising from special circumstances, former Line 60 was rarely used.

Form 22C-2 also reflects the Supreme Court's decision in *Hamilton v. Lanning*, 130 S. Ct. 2464 (2010). The Court held in *Lanning* that the calculation of a chapter 13 debtor's projected disposable income under § 1325(b) required consideration of changes to income or expenses reported elsewhere on former Official Form 22C that, at the time of plan confirmation, had occurred or were virtually certain to occur. Those changes could result in either an increased or decreased projected disposable income. Because only debtors whose annualized current monthly income exceeds the applicable median family income have their projected disposable income determined by the information provided on both parts of Official Form 22C, only these debtors are required to provide the information about changes to income and expenses on Official Form 22C-1 and 22C-2. Part 3 of Official Form 22C-2 provides for the reporting of those changes.

In reporting changes to income a debtor must indicate whether the amounts reported in Official Form 22C-1—which are monthly averages of various types of income received during the six months prior to the filing of the bankruptcy case—have already changed or are virtually certain to change during the pendency of the case. For each change, the debtor must indicate the line of Official Form 22C-1 on which the amount to be changed was reported, the reason for the change, the date of its occurrence, whether the change is an increase or decrease of income, and the amount of the change. Similarly, in reporting changes to expenses, a debtor must list changes to the debtor's actual expenditures reported in Part 1 of Official Form 22C-2 that are virtually certain to occur while the case is pending. With respect to the deductible amounts reported in Part 1 that are determined by the IRS national and local standards, only changed amounts that result from changed circumstances in the debtor's life—such as the addition of a family member or the surrender of a vehicle—should be reported. For each change in expenses, the same information required to be provided for income changes must be reported.

Unlike former Official Forms 22A and 22C, Official Forms 22A-2 and 22C-2 permit, at line 23, the deduction of cell phone

April 22-23, 2014 Page 56 of 596

expenses necessary for the production of income if those expenses have not been reimbursed by the debtor's employer or deducted by the debtor in calculating net self-employment income. The same line also states that expenses for internet service may be deducted as a telecommunication services expense only if necessary for the production of income. Under IRS guidelines adopted in 2011, expenses for home internet service used for other purposes are included in the Local Standards for Housing and utilities—
Insurance and operating expenses. Also, Official Forms 22A-2 and 22C-2 now provide, at line 18, for deductions of the premiums paid by one jointly filing debtor on term life insurance policies of the other joint debtor as well for premium payments on the debtor's own policies.

Official Form 22B now includes, at lines 12-14, space for a married debtor whose chapter 11 case is not a joint case to claim an adjustment to current monthly income for any of the spouse's income that is not contributed to pay for household expenses of the debtor or the debtor's dependents. Because § 1129(a)(15)(B) requires that disposable income be calculated as provided by § 1325(b)(2), the same marital adjustment that is included in Official Form 22C-1 has been added to Official Form 22B.

April 22-23, 2014 Page 57 of 596

April 22-23, 2014 Page 58 of 596

Appendix A Forms for Final Approval Effective Date 2015

April 22-23, 2014 Page 59 of 596

April 22-23, 2014 Page 60 of 596

Fill in this information to identify your case:				
United States Bankruptcy Court for the:				
District of (State)				
Case number (If known):	Chapter you are filing under: Chapter 7 Chapter 11 Chapter 12 Chapter 13			

☐ Check if this is an amended filing

Official Form 101

Voluntary Petition for Individuals Filing for Bankruptcy

12/15

The bankruptcy forms use you and Debtor 1 to refer to a debtor filing alone. A married couple may file a bankruptcy case together—called a joint case—and in joint cases, these forms use you to ask for information from both debtors. For example, if a form asks, "Do you own a car," the answer would be yes if either debtor owns a car. When information is needed about the spouses separately, the form uses Debtor 1 and Debtor 2 to distinguish between them. In joint cases, one of the spouses must report information as Debtor 1 and the other as Debtor 2. The same person must be Debtor 1 in all of the forms.

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

Part 1:	Identify	Yourself

	About Debtor 1:	About Debtor 2 (Spouse Only in a Joint Case):
1. Your full name		
Write the name that is on your		
government-issued picture identification (for example, your driver's license or	First name	First name
passport).	Middle name	Middle name
Bring your picture identification to your meeting with the trustee.	Last name	Last name
	Suffix (Sr., Jr., II, III)	Suffix (Sr., Jr., II, III)
2. All other names you		
have used in the last 8 years	First name	First name
Include your married or maiden names.	Middle name	Middle name
	Last name	Last name
	First name	First name
	Middle name	Middle name
	Last name	Last name
3. Only the last 4 digits of	xxx - xx	xxx - xx
your Social Security number or federal	OR	OR
Individual Taxpayer Identification number	9 xx - xx	9 xx - xx
(ITIN)		

		About Debtor 1:	About Debtor 2 (Spouse Only in a Joint Case):
4.	Any business names and Employer Identification Numbers (EIN) you have used in	☐ I have not used any business names or EINs.	☐ I have not used any business names or EINs.
	the last 8 years	Business name	Business name
	Include trade names and doing business as names	Business name	Business name
		EIN	EIN
		EIN	EIN — — — — — — — —
5.	Where you live		If Debtor 2 lives at a different address:
		Number Street	Number Street
		City State ZIP Code	City State ZIP Code
		County If your mailing address is different from the one above, fill it in here. Note that the court will send any notices to you at this mailing address.	County If Debtor 2's mailing address is different from yours, fill it in here. Note that the court will send any notices to this mailing address.
		Number Street	Number Street
		P.O. Box	P.O. Box
		City State ZIP Code	City State ZIP Code
6.	Why you are choosing this district to file for bankruptcy	Check one: Over the last 180 days before filing this petition, I have lived in this district longer than in any other district. I have another reason. Explain. (See 28 U.S.C. § 1408.)	Check one: ☐ Over the last 180 days before filing this petition, I have lived in this district longer than in any other district. ☐ I have another reason. Explain. (See 28 U.S.C. § 1408.)

Part 2:

Tell the Court About Your Bankruptcy Case

7.	The chapter of the Bankruptcy Code you are choosing to file under		ruptcy (Forroter 7 oter 11 oter 12	rief description of eac m B2010)). Also, go to			U.S.C. § 342(b) for Individuals Filing the appropriate box.
8.	How you will pay the fee	local yours subn with I nee Appl I req By la less pay t	court for itself, you in nitting you a pre-print of to pay it ication for uest that w, a judge than 150% he fee in itself.	more details about nay pay with cash, ir payment on your ted address. the fee in installm individuals to Paymy fee be waived a may, but is not read the of the official pov	how you m cashier's cl behalf, you nents. If you Your Filing I (You may equired to, werty line that a choose th	ay pay. Typicall heck, or money in attorney may put choose this operate in Installment request this optivative your fee, at applies to you is option, you m	eck with the clerk's office in your y, if you are paying the fee order. If your attorney is pay with a credit card or check tion, sign and attach the ents (Official Form 103A). Ion only if you are filing for Chapter 7. and may do so only if your income is r family size and you are unable to ust fill out the Application to Have the with your petition.
9.	Have you filed for bankruptcy within the last 8 years?	□ No □ Yes.	District		When	MM / DD / YYYY MM / DD / YYYY MM / DD / YYYY	Case number Case number Case number
10.	Are any bankruptcy cases pending or being filed by a spouse who is not filing this case with you, or by a business partner, or by an affiliate?	☐ No☐ Yes.	District				Relationship to you Case number, if known Relationship to you Case number, if known
11.	Do you rent your residence?	☐ No. ☐ Yes.	residence	andlord obtained an ?? o to line 12.			and do you want to stay in your *Against You (Form 101A) and file it with

Part 3: Report About Any Businesses You Own as a Sole Proprietor

business?

A sole proprietorship is a business you operate as an individual, and is not a separate legal entity such as

12. Are you a sole proprietor

of any full- or part-time

separate legal entity such as a corporation, partnership, or LLC.

If you have more than one

If you have more than one sole proprietorship, use a separate sheet and attach it to this petition.

☐ No. Go to Part 4.	
_	

Yes. Name and location of business

Name of business, if any

Number Street

City State ZIP Code

Check the appropriate box to describe your business:

- ☐ Health Care Business (as defined in 11 U.S.C. § 101(27A))
- ☐ Single Asset Real Estate (as defined in 11 U.S.C. § 101(51B))
- ☐ Stockbroker (as defined in 11 U.S.C. § 101(53A))
- ☐ Commodity Broker (as defined in 11 U.S.C. § 101(6))
- None of the above

13. Are you filing under Chapter 11 of the Bankruptcy Code and are you a small business debtor?

For a definition of *small* business debtor, see 11 U.S.C. § 101(51D).

If you are filing under Chapter 11, the court must know whether you are a small business debtor so that it can set appropriate deadlines. If you indicate that you are a small business debtor, you must attach your most recent balance sheet, statement of operations, cash-flow statement, and federal income tax return or if any of these documents do not exist, follow the procedure in 11 U.S.C. 1116(1)(B).

- ☐ No. I am not filing under Chapter 11.
- No. I am filing under Chapter 11, but I am NOT a small business debtor according to the definition in the Bankruptcy Code.
- ☐ Yes. I am filing under Chapter 11 and I am a small business debtor according to the definition in the Bankruptcy Code.

Part 4:

Report if You Own or Have Any Hazardous Property or Any Property That Needs Immediate Attention

14. Do you own or have any property that poses or is alleged to pose a threat of imminent and identifiable hazard to public health or safety? Or do you own any property that needs immediate attention?

> For example, do you own perishable goods, or livestock that must be fed, or a building that needs urgent repairs?

☐ No

☐ Yes. What is the hazard?

If immediate attention is needed, why is it needed?

Where is the property? Number Street

City State ZIP Code

Official Form 101

Part 5:

Explain Your Efforts to Receive a Briefing About Credit Counseling

15. Tell the court whether you have received briefing about credit counseling.

The law requires that you receive a briefing about credit counseling before you file for bankruptcy. You must truthfully check one of the following choices. If you cannot do so, you are not eligible to file.

If you file anyway, the court can dismiss your case, you will lose whatever filing fee you paid, and your creditors can begin collection activities again.

About Debtor 1	Abou	tυ	ep	tor		
----------------	------	----	----	-----	--	--

You must check one:

☐ I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, and I received a certificate of completion.

Attach a copy of the certificate and the payment plan, if any, that you developed with the agency.

□ I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, but I do not have a certificate of completion.

Within 14 days after you file this bankruptcy petition, you MUST file a copy of the certificate and payment plan, if any.

□ I certify that I asked for credit counseling services from an approved agency, but was unable to obtain those services during the 7 days after I made my request, and exigent circumstances merit a 30-day temporary waiver of the requirement.

To ask for a 30-day temporary waiver of the requirement, attach a separate sheet explaining what efforts you made to obtain the briefing, why you were unable to obtain it before you filed for bankruptcy, and what exigent circumstances required you to file this case.

Your case may be dismissed if the court is dissatisfied with your reasons for not receiving a briefing before you filed for bankruptcy.

If the court is satisfied with your reasons, you must still receive a briefing within 30 days after you file. You must file a certificate from the approved agency, along with a copy of the payment plan you developed, if any. If you do not do so, your case may be dismissed.

Any extension of the 30-day deadline is granted only for cause and is limited to a maximum of 15 days.

I am not required to receive a briefing ab	out
credit counseling because of:	

☐ Incapacity. I have a mental illness or a mental deficiency that makes me

incapable of realizing or making rational decisions about finances.

☐ **Disability.** My physical disability causes me to be unable to participate in a

briefing in person, by phone, or through the internet, even after I reasonably tried to do so.

Active duty. I am currently on active military duty in a military combat zone.

If you believe you are not required to receive a briefing about credit counseling, you must file a motion for waiver of credit counseling with the court.

About Debtor 2 (Spouse Only in a Joint Case):

You must check one:

 ☐ I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, and I received a certificate of completion.

Attach a copy of the certificate and the payment plan, if any, that you developed with the agency.

☐ I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, but I do not have a certificate of completion.

Within 14 days after you file this bankruptcy petition, you MUST file a copy of the certificate and payment plan, if any.

□ I certify that I asked for credit counseling services from an approved agency, but was unable to obtain those services during the 7 days after I made my request, and exigent circumstances merit a 30-day temporary waiver of the requirement.

To ask for a 30-day temporary waiver of the requirement, attach a separate sheet explaining what efforts you made to obtain the briefing, why you were unable to obtain it before you filed for bankruptcy, and what exigent circumstances required you to file this case.

Your case may be dismissed if the court is dissatisfied with your reasons for not receiving a briefing before you filed for bankruptcy.

If the court is satisfied with your reasons, you must still receive a briefing within 30 days after you file. You must file a certificate from the approved agency, along with a copy of the payment plan you developed, if any. If you do not do so, your case may be dismissed.

Any extension of the 30-day deadline is granted only for cause and is limited to a maximum of 15 days.

I am not required to	receive a	a briefing	about
credit counseling b	ecause o	f:	

☐ Incapacity. I have a mental illness or a mental

deficiency that makes me incapable of realizing or making rational decisions about finances.

Disability. My physical disability causes me to be unable to participate in a

briefing in person, by phone, or through the internet, even after I reasonably tried to do so.

Active duty. I am currently on active military duty in a military combat zone.

If you believe you are not required to receive a briefing about credit counseling, you must file a motion for waiver of credit counseling with the court.

April 22-23, 2014

Part 6: Answer These Ques	stions for Reporting Purpos	es			
16. What kind of debts do you have?	as "incurred by an individual No. Go to line 16b. Yes. Go to line 17. 16b. Are your debts primar money for a business or in No. Go to line 16c. Yes. Go to line 17.	al primarily for a personal, family, or	debts are debts that you incurred to obtain of the business or investment.		
17. Are you filing under Chapter 7? Do you estimate that after any exempt property is excluded and administrative expenses are paid that funds will be available for distribution to unsecured creditors?	□ No. I am not filing under Ch □ Yes. I am filing under Chapte administrative expense □ No □ Yes	er 7. Do you estimate that after any	exempt property is excluded and ble to distribute to unsecured creditors?		
18. How many creditors do you estimate that you owe?	☐ 1-49 ☐ 50-99 ☐ 100-199 ☐ 200-999	☐ 1,000-5,000 ☐ 5,001-10,000 ☐ 10,001-25,000	25,001-50,000 50,001-100,000 More than 100,000		
19. How much do you estimate your assets to be worth?	\$0-\$50,000 \$50,001-\$100,000 \$100,001-\$500,000 \$500,001-\$1 million	\$1,000,001-\$10 million \$10,000,001-\$50 million \$50,000,001-\$100 million \$100,000,001-\$500 million	\$500,000,001-\$1 billion \$1,000,000,001-\$10 billion \$10,000,000,001-\$50 billion More than \$50 billion		
20. How much do you estimate your liabilities to be? Part 7: Sign Below	\$0-\$50,000 \$50,001-\$100,000 \$100,001-\$500,000 \$500,001-\$1 million	\$1,000,001-\$10 million \$10,000,001-\$50 million \$50,000,001-\$100 million \$100,000,001-\$500 million	\$500,000,001-\$1 billion \$1,000,000,001-\$10 billion \$10,000,000,001-\$50 billion More than \$50 billion		
<u> </u>	I have examined this petition, ar	nd I declare under penalty of perjury	y that the information provided is true and		
For you	correct. If I have chosen to file under Chapter 7, I am aware that I may proceed, if eligible, under Chapter 7, 11,12, or 13 of title 11, United States Code. I understand the relief available under each chapter, and I choose to proceed under Chapter 7.				
		d I did not pay or agree to pay some	eone who is not an attorney to help me fill out U.S.C. § 342(b).		
	·	th the chapter of title 11, United Sta			
		ult in fines up to \$250,000, or impris	aining money or property by fraud in connection conment for up to 20 years, or both.		
	*	×			
	Signature of Debtor 1	Sign	nature of Debtor 2		
	Executed on		ecuted on		

Official Form 101

n	^	h	1	r	1

First Name Middle Name Last Name

Case number (if known)_____

For your attorney, if you are represented by one

If you are not represented by an attorney, you do not need to file this page. I, the attorney for the debtor(s) named in this petition, declare that I have informed the debtor(s) about eligibility to proceed under Chapter 7, 11, 12, or 13 of title 11, United States Code, and have explained the relief available under each chapter for which the person is eligible. I also certify that I have delivered to the debtor(s) the notice required by 11 U.S.C. § 342(b) and, in a case in which § 707(b)(4)(D) applies, certify that I have no knowledge after an inquiry that the information in the schedules filed with the petition is incorrect.

	Date		
Signature of Attorney for Debtor		MM /	DD /YYYY
Printed name			
Firm name			
Number Street			
City	State	ZIP Code	
Contact phone	Email address	S	
Bar number	State	_	

For you if you are filing this bankruptcy without an attorney

If you are represented by an attorney, you do not need to file this page.

The law allows you, as an individual, to represent yourself in bankruptcy court, but you should understand that many people find it extremely difficult to represent themselves successfully. Because bankruptcy has long-term financial and legal consequences, you are strongly urged to hire a qualified attorney.

To be successful, you must correctly file and handle your bankruptcy case. The rules are very technical, and a mistake or inaction may affect your rights. For example, your case may be dismissed because you did not file a required document, pay a fee on time, attend a meeting or hearing, or cooperate with the court, case trustee, U.S. trustee, bankruptcy administrator, or audit firm if your case is selected for audit. If that happens, you could lose your right to file another case, or you may lose protections, including the benefit of the automatic stay.

You must list all your property and debts in the schedules that you are required to file with the court. Even if you plan to pay a particular debt outside of your bankruptcy, you must list that debt in your schedules. If you do not list a debt, the debt may not be discharged. If you do not list property or properly claim it as exempt, you may not be able to keep the property. The judge can also deny you a discharge of all your debts if you do something dishonest in your bankruptcy case, such as destroying or hiding property, falsifying records, or lying. Individual bankruptcy cases are randomly audited to determine if debtors have been accurate, truthful, and complete. Bankruptcy fraud is a serious crime; you could be fined and imprisoned.

If you decide to file without an attorney, the court expects you to follow the rules as if you had hired an attorney. The court will not treat you differently because you are filing for yourself. To be successful, you must be familiar with the United States Bankruptcy Code, the Federal Rules of Bankruptcy Procedure, and the local rules of the court in which your case is filed. You must also be familiar with any state exemption laws that apply.

Are you aware that filing for bankruptcy is a serious acconsequences? No Yes	ction with long-term financial and legal
Are you aware that bankruptcy fraud is a serious crim inaccurate or incomplete, you could be fined or impris No Yes	. ,
Did you pay or agree to pay someone who is not an all No Yes. Name of Person Attach Bankruptcy Petition Preparer's Notice, Do	
By signing here, I acknowledge that I understand the have read and understood this notice, and I am aware attorney may cause me to lose my rights or property i	e that filing a bankruptcy case without an
Signature of Debtor 1	Signature of Debtor 2
Date	Date
Contact phone	Contact phone
Cell phone	Cell phone
Email address	Email address

Official Form 101

COMMITTEE NOTE

Official Form 101, *Voluntary Petition for Individuals*Filing for Bankruptcy, applies only in cases of individual debtors.
Form 101 replaces Official Form 1, Voluntary Petition. It is renumbered to distinguish it from the forms used by non-individual debtors, such as corporations, and includes stylistic changes throughout the form. It is revised as part of the Forms
Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. Because the goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats, many of the open-ended questions and multiple-part instructions have been replaced with more specific questions.

Official Form 101 has been substantially reorganized. References to Exhibits A, B, C, and D, and the exhibits themselves, have been eliminated because the requested information is now asked in the form or is not applicable to individual debtors.

Part 1, *Identify Yourself*, line 6, replaces the venue box from page 2 of Official Form 1 and deletes venue questions that pertain only to non-individuals.

Part 2, *Tell the Court About Your Bankruptcy Case*, line 7, removes choices for chapters 9 and 15 filings because they do not pertain to individuals. The status of "being filed" is added to the question regarding bankruptcy cases pending or filed by a spouse, business partner, or affiliate (line 10). Lastly, the question "Do you rent your residence?" (line 11) and Official Forms 101A, *Initial Statement About an Eviction Judgment Against You*, and 101B, *Statement About Payment of An Eviction Judgment Against You*, replace "certification by a debtor who resides as a tenant of residential property," on page 2 of Official Form 1.

Part 3, Report About Any Businesses You Own as a Sole Proprietor, line 12, incorporates options from the "nature of

April 22-23, 2014 Page 69 of 596

business" box from page 1 of Official Form 1 that would apply to individual debtors, thus eliminating checkboxes for railroads and clearing banks. Part 3, line 13, also eliminates a checkbox to report whether a plan was filed with the petition, or if plan acceptances were solicited prepetition. Additionally, line 13 rephrases the question relating to whether a debtor filing under Chapter 11 is a small business debtor.

Part 4, Report if You Own or Have Any Hazardous Property or Any Property That Needs Immediate Attention, line 14, replaces Exhibit C from Official Form 1 and adds the category of "property that needs immediate attention."

Part 5, Explain Your Efforts to Receive Credit Counseling (line 15), replaces Exhibit D from Official Form 1. Additionally, this part describes incapacity and disability using a simplified definition, tells the debtor of the ability to file a motion for a waiver, and eliminates statutory reference about districts where credit counseling does not apply because such districts are rare.

Part 6, Answer These Questions for Reporting Purposes (line 16c), provides a text field for the debtor to describe the type of debts owed if the debtor believes they are neither primarily consumer nor business debts.

Part 7, *Sign Here*, deletes from the debtor's declaration the phrase "to the best of my knowledge, information, and belief" in order to conform to the language of 28 U.S.C. § 1746. *See* Rule 1008. This part combines the two attorney signature blocks into one certification and eliminates signature lines for corporations/partnerships and chapter 15 Foreign Representative. The declaration and signature section for a non-attorney bankruptcy petition preparer (BPP) has also been removed as unnecessary. The same declaration, required under 11 U.S.C. § 110, is contained in Official Form 119. That form must be completed and signed by the BPP and filed with each document prepared by a BPP.

April 22-23, 2014 Page 70 of 596

Fill in this information to identify your case:				
Debtor 1	First Name	Middle Name	Last Name	
Debtor 2	Filst Name	widdle Name	Last Name	
(Spouse, if filing)	First Name	Middle Name	Last Name	
United States E	Sankruptcy Court for the:		District of(State)	
Case number (If known)				

Draft February 28, 2014

Official Form 101A

Initial Statement About an Eviction Judgment Against You

12/15

File this form with the court and serve a copy on your landlord when you first file bankruptcy only if:

- you rent your residence; and
- your landlord has obtained a judgment for possession in an eviction, unlawful detainer action, or similar proceeding (called eviction judgment) against you to possess your residence.

Landlord's name				
Landlord's address	Number	Street		
	City		State	

If you want to stay in your rented residence after you file your case for bankruptcy, also complete the certification below.

I certify under penalty of	neriury that:	
_ , , ,	. , ,	
		ies to the judgment for possession (eviction judgment), landlord the entire delinquent amount.
the Voluntary Petitic	kruptcy court clerk a deposit for t n for Individuals Filing for Bankru	
Signature of Deb	or 1	Signature of Debtor 2
Date		Date
Stay of Eviction: (a)	and served your landlord with a	y. If you checked both boxes above, signed the form to certify that both apply, copy of this statement, the automatic stay under 11 U.S.C. § 362(a)(3) will eviction against you for 30 days after you file your Voluntary Petition for (Official Form 101).
(b)	receive the protection of the aut amount to your landlord as state	If you wish to stay in your residence after that 30-day period and continue to tomatic stay under 11 U.S.C. § 362(a)(3), you must pay the entire delinquent ed in the eviction judgment before the 30-day period ends. You must also fill of an Eviction Judgment Against You (Official Form 101B), file it with the

Check the Bankruptcy Rules (www.uscourts.gov/rulesandpolicies/rules.aspx) and the local court's website (to find your court's website, go to www.uscourts.gov/Court_Locator.aspx) for any specific requirements that you might have to meet to serve this statement.

11 U.S.C. §§ 362(b)(22) and 362(l)

Official Form 101A

April 22-23, 2014 Page 71 of 596

April 22-23, 2014 Page 72 of 596

Fill in this information to identify your case:				
Debtor 1	First Name	Middle Name	Last Name	
Debtor 2				
(Spouse, if filing)	First Name	Middle Name	Last Name	
United States E	Bankruptcy Court for the	e:		
Case number		(State)		

Draft February 28, 2014

Official Form 101B

Statement About Payment of an Eviction Judgment Against You

12/15

Fill out this form only if:

- you filed Initial Statement About an Eviction Judgment Against You (Official Form 101A); and
- you served a copy of Form 101A on your landlord; and
- you want to stay in your rented residence for more than 30 days after you file your Voluntary Petition for Individuals Filing for Bankruptcy (Official Form 101).

File this form within 30 days after you file your *Voluntary Petition for Individuals Filing for Bankruptcy* (Official Form 101). Also serve a copy on your landlord within that same time period.

Certification About Applicable Law and Paymo	ent of Eviction Suugment
certify under penalty of perjury that (Check all that apply)):
Under the state or other nonbankruptcy law that applies <i>judgment</i>), I have the right to stay in my residence by page 1.	
Within 30 days after I filed my <i>Voluntary Petition for Indi</i> Form 101), I have paid my landlord the entire amount I of (eviction judgment).	
~	✓
Signature of Debtor 1	Signature of Debtor 2
Signature of Debtor 1 Date MM / DD / YYYY	Signature of Debtor 2 Date
Date	Date

You must serve your landlord with a copy of this form.

Check the Bankruptcy Rules (www.uscourts.gov/rulesandpolicies/rules.aspx) and the court's local website (go to http://www.uscourts.gov/Court_Locator.aspx to find your court's website) for any specific requirements that you might have to meet to serve this statement.

Official Form 101B

Statement About Payment of an Eviction Judgment Against You

April 22-23, 2014 Page 73 of 596

April 22-23, 2014 Page 74 of 596

COMMITTEE NOTE

Official Form 101A, Initial Statement About an Eviction Judgment Against You, and Official Form 101B, Statement About Payment of an Eviction Judgment Against You, are new forms promulgated as part of the Forms Modernization Project. They replace the "Certification by a Debtor Who Resides as a Tenant of Residential Property" section on Official Form 1, Voluntary Petition. The forms apply only in cases of individual debtors.

Official Form 101A explains that debtors need to complete and file the form only if their landlord has a judgment for possession or an eviction judgment against them and they rent their residence. The form further explains that if the debtor wishes to stay in their residence for 30 days after filing their bankruptcy petition, the certification must be completed. The form adds references to the provisions in the Bankruptcy Code that specify when debtor-tenants subject to eviction may remain in their residence after filing for bankruptcy.

The form eliminates the checkboxes that the debtor has served the landlord with the certification and paid the court the rent that would be due during the 30 days after the filing of the bankruptcy petition. Instead, debtors are required to certify under penalty of perjury that the rent has been paid to the court, and the instructions direct debtors to serve a copy of the statement on the landlord.

The form eliminates the checkbox that the debtor claims there are circumstances under applicable nonbankruptcy law under which the debtor would be permitted to cure the monetary default that gave rise to the judgment for possession (or eviction judgment) and remain in residence. Instead, debtors are required to certify under penalty of perjury that they have the right to stay in their residence under state law or other nonbankruptcy law by paying their landlord the entire delinquent amount.

Official Form 101B is new. If debtors wish to stay in their residence for more than 30 days after filing the petition, they must complete, file, and serve the form within 30 days after the petition is filed. Under Official Form 101B, debtors certify under penalty of perjury that

April 22-23, 2014 Page 75 of 596

they have the right to stay in their residence under state law or other nonbankruptcy law by paying their landlord the entire delinquent amount and that they have paid their landlord the entire amount owed as stated in the judgment for possession or in the eviction judgment.

April 22-23, 2014 Page 76 of 596

Draft May 7, 2013

Fill in this information to identify your case:					
Debtor 1					
	First Name	Middle Name	Last Name		
Debtor 2					
(Spouse, if filing)	First Name	Middle Name	Last Name		
United States Bankruptcy Court for the:			District of(State)		
Case number (If known)					

☐ Check if this is an amended filing

Official Form 104

For Individual Chapter 11 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims Against You and Are Not Insiders

If you are an individual filing for bankruptcy under Chapter 11, you must fill out this form. If you are filing under Chapter 7, Chapter 12, or Chapter 13, do not fill out this form. Do not include claims by anyone who is an *insider*. Insiders include your relatives; any general partners; relatives of any general partners; partnerships of which you are a general partner; corporations of which you are an officer, director, person in control, or owner of 20 percent or more of their voting securities; and any managing agent, including one for a business you operate as a sole proprietor. 11 U.S.C. § 101. Also, do not include claims by secured creditors unless the unsecured claim resulting from inadequate collateral value places the creditor among the holders of the 20 largest unsecured claims.

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information.

Part 1:

List the 20 Unsecured Claims in Order from Largest to Smallest. Do Not Include Claims by Insiders.

				Unsecured claim
		What is the nature of the claim?		\$
		As of the date you file, the claim is: Check	call that apply.	
		☐ Contingent		
		☐ Unliquidated		
		☐ Disputed		
		☐ None of the above apply		
State	ZIP Code	☐ No		
		Yes. Total claim (secured and unsecured):	\$	_
		Value of security:	· \$	-
		Unsecured claim	\$	-
		What is the nature of the claim?		\$
		As of the date you file, the claim is: Check	call that apply.	
		☐ Contingent		
		☐ Unliquidated		
		☐ Disputed		
		☐ None of the above apply		
State	ZIP Code	Does the creditor have a lien on your pro ☐ No	perty?	
		Yes. Total claim (secured and unsecured):	\$	-
		Value of security:	\$	_
			As of the date you file, the claim is: Check Contingent Unliquidated Disputed None of the above apply State	Unliquidated Disputed None of the above apply Does the creditor have a lien on your property? No Yes. Total claim (secured and unsecured): Value of security: Unsecured claim What is the nature of the claim? As of the date you file, the claim is: Check all that apply. Contingent Unliquidated Disputed None of the above apply State ZIP Code Does the creditor have a lien on your property? No Yes. Total claim (secured and unsecured): \$

Debtor 1				Case number (if known)
	First Name	Middle Name	Last Name	

				Unsecured claim
3			What is the nature of the claim?	
ŭ	Creditor's Name		What is the nature of the claim:	 \$
			As of the date you file, the claim is: Check all that apply.	
	Number Street		☐ Contingent	
			☐ Unliquidated	
			☐ Disputed	
			☐ None of the above apply	
	City Si	ate ZIP Code	Does the creditor have a lien on your property?	
			□ No	
	Contact		☐ Yes. Total claim (secured and unsecured): \$	
			Value of security:	
	Contact phone		Unsecured claim \$	
			Ψ	
4			What is the nature of the claim?	
	Creditor's Name		As of the date you file, the claim is: Check all that apply.	
			☐ Contingent	
	Number Street		☐ Unliquidated	
			☐ Disputed	
			☐ None of the above apply	
	Cit.	ate ZIP Code	Does the creditor have a lien on your property?	
	City	ate ZIP Code	□ No	
			☐ Yes. Total claim (secured and unsecured): \$	
	Contact		Value of security:	
			Unsecured claim \$	
	Contact phone		Onsecured ciaim	
5				\$
Y	Creditor's Name		What is the nature of the claim?	
			As of the date you file, the claim is: Check all that apply.	
	Number Street		Contingent	
			Unliquidated	
			Disputed	
			☐ None of the above apply	
	City Si	ate ZIP Code	Does the creditor have a lien on your property?	
			□ No	
	Contact		☐ Yes. Total claim (secured and unsecured): \$	
			Value of security:	
	Contact phone		Unsecured claim \$	
6				\$
J	Creditor's Name		What is the nature of the claim?	Φ
			As of the date you file, the claim is: Check all that apply.	
	Number Street		Contingent	
			☐ Unliquidated	
			Disputed	
			☐ None of the above apply	
	City Si	ate ZIP Code	Does the creditor have a lien on your property?	
			No	
	Contact		☐ Yes. Total claim (secured and unsecured): \$	
			Value of security:	
	Contact phone		Unsecured claim \$	
7			What is the nature of the claim?	\$
	Creditor's Name		As of the date you file, the claim is: Check all that apply.	
			Contingent	
	Number Street		☐ Unliquidated	
			☐ Disputed	
			☐ None of the above apply	
	City	ate ZIP Code	Does the creditor have a lien on your property?	
			No	
	Contact		☐ Yes. Total claim (secured and unsecured): \$	
			Value of security:	
	Contact phone		Unsecured claim \$	

Debtor 1				Case number (if known)
	First Name	Middle Name	Last Name	

							Unsecured claim
8					What is the nature of the claim?		\$
	Creditor's Name				As of the date you file, the claim is: Check		·
	Number	Street			☐ Contingent		
	rambo.	Ciloti			Unliquidated		
					☐ Disputed		
					None of the above apply		
	City		State	ZIP Code	Does the creditor have a lien on your prop	erty?	
					☐ Yes. Total claim (secured and unsecured):	\$	
	Contact				Value of security:	\$	
	Contact phone				Unsecured claim	\$	•
9					What is the nature of the claim?		\$
	Creditor's Name				As of the date you file, the claim is: Check		·
	Number	Street			☐ Contingent		
	Number	Officer			☐ Unliquidated		
					☐ Disputed		
					None of the above apply		
	City		State	ZIP Code	Does the creditor have a lien on your prop	erty?	
					=	\$	
	Contact				Value of security:	\$	
	Contact phone				Unsecured claim	\$	_
10	Comact prioric						•
IV	Creditor's Name				What is the nature of the claim?		\$
	Orealtor 3 Harrie				As of the date you file, the claim is: Check a	all that apply.	
	Number	Street			☐ Contingent☐ Unliquidated		
					☐ Disputed		
					☐ None of the above apply		
	City		State	ZIP Code	Does the creditor have a lien on your prop	erty?	
					□ No		
	Contact				Yes. Total claim (secured and unsecured):		-
					Value of security:	\$	
	Contact phone		_		Unsecured claim	\$	
11					What is the nature of the claim?		\$
	Creditor's Name				As of the date you file, the claim is: Check		Y
	Number	Street			Contingent	ан шасарріу.	
	Number	Street			☐ Unliquidated		
					Disputed		
					☐ None of the above apply		
	City		State	ZIP Code	Does the creditor have a lien on your prop	erty?	
					Yes. Total claim (secured and unsecured):	\$	
	Contact				Value of security:	\$	
	Contact phone				Unsecured claim	\$	
12					What is the nature of the claim?		\$
	Creditor's Name				As of the date you file, the claim is: Check		
	Orealtor 3 Harrie				☐ Contingent		
	Number	Street			☐ Unliquidated		
					Disputed		
					None of the above apply		
	City		State	ZIP Code	Does the creditor have a lien on your prop	ету?	
					☐ Yes. Total claim (secured and unsecured):	\$	
	Contact				Value of security:	\$	•
					Unsecured claim	\$	
	Contact phone						

ebtor 1				Case number (if known)
	First Name	Middle Name	Last Name	

						Unsecured claim
13				Wilestin the material of the Co		\$
-	Creditor's Name			What is the nature of the claim?		Ψ
				As of the date you file, the claim is: Check a	all that apply.	
	Number Street			☐ Contingent ☐ Unliquidated		
				☐ Disputed		
				☐ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your prope	arty?	
	Oity	Otate	ZII GOGC	No	orty:	
					\$	
	Contact			Value of security:	\$	•
	0			Unsecured claim	\$	
	Contact phone			Chicocarca diann	Ψ	
14				What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Check a		
	N			☐ Contingent	tat app.y.	
	Number Street			☐ Unliquidated		
				Disputed		
				☐ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your prope	erty?	
				□ No	•	
					\$	
	Contact			Value of security:	\$	-
	Contact phone	_		•	\$	•
	Contact phone			Onscoured dam	Ψ	
15				What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Check a	all that apply.	
				☐ Contingent	11.7	
	Number Street			☐ Unliquidated		
				☐ Disputed		
				☐ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your prope	erty?	
	•			☐ No		
				☐ Yes. Total claim (secured and unsecured):	\$	-
	Contact			Value of security:	\$	
	Contact phone			Unsecured claim	\$	-
	Contact prioric					
16				What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Check a	all that apply.	
	Number Street			☐ Contingent	* * *	
				☐ Unliquidated		
				☐ Disputed		
				☐ None of the above apply		
	City	State	ZIP Code	Does the creditor have a lien on your proper	erty?	
				□ No		
	Contact			☐ Yes. Total claim (secured and unsecured):	\$	
				Value of security:	\$	
	Contact phone	_		Unsecured claim	\$	
17				What is the nature of the claim?		\$
	Creditor's Name			As of the date you file, the claim is: Check a		
				Contingent	ш шасарру.	
	Number Street			☐ Unliquidated		
				Disputed		
				☐ None of the above apply		
				Does the creditor have a lien on your prope	erty?	
	City	State	ZIP Code	□ No	•	
				Yes. Total claim (secured and unsecured):	\$	_
	Contact				\$	
				Value of security:		-
	Contact phone			Unsecured claim	\$	-

1	First Name	Middle Name	Last Na		e number (if known)	
						Unsecured claim
				What is the nature of the claim?		\$
Creditor's	Name			As of the date you file, the claim is: Check	all that apply.	
Number	Street			☐ Contingent☐ Unliquidated		
				☐ Disputed		
				☐ None of the above apply		
City		State	ZIP Code	Does the creditor have a lien on your prop	perty?	
				□ No		
Contact				Yes. Total claim (secured and unsecured):	\$	
				Value of security:	\$	
Contact pl	hone			Onsecured daim	\$	
				What is the nature of the claim?		
				What is the nature of the claim?		 \$
Creditor's	Name			As of the date you file, the claim is: Check	all that apply.	·
Number	Street			☐ Contingent	1 1 .5.	
				Unliquidated		
				☐ Disputed ☐ None of the above apply		
City		State	ZIP Code			
O.I.y		Ciaio	2.1 0000	Does the creditor have a lien on your prop No	perty?	
Contact				Yes. Total claim (secured and unsecured):	\$	
				Value of security:	\$	
Contact pl	hone			Unsecured claim	\$	
				What is the nature of the claim?		
						\$
				As of the date you file, the claim is: Check Contingent	all that apply.	
Creditor's	Name			☐ Unliquidated		
				☐ Disputed		
Number	Street			☐ None of the above apply		
				Does the creditor have a lien on your prop	nerty?	
				No		
City		State	ZIP Code	☐ Yes. Total claim (secured and unsecured):	\$	
				Value of security:	\$	
Contact				Unsecured claim	\$	
Contact pl	hone					
Contact p						
t 2:	Sign Belov	A/				
l 2:	Sign Belov					
Indo	onalty of no	riury I deales	that the in	formation provided in this form is true on	d correct	
muer p	enaity of pe	ı jui y, i deciare	a uiat the In	formation provided in this form is true an	u correct.	
				4.5		
				×		
Signatu	re of Debtor 1			Signature of Debtor 2		
natu	re of Debtor 1			Signature of Debtor 2		

Date MM / DD / YYYY

Date MM / DD / YYYY

April 22-23, 2014 Page 82 of 596

COMMITTEE NOTE

Official Form 104, For Individual Chapter 11 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims Against You and Are Not Insiders, is revised as part of the Forms Modernization Project. It replaces Official Form 4, List of Creditors Holding 20 Largest Unsecured Claims in chapter 11 cases filed by individuals or joint debtors. The form is renumbered to distinguish it from the version to be used in chapter 11 cases filed by non-individuals, such as corporations and partnerships, and in chapter 9 cases.

Form 104 is reformatted to make it easier to complete and understand and to be more visually appealing. Blanks and checkboxes are provided for specific information about each claim, replacing columns for listing information. A separate, numbered section is provided for each of the 20 claims.

The instruction not to include fully secured claims is restated in less technical terms. Debtors are instructed to include a secured creditor only if the creditor has an unsecured claim resulting from inadequate collateral value that is among the 20 largest unsecured claims. Blanks are provided to calculate the value of the unsecured portion of a partially secured claim.

Examples of "insiders" are provided in addition to the statutory reference. The form adds an explicit instruction not to file the form in a chapter 7, chapter 12, or chapter 13 case. An instruction to be as complete and accurate as possible is added, along with a warning that, if two married people are filing jointly, both are equally responsible for supplying correct information.

With respect to children who may be creditors, the direction to state only the initials of a minor child and the name and address of the child's parent or guardian, rather than the child's full name, is moved to the general instruction booklet for the forms because it applies to all of the forms.

April 22-23, 2014 Page 83 of 596

April 22-23, 2014 Page 84 of 596

Fill in this information to identify the case:				
United States Bankruptcy Court for the:				
District of(State)				
Case number (If known):	Chapter			

☐ Check if this is an amended filing

Official Form 105

Involuntary Petition Against an Individual

12/15

Use this form to begin a bankruptcy case against an individual you allege to be a debtor subject to an involuntary case. If you want to begin a case against a non-individual, use the *Involuntary Petition Against a Non-individual* (Official Form 205). Be as complete and accurate as possible. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write name and case number (if known).

Part 1: Identify the Cha	pter of the Bankruptcy Code Under Which Petition Is Filed
Chapter of the Bankruptcy Code	Check one: Chapter 7 Chapter 11
Part 2: Identify the Debi	
2. Debtor's full name	First name
	Middle name
	Last name
	Suffix (Sr., Jr., II, III)
3. Other names you know the debtor has used in the last 8 years	
Include any assumed, married, maiden, or trade names, or doing business a names.	
Only the last 4 digits of debtor's Social Security Number or federal	ty
Individual Taxpayer Identification Number (ITIN)	xxx - xx OR 9 xx - xx
5. Any Employer Identification Numbers (EINs) used in the last 8	
years	EIN
	EIN

Official Form 105

6.	Debtor's address	Prir	ncipal residence		Mailing addr	ess, if different	from residen	ce
		Num	ber Street		Number S	treet		
		City	State ZIP	Code	City		State	ZIP Code
		Cou	nty					
		Prir	ncipal place of business					
		Num	ber Street					
		City	State ZIP	Code				
		Cou	nty					
		_						
7.	Type of business	Ц	Debtor does not operate a business					
		Che	eck one if the debtor operates a business:					
			Health Care Business (as defined in 11 U.S	S.C. § 101	(27A))			
			Single Asset Real Estate (as defined in 11	U.S.C. §	101(51B))			
			Stockbroker (as defined in 11 U.S.C. § 101					
			Commodity Broker (as defined in 11 U.S.C.	. § 101(6))			
		Ц	None of the above					
8.	Type of debt	Eac	ch petitioner believes:					
			Debts are primarily consumer debts	S. Consun	ner debts are d	defined in 11 U.S	S.C. § 101(8) a	as.
			"incurred by an individual primarily for a per				3 (0)	-
			Debts are primarily business debts.					money
			for a business or investment or through the	operation	n of the busine	ss or investmen	t.	
9.	Do you know of any		No					
•	bankruptcy cases	_				5.00		
	pending by or against any partner, spouse, or	Ц	Yes. Debtor					
	affiliate of this debtor?		DistrictDat		1/DD / YYYY	Case number, if k	nown	
			Debtor			Relationship		
			District Date	e filed		Case number, if k	nown	
					I / DD / YYYY	,		

Part 3: Report About the	Case				
10. Venue	Check one:				
Reason for filing in this court.	Over the last 180 days before the filing of this bankruptcy, the debtor has resided, had the p business, or had principal assets in this district longer than in any other district.	rincipal place of			
	☐ A bankruptcy case concerning debtor's affiliates, general partner, or partnership is pending i	n this district.			
	Other reason. Explain. (See 28 U.S.C. § 1408.)				
11. Allegations	Each petitioner is eligible to file this petition under 11 U.S.C. § 303(b).				
	The debtor may be the subject of an involuntary case under 11 U.S.C. § 303(a).				
	At least one box must be checked:				
	☐ The debtor is generally not paying such debts as they become due, unless they are the subj dispute as to liability or amount.	ect of a bona fide			
	☐ Within 120 days before the filing of this petition, a custodian, other than a trustee, receiver, of authorized to take charge of less than substantially all of the property of the debtor for the pulien against such property, was appointed or took possession.	or agent appointed or urpose of enforcing a			
12. Has there been a transfer of any claim against the debtor by or to any petitioner?	☐ No ☐ Yes. Attach all documents that evidence the transfer and any statements required under Bar 1003(a).	nkruptcy Rule			
13. Each petitioner's claim	Name of petitioner Nature of petitioner's claim	Amount of the claim above the value of any lien			
		\$			
		\$ 			
		\$ 			
	Total	\$			
	If more than 3 petitioners, attach additional sheets with the statement under penalty of perjury, each petitioner's (or representative's) signature under the statement, along with the signature of the petitioner's attorney, and the information on the petitioning creditor, the petitioner's claim, the petitioner's representative, and the attorney following the format on this form.				

Part 4:

Request for Relief

Petitioners request that an order for relief be entered against the debtor under the chapter specified in Part 1 of this petition. If a petitioning creditor is a corporation, attach the corporate ownership statement required by Bankruptcy Rule 1010(b). If any petitioner is a foreign representative appointed in a foreign proceeding, a certified copy of the order of the court granting recognition is attached.

Petitioners declare under penalty of perjury that the information provided in this petition is true and correct to the best of their knowledge, information, and belief. Petitioners understand that if they make a false statement, they could be fined up to \$250,000 or imprisoned for up to 5 years, or both.

18 U.S.C. §§ 152 and 3571. If relief is not ordered, the court may award attorneys' fees, costs, damages, and punitive damages. 11 U.S.C. § 303(i).

Petitioners or Petitioners' Representative	Attorneys
×	×
Signature of petitioner or representative, including representative's title	Signature of attorney
Printed name of petitioner	Printed name
Date signed MM / DD / YYYY	Firm name, if any
Mailing address of petitioner	Number Street
Number Street	City State ZIP Code
City State ZIP Code	Date signed MM / DD / YYYY
	Contact phone Email
If petitioner is an individual and is not represented by an attorney:	
Contact phone	
Email	
Name and mailing address of petitioner's representative, if any	
Name	
Number Street	
City State ZIP Code	

Signature of petitioner	or representative, including repres	entative's title
Printed name of petitio	ner	
Date signed MM	/ DD /YYYY	
Mailing address of	petitioner	
Number Street		
City	State	ZIP Co
Name and mailing	address of petitioner's repre	esentative, if an
Name		
Name Number Street		

Signature of Attorney		
Printed name		
Firm name, if any		
Number Street		
City	State	ZIP Code
Date signed MM / DD / YYYY		
Contact phone	Email	

Signature of	petitioner or representa	tive, including repres	entative's title
Printed nam	e of petitioner		
Date signed	MM / DD /YY	/Y	
Mailing ad	dress of petitioner		
Number S	Street		
City		State	ZIP Code
Name and	mailing address of	petitioner's repre	esentative, if any
Number S	itreet		
City	State ZIP Code		

Signature of A	ttorney		
Printed name			
Firm name, if	any		
Number Str	eet		
City		State	ZIP Code
Date signed	MM / DD / YYYY		

April 22-23, 2014 Page 90 of 596

COMMITTEE NOTE

Official Form 105, *Involuntary Petition Against an Individual*, which is used only in cases of individual debtors, is revised in its entirety as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. In addition, the form is renumbered to distinguish it from the version to be used in non-individual cases, and stylistic changes were made throughout the form.

The form is derived from Official Form 5, *Involuntary Petition*. The new form separates questions into four parts likely to be more familiar to non-lawyers, groups questions of a similar nature together, and eliminates questions unrelated to individual debtors.

Part 1, *Identify the Chapter of the Bankruptcy Code Under Which Petition is Filed*, moves to the beginning of the form the question regarding the chapter of the Bankruptcy Code under which the petition is filed.

Part 2, *Identify the Debtor*, includes the questions regarding the debtor's name, prior names, Social Security Number, Individual Taxpayer Identification Number and Employer Identification Number. Petitioners must list the address for the debtor's principal residence, mailing address (if different), and principal place of business. Petitioners must indicate whether the debtor operates a business, and, if so, use checkboxes to indicate whether the business falls into certain categories. The statutory definition of "consumer debts" is provided, as well as a definition of "business debts."

Part 3, *Report About the Case*, amends the question regarding venue to advise that venue is "the reason to file in this court" and amends the choices for venue. The first option is revised to read: "Over the last 180 days before the filing of this bankruptcy, the debtor has resided, had the principal place of business, or had principal assets in this district longer than any other district." Also, the form adds an option for "Other reason.

April 22-23, 2014 Page 91 of 596

Explain," with a statutory reference. The filing fee amounts are provided. In the question for Allegations, the exact citation to the Bankruptcy Code is provided for the second allegation, and checkboxes are provided for the last allegation. Petitioners must check "yes" or "no" to answer whether there has been any transfer of any claim against the debtor by or to a petitioner. The information regarding the petitioner's claims is moved to this part of the form, and the portion listing the amount of the claim is amended to ask about the amount of the claim that exceeds the value of the lien, if any.

Part 4, *Request for Relief*, amends the instructions to include a warning about making a false statement, and adds a separate requirement for each petitioner's mailing address. Also, petitioners' attorneys must provide their email addresses, or if a petitioner is an individual and not represented by an attorney, the contact phone and email address of that petitioner must be provided.

April 22-23, 2014 Page 92 of 596

Draft March 14, 2014

Fill in this in	formation to iden	tify your case and this	filing:
Debtor 1			
	First Name	Middle Name	Last Name
Debtor 2			
(Spouse, if filing)	First Name	Middle Name	Last Name
United States E	Bankruptcy Court for	the:	District of
Case number			

☐ Check if this is an amended filing

Official Form 106A/B

Schedule A/B: Property

12/15

In each category, separately list and describe items. List an asset only once. If an asset fits in more than one category, list the asset in the category where you think it fits best. Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

Part 1: Describe Each Residence, Building, Land, or Other Real Estate You Own or Have an Interest In

1 1	/here is the property?	What is the property? Check all that apply. ☐ Single-family home ☐ Duplex or multi-unit building	Do not deduct secured clain amount of any secured clain Creditors Who Have Claims	ns on <i>Schedule D:</i>
Stree	et address, if available, or other description	☐ Condominium or cooperative ☐ Manufactured or mobile home ☐ Land ☐ Investment property	Current value of the entire property?	Current value of to portion you own?
City	State ZIP Code	☐ Timeshare ☐ Other Who is an owner of the property? Check one.	Describe the nature of interest (such as fee si entireties, or a life esta	mple, tenancy by th
Cour	nty	☐ Debtor 1 only ☐ Debtor 2 only ☐ Debtor 1 and Debtor 2 only ☐ At least one of the debtors and another	Check if this is cominstructions)	nmunity property (s
		Other information you wish to add about this		
you own	or have more than one, list here:	property identification number:		
1.2		what is the property? Check all that apply. Single-family home		ns on Schedule D:
2	or have more than one, list here:	what is the property? Check all that apply. Single-family home Duplex or multi-unit building Condominium or cooperative Manufactured or mobile home	Do not deduct secured clain amount of any secured clain	ns on Schedule D:
1.2		what is the property? Check all that apply. Single-family home Duplex or multi-unit building Condominium or cooperative	Do not deduct secured clain amount of any secured clain Creditors Who Have Claims Current value of the	ns on Schedule D: Secured by Property. Current value of portion you own your ownership mple, tenancy by tena

Official Form 106A/B April 22-23, 2014

Debtor 1	First Name Middle Name Last Na	Case numbe	Pr (if known)	
	THO NAME MIGGO NAME CONTROL OF THE PARTY OF			
1.3.		What is the property? Check all that apply. ☐ Single-family home	Do not deduct secured clair amount of any secured clair Creditors Who Hold Claims	ms on Schedule D:
	Street address, if available, or other description	Duplex or multi-unit building	Occurrent control of the	0
		Condominium or cooperative	Current value of the entire property?	Current value of the portion you own?
		 Manufactured or mobile home I and 		p
			\$. Φ
		☐ Investment property ☐ ☐ Timeshare	Describe the nature of	vour ownership
	City State ZIP Coo	le	interest (such as fee s	
		Other	entireties, or a life esta	ate), if known.
		Who is an owner of the property? Check one.		
		Debtor 1 only		
	County	Debtor 2 only		
		Debtor 1 and Debtor 2 only	☐ Check if this is con	nmunity property (see
		At least one of the debtors and another	instructions)	
		Other information you wish to add about this property identification number:		
2 Add t	he dollar value of the portion you own to	r all of your entries from Part 1, including any ent	tries for names	
vou t	have attached for Part 1. Write that number	er here.		\$
•				
	Describe Your Vehicles		an and 2 land and a manabial	
Do you o you own	own, lease, or have legal or equitable inte that someone else drives. If you lease a vel	erest in any vehicles, whether they are registered nicle, also report it on Schedule G: Executory Contractions.	-	es
you own	own, lease, or have legal or equitable interest that someone else drives. If you lease a velous, trucks, tractors, sport utility vehic	nicle, also report it on Schedule G: Executory Contract	-	es
Do you oyou own 3. Cars,	own, lease, or have legal or equitable interest that someone else drives. If you lease a veh vans, trucks, tractors, sport utility vehice	nicle, also report it on Schedule G: Executory Contract	-	es
Do you oyou own	own, lease, or have legal or equitable interest that someone else drives. If you lease a veh vans, trucks, tractors, sport utility vehice	nicle, also report it on Schedule G: Executory Contract	-	es
Do you oyou own 3. Cars, \textsize N \textsize Y	own, lease, or have legal or equitable interthat someone else drives. If you lease a velowans, trucks, tractors, sport utility vehicoes	nicle, also report it on Schedule G: Executory Contractions, motorcycles	cts and Unexpired Leases.	
Do you oyou own 3. Cars,	own, lease, or have legal or equitable interested that someone else drives. If you lease a vehicle vans, trucks, tractors, sport utility vehicle oes Make:	who is an owner of the property? Check one.	cts and Unexpired Leases. Do not deduct secured claim amount of any secured claim	s or exemptions. Put the
Do you oyou own 3. Cars, \textsize N \textsize Y	own, lease, or have legal or equitable interthat someone else drives. If you lease a velowans, trucks, tractors, sport utility vehicoes	who is an owner of the property? Check one.	cts and Unexpired Leases. Do not deduct secured claim	s or exemptions. Put the
Do you oyou own 3. Cars, \textsize N \textsize Y	own, lease, or have legal or equitable interested that someone else drives. If you lease a vehicle vans, trucks, tractors, sport utility vehicle oes Make:	who is an owner of the property? Check one. Debtor 1 only Debtor 2 only	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the	s or exemptions. Put the is on <i>Schedule D:</i> Secured by Property. Current value of th
Do you oyou own 3. Cars, \textsize N \textsize Y	own, lease, or have legal or equitable interest that someone else drives. If you lease a velous vans, trucks, tractors, sport utility vehico es Make: Model:	who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only	Cts and Unexpired Leases. Do not deduct secured claim amount of any secured claims Creditors Who Have Claims	s or exemptions. Put the is on Schedule D: Secured by Property.
Do you oyou own 3. Cars, \textsize N \textsize Y	own, lease, or have legal or equitable interest that someone else drives. If you lease a vertex vans, trucks, tractors, sport utility vehice of es Make: Model: Year:	who is an owner of the property? Check one. Debtor 1 only Debtor 2 only	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the	s or exemptions. Put the is on <i>Schedule D:</i> Secured by Property. Current value of th
Do you oyou own 3. Cars, \textsize N \textsize Y	own, lease, or have legal or equitable interest that someone else drives. If you lease a velous vans, trucks, tractors, sport utility vehico es Make: Model: Year: Approximate mileage:	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only At least one of the debtors and another	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the	s or exemptions. Put the is on <i>Schedule D:</i> Secured by Property. Current value of th
Do you oyou own 3. Cars, \textsize N \textsize Y	own, lease, or have legal or equitable interest that someone else drives. If you lease a velous vans, trucks, tractors, sport utility vehico es Make: Model: Year: Approximate mileage:	who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only	Do not deduct secured claim amount of any secured claims Creditors Who Have Claims Current value of the entire property?	s or exemptions. Put the is on <i>Schedule D:</i> Secured by Property. Current value of th portion you own?
Do you oyou own 3. Cars, N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle of the someone else drives, sport utility vehicle of es. Make: Model: Year: Approximate mileage: Other information:	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions)	Do not deduct secured claim amount of any secured claims Creditors Who Have Claims Current value of the entire property?	s or exemptions. Put the is on <i>Schedule D:</i> Secured by Property. Current value of th portion you own?
Do you o you own 3. Cars, N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a velous vans, trucks, tractors, sport utility vehico es Make: Model: Year: Approximate mileage:	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions)	Do not deduct secured claim amount of any secured claims Creditors Who Have Claims Current value of the entire property?	s or exemptions. Put the is on <i>Schedule D:</i> Secured by Property. Current value of th portion you own?
Do you o you own 3. Cars, N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle of the someone else drives, sport utility vehicle of es. Make: Model: Year: Approximate mileage: Other information:	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions)	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim	s or exemptions. Put the is on Schedule D: Secured by Property. Current value of th portion you own? \$
Do you oyou own 3. Cars, N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle of the someone else drives, sport utility vehicle of es. Make: Model: Year: Approximate mileage: Other information:	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions)	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the entire property?	s or exemptions. Put the is on Schedule D: Secured by Property. Current value of th portion you own? \$ s or exemptions. Put the is on Schedule D:
Do you oyou own 3. Cars, N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle of the someone else drives, if you lease a vehicle of the someone else drives, sport utility vehicle of es. Make: Model: Year: Approximate mileage: Other information: own or have more than one, describe here: Make: Model: Model:	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions)	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim amount of any secured claim Creditors Who Have Claims	s or exemptions. Put the is on Schedule D: Secured by Property. Current value of th portion you own? \$
Do you oyou own 3. Cars, N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vertex vans, trucks, tractors, sport utility vehice of es. Make: Model: Year: Approximate mileage: Other information: own or have more than one, describe here: Make: Model: Year:	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who is an owner of the property? Check one. Debtor 1 only	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim amount of any secured claim creditors Who Have Claims Current value of the	s or exemptions. Put the is on Schedule D: Secured by Property. Current value of th portion you own? \$
Do you oyou own 3. Cars, N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle of the someone else drives, if you lease a vehicle of the someone else drives, sport utility vehicle of es. Make: Model: Year: Approximate mileage: Other information: own or have more than one, describe here: Make: Model: Model:	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim amount of any secured claim Creditors Who Have Claims	s or exemptions. Put the is on Schedule D: Secured by Property. Current value of th portion you own? \$
Do you oyou own 3. Cars, N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vertex vans, trucks, tractors, sport utility vehice of es. Make: Model: Year: Approximate mileage: Other information: own or have more than one, describe here: Make: Model: Year:	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Obebtor 2 only At least one of the debtors and another	Do not deduct secured claim amount of any secured claims Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the entire property?	s or exemptions. Put the is on Schedule D: Secured by Property. Current value of the portion you own? \$ s or exemptions. Put the is on Schedule D: Secured by Property. Current value of the portion you own?
Do you oyou own 3. Cars, N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle of the content	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Obebtor 2 only At least one of the debtors and another	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim amount of any secured claim creditors Who Have Claims Current value of the	s or exemptions. Put the is on Schedule D: Secured by Property. Current value of the portion you own? \$ s or exemptions. Put the is on Schedule D: Secured by Property. Current value of the

	First Name Middle Na	ame Last Name	Case number	r (if known)	
	FIIST NAME MIGGIE NA	ame Last Name			
3.3.	Make: _		Who is an owner of the property? Check one.	Do not deduct secured claim amount of any secured claim	
	Model: _		Debtor 1 only	Creditors Who Have Claims	
	Year: _		Debtor 2 only Debtor 1 and Debtor 2 only	Current value of the	Current value of the
	Approximate mileage: _		At least one of the debtors and another	entire property?	portion you own?
	Other information:			¢	\$
			☐ Check if this is community property (see instructions)	Φ	Φ
3.4.	Make: _		Who is an owner of the property? Check one.	Do not deduct secured claim amount of any secured claim	
	Model: _		Debtor 1 only	Creditors Who Have Claims	
	Year: _		Debtor 2 only Debtor 1 and Debtor 2 only	Current value of the	Current value of the
	Approximate mileage: _		At least one of the debtors and another	entire property?	portion you own?
	Other information:			¢.	c
			Check if this is community property (see instructions)	Φ	\$
			r recreational vehicles, other vehicles, and ac		
	<i>nples:</i> Boats, trailers, moto lo		er recreational vehicles, other vehicles, and ac ft, fishing vessels, snowmobiles, motorcycle acce		
□ N	<i>nples:</i> Boats, trailers, moto lo es	ors, personal watercra	ft, fishing vessels, snowmobiles, motorcycle acce	essories	s or exemptions. But the
	nples: Boats, trailers, moto lo res Make:	ors, personal watercra	ft, fishing vessels, snowmobiles, motorcycle acce Who is an owner of the property? Check one.	Do not deduct secured claim amount of any secured claim	s on Schedule D:
□ N	mples: Boats, trailers, moto lo les Make: Model:	ors, personal watercra	ft, fishing vessels, snowmobiles, motorcycle acce Who is an owner of the property? Check one. Debtor 1 only	essories Do not deduct secured claim	s on Schedule D:
□ N	mples: Boats, trailers, moto lo les Make: Model: Year:	ors, personal watercra	ft, fishing vessels, snowmobiles, motorcycle acce Who is an owner of the property? Check one.	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the	s on Schedule D: Secured by Property. Current value of the
□ N	mples: Boats, trailers, moto lo les Make: Model:	ors, personal watercra	th, fishing vessels, snowmobiles, motorcycle accessification. Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims	s on Schedule D: Secured by Property.
□ N	mples: Boats, trailers, moto lo les Make: Model: Year:	ors, personal watercra	th, fishing vessels, snowmobiles, motorcycle accessification. Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the	s on Schedule D: Secured by Property. Current value of the
□ N	mples: Boats, trailers, moto lo les Make: Model: Year:	ors, personal watercra	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the entire property?	s on Schedule D: Secured by Property. Current value of the portion you own?
□ N □ Y 4.1.	mples: Boats, trailers, moto lo les Make: Model: Year:	ors, personal watercra	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the entire property?	s on Schedule D: Secured by Property. Current value of the portion you own?
□ N □ Y 4.1.	mples: Boats, trailers, motoriolo res Make: Model: Year: Other information:	ors, personal watercra	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property	Do not deduct secured claim amount of any secured claims Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim	s on Schedule D: Secured by Property. Current value of the portion you own? \$
 N Y4.1.	mples: Boats, trailers, motoriolo res Make: Model: Year: Other information:	ors, personal watercra	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions)	Do not deduct secured claim amount of any secured claims Creditors Who Have Claims Current value of the entire property?	s on Schedule D: Secured by Property. Current value of the portion you own? \$
□ N□ Y4.1.	mples: Boats, trailers, motor location in the second secon	ors, personal watercra	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions)	Do not deduct secured claim amount of any secured claims Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim amount of any secured claim Who Have Claims Secured II	s on Schedule D: Secured by Property. Current value of the portion you own? \$
If you	mples: Boats, trailers, motor lookes Make: Model: Year: Other information: Jown or have more than of Make: Model: Year: Model: Year:	ors, personal watercra	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 only Debtor 2 only	Do not deduct secured claim amount of any secured claim Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim amount of any secured claim	s on Schedule D: Secured by Property. Current value of the portion you own? \$
If you	mples: Boats, trailers, motor lookes Make: Model: Year: Other information: I own or have more than of Make: Model: Model:	ors, personal watercra	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only	Do not deduct secured claim amount of any secured claims Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim amount of any secured claim Who Have Claims Secured It	s on Schedule D: Secured by Property. Current value of the portion you own? \$
If you	mples: Boats, trailers, motor lookes Make: Model: Year: Other information: Jown or have more than of Make: Model: Year: Model: Year:	ors, personal watercra	Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who is an owner of the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 only Debtor 2 only	Do not deduct secured claim amount of any secured claims Creditors Who Have Claims Current value of the entire property? \$ Do not deduct secured claim amount of any secured claim Who Have Claims Secured It	s on Schedule D: Secured by Property. Current value of the portion you own? \$

5. Add the dollar value of the portion you own for all of your entries from Part 2, including any entries for pages you have attached for Part 2. Write that number here

(see instructions)

_		
De	htor	1

irot Nomo	Middle Neme	Last Name	

Case number	if known)
-------------	-----------

Part 3: Describe Your Personal and Household Items

Do you own or have any legal or equitable interest in any of the following items?	Current value of the portion you own? Do not deduct secured claims or exemptions.
6. Household goods and furnishings	
Examples: Major appliances, furniture, linens, china, kitchenware	
□ No	
Yes. Describe	\$
	·
7. Electronics	
Examples: Televisions and radios; audio, video, stereo, and digital equipment; computers, printers, scanners; music	
collections; electronic devices including cell phones, cameras, media players, games	
□ No	
Yes. Describe	\$
8. Collectibles of value	
Examples: Antiques and figurines; paintings, prints, or other artwork; books, pictures, or other art objects; stamp, coin, or baseball card collections; other collections, memorabilia, collectibles	
No	
Yes. Describe	\$
	Ψ
9. Equipment for sports and hobbies	
Examples: Sports, photographic, exercise, and other hobby equipment; bicycles, pool tables, golf clubs, skis; canoes	
and kayaks; carpentry tools; musical instruments	
No No	
Yes. Describe	\$
10. Firearms	
Examples: Pistols, rifles, shotguns, ammunition, and related equipment	
□ No	
☐ Yes. Describe	\$
11. Clothes	
Examples: Everyday clothes, furs, leather coats, designer wear, shoes, accessories	
□ No	
Yes. Describe	\$
40 January	
12. Jewelry Examples: Everyday jewelry, costume jewelry, engagement rings, wedding rings, heirloom jewelry, watches, gems,	
examples: Everyday jeweiry, costume jeweiry, engagement rings, wedding rings, neirioom jeweiry, watches, gems, gold, silver	
□ No	
Yes. Describe	\$
13. Non-farm animals	
Examples: Dogs, cats, birds, horses	
□ No	
Yes. Describe	\$
14. Any other personal and household items you did not already list, including any health aids you did not list	
□ No	
☐ Yes. Give specific	
information	\$
15. Add the dollar value of all of your entries from Part 3, including any entries for pages you have attached	_
for Part 3. Write that number here	*

_		
De	btor	1

			Case number (if known)
st Name	Middle Name	Last Name	

Part 4:	Describe	Your	Financial	Assets

Do you own or have any l	egal or equitable interest in a	any of the following?		Current value of the portion you own? Do not deduct secured claims or exemptions.
16. Cash Examples: Money you h	nave in your wallet, in your hom	ne, in a safe deposit box, and on hand when you	file your petition	
			Cash:	\$
		nts; certificates of deposit; shares in credit union ultiple accounts with the same institution, list eac		
☐ No ☐ Yes		Institution name:		
	17.1. Checking account:			\$
	17.2. Checking account:			\$
	17.3. Savings account:			\$
	17.4. Savings account:			\$
	17.5. Certificates of deposit:			\$
	17.6. Other financial account:			\$
	17.7. Other financial account:			\$
	17.8. Other financial account:			\$
	17.9. Other financial account:			\$
18. Bonds, mutual funds, Examples: Bond funds,		erage firms, money market accounts		
☐ Yes	Institution or issuer name:			
				\$
				*
				\$
19. Non-publicly traded st an LLC, partnership, a		rated and unincorporated businesses, includi	ing an interest in	
□ No	Name of entity:		% of ownership:	
Yes. Give specific information about				\$
them				\$
			%	\$

Government and corporal Negotiable instruments incl Non-negotiable instruments	ate bonds and other	s, cashiers' checks, pro	negotiable instrume	nts noney orders.	
Negotiable instruments incl Non-negotiable instruments	clude personal checks ts are those you cann	s, cashiers' checks, pro	omissory notes, and n	noney orders.	
Negotiable instruments incl Non-negotiable instruments	clude personal checks ts are those you cann	s, cashiers' checks, pro	omissory notes, and n	noney orders.	
	ssuer name:				
	ssuer name:				
information about them					\$
_					\$
_					\$
Retirement or pension ac	ccounts				
<u>-</u>		(k), 403(b), thrift savin	ngs accounts, or other	pension or profit-sharing pla	ans
□ No					
Yes. List each account separately Ty	Гуре of account: I	nstitution name:			
41	101(k) or similar plan:				\$
P	Pension plan:				\$
	RA:				
	Retirement account:				_
K	Ceogh:				\$
	Additional account:				 \$
	Additional account:				
7.0	tadiionar account.				Ψ
Security denocite and pro	on avmonto				
Security deposits and pre Your share of all unused de		de so that you may co	ntinue service or use	from a company	
Examples: Agreements with companies, or others	th landlords, prepaid	rent, public utilities (ele	ectric, gas, water), tele	ecommunications	
□ No					
☐ Yes	Instit	ution name or individua	al:		
E	Electric:				
G	Gas:				- \$
н	Heating oil:				- \$
S	Security deposit on renta	I unit:			- \$
	Prepaid rent:				- \$
	Γelephone:				- \$
	Nater:				- \$
	Rented furniture: Other:				- \$

Debtor 1		Case number (##	nown)	
First Name Middle Name	Last Name			
24. Interests in an education IRA as d 26 U.S.C. § 529(b)(1).	lefined in 26 U.S.C. § 530(b)(1) or	under a qualified state tuition plan a	as defined in	
□ No □ YesInsti	itution name and description. Sepa	arately file the records of any interes	ts.11 U.S.C. § 521(c):	
				\$
				\$
				\$
25. Trusts, equitable or future interes exercisable for your benefit	ts in property (other than anyth	ing listed in line 1), and rights or p	powers	
□ No				
Yes. Give specific information about them				\$
26. Patents, copyrights, trademarks,	trade secrets, and other intellec	tual property		
Examples: Internet domain names,	websites, proceeds from royalties	and licensing agreements		
☐ No☐ Yes. Give specific				
information about them				\$
27. Licenses, franchises, and other g <i>Examples</i> : Building permits, exclusive	=	on holdings liquor licenses professi	ional licenses	
□ No	To noonlood, cooperative accordant			
☐ Yes. Give specific				_
information about them				\$
Money or property owed to you?				Current value of the portion you own? Do not deduct secured claims or exemptions.
28. Tax refunds owed to you				,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
☐ No				
Yes. Give specific information about them, including whet	thor		Federal:	\$
you already filed the return	ns		State:	\$
and the tax years			Local:	\$
29. Family support <i>Examples:</i> Past due or lump sum ali	imony, spousal support, child supp	port, maintenance, divorce settleme	nt, property settlement	
□ No		1		
☐ Yes. Give specific information			Alimony:	\$
			Maintenance:	\$
			Support:	\$
			Divorce settlement:	\$
			Property settlement:	\$

30. **Other amounts someone owes you**Examples: Unpaid wages, disability insurance payments, disability benefits, sick pay, vacation pay, workers' compensation, Social Security benefits; unpaid loans you made to someone else

☐ No

☐ Yes. Give specific information.....

Debtor 1				Case number (if known)	
	First Name	Middle Name	Last Name		
	s in insuranc es: Health dis	•	ce: health savings accoun	t (HSA); credit, homeowner's, or renter's insurance	
□ No	, , , , , , , , , , , , , , , , , , ,	ac, ccca.a.	oo, noam oarnige accoun	. (
		surance company	Company name:	Beneficiary:	Surrender or refund value:
	of each polic	y and list its value			\$
					\$
					\$
32 Any inte	erest in nrone	erty that is due you	from someone who has	died	,
If you ar property	e the benefici			insurance policy, or are currently entitled to receive	
☐ No	Cius sassifis	info monations			
☐ Yes.	Give specific	information			\$
oo Claima	awainat third	martina whathar a	net yev beye filed a leve	auit au mada a damand fau naumant	
	_		s, insurance claims, or rigl	suit or made a demand for payment hts to sue	
☐ No		, , , ,			
Yes.	Describe eac	ch claim			Φ.
04					\$
to set o	ontingent and ff claims	a uniiquidated ciain	is of every nature, includ	ling counterclaims of the debtor and rights	
☐ No		r			
☐ Yes.	Describe eac	ch claim			\$
		ı			
4					
-	inciai assets	you did not already	/ IISt		
☐ No	Give specific	information			
— 103.	Oive specific	inionnation			\$
as Add the	dollar value	of all of your entrie	se from Part / including	any entries for pages you have attached	
		number here	s nom rart 4, moruumg		\$
Part 5:	Describe	Any Business-	Related Property Ye	ou Own or Have an Interest In. List any r	eal estate in Part 1.
37 Do vo u 6	own or have	any legal or equital	ole interest in any busine	ess-related property?	
	Go to Part 6.	any logar or equitar	ne interest in any basine	iss related property.	
	Go to line 38				
					Current value of the portion you own? Do not deduct secured claims
					or exemptions.
38. Account	ts receivable	or commissions yo	ou already earned		
☐ No					-
Yes.	Describe				\$
20 Office -	auinment fo	rnichings and sur	nline		
		rnishings, and sup ted computers, software		ax machines, rugs, telephones, desks, chairs, electronic devices	
☐ No					_
Yes.	Describe				\$

Debtor 1	Case number (if known)

40. Machinery, fixtures,	equipment, supplies you use in business, and tools of your trade		
☐ No☐ Yes. Describe			
			\$
41. Inventory No			
Yes. Describe			\$
42. Interests in partnersl			_
□ No	ipo or joint voltareo		
Yes. Describe	Name of entity:	% of ownership:	
		, o o. o	\$
		%	\$
			\$
	ng lists, or other compilations		
□ No	to the land of the transfer land of the transfer of the transf	\\ ^	
	s include personally identifiable information (as defined in 11 U.S.C. § 101(41A	.))?	
☐ No ☐ Yes. Des	and a		
☐ Yes. Des	Cribe		\$
	I property you did not already list		
□ No			
Yes. Give specific information			\$
illioilliation			\$
			\$
			\$
			\$
			\$
	of all of your entries from Part 5, including any entries for pages you have at number here		\$
ioi i ait 5. Wille that	Tuttiber fiere		
	Any Farm- and Commercial Fishing-Related Property You Own or Ha or have an interest in farmland, list it in Part 1.	ve an Interest Ir	ı.
,	***		
No. Go to Part 7.	any legal or equitable interest in any farm- or commercial fishing-related pro	perty?	
Yes. Go to line 47			
			Current value of the
			portion you own? Do not deduct secured claims
_			or exemptions.
47. Farm animals	and the Comment of Calif		
	poultry, farm-raised fish		
□ No □ Yes			
■ res			
			\$

Dehtor	1	

			Case number (if known)_
First Name	Middle Name	Last Name	

48. Crops—either growing or harvested			
☐ No ☐ Yes. Give specific information			\$
49. Farm and fishing equipment, implements, machinery, fixtur	res, and tools of trade		_
☐ Yes			
			\$
50. Farm and fishing supplies, chemicals, and feed No			
☐ Yes			1
			\$
51. Any farm- and commercial fishing-related property you did	l not already list		
☐ Yes. Give specific			
information			\$
52. Add the dollar value of all of your entries from Part 6, inclu for Part 6. Write that number here	• • •	_	\$
Part 7: Describe All Property You Own or Have	e an Interest in Th	at You Did Not List Above	
53. Do you have other property of any kind you did not already	y list?		
Examples: Season tickets, country club membership			
☐ No☐ Yes. Give specific			\$
information			\$
			\$
54. Add the dollar value of all of your entries from Part 7. Write	e that number here	→	\$
Part 8: List the Totals of Each Part of this Form	m		
55. Part 1: Total real estate, line 2		→	\$
56. Part 2: Total vehicles, line 5	\$		
57. Part 3: Total personal and household items, line 15	\$		
58. Part 4: Total financial assets, line 36	\$	_	
59. Part 5: Total business-related property, line 45	\$	_	
60. Part 6: Total farm- and fishing-related property, line 52	\$	_	
61. Part 7: Total other property not listed, line 54	+\$	_	
62. Total personal property. Add lines 56 through 61	. \$	Copy personal property total ->	+\$
63. Total of all property on Schedule A/B. Add line 55 + line 62			\$

Draft March 7, 2014

Fill in this information to identify your case:					
Debtor 1					
	First Name	Middle Name	Last Name		
Debtor 2					
(Spouse, if filing)	First Name	Middle Name	Last Name		
United States Bankruptcy Court for the:		District of (State)			
Case number (If known)					

☐ Check if this is an amended filing

Official Form 106C

Schedule C: The Property You Claim as Exempt

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. Using the property you listed on *Schedule A/B: Property* (Official Form 106A/B) as your source, list the property that you claim as exempt. If more space is needed, fill out and attach to this page as many copies of *Part 2: Additional Page* as necessary. On the top of any additional pages, write your name and case number (if known).

For each item of property you claim as exempt, you must specify the amount of the exemption you claim. One way of doing so is to state a specific dollar amount as exempt. Alternatively, you may claim the full fair market value of the property being exempted up to the amount of any applicable statutory limit. Some exemptions—such as those for health aids, rights to receive certain benefits, and tax-exempt retirement funds—may be unlimited in dollar amount. However, if you claim an exemption of 100% of fair market value under a law that limits the exemption to a particular dollar amount and the value of the property is determined to exceed that amount, your exemption would be limited to the applicable statutory amount.

Part 1: **Identify the Property You Claim as Exempt** 1. Which set of exemptions are you claiming? Check one only, even if your spouse is filing with you. ☐ You are claiming state and federal nonbankruptcy exemptions. 11 U.S.C. § 522(b)(3) ☐ You are claiming federal exemptions. 11 U.S.C. § 522(b)(2) 2. For any property you list on Schedule A/B that you claim as exempt, fill in the information below. Brief description of the property and line on Amount of the exemption you claim Specific laws that allow exemption Current value of the Schedule A/B that lists this property portion you own Check only one box for each exemption. Copy the value from Schedule A/B Brief description: ☐ 100% of fair market value, up to Line from any applicable statutory limit Schedule A/B: Brief **□** \$ description: ☐ 100% of fair market value, up to Line from any applicable statutory limit Schedule A/B: **\$** description: ☐ 100% of fair market value, up to I ine from any applicable statutory limit 3. Are you claiming a homestead exemption of more than \$155,675? (Subject to adjustment on 4/01/16 and every 3 years after that for cases filed on or after the date of adjustment.) ☐ No Yes. Did you acquire the property covered by the exemption within 1,215 days before you filed this case? Yes

Case number	(if known)	

Part 2: Additional Page

Brief description: Line from Schedule A/B:	arket value, up tostatutory limit
description: Line from Schedule A/B: Brief description: Line from Schedule A/B: Select Se	arket value, up tostatutory limit
Brief description: Line from Schedule A/B: Brief	·
description: Line from Schedule A/B: Brief	
Schedule A/B: any applicable	
Brief	
description: \$ \$	
Line from Schedule A/B: 100% of fair m any applicable	
Brief	
Line from 100% of fair m any applicable	
Brief description: \$\$ \$	
Line from Schedule A/B: 100% of fair m any applicable	
Brief	
Line from Schedule A/B: 100% of fair m any applicable	
Brief description: \$\$ \$	
Line from any applicable	
Brief description: \$\$	
Line from Schedule A/B: any applicable	
Brief	
Line from Schedule A/B: 100% of fair m any applicable	
Brief	
Line from Schedule A/B: 100% of fair m any applicable	
Brief description: \$ \$	
Line from Schedule A/B: 100% of fair m any applicable	
Brief	
Line from Schedule A/B: 100% of fair m any applicable	

Draft March 14, 2014

Fill in this information to identify your case:					
Debtor 1	First Name	Middle Name	Last Name		
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name		
United States E	Bankruptcy Court fo	District of (State)			
Case number (If known)					

☐ Check if this is an amended filing

Official Form 106D

Schedule D: Creditors Who Have Claims Secured by Property

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct

	nformation. If more space is needed, copy dditional pages, write your name and cas	the Additional Page, fill it out, number the entries, e number (if known).	and attach it to this	form. On the top of	any
1.	Do any creditors have claims secured by ☐ No. Check this box and submit this form ☐ Yes. Fill in all of the information below.	y your property? In to the court with your other schedules. You have nothing	ng else to report on	this form.	
Pa	art 1: List All Secured Claims				
2.	for each claim. If more than one creditor ha	ore than one secured claim, list the creditor separately as a particular claim, list the other creditors in Part 2. abetical order according to the creditor's name.	Column A Amount of claim Do not deduct the value of collateral.	Column B Value of collateral that supports this claim	Column C Unsecured portion If any
2.1]	Describe the property that secures the claim:	\$	\$	\$
	Creditor's Name				
	Number Street	As of the date you file, the claim is: Check all that apply.			
		Contingent			
		☐ Unliquidated			

		Contingent
	0.1.7100.1	_ Unliquidated
City	State ZIP Code	Disputed
Who owes the deb	ot? Check one.	☐ Liquidated and neither contingent nor disputed
Debtor 1 only		Nature of lien. Check all that apply.
Debtor 2 only		An agreement you made (such as mortgage or secured
Debtor 1 and De	btor 2 only	car loan)
At least one of the	ne debtors and another	Statutory lien (such as tax lien, mechanic's lien)
□ o bsects ###:	la!lataa ta a	Judgment lien from a lawsuit
community de	laim relates to a bt	Other (including a right to offset)
Date debt was inc		Last 4 digits of account number
2.2		Describe the property that secures the claim: \$\$\$
Creditor's Name		
Number Stree	t	
		As of the date you file, the claim is: Check all that apply.
		Contingent
		Unliquidated
City	State ZIP Code	☐ Disputed
Who owes the deb	ot? Check one.	☐ Liquidated and neither contingent nor disputed
Debtor 1 only		Nature of lien. Check all that apply.
Debtor 2 only		An agreement you made (such as mortgage or secured
Debtor 1 and De	btor 2 only	car loan)
At least one of the	ne debtors and another	Statutory lien (such as tax lien, mechanic's lien)
□ o bsects ###:	la!lataa ta a	Judgment lien from a lawsuit
community de	laim relates to a bt	Other (including a right to offset)
Date debt was inc	urred	Last 4 digits of account number
Add the dollar	value of your entries in	Column A on this page. Write that number here:

Debtor 1				Case number (if known)
	First Name	Middle Name	Last Name	

Part 1:	Additional Page After listing any entries on this p by 2.4, and so forth.	Column A Amount of claim Do not deduct the value of collateral.	Column B Value of collateral that supports this claim	Column C Unsecured portion If any	
		Describe the property that secures the claim:	\$	\$	6
Creditor	r's Name			·	
Number	Street				
		As of the date you file, the claim is: Check all that apply.			
0::	7100	Contingent			
City	State ZIP Code	Unliquidated			
		Disputed			
Who ow	ves the debt? Check one.	☐ Liquidated and neither contingent nor disputed			
	tor 1 only	Nature of lien. Check all that apply.			
	tor 2 only	☐ An agreement you made (such as mortgage or secured			
Debt	tor 1 and Debtor 2 only	car loan)			
☐ At le	ast one of the debtors and another	Statutory lien (such as tax lien, mechanic's lien)			
☐ Che	ck if this claim relates to a	Judgment lien from a lawsuit			
	munity debt	Other (including a right to offset)			
Date de	bt was incurred	Last 4 digits of account number			
		Describe the manufactual entrement that a comment the algebra	Φ.	\$ 5	<u> </u>
Creditor	r's Name	Describe the property that secures the claim:	\$	\$;	Þ
O. Gaito.	· · · · · · · · · · · · · · · · · · ·				
Number	r Street				
		As of the date you file, the claim is: Check all that apply.	l		
		☐ Contingent			
		☐ Unliquidated			
City	State ZIP Code	☐ Disputed			
Who ow	ves the debt? Check one.	☐ Liquidated and neither contingent nor disputed			
_	tor 1 only				
	tor 2 only	Nature of lien. Check all that apply.			
	tor 1 and Debtor 2 only	An agreement you made (such as mortgage or secured			
	east one of the debtors and another	car loan) Statutory lien (such as tax lien, mechanic's lien)			
	ast one of the debtors and unother	Judgment lien from a lawsuit			
	ck if this claim relates to a	Other (including a right to offset)			
com	nmunity debt	Girler (including a right to onset)			
Date de	bt was incurred	Last 4 digits of account number			
		Describe the property that secures the claim:	\$	\$	\$
Creditor	's Name				
Number	r Street				
Number	Sueet				
		As of the date you file, the claim is: Check all that apply.			
		☐ Contingent			
City	State ZIP Code	☐ Unliquidated			
,		☐ Disputed			
Who ou	ves the debt? Check one.	☐ Liquidated and neither contingent nor disputed			
_					
	tor 1 only	Nature of lien. Check all that apply.			
	tor 2 only	An agreement you made (such as mortgage or secured			
_	tor 1 and Debtor 2 only	car loan) Statutory lien (such as tax lien, mechanic's lien)			
☐ At le	ast one of the debtors and another	Judgment lien from a lawsuit			
☐ Check if this claim relates to a		Other (including a right to offset)			
com	nmunity debt	— Caron (moradang a right to onsor)			
Date de	bt was incurred	Last 4 digits of account number			
	Add the deller velve of very anti-	in Column A on this page Write that number have			
		s in Column A on this page. Write that number here:	\$		
	this is the last page of your form,	add the dollar value totals from all pages.	\$		

Debtor 1	 Case number (if known)

First Name Middle Name Last Name Part 2: List Others to Be Notified for a Debt That You Already Listed Use this page only if you have others to be notified about your bankruptcy for a debt that you already listed in Part 1. For example, if a collection agency is trying to collect from you for a debt you owe to someone else, list the creditor in Part 1, and then list the collection agency here. Similarly, if you have more than one creditor for any of the debts that you listed in Part 1, list the additional creditors here. If you do not have additional persons to be notified for any debts in Part 1, do not fill out or submit this page. On which line in Part 1 did you enter the creditor? _ Name Last 4 digits of account number Number Street City State ZIP Code On which line in Part 1 did you enter the creditor? __ Name Last 4 digits of account number _ Number Street City ZIP Code State On which line in Part 1 did you enter the creditor? _ Name Last 4 digits of account number _ Number Street City State ZIP Code On which line in Part 1 did you enter the creditor? _ Name Last 4 digits of account number _ Number Street City State ZIP Code On which line in Part 1 did you enter the creditor? _ Name Last 4 digits of account number _ Number Street City State ZIP Code On which line in Part 1 did you enter the creditor? _ Name Last 4 digits of account number _

Number

City

Street

ZIP Code

April 22-23, 2014 Page 108 of 596

Fill in this information to identify your case:					
Debtor 1					
	First Name	Middle Name	Last Name		
Debtor 2					
(Spouse, if filing)	First Name	Middle Name	Last Name		
United States Bankruptcy Court for the:			District of		
Case number (If known)					

Draft March 14, 2014

Check if this is an amended filing

Official Form 106E/F

Schedule E/F: Creditors Who Have Unsecured Claims

12/15

Be as complete and accurate as possible. Use Part 1 for creditors with PRIORITY claims and Part 2 for creditors with NONPRIORITY claims. List the other party to any executory contracts or unexpired leases that could result in a claim. Also list executory contracts on Schedule A/B: Property (Official Form 106A/B) and on Schedule G: Executory Contracts and Unexpired Leases (Official Form 106G). Do not include any creditors with partially secured claims that are listed in Schedule D: Creditors Who Hold Claims Secured by Property. If more space is needed, copy the Part you need, fill it out, number the entries in the boxes on the left. Attach the Continuation Page to this page. If you have no information to report in a Part, do not file that Part. On the top of any additional pages, write your name and case number (if known).

t 1: List All of Your PRIORITY Unsecur	ed Claims			
Oo any creditors have priority unsecured claim No. Go to Part 2.	s against you?			
Yes.				
each claim listed, identify what type of claim it is. If nonpriority amounts. As much as possible, list the unsecured claims, fill out the Continuation Page of	editor has more than one priority unsecured claim, list the a claim has both priority and nonpriority amounts, list the claims in alphabetical order according to the creditor's neart 1. If more than one creditor holds a particular claim instructions for this form in the instruction booklet.)	at claim here a ame. If you hav	nd sȟow both re more than t	priority and wo priority
, or all oxplaination of odoll type of stalling occurren		Total claim	Priority amount	Nonpriorit amount
		•	•	•
Priority Creditor's Name	Last 4 digits of account number	\$	_ \$	\$
Filolity Greditor's Name	When was the debt incurred?			
Number Street	When was the dept mounted:			
	As of the date you file, the claim is: Check all that apply	,		
		/.		
City State ZIP Code	☐ Contingent			
Who incurred the debt? Check one.	☐ Unliquidated			
Debtor 1 only	Disputed			
Debtor 2 only	Liquidated and neither contingent nor disputed			
Debtor 1 and Debtor 2 only	Type of PRIORITY unsecured claim:			
☐ At least one of the debtors and another	☐ Domestic support obligations			
☐ Check if this claim is for a community debt	☐ Taxes and certain other debts you owe the government			
Is the claim subject to offset?	Claims for death or personal injury while you were			
No	intoxicated			
☐ Yes	Other. Specify	_		
	Last 4 digits of account number	¢	•	œ.
Priority Creditor's Name	When was the debt incurred?	Ψ	Ψ	Φ
	when was the debt incurred?			
Number Street	As of the date you file, the claim is: Check all that apply	,		
	☐ Contingent	,		
City State ZIP Code	Unliquidated			
	☐ Disputed			
Who incurred the debt? Check one.	☐ Liquidated and neither contingent nor disputed			
☐ Debtor 1 only ☐ Debtor 2 only				
Debtor 2 only Debtor 1 and Debtor 2 only	Type of PRIORITY unsecured claim:			
At least one of the debtors and another	■ Domestic support obligations			
_	☐ Taxes and certain other debts you owe the government			
☐ Check if this claim is for a community debt	☐ Claims for death or personal injury while you were			
Is the claim subject to offset?	intoxicated			
□ No	Other. Specify	-		
☐ Yes				

-	htn	4

First Name Middle Name Last Name

Case number (if known)	
------------------------	--

Part 1: Your PRIORITY Unsecured Claims – Continuation Page

Afte	er listing any entries on this page, number them	beginning with 2.3, followed by 2.4, and so forth.	Total claim	Priority amount	Nonpriority amount
	Priority Creditor's Name	Last 4 digits of account number	\$	\$	\$
	Number Street	When was the debt incurred?			
	Number Street	As of the date you file, the claim is: Check all that apply.			
		☐ Contingent			
	City State ZIP Code	☐ Unliquidated ☐ Disputed			
	Who incurred the debt? Check one.	☐ Liquidated and neither contingent nor disputed			
	☐ Debtor 1 only ☐ Debtor 2 only	Type of PRIORITY unsecured claim:			
	Debtor 2 only Debtor 1 and Debtor 2 only				
	☐ At least one of the debtors and another	Domestic support obligationsTaxes and certain other debts you owe the government			
	☐ Check if this claim is for a community debt	 Claims for death or personal injury while you were intoxicated 			
	Is the claim subject to offset?	Other. Specify			
	□ No □ Yes				
		Last 4 digits of account number	\$	\$	\$
	Priority Creditor's Name			· · ·	·
	Number Street	When was the debt incurred?			
		As of the date you file, the claim is: Check all that apply.			
	City State ZIP Code	☐ Contingent ☐ Unliquidated			
	·	☐ Disputed			
	Who incurred the debt? Check one. Debtor 1 only	☐ Liquidated and neither contingent nor disputed			
	Debtor 2 only	Type of PRIORITY unsecured claim:			
	☐ Debtor 1 and Debtor 2 only ☐ At least one of the debtors and another	☐ Domestic support obligations			
	☐ Check if this claim is for a community debt	☐ Taxes and certain other debts you owe the government☐ Claims for death or personal injury while you were			
	•	intoxicated Other. Specify			
	Is the claim subject to offset?	Other. Specify			
	Yes				
		Last 4 digits of account number	\$	\$	\$
	Priority Creditor's Name	When was the debt incurred?			
	Number Street				
		As of the date you file, the claim is: Check all that apply.			
	City State ZIP Code	☐ Contingent ☐ Unliquidated			
	Who incurred the debt? Check one.	Disputed			
	Debtor 1 only	☐ Liquidated and neither contingent nor disputed			
	Debtor 2 only	Type of PRIORITY unsecured claim:			
	□ Debtor 1 and Debtor 2 only□ At least one of the debtors and another	Domestic support obligationsTaxes and certain other debts you owe the government			
	☐ Check if this claim is for a community debt	☐ Claims for death or personal injury while you were			
	•	intoxicated Other. Specify			
	Is the claim subject to offset?	- 11			
	☐ Yes				

_		
De	ht∩r	1

First Name Middle Name Last Name

Part 2:

List All of Your NONPRIORITY Unsecured Claims

3.	Do any creditors have nonpriority unsecured claims against you? No. You have nothing to report in this part. Submit this form to the court with your other schedules. Yes						
	List all of your nonpriority unsecured claims in the alphabetical order of the creditor who holds each claim. If a creditor has more than one priority unsecured claim, list the creditor separately for each claim. For each claim listed, identify what type of claim it is. Do not list claims already included in Part 1. If more than one creditor holds a particular claim, list the other creditors in Part 3.If you have more than four priority unsecured claims fill out the Continuation Page of Part 2.						
			Total claim				
1.1		Last 4 digits of account number					
	Nonpriority Creditor's Name	-	\$				
	Number Street	When was the debt incurred?					
	City State ZIP Code	As of the date you file, the claim is: Check all that apply.					
	City State ZIP Code	☐ Contingent					
	Who incurred the debt? Check one.	☐ Unliquidated					
	Debtor 1 only	☐ Disputed					
	Debtor 2 only	☐ Liquidated and neither contingent nor disputed					
	Debtor 1 and Debtor 2 only						
	☐ At least one of the debtors and another	Type of NONPRIORITY unsecured claim:					
		☐ Student loans					
	☐ Check if this claim is for a community debt	Obligations arising out of a separation agreement or divorce					
	Is the claim subject to offset?	that you did not report as priority claims Debts to pension or profit-sharing plans, and other similar debts					
	□ No	Other. Specify					
	Yes						
1.2		Last 4 digits of account number	\$				
	Nonpriority Creditor's Name	When was the debt incurred?					
	Number Street	As of the date you file, the claim is: Check all that apply.					
	City State ZIP Code	Contingent					
	Who incurred the debt? Check one.	☐ Unliquidated ☐ Disputed					
	☐ Debtor 1 only	☐ Liquidated and neither contingent nor disputed					
	☐ Debtor 2 only	Enquirated and nettres contangent not disputed					
	☐ Debtor 1 and Debtor 2 only	Type of NONPRIORITY unsecured claim:					
	☐ At least one of the debtors and another	☐ Student loans					
	☐ Check if this claim is for a community debt	Obligations arising out of a separation agreement or divorce that you did not report as priority claims					
	Is the claim subject to offset?	Debts to pension or profit-sharing plans, and other similar debts					
	□ No	Other. Specify					
	Yes						
1.3		Last 4 digits of account number	\$				
	Nonpriority Creditor's Name	When was the debt incurred?	\$				
	Number Street						
		As of the date you file, the claim is: Check all that apply.					
	City State ZIP Code	☐ Contingent					
	Who incurred the debt? Check one.	☐ Unliquidated					
	Debtor 1 only	☐ Disputed					
	Debtor 2 only	Liquidated and neither contingent nor disputed					
	☐ Debtor 1 and Debtor 2 only						
	☐ At least one of the debtors and another	Type of NONPRIORITY unsecured claim:					
	☐ Check if this claim is for a community debt	☐ Student loans					
		Obligations arising out of a separation agreement or divorce					
	Is the claim subject to offset?	that you did not report as priority claims					
	□ No	Debts to pension or profit-sharing plans, and other similar debts					
	☐ Yes	Other. Specify					

Debtor 1

First Name Middle Name

Last Name

Case number (if known)

Part 2:

Your NONPRIORITY Unsecured Claims — Continuation Page

Last 4 digits of account number	
When was the debt incurred?	
Number Street As of the date you file, the claim is: Check all that apply	
As of the date you me, the dam is. Once an that apply.	
City State ZIP Code Contingent Unliquidated Who incurred the debt? Check one.	
☐ Debtor 1 only ☐ Debtor 2 only ☐ Liquidated and neither contingent nor disputed	
☐ Debtor 1 and Debtor 2 only ☐ At least one of the debtors and another ☐ Student loans ☐ Student loans	
☐ Check if this claim is for a community debt ☐ Obligations arising out of a separation agreement or divorce that you did not report as priority claims ☐ Debts to pension or profit-sharing plans, and other similar debts	
□ No □ Other. Specify	
Last 4 digits of account number	
Nonpriority Creditor's Name When was the debt incurred?	
Number Street As of the date you file, the claim is: Check all that apply.	
City State ZIP Code Contingent Unliquidated Who incurred the debt? Check one.	
Who incurred the debt? Check one. ☐ Disputed ☐ Debtor 1 only ☐ Debtor 2 only ☐ Liquidated and neither contingent nor disputed	
☐ Debtor 1 and Debtor 2 only ☐ At least one of the debtors and another ☐ Student loans ☐ Student loans	
□ Check if this claim is for a community debt Is the claim subject to offset? □ Obligations arising out of a separation agreement or divorce that you did not report as priority claims □ Debts to pension or profit-sharing plans, and other similar debts	
□ No □ Other. Specify □ Yes □ Other. Specify	
Last 4 digits of account number	
Nonpriority Creditor's Name When was the debt incurred?	
Number Street As of the date you file, the claim is: Check all that apply.	
City State ZIP Code Contingent Unliquidated Who incurred the debt? Check one.	
Who incurred the debt? Check one. ☐ Disputed ☐ Debtor 1 only ☐ Debtor 2 only ☐ Debtor 2 only	
☐ Debtor 1 and Debtor 2 only Type of NONPRIORITY unsecured claim:	
☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim is for a community debt ☐ Check if this claim	
Is the claim subject to offset? □ Debts to pension or profit-sharing plans, and other similar debts □ No □ Other. Specify	

Debtor 1

First Name Middle Name Last Name

Case number (if known)_____

Part 3:

List Others to Be Notified About a Debt That You Already Listed

	•		•	ons to be notified for any debts in Parts 1 or 2, do not fill out or submit this page.
Now-				On which entry in Part 1 or Part 2 did you list the original creditor?
Name				Line of (Check one): ☐ Part 1: Creditors with Priority Unsecured Claims
Number	Street			Part 2: Creditors with Nonpriority Unsecured Claims
				Trait 2. Greaters with Non-phoney ensecured claim
				Last 4 digits of account number
City		State	ZIP Code	
				On which entry in Part 1 or Part 2 did you list the original creditor?
Name				
Number	Street			Line of (Check one): Part 1: Creditors with Priority Unsecured Claims
4011IDEI	Gueet			Part 2: Creditors with Nonpriority Unsecured Claims
				Last 4 digits of account number
City		State	ZIP Code	Last 4 digits of account number
				On which entry in Part 1 or Part 2 did you list the original creditor?
Name				Line of (Check one): Part 1: Creditors with Priority Unsecured Claims
Number	Street			Part 2: Creditors with Nonpriority Unsecured
				Claims
				Last 4 digits of account number
City		State	ZIP Code	
Name				On which entry in Part 1 or Part 2 did you list the original creditor?
·amo				Line of (Check one): Part 1: Creditors with Priority Unsecured Claims
Number	Street			□ Part 2: Creditors with Nonpriority Unsecured
				Claims
City		State	ZIP Code	Last 4 digits of account number
				On which entry in Part 1 or Part 2 did you list the original creditor?
Name				,
				Line of (Check one): ☐ Part 1: Creditors with Priority Unsecured Claims
Number	Street			☐ Part 2: Creditors with Nonpriority Unsecured Claims
				Ciaillis
City		State	ZIP Code	Last 4 digits of account number
,		5.00	5500	On which entry in Part 1 or Part 2 did you list the original creditor?
Name				
				Line of (Check one): Part 1: Creditors with Priority Unsecured Claims
Number	Street			☐ Part 2: Creditors with Nonpriority Unsecured Claims
				Cidano
City		State	ZIP Code	Last 4 digits of account number
				On which entry in Part 1 or Part 2 did you list the original graditor?
Name				On which entry in Part 1 or Part 2 did you list the original creditor?
Number	Street			Line of (Check one): Part 1: Creditors with Priority Unsecured Claims
INUITIDEI [*]	Sueel			☐ Part 2: Creditors with Nonpriority Unsecured Claims
City		State	ZIP Code	Last 4 digits of account number

Middle Name Last Name

Part 4:

Add the Amounts for Each Type of Unsecured Claim

6. Total the amounts of certain types of unsecured claims. This information is for statistical reporting purposes only. 28 U.S.C. §159. Add the amounts for each type of unsecured claim.

Total claims from Part 1

- 6a. Domestic support obligations
- 6b. Taxes and certain other debts you owe the government
- 6c. Claims for death or personal injury while you were intoxicated
- 6d. Other. Add all other priority unsecured claims. Write that amount here.
- 6e. Total. Add lines 6a through 6d.

Total claim

- 6a. _{\$}
- 6b. _{\$}
- 6c.
- 6d. + s
- 6e. \$______

Total claims from Part 2

- 6f. Student loans
- 6g. Obligations arising out of a separation agreement or divorce that you did not report as priority claims
- 6h. Debts to pension or profit-sharing plans, and other similar debts
- 6i. **Other.** Add all other nonpriority unsecured claims. Write that amount here.
- 6j. Total. Add lines 6f through 6i.

Total claim

- 6f.
- 6g. \$_____
- 6h. _{\$}
- 6i **→** ¢
- 6j. \$_____

Draft April 7, 2013

Fill in this information to identify your case:						
Debtor						
·	First Name	Middle Name	Last Name			
Debtor 2						
(Spouse If filing)	First Name	Middle Name	Last Name			
United States E	Bankruptcy Court	District of	 State)			
Case number (If known)			•			

☐ Check if this is an amended filing

Official Form 106G

Schedule G: Executory Contracts and Unexpired Leases

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, copy the additional page, fill it out, number the entries, and attach it to this page. On the top of any additional pages, write your name and case number (if known).

- 1. Do you have any executory contracts or unexpired leases?
 - ☐ No. Check this box and file this form with the court with your other schedules. You have nothing else to report on this form.
 - ☐ Yes. Fill in all of the information below even if the contracts or leases are listed on Schedule A/B: Property (Official Form 106A/B).
- 2. List separately each person or company with whom you have the contract or lease. Then state what each contract or lease is for (for example, rent, vehicle lease, cell phone). See the instructions for this form in the instruction booklet for more examples of executory contracts and unexpired leases.

	Person or	company wit	h whom you h	ave the contract or lease	State what the contract or lease is for
1					
	Name				
	Number	Street			
	City		State	ZIP Code	_
2					
	Name				
	Number	Street			
	City		State	ZIP Code	_
3					
	Name				
	Number	Street			
	City		State	ZIP Code	
4					
	Name				
	Number	Street			
	City		State	ZIP Code	
5					
	Name				
	Number	Street			
	City		State	ZIP Code	

Debtor 1

First Name	Middle Nome	Lact Namo	

Case number	(if known)	

Ad	dditional Paç	ge if You Ha	ve More Contracts or Leases	s ·
Person or	company wit	h whom you h	ave the contract or lease	What the contract or lease is for
Name				_
Number	Street			_
City		State	ZIP Code	_
				_
Name				
Number	Street			_
City		State	ZIP Code	_
				_
Name				
Number	Street			_
City		State	ZIP Code	_
Name				_
Number	Street			_
City		State	ZIP Code	_
Name				
Number	Street			_
City		State	ZIP Code	_
Name				_
Number	Street			_
City		State	ZIP Code	_
Name				_
Number	Street			_
City		State	ZIP Code	_
Name				_

Number

City

Street

State

ZIP Code

Draft March 24, 2014

Fill in this information to identify your case:					
Debtor 1					
_	First Name	Middle Name	Last Name		
Debtor 2					
(Spouse, if filing)	First Name	Middle Name	Last Name		
United States B	Bankruptcy Court for the:		_ District of(State)		
Case number (If known)					

☐ Check if this is an amended filing

Official Form 106H

Schedule H: Your Codebtors

12/15

Codebtors are people or entities who are also liable for any debts you may have. Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, copy the Additional Page, fill it out, and number the entries in the boxes on the left. Attach the Additional Page to this page. On the top of any Additional Pages, write your name and case number (if known). Answer every question.

1.	Do you No	have any codebto	rs? (If you are filing a joint case, do not	list either spouse as a	codebtor.)				
	☐ Yes	S							
2.		•	ave you lived in a community property Louisiana, Nevada, New Mexico, Puerto	Community property states and territories include gton, and Wisconsin.)					
	☐ No.	. Go to line 3.							
	☐ Yes	s. Did your spouse,	former spouse, or legal equivalent live w	vith you at the time?					
		No							
		Yes. In which comm	nunity state or territory did you live?	Fi	ill in the name and current address of that person.				
		Name of your spouse, for	ormer spouse, or legal equivalent						
		Number Street							
		City	State	ZIP Code					
3.	In Colu	ımn 1, list all of you	ır codebtors. Do not include your spo	use as a codebtor if	your spouse is filing with you. List the person				
		wn in line 2 again as a codebtor only if that person is a guarantor or cosigner. Make sure you have listed the creditor on							
				06E/F), or Schedule	G (Official Form 106G). Use Schedule D,				
	Scried	ule E/F, or Scriedu	le G to fill out Column 2.						
	Colun	nn 1: Your codebto	r		Column 2: The creditor to whom you owe the debt				
					Check all schedules that apply:				
3.1									
	Name				Schedule D, line				
					☐ Schedule E/F, line				
	Numbe	er Street			☐ Schedule G, line				
	City		State	ZIP Code	_				
3.2	· ·		5.2.0						
J.,_	Name				Schedule D, line				
	1100				☐ Schedule E/F, line				
	Numbe	er Street			☐ Schedule G, line				
	City		State	ZIP Code	_				
3.3			Otate	Zii Oode					
0.0	Name				Schedule D, line				
	ivanie				☐ Schedule E/F, line				
	Numbe	er Street			Schedule G, line				
	City		State	ZIP Code	_				
	City		State	ZIF Code					

D_{ℓ}	htor	- 1

iret Namo	Middle Name	Last Namo

Case number (i	f known)	
Case Hullibel (/	i Kriowrii	

	: Your codebtor			Column 2: The creditor to whom you owe the debt
Column	Tour couchion			Check all schedules that apply:
Name				Schedule D, line
				Schedule E/F, line
Number	Street			Genedule G, line
City		State	ZIP Code	_
				Schedule D, line
Name				☐ Schedule E/F, line
Number	Street			Schedule G, line
				_
City		State	ZIP Code	
Name				Schedule D, line
ivame				☐ Schedule E/F, line
Number	Street			Schedule G, line
City		Chala	ZIP Code	_
City		State	ZIP Code	
Name				Schedule D, line
ranio				☐ Schedule E/F, line
Number	Street			Schedule G, line
City		State	ZIP Code	_
Oity		Otato	Zii Oode	
 Name				Schedule D, line
				Schedule E/F, line
Number	Street			Schedule G, line
City		State	ZIP Code	_
 Name				Schedule D, line
				Schedule E/F, line
Number	Street			Schedule G, line
City		State	ZIP Code	
				Schedule D, line
 Name				Schedule E/F, line
				Schedule G, line
Number	Street			- outcodic o, line
 City		State	ZIP Code	
				Schedule D, line
Name				Schedule E/F, line
				Schedule G, line
Number	Street			Confedure O, line
City		State	ZIP Code	_

Draft March 24, 2014

Fill in this information to identify your case:						
Debtor 1	First Name	Middle Name	Last Name			
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name			
United States E	Bankruptcy Court for	District of (State)				
Case number (If known)		(State)				

☐ Check if this is an amended filing

Official Form 106Dec

Declaration About an Individual Debtor's Schedules

12/15

If two married people are filing together, both are equally responsible for supplying correct information.

You must file this form whenever you file bankruptcy schedules or amended schedules. Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

No	
Yes. Name of person	. Attach Bankruptcy Petition Preparer's Notice, Declaration, and
	Signature (Official Form 119).
	nave read the summary and schedules filed with this declaration and
	ave read the summary and schedules filed with this declaration and
der penalty of perjury, I declare that I l t they are true and correct.	ave read the summary and schedules filed with this declaration and
	nave read the summary and schedules filed with this declaration and
	nave read the summary and schedules filed with this declaration and

April 22-23, 2014 Page 120 of 596

Fill in this in	formation to identify	your case:	
Debtor 1			
_	First Name	Middle Name	Last Name
Debtor 2			
(Spouse, if filing)	First Name	Middle Name	Last Name
United States E	Bankruptcy Court for the:		District of
Case number	(If known)		

☐ Check if this is an amended filing

Official Form 106Sum

Summary of Your Assets and Liabilities and Certain Statistical Information 12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. Fill out all of your schedules first; then complete the information on this form. If you are filing amended schedules after you file your original forms, you must fill out a new *Summary* and check the box at the top of this page.

Part 1: Summarize Your Assets	
	Your assets Value of what you own
Schedule A/B: Property (Official Form 106A/B) 1a. Copy line 55, Total real estate, from Schedule A/B	\$
1b. Copy line 62, Total personal property, from Schedule A/B	\$
1c. Copy line 63, Total of all property on Schedule A/B	\$
Part 2: Summarize Your Liabilities	
	Your liabilities Amount you owe
 Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D) Copy the total you listed in Column A, Amount of claim, at the bottom of the last page of Part 1 of Schedule D 	\$
3. Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F) 3a. Copy the total claims from Part 1 (priority unsecured claims) from line 6e of Schedule E/F	\$
3b. Copy the total claims from Part 2 (nonpriority unsecured claims) from line 6j of Schedule E/F	+ \$
Your total liabilities	\$
Part 3: Summarize Your Income and Expenses	
4. Schedule I: Your Income (Official Form 106I) Copy your combined monthly income from line 12 of Schedule I	\$
Schedule J: Your Expenses (Official Form 106J) Copy your monthly expenses from line 22, Column A, of Schedule J	\$

ח	ρh	tο	r	1

First Name Middle Name Last Name Case number (if known)______

Pá	art 4: Answer These Questions for Administrative and Statistical Records	
6.	Are you filing for bankruptcy under Chapters 7, 11, or 13? No. You have nothing to report on this part of the form. Check this box and submit this form. Yes	orm to the court with your other schedules.
7.	What kind of debt do you have? Your debts are primarily consumer debts. Consumer debts are those "incurred by an family, or household purpose." 11 U.S.C. § 101(8). Fill out lines 8-10 for statistical purpo Your debts are not primarily consumer debts. You have nothing to report on this part this form to the court with your other schedules.	ses. 28 U.S.C. § 159.
8.	From the Statement of Your Current Monthly Income (Official Form 108-1, 109, or 110-1 Copy your total current monthly income from line 11.	\$
9.	Copy the following special categories of claims from Part 4, line 6 of <i>Schedule E/F</i> : From Part 4 on <i>Schedule E/F</i> , copy the following:	Total claim
	9a. Domestic support obligations (Copy line 6a.)	\$
	9b. Taxes and certain other debts you owe the government. (Copy line 6b.)9c. Claims for death or personal injury while you were intoxicated. (Copy line 6c.)	\$ \$
	9d. Student loans. (Copy line 6f.)9e. Obligations arising out of a separation agreement or divorce that you did not report as priority claims. (Copy line 6g.)	\$ \$
	9f. Debts to pension or profit-sharing plans, and other similar debts. (Copy line 6h.) 9g. Total. Add lines 9a through 9f.	+ \$ \$

COMMITTEE NOTE

The schedules to be used in cases of individual debtors are revised as part of the Forms Modernization Project, making them easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats. Therefore, many of the open-ended questions and multiple-part instructions have been replaced with more specific questions. The individual debtor schedules are also renumbered, starting with the number 106 and followed by the letter or name of the schedule to distinguish them from the versions to be used in non-individual cases.

Official Form 106Sum, Summary of Your Assets and Liabilities and Certain Statistical Information, replaces Official Form 6, Summary of Schedules and Statistical Summary of Certain Liability and Related Data (28 U.S.C. § 159), in cases of individual debtors.

The form is reformatted and updated with cross-references indicating the line numbers of specific schedules from which the summary information is to be gathered. In addition, because most filings are now done electronically, the form no longer requires the debtor to indicate which schedules are attached or to state the number of sheets of paper used for the schedules.

Official Form 106A/B, Schedule A/B: Property, consolidates information about an individual debtor's real and personal property into a single form. It replaces Official Form 6A, Real Property, and Official Form 6B, Personal Property, in cases of individual debtors. In addition to specific questions about the assets, the form also includes open text fields for providing additional information regarding particular assets when appropriate.

The layout and categories of property on Official Form 106A/B have changed. Instead of dividing property interests into

April 22-23, 2014 Page 123 of 596

two categories (real or personal property), the new form uses seven categories likely to be more familiar to non-lawyers: real estate, vehicles, personal household items, financial assets, business-related property, farm- and commercial fishing-related property, and a catch-all category for property that was not listed elsewhere in the form. The new form categories and the examples provided in many of the categories are designed to prompt debtors to be thorough and list all of their interests in property. The debtor may describe generally items of minimal value (such as children's clothes) by adding the value of the items and reporting the total.

Although a particular item of property may fit into more than one category, the instructions for the form explain that it should be listed only once.

In addition, because property that falls within a particular category may not be specifically elicited by the particular line items on the form, the debtor is asked in Parts 3–6 (lines 14, 35, 44, and 51) to specifically identify and value any other property in the category.

In Part 1, Describe Each Residence, Building, Land, or Other Real Estate You Own or Have an Interest In, the debtor is asked to state the "current value of the portion you own," and to also state whether ownership is shared with someone else. In addition, the debtor is asked for the nature of the ownership interest, if known by the debtor. Furthermore, instead of asking for an open-ended description of the property, the form guides the debtor in answering the description question by providing eight options from which to choose: single-family home, duplex or multi-unit building, condominium or cooperative, manufactured or mobile home, land, investment property, timeshare, and other.

Part 2, *Describe Your Vehicles*, also guides the debtor in answering the question, asking for the make, model, year, and mileage of the car or other vehicle. Because mileage is just a general indication of vehicle value, the debtor is not required to list the exact mileage, but instead is prompted to provide the approximate mileage.

April 22-23, 2014 Page 124 of 596

Part 3, *Describe Your Personal and Household Items*, simplifies wording, updates categories, and uses more common terms. For example, "Wearing apparel" is changed to "Clothes" and examples include furs, which were previously grouped with jewelry. Firearms, on the other hand, which were previously grouped with sports and other hobbies, are now set out as a separate category. Additionally, because a new Part 6 has been added to separately describe-farm related property, Part 3 includes a category for "Non-farm animals."

Part 4, *Describe Your Financial Assets*, prompts a listing of the debtor's financial assets through several questions providing separate space, after each listed type of account or deposit, for the institution or issuer name and the value of the debtor's interest in the asset. Two new categories of financial assets are added: "Bonds, mutual funds, or publicly traded stocks" and "Claims against third parties, whether or not you have filed a lawsuit or made a demand for payment."

Part 5, *Describe Any Business-Related Property You Own or Have an Interest In*, provides prompts for listing business-related property such as accounts receivable, inventory, and machinery, and includes a direction to list business-related real estate in Part 1, to avoid listing real estate twice.

Part 6, *Describe Any Farm- and Commercial Fishing-Related Property You Own or Have an Interest In*, provides prompts for listing farm- or commercial fishing-related property, such as farm animals, crops, and feed. It also includes a direction to list any farm- or commercial fishing-related real estate in Part 1.

Part 7, Describe All Property You Own or Have an Interest in That You Did Not List Above, is a catch-all provision that allows the debtor to report property that is difficult to categorize.

Part 8, *List the Totals of Each Part of this Form*, tabulates the total value of the debtor's interest in the listed property. The tabulation includes two subtotals, one for real estate, which corresponds to the real property total that was reported on former Official Form 6A. The second subtotal is of Parts 2-7, which

April 22-23, 2014 Page 125 of 596

corresponds to the personal property total that was reported on former Official Form 6B.

Official Form 106C, Schedule C: The Property You Claim as Exempt, replaces Official Form 6C, Property Claimed as Exempt, in cases of individual debtors.

Part 1, *Identify the Property You Claim as Exempt*, includes a table to list the property the debtor seeks to exempt, the value of the property owned by the debtor, the amount of the claimed exemption, and the law that allows the exemption. The first column asks for a brief description of the exempt property, and it also asks for the line number where the property is listed on Schedule A/B. The second column asks for the value of the portion of the asset owned by the debtor, rather than the entire asset. The third column asks for the amount, rather than the value, of the exemption claim.

The form has also been changed in light of the Supreme Court's ruling in *Schwab v. Reilly*, 130 S. Ct. 2652 (2010). Entries in the "amount of the exemption you claim" column may now be listed as either a dollar limited amount or as 100% of fair market value, up to any applicable statutory limit. For example, a debtor might claim 100% of fair market value for a home covered by an exemption capped at \$15,000, and that limit would be applicable. This choice would impose no dollar limit where the exemption is unlimited in dollar amount, such as some exemptions for health aids, certain governmental benefits, and tax-exempt retirement funds.

Official Form 106D, Schedule D: Creditors Who Hold Claims Secured by Property, replaces Official Form 6D, Creditors Holding Secured Claims, in cases of individual debtors.

Part 1, *List Your Secured Claims*, now directs the debtor to list only the last four digits of the account number. Part 1 also adds four checkboxes with which to describe the nature of the lien: an agreement the debtor made (such as mortgage or secured car loan); statutory lien (such as tax lien, mechanic's lien); judgment lien from a lawsuit; and other.

April 22-23, 2014 Page 126 of 596

The form adds Part 2, *List Others to Be Notified for a Debt That You Already Listed*. The debtor is instructed to use Part 2 if there is a need to notify someone about the bankruptcy filing other than the creditor for a debt listed in Part 1. For example, if a collection agency is trying to collect for a creditor listed in Part 1, the collection agency would be listed in Part 2.

Official Form 106E/F, Schedule E/F: Creditors Who Have Unsecured Claims, consolidates information about priority and nonpriority unsecured claims into a single form. It replaces Official Form 6E, Creditors Holding Unsecured Priority Claims, and Official Form 6F, Creditors Holding Unsecured Nonpriority Claims, in cases of individual debtors.

Although both priority and nonpriority unsecured claims are reported in Official Form 106E/F, the two types of claims are separately grouped so that the total for each type can be reported for case administration and statistical purposes. The form eliminates the question "consideration for claim" and instructs debtors to list claims in the alphabetical order of creditors as much as possible.

Part 1, *List All of Your PRIORITY Unsecured Claims*, includes four checkboxes for identifying the type of priority that applies to the claim: domestic support obligations; taxes and certain other debts owed to the government; claims for death or personal injury while intoxicated; and "other." The first three categories are required to be separately reported for statistical purposes. If the debtor selects "other," the debtor must specify the basis of the priority, *e.g.*, wages or employee benefit plan contribution.

Part 2, *List All of Your NONPRIORITY Unsecured Claims*, contains four checkboxes, including three for types of claims that must be separately reported for statistical purposes: student loans; obligations arising out of a separation agreement or divorce not listed as priority claims; and debts to pension or profit-sharing plans and other similar debts. The remaining "other" checkbox treats claims not subject to separate reporting. If the debtor selects "other," the debtor must specify the basis of the claim.

April 22-23, 2014 Page 127 of 596

Part 3, *List Others to Be Notified About a Debt That You Already Listed*, is new. The debtor is instructed to use Part 3 only if there is a need to give notice of the bankruptcy to someone other than a creditor listed in Parts 1 and 2. For example, if a collection agency is trying to collect for a creditor listed in Part 1, the collection agency would be listed in Part 3.

Finally, Part 4, *Add the Amounts for Each Type of Unsecured Claim*, requires the debtor to provide the total amounts of particular types of unsecured claims for statistical reporting purposes and the overall totals of the priority and nonpriority unsecured claims reported in this form.

Official Form 106G, Schedule G: Executory Contracts and Unexpired Leases, replaces Official Form 6G, Executory Contracts and Unexpired Leases, in cases of individual debtors.

The form is simplified. Instead of requiring the debtor to make multiple assertions about each potential executory contract or unexpired lease, the form simply requires the debtor to identify the name and address of the other party to the contract or lease, and to state what the contract or lease deals with. Definitions and examples of executory contracts and unexpired leases are included in the separate instructions for the form.

An additional page is provided in case the debtor has so many executory contracts and unexpired leases that the available page is not adequate. If the debtor needs to use the additional page, the debtor is required to fill in the entry number.

Official Form 106H, *Schedule H: Your Codebtors*, replaces Official Form 6H, *Codebtors*, in cases of individual debtors.

The form breaks out the questions about whether there are any codebters, and whether the debtor has lived with a spouse, former spouse or legal equivalent in a community property state in the prior eight years. It also removes Alaska from the listed community property states. Finally, it asks the debtor to indicate where the debt is listed on Schedule D, Schedule E/F or Schedule

April 22-23, 2014 Page 128 of 596

G, thereby eliminating the need to list the name and address of the creditor.

Official Form 106I, *Schedule I: Your Income*, replaces Official Form 6I, *Your Income*, in cases of individual debtors.

The form is one of an initial set of forms that were published as part of the Forms Modernization Project in 2012. It is renumbered and internal cross references are updated to conform to the new numbering system now being introduced by the Forms Modernization Project.

Official Form 106J, *Schedule J: Your Expenses*, replaces Official Form 6J, *Your Expenses*, in cases of individual debtors.

The form is one of an initial set of forms that were published as part of the Forms Modernization Project in 2012. It is renumbered and internal cross references are updated to conform to the new numbering system now being introduced by the Forms Modernization Project.

Official Form 106Dec, Declaration About an Individual Debtor's Schedules, replaces Official Form 6, Declaration Concerning Debtor's Schedules, in cases of individual debtors.

The form, which is to be signed by the debtor and filed with the debtor's schedules, deletes the Declaration and Signature of Bankruptcy Petition Preparer (BPP). Instead, the debtor is directed to complete and file Official Form 119, *Bankruptcy Petition Preparer's Notice, Declaration, and Signature*, if a BPP helped fill out the bankruptcy forms.

Because the form applies only to individual debtors, it no longer contains the Declaration Under Penalty of Perjury on Behalf of a Corporation or Partnership. It also deletes from the declaration the phrase "to the best of my knowledge, information, and belief" in order to conform to the language of 28 U.S.C. § 1746. *See* Rule 1008.

April 22-23, 2014 Page 129 of 596

April 22-23, 2014 Page 130 of 596

Draft March 14, 2014

Fill in this in	formation to identif	y your case:	
Debtor 1	First Name	Middle Name	Last Name
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name
United States E	Bankruptcy Court for the	:	District of(State)
Case number (If known)			

☐ Check if this is an amended filing

Official Form 107

Statement of Financial Affairs for Individuals Filing for Bankruptcy

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

rt 1: Give Details About Your M What is your current marital status?	arital Status and Where Yo	ou Lived Before	
☐ Married			
☐ Not married			
During the last 3 years, have you lived	anywhere other than where yo	ou live now?	
□ No□ Yes. List all of the places you lived in	the last 3 years. Do not include	where you live now.	
Debtor 1:	Dates Debtor 1 lived there	Debtor 2:	Dates Debtor 2 lived there
		☐ Same as Debtor 1	☐ Same as Debtor
Number Street	From	Number Street	From
	To		To
City State 2	IP Code	City State ZIP Code	
		☐ Same as Debtor 1	☐ Same as Debtor
	From		From
Number Street	To	Number Street	To
City State 2	ZIP Code	City State ZIP Code	
Within the last 8 years, did you ever livand territories include Arizona, California	ve with a spouse or legal equiv ı, Idaho, Louisiana, Nevada, Nev	ralent in a community property state or territory? (v v Mexico, Puerto Rico, Texas, Washington, and Wisc	Community property sta
□ No			,

Part 2: Explain the Sources of Your Income

Statement of Financial Affairs for Individuals Filing for Bankruptcy

page 1

irot Nomo			
	Middle Name	Lact Namo	

Did you have any income from employmen Fill in the total amount of income you received If you are filing a joint case and you have income	from all jobs and all busing	nesses, including part-tir		·
□ No□ Yes. Fill in the details.				
	Debtor 1		Debtor 2	
	Sources of income Check all that apply.	Gross income (before deductions and exclusions)	Sources of income Check all that apply.	Gross income (before deductions and exclusions)
From January 1 of current year until the date you filed for bankruptcy:	Wages, commissions, bonuses, tipsOperating a business	\$	☐ Wages, commissions, bonuses, tips☐ Operating a business	\$
For last calendar year:	Wages, commissions, bonuses, tips	\$	Wages, commissions, bonuses, tips	\$
(January 1 to December 31,)	Operating a business		Operating a business	
For the calendar year before that:	Wages, commissions, bonuses, tips	\$	Wages, commissions, bonuses, tips	\$
(January 1 to December 31,)	Operating a business	\$	Operating a business	Φ
Include income regardless of whether that include and other public benefit payments; pensions; winnings. If you are filing a joint case and you List each source and the gross income from e	ome is taxable. Examples rental income; interest; div have income that you rec	of other income are aliminated as the income	d from lawsuits; royalties; ar y once under Debtor 1.	
Did you receive any other income during the Include income regardless of whether that include and other public benefit payments; pensions; winnings. If you are filing a joint case and you blist each source and the gross income from the Impact of the Included Property of	ome is taxable. Examples rental income; interest; div have income that you rec	of other income are aliminated as the income	d from lawsuits; royalties; ar y once under Debtor 1.	
Include income regardless of whether that include and other public benefit payments; pensions; winnings. If you are filing a joint case and you List each source and the gross income from a No	ome is taxable. Examples rental income; interest; div have income that you rec	of other income are aliminated as the income	d from lawsuits; royalties; ar y once under Debtor 1.	
Include income regardless of whether that include income regardless of whether that included and other public benefit payments; pensions; winnings. If you are filing a joint case and you List each source and the gross income from e	ome is taxable. Examples rental income; interest; div have income that you receath source separately. Do	of other income are aliminated as the income	d from lawsuits; royalties; ar y once under Debtor 1. t you listed in line 4.	Gross income from each source
Include income regardless of whether that include income regardless of whether that included and other public benefit payments; pensions; winnings. If you are filing a joint case and you be a like the case and you are filing a joint case and you have a like a like and you have a like a li	ome is taxable. Examples rental income; interest; div have income that you receath source separately. Do Debtor 1 Sources of income	of other income are alimited of other income are alimited of the control of the c	d from lawsuits; royalties; are yonce under Debtor 1. It you listed in line 4. Debtor 2 Sources of income Describe below.	Gross income from each source (before deductions and exclusions)
Include income regardless of whether that include income regardless of whether that included and other public benefit payments; pensions; winnings. If you are filing a joint case and you have a source and the gross income from a No	ome is taxable. Examples rental income; interest; div have income that you receath source separately. Do Debtor 1 Sources of income	of other income are alimitidends; money collected eived together, list it only to not include income that to not include income that the control of the cont	d from lawsuits; royalties; are yonce under Debtor 1. It you listed in line 4. Debtor 2 Sources of income Describe below.	Gross income from each source (before deductions and
Include income regardless of whether that include income regardless of whether that included and other public benefit payments; pensions; winnings. If you are filing a joint case and you list each source and the gross income from a No Yes. Fill in the details. From January 1 of current year until	ome is taxable. Examples rental income; interest; div have income that you receath source separately. Do Debtor 1 Sources of income	of other income are alimitidends; money collected eived together, list it only to not include income that to not include income that the control of the cont	d from lawsuits; royalties; are yonce under Debtor 1. It you listed in line 4. Debtor 2 Sources of income Describe below.	Gross income from each source (before deductions and exclusions)
Include income regardless of whether that include income regardless of whether that included and other public benefit payments; pensions; winnings. If you are filing a joint case and you with List each source and the gross income from each of the No Yes. Fill in the details. From January 1 of current year until the date you filed for bankruptcy:	ome is taxable. Examples rental income; interest; div have income that you receath source separately. Do Debtor 1 Sources of income	of other income are alimitidends; money collected eived together, list it only to not include income that the not include income the not include income the not include income that the not include income that the not include income the n	d from lawsuits; royalties; are yonce under Debtor 1. It you listed in line 4. Debtor 2 Sources of income Describe below.	Gross income from each source (before deductions and exclusions)
Include income regardless of whether that include income regardless of whether that included and other public benefit payments; pensions; winnings. If you are filing a joint case and you list each source and the gross income from a No Yes. Fill in the details. From January 1 of current year until the date you filed for bankruptcy: For last calendar year:	ome is taxable. Examples rental income; interest; div have income that you receath source separately. Do Debtor 1 Sources of income	of other income are alimitidends; money collected eived together, list it only to not include income that the not include income the not include income that the not include income that the not include income the n	d from lawsuits; royalties; are yonce under Debtor 1. It you listed in line 4. Debtor 2 Sources of income Describe below.	Gross income from each source (before deductions and exclusions)
Include income regardless of whether that include and other public benefit payments; pensions; winnings. If you are filing a joint case and you List each source and the gross income from each of the proof of the p	ome is taxable. Examples rental income; interest; div have income that you receath source separately. Do Debtor 1 Sources of income	of other income are alimitidends; money collected eived together, list it only to not include income that the not inclu	d from lawsuits; royalties; ary once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income Describe below.	Gross income from each source (before deductions and exclusions)
Include income regardless of whether that include and other public benefit payments; pensions; winnings. If you are filing a joint case and you List each source and the gross income from each of the list of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list each source and the gross income from each of the list eac	ome is taxable. Examples rental income; interest; div have income that you receath source separately. Do Debtor 1 Sources of income	of other income are alimitidends; money collected eived together, list it only to not include income that the not include i	d from lawsuits; royalties; ar y once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income Describe below.	Gross income from each source (before deductions and exclusions) \$
Include income regardless of whether that include and other public benefit payments; pensions; winnings. If you are filing a joint case and you List each source and the gross income from a No Yes. Fill in the details. From January 1 of current year until the date you filed for bankruptcy: For last calendar year: (January 1 to December 31,)	ome is taxable. Examples rental income; interest; div have income that you receath source separately. Do Debtor 1 Sources of income	of other income are alimitidends; money collected eived together, list it only to not include income that the not include	d from lawsuits; royalties; ar y once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income Describe below.	Gross income from each source (before deductions and exclusions) \$

April 22-23, 2014 Page 132 of 596

Statement of Financial Affairs for Individuals Filing for Bankruptcy

Eiret Namo	Middle Name	Last Namo

Creditor's Name S S Mortgage Car Credit card Loan repayment Suppliers or vende City State ZIP Code S S Mortgage Creditor's Name City State ZIP Code S S Mortgage Creditor's Name City State ZIP Code S S Mortgage Credit card Loan repayment Suppliers or vende City State ZIP Code City State ZIP Code S S Mortgage Creditor's Name Creditor's Name S S Mortgage Credit card Loan repayment Suppliers or vende City State ZIP Code S S Mortgage Creditor's Name Creditor's Name S S Mortgage Credit card Loan repayment Suppliers or vende Code Credit card Code Cre	Part 3:	List	Certain Paym	ents You	Made Befor	e You Filed	for Bankruptcy		
No. Neither Debtor 1 nor Debtor 2 has primarily consumer debts. Consumer debts are defined in 11 U.S.C. § 101(8) as 'incurred by an individual primarily for a personal, family, or household purpose.' During the 90 days before you filled for bankruptcy, did you pay any creditor a total of \$6,225° or more in one or more payments and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments to an attorney for this bankruptcy case. Subject to adjustment on 4/01/16 and every 3 years after that for cases filed on or after the date of adjustment. Yes. Debtor 1 or Debtor 2 or both have primarily consumer debts. During the 90 days before you filed for bankruptcy, did you pay any creditor a total of \$600 or more? No. Go to line 7. Yes. List below each creditor to whom you paid a total of \$600 or more and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments to an attorney for this bankruptcy case. Dates of payment Total amount paid Amount you still owe Mas this payment to creditor. Name Size Mortgage Car Car Creditor's Name Size Mortgage Creditor's Name Size Mortgage Credit card Car Car									
"incurred by an individual primarily for a personal, family, or household purpose." During the 90 days before you filed for bankruptcy, did you pay any creditor a total of \$6,225' or more? No. Go to line 7. Yes. List below each creditor to whom you paid a total of \$6,225' or more in one or more payments and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and allamony. Also, do not include payments to an attorney for this bankruptcy case. * Subject to adjustment on 40/1/6 and every 3 years after that for cases flied on or after the date of adjustment. Yes. Debtor 1 or Debtor 2 or both have primarily consumer debts. During the 90 days before you filed for bankruptcy, did you pay any creditor a total of \$600 or more? No. Go to line 7. Yes. List below each creditor to whom you paid a total of \$600 or more and the total amount you paid that creditor. Do not include payments for atomestic support obligations, such as child support and alimony. Also, do not include payments to an attorney for this bankruptcy case. Dates of payment Dates of payment Total amount paid Amount you still owe Was this payment for payment.	6. Are eitl	her De	ebtor 1's or Debt	or 2's debt	s primarily co	nsumer debt	s?		
No. Go to line 7. Yes. List below each creditor to whom you paid a total of \$6,225° or more in one or more payments and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and allimony. Also, do not include payments for anothers for this barkruptcy case. * Subject to adjustment on 40/176 and every 3 years after that for cases filed on or after the date of adjustment. Yes. Debtor 1 or Debtor 2 or both have primarily consumer debts. During the 90 days before you filed for bankruptcy, did you pay any creditor a total of \$600 or more? No. Go to line 7. Yes. List below each creditor to whom you paid a total of \$600 or more and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments for domestic support obligations, such as child support and the total amount you paid that credit and under the foliation of the payment for paymen	☐ No							e defined in 11 U.S.C. § 101	(8) as
Yes. List below each creditor to whom you paid a total of \$6,225' or more in one or more payments and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and allmony. Also, do not include payments for a attorney for this bankruptcy case. * Subject to adjustment on 4/01/16 and every 3 years after that for cases filed on or after the date of adjustment. Yes. Debtor 1 or Debtor 2 or both have primarily consumer debts. During the 90 days before you filed for bankruptcy, did you pay any creditor a total of \$600 or more? No. Go to line 7. Yes. List below each creditor to whom you paid a total of \$600 or more and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments to an attorney for this bankruptcy case. Dates of Total amount paid		Duri	ng the 90 days be	efore you fil	ed for bankrup	tcy, did you pa	ay any creditor a total of	\$6,225* or more?	
total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and allmony. Also, do not include payments or an attorney for this bankruptor cases. * Subject to adjustment on 4/01/16 and every 3 years after that for cases filed on or after the date of adjustment. Yes. Debtor 1 or Debtor 2 or both have primarily consumer debts. During the 90 days before you filed for bankruptcy, did you pay any creditor a total of \$600 or more? No. Go to line 7. Yes. List below each creditor to whom you paid a total of \$600 or more and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments to an attorney for this bankruptcy case. Dates of payment Total amount paid Amount you still owe Was this payment for payment.			No. Go to line 7.						
* Subject to adjustment on 4/01/16 and every 3 years after that for cases filed on or after the date of adjustment. Yes. Debtor 1 or Debtor 2 or both have primarily consumer debts. During the 90 days before you filed for bankruptcy, did you pay any creditor a total of \$600 or more? No. Go to line 7. Yes. List below each creditor to whorn you paid a total of \$600 or more and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments to an attorney for this bankruptcy case. Dates of payment Total amount paid			total amount	t you paid th	at creditor. Do	not include pa	ayments for domestic su	pport obligations, such as	
Yes. Debtor 1 or Debtor 2 or both have primarily consumer debts. During the 90 days before you filed for bankruptcy, did you pay any creditor a total of \$600 or more? No. Go to line 7. Yes. List below each creditor to whom you paid a total of \$600 or more and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments to an attorney for this bankruptcy case. Dates of payment Total amount paid Amount you still owe Was this payment for payment Total amount paid Amount you still owe Was this payment for payment Street Greditor's Name Mortgage Car Credit card Loan repayment Suppliers or vender Car Car Creditor's Name Mortgage Car Creditor's Name Creditor's Name Street Car Creditor's Name Car Creditor's Name Street Car Creditor's Name Car Car Creditor's Name Car Creditor's Name Car Car Creditor's Name Car Car Credit card Car Car Credit card Car		* Su							
During the 90 days before you filed for bankruptcy, did you pay any creditor a total of \$600 or more? No. Go to line 7. Yes. List below each creditor to whom you paid a total of \$600 or more and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and allimony. Also, do not include payments to an attorney for this bankruptcy case. Dates of payment Total amount paid Amount you still owe Was this payment for payment. Suppliers or vendule and the total amount you paid that creditor's Name Street Number Street	□ v _o ,				-			,	
No. Go to line 7. Yes. List below each creditor to whom you paid a total of \$600 or more and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments to an attorney for this bankruptcy case. Dates of payment Total amount paid Amount you still owe Was this payment for payment	■ Yes							\$600 or more?	
Yes. List below each creditor to whom you paid a total of \$600 or more and the total amount you paid that creditor. Do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments to an attorney for this bankruptcy case. Dates of payment Total amount paid Amount you still owe Was this payment for payment Total amount paid Amount you still owe Was this payment for payment Total amount paid Amount you still owe Was this payment for payment Total amount paid Amount you still owe Was this payment for payment Car Creditors Name S Mortgage Car Credit card Loan repayment Suppliers or vended Other Car Credit card Loan repayment Suppliers or vended Other Credit card Car Car Credit card Car Car Credit card Car				cioic you iii	ca for bankrup	noy, ala you pe	ay arry creditor a total of	φοσο οι more:	
creditor. Do not include payments for domestic support obligations, such as child support and alimony. Also, do not include payments to an attorney for this bankruptcy case. Dates of payment		U 1	No. Go to line 7.						
Creditor's Name S		,	creditor. Do	not include	payments for d	domestic supp	ort obligations, such as	child support and	
Creditor's Name Car Credit card Loan repayment Suppliers or vends Credit card Credit card Credit card Condition Credit card Condition Credit card Credit card Credit card Credit card Credit card Car Credit card Can repayment Suppliers or vends City State ZIP Code Same Car Credit card Condition Credit card Condition Credit card Car Car Credit card Car Credit card Car Credit card Car C							Total amount paid	Amount you still owe	Was this payment for
Creditor's Name Car Credit card Loan repayment Suppliers or vends Credit card Credit card Credit card Condition Credit card Condition Credit card Credit card Credit card Credit card Credit card Car Credit card Can repayment Suppliers or vends City State ZIP Code Same Car Credit card Condition Credit card Condition Credit card Car Car Credit card Car Credit card Car Credit card Car C							\$	\$	☐ Martana
Number Street Credit card Loan repayment Suppliers or vends Other			Creditor's Name				Ψ		
Loan repayment Suppliers or vends City State ZIP Code S Mortgage Car Creditor's Name Suppliers or vends City State ZIP Code S Mortgage Car Credit card Loan repayment Suppliers or vends City State ZIP Code S Mortgage Car Credit card Can repayment Can repa									
City State ZIP Code \$ \$ \$ Mortgage Creditor's Name Number Street City State ZIP Code \$ \$ Mortgage Credit card Loan repayment Suppliers or vende City State ZIP Code \$ \$ \$ Mortgage Credit card Loan repayment City State ZIP Code \$ \$ \$ Mortgage Credit card Loan repayment Suppliers or vende Car Credit card Loan repayment Suppliers or vende Car Credit card Loan repayment Suppliers or vende			Number Street						
City State ZIP Code \$ \$ \$ Mortgage Creditor's Name Number Street City State ZIP Code \$ \$ Mortgage Creditor's Name Suppliers or vende Creditor's Name Creditor's Name Suppliers or vende Creditor's Name Suppliers or vende Creditor's Name Creditor's Name Suppliers or vende Creditor's Name Other									
Creditor's Name Number Street Mortgage Car Credit card Loan repayment Suppliers or vender Car Other Car Credit card Car Credit card Car Credit card Car									Other
Creditor's Name Car Credit card Loan repayment Suppliers or vende Creditor's Name Creditor's Name Creditor's Name Suppliers or vende Creditor's Name Creditor's Name Suppliers or vende Car			City	State	ZIP Code				
Creditor's Name Car Credit card Loan repayment Suppliers or vende Creditor's Name Creditor's Name Creditor's Name Suppliers or vende Car Creditor's Name Suppliers or vende Car Car							¢	¢	
Number Street Credit card Loan repayment Suppliers or vends Other			Creditor's Name				Φ	Φ	
City State ZIP Code Suppliers or vender Creditor's Name Creditor's Name Number Street Suppliers or vender Creditor's Name Suppliers or vender Car Credit card Loan repayment Suppliers or vender Suppliers or vender Other									
City State ZIP Code \$ \$ \$ Mortgage Creditor's Name Number Street Code Car Credit card Code Car Code Code Code Car Code Car Code Code Car Code Code Car Code			Number Street						
City State ZIP Code \$\$									
City State ZIP Code State ZIP Code Mortgage Creditor's Name Car Credit card Loan repayment Suppliers or vende									* *
Creditor's Name Car Number Street Loan repayment Suppliers or vender Other			City	State	ZIP Code				Uther
Creditor's Name Car Number Street Loan repayment Suppliers or vender Other		-							
Creditor's Name Car Number Street Loan repayment Suppliers or vendor Other							\$	\$	☐ Mortgage
Number Street Loan repayment Suppliers or vendo			Creditor's Name						
Loan repayment Suppliers or vendo Other									
Suppliers or vendo			Number Street						
Other									
City State 7IP Code									
5.9			City	State	ZIP Code				_ 0

Official Form 107

City

Insider's Name

Number Street

State

ZIP Code

First Name

Middle Name

Last Name

thin 1 year before you filed fo at all such matters, including per d contract disputes.				-	_
No Yes. Fill in the details.					
	Natur	e of the case	Court or agency		Status of the case
Case title			Court Name		Pending On appeal
Case number			Number Street		Concluded
			City State	e ZIP Code	
Case title			Court Name		Pending On appeal
Case number			Number Street		Concluded
			City State	e ZIP Code	
No. Go to line 11. Yes. Fill in the information belo	details below.			nished, attache	, ,
		Describe the propert	у	Date	
		Describe the propert	у		
Yes. Fill in the information belo		Describe the propert			Value of the property
Yes. Fill in the information belonger of the information b		_	ned		Value of the property
Yes. Fill in the information belonger of the information b		Explain what happer Property was r Property was f	epossessed. oreclosed.		Value of the property
Yes. Fill in the information belonger Creditor's Name Number Street	ow.	Explain what happer Property was f Property was f Property was g	epossessed. oreclosed. garnished.		Value of the property
Yes. Fill in the information belonger of the information b		Explain what happer Property was r Property was f Property was g Property was a	epossessed. oreclosed. garnished. attached, seized, or levied.		Value of the property
Yes. Fill in the information belonger Creditor's Name Number Street	ow.	Explain what happer Property was f Property was f Property was g	epossessed. oreclosed. garnished. attached, seized, or levied.	Date	Value of the property
Yes. Fill in the information belonger Creditor's Name Number Street	ow.	Explain what happer Property was r Property was f Property was g Property was a	epossessed. oreclosed. garnished. attached, seized, or levied.	Date	Value of the property
Yes. Fill in the information belonger Creditor's Name Number Street	ow.	Explain what happer Property was r Property was f Property was g Property was a	epossessed. oreclosed. garnished. attached, seized, or levied.	Date	Value of the property \$ Value of the proper
Yes. Fill in the information belonger Creditor's Name Number Street	ow.	Explain what happer Property was r Property was f Property was g Property was a	epossessed. oreclosed. garnished. attached, seized, or levied.	Date	Value of the property \$ Value of the proper
Creditor's Name Number Street City Creditor's Name	ow.	Explain what happer Property was r Property was f Property was a Property was a Describe the propert	epossessed. oreclosed. garnished. attached, seized, or levied. y	Date	Value of the property \$ Value of the proper
Creditor's Name Number Street City Creditor's Name	ow.	Explain what happer Property was r Property was g Property was a Property was a Describe the propert Explain what happer Property was r Property was r	epossessed. oreclosed. garnished. attached, seized, or levied. y ned epossessed. oreclosed.	Date	Value of the property \$ Value of the proper
Creditor's Name Number Street City Creditor's Name	ow.	Explain what happer Property was f Property was g Property was a Property was a Describe the propert Explain what happer Property was f Property was f Property was f	epossessed. oreclosed. garnished. attached, seized, or levied. y ned epossessed. oreclosed.	Date	Value of the property \$ Value of the proper

Official Form 107

Person's relationship to you _

Statement of Financial Affairs for Individuals Filing for Bankruptcy

page 6

April 22-23, 2014 Page 136 of 596

FIR	rst Name Middle Name	Last Na		Case number (if known)		
	st Name i wildlie Name	Lastina	ne			
ithin 2 ye	ears before you filed	for bankruptc	y, did you give any gifts or co	ntributions with a total value	of more than \$600	to any charity?
□ No						
■ Yes. Fi	ill in the details for eac	ch gift or contrib	oution.			
	or contributions to char	rities	Describe what you contributed		Date you	Value
that to	otal more than \$600				contributed	
Charity's	Name					\$
, .						
						\$
Number	Street					
City	State ZIP Code					
i 6: L	ist Certain Losses	S				
	ribe the property you los ss occurred	st and how	Describe any insurance coverage Include the amount that insurance	has paid. List pending insurance	Date of your loss	Value of property lost
			claims on line 33 of Schedule A/B	: Property.		
						\$
						\$
						\$
: 7: Lis	st Certain Paymen	nts or Transf	ers			\$
Vithin 1 y	ear before you filed f	for bankruptcy	, did you or anyone else actin	ng on your behalf pay or trans	sfer any property to	·
Vithin 1 yo	ear before you filed f I about seeking bank	for bankruptcy	, did you or anyone else actinoring a bankruptcy petition?			·
Vithin 1 your onsulted nolude an	ear before you filed f I about seeking bank	for bankruptcy	, did you or anyone else actin			·
Within 1 yes consulted nolude and	rear before you filed f I about seeking bank by attorneys, bankrupto	for bankruptcy	, did you or anyone else actinoring a bankruptcy petition?			·
Within 1 yes consulted nolude and	ear before you filed f I about seeking bank	for bankruptcy	y, did you or anyone else actin paring a bankruptcy petition? arers, or credit counseling agen	cies for services required in yo	our bankruptcy.	anyone you
Vithin 1 yesonsulted nolude and	rear before you filed f I about seeking bank by attorneys, bankrupto	for bankruptcy	, did you or anyone else actinoring a bankruptcy petition?	cies for services required in yo		·
Vithin 1 yo consulted nclude and ☐ No ☐ Yes. Fi	rear before you filed f I about seeking bank by attorneys, bankrupto	for bankruptcy	y, did you or anyone else actin paring a bankruptcy petition? arers, or credit counseling agen	cies for services required in yo	our bankruptcy. Date payment or	anyone you
Vithin 1 yesonsulted and No Yes. Fi	rear before you filed for about seeking bank by attorneys, bankruptonill in the details.	for bankruptcy	y, did you or anyone else actin paring a bankruptcy petition? arers, or credit counseling agen	cies for services required in yo	our bankruptcy. Date payment or	anyone you
Vithin 1 yesonsulted nclude and No ☐ Yes. Fi	rear before you filed for about seeking bank by attorneys, bankruptonill in the details.	for bankruptcy	y, did you or anyone else actin paring a bankruptcy petition? arers, or credit counseling agen	cies for services required in yo	our bankruptcy. Date payment or	anyone you
Within 1 yesonsulted notude and No Yes. Fi	rear before you filed for about seeking bank by attorneys, bankruptonill in the details.	for bankruptcy	y, did you or anyone else actin paring a bankruptcy petition? arers, or credit counseling agen	cies for services required in yo	our bankruptcy. Date payment or	anyone you
Nithin 1 yes consulted include and No Yes. Fi	rear before you filed for about seeking bank by attorneys, bankruptonill in the details. Who Was Paid	for bankruptcy kruptcy or prepa	y, did you or anyone else actin paring a bankruptcy petition? arers, or credit counseling agen	cies for services required in yo	our bankruptcy. Date payment or	Amount of payme
Within 1 yesonsulted notude and No Yes. Fi	rear before you filed for about seeking bank by attorneys, bankruptonill in the details.	for bankruptcy	y, did you or anyone else actin paring a bankruptcy petition? arers, or credit counseling agen	cies for services required in yo	our bankruptcy. Date payment or	Amount of payme
No Yes. Fi	rear before you filed for about seeking bank by attorneys, bankruptonill in the details. Who Was Paid	for bankruptcy kruptcy or prepa	y, did you or anyone else actin paring a bankruptcy petition? arers, or credit counseling agen	cies for services required in yo	our bankruptcy. Date payment or	Amount of payme
Nithin 1 yes consulted nelude and No Yes. Fi	rear before you filed for about seeking bank by attorneys, bankruptonill in the details. Who Was Paid State	for bankruptcy kruptcy or prep cy petition prepa ZIP Code	y, did you or anyone else actin paring a bankruptcy petition? arers, or credit counseling agen	cies for services required in yo	our bankruptcy. Date payment or	Amount of payme

April 22-23, 2014 Page 137 of 596

Statement of Financial Affairs for Individuals Filing for Bankruptcy

page 7

Official Form 107

Number

City

Street

Person's relationship to you _

Statement of Financial Affairs for Individuals Filing for Bankruptcy

page 8

ZIP Code

Middle Name

Last Name

Case number (if kn	nown)
--------------------	-------

	nin 10 years before you filed for bankrupt a beneficiary? (These are often called ass		to a self-settled trust o	or similar device of wh	ich you
<u> </u>	No Yes. Fill in the details.				
		Description and value of the proper	ty transferred		Date transfer was made
-	Name of trust				
Part 8	List Certain Financial Accounts	Instruments, Safe Deposit	Boxes, and Storage	e Units	
clos Inclu brok	nin 1 year before you filed for bankruptcy sed, sold, moved, or transferred? ude checking, savings, money market, o kerage houses, pension funds, cooperati No Yes. Fill in the details.	r other financial accounts; certifi	cates of deposit; share	-	
		Last 4 digits of account number	Type of account or instrument	Date account was closed, sold, moved, or transferred	Last balance before closing or transfer
	Name of Financial Institution Number Street	xxxx	☐ Checking ☐ Savings ☐ Money market ☐ Brokerage		\$
_	City State ZIP Code		Other		
	Name of Financial Institution	xxxx	☐ Checking ☐ Savings		\$
	Number Street		☐ Money market ☐ Brokerage ☐ Other		
seci	City State ZIP Code you now have, or did you have within 1 yourities, cash, or other valuables? No Yes. Fill in the details.	ear before you filed for bankrupt		x or other depository t	ior
		Who else had access to it?	Describe the	contents	Do you still have it?
	Name of Financial Institution	Name			☐ No☐ Yes
	Number Street	Number Street			
	City State ZIP Code	City State ZIP Code			

Official Form 107

Statement of Financial Affairs for Individuals Filing for Bankruptcy

page 9

April 22-23, 2014 Page 139 of 596

First Name	e Middle Name	Last Name	Case number (if known)	
Have you stored ☑ No	d property in a storage	e unit or place other than your home withi	n 1 year before you filed for bankruptcy	?
Yes. Fill in t	he details.			
		Who else has or had access to it?	Describe the contents	Do you st have it?
				□ No
Name of Stor	rage Facility	Name		Yes
Number St	root	Number Street		
Number 30	1661	Number Sueet		
		City State ZIP Code		
City	State ZIP (Code		
rt 9: Ider	ntify Property You	Hold or Control for Someone Else		
-		that someone else owns? Include any pro	pperty you borrowed from, are storing fo	or,
or hold in trust	for someone.			
Yes. Fill in	the details.			
		Where is the property?	Describe the property	Value
				\$
Owner's Nan				Ψ
Owner's Nan		Number Street		
Owner's Nan		Number Street		
	reet		Code	
Number St	State ZIP (City State ZIP	Code	
Number St City Give	State ZIP (Code City State ZIP	Code	
Number St City Tt 10: Give	State ZIP of Details About En	Code City State ZIP		ses of
Number St City Tt 10: Give T the purpose of Environmental hazardous or t	State ZIP of Part 10, the following I law means any feder toxic substances, was	City State ZIP Invironmental Information Ing definitions apply: In rail, state, or local statute or regulation constes, or material into the air, land, soil, sur	cerning pollution, contamination, releas	
City Tt 10: Give the purpose of Environmental hazardous or to including statu	State ZIP of the property of Part 10, the following law means any feder toxic substances, was utes or regulations co	City State ZIP avironmental Information ng definitions apply: ral, state, or local statute or regulation constes, or material into the air, land, soil, surportrolling the cleanup of these substances	cerning pollution, contamination, releas face water, groundwater, or other medit , wastes, or material.	ım,
City Tt 10: Give The purpose of Environmental hazardous or to including status. Site means any	State ZIP of the Details About En of Part 10, the following I law means any feder toxic substances, was utes or regulations coy location, facility, or	City State ZIP Invironmental Information Ing definitions apply: In rail, state, or local statute or regulation constes, or material into the air, land, soil, sur	cerning pollution, contamination, releas face water, groundwater, or other medit , wastes, or material.	ım,
City Tt 10: Give The purpose of Environmental hazardous or to including status Site means any it or used to out the purpose of the purpose	State ZIP of Part 10, the following I law means any feder toxic substances, was utes or regulations con y location, facility, or won, operate, or utilize the state of the sta	City State ZIP nvironmental Information ng definitions apply: ral, state, or local statute or regulation constes, or material into the air, land, soil, sure ontrolling the cleanup of these substances property as defined under any environment it, including disposal sites. g an environmental law defines as a hazard	cerning pollution, contamination, releas face water, groundwater, or other medit , wastes, or material. ntal law, whether you now own, operate,	um, or utilize
City Tt 10: Give T the purpose of Environmental hazardous or tincluding statu Site means any it or used to out the substance, hazardous masubstance, hazardou	State ZIP of Part 10, the following I law means any feder toxic substances, was utes or regulations concept location, facility, or lawn, operate, or utilize the standard means anything the standard	City State ZIP avironmental Information and definitions apply: ral, state, or local statute or regulation constes, or material into the air, land, soil, surportrolling the cleanup of these substances property as defined under any environment it, including disposal sites. g an environmental law defines as a hazard lutant, contaminant, or similar term.	cerning pollution, contamination, releas face water, groundwater, or other medit , wastes, or material. ntal law, whether you now own, operate, dous waste, hazardous substance, toxic	um, or utilize
City The purpose of Environmental hazardous or tincluding status Site means any it or used to out the Hazardous masubstance, hazardous m	State ZIP of Part 10, the following I law means any feder toxic substances, was utes or regulations concept location, facility, or lawn, operate, or utilize the standard means anything the standard	City State ZIP nvironmental Information ng definitions apply: ral, state, or local statute or regulation constes, or material into the air, land, soil, sure ontrolling the cleanup of these substances property as defined under any environment it, including disposal sites. g an environmental law defines as a hazard	cerning pollution, contamination, releas face water, groundwater, or other medit , wastes, or material. ntal law, whether you now own, operate, dous waste, hazardous substance, toxic	um, or utilize
City Trit 10: Give	State ZIP of Part 10, the following I law means any feder toxic substances, was attes or regulations concept location, facility, or lawn, operate, or utilize the state of the	City State ZIP avironmental Information and definitions apply: ral, state, or local statute or regulation constes, or material into the air, land, soil, surportrolling the cleanup of these substances property as defined under any environment it, including disposal sites. g an environmental law defines as a hazard lutant, contaminant, or similar term.	cerning pollution, contamination, releas face water, groundwater, or other medic , wastes, or material. htal law, whether you now own, operate, dous waste, hazardous substance, toxic	um, or utilize
City The purpose of Environmental hazardous or tincluding status of the control	State ZIP of Part 10, the following I law means any feder toxic substances, was attes or regulations concept location, facility, or lawn, operate, or utilize the state of the	City State ZIP Evironmental Information Ing definitions apply: In training the cleanup of these substances of the cleanup of these substances of the cleanup of these substances of the cleanup of the	cerning pollution, contamination, releas face water, groundwater, or other medic , wastes, or material. htal law, whether you now own, operate, dous waste, hazardous substance, toxic	um, or utilize
City The purpose of Environmental hazardous or tincluding status or used to out the substance, hazardous masubstance, hazardous masubsta	State ZIP of Part 10, the following of Part	City State ZIP Evironmental Information Ing definitions apply: In training the cleanup of these substances of the cleanup of these substances of the cleanup of these substances of the cleanup of the	cerning pollution, contamination, releas face water, groundwater, or other medic , wastes, or material. htal law, whether you now own, operate, dous waste, hazardous substance, toxic	um, or utilize
City The purpose of Environmental hazardous or tincluding status. Site means any it or used to out the purpose of the purpose	State ZIP of Part 10, the following of Part	City State ZIP Evironmental Information Ing definitions apply: In training the cleanup of these substances of the cleanup of these substances of the cleanup of these substances of the cleanup of the	cerning pollution, contamination, releas face water, groundwater, or other medic , wastes, or material. htal law, whether you now own, operate, dous waste, hazardous substance, toxic	um, or utilize
City The purpose of Environmental hazardous or tincluding status. Site means any it or used to out the purpose of the purpose	State ZIP of Part 10, the following of Part	City State ZIP avironmental Information and definitions apply: ral, state, or local statute or regulation constes, or material into the air, land, soil, surportrolling the cleanup of these substances property as defined under any environment it, including disposal sites. g an environmental law defines as a hazard lutant, contaminant, or similar term. redings that you know about, regardless of you that you may be liable or potentially liable.	cerning pollution, contamination, release face water, groundwater, or other medic, wastes, or material. Intal law, whether you now own, operate, dous waste, hazardous substance, toxic when they occurred.	um, , or utilize : nental law?
City The purpose of Environmental hazardous or tincluding status. Site means any it or used to out the purpose of the purpose	State ZIP of Part 10, the following of Part	City State ZIP avironmental Information and definitions apply: ral, state, or local statute or regulation constes, or material into the air, land, soil, surportrolling the cleanup of these substances property as defined under any environment it, including disposal sites. g an environmental law defines as a hazard lutant, contaminant, or similar term. redings that you know about, regardless of you that you may be liable or potentially liable.	cerning pollution, contamination, release face water, groundwater, or other medic, wastes, or material. Intal law, whether you now own, operate, dous waste, hazardous substance, toxic when they occurred.	um, , or utilize : nental law?
Rumber Standard Number Standar	State ZIP of the Details About En of Part 10, the following I law means any feder toxic substances, was utes or regulations concept location, facility, or lawn, operate, or utilized the means anything the zardous material, pollowing the concept location, releases, and proceed the mental unit notified the details.	City State ZIP Invironmental Information Ing definitions apply: Iral, state, or local statute or regulation constes, or material into the air, land, soil, sure portrolling the cleanup of these substances property as defined under any environment it, including disposal sites. Ing an environmental law defines as a hazard lutant, contaminant, or similar term. Intelligent the contaminant is defined to potentially liable or potentially	cerning pollution, contamination, release face water, groundwater, or other medic, wastes, or material. Intal law, whether you now own, operate, dous waste, hazardous substance, toxic when they occurred.	um, , or utilize : nental law?
Number Standard Number	State ZIP of the Details About En of Part 10, the following I law means any feder toxic substances, was utes or regulations concept location, facility, or lawn, operate, or utilized the means anything the zardous material, pollowing the concept location, releases, and proceed the mental unit notified the details.	Code City State ZIP Code City State City Comparison Compar	cerning pollution, contamination, release face water, groundwater, or other medic, wastes, or material. Intal law, whether you now own, operate, dous waste, hazardous substance, toxic when they occurred.	um, , or utilize : nental law?

April 22-23, 2014 Page 140 of 596

Statement of Financial Affairs for Individuals Filing for Bankruptcy

page 10

Have you notified any governmental unit o □ No □ Yes. Fill in the details.	of any release of hazardous mater	ial?	
Tes. Fill III the details.			
	Governmental unit	Environmental law, if you know it	Date of notice
	_		
Name of site	Governmental unit		
Number Street	Number Street		
	City State ZIP Code	_	
City State ZIP Code	_		
lave you been a party in any judicial or a	dministrative proceeding under an	ny environmental law? Include settlement	s and orders.
☐ No ☐ Yes. Fill in the details.			
	Court or agency	Nature of the case	Status of the case
Case title	_		☐ Pending
	Court Name		On appeal
	Number Street		Concluded
Case number	City State ZIP C	ode	
rt 11: Give Details About Your B	usiness or Connections to An	ny Business	
		ave any of the following connections to a	any business?
□ A sole proprietor or self-employed□ A member of a limited liability con			
☐ A partner in a partnership			
An officer, director, or managing	-		
An owner of at least 5% of the vot		ration	
No. None of the above applies. Go toYes. Check all that apply above and fi		siness.	
	Describe the nature of the busine		n number
Business Name	-	Do not include Social	Security number or ITIN.
	_	EIN:	
Number Street	Name of accountant or bookkeep	Dates business exister	d
	_	From To	o
	-		
City State ZIP Code	Describe the nature of the busine	ess Employer Identification	n number
City State ZIP Code Business Name	Describe the nature of the busine		n number Security number or ITIN.
	Describe the nature of the busine	Do not include Social	

Official Form 107

City

State

ZIP Code

From _____ To ____

First Name Middle Na	me Last Na	nme	Case number (if known)
		Describe the nature of the business	Employer Identification number Do not include Social Security number or ITIN
Business Name			EIN:
Number Street		Name of accountant or bookkeeper	Dates business existed
			From To
City St	ate ZIP Code		
ithin 2 years before you fi	led for bankrupto	cy, did you give a financial statement to	o anyone about your business? Include all financial
stitutions, creditors, or ot			•
No Yes. Fill in the details be	elow.		
		Date issued	
Name		MM / DD / YYYY	
Number Street			
City St.	ate ZIP Code		
12: Sign Below			
			nts, and I declare under penalty of perjury that the aling property, or obtaining money or property by frauc
	ruptcy case can i	result in fines up to \$250,000, or impris	
		*	
×		•	

☐ Yes

Did you attach additional pages to Your Statement of Financial Affairs for Individuals Filing for Bankruptcy (Official Form 107)?

Did you pay or agree to pay someone who is not an attorney to help you fill out bankruptcy forms?

☐ No ☐ Yes. Name of person_

. Attach the Bankruptcy Petition Preparer's Notice, Declaration, and Signature (Official Form 119).

Page 142 of 596 April 22-23, 2014

Statement of Financial Affairs for Individuals Filing for Bankruptcy

Official Form 107

■ No

COMMITTEE NOTE

Official Form 107, Statement of Financial Affairs for Individuals Filing for Bankruptcy, which applies only in cases of individual debtors, is revised in its entirety as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats. Therefore, many of the open-ended questions and multiple-part instructions have been replaced with more specific questions. In addition, the form is renumbered to distinguish it from the version to be used in non-individual cases, and stylistic changes were made throughout the form.

The form is derived from former Official Form 7, *Statement of Financial Affairs*. The new form uses eleven sections likely to be more understandable to non-lawyers, groups questions of a similar nature together, and eliminates questions unrelated to individual debtors. The new form deletes the instruction, previously found in many questions, that married debtors filing under chapter 12 or chapter 13 must include information applicable to their spouse, even if their spouse is not filing with them, unless the spouses are separated. This change was made because a non-filing spouse's general financial affairs are not relevant to the debtor's bankruptcy case.

Part 1, Give Details About Where You Lived Before, moves the questions regarding the debtor's prior addresses, as well as residences in a community property state, to the beginning of the form. The form eliminates the "name used" question in reference to prior addresses. Also, the debtor is no longer required to list the name of a spouse or former spouse who lived with the debtor in a community property state since that information will be provided in Official Form 106F.

Part 2, *Explain the Sources of Your Income*, consolidates the questions regarding income, adding "wages, commissions, bonuses, tips" as a category for sources of income, and it

April 22-23, 2014 Page 143 of 596

eliminates the option to report income on a fiscal year basis. In addition, the form provides examples of types of "other income." The time period is clarified to indicate that the prior two years means two calendar years, plus the portion of the calendar year in which the bankruptcy is filed.

Part 3, List Certain Payments You Made Before You Filed for Bankruptcy, includes questions related to payments made in the 90 days prior to bankruptcy, with a separate question for payments made to insiders within one year before filing for bankruptcy. The statutory definition of consumer debt is provided. The question regarding the nature of the debtor's debts requires the debtor to use checkboxes to indicate whether or not they are primarily consumer debts. The form instructs debtors not to include payments for domestic support obligations in the section regarding insider payments. The form provides a separate question regarding payments or transfers on account of a debt that benefited an insider. For both questions regarding payments to insiders, the debtor is required to provide a reason for the payment. Partnerships of which the debtor is a general partner have been added to the examples of "insiders."

Part 4, *Identify Legal Actions, Repossessions, and Foreclosures*, consolidates questions regarding actions against the debtor's property. The form provides examples of types of legal actions, and requires the debtor to indicate the status of any action. The form adds the requirements that a debtor include any property levied on within a year of filing for bankruptcy and that the debtor provide the last four digits of any account number for any setoffs. Also, a debtor must list any assignment for the benefit of creditors made within one year of filing for bankruptcy.

Part 5, *List Certain Gifts and Contributions*, changes the reporting threshold to \$600 per person or charity and increases the look-back period from one to two years.

Part 6, *List Certain Losses*, clarifies how to report insurance coverage for losses. It provides that the debtor must include on this form amounts of insurance that have been paid, but must list pending insurance claims on Official Form 106A/B.

April 22-23, 2014 Page 144 of 596

Part 7, List Certain Payments or Transfers, includes questions regarding payments or transfers of property by the debtor. The question regarding payments or transfers to anyone who was consulted about seeking bankruptcy or preparing a bankruptcy petition requires the email or website address of the person who was paid, as well as the name of the person who made the payment if it was not the debtor. There is a separate question asked about payments or transfers to anyone who promised to help the debtor deal with creditors or make payments to creditors, reminding the debtor not to include any payments or transfers already listed. Also, the debtor must list any transfers of property, outright or for security purposes, made within two years of filing for bankruptcy, unless the transfer was made in the ordinary course of the debtor's business. There is a reminder not to list gifts or other transfers already included elsewhere on the form. The question regarding self-settled trusts adds an explanation that such trusts are often referred to as asset-protection devices.

Part 8, List Certain Financial Accounts, Safety Deposit Boxes, and Storage Units, adds money market accounts to the examples provided for the question regarding financial accounts or instruments and removes "other instruments" from the examples. Also, the form adds a question about whether the debtor has or had property stored in a storage unit within one year of filing for bankruptcy. The debtor must provide the name and address of the storage facility and anyone who has or had access to the unit, as well as a description of the contents and whether the debtor still has access to the storage unit. Storage units that are part of the building in which the debtor resides are excluded.

Part 9, *Identify Property You Hold or Control for Someone Else*, instructs that the debtor should include any property that the debtor borrowed from, is storing for, or is holding in trust for someone.

Part 10, *Give Details About Environmental Information*, requires the debtor to list the case title and nature of the case for any judicial or administrative proceeding under any environmental law and to indicate the status of the case.

April 22-23, 2014 Page 145 of 596

Part 11, *Give Details About Your Business or Connections to Any Business*, eliminates instructions that apply only to corporations and partnerships. The debtor must indicate if, within four years (previously six years) before filing for bankruptcy, the debtor owned a business or had certain connections to a business, with five categories of businesses provided as checkboxes. If the debtor has a connection to a business, the debtor must list the name, address, nature, and Employer Identification number of the business, the dates the business existed, and the name of an accountant or bookkeeper for the business. Accounting information requested is truncated; the debtor is simply required to provide the name of the business bookkeeper or accountant.

Part 12, *Sign Below*, eliminates the signature boxes for a partnership or corporation and a non-attorney bankruptcy petition preparer. Also, the debtor is asked to indicate through checkboxes whether additional pages are attached to the form.

April 22-23, 2014 Page 146 of 596

Draft March 7, 2014

Fill in this information to identify your case:			
Debtor 1	First Name	Middle Name	Last Name
Debtor 2			
(Spouse, if filing)	First Name	Middle Name	Last Name
United States E	Bankruptcy Court fo	r the:	District of
Case number			(State)
(If known)			

☐ Check if this is an amended filing

12/15

Official Form 112

Statement of Intention for Individuals Filing Under Chapter 7

If you are an individual filing under chapter 7, you must fill out this form if:

- creditors have claims secured by your property, or
- you have leased personal property and the lease has not expired.

You must file this form with the court within 30 days after you file your bankruptcy petition or by the date set for the meeting of creditors, whichever is earlier, unless the court extends the time for cause. You must also send copies to the creditors and lessors you list on the form.

If two married people are filing together in a joint case, both are equally responsible for supplying correct information. Both debtors must sign and date the form.

Be as complete and accurate as possible. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known).

Part 1: List Your Creditors Who Hold Secured Claims

For any creditors that you listed in Part 1 of Schedule D: 0 information below.	al Form 106D), fill in the	
Identify the creditor and the property that is collateral	What do you intend to do with the property that secures a debt?	Did you claim the property as exempt on Schedule C?
Creditor's name: Description of property securing debt:	 ☐ Surrender the property. ☐ Retain the property and redeem it. ☐ Retain the property and enter into a Reaffirmation Agreement. ☐ Retain the property and [explain]: 	□ No □ Yes
Creditor's name: Description of property securing debt:	 □ Surrender the property. □ Retain the property and redeem it. □ Retain the property and enter into a Reaffirmation Agreement. □ Retain the property and [explain]: 	□ No □ Yes
Creditor's name: Description of property securing debt:	□ Surrender the property. □ Retain the property and redeem it. □ Retain the property and enter into a Reaffirmation Agreement. □ Retain the property and [explain]:	□ No □ Yes
Creditor's name: Description of property securing debt:	□ Surrender the property. □ Retain the property and redeem it. □ Retain the property and enter into a Reaffirmation Agreement. □ Retain the property and [explain]:	□ No □ Yes

			Case number (If known)
First Name	Middle Name	Last Name	,		

Part 2:	List Your Unexpired Po	ersonal Property	/ Leases

fill	r any unexpired personal property lease that you listed in <i>Schedule G: Executory Contracts and Unexpir</i> in the information below. Do not list real estate leases. <i>Unexpired leases</i> are leases that are still in effected. You may assume an unexpired personal property lease if the trustee does not assume it. 11 U.S.C.	t; the lease period has not yet
	Describe your unexpired personal property leases	Will the lease be assumed?
	Lessor's name:	□ Na

Describe your unexpired personal property leases	Will the lease be assumed?
Lessor's name:	□ No
Description of leased property:	☐ Yes
Lessor's name:	□ No
Description of leased property:	☐ Yes
Lessor's name:	□ No
Description of leased property:	☐ Yes
Lessor's name:	□ No
Description of leased property:	Yes
Lessor's name:	□ No
Description of leased property:	☐ Yes
Lessor's name:	□ No
Description of leased property:	☐ Yes
Lessor's name:	□ No
Description of leased	☐ Yes

Part 3: Sign Below

Under penalty of perjury, I declare that I have indicated my intention about any property of my estate that secures a debt and any personal property that is subject to an unexpired lease.

×	×		
Signature of Debtor 1	Signature of Debtor 2		
Date	Date		

COMMITTEE NOTE

Official Form 112, *Statement of Intention for Individuals Filing Under Chapter* 7, is revised in its entirety as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. In addition, the form is renumbered, and stylistic changes are made throughout the form.

The form is derived from former Official Form 8, *Chapter 7 Individual Debtor's Statement of Intention.* The new form uses language likely to be understandable to non-lawyers. In addition, the instructions are more extensive, advising an individual Chapter 7 debtor that the form must be completed and filed within 30 days and that the debtor must deliver copies of the form to creditors and lessors listed on the form.

Part 1, *Your Creditors Who Hold Secured Claims*, refers to entering into a "Reaffirmation Agreement" rather than asking whether the debtor intends to "reaffirm the debt." In addition, the debtor is asked if the property is claimed as exempt on Schedule C (Official Form 106C).

Part 2, *List Your Unexpired Personal Property Leases*, defines unexpired leases and explains that a debtor may assume an unexpired personal property lease if the trustee does not assume it.

April 22-23, 2014 Page 149 of 596

April 22-23, 2014 Page 150 of 596

Fill in this information to identify the case:				
Debtor 1	First Name	Middle Name	Last Name	
Debtor 2 (Spouse, if filing)		Middle Name	Last Name	
United States E	Bankruptcy Court fo	or the:		
Case number			(State) Chapter	

Draft March 14, 2014

Official Form 119

Bankruptcy Petition Preparer's Notice, Declaration, and Signature

12/15

Bankruptcy petition preparers as defined in 11 U.S.C. § 110 must fill out this form every time they help prepare documents that are filed in the case. If more than one bankruptcy petition preparer helps with the documents, each must sign in Part 3. A bankruptcy petition preparer who does not comply with the provisions of title 11 of the United States Code and the Federal Rules of Bankruptcy Procedure may be fined, imprisoned, or both. 11 U.S.C. § 110; 18 U.S.C. § 156.

Part 1:

Notice to Debtor

Bankruptcy petition preparers must give the debtor a copy of this form and have the debtor sign it before they prepare any documents for filing or accept any compensation. A signed copy of this form must be filed with any document prepared.

Bankruptcy petition preparers are not attorneys and may not practice law or give you legal advice, including the following:

- whether to file a petition under the Bankruptcy Code (11 U.S.C. § 101 et seq.);
- whether filing a case under chapter 7, 11, 12, or 13 is appropriate;
- whether your debts will be eliminated or discharged in a case under the Bankruptcy Code;
- whether you will be able to keep your home, car, or other property after filing a case under the Bankruptcy Code;
- what tax consequences may arise because a case is filed under the Bankruptcy Code;
- whether any tax claims may be discharged;
- whether you may or should promise to repay debts to a creditor or enter into a reaffirmation agreement;
- how to characterize the nature of your interests in property or your debts; or
- what procedures and rights apply in a bankruptcy case.

	Name		has notified me of
any maximum allowable fee before	e preparing any document for filing o	or accepting any fee.	
	· · · · · · · · · · · · · · · · · · ·	Date	
Signature of Debtor 1 acknowledging reco	eipt of this notice	MM/DD /YYYY	,
Signature of Debtor 2, acknowledging rec	ceipt of this notice	Date MM / DD / YYYY	

Official Form 119

Debtor	1
--------	---

First Name	Middle Name	Last Name

Case number	(if known)	

Part 2:

Declaration and Signature of the Bankruptcy Petition Preparer

	nder penalty of perjury, I declare that:								
	I am a bankruptcy petition preparer or the offi	-		•					
	I or my firm prepared the documents listed be <i>Preparer</i> as required by 11 U.S.C. §§ 110(b),		•		e No	ntice to Debtor by Bankruptcy Petition			
	if rules or guidelines are established according preparers may charge, I or my firm notified the accepting any fee from the debtor.								
	Printed name Title, if a	ny		Firm name, if it applies					
	Number Street								
	City State		ZIP Code	Contact phone		_			
	I or my firm prepared the documents checked below and the completed declaration is made a part of each document (Check all that apply.)								
	☐ Voluntary Petition (Form 101)		Schedule I (Fo	,	Ц	Chapter 11 Statement of Your Current Monthly Income (Form 109)			
	Statement About Your Social Security Numbers (Form 121) Your Assets and Liabilities and Certain			About an Individual Debtor's		Chapter 13 Statement of Your Current Monthly Income and Calculation of Commitment Period (Form 110-1)			
	Statistical Information (Form 106Sum) Schedule A/B (Form 106A/B)			Financial Affairs (Form 107) Intention for Individuals Filing		,			
						Income (Form 110-2)			
	Schedule C (Form 106C)		Under Chapter 7 (Form 112)			Application to Pay Filing Fee in Installments (Form 103A)			
	☐ Schedule D (Form 106D) ☐ Schedule E/F (Form 106E/F)	Ц	Chapter 7 Statement of Your Current Monthly Income (Form 108-1)	Application to Have Chapter 7 Filing Fee					
	Schedule G (Form 106G)			atement of Exemption from Presumption Abuse Under § 707(b)(2) orm 108-1Supp) apter 7 Means Test Calculation orm 108-2)		Waived (Form 103B)			
	Schedule H (Form 106H)				_	A list of names and addresses of all creditors (creditor or mailing matrix)			
			Chapter 7 Me (Form 108-2)			Other			
	Bankruptcy petition preparers must sign and give their Social Security numbers. If more than one bankruptcy petition preparer prepared the documents to which this declaration applies, the signature and Social Security number of each preparer must be provided. 11 U.S.C. § 110.								
	Signature of bankruptcy petition preparer or officer, pringerson, or partner	icipal, i	responsible	Social Security number of po	erson	Date n who signed MM / DD / YYYY			
	Printed name								
	Signature of bankruptcy petition preparer or officer, pringerson, or partner	responsible	Social Security number of p	 erson	Date u who signed MM / DD / YYYY				
	Printed name								

Official Form 119

COMMITTEE NOTE

Official Form 119, *Bankruptcy Petition Preparer's Notice*, *Declaration*, *and Signature*, applies only in cases of individual debtors. It is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. In addition, the form is renumbered, and stylistic changes are made throughout the form.

The form is derived from former Official Form 19, Declaration and Signature of Non-Attorney Bankruptcy Petition Preparer. An instruction is added to the form that provides statutory citations. Filers are advised that if more than one bankruptcy petition preparer helped with the documents, each must sign the form.

Part 1, *Notice to Debtor*, is moved to the beginning of the form and revised. An instruction is added that bankruptcy petition preparers must give the debtor a copy of the form and have the debtor sign it before they prepare any documents for filing or accept compensation, and that the form must be filed with any document prepared. It warns the debtor that bankruptcy petition preparers are not attorneys and may not practice law or give legal advice, with a list of examples of advice that may not be provided by a bankruptcy petition preparer. The signature line of this part includes a statement that the debtor acknowledges receipt of the notice.

Part 2, *Declaration and Signature of the Bankruptcy Petition Preparer*, revises the declaration by the bankruptcy
petition preparer to include an officer, principal, responsible
person, or partner of a bankruptcy petition preparer. The
bankruptcy petition preparer must provide a firm name, if
applicable, as well as a contact phone, and must indicate which
documents the bankruptcy petition preparer prepared from a list of
documents. An "other" option is provided for any additional
documents. The signature line includes spaces for the bankruptcy
petition preparer to enter a social security number, and language

April 22-23, 2014 Page 153 of 596

regarding an officer, principal, responsible person, or partner of the bankruptcy petition preparer.

April 22-23, 2014 Page 154 of 596

Fill in this information to identify your case:						
United States Bankruptcy Court for the:						
District of						
Case number (If known):						

Draft March 14, 2014

Official Form 121

Statement About Your Social Security Numbers

12/15

Use this form to tell the court about any Social Security or federal Individual Taxpayer Identification numbers you have used. Do not file this form as part of the public case file. This form must be submitted separately and must not be included in the court's public electronic records. Please consult local court procedures for submission requirements.

To protect your privacy, the court will not make this form available to the public. You should not include a full Social Security Number or Individual Taxpayer Number on any other document filed with the court. The court will make only the last four digits of your numbers known to the public. However, the full numbers will be available to your creditors, the U.S. Trustee or bankruptcy administrator, and the trustee assigned to your case.

Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Part 1: Tell the Court	About Yourself and Your spouse if Your Spouse i	s Filing With You
	For Debtor 1:	For Debtor 2 (Only If Spouse Is Filing):
1. Your name		
	First name	First name
	Middle name	Middle name
	Last name	Last name
Part 2: Tell the Court	About all of Your Social Security or Federal Indiv	idual Taxpayer Identification Numbers
2. All Social Security Numbers you have used		
usea		
	☐ You do not have a Social Security number.	☐ You do not have a Social Security number.
3. All federal Individual Taxpayer Identification	9	9
Numbers (ITIN) you have used	9	9
Part 3: Sign Below	☐ You do not have an ITIN.	☐ You do not have an ITIN.
	Under penalty of perjury, I declare that the information I have provided in this form is true and correct.	Under penalty of perjury, I declare that the information I have provided in this form is true and correct.
	x	x
	Signature of Debtor 1	Signature of Debtor 2
	Date	Date

April 22-23, 2014 Page 155 of 596

April 22-23, 2014 Page 156 of 596

COMMITTEE NOTE

Official Form 121, *Statement About Your Social Security Numbers*, is revised as part of the Forms Modernization Project. The form, which applies only in cases of individual debtors, replaces former Official Form 21, *Statement of Social Security Number(s)*. It is renumbered to distinguish it from the forms used by non-individual debtors, such as corporations and partnerships.

To make Form 121 easier to understand and complete, the form is divided into three sections, and directions on the form are simplified. The debtors' Employer Tax-Identification number (EIN) is eliminated from the form, and the debtor's name is moved from the caption to the body of the form.

April 22-23, 2014 Page 157 of 596

April 22-23, 2014 Page 158 of 596

Draft February 26, 2014

Information to identify the case:							
Debtor 1				Last 4 digits of Social Security number or ITIN			
	First Name	Middle Name	Last Name	EIN			
Debtor 2				Last 4 digits of Social Security number or ITIN			
(Spouse, if filing) First Name	Middle Name	Last Name				
				EIN			
United States Bankruptcy Court for the:							
			(State)				
Case number:							

Order of Discharge

IT IS ORDERED: A discharge under	11 U.S.C. § 727 is granted to:	
[include all names used] ears prior to the filing of the petition]
MM / DD / YYYY	By the court: United S	States Bankruptcy Judge

Explanation of Bankruptcy Discharge in a Chapter 7 Case

This order does not close or dismiss the case, and it does not determine how much money, if any, the trustee will pay creditors.

Creditors cannot collect discharged debts

This order means that no one may make any attempt to collect a discharged debt from the debtors personally. For example, creditors cannot sue, garnish wages, assert a deficiency, or otherwise try to collect from the debtors personally on discharged debts. Creditors cannot contact the debtors by mail, phone, or otherwise in any attempt to collect the debt personally. Creditors who violate this order can be required to pay debtors damages and attorney's fees.

However, a creditor with a lien may enforce a claim against the debtors' property subject to that lien unless the lien was avoided or eliminated. For example, a creditor may have the right to foreclose a home mortgage or repossess an automobile.

This order does not prevent debtors from paying any debt voluntarily or from paying reaffirmed debts according to the reaffirmation agreement.

11 U.S.C. § 524(c), (f).

Most debts are discharged

Most debts are covered by the discharge, but not all. Generally a discharge removes the debtors' personal liability for debts owed before the debtors' bankruptcy case was filed.

Also, if this case began under a different chapter of the Bankruptcy Code and was later converted to chapter 7, debts owed before the conversion are discharged.

In a case involving community property: Special rules protect certain community property owned by the debtor's spouse, even if that spouse did not file a bankruptcy case.

For more information, see page 2 ▶

Official Form B318 Order of Discharge page 1

Some debts are not discharged

Examples of some debts that are not discharged are:

debts that are domestic support obligations;

debts for most student loans;

debts for most taxes;

debts that the bankruptcy court has decided or will decide are not discharged in this bankruptcy case;

debts for most fines, penalties, forfeitures, or criminal restitution obligations;

some debts which the debtors did not properly list;

debts for certain types of loans owed to pension, profit sharing, stock bonus, or retirement plans; and

debts for death or personal injury caused by operating a vehicle while intoxicated.

Also, debts covered by a valid reaffirmation agreement are not discharged.

In addition, this discharge does not stop creditors from collecting from anyone else who is also liable on the debt, such as an insurance company or a person who cosigned or guaranteed a loan.

This information is only a general summary of the bankruptcy discharge; some exceptions exist. Because the law is complicated, you should consult an attorney to determine the exact effect of the discharge in this case.

Official Form B318 Order of Discharge page 2

COMMITTEE NOTE

Official Form 318, *Order of Discharge*, is revised and renumbered as part of the Forms Modernization Project. The form is used to issue a discharge in chapter 7 cases filed by individuals or joint debtors. It replaces former Official Form 18, *Discharge of Debtor*, Director's Procedural Form 18J, *Discharge of Joint Debtors*, and Director's Procedural Form 18JO, *Discharge of One Joint Debtor*.

To make the discharge order and the explanation of it easier to read and understand, legal terms are explained more fully or replaced with commonly understood terms, and the form is reformatted.

Reaffirmed debts are explained more fully, and readers are informed that a discharge will not stop creditors from collecting debts from any property in which they have a valid lien. In addition, readers are advised that the discharge does not stop creditors from collecting from anyone else who is liable on the debt, such as a cosigner on the loan or an insurance company.

Director's Procedural Forms 18J and 18JO are no longer needed because Form 318 specifies the names of the debtors, or debtor, to whom the discharge is issued. Any alternate names of the debtor or debtors appear in the order not in the information box at the top of the form.

April 22-23, 2014 Page 161 of 596

April 22-23, 2014 Page 162 of 596

Fill in this information to identify the case:							
Debtor 1	First Name	Middle Name	Last Name				
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name				
United States E	Bankruptcy Court fo	r the:	District of(State)				
Case number (If known)			(State)				

Draft February 25, 2014

Official Form 423

Certification About a Financial Management Course

12/15

If you are an individual, you must take an approved course about personal financial management if:

- you filed for bankruptcy under chapter 7 or 13, or
- you filed for bankruptcy under chapter 11 and § 1141 (d)(3) does not apply.

In a joint case, each debtor must take the course. 11 U.S.C. §§ 727(a)(11) and 1328(g).

After you finish the course, the provider will give you a certificate. The provider may notify the court that you have completed the course. If the provider does not ify the court, you need not file this form. If the provider does not notify the court, then Debtor 1 and Debtor 2 must each file this form with the certificate number before your debts will be discharged.

- If you filed under chapter 7 and you need to file this form, file it within 60 days after the first date set for the meeting of creditors under § 341 of the Bankruptcy Code.
- If you filed under chapter 11 or 13 and you need to file this form, file it before you make the last payment that your plan requires or before you file a motion for a discharge under § 1141(d)(5)(B) or § 1328(b) of the Bankruptcy Code. Fed. R. Bankr. P. 1007(c).

In some cases, the court can waive the requirement to take the financial management course. To have the requirement waived, you must file a motion with the court and obtain a court order.

Part 1: Tell the Court About the Required Course

You n	You must check one:							
	I completed an approved course in personal financial management:							
	Date	e I took the cou	MM / DD / YYYY					
	Nam	ne of approved	provider					
	Cert	ificate number						
			to complete a course in personal financial management because the court has granted my motion for a uirement based on (check one):					
		Incapacity.	I have a mental illness or a mental deficiency that makes me incapable of realizing or making rational decisions about finances.					
		Disability.	My physical disability causes me to be unable to complete a course in personal financial management in person, by phone, or through the internet, even after I reasonably tried to do so.					
		Active duty.	I am currently on active military duty in a military combat zone.					
		Residence.	I live in a district in which the United States trustee (or bankruptcy administrator) has determined that the approved instructional courses cannot adequately meet my needs.					
Part	2:	Sign Here						
	I certify that the information I have provided is true and correct.							
	Signature of debtor named on certificate Printed name of debtor MM / DD / YYYY							

Official Form 423

April 22-23, 2014 Page 163 of 596

April 22-23, 2014 Page 164 of 596

COMMITTEE NOTE

Official Form 423, Certification About a Financial Management Course, is revised as part of the Forms Modernization Project. The form replaces former Official Form 23, Debtor's Certification of Completion of Postpetition Instructional Course Concerning Personal Financial Management. Form 423 is renumbered to distinguish it from the forms used by non-individual debtors, such as corporations and partnerships.

To make Form 423 easier to understand, legal terms are explained more fully or replaced with commonly understood terms, and the form is reformatted. Part 1, *Tell the Court About the Required Course*, provides definitions for "incapacity" and "disability," rather than providing statutory citations.

A statement is added that, in some cases, the court can waive the requirement to complete the financial management course. To have the requirement waived, the debtor must file a motion with the court and obtain a court order.

April 22-23, 2014 Page 165 of 596

April 22-23, 2014 Page 166 of 596

Fill in this information to identify your case:							
Debtor 1	First Name	Middle Name	Last Name				
Debtor 2 (Spouse, if filing)		Middle Name	Last Name				
()		Widdle Name					
Case number (If known)			(State)				

Draft March 31, 2014

Official Form 427

Cover Sheet for Reaffirmation Agreement

12/14

Anyone who is a party to a reaffirmation agreement may fill out and file this form. Fill it out completely, attach it to the reaffirmation agreement, and file the documents within the time set under Bankruptcy Rule 4008.

	•	epayment Terms of the Reaffirmation Agreement	
1.	Who is the creditor?	Name of the creditor	
2.	How much is the debt?	On the date that the bankruptcy case is filed \$	
		To be paid under the reaffirmation agreement \$	
		\$ per month for months (if fixed inte	erest rate)
3.	What is the Annual Percentage Rate (APR) of interest? (See	Before the bankruptcy case was filed	%
	Bankruptcy Code § 524(k)(3)(E).)	Under the reaffirmation agreement	%
4.	Does collateral secure the debt?	☐ No☐ Yes. Describe the collateral.	
		Current market value \$	_
5.	Does the creditor assert that the debt is nondischargeable?	☐ No☐ Yes. Attach an explanation of the nature of the debt a	and the basis for contending that the debt is nondischargeable.
6.	Using information from Schedule G: Your	Income and expenses reported on Schedules G and H	Income and expenses stated on the reaffirmation agreement
	Income (Official Form 106G) and Schedule H: Your Expenses (Official Form 106H), fill in the	6a. Combined monthly income from \$line 12 of Schedule G	6e. Monthly income from all sources \$s
	amounts.	6b. Monthly expenses from Column A, line 22 of Schedule H - \$	6f. Monthly expenses — \$
		6c. Monthly payments on all reaffirmed debts not listed on Schedule H	6g. Monthly payments on all reaffirmed debts not included in monthly expenses
		6d. Scheduled net monthly income \$	6h. Present net monthly income \$
		Subtract lines 6b and 6c from 6a.	Subtract lines 6f and 6g from 6e.
		If the total is less than 0, put the number in brackets.	If the total is less than 0, put the number in brackets.

7.	Are the income amounts on lines 6a and 6e different?	No Yes.	Explain why they are different and comple	te line 10	
3.	Are the expense amounts on lines 6b and 6f different?	☐ No ☐ Yes.	Explain why they are different and comple	te line 10	
).	Is the net monthly income in line 6h less than 0?	□ No □ Yes.	A presumption of hardship arises (unless to Explain how the debtor will make monthly Complete line 10.	he creditor is a credit union). payments on the reaffirmed debt a	nd pay other living expenses.
10.	Debtor's certification about lines 7-9		I certify that each explanation on lines 7-9	is true and correct.	
	If any answer on lines 7-9 is Yes, the debtor must sign here.		×	*	
	If all the answers on lines 7-9 are <i>No</i> , go to line 11.		Signature of Debtor 1	Signature of Debtor 2 ((Spouse Only in a Joint Case)
11.	Did an attorney represent the debtor in negotiating the reaffirmation agreement?	☐ Yes.	Has the attorney executed a declaration or No Yes	an affidavit to support the reaffirma	tion agreement?
P	art 2: Sign Here				
W	hoever fills out this form ust sign here.	I certify the	nat the attached agreement is a true and entified on this Cover Sheet for Reaffirm	correct copy of the reaffirmation ation Agreement.	agreement between the
		×			
		Signatu	ле	Date	MM / DD / YYYY
		Printed	I Name		
		_	ek one: Debtor or Debtor's Attorney		
			Creditor or Creditor's Attorney		

Case number (if known)_

Debtor 1

First Name

Middle Name

Last Name

COMMITTEE NOTE

Line 3 of the form has been changed to clarify the requirement to disclose an annual percentage rate of interest. Section 524(k)(3)(E) of the Bankruptcy Code defines the "Annual Percentage Rate" to be disclosed in connection with a reaffirmation agreement. Line 3 of the form now includes a reference to that Code provision, which in appropriate circumstances permits disclosure of the simple interest rate as the Annual Percentage Rate.

April 22-23, 2014 Page 169 of 596

April 22-23, 2014 Page 170 of 596

Instructions

Bankruptcy Forms for Individuals

U.S. Bankruptcy Court

December 2015

About this Booklet of Instructions

About this Booklet of Instructions	2
About the bankruptcy forms and filing bankruptcy	
Things to remember when filling out these forms	4
About the Process for Filing a Bankruptcy Case for Individuals	
Notice Required by 11 U.S.C. § 342(b) for Individuals Filing for Bankruptcy (Form 2010)	
The types of bankruptcy that are available to individuals	
Bankruptcy crimes have serious consequences	12
Understand which services you could receive from credit counseling agencies	12
Instructions for Selected Forms	13
Schedule A/B: Property (Official Form 106A/B)	14
Schedule C: The Property You Claim as Exempt (Official Form 106C)	16
Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D)	17
Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F)	20
Schedule G: Executory Contracts and Unexpired Leases (Official Form 106G)	24
Schedule H: Your Codebtors (Official Form 106H)	25
Schedule I: Your Income (Official Form 106I)	26
Schedule J: Your Expenses (Official Form 106J)	28
Summary of Your Assets and Liabilities and Certain Statistical Information (Official Form 106Sum)	29
Statement of Financial Affairs for Individuals Filing for Bankruptcy (Official Form 107)	
Chapter 7 Statement of Your Current Monthly Income and Means Test Calculation (Official	
Forms 108-1, 108A-1Supp, and 108-2)	
Chapter 11 Statement of Your Current Monthly Income (Official Form 109)	33
Chapter 13 Statement of Your Current Monthly Income, Calculation of Commitment Period and Chapter 13 Calculation of Your Disposable Income (Official Forms 110–1 and 110–2)	34
Statement of Intention for Individuals Filing Under Chapter 7 (Official Form 112)	35
Application for Individuals to Pay the Filing Fee in Installments (Official Form 103A)	37
Application to Have the Chapter 7 Filing Fee Waived (Official Form 103B)	38
For Individual Chapter 11 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims Against You and Are Not Insiders (Official Form 104)	38
Glossary	40
Definitions of Some Terms Used in the Forms for Individuals Filing for Bankruptcy	41

About this Booklet of Instructions

About this Booklet of Instructions

This booklet provides instructions for completing selected forms that individuals filing for bankruptcy must submit to the U.S. Bankruptcy Court. You can download all of the required forms without charge

from: http://www.uscourts.gov/FormsAndFees/Forms/BankruptcyForms.aspx.

The instructions are designed to accompany the forms and are intended to help you understand what information is required to properly file. You are responsible for properly completing the forms. These instructions are not intended to provide, and should not be understood to provide, legal advice. They are not designed to fully explain, or to be relied upon in interpreting, the law.

Completing the forms is only a part of the bankruptcy process. You are strongly encouraged to hire a qualified attorney not only to help you complete the forms but also to give you general advice about bankruptcy and to represent you in your bankruptcy case. If you cannot afford to pay an attorney, you might qualify for free legal services if they are provided in your area. Contact your state or local bar association for help in obtaining free legal services or in hiring an attorney.

Note: It is extremely difficult to succeed in a chapter 11, 12, or 13 case without an attorney.

If an attorney represents you, you must provide information so the attorney can prepare your forms. Once the attorney prepares the forms, you must make sure that the forms are accurate and complete. These instructions may help you perform those tasks. If you are filing for bankruptcy without the help of an attorney, this booklet tells you which forms must be filed and provides information about them.

You should carefully read this booklet and keep it with your records. Review the individual forms as you read the instructions for each.

Although bankruptcy petition preparers can help you type the bankruptcy forms, they cannot tell you how to complete the forms, they cannot file the documents for you, and they cannot give you legal advice. Court employees cannot give you legal advice, either.

About this Booklet of Instructions

Read These Important Warnings

Because bankruptcy can have serious long-term financial and legal consequences, including loss of your property, you should hire an attorney and carefully consider all of your options before you file. Only an attorney can give you legal advice about what can happen as a result of filing for bankruptcy and what your options are. If you do file for bankruptcy, an attorney can help you fill out the forms properly and protect you, your family, your home, and your possessions.

Although the law allows you to represent yourself in bankruptcy court, you should understand that many people find it difficult to represent themselves successfully. The rules are technical, and a mistake or inaction may harm you. If you file without an attorney, you are still responsible for knowing and following all of the legal requirements.

You should not file for bankruptcy if you are not eligible to file or if you do not intend to file the necessary documents.

Bankruptcy fraud is a serious crime; you could be fined and imprisoned if you commit fraud in your bankruptcy case. Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

About the bankruptcy forms and filing bankruptcy

Use the forms that are numbered in the 100 series to file bankruptcy for individuals or married couples. Use the forms that are numbered in the 200 series if you are preparing a bankruptcy on behalf of a nonindividual, such as a corporation, partnership, or limited liability company (LLC). Sole proprietors must use the forms that are numbered in the 100 series.

When a bankruptcy is filed, the U.S. Bankruptcy Court opens a case. It is important that the answers to the questions on the forms be complete and accurate so that the case proceeds smoothly. A person filing bankruptcy who gives false information could be charged with a federal crime or could lose all the benefits of filing for bankruptcy.

You should understand that filing a bankruptcy case is not private. Anyone has a right to see your bankruptcy forms after you file them, unless the court orders otherwise under 11 U.S.C. § 107. Certain information in court filings, however, must be protected from public disclosure under Bankruptcy Rule 9037.

Understand the terms used in the forms

The forms for individuals use *you* and *Debtor 1* to refer to a debtor filing alone. A married couple may file a bankruptcy case together—called a *joint case*—and in joint cases, these forms use *you* to ask for information from both debtors.

About this Booklet of Instructions

For example, if a form asks, "Do you own a car?" the answer would be *yes* if either debtor owns a car. When information is needed about the spouses separately, the forms use *Debtor 1* and *Debtor 2* to distinguish between them. In joint cases, one of the spouses must report information as *Debtor 1* and the other as *Debtor 2*. The same person must be *Debtor 1* in all of the forms.

To understand other terms used in the forms and the instructions, see the *Glossary* at the end of this booklet.

Things to remember when filling out these forms

- Do not file these instructions with the bankruptcy forms that you file with the court.
- Be as complete and accurate as possible.
- If more space is needed, attach a separate sheet to the form. On the top of any additional pages, write your name and case number (if known). Also identify the form and line number to which the additional information applies.

- If two married people are filing together, both are equally responsible for supplying correct information.
- Do not list a minor child's full name. Instead, fill in only the child's initials and the full name and address of the child's parent or guardian. For example, write A.B., a minor child (*John Doe, parent, 123 Main St., City, State*). 11 U.S.C. § 112; Bankruptcy Rule 1007(m) and 9037.
- For your records, be sure to keep a copy of your bankruptcy documents and all attachments that you file.

On what date was a debt incurred?

When a debt was incurred on a single date, fill in the actual date that the debt was incurred.

When a debt was incurred on multiple dates, fill in the range of dates. For example, if the debt is from a credit card, fill in the month and year of the first and last transaction.

About this Booklet of Instructions

About the Process for Filing a Bankruptcy Case for Individuals

Before you file your bankruptcy case

Before you file for bankruptcy, you must do several things:

□ Receive a briefing about credit counseling from an approved agency within 180 days before you file. (If you and your spouse are filing together, each of you must receive a briefing before you file. Failure to do so may result in the dismissal of your case.) You may have a briefing about credit counseling one-on-one or in a group, by telephone, or by internet.

For a list of approved providers, go to: http://www.justice.gov/ust/eo/bapcpa/ccde/cc_a pproved.htm

In Alabama and North Carolina, go to: http://www.uscourts.gov.

After you finish the briefing, you will receive a certificate that you will need to file in your bankruptcy case.

☐ Find out in which bankruptcy court you must file your bankruptcy case. It is important that you file in the correct district within your state. To find out which district you are in, go to:

http://www.uscourts.gov/courtlinks

- □ Check the local court's website for any specific local requirements that you might have to meet. Go to:

 http://www.uscourts.gov/courtlinks
- □ Find out which chapters of the Bankruptcy
 Code you are eligible for. For descriptions of
 each chapter, review the information
 contained in the notice, *Notice Required by*11 U.S.C. § 342(b) for Individuals Filing for
 Bankruptcy (Form B2010), which is included
 in this booklet.

When you file your bankruptcy case

There are several forms and documents that you must give the court at the time you file.

Additional forms and documents must be filed no later than 14 days after you file your bankruptcy case, although they may be filed at the same time you file your case.

You must file the forms listed below on the date you open your bankruptcy case. For copies of the forms listed here, go to http://www.uscourts.gov. (The list continues on the next page.):

- □ Voluntary Petition for Individuals Filing for Bankruptcy (Official Form 101). This form opens the case. Directions for filling it out are included in the form itself.
- □ Statement About Your Social Security
 Numbers (Official Form 121). This form gives the court your full Social Security number or federal Individual Taxpayer Identification number. To protect your privacy, the court will make only the last four digits of your number known to the general public.
 However, the court will make your full number available to your creditors, the U.S. trustee or bankruptcy administrator, and the trustee assigned to your case. This form has no separate instructions.
- ☐ Your filing fee. If you cannot pay the entire filing fee, you must also include:
 - ☐ Application for Individuals to Pay the Filing Fee in Installments (Official Form 103A), or
 - ☐ Application to Have the Chapter 7 Filing Fee Waived (Official Form 103B). Use this form only if you are filing under chapter 7 and you meet the criteria to have the chapter 7 filing fee waived.
- □ A list of names and addresses of all of your creditors, formatted as a mailing list according

- to instructions from the bankruptcy court in which you file. (Your court may call this a *creditor matrix* or *mailing matrix*.)
- □ Your credit counseling certificate from an approved credit counseling agency. (See *Before you file your bankruptcy case*, above). If you have received the briefing about credit counseling but have not yet received the certificate, file it no later than 14 days after you file for bankruptcy. If you have not already received the briefing and believe you are entitled to a temporary waiver from receiving it or that you are not required to receive the briefing, see line 15 of the *Voluntary Petition for Individuals Filing for Bankruptcy* (Official Form 101).
- ☐ For Individual Chapter 11 Cases: List of
 Creditors Who Have the 20 Largest
 Unsecured Claims Against You and Are Not
 Insiders (Official Form 104). Fill out this form
 only if you file under chapter 11.
- ☐ Initial Statement About an Eviction Judgment Against You (Official Form 101A) and Statement About Payment of an Eviction Judgment Against You (Official Form 101B). Use Form 101A if your landlord has an eviction judgment against you. If you complete Form 101A and you want to stay in your residence for the first 30 days after you file, you must indicate that on the form. Use Form 101B if you have completed Form 101A and you want to stay in your rented residence form more than 30 days after you file for bankruptcy.
- □ Bankruptcy Petition Preparer's Notice, Declaration, and Signature (Official Form 119) and Disclosure of Compensation of Bankruptcy Petition Preparer (Form 2800). Use these forms if a bankruptcy petition preparer typed your forms.

About the Process for Filing a Bankruptcy Case for Individuals

When you file your bankruptcy case or within 14 days after you file

You must file the forms listed below either when you file your bankruptcy case or within 14 days after you file your *Voluntary Petition for Individuals Filing for Bankruptcy* (Official Form 101). If you do not do so, your case may be dismissed. Although it is possible to open your case by submitting only the documents that are listed under *When you file your bankruptcy case*, you should file the entire set of forms at one time to help your case proceed smoothly.

Although some forms may ask you similar questions, you must fill out all of the forms completely to protect your legal rights.

The list below shows the forms that all individuals must file as well as the forms that are specific to each chapter. For copies of the official forms listed here, go to http://www.uscourts.gov.

All individuals who file for bankruptcy must file these forms and the forms for the specific chapter:

- □ *Schedules of Assets and Liabilities* (Official Form 106) which includes these forms:
 - □ *Schedule A/B: Property* (Official Form 106A/B)
 - □ Schedule C: The Property You Claim as Exempt (Official Form 106C)
 - □ Schedule D: Creditors Who Have Claims Secured by Your Property (Official Form 106D)
 - ☐ Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F)
 - ☐ Schedule G: Executory Contracts and Unexpired Leases (Official Form 106G)
 - □ *Schedule H: Your Codebtors* (Official Form 106H)
 - □ *Schedule I: Your Income* (Official Form 106I)
 - □ *Schedule J: Your Expenses* (Official Form 106J)

- ☐ Summary of Your Assets and Liabilities and Certain Statistical Information
 (Official Form 106Sum). This form gives an overview of the totals on the schedules
- □ Declaration About an Individual Debtor's Schedules (Official Form 106Dec)
- □ Statement of Financial Affairs for Individuals Filing for Bankruptcy (Official Form 107)
- □ Disclosure of Compensation to Debtor's

 Attorney Unless local rules provide
 otherwise, Director's Form 2030 may be used.
- ☐ Credit counseling certificate that you received from an approved credit counseling agency
- □ Copies of all payment advices (*pay stubs*) or other evidence of payment that you received within 60 days before you filed your bankruptcy case. Some local courts may require that you submit these documents to the trustee assigned to your case rather than filing them with the court. Check the local court's website to find out if local requirements apply. Go to http://www.uscourts.gov/courtlinks.

About the Process for Filing a Bankruptcy Case for Individuals

If you file under chapter 7, you must also file:

- □ Statement of Intention for Individuals Filing Under Chapter 7 (Official Form 112)
- □ Chapter 7 Statement of Your Current Monthly Income (Official Form 108-1)
- ☐ If necessary, Chapter 7 Means Test Calculation (Official Form 108-2)
- ☐ If necessary, Statement of Exemption from Presumption of Abuse Under § 707(b)(2) (Official Form 22A-1Supp)

If you file under chapter 11, you must also file:

□ Chapter 11 Statement of Your Current Monthly Income (Official Form 109)

If you file under chapter 11 and are a small business debtor (that is, if you are self-employed and your debts are less than \$2,490,925*), within 7 days after you file your bankruptcy forms to open your case, you must also file your most recent:

- □ Balance sheet
- □ Statement of operations
- □ Cash-flow statement
- □ Federal income tax return

If you do not have these documents, you must file a statement made under penalty of perjury that you have not prepared either a balance sheet, statement of operations, or cash-flow statement or you have not filed a federal tax return.

If you file under chapter 11, you must file additional documents beyond the scope of these instructions. You should consult your attorney.

* Subject to adjustment on 4/01/16, and every 3 years after that for cases begun on or after the date of adjustment.

If you file under chapter 12, you must also file:

□ Chapter 12 Plan (within 90 days after you file your bankruptcy forms to open your case)

If you file under chapter 13, you must also file:

- □ Chapter 13 Statement of Your Current Monthly Income and Calculation of Commitment Period (Official Form 110-1)
- ☐ If necessary, Chapter 13 Calculation of Your Disposable Income (Official Form 110-2)
- □ Chapter 13 Plan (Official Form 113, if in effect) If Official Form 113 is not effective when you file, many bankruptcy courts require you to use a local form plan. Check the local court's website for any specific form that you might have to use. Go to http://www.uscourts.gov/courtlinks.)

About the Process for Filing a Bankruptcy Case for Individuals

Notice Required by 11 U.S.C. § 342(b) for Individuals Filing for Bankruptcy (Form 2010)

This notice is for you if:

You are an individual filing for bankruptcy, and

Your debts are primarily consumer debts.

Consumer debts are defined in 11 U.S.C. § 101(8) as "incurred by an individual primarily for a personal, family, or household purpose."

The types of bankruptcy that are available to individuals

Individuals who meet the qualifications may file under one of four different chapters of the Bankruptcy Code:

- Chapter 7 Liquidation
- Chapter 11— Reorganization
- Chapter 12— Voluntary repayment plan for family farmers or fishermen
- Chapter 13— Voluntary repayment plan for individuals with regular income

You should have an attorney review your decision to file for bankruptcy and the choice of chapter.

Chap	ter 7	' :	Liqui	ida	tion

	\$245	filing fee
	\$75	administrative fee
+	\$15	trustee surcharge
	\$335	total fee

Chapter 7 is for individuals who have financial difficulty preventing them from paying their debts and who are willing to allow their non-exempt property to be used to pay their creditors. The primary purpose of filing under chapter 7 is to have your debts discharged. The bankruptcy discharge relieves you after bankruptcy from having to pay many of your pre-bankruptcy debts. Exceptions exist for particular debts, and liens on property may still be enforced after discharge. For example, a creditor may have the right to foreclose a home mortgage or repossess an automobile.

However, if the court finds that you have committed certain kinds of improper conduct described in the Bankruptcy Code, the court may deny your discharge.

You should know that even if you file chapter 7 and you receive a discharge, some debts are not discharged under the law.

Therefore, you may still be responsible to pay:

- most taxes;
- most student loans;
- domestic support and property settlement obligations;

Notice Required by 11 U.S.C. U.S.C. § 342(b) for Individuals Filing for Bankruptcy (Form 2010)

- most fines, penalties, forfeitures, and criminal restitution obligations; and
- certain debts that are not listed in your bankruptcy papers.

You may also be required to pay debts arising from:

- fraud or theft;
- fraud or defalcation while acting in breach of fiduciary capacity;
- intentional injuries that you inflicted; and
- death or personal injury caused by operating a motor vehicle, vessel, or aircraft while intoxicated from alcohol or drugs.

If your debts are primarily consumer debts, the court can dismiss your chapter 7 case if it finds that you have enough income to repay creditors a certain amount. You must file *Chapter 7 Statement of Your Current Monthly Income* (Official Form 108–1) if you are an individual filing for bankruptcy under chapter 7. This form will determine your current monthly income and compare whether your income is more than the median income that applies in your state.

If your income is not above the median for your state, you will not have to complete the other chapter 7 form, the *Chapter 7 Means Test Calculation* (Official Form 108–2).

If your income is above the median for your state, you must file a second form —the *Chapter 7 Means Test Calculation* (Official Form 108–2). The calculations on the form—sometimes called the *Means Test*—deduct from your income living expenses and payments on certain debts to determine any amount available to pay unsecured creditors. If

your income is more than the median income for your state of residence and family size, depending on the results of the *Means Test*, the U.S. trustee, bankruptcy administrator, or creditors can file a motion to dismiss your case under § 707(b) of the Bankruptcy Code. If a motion is filed, the court will decide if your case should be dismissed. To avoid dismissal, you may choose to proceed under another chapter of the Bankruptcy Code.

If you are an individual filing for chapter 7 bankruptcy, the trustee may sell your property to pay your debts, subject to your right to exempt the property or a portion of the proceeds from the sale of the property. The property, and the proceeds from property that your bankruptcy trustee sells or liquidates that you are entitled to, is called *exempt property*. Exemptions may enable you to keep your home, a car, clothing, and household items or to receive some of the proceeds if the property is sold.

Exemptions are not automatic. To exempt property, you must list it on *Schedule C: The Property You Claim as Exempt* (Official Form 106C). If you do not list the property, the trustee may sell it and pay all of the proceeds to your creditors.

Chapter 11: Reorganization

\$1,167 filing fee

+ \$550 administrative fee

\$1,717 total fee

Chapter 11 is often used for reorganizing a business, but is also available to individuals. The provisions of chapter 11 are too complicated to summarize briefly.

April 22-23, 2014 Page 181 of 596

Notice Required by 11 U.S.C. U.S.C. § 342(b) for Individuals Filing for Bankruptcy (Form 2010)

Read These Important Warnings

Because bankruptcy can have serious long-term financial and legal consequences, including loss of your property, you should hire an attorney and carefully consider all of your options before you file. Only an attorney can give you legal advice about what can happen as a result of filing for bankruptcy and what your options are. If you do file for bankruptcy, an attorney can help you fill out the forms properly and protect you, your family, your home, and your possessions.

Although the law allows you to represent yourself in bankruptcy court, you should understand that many people find it difficult to represent themselves successfully. The rules are technical, and a mistake or inaction may harm you. If you file without an attorney, you are still responsible for knowing and following all of the legal requirements.

You should not file for bankruptcy if you are not eligible to file or if you do not intend to file the necessary documents.

Bankruptcy fraud is a serious crime; you could be fined and imprisoned if you commit fraud in your bankruptcy case. Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Chapter 12: Repayment plan for family farmers or fishermen

	\$200	filing fee
+	\$75	administrative fee
	\$275	total fee

Similar to chapter 13, chapter 12 permits family farmers and fishermen to repay their debts over a period of time using future earnings and to discharge some debts that are not paid.

Chapter 13: Repayment plan for individuals with regular income

	\$235	filing fee
+	\$75	administrative fee
	¢210	total foo

Chapter 13 is for individuals who have regular income and would like to pay all or part of their debts in installments over a period of time and to discharge some debts that are not paid. You are eligible for chapter 13 only if your debts are not more than certain dollar amounts set forth in 11 U.S.C. § 109.

Under chapter 13, you must file with the court a plan to repay your creditors all or part of the money that you owe them, usually using your future earnings. If the court approves your plan, the court will allow you to repay your debts, as adjusted by the plan, within 3 years or 5 years, depending on your income and other factors.

After you make all the payments under your plan, many of your debts are discharged. The debts that are not discharged and that you may still be responsible to pay include:

- domestic support obligations,
- most student loans,
- certain taxes,
- debts for fraud or theft,
- debts for fraud or defalcation while acting in a fiduciary capacity,
- most criminal fines and restitution obligations,
- certain debts that are not listed in your bankruptcy papers,
- certain debts for acts that caused death or personal injury, and
- certain long-term secured debts.

April 22-23, 2014 Page 182 of 596

Notice Required by 11 U.S.C. U.S.C. § 342(b) for Individuals Filing for Bankruptcy (Form 2010)

Warning: File Your Forms on Time

Section 521(a)(1) of the Bankruptcy Code requires that you promptly file detailed information about your creditors, assets, liabilities, income, expenses and general financial condition. The court may dismiss your bankruptcy case if you do not file this information within the deadlines set by the Bankruptcy Code, the Bankruptcy Rules, and the local rules of the court.

For more information about the documents and their deadlines, go

to: http://www.uscourts.gov/bkforms/bankruptcy_f orms.html#procedure.

Bankruptcy crimes have serious consequences

- If you knowingly and fraudulently conceal assets or make a false oath or statement under penalty of perjury—either orally or in writing—in connection with a bankruptcy case, you may be fined, imprisoned, or both.
- All information you supply in connection with a bankruptcy case is subject to examination by the Attorney General acting through the Office of the U.S. Trustee, the Office of the U.S. Attorney, and other offices and employees of the U.S. Department of Justice.

Make sure the court has your mailing address

The bankruptcy court sends notices to the mailing address you list on *Voluntary Petition* for *Individuals Filing for Bankruptcy* (Official Form 101). To ensure that you receive information about your case, Bankruptcy Rule 4002 requires that you notify the court of any changes in your address.

A married couple may file a bankruptcy case together—called a *joint case*. If you file a joint case and each spouse lists the same mailing address on the bankruptcy petition, the bankruptcy court generally will mail you and your spouse one copy of each notice, unless you file a statement with the court asking that each spouse receive separate copies.

Understand which services you could receive from credit counseling agencies

The law generally requires that you receive a credit counseling briefing from an approved credit counseling agency. 11 U.S.C. § 109(h). If you are filing a joint case, both spouses must receive the briefing. With limited exceptions, you must receive it within the 180 days **before** you file your bankruptcy petition. This briefing is usually conducted by telephone or on the Internet.

In addition, after filing a bankruptcy case, you generally must complete a financial management instructional course before you can receive a discharge. If you are filing a joint case, both spouses must complete the course.

You can obtain the list of agencies approved to provide both the briefing and the instructional course

from: http://justice.gov/ust/eo/hapcpa/ccde/cc_approved.html.

In Alabama and North Carolina, go to: <a href="http://www.uscourts.gov/FederalCourts/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Bankruptcy/Ban

If you do not have access to a computer, the clerk of the bankruptcy court may be able to help you obtain the list.

April 22-23, 2014 Page 183 of 596

Notice Required by 11 U.S.C. U.S.C. § 342(b) for Individuals Filing for Bankruptcy (Form 2010)

Instructions for Selected Forms

Instructions for Selected Forms

Schedule A/B: Property (Official Form 106A/B)

Schedule A/B: Property (Official Form 106A/B) lists property interests that are involved in a bankruptcy case. All individuals filing for bankruptcy must list everything they own or have a legal or equitable interest in. Legal or equitable interest is a broad term and includes all kinds of property interests in both tangible and intangible property, whether or not anyone else has an interest in that property.

The information in this form is grouped by category and includes several examples for many items. Note that those examples are meant to give you an idea of what to include in the categories. They are not intended to be complete lists of everything within that category. Make sure you list everything you own or have an interest in.

You must verify under penalty of perjury that the information you provide is complete and accurate. Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Understand the terms used in this form

Community property — Type of property ownership available in certain states for property owned by spouses and, in some instances, legal equivalents of spouses. Community property states and territories include Arizona, California, Idaho, Louisiana, Nevada, New Mexico, Puerto Rico, Texas, Washington, and Wisconsin.

Current value — In this form, report the current value of the property that you own in each category. Current value is sometimes called fair market value and, for this form, is the fair market value as of the date of the filing of the petition. Current value is how much the property is worth, which may be more or less than when you purchased the property. Property you own includes property you have purchased, even if you owe money on it, such as a home with a mortgage or an automobile with a lien.

Report the current value of the portion you own

For each question, report the current value of the portion of the property that you own. To do this, you would usually determine the current value of the entire property and the percentage of the property that you own. Multiply the current value of the property by the percentage that you own. Report the result where the form asks for *Current value of the portion you own*. For example:

- If you own a house by yourself, you own 100% of that house. Report the entire current value of the house.
- If you and a sister own the house equally, report 50% of the value of the house (or half of the value of the house).

In certain categories, current value may be difficult to figure out. When you cannot find the value from a reputable source (such as a pricing guide for your car), estimate the value and be prepared to explain how you determined it.

Schedule A/B: Property (Official Form 106A/B)

List items once on this form

List items only once on this form; do not list them in more than one category. List all real estate in Part 1 and other property in the other parts of the form.

Where you list similar items of minimal value (such as clothing), add the value of the items and report a total.

Be specific when you describe each item. If you have an item that you think could fit into more than one category, select the most suitable category and list the item there.

Separately describe and list individual items worth more than \$500.

Match the values to the other schedules

Make sure that the values you report on this form match the values you report on *Schedule D:*Creditors Who Have Claims Secured by Your

Property (Official Form 106D) and Schedule C:

The Property You Claim as Exempt (Official
Form 106C).

Schedule A/B: Property (Official Form 106A/B)

Schedule C: The Property You Claim as Exempt

(Official Form 106C)

How exemptions work

If you are an individual filing for bankruptcy, the law may allow you to keep some property, or it may entitle you to part of the proceeds if the property is sold after your case is filed. Property that the law permits you to keep is called *exempt* property. For example, exemptions may enable you to keep your home, a car, clothing, and household items.

Exemptions are not automatic. For property to be considered exempt, you must list the property on *Schedule C: The Property You Claim as Exempt* (Official Form 106C). If you do not list the property, the trustee may sell it and pay all of the proceeds to your creditors.

You may unnecessarily lose property if you do not claim exemptions to which you are entitled. You are strongly encouraged to hire a qualified attorney to advise you.

Determine which set of exemptions you will use

Before you fill out this form, you must learn which set of exemptions you can use. In general, exemptions are determined on a state-by-state basis. Some states permit you to use the exemptions provided by the Bankruptcy Code. 11 U.S.C. § 522.

The Bankruptcy Code provides that you use the exemptions in the law of the state where you had your legal home for 730 days before you file for bankruptcy. Special rules may apply if you did not have the same home state for 730 days before you file.

You may lose property if you do not use the best set of exemptions for your situation.

If your spouse is filing with you and you are filing in a state in which you may choose between state and federal sets of bankruptcy exemptions, you both must use the same set of exemptions.

Claiming exemptions

Using the property and values that you listed on *Schedule A/B: Property* (Official Form 106A/B) as your source, list on this form the property that you claim as exempt.

Listing the amount of each exemption

For each item of property you claim as exempt, you must specify the amount of the exemption you claim. Usually, a specific dollar amount is claimed as exempt, but in some circumstances, the amount of the exemption claimed might be indicated as 100% of fair market value. For example, a debtor might claim 100% of fair market value for an exemption that is unlimited in dollar amount, such as some exemptions for health aids.

Listing which laws apply

In the last column of the form, you must identify the laws that allow you to claim the property as exempt. If you have questions about exemptions, consult a qualified attorney.

Schedule C: The Property You Claim as Exempt (Official Form 106C)

Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D)

The people or organizations to whom you owe money are called your *creditors*. A *claim* is a creditor's right to payment. When you file for bankruptcy, the court needs to know who all your creditors are and what types of claims they have against you.

Typically in bankruptcy cases, there are more debts than assets to pay those debts. The court must know as much as possible about your creditors to make sure that their claims are properly treated according to the rules.

Creditors may have different types of claims:

- Secured claims. Report these on *Schedule D:* Creditors Who Have Claims Secured by Property (Official Form 106D).
- Unsecured claims. Report these on Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F).

If your debts are not paid, a creditor with a secured claim may be able to get paid from specific property in which that creditor has an interest, such as a mortgage or a lien. That property is sometimes called *collateral* for your debt and could include items such as your house, your car, or your furniture. Creditors with unsecured claims do not have rights against specific property.

Many creditors' claims have a specific amount, which you do not dispute. However, some claims are uncertain when you file for bankruptcy, or they become due only after you file. You must list the claims of all your creditors in your schedules, even if the claims are contingent, unliquidated, or disputed.

Claims may be contingent, unliquidated, or disputed

Claims may be:

- Contingent claims,
- Unliquidated claims, or
- Disputed claims.

A claim is *contingent* if you are not obligated to pay it unless a particular event occurs after you file for bankruptcy. For example, if you cosigned someone else's note, you may not have to pay unless that other person later fails to repay the loan.

A claim is *unliquidated* if the amount of the debt cannot be readily determined, such as by referring to an agreement or by a simple computation. An unliquidated claim is one for which there may be a definite liability but where the value has not been set. For instance, if you were involved in a car accident, the victim may have an unliquidated claim against you because the amount of damages has not been determined.

A claim is *disputed* if you disagree about whether you owe the debt. For instance, if a bill collector demands payment for a bill you believe you already fully paid, you may describe the claim as disputed.

A single claim can have one, more than one, or none of these characteristics.

On Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D), list all creditors who have a claim that is secured by your property.

Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D)

Do not leave out any secured creditors

In alphabetical order (as much as possible), list anyone who has judgment liens, garnishments, statutory liens, mortgages, deeds of trust, and other security interests against your property. When listing creditors who have secured claims, be sure to include all of them. For example, include the following:

- Your relatives or friends who have a lien or security interest in your property;
- Car or truck lenders, stores, banks, credit unions, and others who made loans to enable you to finance the purchase of property and who have a lien against that property;
- Anyone who has a mortgage or deed of trust on real estate that you own;
- Contractors or mechanics who have liens on property you own because they did work on the property and were not paid;
- Someone who won a lawsuit against you and has a judgment lien;
- Another parent or a government agency that has a lien for unpaid child support;
- Doctors or attorneys who have liens on the outcome of a lawsuit;
- Federal, state, or local government agencies such as the IRS that have tax liens against property for unpaid taxes; and
- Anyone who is trying to collect a secured debt from you, such as collection agencies and attorneys.

List the debt in Part 1 only once and list any others that should be notified about that debt in Part 2. For example, if a collection agency or an attorney is trying to collect from you for a debt you owe to someone else, list the person to whom you owe the debt in Part 1, and list the collection agency in Part 2. If you are not sure who the creditor is, list the person you are paying in Part 1 and list anyone else who has contacted you about this debt in Part 2.

If a creditor's full claim is more than the value of your property securing that claim—for instance, a car loan in an amount greater than the value of the car—the creditor's claim may be partly secured and partly unsecured. In that situation, list the claim only once on *Schedule D: Creditors Who Have Claims Secured by Property* (Official Form 106D). Do not repeat it on *Schedule E/F: Creditors Who Have Unsecured Claims* (Official Form 106E/F). List a creditor in *Schedule D* even if it appears that there is no value to support that creditor's secured claim.

Determine the unsecured portion of secured claims

To determine the amount of a secured claim, compare the amount of the claim to the value of your portion of the property that supports the claim. If that value is greater than the amount of the claim, then the entire amount of the claim is secured. But if that value is less than the amount of the claim, the difference is an *unsecured portion*. For example, if the outstanding balance of a car loan is \$10,000 and the car is worth \$8,000, the car loan has a \$2,000 unsecured portion.

If there is more than one secured claim against the same property, the claim that is entitled to be paid first must be subtracted from the property value to determine how much value remains for the next claim. For example, if a home worth \$300,000 has a first mortgage of \$200,000 and a second mortgage of \$150,000, the first mortgage would be fully secured, and there would be \$100,000 of property value for the second mortgage, which would have an unsecured portion of \$50,000.

	\$300,000	value of a home
<u>-</u>	\$200,000	first mortgage
	\$100,000	remaining property value
	\$150,000	second mortgage
_	\$100,000	remaining property value
	\$ 50,000	unsecured portion of second mortgage

Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D)

Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F)

The people or organizations to whom you owe money are called your *creditors*. A *claim* is a creditor's right to payment. When you file for bankruptcy, the court needs to know who all your creditors are and what types of claims they have against you.

Typically in bankruptcy cases, there are more debts than assets to pay those debts. The court must know as much as possible about your creditors to make sure that their claims are properly treated according to the rules.

Use Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F) to identify everyone who has an unsecured claim against you when you file your bankruptcy petition, unless you have already listed them on Schedule D: Creditors Who Have Claims Secured by Your Property (Official Form 106D).

Creditors may have different types of claims:

- **Secured claims**. Report these on *Schedule D: Creditors Who Have Claims Secured by Property* (Official Form 106D).
- Unsecured claims. Report these on Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F).

If your debts are not paid, creditors with secured claims may be able to get paid from specific property in which that creditor has an interest, such as a mortgage or a lien. That property is sometimes called *collateral* for your debt and could include items such as your house, your car, or your furniture. Creditors

with unsecured claims do not have rights against specific property.

Many creditors' claims have a specific amount, which you do not dispute. However, some claims are uncertain when you file for bankruptcy, or they become due only after you file. You must list the claims of all your creditors in your schedules, even if the claims are contingent, unliquidated, or disputed.

Claims may be contingent, unliquidated, or disputed

Claims may be:

- Contingent claims,
- Unliquidated claims, or
- Disputed claims.

A claim is *contingent* if you are not obligated to pay it unless a particular event occurs after you file for bankruptcy. For example, if you cosigned someone else's note, you may not have to pay unless that person later fails to repay the loan.

A claim is *unliquidated* if the amount of the debt cannot be readily determined, such as by referring to an agreement or by a simple computation. An unliquidated claim is one for which there may be a definite liability but where the value has not been set. For instance, if you were involved in a car accident, the victim may have an unliquidated claim against you because the amount of damages has not been set.

Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F)

A claim is *disputed* if you disagree about whether you owe the debt. For instance, if a bill collector demands payment for a bill you believe you already fully paid, you may describe the claim as disputed.

A single claim can have one, more than one, or none of these characteristics.

Creditors with unsecured claims do not have liens on or other security interests in your property. Secured creditors have a right to take property if you do not pay them. Common examples are lenders for your car, your home, or your furniture.

Do not leave out any unsecured creditors

List all unsecured creditors in each part of the form in alphabetical order as much as possible. Even if you plan to pay a creditor, you must list that creditor. When listing creditors who have unsecured claims, be sure to include all of them. For instance, include the following:

- Your relatives or friends to whom you owe money;
- Your ex-spouse, if you are still obligated under a divorce decree or settlement agreement to pay joint debts;
- A credit card company, even if you intend to fully pay your credit card bill;
- A lender, even if the loan is cosigned;
- Anyone who has a loan or promissory note that you cosigned for someone else;
- Anyone who has sued or may sue you because of an accident, dispute, or similar event that has occurred: or
- Anyone who is trying to collect a debt from you such as a bill collector or attorney.

Unsecured claims could be priority or nonpriority claims

What are priority unsecured claims?

In bankruptcy cases, *priority unsecured claims* are those debts that the Bankruptcy Code requires to be paid before most other unsecured claims are paid. The most common priority unsecured claims are certain income tax debts and past due alimony or child support. Priority unsecured claims include those you owe for:

- Domestic support obligations—If you owe domestic support to a spouse or former spouse; a child or the parent, legal guardian, or responsible relative of a child; or a governmental unit to whom such a domestic support claim has been assigned.

 11 U.S.C. § 507(a)(1).
- Taxes and certain other debts you owe the government—If you owe certain federal, state, or local government taxes, customs duties, or penalties.

 11 U.S.C. § 507(a)(8).
- Claims for death or personal injury that you caused while you were intoxicated—If you have a claim against you for death or personal injury that resulted from your unlawfully operating a motor vehicle or vessel while you were unlawfully intoxicated from alcohol, drugs, or another substance. This priority does not apply to claims for property damage.

 11 U.S.C. § 507(a)(10).

Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F)

Other:

- □ **Deposits by individuals**—If you received money from someone for the purchase, lease, or rental of your property or the use of your services but you never delivered or performed. For the debt to have priority, the property or services must have been intended for personal, family, or household use (only the first \$2,775* per person is a priority debt). 11 U.S.C. § 507(a)(7).
- □ Wages, salaries, and commissions—If you owe wages, salaries, and commissions, including vacation, severance, and sick leave pay and those amounts were earned within 180 days before you filed your bankruptcy petition or ceased business. In either instance, only the first \$12,475* per claim is a priority debt. 11 U.S.C. § 507(a)(4).
- □ Contributions to employee benefit plans—If you owe contributions to an employee benefit plan for services an employee rendered within 180 days before you file your bankruptcy petition, or within 180 days before your business ends. Count only the first \$12,475* per employee, less any amounts owed for wages, salaries, and commissions.

 11 U.S.C. § 507(a)(5).

What are nonpriority unsecured claims?

Nonpriority unsecured claims are those debts that generally will be paid after priority unsecured claims are paid. The most common examples of nonpriority unsecured claims are credit card bills, medical bills, and educational loans.

What if a claim has both priority and nonpriority amounts?

If a claim has both priority and nonpriority amounts, list that claim in Part 2 and show both priority and nonpriority amounts. Do not list it again in Part 3.

In Part 3, list all of the creditors you have not listed before. You must list every creditor that you owe, regardless of the amount you owe and even if you plan to pay a particular debt. If you do not list a debt, it may not be discharged.

What is needed for statistical purposes?

For statistical reasons, the court must collect information about some specific categories of unsecured claims.

The categories for priority unsecured claims are:

- Domestic support obligations
- Taxes and certain other debts you owe the government
- Claims for death or personal injury that you caused while you were intoxicated

Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F)

The categories for nonpriority unsecured claims are:

- **Student loans**—If you owe money for any loans that you used to pay for your education;
- Obligations arising out of a separation agreement or divorce that you did not report as priority claims—If you owe debts for separation or divorce agreements or for domestic support and you did not report those debts in Part 2; and
- Debts to pension or profit-sharing plans and other similar debts—If you owe money to a pension or profit-sharing plan.

Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F)

Schedule G: Executory Contracts and Unexpired Leases (Official Form 106G)

Use Schedule G: Executory Contracts and Unexpired Leases (Official Form 106G) to identify your ongoing leases and certain contracts. List all of your executory contracts and unexpired leases.

Executory contracts are contracts between you and someone else in which neither you nor the other party has performed all of the requirements by the time you file for bankruptcy. *Unexpired leases* are leases that are still in effect; the lease period has not yet ended.

You must list all agreements that may be executory contracts or unexpired leases, even if they are listed on *Schedule A/B: Property* (Official Form 106A/B), including the following:

- Residential leases (for example, a rental agreement for a place where you live or vacation, even if it is only a verbal or month-to-month arrangement);
- Service provider agreements (for example, contracts for cell phones and personal electronic devices);

- Internet and cable contracts;
- Vehicle leases;
- Supplier or service contracts (for example, contracts for lawn care or home alarm or security systems);
- Timeshare contracts or leases;
- Rent-to-own contracts;
- Employment contracts;
- Real estate listing agreements;
- Contracts to sell a residence, building, land, or other real property;
- Equipment leases;
- Leases for business or investment property;
- Supplier and service contracts for your business;
- Copyright and patent license agreements; and
- Development contracts.

Schedule H: Your Codebtors (Official Form 106H)

If you have any debts that someone else may also be responsible for paying, these people or entities are called *codebtors*. Use *Schedule H: Your Codebtors* (Official Form 106H) to list any codebtors who are responsible for any debts you have listed on the other schedules.

To help fill out this form, use both *Schedule D: Creditors Who Have Claims Secured by Property* (Official Form 106D) and *Schedule E/F: Creditors Who Have Unsecured Claims* (Official Form 106E/F).

List all of your codebtors and the creditors to whom you owe the debt. For example, if someone cosigned for the car loan that you owe, you must list that person on this form. If you are filing a joint case, do not list either spouse as a codebtor.

Other codebtors could include the following:

- Cosigner;
- Guarantor;
- Former spouse;
- Unmarried partner;
- Joint contractor; or
- Nonfiling spouse—even if the spouse is not a cosigner—where the debt is for necessities (such as food or medical care) if state law makes the nonfiling spouse legally responsible for debts for necessities.

Schedule H: Your Codebtors (Official Form 106H)

Schedule I: Your Income (Official Form 106I)

In *Schedule I: Your Income* (Official Form 106I), you will give the details about your employment and monthly income as of the date you file this form. If you are married and your spouse is living with you, include information about your spouse even if your spouse is not filing with you. If you are separated and your spouse is not filing with you, do not include information about your spouse.

How to report employment and income

If you have nothing to report for a line, write \$0.

In Part 1, line 1, fill in employment information for you and, if appropriate, for a non-filing spouse. If either person has more than one employer, attach a separate page with information about the additional employment.

In Part 2, give details about the monthly income you currently expect to receive. Show all totals as monthly payments, even if income is not received in monthly payments.

If your income is received in another time period, such as daily, weekly, quarterly, annually, or irregularly, calculate how much income would be by month, as described below.

If either you or a non-filing spouse has more than one employer, calculate the monthly amount for each employer separately, and then combine the income information for all employers for that person on lines 2-7. One easy way to calculate how much income per month is to total the payments earned in a year, then divide by 12 to get a monthly figure. For example, if you are paid seasonally, you would simply divide the amount you expect to earn in a year by 12 to get the monthly amount

Below are other examples of how to calculate monthly amount.

Example for weekly payments:

If you are paid \$1,000 every week, figure your monthly income in this way:

\$1,000		income every week
Χ	52	number of pay periods in the year
\$52,	000	total income for the year

\$52,000 (income for year) = \$4,333 monthly income 12 (number of months in year)

Example for bi-weekly payments:

If you are paid \$2,500 every other week, figure your monthly income in this way:

\$2,500 income every other week

X 26 number of pay periods in the year

\$65,000 total income for the year

\$65,000 (income for year) = \$5,417 monthly income
12 (number of months in year)

page **26**

Schedule I: Your Income (Official Form 106I)

April 22-23, 2014 Page 197 of 596

Example for daily payments:

If you are paid \$75 a day and you work about 8 days a month, figure your monthly income in this way:

	\$75	income a day
X	96	days a year
	\$7,200	total income for the year

\$7,200 (income for year) = \$600 monthly income
12 (number of months in year)

or this way:

	\$600	income for the month
X	8	payments a month
	\$75	income a day

Example for quarterly payments:

If you are paid \$15,000 every quarter, figure your monthly income in this way:

V	\$15,000	income every quarter
<u>X</u>	<u>4</u>	pay periods in the year
	\$60,000	total income for the year

\$60,000 (income for year) = \$5,000 (number of months in year)
12 monthly income

Example for irregular payments:

If you are paid \$4,000 8 times a year, figure your monthly income in this way:

\$4,000 income a payment

X 8 payments a year

\$32,000 income for the year

\$32,000 (income for year) = \$2,667 monthly income 12 (number of months in year)

In Part 2, line 11, fill in amounts that other people provide to pay the expenses you list on *Schedule J: Your Expenses*. For example, if you and a person to whom you are not married pay all household expenses together and you list all your joint household expenses on Schedule J, you must list the amounts that person contributes monthly to pay the household expenses on line 11. If you have a roommate and you divide the rent and utilities, do not list the amounts your roommate pays on line 11 if you have listed only your share of those expenses on Schedule J. Do not list on line 11 contributions that you already disclosed elsewhere on the form.

Note that the income you report on *Schedule I* may be different from the income you report on other bankruptcy forms. For example, the Chapter 7 Statement of Your Current Monthly Income (Official Form 108-1), Chapter 11 Statement of Your Current Monthly Income (Official Form 109), and the *Chapter 13* Statement of Your Current Monthly Income and Calculation of Commitment Period (Official Form 110-1) all use a different definition of income and apply that definition to a different period of time. Schedule I asks about the income that you are now receiving, while the other forms ask about income you received in the applicable time period before filing. So the amount of income reported in any of those forms may be different from the amount reported here.

If, after filing Schedule I, you need to file an estimate of income in a chapter 13 case for a date after your bankruptcy, you may complete a supplemental Schedule I. To do so you must check the "supplement" box at the top of the form and fill in the date.

page **27**

Schedule I: Your Income (Official Form 106I)

April 22-23, 2014 Page 198 of 596

Schedule J: Your Expenses (Official Form 106J)

Schedule J: Your Expenses (Official Form 106J) provides an estimate of the monthly expenses, as of the date you file for bankruptcy, for you, your dependents, and the other people in your household whose income is included on Schedule I: Your Income (Official Form 106I).

If you are married and are filing individually, include your non-filing spouse's expenses unless you are separated.

If you are filing jointly and Debtor 1 and Debtor 2 keep separate households, fill out a separate *Schedule J* for each debtor. Check the box at the top of page 1 of the form for Debtor 2 to show that a separate form is being filed.

Do not include expenses that other members of your household pay directly from their income if you did not include that income on *Schedule I*. For example, if you have a roommate and you divide the rent and utilities and you have not listed your roommate's contribution to household expenses in line 11 of *Schedule I*, you would list only your share of these expenses on *Schedule J*.

Show all totals as monthly payments. If you have weekly, quarterly, or annual payments,

calculate how much you would spend on those items every month.

Do not list as expenses any payments on credit card debts incurred before filing bankruptcy.

Do not include business expenses on this form. You have already accounted for those expenses as part of determining net business income on *Schedule I*.

On line 20, do not include expenses for your residence or for any rental or business property. You have already listed expenses for your residence on lines 4 and 5 of this form. You listed the expenses for your rental and business property as part of the process of determining your net income from that property on *Schedule I* (line 8a).

If you have nothing to report for a line, write \$0.

If, after filing *Schedule J*, you need to file an estimate of expenses in a chapter 13 case for a date after your bankruptcy, you may complete a supplemental *Schedule J*. To do so you must check the "supplement" box at the top of the form and fill in the date.

Summary of Your Assets and Liabilities and Certain Statistical Information (Official Form 106Sum)

When you file for bankruptcy, you must summarize certain information from the following forms:

- *Schedule A/B: Property* (Official Form 106A/B)
- Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D)
- Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F)
- Schedule I: Your Income (Official Form 106I)
- Schedule J: Your Expenses (Official Form 106J)
- Chapter 7 Statement of Your Current Monthly Income (Official Form 108-1), Chapter 11 Statement of Your Current Monthly Income (Official Form 109), or Chapter 13 Statement of Your Current Monthly Income and Calculation of Commitment Period (Official Form 110-1)

After you fill out all of the forms, complete *Summary of Your Assets and Liabilities and Certain Statistical Information* (Official Form 106Sum) to report the totals of certain information that you listed in the forms.

If you are filing an amended version of any of these forms at some time after you file your original forms, you must fill out a new *Summary* to ensure that your information is up to date and you must check the box at the top.

Statement of Financial Affairs for Individuals Filing for Bankruptcy (Official Form 107)

Your Statement of Financial Affairs for Individuals Filing for Bankruptcy, provides a summary of your financial history over certain periods of time before you file for bankruptcy. If you are an individual in a bankruptcy case, you must fill out this statement.

11 U.S.C. § 521(a) and Bankruptcy Rule 1007(b)(1).

If you are married and your spouse is not filing this case with you, you need only provide information on this form about your spouse if you are filing under chapter 12 or chapter 13 and are not separated from your spouse.

If you are in business as a sole proprietor, partner, family farmer, or self-employed

professional, you must provide the information about all of your business and personal financial activities.

Although this statement may ask you questions that are similar to some questions on the schedules, you must fill out all of the forms completely to protect your legal rights.

Understand the terms used in this form

Legal equivalent of a spouse — A person whom applicable nonfederal law recognizes as having a relationship with the debtor that grants legal rights and responsibilities equivalent, in whole or in part, to those granted to a spouse.

Statement of Financial Affairs for Individuals Filing for Bankruptcy (Official Form 107)

Chapter 7 Statement of Your Current Monthly Income and Means Test Calculation (Official Forms 108-1, 108A-1Supp, and 108-2)

If you are filing under chapter 11, 12, or 13, do not fill out this form.

Official Forms 108-1 and 108-2 determine whether your income and expenses create a presumption of abuse that may prevent you from obtaining relief from your debts under chapter 7 of the Bankruptcy Code. Chapter 7 relief can be denied to a person who has primarily consumer debts if the court finds that the person has enough income to repay creditors an amount that, under the Bankruptcy Code, would be a sufficient portion of their claims.

You must file Chapter 7 Statement of Your Current Monthly Income (Official Form 108–1) if you are an individual filing for bankruptcy under chapter 7. This form will determine your current monthly income and compare whether your income is more than the median income for households of the same size in your state. If your income is not above the median, there is no presumption of abuse and you will not have to fill out the second form.

Similarly, Statement of Exemption from *Presumption of Abuse Under § 707(b)(2)* (Official Form 108A-1Supp) determines whether you may be exempted from the presumption of abuse because you do not have primarily consumer debts or because you have provided certain military or homeland defense services. If one of these exemptions applies, you should file a supplement, Form 108A-1Supp, and verify the supplement by completing Part 3 of Form 108A-1. If you qualify for an exemption, you are not required to fill out any part of Form 108A-1 other than the verification. If the exemptions do not apply, you should complete all of the parts of Form 108A-1 and file it without the supplemental form.

If your income is above the median, you must file the second form, Chapter 7 Means Test Calculation (Official Form 108 –2). The calculations on this form—sometimes called the *Means Test*—reduce your income by living expenses and payment of certain debts, resulting in an amount available to pay other debts.

If this amount is high enough, it will give rise to a presumption of abuse. A presumption of abuse does not mean you are actually trying to abuse the bankruptcy system. Rather, the presumption simply means that you are presumed to have enough income that you should not be granted relief under chapter 7. You may overcome the presumption by showing special circumstances that reduce your income or increase your expenses.

If you cannot obtain relief under chapter 7, you may be eligible to continue under another chapter of the Bankruptcy Code and pay creditors over a period of time.

Read each question carefully. You may not be required to answer every question on this form. For example, your military status may

Chapter 7 Statement of Current Monthly Income; Means Test Calculation (Official Forms 108-1, 108-2)

determine whether you must fill out the entire form. The instructions will alert you if you may skip questions.

If you have nothing to report for a line, write \$0.

If you and your spouse are filing together, you and your spouse may file a single statement. However, if an exclusion in Parts 1 or 2 applies to either of you, separate statements may be required. 11 U.S.C. § 707(b)(2)(C).

Information for completing the forms

To fill out several lines of the forms, you must look up information provided on websites or from other sources. For information to complete line 13 of Form 108-1 and lines 6-15, 30, and 36 of Form 108-2, go

to: www.justice.gov/ust/eo/bapcpa/meanstesting.ht m

For line 36 of Form 108-2, also go to that site unless your case is filed in Alabama or North Carolina. If your case is filed in one of those states, go to:

www.uscourts.gov/FederalCourts/Bankruptcy/BankruptcyResources/AdministrativeExpensesMultiplier.a spx

For the *Bankruptcy Basics* information referred to on line 36 of Form 22A-2, go to: www.uscourts.gov/FederalCourts/Bankruptcy/Bankruptcy/Bankruptcy/Basics.aspx.

If you do not have a computer with internet access, you may be able to use a public computer at the bankruptcy clerk's office or at a public library.

April 22-23, 2014 Page 203 of 596

Chapter 11 Statement of Your Current Monthly

Income (Official Form 109)

If you are filing under chapter 7, 12, or 13, do not fill out this form.

You must file the *Chapter 11 Statement of Your Current Monthly Income* (Official Form 109) if you are an individual filing for bankruptcy under chapter 11.

If you have nothing to report for a line, write \$0.

Chapter 11 Statement of Your Current Monthly Income (Official Form 109)

Chapter 13 Statement of Your Current Monthly Income, Calculation of Commitment Period and Chapter 13 Calculation of Your Disposable Income

(Official Forms 110-1 and 110-2)

If you are filing under chapter 7, 11, or 12, do not fill out this form.

Official Forms 110—1 and 110—2 determine the commitment period for your payments to creditors, how the amount you may be required to pay to creditors is established, and, in some situations, how much you must pay.

You must file the Chapter 13 Statement of Your Current Monthly Income and Calculation of Commitment Period (Official Form 110—1) if you are an individual and you are filing under chapter 13. This form will report your current monthly income and determine whether your income is at or below the median income for households of the same size in your state. If your income is equal to or less than the median, you will not have to fill out the second form. Form 110-1 also will determine your applicable commitment period—the time period for making payments to your creditors, unless the court orders otherwise.

If your income is above the median, you must file the second form, *Chapter13 Calculation of Your Disposable Income* (Official Form 110—2). The calculations on this form—

sometimes called the *Means Test*—reduce your income by living expenses and payment of certain debts, resulting in an amount available to pay unsecured debts. Your chapter 13 plan may be required to provide for payment of this amount toward unsecured debts.

Read each question carefully. You may not be required to answer every question on this form. The instructions will alert you if you may skip questions.

Generally, if you and your spouse are filing together, you should file one statement together.

Information for completing the forms

To fill out several lines of the forms, you must look up information provided on websites or from other sources. For information to complete line 16c of Form 110-1 and lines 6-15, 30, and 36 of Form 110-2, go

to: www.justice.gov/ust/eo/bapcpa/meanstesting.htm.

If you do not have a computer with internet access, you may be able to use a public computer at the bankruptcy clerk's office or at a public library.

Chapter 13 Statement of Current Monthly Income, Calculation of Commitment Period and Chapter 13 Calculation of Your Disposable Income (Official Forms 110-1, 110-2)

page 34

April 22-23, 2014 Page 205 of 596

Statement of Intention for Individuals Filing Under Chapter 7 (Official Form 112)

If you are filing under chapter 11, 12, or 13, do not fill out this form.

If you are an individual filing under chapter 7, you must fill out the *Statement of Intention for Individuals Filing Under Chapter 7* (Official Form 112) if:

- creditors have claims secured by your property, or
- you have leased personal property and the lease has not expired.

The Bankruptcy Code requires you to state your intentions about such claims and provides for early termination of the automatic stay as to personal property if the statement is not timely filed. The same early termination of the automatic stay applies to any unexpired lease of personal property unless you state that you intend to assume the unexpired lease if the trustee does not do so.

To help fill out this form, use the information you have already provided on the following forms:

- Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D),
- Schedule C: The Property You Claim as Exempt (Official Form 106C), and
- Schedule G: Executory Contracts and Unexpired Leases (Official Form 106G).

Explain what you intend to do with your property that is collateral for a claim

If you have property that is collateral (or security) for a claim, you must state what you intend to do with that property.

You may choose either to surrender the property to the creditor, or retain the property. Below is more information about each of these options.

You may surrender the property to the creditor. If you surrender the property to the creditor, your bankruptcy discharge will protect you from any claim for the difference between what you owe the creditor and what the creditor receives from a sale of the property, unless the court determines that the debt is nondischargeable.

You may want to retain the property. If you want to retain your secured personal property, you may be able to reaffirm the debt, redeem the property, or take other action (for example, avoid a lien using 11 U.S.C. § 522(f)).

decide to remain legally obligated to pay a debt so that you can keep the property securing the debt. This is called *reaffirming a debt*. You may reaffirm the debt in full on its original terms or you and the creditor may agree to change the terms. For example, if you want to keep your car, you may reaffirm a car loan, stating that you will continue to make monthly payments for it. Only reaffirm those debts that you are confident you can repay. You may seek to reaffirm the debt if you sign a *Reaffirmation Agreement*, which is a contract between you and a creditor, and

Statement of Intention for Individuals Filing Under Chapter 7 (Official Form 112)

you follow the proper procedure for the *Reaffirmation Agreement*. 11 U.S.C. § 524. The procedure is explained in greater detail in the Disclosures that are part of the reaffirmation documents.

- You may be able to redeem your property. 11 U.S.C. § 722. You can redeem property only if all of the following apply:
 - ☐ The property secures a debt that is a consumer debt — you incurred the debt primarily for personal, family, or household use.
 - ☐ The property is *tangible personal*property the property is physical, such as furniture, appliances, and cars.
 - ☐ You are either claiming the property as exempt or the trustee has abandoned it.

To obtain court authorization to redeem your property, you must file a motion with the court. If the court grants your motion, you pay the creditor the value of the property or the amount of the claim, whichever is less. The payment will be a single lump-sum payment.

Explain what you intend to do with your leased personal property

If you lease personal property such as your car, you may be able to continue your lease if the trustee does not assume the lease. To continue your lease, you can write to the lessor that you want to assume your lease. The creditor may, at its option, notify you that it is willing to have you assume the lease and may condition the assumption on cure of any outstanding default. If the lessor notifies you that it is willing to have you assume the lease, you must write to the lessor within 30 days stating that you assume the lease. 11 U.S.C. § 365(p)(2).

File the *Statement of Intention* before the deadline

You must file this form either within 30 days after you file your bankruptcy petition or by the date set for the meeting of creditors, whichever is earlier. You must also deliver copies of this statement to the creditors and lessors you listed on the form. Bankruptcy Rule 1007(b)(2).

If two married people are filing together in a joint case, both are equally responsible for supplying correct information. Both debtors must sign and date the form.

Statement of Intention for Individuals Filing Under Chapter 7 (Official Form 112)

Application for Individuals to Pay the Filing Fee in Installments (Official Form 103A)

If you cannot afford to pay the full filing fee when you first file for bankruptcy, you may pay the fee in installments. However, in most cases, you must pay the entire fee within 120 days after you file, and the court must approve your payment timetable. Your debts will not be discharged until you pay your entire fee.

Do not file this form if you can afford to pay your full fee when you file.

If you are filing under chapter 7 and cannot afford to pay the full filing fee at all, you may be qualified to ask the court to waive your filing fee.

See Application to Have Your Chapter 7 Filing Fee Waived (Official Form 103B).

If a bankruptcy petition preparer helped you complete this form, make sure that person fills out the *Bankruptcy Petition Preparer's Notice*, *Declaration, and Signature* (Official Form 119); include a copy of it when you file this application.

Application for Individuals to Pay the Filing Fee in Installments (Official Form 103A)

Application to Have the Chapter 7 Filing Fee

Waived (Official Form 103B)

The fee for filing a bankruptcy case under chapter 7 is \$335. If you cannot afford to pay the entire fee now in full or in installments within 120 days, use this form. If you can afford to pay your filing fee in installments, see *Application for Individuals to Pay the Filing Fee in Installments* (Official Form 103A).

If you file this form, you are asking the court to waive your fee. After reviewing your application, the court may waive your fee, set a hearing for further investigation, or require you to pay the fee in installments or in full.

For your fee to be waived, all of these statements must be true:

- You are filing for bankruptcy under chapter 7.
- You are an individual.
- The total combined monthly income for your family is less than 150% of the official poverty guideline last published by the U.S. Department of Health and Human Services (DHHS). (For more information about the guidelines, go to http://www.uscourts.gov.)
- You cannot afford to pay the fee in installments.

Your family includes you, your spouse, and any dependents listed on *Schedule I*. Your family may be different from your *household*, referenced on *Schedules I* and *J*. Your household may include your unmarried partner and others who live with you and with whom you share income and expenses.

If a bankruptcy petition preparer helped you complete this form, make sure that person fills out *Bankruptcy Petition Preparer's Notice*, *Declaration, and Signature* (Official Form 119); include a copy of it when you file this application.

If you have already completed the following forms, the information on them may help you when you fill out this application:

- *Schedule A/B: Property* (Official Form 106A/B)
- Schedule I: Your Income (Official Form 106I)
- Schedule J: Your Expenses (Official Form 106J)

Application to Have the Chapter 7 Filing Fee Waived (Official Form 103B)

For Individual Chapter 11 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims Against You and Are Not Insiders (Official Form 104)

If you are filing under chapter 7, 12, or 13, do not fill out this form.

The people or organizations to whom you owe money are called your *creditors*. A *claim* is a creditor's right to payment. If you are an individual filing for bankruptcy under chapter 11, you must fill out *For Individual Chapter 11 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims Against You and Are Not Insiders* (Official Form 104).

Creditors may have different types of claims:

- Secured claims, or
- Unsecured claims.

If your debts are not paid, creditors with secured claims may be able to get paid from specific property in which that creditor has an interest, such as a mortgage or a lien. If a creditor has security interest in your property, but the value of the property available to pay the creditor is less than the amount you owe the creditor, the creditor has both a secured and an unsecured claim against you. The amount of the unsecured claim is the total claim minus the value of the property that is available to pay the creditor.

Generally, creditors with unsecured claims do not have rights against specific property, or the specific property in which the creditor has rights is not worth enough to pay the creditor in full. For example, if you owe a creditor \$30,000 for your car and the creditor has a security interest in your car but the car is worth only \$20,000, the creditor has a \$20,000 secured claim and a \$10,000 unsecured claim.

_	\$30,000 \$20,000	Total amount you owe creditor Amount your car is worth (amount of secured claim)
	\$10.000	Amount of unsecured claim

Many claims have a specific amount, and you clearly owe them. However, some claims are uncertain when you file for bankruptcy, or they become due only after you file. You must include such claims when listing your 20 largest unsecured claims on this list.

Claims may be contingent, unliquidated, or disputed.

The form asks you to identify claims that are:

- Contingent claims,
- Unliquidated claims, or
- Disputed claims.

A claim is *contingent* if you are not obligated to pay it unless a particular event occurs after you file for bankruptcy. You owe a contingent claim, for example, if you cosigned someone else's loan. You may not have to pay unless that person later fails to repay the loan.

Individual Chapter 11 Cases: Creditors Who Have 20 Largest Unsecured Claims (Official Form 104)

A claim is *unliquidated* if the amount of the debt cannot be readily determined, such as by referring to an agreement or by a simple computation. An unliquidated claim is one for which there may be a definite liability but where the amount has not been set. For instance, if you were involved in a car accident, the victim may have an unliquidated claim against you because the amount of damages has not been set.

A claim is *disputed* if you do not agree that you owe the debt. For instance, your claim is disputed if a bill collector demands payment for a bill you believe you already fully paid.

A single claim can have one, more than one, or none of these characteristics.

On this form, list the creditors with the 20 largest unsecured claims who are not insiders

You must file this form when you file your chapter 11 bankruptcy case with the court.

When you list the 20 largest unsecured creditors, include all unsecured creditors, except for the following two types of creditors, even if you plan to pay them. Do not include:

- Anyone who is an *insider*. *Insiders* include relatives; general partners of you or your relatives; corporations of which you are an officer, director, or person in control; and any managing agent. 11 U.S.C. § 101(31).
- Secured creditors, unless the unsecured claim resulting from inadequate collateral value places the creditor among the holders of the 20 largest unsecured claims.

Make sure that all of the creditors listed on this form are also listed on either *Schedule D: Creditors Who Have Claims Secured by Property* (Official Form 106D) or *Schedule E/F: Creditors Who Have Unsecured Claims* (Official Form 106E/F).

On the form, you will fill in what the claim is for. Examples include trade debts, bank loans, professional services, and government contracts.

Individual Chapter 11 Cases: Creditors Who Have 20 Largest Unsecured Claims (Official Form 104)

Glossary

Glossary page 40

April 22-23, 2014 Page 212 of 596

Definitions of Some Terms Used in the Forms for Individuals Filing for Bankruptcy

Here are definitions of some of the important terms used in the forms for individuals who are filing for bankruptcy. See *Bankruptcy Basics* (www.uscourts.gov/FederalCourts) for more information about filing for bankruptcy and other important terms you should know. These definitions are intended only to provide guidance. They are not a substitute for legal advice.

Annuity — A contract for the periodic payment of money to you, either for life or for a number of years.

Bankruptcy petition preparer — A person or business, other than a lawyer or someone who works for a lawyer, that charges a fee to prepare bankruptcy documents. Under your direction and control, the bankruptcy petition preparer generates bankruptcy forms for you to file by typing them. Because they are not attorneys, they cannot give legal advice or represent you in bankruptcy court. Also called *typing services*.

Business debt — A debt that you incurred to obtain money for a business or investment or incurred through the operation of the business or investment.

Claim — A creditor's right to payment, even if contingent, disputed, unliquidated, or unmatured.

Codebtor — A person or entity that may also be responsible for paying a claim against the debtor.

Collateral — Specific property subject to a lien from which a creditor may be paid ahead of other creditors without liens on that property. Includes a mortgage, security interest, judgment lien, statutory lien, or other lien.

Community property — A type of property ownership available in certain states for property owned by spouses and, in some instances, legal equivalents of spouses. Community property states and territories include Arizona, California, Idaho, Louisiana, Nevada, New Mexico, Puerto Rico, Texas, Washington, and Wisconsin.

Consumer debt — A debt you incurred primarily for a personal, family, or household purpose.

Contingent claim — A debt you are not obligated to pay unless a particular event occurs after you file for bankruptcy. You owe a contingent claim, for example, if you cosigned someone else's loan. You may not have to pay unless that person later fails to repay the loan.

Creditor matrix or mailing matrix — A list of names and addresses of all of your creditors, formatted as a mailing list according to instructions from the bankruptcy court in which you file.

Creditor — A person or organization to whom you owe money or who claims that you owe it money.

Current value, fair market value, or value —

The amount property is worth, which may be more or less than when you purchased the property. Absent specific instruction, the value should be the price that could be realized from a cash sale or liquidation without duress within a reasonable time. See the instructions for specific forms regarding whether the value requested is as of the date of the filing of the petition, the date you complete the form, or some other date.

Debtor 1 — A debtor filing alone or one person in married couple who is filing a bankruptcy case with a spouse. The same person retains this designation in all of the forms.

Debtor 2 — A second person in a married couple who is filing a bankruptcy case with a spouse.

Dependent — A person who is economically dependent on you regardless of whether the person can be claimed as a dependent on your federal tax return. However, *Chapter 7 Means Test Calculation* (Official Form 108-2) and *Chapter 13 Calculation of Your Disposable Income* (Official Form 110-2) use the term in a more limited way. See the instructions on those forms.

Discharge — A discharge in bankruptcy relieves you after your bankruptcy case is over from having to pay debts that you owed before you filed your bankruptcy case. Most debts are covered by the discharge, but not all. (The instruction booklet explains more about common debts that are not discharged in bankruptcy.) Only your personal liability is removed by the discharge.

Disputed claim — A debt you do not agree that you owe. For instance, your claim is disputed if a bill collector demands payment for a bill you believe you already fully paid.

Eviction judgment — A judgment for possession that your landlord has obtained in an eviction, unlawful detainer action, or similar proceeding.

Executory contract — A contract between you and someone else in which both of you still have obligations to perform under the contract at the time you file for bankruptcy.

Exempt property — Property, or the value of a portion of it, that the law allows you to keep for your use rather than surrender it for the payment of your debts, provided that you follow the correct procedure to claim the exemption.

Garnishment — A procedure by which a creditor can reach money of yours that is in the hands of a third party to satisfy a debt. Garnishments are sometimes used by creditors to obtain money from your wages or bank account.

Definitions Used in the Forms for Individuals Filing for Bankruptcy

Individual debtor — A human being who is filing for bankruptcy either alone or with a spouse, whether or not the individual owns a business.

Joint case — A single case filed by a married couple.

Judgment lien — A lien that arises as a result of a judgment against you.

Legal equivalent of a spouse — A person recognized by applicable nonfederal law as having a relationship with the debtor that grants legal rights and responsibilities equivalent, in whole or in part, to those granted to a spouse.

Legal or equitable interest — A broad term that includes all kinds of property interests in both tangible and intangible property, whether or not anyone else has an interest in that property.

Negotiable instrument — A financial instrument that you can transfer to someone by signing or delivering it, including personal checks, cashiers' checks, promissory notes, and money orders.

Non-individual debtor — A debtor that is not a human being — for example, an artificial entity such as a corporation, partnership, or limited liability company (LLC).

Non-negotiable instrument — A financial instrument that you cannot transfer to someone by signing or delivering it.

Nonpriority unsecured claim — A debt that generally will be paid after priority unsecured claims are paid. The most common examples are credit card bills, medical bills, and educational loans.

Payment advice — A statement such as a pay stub or earnings statement from your employer that shows all earnings and deductions from your pay.

Presumption of abuse — A rebuttable legal presumption that you have too much income after allowed expenses to be granted relief under chapter 7.

Priority unsecured claim — A debt that the Bankruptcy Code requires to be paid before most other unsecured claims are paid. The most common examples are certain income tax debts and past due alimony or child support.

Property you own — Includes property you have purchased, even if you owe money on it, such as a home with a mortgage or an automobile with a lien.

Reaffirming a debt — Agreeing to repay a debt that would otherwise be discharged by entering into a new written agreement with the creditor. A reaffirmation agreement may allow you to keep property that a creditor has the right to take from you because it secures the debt being reaffirmed. For a reaffirmation agreement to be effective, there are many procedural and legal requirements that must be satisfied during the bankruptcy case.

Secured claim — A claim that may be satisfied in whole or in part either

- by a charge against or an interest in specific property of the debtor, or
- by a right of setoff.

Common examples of creditors who have secured claims are lenders from your car, your home, or your furniture.

Sole proprietorship — A business you own as an individual that is not a separate legal entity such as a corporation, partnership, or LLC. Sole proprietors must use the bankruptcy forms that are numbered in the 100 series.

Statutory lien — A lien that arises as a result of a statute.

Unexpired lease —A lease that is in effect at the time you filed for bankruptcy.

Unliquidated claim — A debt with an amount cannot be readily determined, such as by referring to an agreement or by a simple computation. An unliquidated claim is one for which there may be a definite liability but where the value has not been set. For instance, if you were involved in a car accident, the victim may have an unliquidated claim against you because the amount of damages has not been determined.

Unsecured claim — A claim held by a creditor who does not have security interest in or other lien on your property or a right of setoff.

You — A debtor filing alone or one person in married couple who is filing a bankruptcy case with a spouse.

Definitions Used in the Forms for Individuals Filing for Bankruptcy

Appendix B Forms for Publication

April 22-23, 2014 Page 217 of 596

April 22-23, 2014 Page 218 of 596

United States Bankruptcy Court

			District Of		_
In re	Debtor	_ ,			
				Case No Chapter	

GENERAL POWER OF ATTORNEY

[Abrogated]

April 22-23, 2014 Page 219 of 596

April 22-23, 2014 Page 220 of 596

Official Form 11A

2015 COMMITTEE NOTE

The form is abrogated. Former Official Form 11A, although abrogated as an Official Form, continues to be available as a Director's Procedural Form.

Parties routinely modify the General Power of Attorney form to conform to state law, the needs of the case, or local practice. The exact language of the form is not needed. The proposed amendment to Rule 9009, however, restricts alteration of the Official Forms, except as provided in the rules or in a particular Official Form.

The Director's Procedural Forms are issued by the Director of the Administrative Office pursuant to Rule 9009 as an accommodation for the courts and parties. The procedural forms may be altered as needed and their use is not mandatory, unless required by local rule.

April 22-23, 2014 Page 221 of 596

April 22-23, 2014 Page 222 of 596

United States Bankruptcy Court

			District Of		
In re	Debtor	,			
				Case No Chapter	

SPECIAL POWER OF ATTORNEY

[Abrogated]

April 22-23, 2014 Page 223 of 596

April 22-23, 2014 Page 224 of 596

Official Form 11B

2015 COMMITTEE NOTE

The form is abrogated. Former Official Form 11B, although abrogated as an Official Form, continues to be available as a Director's Procedural Form.

Parties routinely modify the Special Power of Attorney form to conform to state law, the needs of the case, or local practice. The exact language of the form is not needed. The proposed amendment to Rule 9009, however, restricts alteration of the Official Forms, except as provided in the rules or in a particular Official Form.

The Director's Procedural Forms are issued by the Director of the Administrative Office pursuant to Rule 9009 as an accommodation for the courts and parties. The procedural forms may be altered as needed and their use is not mandatory, unless required by local rule.

April 22-23, 2014 Page 225 of 596

April 22-23, 2014 Page 226 of 596

Fill in this information to iden	tify your case:			
Debtor 1		Check if this is	••	
First Name Debtor 2	Middle Name Last Name	———— An amende		
(Spouse, if filing) First Name	Middle Name Last Name		•	-petition chapter 13
United States Bankruptcy Court for t			as of the following	
Case number(If known)		MM / DD / Y	YYYY	
Official Form 106J				
Schedule J: Y	our Expenses			12/15
	s possible. If two married people are fili eeded, attach another sheet to this form ion.			
Part 1: Describe Your I	Household			
1. Is this a joint case?				
□ No. Go to line 2.□ Yes. Does Debtor 2 live in	n a separate household?			
☐ No				
Yes. Debtor 2 mus	st file Official Forms 106J-2, Expenses for	Separate Household of Debtor 2.		
 Do you have dependents? Do not list Debtor 1 and Debtor 2. 	☐ No☐ Yes. Fill out this information for	Dependent's relationship to Debtor 1 or Debtor 2	Dependent's age	Does dependent live with you?
Do not state the dependents'	each dependent			☐ No
names.				☐ Yes
				☐ Yes
				☐ No
				☐ Yes
				□ No
				Yes
				□ No □ Yes
Do your expenses include expenses of people other that yourself and your dependent.				
	ngoing Monthly Expenses			
	our bankruptcy filing date unless you a bankruptcy is filed. If this is a supplem	-		
	non-cash government assistance if you	u know the value of		
such assistance and have inclu	uded it on Schedule I: Your Income (Offi	icial Form B 6l.)	Your expe	nses
 The rental or home ownersh any rent for the ground or lot. 	nip expenses for your residence. Include		4. \$	
If not included in line 4:				
4a. Real estate taxes				
4b. Property, homeowner's,			_	
	pair, and upkeep expenses		4c. \$	
4d. Homeowner's association	on or condominium dues		4d. \$	

First Name Middle Name Last Name

Case number (if known)______

			Your expenses
5.	Additional mortgage payments for your residence, such as home equity loans	5.	\$
	Utilities:		
б.		60	¢
	6a. Electricity, heat, natural gas	6a.	\$
	6b. Water, sewer, garbage collection	6b.	\$
	6c. Telephone, cell phone, Internet, satellite, and cable services	6c.	\$
	6d. Other. Specify:	6d.	\$
7.	Food and housekeeping supplies	7.	\$
8.	Childcare and children's education costs	8.	\$
9.	Clothing, laundry, and dry cleaning	9.	\$
10.	Personal care products and services	10.	\$
11.	Medical and dental expenses	11.	\$
12.	Transportation. Include gas, maintenance, bus or train fare. Do not include car payments.	12.	\$
13.	Entertainment, clubs, recreation, newspapers, magazines, and books	13.	\$
14.	Charitable contributions and religious donations	14.	\$
15.	Insurance. Do not include insurance deducted from your pay or included in lines 4 or 20.		
	15a. Life insurance	15a.	\$
	15b. Health insurance	15b.	\$
	15c. Vehicle insurance	15c.	\$
	15d. Other insurance. Specify:	15d.	\$
	outer insurance. Openly.	150.	Ψ
16.	Taxes. Do not include taxes deducted from your pay or included in lines 4 or 20. Specify:	16.	\$
17.	Installment or lease payments:		
	17a. Car payments for Vehicle 1	17a.	\$
	17b. Car payments for Vehicle 2	17b.	\$
	17c. Other. Specify:	17c.	\$
	17d. Other. Specify:	17d.	\$
18.	Your payments of alimony, maintenance, and support that you did not report as deducted from		—
	your pay on line 5, Schedule I, Your Income (Official Form 6I).	18.	\$
19.	Other payments you make to support others who do not live with you.		
	Specify:	19.	\$
20.	Other real property expenses not included in lines 4 or 5 of this form or on Schedule I: Your Incom	ne.	
	20a. Mortgages on other property	20a.	\$
	20b. Real estate taxes	20b.	\$
	20c. Property, homeowner's, or renter's insurance	20c.	\$
	20d. Maintenance, repair, and upkeep expenses	20d.	\$
	20e. Homeowner's association or condominium dues	20e.	\$

Debto	r 1 First Name Middle Name Last Name	Case number (if known)				
:1. C	Other. Specify:	21.	+\$			
2:	calculate your monthly expenses. 2a. Add lines 4 through 21. 2b. Copy line (monthly expenses for Debtor 2), if any, from Official Form 10. 2c. Add line 22a and 22b. The result is your monthly expenses.	06J-2 22.	\$ \$ \$			
23. C a	lculate your monthly net income.					
23	a. Copy line 12 (your combined monthly income) from Schedule I.	23a.	\$			
231	c. Copy your monthly expenses from line 22 above.	23b.	-\$			
230	c. Subtract your monthly expenses from your monthly income. The result is your <i>monthly net income</i> .	23c.	\$			
4. D o	o you expect an increase or decrease in your expenses within the year afte	er you file this form?				
	or example, do you expect to finish paying for your car loan within the year or do ortgage payment to increase or decrease because of a modification to the term	, , ,				
	No.					

☐ Yes.

Explain here:

April 22-23, 2014 Page 230 of 596

Fill in this information to identify your case:			
Debtor 1 First Name Last Name Check if this	e io:		
This ratio			
(Spouse, if filing) First Name Middle Name Last Name		•	-petition chapter 13
		the following	
Case number (If known)	/ YYYY	_	
(i. i.i.d. iii)			
Official Form 106J-2			
Schedule J-2: Expenses for Separate Household	of D	ebtor 2	12/15
Use this form for Debtor 2's separate household expenses ONLY IF Debtor 1 and Debtor 2 maintain se questions on this form with respect to Debtor 2 only. If Debtor 1 and Debtor 2 have one or more dependent on the Schedule J and Schedule J-2. Be as complete and accurate as possible. If more space is needed the top of any additional pages, write your name and case number (if known). Answer every question. Part 1: Describe Your Household	<i>ndents i</i> I, attach	n common, li	st the dependent on
Do you and Debtor 1 maintain separate households? Yes No (Do not complete this form)j			
1. Do you have dependents?	·	Dependent's	Does dependent live
Do not list Debtor 1 but list all Yes. Fill out this information for Debtor 2:		ige	with you?
regardless of whether listed as a			☐ No
dependent of Debtor 1 on Schedule J.	· <u>-</u>		Yes
Only list dependents	. <u> </u>		□ No □ Yes
Do not state the dependents'			☐ No
names.	_		☐ Yes
	. <u> </u>		□ No
			☐ Yes
	_		☐ No ☐ Yes
2. Do your expenses include expenses of people other than yourself, your dependents, and Debtor 1?			
Part 2: Estimate Your Ongoing Monthly Expenses			
	ant in a	Chapter 12 a	ages to veneut
Estimate your expenses as of your bankruptcy filing date unless you are using this form as a supplem expenses as of a date after the bankruptcy is filed.	ient in a	Chapter 13 c	case to report
Include expenses paid for with non-cash government assistance if you know the value of such assistance and have included it on <i>Schedule I: Your Income</i> (Official Form 106I.)		Your expe	nses
3. The rental or home ownership expenses for your residence. Include first mortgage payments and any rent for the ground or lot.	4.	\$	
If not included in line 4:			
If not included in line 4: 4a. Real estate taxes	4a.	\$	
	4a. 4b.	_	
4a. Real estate taxes		\$	

First Name	Middle Name	Last Name	

Case number (if known)_____

			Your expenses
4.	Additional mortgage payments for your residence, such as home equity loans	5.	\$
	Utilities:		
Э.	6a. Electricity, heat, natural gas	6a.	\$
	6b. Water, sewer, garbage collection	6b.	\$
	6c. Telephone, cell phone, Internet, satellite, and cable services	6c.	\$
	6d. Other. Specify:	6d.	\$
6.	Food and housekeeping supplies	7.	\$
7.		8.	\$
8.	Clothing, laundry, and dry cleaning	9.	\$
9.	Personal care products and services	10.	\$
10.	Medical and dental expenses	11.	\$
	Transportation. Include gas, maintenance, bus or train fare. Do not include car payments.	12.	\$
12.	Entertainment, clubs, recreation, newspapers, magazines, and books	13.	\$
13.	Charitable contributions and religious donations	14.	\$
		• • • •	Ψ
14.	Insurance. Do not include insurance deducted from your pay or included in lines 4 or 20.		
	15a. Life insurance	15a.	\$
	15b. Health insurance	15b.	\$
	15c. Vehicle insurance	15c.	\$
	15d. Other insurance. Specify:	15d.	\$
16.	Taxes. Do not include taxes deducted from your pay or included in lines 4 or 20. Specify:	16.	\$
17.	Installment or lease payments:		
	17a. Car payments for Vehicle 1	17a.	\$
	17b. Car payments for Vehicle 2	17b.	\$
	17c. Other. Specify:	17c.	\$
	17d. Other. Specify:	17d.	\$
18.	Your payments of alimony, maintenance, and support that you did not report as deducted from your pay on line 5, <i>Schedule I, Your Income</i> (Official Form 106I).	18.	\$
19.	Other payments you make to support others who do not live with you.		·
	Specify:	19.	\$
20.	Other real property expenses not included in lines 4 or 5 of this form or on Schedule I: Your Inco.	me.	
	20a. Mortgages on other property	20a.	\$
	20b. Real estate taxes	20b.	\$
	20c. Property, homeowner's, or renter's insurance	20c.	\$
	20d. Maintenance, repair, and upkeep expenses	20d.	\$
	20e. Homeowner's association or condominium dues	20e.	\$

otor 1					Case number (if known)				
	First Name	Middle Name	Last Name						
Other. Sp	ecify:						21.	+\$	
Your mor	nthly expen	ses. Add lines 4	through 21.					.	
The result	t is the mont	hly expenses of		result to line 22b	of Schedule	J to calculate the		•	
expenses	for Debtor 1	and Debtor 2.					22.		
Line not us	sed on this fo	orm.							
									
Do vou ex	pect an inc	rease or decrea	se in vour expense	s within the vea	ar after vou f	ile this form?			
-				-	-				
mortgage p	payment to i	ncrease or decre	ease because of a mo	odification to the	e terms of you	r mortgage?			
□ No.									
	Evaloia be								
103.	Explain ne	ere.							
	Other. Sp Your mon The result expenses Line not us Do you ex For examp	Other. Specify: Your monthly expensions for Debtor 1 Line not used on this for Debtor 1 Do you expect an incompression of Debtor 1 For example, do you expect an incompression of Debtor 1	Other. Specify: Your monthly expenses. Add lines 4 The result is the monthly expenses of expenses for Debtor 1 and Debtor 2. Line not used on this form. Do you expect an increase or decrease. For example, do you expect to finish parmortgage payment to increase or decrease. No.	Other. Specify: Your monthly expenses. Add lines 4 through 21. The result is the monthly expenses of Debtor 2. Copy the expenses for Debtor 1 and Debtor 2. Line not used on this form. Do you expect an increase or decrease in your expenses for example, do you expect to finish paying for your car load mortgage payment to increase or decrease because of a mile.	Other. Specify: Your monthly expenses. Add lines 4 through 21. The result is the monthly expenses of Debtor 2. Copy the result to line 22th expenses for Debtor 1 and Debtor 2. Line not used on this form. Do you expect an increase or decrease in your expenses within the year mortgage payment to increase or decrease because of a modification to the No.	Other. Specify: Your monthly expenses. Add lines 4 through 21. The result is the monthly expenses of Debtor 2. Copy the result to line 22b of Schedule expenses for Debtor 1 and Debtor 2. Line not used on this form. Do you expect an increase or decrease in your expenses within the year after you for example, do you expect to finish paying for your car loan within the year or do you expended payment to increase or decrease because of a modification to the terms of your loan.	Other. Specify: Your monthly expenses. Add lines 4 through 21. The result is the monthly expenses of Debtor 2. Copy the result to line 22b of Schedule J to calculate the expenses for Debtor 1 and Debtor 2. Line not used on this form. Do you expect an increase or decrease in your expenses within the year after you file this form? For example, do you expect to finish paying for your car loan within the year or do you expect your mortgage payment to increase or decrease because of a modification to the terms of your mortgage? No.	Other. Specify:	

April 22-23, 2014 Page 234 of 596

COMMITTEE NOTE

Schedule J: Your Expenses (Official Form 106J), formally Official Form 6J, has been revised to include references to new Schedule J-2: Expenses for Separate Household of Debtor 2 (Official Form 106J-2) at line 1 and new line 22b. The revisions clarify how to calculate monthly net income in joint cases where Debtor 1 and Debtor 2 maintain separate households. Line 22b is added so Schedule J and Schedule J-2 are easily coordinated. Schedule J is also renumbered to conform to the three digit numbering system that was developed for official bankruptcy forms as part of the Forms Modernization Project.

Schedule J-2 is new. It is used to report the monthly expenses of Debtor 2 in a joint debtor case only if Debtor 1 and Debtor 2 maintain separate households.

April 22-23, 2014 Page 235 of 596

April 22-23, 2014 Page 236 of 596

2013 Published Form, with working group edits 3-31-2014

Debtor		
United States Ba	Bankruptcy Court for the:	[Bankruptcy district]
Case number:		Chapter you are filing under:
caoo nambor.		. ,
		☐ Check if this is an amended plan
Official	Form 113	·
Chapt	ter 13 Plan	12/1
Part 1:	Notices	
To Debtors:	This form sets out options that may be appropriate in som indicate that the option is appropriate in your circumstance. In the following notice to creditors, you must check each box the	
To Creditors	rs: Your rights may be affected by this plan. Your claim may b	pe reduced, modified, or eliminated.
	You should read this plan carefully and discuss it with your atto have an attorney, you may wish to consult one.	orney, if you have one in this bankruptcy case. If you do not
	If you oppose the plan's treatment of your claim or any provisic confirmation at least 7 days before the date set for the hearing Court. The Bankruptcy Court may confirm this plan without furt Bankruptcy Rule 3015. In addition, you may need to file a time	on confirmation, unless otherwise ordered by the Bankruptcy ther notice if no objection to confirmation is filed. See
	The following matters may be of particular importance to you.	
	☐ The plan seeks to limit the amount of a secured claim, may result in a partial payment or no payment at all to	
	☐ The plan requests the avoidance of a judicial lien or no	
	security interest as set out in Part 3, section 3.4.	
	☐ The plan sets out nonstandard provisions in Part 9.	
Part 2:	Plan Payments and Length of Plan	
2.1 Debtor	r(s) will make regular payments to the trustee as follows:	
\$	per for months	
	\$ per for months.] <i>Insert additi</i>	ional lines if needed
If fewe the pay	er than 60 months of payments are specified, additional monthly pa ayments to creditors specified in Parts 3 through 6 of this plan.	yments will be made to the extent necessary to make
2.2 Regula	ar payments to the trustee will be made from future earnings ir	the following manner:
Check a	call that apply.	
☐ De	bebtor(s) will make payments pursuant to a payroll deduction order.	
☐ De	bebtor(s) will make payments directly to the trustee.	
Otl	Other (specify method of payment):	<u>_</u> .
2.3 Federa	al income tax refunds.	
Check of	cone.	
☐ De	ebtor(s) will retain any federal tax refunds received during the plant	term.
☐ De	bebtor(s) will supply the trustee with a copy of each federal tax return nd will turn over to the trustee all federal income tax refunds, other to	n filed during the plan term within 14 days of filing the return
	bebtor(s) will supply the trustee with federal tax returns filed during the deral income tax refunds received during the plan term as specified	
_		

Official Form 113 April 22-23, 2014

Chapter 13 Plan

Page 237 of 596

Debte	or	Case number									
2.4	Additional payments.										
	Check one.										
	☐ None. If None is chec	ked, the rest of section	on 2.4 need not be compl	eted or reprodu	ced.						
	Debtor(s) will make ac amount, and date of e			sources, as spe	cified below. Descri	be the source, e	stimated				
2.5	The total amount of estin	nated payments to t	he trustee provided for	in §§ 2.1 and 2	2.4 is \$						
Par	Treatment of Se	ecured Claims									
3.1	Maintenance of payment	s and cure of any de	efault.								
	Check one.										
	☐ None. If None is chec	ked, the rest of section	on 3.1 need not be compl	eted or reprodu	ced.						
The debtor(s) will maintain the contractual installment payments on the claims listed below, with any changes recapplicable contract, and cure any default in payments on the secured claims listed below. The allowed claim for a amount will be paid under the plan, with interest, if any, at the rate stated. Unless otherwise ordered by the court, listed on a proof of claim or modification of a proof of claim filed before the filing deadline under Bankruptcy Rule over any contrary amounts listed below as to the current installment payment and arrearage. If relief from the aut ordered as to any item of collateral listed in this paragraph, then, unless otherwise ordered by the court, all paym paragraph as to that collateral will cease and all secured claims based on that collateral will no longer be treated final column includes only payments disbursed by the trustee rather than by the debtor.							earage mounts c) control stay is nder this				
	Name of creditor	Collateral	Current installment payment (including escrow)	Amount of arrearage	Interest rate on arrearage (if applicable)	Monthly plan payment on arrearage	Estimated total payments by trustee				
			\$	\$	%	\$	\$				
			Disbursed by: Trustee Debtor(s)								
			c	c	0/	c	\$				
		_		\$	%	\$	Φ				

Insert additional claims as needed.

☐ Debtor(s)

Debtor	Case number	
20010.	 ouco mumbon	

3.2 Request for valuation of security and claim modification. Check one.

■ None. If None is checked, the rest of section 3.2 need not be completed or reproduced.

The remainder of this paragraph will be effective only if the applicable box in Part 1 of this plan is checked.

The debtor(s) request that the court determine the value of the secured claims listed below. For each non-governmental secured claim listed below, the debtor(s) state that the value of the secured claim should be as stated below in the column headed *Amount of secured* claim. For secured claims of governmental units, unless otherwise ordered by the court, the amounts listed in proofs of claim filed in accordance with the Bankruptcy Rules control over any contrary amounts listed below. For each listed secured claim, the controlling amount of the claim will be paid in full under the plan with interest at the rate stated below.

The portion of any allowed claim that exceeds the amount of the secured claim will be treated as an unsecured claim under Part 5 of this plan. If the amount of a creditor's secured claim is listed below as having no value, the creditor's allowed claim will be treated in its entirety as an unsecured claim under Part 5 of this plan. Unless otherwise ordered by the court, the amount of the creditor's total claim listed on the proof of claim controls over any contrary amounts listed in this paragraph.

The holder of any claim listed below as having value in the column headed Amount of secured claim will retain the lien until the earlier of:

- (a) payment of the underlying debt determined under nonbankruptcy law, or
- (b) discharge under 11 U.S.C. § 1328, at which time the lien will terminate and be released by the creditor. See Bankruptcy Rule 3015.

Name of creditor	Estimated amount Color creditor's total claim	Value of collateral	Amount of claims senior to creditor's claim	Amount of secured claim	Interest rate	Monthly payment to creditor	Estimated total of monthly payments
	\$	 \$	\$	\$	%	\$	\$
	\$	 \$	\$	\$	%	\$	\$
	\$	 \$	\$	\$	%	\$	\$

Insert additional claims as needed.

3.3 Secured claims excluded from 11 U.S.C. § 506.

Check one.

☐ None. If None is checked, the rest of section 3.3 need not b	be completed or reproduced
----------------------------------------------------------------	----------------------------

☐ The claims listed below were either:

- (1) incurred within 910 days before the petition date and secured by a purchase money security interest in a motor vehicle acquired for the personal use of the debtor(s), or
- (2) incurred within 1 year of the petition date and secured by a purchase money security interest in any other thing of value.

These claims will be paid in full under the plan with interest at the rate stated below. Unless otherwise ordered by the court, the claim amount stated on a proof of claim or modification of a proof of claim filed before the filing deadline under Bankruptcy Rule 3002(c) controls over any contrary amount listed below. The final column includes only payments disbursed by the trustee rather than by the debtor.

Name of creditor	Collateral	Amount of claim	Interest rate	Monthly plan payment	Estimated total payments by trustee
		\$	%	\$ Disbursed by: ☐ Trustee ☐ Debtor(s)	\$
		\$	70	\$ Disbursed by: ☐ Trustee ☐ Debtor(s)	\$

Insert additional claims as needed.

			Case number						
Lien a	avoidance.								
Check	k one.								
	None. If None is checked, the re	est of section 3.4 need not be completed or	reproduced.						
		will be effective only if the applicable b		checked.					
	☐ The judicial liens or nonpossessory, nonpurchase money security interests securing the claims listed below impair exemptions to which								
	the debtor(s) would have been e avoided to the extent that it impose security interest that is avoided interest that is not avoided will b	entitled under 11 U.S.C. § 522(b). A judicia airs such exemptions upon entry of the ord will be treated as an unsecured claim in Pape paid in full as a secured claim under the avoided, provide the information separate	I lien or security interest sec er confirming the plan. The a art 5. The amount, if any, of t plan. See 11 U.S.C. § 522(f	uring a claim listed below will be amount of the judicial lien or he judicial lien or security					
	Information regarding judicial lien or security interest	al Calculation of lien avoidance		Treatment of remaining secured claim					
	Name of creditor	a. Amount of lien	\$	Amount of secured claim after avoidance (line a minus line f)					
		b. Amount of all other liens	\$	\$					
	Collateral	c. Value of claimed exemptions	+ \$	Interest rate (if applicable)					
		d. Total of adding lines a, b, and c	\$	%					
	Lien identification (such as judgment date, date of lien recording book and page number)	e. Value of debtor's interest in property	- \$	Monthly plan payment \$					
		f. Subtract line e from line d.	\$	Estimated total payments on secured claim					
		Extent of exemption impairment (Check applicable box):		-					
		☐ Line f is equal to or greater tha The entire lien is avoided. (Do no							
	Insert additional claims as nee	ded.							
Surre	nder of collateral.								
Check		est of section 3.5 need not be completed or	reproduced.						
	termination of the stay under 11	to each creditor listed below the collateral U.S.C. § 362(a) and § 1301 with respect to the disposition of the collateral will be treated	o the collateral upon confirm						

Insert additional claims as needed.

3.5

Debtor			

Case number			

Part 4:	Treatment of Trustee's Fees and Priority Claims

4.1 General

Trustee's fees and all allowed priority claims other than those treated in section 4.5 will be paid in full without interest.

4.2 Trustee's fees

Trustee's fees are estimated to be ______ % of plan payments; and during the plan term, they are estimated to total \$______

4.3 Attorney's fees

The balance of the fees owed to the attorney for the debtor(s) is estimated to be \$______

4.4 Priority claims other than attorney's fees and those treated in section 4.5.

Check one.

- None. If None is checked, the rest of section 4.4 need not be completed or reproduced.
- ☐ The debtor estimates the total amount of other priority claims to be ______.

4.5 Domestic support obligations assigned or owed to a governmental unit and paid less than full amount.

Check one.

- None. If *None* is checked, the rest of section 4.5 need not be completed or reproduced.
- The allowed priority claims listed below are based on a domestic support obligation that has been assigned to or is owed to a governmental unit and will be paid less than the full amount of the claim under 11 U.S.C. § 1322(a)(4), but not less than the amount that would have been paid on such claim if the estate of the debtor were liquidated under chapter 7, see 11 U.S.C. § 1325(a)(4).

a.	Name of creditor	Amount of claim to be paid
		\$
c.		\$

Insert additional claims as needed.

Part 5: Treatment of Nonpriority Unsecured Claims

5.1 General

Nonpriority unsecured claims will be paid to the extent allowed as specified in this Part.

5.2 Nonpriority unsecured claims not separately classified.

Allowed nonpriority unsecured claims that are not separately classified will be paid, pro rata. If more than one option is checked, the option providing the largest payment will be effective. Check all that apply.

The sum of \$	

- ☐ ______% of the total amount of these claims.
- The funds remaining after disbursements have been made to all other creditors provided for in this plan.

Debt	Debtor			Case number					
5.3	Inter	est on allowed nonpriority unsecured cl	aims not separately clas	sified. Ched	ck one.				
	☐ None. If None is checked, the rest of section 5.3 need not be completed or reproduced.								
	Interest on allowed nonpriority unsecured claims that are not separately classified will be paid at an annual percentage rate of % under 11 U.S.C. §1325(a)(4), and is estimated to total \$								
5.4	5.4 Maintenance of payments and cure of any default on nonpriority unsecured claims. Check one.								
		None. If None is checked, the rest of section	ion 5.4 need not be comple	eted or repro	oduced.				
	☐ The debtor(s) will maintain the contractual installment payments and cure any default in payments on the unsecured claims listed below on which the last payment is due after the final plan payment. The allowed claim for the arrearage amount will be paid under the plan.								
		Name of creditor		Current inspayment	stallment	Estima	ted arrearage	Estimated total payments by trustee	
				Disbursed Truste	ee	\$		\$	
		Insert additional plains as product		Disbursed Trust	ee	\$		\$	
	,	Insert additional claims as needed.							
5.5	Othe	r separately classified nonpriority unsec	cured claims. Check one.						
	□N	one. If None is checked, the rest of section	5.5 need not be complete	d or reprodu	uced.				
	□т	he nonpriority unsecured allowed claims lis	sted below are separately o	classified an	d will be treated	d as follo	ows:		
		Name of creditor	Basis for separate class and treatment	ification	Estimated am to be paid	ount	Interest rate (if applicable)	Estimated total amount of	

Insert additional claims as needed.

Debto	or				Case number		
Par	t 6:	Executory Contrac	ts and Unexpired Lea	ises			
6.1		executory contracts and tracts and unexpired leas	_		l will be treated as sp	pecified. All other e	xecutory
		None. If checked, the rest of	of section 6.1 need not be	completed or reproduced			
☐ Assumed items. The final column includes only payments disbursed by the trustee rather than by the debto						ne debtor.	
		Name of creditor	Property description	Treatment (Refer to other plan section if applicable)	Current installment payment	Amount of arrearage to be paid	Estimated total payments by trustee
					\$ Disbursed by: ☐ Trustee ☐ Debtor(s)	\$	\$
					\$ Disbursed by: □ Trustee □ Debtor(s)	\$	\$
Par		Order of Distribution	on of Trustee Paymen		ning and a		
7.1		Trustee's fees	ments required in raits	5 through 6 in the follow	virig order.		
	d.						
	е.						
	f.						
	g.						
	h.						
Par	t 8:	Vesting of Property	of the Estate				
8.1	Pro	perty of the estate shall re	evest in the debtor(s) up				
		ck the applicable box:	, and an				
		Plan confirmation.					
		Closing of the case.					

Debtor		Case number
Part	9: Nonstandard Plan Provisions	
Unde	r Bankruptcy Rule 3015(c), nonstandard provisions are requ	uired to be set forth below.
Thes	e plan provisions will be effective only if the applicable	box in Part 1 of this plan is checked.
_		
-		
_		
_		
Part	10: Signatures	
4.0		
* _	Dignature of Attorney for Debtor(s)	ate
(Signature(s) of Debtor(s) (required if not represent	ted by an attorney; otherwise optional)
×	D	ata

Signature(s) of Debtor(s)

Signature(s) of Debtor(s)

X

Date_

Debtor			

Case number		
Case number		

Chapter 13 Plan Exhibit: Estimated Amounts of Trustee Payments

The trustee will make the following estimated payments on allowed claims in the order set forth in section 7.1:

a.	Maintenance and cure payments on secured claims (Part 3, section 3.1 total):	\$				
b.	Modified secured claims (Part 3, section 3.2 total):	\$				
c.	Secured claims excluded from 11 U.S.C. § 506 (Part 3, section 3.3 total):	\$				
d.	Judicial liens or security interests partially avoided (Part 3, section 3.4 total):					
e.	Administrative and other priority claims (Part 4 total):	\$				
f.	Nonpriority unsecured claims (Part 5, section 5.2 total):	\$				
g.	Interest on allowed unsecured claims (Part 5, section 5.3 total)	\$				
h.	Maintenance and cure payments on unsecured claims (Part 5, section 5.4 total)	\$				
i.	Separately classified unsecured claims (Part 5, section 5.5 total)	\$				
j.	Arrearage payments on executory contracts and unexpired leases (Part 6, section 6.1 total) +	\$				
Tot	\$					

April 22-23, 2014 Page 246 of 596

COMMITTEE NOTE

Official Form 113 is new and is the required plan form in all chapter 13 cases. See Bankruptcy Rule 3015. Alterations to the text of the form or the order of its provisions, except as permitted by the form itself, must comply with Bankruptcy Rule 9009. As the form explains, spaces for responses may be expanded or collapsed as appropriate, and sections that are inapplicable do not need to be reproduced. Portions of the form provide multiple options for provisions of a debtor's plan, but some of those options may not be accepted or recognized in the debtor's judicial district. Debtors are advised to refer to the applicable local rules and standing orders of the bankruptcy court.

Amounts listed in the plan form are controlling, unless the form requests an estimated or approximate amount.

Part 1. This part is intended to highlight some provisions of the plan for the benefit of interested parties and the court. For that reason, if the plan includes one or more of the provisions listed in this part, the appropriate boxes must be checked. For example, if Part 9 of the plan proposes a provision not included in, or contrary to, the Official Form, then that nonstandard provision will be ineffective if the appropriate check box is not selected.

This part states the proposed periodic plan payments, the Part 2. estimated total plan payments, and sources of funding for the plan. Section 2.1 allows the debtor or debtors to propose periodic payments in other than monthly intervals. For example, if the debtor receives a paycheck every week and wishes to make plan payments accordingly, that should be indicated in § 2.1. If the debtor proposes to make payments according to different "steps," the amounts and intervals of those payments should also be indicated in § 2.1. provides for the manner in which the debtor will make regular payments to the trustee. If the debtor selects the option of making payments pursuant to a payroll deduction order, that selection serves as a request by the debtor for entry of the order. Whether to enter a payroll deduction order is determined by the court. See Code § 1325(c). If the debtor selects the option of making payments other than by direct payments to the trustee or by a payroll deduction order, the alternative method (e.g., a designated third party electronic funds transfer program) must be specified.

Part 3. This part provides for the treatment of secured claims.

Section 3.1 provides for the treatment of claims under Code § 1322(b)(5) (maintaining current payments and curing any arrearage). For the claim of a secured creditor listed in § 3.1, an estimated arrearage amount should be given. A contrary arrearage amount listed on the creditor's proof of claim, unless contested by objection or motion, will control over the amount given in the plan.

In § 3.2, the plan may propose to determine under Code § 506(a) the value of a secured claim. For example, the plan could seek to reduce the secured portion of a creditor's claim to the value of the collateral securing it. For the secured claim of a nongovernmental creditor, that determination would be binding upon confirmation of the plan. For the secured claim of a governmental unit, however, a contrary valuation listed on the creditor's proof of claim, unless contested by objection or motion, would control over the valuation given in the

April 22-23, 2014 Page 247 of 596

plan. See Bankruptcy Rule 3012. Bankruptcy Rule 3002 contemplates that a debtor, the trustee, or another entity may file a proof of claim if the creditor does not do so in a timely manner. See Bankruptcy Rules 3004 and 3005. Section 3.2 will not be effective unless the appropriate check box in Part 1 is selected.

Section 3.3 deals with secured claims that may not be bifurcated into secured and unsecured portions under Code § 506(a), but it allows for an interest rate other than the contract rate to be applied to payments on such a claim. If appropriate, a claim may be treated under § 3.1 instead of § 3.3.

In § 3.4, the plan may propose to avoid certain judicial liens or security interests encumbering exempt property in accordance with Code § 522(f). This section includes space for the calculation of the amount of the judicial lien or security interest that is avoided. A plan proposing avoidance in § 3.4 must be served in the manner provided by Bankruptcy Rule 7004 for service of a summons and complaint. See Bankruptcy Rule 4003. Section 3.4 will not be effective unless the appropriate check box in Part 1 is selected.

Section 3.5 provides for elections to surrender collateral and consent to termination of the stay under § 362(a) and § 1301 with respect to the collateral surrendered. Termination will be effective upon confirmation of the plan.

Part 4. This part provides for the treatment of trustee's fees and claims entitled to priority status. Section 4.1 provides that trustee's fees and all allowed priority claims (other than those domestic support obligations treated in § 4.5) will be paid in full. In § 4.2, the plan lists an estimate of the trustee's fees. Although the estimate may indicate whether the plan will be feasible, it does not affect the trustee's entitlement to fees as determined by statute. In § 4.3, the form requests the balance of attorney's fees owed. Additional details about payments of attorney's fees, including information about their timing and approval, are left to the requirements of local practice. In § 4.4, the plan calls for an estimated amount of priority claims. A contrary amount listed on the creditor's proof of claim, unless changed by court order in response to an objection or motion, will control over the amount given in § 4.4. In § 4.5, the plan may propose to pay less than the full amount of a domestic support obligation that has been assigned to, or is owed to, a governmental unit, but not less than the amount that claim would have received in a chapter 7 liquidation.

Part 5. This part provides for the treatment of unsecured claims that are not entitled to priority status. Section 5.1 provides generally that the amounts of such claims listed on creditors' proofs of claim will control, unless otherwise ordered by the court. In § 5.2, the plan may propose to pay nonpriority unsecured claims in accordance with several options. One or more options may be selected. For example, the plan could propose simply to pay unsecured creditors any funds remaining after disbursements to other creditors, or also provide that a defined percentage of the total amount of unsecured claims will be paid. In § 5.4, the plan may provide for the separate classification of nonpriority unsecured claims (such as co-debtor claims) as permitted under Code § 1322(b)(1).

Part 6. This part provides for executory contracts and unexpired leases. An executory contract or unexpired lease is rejected unless it is listed in this part. If the plan proposes neither to assume nor reject an executory contract or

April 22-23, 2014 Page 248 of 596

unexpired lease, that treatment should be set forth as a nonstandard provision in Part 9.

- Part 7. This part provides an order of distribution of payments under the plan. Other than the trustee's fees, the order of distribution is left to be completed by the debtor in keeping with the requirements of the Code. The debtor may instead elect to have the trustee direct the order of distribution.
- Part 8. This part defines when property of the estate will revest in the debtor or debtors. One choice must be selected—upon plan confirmation, upon closing the case, or upon some other specified event. This plan provision is subject to a contrary court order under Code § 1327(b).
- Part 9. This part gives the debtor or debtors the opportunity to propose provisions that are not otherwise in, or are contrary to, the Official Form. All such nonstandard provisions must be set forth in this part and nowhere else in the plan. This part will not be effective unless the appropriate check box in Part 1 is selected. See Bankruptcy Rule 3015.
- Part 10. The plan must be signed by the attorney for the debtor or debtors. If the debtor or debtors are not represented by an attorney, they may signthe plan but are not required to do so.

April 22-23, 2014 Page 249 of 596

April 22-23, 2014 Page 250 of 596

Draft 9 Nov.12, 2013

Fill in this information to identify the case:
United States Bankruptcy Court for the:
District of
Case number (If known): Chapter

Official Form 201

Voluntary Petition for Non-Individuals Filing for Bankruptcy

12/15

If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write the debtor's name and the case number (if known). For more information, a separate document, *Instructions for Bankruptcy Forms for Non-Individuals*, is available.

i. Debtor's name						
2. All other names debtor used in the last 8 years						
Include any assumed names, trade names and <i>doing business</i> as names						
Debtor's federal Employer Identification Number (EIN)						
Debtor's address	Principal place of bus	siness	Mailing of busi		ifferent from p	orincipal place
	Number Street		Number	Street		
			P.O. Box	x		
	City	State ZIP	Code City		State	ZIP Code
			Locatio princip	Location of principal assets, if different from principal place of business		
	County		Number	Street		
			City		State	ZIP Code
Debtor's website (URL)						
Type of debtor	☐ Corporation (includi☐ Partnership (excludi		empany (LLC) and Li	mited Liability F	Partnership (LL	P))

Official Form B201

Del	btor			Case number (if kr	nown)			
	Name							
7.	Describe debtor's business	A. Check	one:					
		☐ Healt	h Care E	Business (as defined in 11 U.S.C. § 101(27A))				
		☐ Single	e Asset I	Real Estate (as defined in 11 U.S.C. § 101(51B))				
		☐ Railro	ad (as c	defined in 11 U.S.C. §101(44))				
		☐ Stock	broker (as defined in 11 U.S.C. § 101(53A))				
				roker (as defined in 11 U.S.C. § 101(6))				
			•	(as defined in 11 U.S.C. §781(3))				
		☐ None	-					
		B. Check	all that	apply:				
		□ тах-е	xempt e	entity (as described in 26 U.S.C. §501)				
			tment co	ompany, including hedge fund or pooled investme	nt vehicle (as defined in 15 U.S.C. §			
		_	•	dvisor (as defined in 15 U.S.C. § 80b-2(a)(11))				
		C. NAICS (North American Industry Classification System) 6-digit code that best describes debtor.						
				cs.com/search.htm.	code that best describes debtor.			
8.	Under which chapter of the	Check or	 ne:	· — — —				
	Bankruptcy Code is the	☐ Chap	ter 7					
	debtor filing?	☐ Chapter 9						
				Check all that apply:				
		<u> </u>		 Debtor's aggregate noncontingent liquidated d 	lehts (excluding dehts owed to			
		insiders or affiliates) are less than \$2,490,925 (amount subject to adjustment on 4/01/16 and every 3 years after that).						
			Ţ	The debtor is a small business debtor as defin	ed in 11 U.S.C. § 101(51D).			
			Ţ	☐ A plan is being filed with this petition.				
			[Acceptances of the plan were solicited prepeti creditors, in accordance with 11 U.S.C. § 1126				
Securities and Exchange Commis Exchange Act of 1934. File the <i>At</i>				Securities and Exchange Commission accordi	periodic reports (for example, 10K and 10Q) with the mmission according to § 13 or 15(d) of the Securities ne Attachment to Voluntary Petition for Non-Individuals Filing er 11 (Official Form 201A) with this form.			
			[☐ The debtor is a shell company as defined in th 12b-2.	e Securities Exchange Act of 1934 Rule			
		☐ Chap	ter 12					
9.	Were prior bankruptcy cases	☐ No						
	filed by or against the debtor within the last 8 years?	☐ Yes.	District	When	Case number			
	If more than 2 cases, attach a separate list.		District	MM / DD / YYYY When MM / DD / YYYY				
10.	Are any bankruptcy cases	□ No						
	pending or being filed by a		Date		Deletionakia			
	business partner or an affiliate of the debtor?	☐ Yes.	Deptor		Keiationsnip			
			District					
	List all cases. If more than 1, attach a separate list.		Case nu	mber, if known	MM / DD /YYYY			

Debtor Name	Case number (if known)				
11. Why is venue proper in <i>this</i> district?	 Check all that apply: Debtor has had its domicile, principal place of business, or principal assets in this district for 180 days immediately preceding the date of this petition or for a longer part of such 180 days than in any other district. A bankruptcy case concerning debtor's affiliate, general partner, or partnership is pending in this district. 				
	■ A bankruptcy case concern	ing debtor's affiliate, general partner, o	or partnership is pending in this district.		
12. Does the debtor own or have possession of any real property or personal property that needs immediate attention?	 No Yes. Answer below for each property that needs immediate attention. Attach additional sheets if needs Why does the property need immediate attention? (Check all that apply.) It poses or is alleged to pose a threat of imminent and identifiable hazard to public health or so What is the hazard? It needs to be physically secured or protected from the weather. It includes perishable goods or assets that could quickly deteriorate or lose value without attention (for example, livestock, seasonal goods, meat, dairy, produce, or securities-related assets or other options). 				
	☐ Other				
	Where is the propert	y? Number Street			
		City	State ZIP Code		
	Is the property insure No Yes. Insurance ager Contact name	ed?			
	Phone				
Statistical and adminis	trative information				
13. Debtor's estimation of available funds		distribution to unsecured creditors. Denses are paid, no funds will be availa	able for distribution to unsecured creditors.		
14. Estimated number of creditors	☐ 1-49 ☐ 50-99 ☐ 100-199 ☐ 200-999	☐ 1,000-5,000 ☐ 5,001-10,000 ☐ 10,001-25,000	☐ 25,001-50,000 ☐ 50,001-100,000 ☐ More than 100,000		
15. Estimated assets	□ \$0-\$50,000 □ \$50,001-\$100,000 □ \$100,001-\$500,000 □ \$500,001-\$1 million	\$1,000,001-\$10 million \$10,000,001-\$50 million \$50,000,001-\$100 million \$100,000,001-\$500 million	\$500,000,001-\$1 billion \$1,000,000,001-\$10 billion \$10,000,000,001-\$50 billion More than \$50 billion		
16. Estimated liabilities	□ \$0-\$50,000 □ \$50,001-\$100,000 □ \$100,001-\$500,000 □ \$500,001-\$1 million	\$1,000,001-\$10 million \$10,000,001-\$50 million \$50,000,001-\$100 million \$100,000,001-\$500 million	\$500,000,001-\$1 billion \$1,000,000,001-\$10 billion \$10,000,000,001-\$50 billion More than \$50 billion		

Debtor		Case number (if known)
	Name	

Request for Relief, Declaration, and Signatures

WARNING -- Bankruptcy fraud is a serious crime. Making a false statement in connection with a bankruptcy case can result in fines up to \$500,000 or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

17.	Declaration and signature of
	authorized representative of
	debtor

The debtor requests relief in accordance with the chapter of title 11, United States Code, specified in this petition.

I have been authorized to file this petition on behalf of the debtor.

I have examined the information in this petition and have a reasonable belief that the information is true and correct.

I declare under penalty of perjury that the foregoing is true and correct.

Executed on	MM / DD / YYYY		
×			
Signature of	authorized representative of debtor	Printed name	

18. Signature of attorney

	Date			
Signature of attorney for debtor	Date	ММ	/DD /YYYY	
Printed name				
Firm name				
Number Street				
City	State		ZIP Code	
Contact phone	Email	addres	ss	
Bar number	State		_	

COMMITTEE NOTE

Official Form 201, *Voluntary Petition for Non-Individuals Filing for Bankruptcy*, replaces Official Form 1, *Voluntary Petition*, for non-individual debtors. It is renumbered to distinguish it from the forms used by individual debtors and includes formatting and stylistic changes throughout the form.

Official Form 201 is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reducing the need to produce the same information in multiple formats.

The Forms Modernization Project made a preliminary decision that separate forms should be created for individual debtors and for non-individual debtors because separate areas of inquiry apply to each group. The forms for non-individuals do not include questions that pertain only to individuals and use a more open-ended response format. Also, where possible, the forms for non-individuals parallel how businesses commonly keep their financial records.

Official Form 201 has been substantially reformatted and reorganized. References to Exhibits B, C, and D, and the exhibits themselves, have been eliminated because the requested information is now asked in the form or is not applicable to non-individual debtors. Official Form 201A, *Attachment to Voluntary Petition for Non-Individuals Filing for Bankruptcy Under Chapter 11*, has replaced Exhibit A. The debtor is instructed to file Official Form 201A if the debtor is filing under chapter 11 and is required to file periodic reports with the Securities and Exchange Commission. A checkbox has been added to the form to indicate whether it is an amended filing.

April 22-23, 2014 Page 255 of 596

In Question 2, *All other names debtor used in the last 8* years, instructions pertaining only to individuals have been deleted, and an instruction to include *doing business as* names and assumed names has been added. In Question 3, *Debtor's federal Employee Identification Number (EIN)*, references to social security numbers and individual taxpayer I.D. numbers have been deleted. In Question 4, *Debtor's address*, the order of listing the various addresses for the debtor has been rearranged, and an address for the location of principal assets is required if different from the principal place of business. Also, the form has been revised to include a space for listing the debtor's website in Question 5.

In Question 6, *Type of Debtor*, options pertaining only to individual debtors have been deleted, and an instruction that the "partnership" option does not include LLPs has been added. Question 7, *Describe debtor's business*, is revised to include a statutory citation for each business type, to add an option for "none of the above," and to delete the option for "other." A new instruction requires the debtor to indicate if the debtor is an investment company, including a hedge fund or pooled investment vehicle; an investment advisor; or a tax exempt entity. The definition of "tax exempt entity" has been removed and replaced with a statutory citation. Additionally, an instruction has been added to require the debtor to list its North American Industry Classification System 6-digit code. A hyperlink is provided for information on finding the correct code.

In *Question 8, Under which chapter of the Bankruptcy Code is the debtor filing,* several separate boxes have been combined, and the options for Chapter 13 and Chapter 15 have been deleted. More detailed options have been added for Chapter 11. The question regarding the nature of the debtor's debts has been removed.

Question 9, Were prior bankruptcy cases filed by or against the debtor within the last 8 years, has been revised to instruct the debtor to include prior bankruptcy cases filed against the debtor and to list the district rather than location of the prior filings. In Question 10, Are any bankruptcy cases pending or being filed by a business partner or an affiliate of the debtor, the reference to

April 22-23, 2014 Page 256 of 596

spouse and the requirement to list the judge in any other cases have been removed.

Question 11, Why is venue proper in this district, has been revised to delete references that pertain only to individuals.

Question 12, Does the debtor own or have possession of any real property or personal property that needs immediate attention, replaces Exhibit C from Official Form 1. The category of "property that needs immediate attention" has been added, as well as options to indicate why the property needs immediate attention. Additionally, the form has been revised to require the debtor to list the location of the property and whether or not the property is insured and, if so, the insurance details.

Statistical and administrative information has been moved to immediately above the signature line, and the reference to exempt property has been removed. The maximum values for "Estimated Assets" and "Estimated Liabilities" have been increased from "more than \$1 billion" to "more than \$50 billion." Request for Relief, Declaration, and Signatures has been reformatted and the signature lines for individual debtors and non-attorney bankruptcy petition preparers have been removed.

April 22-23, 2014 Page 257 of 596

April 22-23, 2014 Page 258 of 596

Fill in this information to identify the case and this filing:			
rict of			
(State)			

Official Form 202

Declaration Under Penalty of Perjury for Non-Individual Debtors

12/15

An individual who is authorized to act on behalf of a non-individual debtor, such as a corporation or partnership, must sign and submit this form for the schedules of assets and liabilities, any other document that requires a declaration that is not included in the document, and any amendments of those documents. This form must state the individual's position or relationship to the debtor, the identity of the document, and the date. Bankruptcy Rules 1008 and 9011.

WARNING -- Bankruptcy fraud is a serious crime. Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$500,000 or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Declaration and signature

I am the president, another officer, or an authorized agent of the corporation; a member or an authorized agent of the partnership; or another individual serving as a representative of the debtor in this case.

I have examined the information in the documents checked below and I have a reasonable belief that the information is true and correct:

i nav	nave examined the information in the documents check	ed below and I have a reasonable belief that the information is true and correct:			
	Schedule A/B: Assets-Real and Personal Property (Official Form 206A/B)				
	Schedule D: Creditors Who Have Claims Secured by Property (Official Form 206D)				
	Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F)				
	Schedule G: Executory Contracts and Unexpired Le	ases (Official Form 206G)			
	Schedule H: Codebtors (Official Form 206H)				
	A Summary of Assets and Liabilities for Non-Individ	uals (Official Form 206–Summary)			
	Amended Schedule				
	Other document that requires a declaration				
I ded	declare under penalty of perjury that the foregoing is tru	e and correct.			
Exec	xecuted on				
	MM / DD / YYYY Sign	nature of individual signing on behalf of debtor			
	 Drin	tod name			

Official Form B202

Position or relationship to debtor

April 22-23, 2014 Page 260 of 596

COMMITTEE NOTE

Official Form 202, Declaration Under Penalty of Perjury for Non-Individual Debtors, replaces Official Form 2, Declaration Under Penalty of Perjury on Behalf of a Corporation or Partnership, and the section of Official Form 6 Declaration, Declaration Concerning Debtor's Schedules containing a corporation's or partnership's declaration. It is renumbered to distinguish it from the forms used by individual debtors and includes formatting and stylistic changes throughout the form.

Official Form 202 is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reducing the need to produce the same information in multiple formats.

Official Form 202 has been substantially reformatted and reorganized with elements from both Official Form 2 and the section of Official Form 6 for a corporation or partnership. Instructions have been added, along with warning language regarding bankruptcy fraud. Checkboxes are provided so the declaration will indicate the schedules included with the declaration or, if the declaration accompanies another document, a description of the attached document. The phrase "to the best of my information and belief" has been deleted from the declaration in order to conform to the language of 28 U.S.C. § 1746. See Rule 1008. The form, however, includes a statement that the person signing the declaration has examined the information in the documents subject to the declaration and has "a reasonable belief that the information is true and correct." Finally, the person signing the declaration must indicate his or her position or relationship to the debtor.

April 22-23, 2014 Page 261 of 596

April 22-23, 2014 Page 262 of 596

Fill in this information to identify the case:	
Debtor name	
United States Bankruptcy Court for the:	District of (State)
Case number (If known):	, ,

☐ Check if this is an amended filing

Official Form 204

Chapter 11 or Chapter 9 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims and Are Not Insiders 12/15

A list of creditors holding the 20 largest unsecured claims must be filed in a Chapter 11 or Chapter 9 case. Include claims which the debtor disputes. Do not include claims by any person or entity who is an *insider*, as defined in 11 U.S.C. § 101(31). Also, do not include claims by secured creditors, unless the unsecured claim resulting from inadequate collateral value places the creditor among the holders of the 20 largest unsecured claims.

	Name of creditor and complete mailing address, including zip code	code email address of creditor contact (for example, trade debts, bank loans, professional claim is continger unliquida		ris ngent, uidated, total claim amount. If claim is partially secured, fill in only unsecured claim amount. If claim is partially secured, fill in uidated, total claim amount and deduction for value of			
				Total claim, if partially secured	Deduction for value of collateral or setoff	Unsecured claim	
1							
2							
3							
4							
5							
6							
7							
8							

Debtor		Case number (if known)
	Name	

Name of creditor and complete mailing address, including zip code	code email address of creditor (for example, trade claim is		Amount of unsecured claim If the claim is fully unsecured, fill in only unsecured claim amount. If claim is partially secured, fill in total claim amount and deduction for value of collateral or setoff to calculate unsecured claim.			
			Total claim, if partially secured	Deduction for value of collateral or setoff	Unsecured claim	
9						
10						
11						
12						
13						
14						
15						
16						
17						
18						
19						
20						

COMMITTEE NOTE

Official Form 204, Chapter 11 or Chapter 9 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims and Are Not Insiders, replaces Official Form 4, List of Creditors Holding 20 Largest Unsecured Claims, for non-individual debtors. It is renumbered to distinguish it from the forms used by individual debtors and includes formatting and stylistic changes throughout the form.

Official Form 204 is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reducing the need to produce the same information in multiple formats.

The Forms Modernization Project made a preliminary decision that separate forms should be created for individual debtors and for non-individual debtors because separate areas of inquiry apply to each group. The forms for non-individuals do not include questions that pertain only to individuals and use a more open-ended response format. Also, where possible, the forms for non-individuals parallel how businesses commonly keep their financial records.

Official Form 204 has been reformatted and reorganized. The instructions have been shortened and revised to include a full cite to the definition of "insider" and a revised explanation of when to include a secured creditor's unsecured claim. The warning regarding the disclosure of a minor child's name has been deleted as a caution has been added to the general instructions for all forms regarding listing a minor child's name.

The heading of the second column of the form has been revised to require the "name, telephone number, and email address of creditor contact," eliminating the need to provide a complete

April 22-23, 2014 Page 265 of 596

mailing address for the creditor contact. Additional examples of "nature of claim" have been provided in the third column. In the fourth column, "subject to setoff" has been removed as an option.

The fifth column has been revised to include three separate potential entries to be used to list the value of the unsecured claim: the total claim, if partially secured; the deduction for value of collateral or setoff; and unsecured claim. The new instructions for the fifth column contain an explanation that if a claim is a fully unsecured claim, only the final sub-column needs to be completed, and that all of the columns must be completed if a claim is partially secured.

The signature line and the instruction to include a declaration have been deleted from the form.

April 22-23, 2014 Page 266 of 596

Fill in this information t	to identify the case:	
United States Bankruptcy	y Court for the:	
	District of (State)	
Case number (If known):		Chapter

☐ Check if this is an amended filing

Official Form 205

Involuntary Petition Against a Non-Individual

12/15

Use this form to begin a bankruptcy case against a non-individual you allege to be a debtor subject to an involuntary case. If you want to begin a case against an individual, use the *Involuntary Petition Against an Individual* (Official Form 105). Be as complete and accurate as possible. If more space is needed, attach any additional sheets to this form. On the top of any additional pages, write debtor's name and case number (if known).

Pa	art 1: Identify the Chapte	r of the Bankruptcy Code Under Which Petition I	s Filed
1.	Chapter of the Bankruptcy Code	Check one: Chapter 7 Chapter 11	
P	art 2: Identify the Debtor		
2.	Debtor's name		
3.	Other names you know the debtor has used in the last 8 years Include any assumed names, trade names, or		
	doing business as names.		
4.	Debtor's federal Employer Identification Number (EIN)	☐ Unknown	
		EIN	
5.	Debtor's address	Principal place of business	Mailing address, if different
		Number Street	Number Street
			P.O. Box
		City State ZIP Code	City State ZIP Code
			Location of principal assets, if different from principal place of business
		County	Number Street
			City State ZIP Code

Official Form 205

De	btor						Case number (if k	:nown)
	Nai	ne						
6.	Debtor's w	vebsite (URL)						
7.	Type of de	btor	Parti	nership (exclud	ling LLP)			ed Liability Partnership (LLP))
8.	Type of de	btor's	Check o	ne:				
			☐ Health	n Care Busines	s (as defined in	11 U.S.C. §	101(27A))	
			☐ Single	Asset Real Es	state (as defined	l in 11 U.S.C	. § 101(51B))	
					in 11 U.S.C. §1			
			☐ Stockl	broker (as defii	ned in 11 U.S.C.	. § 101(53A)))	
			☐ Comm	nodity Broker (a	as defined in 11	U.S.C. § 10	1(6))	
			☐ Cleari	ng Bank (as de	efined in 11 U.S.	.C. §781(3))		
			☐ None	of the types of	business listed.			
			☐ Unkno	own type of bus	siness.			
9.	To the bes		□ No					
	knowledge bankruptc		☐ Yes.	Debtor				Relationship
	pending b	y or against er or affiliate		District		Date filed	l	Case number, if known
	of this deb						MM / DD / YYYY	
				Debtor				Relationship
				District		Date filed	MM / DD / YYYY	Case number, if known
							MM/DD/YYYY	
Pá	art 3: R	eport About the	Case					
10.	Venue		Check on	ne:				
							nkruptcy, the debtor than in any other	or had a domicile, principal place of
						· ·	•	
			A pan	ikrupicy case c	oncerning debic	or s anniates,	general partner, o	or partnership is pending in this district.
11.	Allegation	s	Each peti	tioner is eligible	e to file this peti	tion under 11	U.S.C. § 303(b).	
	3		The debto	or may be the s	subject of an inv	oluntary case	e under 11 U.S.C.	§ 303(a).
			At least o	one box must b	e checked:			
					ally not paying its		ey become due, ui	nless they are the subject of a bona
			_	·	•			
			agent	appointed or a	uthorized to tak	e charge of I	ess than substanti	than a trustee, receiver, or an ially all of the property of the as appointed or took possession.
12.	Has there		☐ No					
	transfer of	any claim e debtor by or	☐ Yes. A	Attach all docur	nents that evide	ence the trans	sfer and any stater	ments required under Bankruptcy
	to any peti			1003(a).				- 1

Official Form 205

13. Each petitioner's claim	Name of petitioner	Nature of petitioner's claim	Amount of the claim above the value of any lien
			\$
			\$
			\$
		Total of petitioners' claims	\$

If more space is needed to list petitioners, attach additional sheets. Write the alleged debtor's name and the case number, if known, at the top of each sheet. Following the format of this form, set out the information required in Parts 3 and 4 of the form for each additional petitioning creditor, the petitioner's claim, the petitioner's representative, and the petitioner's attorney. Include the statement under penalty of perjury set out in Part 4 of the form, followed by each additional petitioner's (or representative's) signature, along with the signature of the petitioner's attorney.

Part 4: Request for Relief

WARNING -- Bankruptcy fraud is a serious crime. Making a false statement in connection with a bankruptcy case can result in fines up to \$500,000 or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Petitioners request that an order for relief be entered against the debtor under the chapter of 11 U.S.C. specified in this petition. If a petitioning creditor is a corporation, attach the corporate ownership statement required by Bankruptcy Rule 1010(b). If any petitioner is a foreign representative appointed in a foreign proceeding, attach a certified copy of the order of the court granting recognition.

I have examined the information in this document and have a reasonable belief that the information is true and correct.

Petitioners or Petitione	rs' Representative		Attorneys		
Name and mailing addr	ress of petitioner				
Name			Printed name		
Number Street			Firm name, if any		
City	State	ZIP Code	Number Street		
Name and mailing addr	ess of petitioner's rep	resentative, if any	City	State	ZIP Code
Name			Contact phone	Email	
Number Street			Bar number		
City	State	ZIP Code	State		
I declare under penalty of	of perjury that the foregoi	ing is true and correct.			
Executed on MM / DD / Y	YYYY -		Signature of attorney		
Signature of petitioner or rep	presentative, including repre	esentative's title	Date signed MM	/ DD /YYYY	

Official Form 205

			Printed name		
Name			riiiled hame		
Number Street			Firm name, if any		
City	State	ZIP Code	Number Street		
			City	 State	ZIP Code
Name and mailing address	s of petitioner's rep	resentative, if any	Contact phone		
Name					
Number Street			Bar number		
			State		
City	State	ZIP Code			
I declare under penalty of pe	erjury that the forego	ing is true and correct.	40		
Executed onMM / DD / YYY	Υ		Signature of attorney		
Signature of petitioner or repres	entative, including repre	esentative's title	Date signedMM / DD /	VVVV	
		Schalive 3 and	WINI 7 DD 7		
Name and mailing address		Schulive 3 une	Printed name		
Name and mailing address					
Name and mailing address Name Number Street		ZIP Code	Printed name		
Name and mailing address Name Number Street	s of petitioner	ZIP Code	Printed name Firm name, if any	State	ZIP Code
Name and mailing address Name Number Street	s of petitioner	ZIP Code	Printed name Firm name, if any Number Street	State	ZIP Code
Name and mailing address Name Number Street City Name and mailing address	s of petitioner	ZIP Code	Printed name Firm name, if any Number Street City Contact phone	State	
Name and mailing address Name Number Street City Name and mailing address Name	s of petitioner	ZIP Code	Printed name Firm name, if any Number Street City Contact phone Bar number	State	
Name and mailing address Name Number Street City Name and mailing address Name	s of petitioner	ZIP Code	Printed name Firm name, if any Number Street City Contact phone	State	
Name and mailing address Name Number Street City Name and mailing address Name Number Street	s of petitioner State s of petitioner's rep State	ZIP Code resentative, if any ZIP Code	Printed name Firm name, if any Number Street City Contact phone Bar number	State	
Name and mailing address Name Number Street City Name and mailing address Name Number Street City I declare under penalty of personal street on second seco	s of petitioner State s of petitioner's rep State State State	ZIP Code resentative, if any ZIP Code	Printed name Firm name, if any Number Street City Contact phone Bar number State	State	
Name and mailing address Name Number Street City Name and mailing address Name	s of petitioner State s of petitioner's rep State State State	ZIP Code resentative, if any ZIP Code	Printed name Firm name, if any Number Street City Contact phone Bar number State	State Email	

COMMITTEE NOTE

Official Form 205, *Involuntary Petition Against a Non-Individual*, replaces Official Form 5, *Involuntary Petition*, for non-individual debtors. It is renumbered to distinguish it from the forms used by individual debtors and includes formatting and stylistic changes throughout the form.

Official Form 205 is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reducing the need to produce the same information in multiple formats.

The Forms Modernization Project made a preliminary decision that separate forms should be created for individual debtors and for non-individual debtors because separate areas of inquiry apply to each group. The forms for non-individuals do not include questions that pertain only to individuals and use a more open-ended response format. Also, where possible, the forms for non-individuals parallel how businesses commonly keep their financial records.

Part 1, *Identify the Chapter of the Bankruptcy Code Under Which Petition is Filed*, has been moved to the beginning of the form.

In Part 2, *Identify the Debtor*, instructions pertaining only to individuals have been deleted, and an instruction to include doing-business-as names and assumed names has been added. The references to social security numbers and individual taxpayer I.D. numbers have been deleted. The order of listing the various addresses for the debtor have been rearranged in Line 5, and an address for the location of principal assets is required if different from the principal place of business. The form has been revised to include a space for listing the debtor's website in Line 6.

April 22-23, 2014 Page 271 of 596

Also in Part 2, the options for type of debtor that pertained only to individuals have been deleted, and an instruction that the "partnership" option does not include LLPs has been added. The options regarding the type of debtor's business have been revised to include a statutory citation for each business type, to add an option for "none of the above," and to delete the option for "other." The question regarding pending bankruptcy cases has been revised to remove the reference to spouse and the requirement to list the judge in any other cases.

In Part 3, *Report About the Case*, the question regarding venue has been revised in Line 10 to read "[o]ver the past 180 days before the filing of this bankruptcy, the debtor had a domicile, principal place or business, or principal assets in this district longer than in any other district." In the question for Allegations, "each" has been added to the first allegation, the exact citation to the Bankruptcy Code has been provided for the second allegation, and checkboxes have been provided for the last allegation. Also, in Line 12, petitioners must check "yes" or "no" to answer whether there has been any transfer of any claim against the debtor by or to a petitioner.

The information regarding the petitioner's claims has been moved to Part 3, and the portion listing the amount of the claim is amended to ask about the amount of the claim that exceeds the value of the lien, if any.

Part 4, *Request Relief*, has been amended to include a warning about making a false statement, and the declaration under penalty of perjury has been revised in order to conform to the language of 28 U.S.C. § 1746. See Rule 1008. A statement has been added that each petitioner, or the petitioner's representative, has reviewed the information in the petition and has "a reasonable belief that the information is true and correct." A requirement has been added for each petitioner's mailing address. Also, petitioners' attorneys must provide their email addresses, bar number, and state of bar membership.

April 22-23, 2014 Page 272 of 596

Fill in this information to identify the case:	
Debtor name	
United States Bankruptcy Court for the: District of (State)	
Case number (If known):	☐ Check if this is an amended filing

Official Form 206A/B

Schedule A/B: Assets — Real and Personal Property

12/15

Disclose all property, real and personal, which the debtor owns or in which the debtor has any other legal, equitable, or future interest. Include all property in which the debtor holds rights and powers exercisable for the debtor's own benefit. Also include assets and properties which have no book value, such as fully depreciated assets or assets that were not capitalized. In Schedule A/B, list any executory contracts or unexpired leases with a net value. Also list them on Schedule G: Executory Contracts and Unexpired Leases (Official Form 206G).

Be as complete and accurate as possible. If more space is needed, attach a separate sheet to this form. At the top of any pages added, write the debtor's name and case number (if known). Also identify the form and line number to which the additional information applies. If an additional sheet is attached, include the amounts from the attachment in the total for the pertinent part.

For Part 1 through Part 11, list each asset under the appropriate category or attach separate supporting schedules, such as a fixed asset schedule or depreciation schedule, that gives the details for each asset in a particular category. List each asset only once. In valuing the debtor's interest, do not deduct the value of secured claims. See the instructions to understand the terms used in this form.

Part 1: Cash and cash equivalents	
1. Does the debtor have any cash or cash equivalents? No. Go to Part 2. Yes. Fill in the information below.	
All cash or cash equivalents owned or controlled by the debtor	Current value of debtor's interest
2. Cash on hand	\$
3. Checking, savings, money market, or financial brokerage accounts (Identify all)	
Name of institution (bank or brokerage firm) 3.1	\$ \$ \$ \$
6. Does the debtor have any deposits or prepayments? No. Go to Part 3. Yes. Fill in the information below.	
	Current value of debtor's interest
7. Deposits, including security deposits and utility deposits	
Description, including name of holder of deposit	
7.1	\$
7.2	\$

Debtor Name		Case	number (if known)	
8. Prepayments, including p	prepayments on executo	ory contracts, leases, insurance, taxes, an	d rent	
Description, including name of	f holder of prepayment			
8.1				\$
8.2				\$
9. Total of Part 2.				\$
Add lines 7 through 8. Cop	by the total to line 81.			Ψ
Part 3: Accounts recei	vable			
10. Does the debtor have ar	ny accounts receivable?)		
☐ No. Go to Part 4.	•			
☐ Yes. Fill in the informa	ation below.			
				Current value of debtor's interest
11. Accounts receivable				
11a. 90 days old or less:		=	·	\$
	face amount	doubtful or uncollectible accounts	_	
11b. Over 90 days old:	face amount	= doubtful or uncollectible accounts	: →	\$
12. Total of Part 3	a + 11b = line 12. Copy th	no total to line 92		\$
Current value on lines 11	a + 11b = IIIIe 12. Copy II	ie total to line oz.		
Part 4: Investments				
13. Does the debtor own an	nv investments?			
☐ No. Go to Part 5.	,			
☐ Yes. Fill in the informa	ation below.			
			Valuation method used for current value	Current value of debtor's interest
14. Mutual funds or publicly	v traded stocks not inclu	ided in Part 1	used for current value	interest
Name of fund or stock:	y traded stocks not more	aded in Fart I		
14.1				\$
14.2				\$
15. Non-publicly traded storincluding any interest in		rporated and unincorporated businesses, r joint venture		
Name of entity:		% of ownership:		
15.1		•		\$
15.2				\$
16. Government bonds, cor instruments not include		r negotiable and non-negotiable		
Describe:	-			
16.1				\$
				\$
17. Total of Part 4				
Add lines 14 through 16.	Copy the total to line 83			\$
	F, 1 total to into 00.			

	ht	^	r

Part 5: Inventory, excluding a	agriculture assets									
18. Does the debtor own any inven	Does the debtor own any inventory (excluding agriculture assets)?									
☐ No. Go to Part 6.										
Yes. Fill in the information be	☐ Yes. Fill in the information below.									
General description	Date of the last physical inventory	Net book value of debtor's interest (Where available)	Valuation method used for current value	Current value of debtor's interest						
19. Raw materials		(
		\$		\$						
20. Work in progress	MM / DD / YYYY	-								
20. Work in progress		\$		\$						
	MM / DD / YYYY	Φ		4						
21. Finished goods, including good	ds held for resale									
		\$		\$						
22. Other inventory or supplies										
	MM / DD / YYYY	\$		\$						
23. Total of Part 5				\$						
Add lines 19 through 22. Copy th	e total to line 84.			Φ						
24. Is any of the property listed in I	Part 5 perishable?									
_	sted in Part 5 subject to or part of a	possible PACA claim?								
☐ Yes										
	Has any of the property listed in Part 5 been purchased within 20 days before the bankruptcy was filed?									
No No Rock value	Valuation method	C	rent value							
Tes. Dook value	valuation method	Curr	ent value							
26. Has any of the property listed in No	n Part 5 been appraised by a prof	essional within the last	year?							
☐ No ☐ Yes										

Part 6:

☐ No. Go to Part 7.

Agricultural assets (other than titled motor vehicles and land)

27. Does the debtor own any agricultural assets (other than titled motor vehicles and land)?

	Yes. Fill in the information below.			
	General description	Net book value of debtor's interest (Where available)	Valuation method used for current value	Current value of debtor's interest
28.	Crops—either planted or harvested	\$		\$
29.	Farm animals Examples: Livestock, poultry, farm-raised fish			

	name			
30.	Farm machinery and equipment (Other than titled motor vehicles)			
		\$		\$
31.	Farm and fishing supplies, chemicals, and feed			
		\$		\$
32.	Other farm-related property not already listed in Part 5.	\$		\$
33.	Total of Part 6.			
	Add lines 28 through 32. Copy the total to line 85.			\$
34.	Is the debtor a member of an agricultural cooperative?			
	□ No			
	☐ Yes. Is any of the debtor's property stored at the cooperative?			
	☐ No			
	☐ Yes			
35.	Has any of the property listed in Part 6 been purchased within 20	days before the bankr	uptcy was filed?	
	No No			
	Yes. Book value \$ Valuation method	Current value	\$	
36.	Is a depreciation schedule available for any of the property listed	in Part 6?		
	No			
	Yes			
37.	Has any of the property listed in Part 6 been appraised by a profe	ssional within the last	year?	
	No			
	Yes			
Pai	office furniture, fixtures, and equipment; and collect	ctibles		
38.	Does the debtor own any office furniture, fixtures, equipment, or	collectibles?		
	☐ No. Go to Part 8.			
	Yes. Fill in the information below.			
	General description	Net book value of debtor's interest	Valuation method used for current value	Current value of debtor's interest
		(Where available)		
39.	Office furniture			
		\$		\$
40.	Office fixtures			
		\$		\$
4.4	Office and to the last of the second of the			
41.	Office equipment, including all computer equipment and communication systems equipment and software			
		\$		\$
42.	Collectibles <i>Examples:</i> Antiques and figurines; paintings, prints, or of artwork; books, pictures, or other art objects; china and crystal; stamp baseball card collections; other collections, memorabilia, or collectible	, coin, or		
	42.1	\$		
	42.2	\$		
	42.3	\$		\$
43.	Total of Part 7.			¢
	Add lines 39 through 42. Copy the total to line 86.			\$

Case number (if known)_

Debtor

De	ebtor	Name	Case n	umber (if known)	
		Name			
4.4	la a -la		Dawi 70		
44.	-	eciation schedule available for any of the property listed i	n Part / ?		
	☐ No				
	Yes				
45	Hae any	of the property listed in Part 7 been appraised by a profes	sional within the last ve	ar?	
чо.	-	of the property listed in rail r been appraised by a profes	isional within the last ye	ai :	
	☐ No				
	Yes				
Pal	tt 89 IVIa	chinery, equipment, and vehicles			
46.	Does the	debtor own any machinery, equipment, or vehicles?			
		Go to Part 9.			
	Yes.	Fill in the information below.			
	General d	escription	Net book value of	Valuation method used	Current value of
	Include ve	ar, make, model, and identification numbers (i.e., VIN,	debtor's interest	for current value	debtor's interest
	HIN, or N-		(Where available)		
47.	Automob	oiles, vans, trucks, motorcycles, trailers, and titled farm ve	ehicles		
	47.1		\$		\$
	47.2		\$		\$
	47.3		\$		\$
					\$
	47.4		Φ		Φ
48.	Watercra	off, trailers, motors, and related accessories Examples: Boa	ats.		
		notors, floating homes, personal watercraft, and fishing vessel			
	40.4		\$		\$
	48.2		\$		\$
40	Airereft a	and accessories			
49.	AllCraft	and accessories			
	49.1		\$		\$
	49.2		\$		\$
50.		achinery, fixtures, and equipment (excluding farm ry and equipment)			
	macmine	ry and equipment)			
			\$		\$
			*		·
ΕA	Total of I	Dart 0			
51.					\$
	Add lines	47 through 50. Copy the total to line 87.			
52	ls a denr	eciation schedule available for any of the property listed i	n Part 8?		
J2.	□ No	constant analogo for any of the property fiction	 -		
	☐ Yes				
	162				
53.	Has any	of the property listed in Part 8 been appraised by a profes	sional within the last ye	ar?	
	☐ No		•		
	☐ Yes				

П	Δ	ht	۸r

Case number (if known)_____

Par	t 9: Real property				
54.	Does the debtor own any real property?				
	☐ No. Go to Part 10.				
	☐ Yes. Fill in the information below.				
55.	Any building, other improved real estate, or la	nd which the debtor o	owns or in which the d	lebtor has an interest	
	Description and location of property Include street address or other description such as Assessor Parcel Number (APN), and type of property (for example, acreage, factory, warehouse, apartment or office building), if available.	Nature and extent of debtor's interest in property	Net book value of debtor's interest (Where available)	Valuation method used for current value	Current value of debtor's interest
	55.1		\$		\$
	55.2		\$		\$
	55.3		\$		\$
	55.4		\$		\$
	55.5		\$		\$
	55.6		\$		\$
56	Total of Part 9.				
50.	Add the current value on lines 55.1 through 55.6	and entries from any a	dditional sheets. Copy t	he total to line 88.	\$
57.	No				
	☐ Yes				
58.	Has any of the property listed in Part 9 been a	ppraised by a profess	ional within the last y	ear?	
	U No □				
	Yes				
Par	t 10: Intangibles and Intellectual Prope	rty			
59.	Does the debtor have any interests in intangit	oles or intellectual pro	pperty?		
	No. Go to Part 11.				
	☐ Yes. Fill in the information below.				
	General description		Net book value of debtor's interest (Where available)	Valuation method used for current value	Current value of debtor's interest
60.	Patents, copyrights, trademarks, and trade se	crets	•		•
61.	Internet domain names and websites		\$		\$
62.	Licenses, franchises, and royalties		\$		\$
63.	Customer lists, mailing lists, or other compila	tions	\$		\$
64.	Other intangibles, or intellectual property		\$		\$
65.	Goodwill		\$		\$
66.	Total of Part 10. Add lines 60 through 65. Copy the total to line 89	ı.			\$

Del	btor Case number (if known)	
	Name	
67.	Do your lists or records include personally identifiable information of customers (as defined in 11 U.S.C. §§ 101(41A) and 107) ?
	□ No	
	☐ Yes	
60	Is there an amortization or other similar schedule available for any of the property listed in Part 10?	
00.		
	U No	
	Yes	
69.	Has any of the property listed in Part 10 been appraised by a professional within the last year?	
	□ No	
	☐ Yes	
Davi	All other coasts	
Pai	rt 11: All other assets	
70.	Does the debtor own any other assets that have not yet been reported on this form?	
	No. Go to Part 12.	
	Yes. Fill in the information below.	
		Current value of
		debtor's interest
71.	Notes receivable	
	Description (include name of obligor)	
	Total face amount doubtful or uncollectible amount	\$
72	Tay refunds and unused not energing leades (NOLs)	
12.	Tax refunds and unused net operating losses (NOLs)	
	Describe (for example, federal, state, local)	
	Tax year	\$
	Tax year	\$
	Tax year	\$
73	Interests in insurance policies or annuities	
75.	interests in insurance policies of annumes	Φ.
		\$
74.	Causes of action against third parties (whether or not a lawsuit	
	has been filed)	
		\$
	Nature of claim	
	Amount requested \$	
75.	Other contingent and unliquidated claims or causes of action of	
	every nature, including counterclaims of the debtor and rights to	
	set off claims	
		\$
	Nature of claim	
	Amount requested_ \$	
76	Trusts, equitable or future interests in property	
. 0.	Tracte, equitable of future interests in property	
	<u> </u>	\$
77.	Other property of any kind not already listed Examples: Season tickets, country club membership	
	Country club membership	
		\$
		\$
78.	Total of Part 11.	
	Add lines 71 through 77. Copy the total to line 90.	\$
79.	Has any of the property listed in Part 11 been appraised by a professional within the last year?	
	U No □	
	Yes	

Part 12:

Summary

In Part 12 copy all of the totals from the earlier parts of the form.

Type of property	Current value of personal property	Current value of real property
80. Cash, cash equivalents, and financial assets. Copy line 45, Part 1.	\$	
81. Deposits and prepayments. Copy line 9, Part 2.	\$	
82. Accounts receivable. Copy line 12, Part 3.	\$	
83. Investments. Copy line 17, Part 4.	\$	
84. Inventory. Copy line 23, Part 5.	\$	
85. Agricultural assets. Copy line 33, Part 6.	\$	
36. Office furniture, fixtures, and equipment, and collectibles. Copy line 43, Part 7.	\$	
7. Machinery, equipment, and vehicles. Copy line 51, Part 8.	\$	
8. Real property. Copy line 56, Part 9	→	\$
9. Intangibles and intellectual property. Copy line 66, Part 10.	\$	
00. All other assets. Copy line 78, Part 11.	+ \$	
91. Total. Add lines 80 through 90 for each column91a.	\$	+ 91b. \$
2. Total of all property on Schedule A/B. Lines 91a + 91b = 92		

Fill in this information to identify the case:		Draft Nov. 1	2, 2013
Debtor name United States Bankruptcy Court for the: Case number (If known):			Check if this is an
Official Form 206D			amended filing
	Vho Have Claims Secured b	v Property	12/15
Be as complete and accurate as possible.		<i>y</i>	12/10
 Do any creditors have claims secured by det No. Check this box and submit page 1 of thi Yes. Fill in all of the information below. 	otor's property? s form to the court with debtor's other schedules. Debtor h	as nothing else to report	on this form.
Part 1: List Creditors Who Have Secure	ed Claims	Column A	Column B
secured claim, list the creditor separately for ea	ave secured claims. If a creditor has more than one ch claim.	Amount of claim Do not deduct the value of lien.	Value of debtor's
2.1 Creditor's name	Describe debtor's property that is subject to a lien	•	
Creditor's mailing address		_ \$	\$
	Describe the lien	-	
Creditor's email address, if known	Is the creditor an insider or related party? No Yes		
Date debt was incurred	Is anyone else liable on this claim? ☐ No		
Last 4 digits of account number	Yes. Fill out <i>Schedule H: Codebtors</i> (Official Form 206H).		
Do multiple creditors have an interest in the same property?	As of the petition filing date, the claim is: Check all that apply.		
☐ No ☐ Yes. Specify each creditor, including this creditor, and its relative priority.	Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed		
2.2 Creditor's name	Describe debtor's property that is subject to a lien	•	•
Creditor's mailing address		_ \$ -	\$
	Describe the lien	-	
Creditor's email address, if known	Is the creditor an insider or related party? No Yes	-	
Date debt was incurred	Is anyone else liable on this claim?		
Last 4 digits of account number	□ No□ Yes. Fill out Schedule H: Codebtors (Official Form 206H).		
Do multiple creditors have an interest in the same property?	As of the petition filing date, the claim is: Check all that apply.		
□ No□ Yes. Have you already specified the relative	Contingent Unliquidated		
priority? No. Specify each creditor, including this creditor, and its relative priority.	Disputed Liquidated and neither contingent nor disputed		
Yes. The relative priority of creditors is specified on lines			

3. Total of the dollar amounts from Part 1, Column A, including the amounts from the Additional Page, if any.

	Case number (if known)	
Name		

Part 1: Additional Page		Amount of claim Do not deduct the value	Value of debtor's property that secures
Copy this page only if more space is needed. Corevious page.	Continue numbering the lines sequentially from the	of lien.	this claim
Creditor's name	Describe debtor's property that is subject to a lien		
Creditor's mailing address			\$
	Describe the lien		
Creditor's email address, if known	Is the creditor an insider or related party? ☐ No ☐ Yes		
Date debt was incurred Last 4 digits of account	Is anyone else liable on this claim?		
number	Yes. Fill out <i>Schedule H: Codebtors</i> (Official Form 206H). As of the petition filing date, the claim is:		
same property? □ No	Check all that apply. Contingent Unliquidated		
☐ Yes. Have you already specified the relative priority?☐ No. Specify each creditor, including this	 Unliquidated Disputed Liquidated and neither contingent nor disputed 		
Yes. The relative priority of creditors is specified on lines	Describe debtor's property that is subject to a lien		
Creditor's mailing address		\$	\$
	Describe the lien		
Creditor's email address, if known	Is the creditor an insider or related party? ☐ No ☐ Yes		
Date debt was incurred Last 4 digits of account number	Is anyone else liable on this claim? No Yes. Fill out Schedule H: Codebtors (Official Form 206H).		
Do multiple creditors have an interest in the same property?	As of the petition filing date, the claim is: Check all that apply.		
☐ No☐ Yes. Have you already specified the relative priority?	Contingent Unliquidated Disputed		
No. Specify each creditor, including this creditor, and its relative priority.	☐ Liquidated and neither contingent nor disputed		
Yes. The relative priority of creditors is specified on lines			

Case number (if known)_____

Part 2:

Name

List Others to Be Notified for a Debt Already Listed in Part 1

List in alphabetical order any others who must be notified for a debt already listed in Part 1. Examples of entities that may be listed are collection agencies, assignees of claims listed above, and attorneys for secured creditors.

If no others need to be notified for the debts listed in Part 1, do not fill out or submit this page. If additional pages are needed, copy this page.

Name and address	On which line in Part 1 did you enter the related creditor?	Last 4 digits of account number for this entity
	Line 2	
	Line 2	

April 22-23, 2014 Page 284 of 596

Draft Nov. 12, 2013

Fill in this information to identify the case:		
Debtor		
United States Bankruptcy Court for the:	District of	
Case number(If known)		

☐ Check if this is an amended filing

Official Form 206E/F

Schedule E/F: Creditors Who Have Unsecured Claims

12/15

Be as complete and accurate as possible. Use Part 1 for creditors with PRIORITY unsecured claims and Part 2 for creditors with NONPRIORITY unsecured claims. List the other party to any executory contracts or unexpired leases that could result in a claim. Also list executory contracts on Schedule A/B: Real and Personal Property (Official Form 206A/B) and on Schedule G: Executory Contracts and Unexpired Leases (Official Form 206G). Number the entries in Parts 1 and 2 in the boxes on the left. If more space is needed for Part 1 or Part 2, fill out and attach the Additional Page of that Part included in this form.

Pá	art 1: List All Creditors with PRIORITY U	nsecured Claims			
1.	Do any creditors have priority unsecured claim No. Go to Part 2. Yes. Go to line 2.	s? (See 11 U.S.C. § 507).			
2.		List in alphabetical order all creditors who have unsecured claims that are at least partially entitled to priority. If the debtor has more than 3 creditors with priority unsecured claims, fill out and attach the Additional Page of Part 1.			
			Total claim	Priority amount	
2.1	Priority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$	\$	
	Date or dates debt was incurred	Basis for the claim:			
	Last 4 digits of account number	Is the claim subject to offset? ☐ No ☐ Yes			
2.2	•	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$	\$	
	Date or dates debt was incurred	Basis for the claim:			
	Last 4 digits of account number	Is the claim subject to offset? ☐ No ☐ Yes			
2.3	Priority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$	\$	
	Date or dates debt was incurred	Basis for the claim:			
	Last 4 digits of account number	Is the claim subject to offset? ☐ No ☐ Yes			

Par	it 1. A	dditional Page			
		age if more space is needed. Continue nu ge. If no additional PRIORITY creditors e	•	Total claim	Priority amount
2	Priority	creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$	\$
	Date or	dates debt was incurred	Basis for the claim:		
	number Specify	ligits of account Code subsection of PRIORITY unsecured 1 U.S.C. § 507(a) ()	Is the claim subject to offset? No Yes		
2		creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$	\$
	Date or	dates debt was incurred	Basis for the claim:		
	number Specify	ligits of account Code subsection of PRIORITY unsecured 1 U.S.C. § 507(a) ()	Is the claim subject to offset? ☐ No ☐ Yes		
2	Priority	creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$	\$
	Date or	dates debt was incurred	Basis for the claim:		
	number Specify	ligits of account Code subsection of PRIORITY unsecured 1 U.S.C. § 507(a) ()	Is the claim subject to offset? ☐ No ☐ Yes		
2	Priority	creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$	\$
	Date or	dates debt was incurred	Basis for the claim:		
	number Specify	ligits of account Code subsection of PRIORITY unsecured 1 U.S.C. § 507(a) ()	Is the claim subject to offset? ☐ No ☐ Yes		

Debtor	
	Name

Caca	number (if known)		

Par	t 2: List All Creditors with NONPRIORITY Unsecured	d Claims	
	Do any creditors have unsecured claims not listed on Part 1 o No. Go to Part 3. Yes. Go to line 4.	f this form or on Schedule G?	
	ist in alphabetical order all of the creditors with nonpriority unsecured claims, fill out and attach the Additional Page of Part 2.		4 creditors with nonpriority
			Amount of claim
4.1 N	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$
_		Basis for the claim:	
г	Date or dates debt was incurred	Is the claim subject to offset?	
	ast 4 digits of account number	□ No □ Yes	
	ast 4 digits of account number	☐ Yes	
4.2 N	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$
		Basis for the claim:	
D	Date or dates debt was incurred	Is the claim subject to offset?	
L	ast 4 digits of account number	☐ No ☐ Yes	
4.3 N	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$
		Basis for the claim:	
D	Date or dates debt was incurred	Is the claim subject to offset?	
L	ast 4 digits of account number	☐ No ☐ Yes	
4.4 N	lonpriority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$
		Basis for the claim:	
п	Date or dates debt was incurred	Is the claim subject to offset?	
_	ast 4 digits of account number	□ No □ Yes	
	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply.	\$
-		☐ Contingent☐ Unliquidated☐ Disputed☐ None of the above	
		Basis for the claim:	
D	Date or dates debt was incurred	Is the claim subject to offset?	
	ast 4 digits of account number	U No □ Voc	

Part 2: Additional Page

	opy this page only if more space is needed. Continue nur evious page. If no additional NONPRIORITY creditors exi		Amount of claim
4	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$
		Basis for the claim:	
	Date or dates debt was incurred	Is the claim subject to offset?	
	Last 4 digits of account number	□ No □ Yes	
4	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$
		Basis for the claim:	
	Date or dates debt was incurred	Is the claim subject to offset?	
	Last 4 digits of account number	□ No □ Yes	
4	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated	\$
		☐ Disputed ☐ Liquidated and neither contingent nor disputed	
		Basis for the claim:	
	Date or dates debt was incurred Last 4 digits of account number	Is the claim subject to offset? ☐ No ☐ Yes	
4	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is: Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor	\$
		disputed	
		Basis for the claim:	
	Date or dates debt was incurred Last 4 digits of account number	Is the claim subject to offset? □ No □ Yes	
4	Nonpriority creditor's name and mailing address	As of the petition filing date, the claim is:	
		Check all that apply. Contingent Unliquidated Disputed Liquidated and neither contingent nor disputed	\$
		Basis for the claim:	
	Date or dates debt was incurred	Is the claim subject to offset?	
	Last 4 digits of account number	□ No □ Yes	
		□ 162	

Debtor	Name	Case number (if known)
--------	------	------------------------

Part 3:

List Others to Be Notified About Unsecured Claims

	ditional parties about the claims listed in Parts ed are collection agencies, assignors or assignee			unsecured creditors.
☐ No. If no others are to be notified☐ Yes. Fill in the information below	d of the debtor's unsecured debts, go to Part 4.			
Name and mailing address		-	which line in Part 1 or Part 2 is the ted creditor (if any) listed?	Last 4 digits of account number, if any
5.1.			Not listed. Explain	
5.2.			Not listed. Explain	
5.3.			Not listed. Explain	
5.4.			Not listed. Explain	
5.1.			Not listed. Explain	
5.5.			Not listed. Explain	
5.6.			Not listed. Explain	
5.7.			Not listed. Explain	
5.8.			Not listed. Explain	
5.9.			Not listed. Explain	
5.10.			Not listed. Explain	
5.11.			Not listed. Explain	
			'	

Part 3: Additional Page for Others to Be Notified About Unsecured Claims

Name and mailing address	On which line in Part 1 or Part 2 is the related creditor (if any) listed?	Last 4 digits of account number, if any
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	
5	Line Not listed. Explain	

Part 4:

Total Amounts of the Priority and Nonpriority Unsecured Claims

6. Add the amounts of priority and nonpriority unsecured claims.

Total of claim amounts

6a. Total claims from Part 1

\$_____

6b. Total claims from Part 2

6b. **+** \$_____

6c. Total of Parts 1 and 2 Lines 6a + 6b = 6c.

6c.

6a.

8_____

April 22-23, 2014 Page 292 of 596

Draft Nov. 20, 2013

	art Nov. 20, 20	J13			
Fill	in this information to identify th	ie case:			
Debt	tor name				
Unite	ed States Bankruptcy Court for the:		District of		
Case	e number (If known):		(State) Chapter		
					Check if this is an amended filing
Offi	icial Form 206G				amended ming
		utory Contra	acts and Ui	nexpired Leases	12/15
Be as	complete and accurate as pos	sible. If more space is	needed, copy and atta	ch the additional page, numbering the	entries consecutively.
4 1	Does the debtor have any execu	utory contracts or unov	mired leases?		
	•	•	-	edules. There is nothing else to report on	this form.
Į				d on Schedule A/B: Real and Personal P	
	ist all contracts and unexpired	leases		State the name and mailing address whom the debtor has an executory of	
2.1	State what the contract or lease is for and the nature				
	of the debtor's interest				
	State the term remaining				
	List the contract number of any government contract				
	State what the contract or				
2.2	lease is for and the nature of the debtor's interest				
	State the term remaining List the contract number of				
	any government contract				
	State what the contract or				
2.3	lease is for and the nature of the debtor's interest				
	State the term remaining				
	List the contract number of				
	any government contract				
2.4	State what the contract or				
	lease is for and the nature of the debtor's interest				
	State the term remaining				
	List the contract number of				
	any government contract				
	State what the contract or				
2.5	lease is for and the nature of the debtor's interest				
	State the term remaining				
	List the contract number of				
	any government contract				

Dalata.		0
Debtor		Case number (if known)
	Name	

	Additional Page if Debto	or Has More Executory Contracts or U	Jnexpired Leases
	Copy this page only if more s	space is needed. Continue numbering the li	ines sequentially from the previous page.
	List all contracts and unexpi	red leases	State the name and mailing address for all other parties with whom the debtor has an executory contract or unexpired lease
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining		
	List the contract number of any government contract		
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining		
	List the contract number of any government contract		
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining		
	List the contract number of any government contract		
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining		
	List the contract number of any government contract		
	gerenmen een aet		
	State what the contract or		
2	lease is for and the nature of the debtor's interest		
	State the term remaining		
	List the contract number of any government contract		
	, 35		
2.	State what the contract or lease is for and the nature		
۷	of the debtor's interest		
	State the term remaining		
	List the contract number of		
	any government contract		
2	State what the contract or lease is for and the nature of the debtor's interest		
	State the term remaining		
	List the contract number of		
	any government contract		

COMMITTEE NOTE

The schedules to be used in cases of non-individual debtors have been revised as part of the Forms Modernization Project, making them easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats.

The Forms Modernization Project made a preliminary decision that separate forms should be created for individual debtors and for non-individual debtors because separate areas of inquiry apply to each group. The forms for non-individuals eliminate questions that pertain only to individuals and use a more open-ended response format. Also, where possible, the forms for non-individuals parallel how businesses commonly keep their financial records. The non-individual debtor schedules are also renumbered, starting with the number 206 and followed by the letter or name of the schedule to distinguish them from the versions to be used in individual cases. Each form includes a checkbox to indicate whether it is an amended filing.

Official Form 206Sum, Summary of Assets and Liabilities for Non-Individuals, replaces Official Form 6, Summary of Schedules and Statistical Summary of Certain Liability and Related Data (28 U.S.C. § 159), in cases of non-individual debtors. The form is reformatted and updated with cross-references indicating the line numbers from specific schedules from which the summary information is to be gathered, and the Statistical Summary is deleted because it only applies to individual debtors. In addition, because most filings are now done electronically, the form no longer requires the debtor to indicate which schedules are attached or to state the number of sheets of paper used for the schedules.

Official Form 206A/B, *Schedule A/B: Assets – Real and Personal Property*, consolidates information about a non-

April 22-23, 2014 Page 295 of 596

individual debtor's real and personal property into a single form and replaces Official Form 6A - *Real Property* and Official Form 6B - *Personal Property*, in cases of non-individual debtors. The layout and categories of property on Official Form 206A/B have changed. Instead of dividing property interests into two categories (real or personal property), the new form uses eleven categories of property types. For each part, the specific items are broken out and debtors are instructed to total the part and list the total on a specific line later in the form.

Part 1: Cash and cash equivalents, includes cash and cash equivalents and a shortened list of examples. All financial assets other than cash or cash equivalents are moved to Part 4: *Investments*. In the section to list checking, savings, money market, or financial brokerage accounts, debtors are instructed to include the name of the institution and the last 4-digits of any account number.

In Part 2: *Deposits and prepayments*, adds prepayments and examples. A requirement has been added to include the name of the holder of any deposit.

Part 3: *Accounts receivable*, has been revised to divide accounts receivable into two categories depending on age and asks for separate values for the two categories.

Part 4: *Investments*, has been expanded and includes more detail.

Part 5: *Inventory, excluding agricultural assets*, has been amended to separate non-agricultural from agricultural assets, and has been expanded to include more detail. Categories of inventory are listed, and debtors must include the last date of physical inventory, the net book value of debtor's interest (if available), the valuation method used for current value, and the current value of debtor's interest. The form has been further amended to require the debtor to indicate whether the properties listed are perishable, whether any of the property was purchased within 20 days of the bankruptcy filing, and whether any of the property was appraised by a professional within the year prior to the bankruptcy filing.

April 22-23, 2014 Page 296 of 596

In Part 6: Agricultural assets (other than titled motor vehicles and land), the form has been amended to require more detailed responses and to require the debtor to indicate the net book value of the debtor's interest, the valuation method used for current value, and the current value of debtor's interest. A requirement to list fishing supplies has been added. The form has been further amended to require the debtor to indicate whether the properties listed are perishable, whether any of the property was purchased within 20 days of the bankruptcy filing, whether a depreciation schedule is available for any of the property listed, and whether any of the property was appraised by a professional within the year prior to the bankruptcy filing.

Part 7: Office furniture, fixtures, and equipment; and collectibles, has been amended to combine several categories of assets and to require more detail, including requiring the debtor to indicate the net book value of the debtor's interest, the valuation method used for current value, and the current value of debtor's interest. Examples of collectibles are provided. The form has been further amended to require the debtor to indicate whether a depreciation schedule is available for any property listed and whether any of the property listed was appraised by a professional within the year prior to the bankruptcy filing.

Part 8: *Machinery, equipment, and vehicles*, has been amended to combine several categories of property and to require more detail, including requiring the debtor to indicate the net book value of the debtor's interest, the valuation method used for current value, and the current value of debtor's interest. More examples are provided for each property type. The form has been further amended to indicate whether a depreciation schedule is available for any property listed and whether any of the property listed was appraised by a professional within the year prior to the bankruptcy filing.

Part 9: *Real property*, includes the elements of Official Form 6A, *Real Property*, and has been amended to expand the required information to include the net book value of the debtor's interest and the valuation method used for current value. Also, an instruction has been added for the description and location of the property. The form has been further amended to indicate whether

April 22-23, 2014 Page 297 of 596

a depreciation schedule is available for any property listed and whether any of the property listed was appraised by a professional within the year prior to the bankruptcy filing.

Part 10: Intangibles and intellectual property, includes amendments to combine several categories of property and to include more property types. The debtor is required to list the net book value of the debtor's interest and the valuation method used for current value. The question regarding personally identifiable information has been revised, and the form has been amended to require the debtor to indicate if there is an amortization schedule or similar schedule available for any property listed and whether any of the property listed was appraised by a professional within the year prior to the bankruptcy filing.

Part 11: *All other assets*, includes a new category for notes receivable, which requires a description, including the name of the obligor, the face amount, and any uncollectible amount. In addition, the form has been amended to combine tax refunds and net operating losses into a single question and to require more detail, to delete the requirement to list the insurance company name for any interests in insurance policies, to expand the question regarding contingent and unliquidated claims, and to include examples of other property. The form has been further amended to include a question regarding whether the property listed was appraised by a professional within the year prior to the bankruptcy filing.

Part 12, *Summary*, has been amended to list relevant line numbers for each type of property.

Official Form 206D, Schedule D: Creditors Who Hold Claims Secured by Property, replaces Official Form 6D, Creditors Holding Secured Claims, for non-individual debtors and has been revised to eliminate instructions that pertain only to individuals. The form has been further amended to instruct debtors that if a creditor has more than one secured claim, to list the creditor separately for each claim; to list the creditor's email address, if known; to indicate if multiple creditors have an interest in the same collateral; to list the order of each creditor's priority interest in the collateral; and to indicate whether the creditor is an insider or

April 22-23, 2014 Page 298 of 596

related party. The debtor is also instructed to describe the lien and to fill out *Schedule H: Codebtors*, if anyone else is liable on the claim. A new category for describing claims has been added—"unliquidated and neither contingent nor disputed". Finally, the form has been amended to require the debtor to list the value of the debtor's property that secures the claim.

A new Part 2: List Others to be Notified for a Debt Already Listed in Part 1 has been added, with instructions to list any others who must be notified about the bankruptcy for a debt listed in Part 1 of the form. Examples are provided. The debtor must include the relevant line from Part 1 and the last 4 digits of the account number for the entity.

A new Part 3: *Total Amounts of Claims and the Unsecured Portion of Claims*, has been added.

Official Form 206E/F, Schedule E/F: Creditors Who Hold Unsecured Claims, has been amended to combine Official Form 6E, Schedule E – Creditors Holding Unsecured Priority Claims and Official Form 6F, Schedule F – Creditors Holding Unsecured Nonpriority Claims for non-individual debtors. Priority unsecured claims are listed in Part 1, and nonpriority unsecured claims are listed in Part 2. The instructions have been revised to require the debtor to list the other party to any executory contract or unexpired lease on this schedule and on Schedule A/B Real and Personal Property and Schedule G: Executory Contracts and Unexpired Leases (Official Forms 206A/B and 206G).

Part 1, List All Creditors with PRIORITY Unsecured Claims, has been revised to delete the requirement to list the amount not entitled to priority and to add requirements to specify the Code section for the priority unsecured claim and whether the claim is subject to offset. A new category of "liquidated and neither contingent nor disputed" has been added to Part 2, List All Creditors with NONPRIORITY Unsecured Claims, along with the requirement to indicate if the claim is subject to offset. The instructions have also been significantly shortened. Part 3, List Others to be Notified About Unsecured Claims, has been added, with instructions to list any others that the debtor wants to notify about claims listed in Parts 1 and 2. Examples are given. The

April 22-23, 2014 Page 299 of 596

debtor must include the relevant line from Part 1 or 2 and the last 4 digits of the account number for the entity. A new Part 4: *Total Amounts of the Priority and Nonpriority Unsecured Claims* has been added.

Official Form 206G, Schedule G: Executory Contracts and Unexpired Leases, replaces Official Form 6G - Executory Contracts and Unexpired Leases for non-individual debtors. The form has been amended to delete the instruction regarding the listing of a minor child's name from the form as a caution is included in the general instructions for all forms regarding listing a minor child's name. A new requirement has been added to state the remaining term for any contract or lease listed.

Official Form 206H, *Schedule H: Codebtors*, replaces Official Form 6H – *Codebtors* for non-individual debtors. The form has been amended to delete the instruction regarding the listing of a minor child's name from the form as a caution is included in the general instructions for all forms regarding listing a minor child's name. A new requirement is added to indicate by checkbox what schedule applies to each co-debtor.

Schedules C, Exemptions, I, Income and J, Expenses.

There are no Official Forms for Schedules C, I, and J in non-individual debtor cases. There is no need for an Official Form 206C for non-individual debtors because exemptions are inapplicable to non-individual debtors. And, although section 521(a) of the Bankruptcy Code requires all debtors, including non-individual debtors, to provide schedules of income and expenses, uncertainty about the state of the debtor's business on the petition date – whether it is operating or not, for example – makes it difficult to create standard income and expense forms for non-individual debtors. Some bankruptcy courts have adopted local rules and forms for reporting the income and expenses of non-individual debtors, and Director's Procedural Forms 2060I and 2060J, can be used and modified as appropriate if there are no applicable local rules and forms.

Declaration. There is no Official Form 206, Declaration. The portion of Official Form 6 Declaration for a declaration on behalf of a corporation or partnership has been replaced by Official

April 22-23, 2014 Page 300 of 596

Form 202, *Declaration Under Penalty of Perjury for Non-Individual Debtors*. Official Form 202 includes checkboxes for the schedules included in Official Form 206.

April 22-23, 2014 Page 301 of 596

April 22-23, 2014 Page 302 of 596

Draft Nov. 20, 2013

Fill in this information to identify the case:	
Debtor name	
United States Bankruptcy Court for the:	District of (State)
Case number (If known):	(5-2-5)

☐ Check if this is an amended filing

Official Form 207

Statement of Financial Affairs for Non-Individuals Filing for Bankruptcy 12/15

The debtor must answer every question. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write the debtor's name and case number (if known).

Part	1: Income					
1. Gr	oss revenue from business					
	None					
	Identify the beginning and en be a calendar year	ding dates of the debtor's	s fiscal :	year, which may	Sources of revenue Check all that apply	Gross revenue (before deductions and exclusions)
	From the beginning of the fiscal year to filing date:	From	to	Filing date	Operating a business Other	\$
	For prior year:	From MM/DD/YYYY	to	MM / DD / YYYY	Operating a business Other	\$
	For the year before that:	From	to	MM / DD / YYYY	Operating a business Other	\$
Inc					e may include interest, dividends, mor ately. Do not include revenue listed in	
					Description of sources of revenue	Gross revenue from each source (before deductions and exclusions)
	From the beginning of the fiscal year to filing date:	From	to	Filing date		\$
	For prior year:	From MM/DD/YYYY	to	MM / DD / YYYY		\$
	For the year before that:	From MM/DD/YYYY	to	MM / DD / YYYY		\$

Name

Case number	if known)
-------------	-----------

Part 2:	List Certain	Transfers	Made Befor	e Filing	for Bankruptcy
---------	--------------	------------------	------------	----------	----------------

	None			
	None			
	Creditor's name and address	Dates	Total amount or value	Reasons for payment or transfer Check all that apply
.1.			\$	☐ Secured debt
	Creditor's name		Ψ	Unsecured loan repayments
	Street			☐ Suppliers or vendors
				☐ Services
	City State ZIP Code			☐ Other
	S.i.y S.i.d. 2.ii S.i.d.			
.2.			Φ.	☐ Secured debt
	Creditor's name		\$	☐ Unsecured loan repayments
	Street			Suppliers or vendors
				Services
	City State ZIP Code			Other
st plants;	ments or other transfers of property made with payments or transfers, including expense reimburs anteed or co-signed by an insider unless the aggregate. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. <i>Insiders</i> includeral partners of a partnership debtor and their related bettor. 11 U.S.C. § 101(31).	sements, made egate value of every 3 years de officers, d	de within 1 year before filing the fall property transferred to ose after that with respect to castirectors, and anyone in control	this case on debts owed to an insider or or or or the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives;
ist puar 6,2 ot i ene	payments or transfers, including expense reimburs ranteed or co-signed by an insider unless the aggregate. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31).	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one safter that with respect to castirectors, and anyone in controls of the debtor and insiders of	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of
ist pluar 6,2 ot i ene	payments or transfers, including expense reimburs ranteed or co-signed by an insider unless the aggregation (25. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. <i>Insiders</i> includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31).	sements, made egate value of every 3 years de officers, d	de within 1 year before filing the fall property transferred to ose after that with respect to castirectors, and anyone in control	this case on debts owed to an insider or or or or the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives;
ist puar 6,2 ot i ene	payments or transfers, including expense reimburs ranteed or co-signed by an insider unless the aggregation 25. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31). None Insider's name and address	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one safter that with respect to castirectors, and anyone in controls of the debtor and insiders of	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of
st pluar (ar) (b),2 (c) (d) (e) (d) (e) (d) (e) (e) (e) (e) (e) (e) (e) (e) (e) (e)	payments or transfers, including expense reimburs ranteed or co-signed by an insider unless the aggregate. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31).	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one safter that with respect to castirectors, and anyone in controls of the debtor and insiders of	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of
ist puar 6,2 ot i ene	payments or transfers, including expense reimburs ranteed or co-signed by an insider unless the aggregation 25. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31). None Insider's name and address	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one safter that with respect to castirectors, and anyone in controls of the debtor and insiders of	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of
st puar bar st i ene e c	payments or transfers, including expense reimburs ranteed or co-signed by an insider unless the aggregation 25. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31). None Insider's name and address Insider's name	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one safter that with respect to castirectors, and anyone in controls of the debtor and insiders of	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of
st pluar (ar) (b),2 (c) (d) (e) (d) (e) (d) (e) (e) (e) (e) (e) (e) (e) (e) (e) (e)	payments or transfers, including expense reimburs ranteed or co-signed by an insider unless the aggregation 25. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31). None Insider's name and address Insider's name Street City State ZIP Code	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one safter that with respect to castirectors, and anyone in controls of the debtor and insiders of	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of
ist puar 6,2 ot i ene	payments or transfers, including expense reimburs ranteed or co-signed by an insider unless the aggregation 25. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31). None Insider's name and address Insider's name	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one safter that with respect to castirectors, and anyone in controls of the debtor and insiders of	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of
st uar 3,2 ot i e o	payments or transfers, including expense reimburs ranteed or co-signed by an insider unless the aggregation 25. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31). None Insider's name and address Insider's name Street City State ZIP Code	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one safter that with respect to castirectors, and anyone in controls of the debtor and insiders of	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of
st uar luar 3,2 ot i e o	payments or transfers, including expense reimburs ranteed or co-signed by an insider unless the aggregation 25. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31). None Insider's name and address Insider's name Street City State ZIP Code	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one safter that with respect to castirectors, and anyone in controls of the debtor and insiders of	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of
st uar 3,2 ot i e o	payments or transfers, including expense reimburs anteed or co-signed by an insider unless the aggregation 25. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their reladebtor. 11 U.S.C. § 101(31). None Insider's name and address Insider's name Street City State ZIP Code	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one after that with respect to callirectors, and anyone in controls of the debtor and insiders of the debtor and inside	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of
ist puar 6,2 ot i ene	payments or transfers, including expense reimburs anteed or co-signed by an insider unless the aggress. (This amount may be adjusted on 4/01/16 and include any payments listed in line 3. Insiders includeral partners of a partnership debtor and their relatebtor. 11 U.S.C. § 101(31). None Insider's name and address Insider's name Street City State ZIP Code Relationship to debtor	sements, mac egate value o every 3 years de officers, d tives; affiliate	de within 1 year before filing the fall property transferred to one after that with respect to callirectors, and anyone in controls of the debtor and insiders of the debtor and inside	this case on debts owed to an insider or or for the benefit of the insider is less than uses filed on or after the date of adjustment.) Do ol of a corporate debtor and their relatives; of such affiliates; and any managing agent of

Official Form 207 Statement of Financial Affairs for Non-Individuals Filing for Bankruptcy

page 2

	Name							0400114	ımber (<i>if known</i>)_				
List	possessions, foreclosures, a all property of the debtor that v d at a foreclosure sale, transfer	was o	btained by			-		-				-	editor,
	None												
	Creditor's name and address				Descrip	tion of the p	roperty			Date		Value	e of property
5.1.												•	
	Creditor's name			-								\$	
	Street			-									
				_									
	City St	ate	ZIP Code	-									
5.1.	,												
	Creditor's name			-								\$_	
				_									
	Street												
				-									
	City Sta	ate	ZIP Code										
Seto	offs												
_	Creditor's name and address				Desc	ription of the	e action c	reditor took			ction was	Α	mount
	Creditor's name and address				Desc	ription of the	e action c	reditor took		Date a taken	ction was		mount
	Creditor's name and address Creditor's name				Desc	ription of the	e action c	reditor took			ction was	A	mount
					Desc	ription of the	e action c	reditor took			ction was		mount
	Creditor's name									taken	ction was		mount
	Creditor's name	State	ZIP Cod	e						taken	ction was		mount
nrt 3	Creditor's name Street City S									taken	ction was		mount
rt S Leg List was	Creditor's name Street City S Legal Actions or Assinal actions, administrative protection of the legal actions, proceedings involved in any capacity—with None	gnm oceed	ents lings, cou stigations, rear before	art act arbitr	Last 4	digits of acc	count nur attachme and aud	nber: XXXX- ents, or gov its by federa	rernmental a I or state age	taken	hich the de	\$ebtor	
Leg List was	Creditor's name Street City S Legal Actions or Assimal actions, administrative prothe legal actions, proceedings involved in any capacity—with	gnm oceed	ents lings, cou stigations, rear before	ı rt act arbitr	Last 4	digits of acc	count nur attachme and aud	nber: XXXX- ents, or gov its by federa	vernmental a	taken	hich the de	\$ebtor	tus of case
Leg List was	Creditor's name Street City S Legal Actions or Assinal actions, administrative protection of the legal actions, proceedings involved in any capacity—with None	gnm oceed	ents lings, cou stigations, rear before	art act arbitr	Last 4	digits of acc	attachme and aud	nber: XXXX- ents, or gov its by federa	rernmental a I or state age	taken	hich the de	\$ebtor	tus of case Pending
Leg List was	Creditor's name Street City S Legal Actions or Assinal actions, administrative protection of the legal actions, proceedings involved in any capacity—with None	gnm oceed	ents lings, cou stigations, rear before	art act arbitr	Last 4	digits of acc	attachmand aud	ents, or govits by federa	rernmental a I or state age	taken	hich the de	\$ebtor	tus of case
rt 3 Leg List was	Creditor's name Street City s Legal Actions or Assi pal actions, administrative proceedings involved in any capacity—with None Case title	gnm oceed	ents lings, cou stigations, rear before	art act arbitr	Last 4	digits of acc	attachmand aud	ents, or govits by federa	rernmental a I or state age	taken	hich the de	\$ebtor	tus of case Pending On appeal
Leg List was	Creditor's name Street City s Legal Actions or Assi pal actions, administrative proceedings involved in any capacity—with None Case title	gnm oceed	ents lings, cou stigations, rear before	art act arbitr	Last 4	digits of acc	attachmand aud	ents, or govits by federa	rernmental a I or state age	taken	hich the de	\$ebtor	tus of case Pending On appeal
rt 3 Leg List was	Creditor's name Street City s 3: Legal Actions or Assi al actions, administrative properties involved in any capacity—with None Case title Case number	gnm oceed	ents lings, cou stigations, rear before	art act arbitr	Last 4	digits of acc	attachmand aud	ents, or govits by federa Court or ager ame treet	vernmental all or state age	audits encies in w	hich the de	\$ebtor	tus of case Pending On appeal Concluded
rt 3 Leg. List was	Creditor's name Street City s Legal Actions or Assi pal actions, administrative proceedings involved in any capacity—with None Case title	gnm oceed	ents lings, cou stigations, rear before	art act arbitr	Last 4	digits of acc	attachmand aud	ents, or govits by federa Court or ager ame treet	vernmental a	audits encies in w	hich the de	sebtor	tus of case Pending On appeal Concluded
rt 3 Leg. List was	Creditor's name Street City s 3: Legal Actions or Assi al actions, administrative properties involved in any capacity—with None Case title Case number	gnm oceed	ents lings, cou stigations, rear before	art act arbitr	Last 4	digits of acc	attachmand aud	ents, or govits by federa Court or ager ame treet	vernmental all or state age	audits encies in w	hich the de	sebtor	tus of case Pending On appeal Concluded Pending On appeal
ırt 3 Leg List was	Creditor's name Street City s 3: Legal Actions or Assi al actions, administrative properties involved in any capacity—with None Case title Case number	gnm oceed	ents lings, cou stigations, rear before	art act arbitr	Last 4	digits of acc	attachmand aud	ents, or govits by federa Court or ager ame treet Court or ager	vernmental all or state age	audits encies in w	hich the de	sebtor	tus of case Pending On appeal Concluded

Statement of Financial Affairs for Non-Individuals Filing for Bankruptcy

City

page 3

ZIP Code

April 22-23, 2014 Page 305 of 596

otor Name	Case no	umber (if known)		
hands of a receiver, custodian, or other court-	for the benefit of creditors during the 120 days bet appointed officer within 1 year before filing this ca	-	and any proper	ty in the
None				
Custodian's name and address	Description of the property	Value		
Custodian's name		\$		
Custodian's name	Case title	Court name	and address	
Street				
	Case number	Name		
City State ZIF	P Code	Street		
	Data of audia an accionance			
	Date of order or assignment	City	State	ZIP Code
art 4: Certain Gifts and Charitable Co	ontributions			
None Recipient's name and address	Description of the gifts or contributions	Da	ates given	Value \$
9.1. Recipient's name				Φ
Street				
City State ZIP (Code			
Recipient's relationship to debtor				
9.2. Recipient's name				\$
Street				
City State ZIP 0	Code			
Recipient's relationship to debtor				
art 5: Certain Losses				
. All losses from fire, theft, or other casualty	within 1 year before filing this case.			
None				
Description of the property lost and how the occurred	If you have received payments to cover the lexample, from insurance, government comp	loss, for	ate of loss	Value of propert lost
	tort liability, list the total received. List unpaid claims on Schedule A/B: Propen	4,		

Name		

-	yments related to bankruptcy t any payments of money or other transfers of prop	erty made by the debtor or person acting on behal	f of the debtor within 1 yea	r before
the	filing of this case to another person or entity, inclu- eking bankruptcy relief, or filing a bankruptcy case.	ding attorneys, that the debtor consulted about del		
	None			
	Who was paid or who received the transfer?	If not money, describe any property transferred	Dates	Total amount or value
1.1.				¢.
	Address			\$
	Street			
	City State ZIP Code			
	Email or website address			
	Who made the payment, if not debtor?			
	Who was paid or who received the transfer?	If not money, describe any property transferred	Dates	Total amount or value
1.2.				\$
	Address			
	Street			
	City State ZIP Code			
	Email or website address			
	Who made the payment, if not debtor?			
Sel	f-settled trusts of which the debtor is a benefic	iary		
self	t any payments or transfers of property made by the f-settled trust or similar device. not include transfers already listed on this stateme		within 10 years before the	filing of this case
	None			
	Name of trust or device	Describe any property transferred	Dates transfers were made	Total amount or value
				\$
	Trustee			

Debtor	Name	Case number (if known)		
List 2 ye Incl	ears before the filing of this case to another persor	, trade, or any other means—made by the debtor or a pe n, other than property transferred in the ordinary course o security. Do not include gifts or transfers previously listed	f business or finan	cial affairs.
	Who received transfer?	Description of property transferred or payments received or debts paid in exchange	Date transfer was made	Total amount or value
13.1.				\$
	Address			
	Street			
	City State ZIP Code			
	Relationship to debtor			
	Who received transfer?			\$
13.2.				
	Address			
	Street			
	City State ZIP Code			
	Relationship to debtor			
Part 7	Previous Locations			
	vious addresses			
		years before filing this case and the dates the addresses	s were used.	
	Does not apply Address	Pates o	f occupancy	
	Address		Госсирансу	_
14.1.	Street	From		То
	Ch	710.0-1		
44.5	City	tate ZIP Code From		To
14.2.	Street	FIOIII		10

Statement of Financial Affairs for Non-Individuals Filing for Bankruptcy

page 6

April 22-23, 2014 Page 308 of 596

ZIP Code

Debtor	 Case number (if known)

Part 8	Healthcare Bankruptcies		
15. Hea	althcare bankruptcies		
Is th	e debtor primarily engaged in offering services a	nd facilities for:	
_	diagnosing or treating injury, deformity, or diseas	se, or	
_	providing any surgical, psychiatric, drug treatmer	nt, or obstetric care?	
	No. Go to Part 9.		
	Yes. Fill in the information below.		
	Facility name and address	Nature of the business operation, including type of services the debtor provides	If debtor provides meals and housing, number of patients in debtor's care
15.1.			
	Facility name		
	Street	Location where patient records are maintained (if different from facility address). If electronic, identify any service provider.	How are records kept?
		address). It decitorine, identity any service provider.	Chook all that apply:
			Check all that apply:
	City State ZIP Code		☐ Electronically☐ Paper
			☐ Paper
	Facility name and address	Nature of the business operation, including type of services the debtor provides	If debtor provides meals and housing, number of patients in debtor's care
15.2.			
13.2.	Facility name		
	Street	Location where patient records are maintained (if different from facility address). If electronic, identify any service provider.	How are records kept?
			Check all that apply:
	770.		☐ Electronically
	City State ZIP Code		☐ Paper
Dowt 0	Daman allu Idamtifiahla Information		
Part 9	Personally Identifiable Information		
16. Doe	s the debtor collect and retain personally ider	ntifiable information of customers?	
	No.		
_	Yes. State the nature of the information collected	and retained.	
	Does the debtor have a privacy policy about	t that information?	
	☐ No		
	☐ Yes		
	nin 6 years before filing this case, have any en sion or profit-sharing plan made available by	nployees of the debtor been participants in any ERISA, 401(k), 403 the debtor as an employee benefit?	(b) or other
	No. Go to Part 10.		
	Yes. Does the debtor serve as plan administrator	r?	
	☐ No. Go to Part 10.		
	Yes. Fill in below:		
	Name of plan	Employer identification n	umber of the plan
		EIN:	
	Has the plan been terminated?		
	No		
	☐ Yes		

Name -	
Name	_

Case number	(if known)	
Case number	(If known)	

Vith nov	ed, or transferred? ude checking, savings,	this case	market, or oth	ancial accounts or instrument or financial accounts; certificated on the rinancial institutions	ates of deposi			efit, closed, sold,
1	None							
	Financial institution na	ime and a	address	Last 4 digits of account number	Type of a	ccount	Date account was closed, sold, moved, or transferred	Last balance before closing or transfer
1.				XXXX-	☐ Check	king		- \$
	Name			/uu	☐ Saving	gs		Ψ
	Street				☐ Mone	y market		
					☐ Broke	rage		
	City	State	ZIP Code		Other			
2.				XXXX-	☐ Check	king		- \$
	Name			/vvv <u></u>	☐ Savin	gs		- Ψ
	Street				☐ Money	y market		
					☐ Broke	rage		
	City	State	ZIP Code		Other			
ist a	e deposit boxes any safe deposit box o None	or other d	lepository for s	securities, cash, or other valu	ables the deb	otor now has or	did have within 1 year b	pefore filing this ca
ist a	any safe deposit box o			securities, cash, or other valu			did have within 1 year b	Does debto
ist a	any safe deposit box o None							Does debto
ist a	any safe deposit box o None Depository institution							Does debto still have it?
ist a	any safe deposit box o None Depository institution Name							Does debto still have it
ist a	any safe deposit box o None Depository institution Name Street			Names of anyone with acce				Does debto still have it
ist a	Name Street City Streenses storage	n name ai	ZIP Code	Names of anyone with acce	ss to it	Description	of the contents	Does debto still have it
ist a	None Depository institution Name Street City seremises storage any property kept in storation the debtor does busing	n name at	ZIP Code	Names of anyone with acce	ss to it	Description	of the contents	Does debto still have it
ist a	None Depository institution Name Street City S remises storage any property kept in storation the debtor does busing the storage and the	n name at	ZIP Code	Names of anyone with acce Address ses within 1 year before filing	ss to it	Description of	of the contents	Does debto still have it
ist a	None Depository institution Name Street City S remises storage any property kept in storation the debtor does busing the storage and the	n name at	ZIP Code	Names of anyone with acce Address ses within 1 year before filing	ss to it	Description of	of the contents	Does debto still have it
ist a	None Depository institution Name Street City S remises storage any property kept in storate debtor does busing the debtor does busin	n name at	ZIP Code	Names of anyone with acce Address ses within 1 year before filing	ss to it	Description of	of the contents	Does debto still have it? No Yes of a building in Does debto still have it?
ist a	Any safe deposit box on None Depository institution Name Street City S remises storage any property kept in storation the debtor does busing the debtor do	n name at	ZIP Code	Names of anyone with acce Address ses within 1 year before filing	ss to it	Description of	of the contents	Does debto still have it No Yes Yes Does debto still have it

Γ	htor

Name

Case number	(if known)

Property held for another List any property that the debtor holds Do not list leased or rented property.	or controls that another entity owns. Include any	property borrowed from, being stored fo	or, or held in trust.
☐ None			
Owner's name and address	Location of the property	Description of the property	Value
Name			\$
Street			
City State	ZIP Code		
rt 12: Details About Environm	nental Information		
formerly owned, operated, or utilized.	operty, including disposal sites, that the debtor no		
or a similarly harmful substance.	hat an environmental law defines as hazardous or edings known, regardless of when they occur		aminant,
or a similarly harmful substance.		red.	
or a similarly harmful substance. port all notices, releases, and proce Has the debtor been a party in any j No	edings known, regardless of when they occuri	red.	
or a similarly harmful substance. port all notices, releases, and proce Has the debtor been a party in any j No Yes. Provide details below. Case title	edings known, regardless of when they occuri udicial or administrative proceeding under any	red. environmental law? Include settleme	nts and orders.
or a similarly harmful substance. port all notices, releases, and proce Has the debtor been a party in any j No Yes. Provide details below.	edings known, regardless of when they occuri udicial or administrative proceeding under any	red. environmental law? Include settleme	Status of case Pending On appeal
or a similarly harmful substance. sport all notices, releases, and proce Has the debtor been a party in any j No Yes. Provide details below. Case title	edings known, regardless of when they occurrudicial or administrative proceeding under any Court or agency name and address	red. environmental law? Include settleme	Status of case
or a similarly harmful substance. port all notices, releases, and proce Has the debtor been a party in any j No Yes. Provide details below. Case title	edings known, regardless of when they occurred udicial or administrative proceeding under any Court or agency name and address Name Street	red. venvironmental law? Include settleme Nature of the case	Status of case Pending On appeal
or a similarly harmful substance. eport all notices, releases, and proce Has the debtor been a party in any j No Yes. Provide details below. Case title	edings known, regardless of when they occurred udicial or administrative proceeding under any Court or agency name and address	red. venvironmental law? Include settleme Nature of the case	Status of cas Pending On appea
eport all notices, releases, and proce Has the debtor been a party in any j No Yes. Provide details below. Case title Case number	edings known, regardless of when they occurred udicial or administrative proceeding under any Court or agency name and address Name Street	red. venvironmental law? Include settleme Nature of the case	Status of case Pending On appeal Concluded
or a similarly harmful substance. port all notices, releases, and proce Has the debtor been a party in any j No Yes. Provide details below. Case title Case number Has any governmental unit otherwisenvironmental law? No	court or agency name and address Name Street City State ZIP Code	red. venvironmental law? Include settleme Nature of the case	Status of case Pending On appeal Concluded
or a similarly harmful substance. sport all notices, releases, and proce Has the debtor been a party in any j No Yes. Provide details below. Case title Case number Has any governmental unit otherwise environmental law? No Yes. Provide details below.	court or agency name and address Name Street City State ZIP Code	Nature of the case while or potentially liable under or in visible under or in visible under or in visible or potentially liable under or in visible un	Status of case Pending On appeal Concluded
eport all notices, releases, and proce Has the debtor been a party in any j No Yes. Provide details below. Case title Case number Case number No Yes. Provide details below. Site name and address	court or agency name and address Name City State ZIP Code See notified the debtor that the debtor may be lia	Nature of the case while or potentially liable under or in visible under or in visible under or in visible or potentially liable under or in visible un	Status of case Pending On appeal Concluded

)	the debtor notified any governmen No Yes. Provide details below.	tal unit of any release of hazardous material?	
	Site name and address	Governmental unit name and address	Environmental law, if known Date of notice
	Name	Name	
	Street	Street	
	City State ZIP Co	de City State ZIP Code	
1	3: Details About the Debtor	's Business or Connections to Any Busin	ness
h	or businesses in which the debter b	nge or hae had an interest	
	er businesses in which the debtor I any business for which the debtor wa		son in control within 6 years before filing this case.
	ude this information even if already lis		
)	None		
	Business name and address	Describe the nature of the business	Employer Identification number Do not include Social Security number or ITIN.
			EIN:
	Name		Dates business existed
	Street		From To
			 From To
	City State ZIP Co	nde	 From To
		Describe the nature of the business	Employer Identification number
	City State ZIP Co		
	City State ZIP Co		Employer Identification number Do not include Social Security number or ITIN.
	City State ZIP Co		Employer Identification number Do not include Social Security number or ITIN. EIN:
	City State ZIP Co	Describe the nature of the business	Employer Identification number Do not include Social Security number or ITIN. EIN:
	City State ZIP Co	Describe the nature of the business	Employer Identification number Do not include Social Security number or ITIN. EIN:
	City State ZIP Co	Describe the nature of the business	Employer Identification number Do not include Social Security number or ITIN. EIN:
	City State ZIP Co	Describe the nature of the business	Employer Identification number Do not include Social Security number or ITIN. EIN:
	City State ZIP Co	Describe the nature of the business	Employer Identification number Do not include Social Security number or ITIN. EIN:
	Business name and address Name Street City State ZIP Co	Describe the nature of the business	Employer Identification number Do not include Social Security number or ITIN. EIN: Dates business existed From To Employer Identification number Do not include Social Security number or ITIN. EIN:
	Business name and address Name Street City State ZIP Co	Describe the nature of the business Describe the nature of the business	Employer Identification number Do not include Social Security number or ITIN. EIN: Dates business existed From To Employer Identification number Do not include Social Security number or ITIN. EIN:

Debtor		Case number (if known)
	Name	

	s, records, and financial state	ements pers who maintained the debtor's boo	ks and records within:	2 years before filing this case.
	☐ None			_ ,
	Name and address			Dates of service
				From To
26a.1.	Name			
	Street			
	City	State	ZIP Code	
	Name and address			Dates of service
00-0				From To
26a.2.	Name			
	Street			
	City	State	ZIP Code	
Į	None Name and address			Dates of service
				From To
26b.	1. Name			
	Street			
	City	State	ZIP Code	
	Sity	Citate	211 0000	
	Name and address			Dates of service
26b.				From To
	Name 			
	Street			
	City	State	ZIP Code	
	Ony	State	ZIF COUR	
		vere in possession of the debtor's boo	ks of account and rec	ords when this case is filed.
ļ	None Name and address			If any books of account and records are
	Name and address			unavailable, explain why
26c.				
	Name 			
	Street		_	
	City	State	ZIP Code	
	City	State	ZIP CODE	

26c.2. Name Street City 26d. List all fina within 2 ye None Name Street City Name Street City Name Street City Name Street	ancial institutions, creditors, and opers before filing this case. The and address The and address The and address	State other parties, including mere	ZIP Code rcantile and trade	If any books of account and records are unavailable, explain why	statem
26c.2. Name Street City 26d. List all fina within 2 ye None Name Street City Name Street City Name Street City Name Street	ancial institutions, creditors, and of ears before filing this case.	other parties, including men	rcantile and trade	unavailable, explain why	statem
Street City 26d. List all fina within 2 ye None Name Street City Name Street City Name Street City Name Street Name Street Name Street Name Street	ancial institutions, creditors, and opers before filing this case. The and address The and address The and address	other parties, including men	rcantile and trade	e agencies, to whom the debtor issued a financial s	statem
Street City 26d. List all fina within 2 ye None Name Street City Name Street City Name Street City Name Street Name Street Name Street Name Street	ancial institutions, creditors, and opers before filing this case. The and address The and address The and address	other parties, including men	rcantile and trade	e agencies, to whom the debtor issued a financial s	statem
City 26d. List all fina within 2 ye None Name 26d.2. Name City Name City Name City Name City Name Street City Name All Parents of the control of the contr	ancial institutions, creditors, and of ears before filing this case. The and address The and address The and address	other parties, including men	rcantile and trade	agencies, to whom the debtor issued a financial s	statem
26d. List all fina within 2 ye None Name 26d.2. Name Street City Name Street City Name And City Name Street Name	ears before filing this case. The and address The and address The and address	other parties, including men	rcantile and trade	agencies, to whom the debtor issued a financial s	statem
26d. List all fina within 2 ye None Name 26d.2. Name Street City Name Street City Name And City Name Street Name	ears before filing this case. The and address The and address The and address	other parties, including men	rcantile and trade	e agencies, to whom the debtor issued a financial s	statem
within 2 ye None Name 26d.2. Name City Name Street City Name Street City Name And City Name Name Name Name Name Name Name Name Name	ears before filing this case. The and address The and address The and address			e agencies, to whom the debtor issued a financial s	statem
26d.2. Name Street City Name Street City Name Street	t e and address	State	ZIP Code		
Name Street City Name 26d.2. Name Street City Name And City Name Name Name Name Name Name Name Nam	t ne and address	State	ZIP Code		
Name Street City Name 26d.2. Name Street City Name And City Name Name Name Name Name Name Name Nam	t ne and address	State	ZIP Code		
City Name 26d.2. Name Street City nventories Have any inven	e and address	State	ZIP Code		
City Name 26d.2. Name Street City nventories Have any inven	e and address	State	ZIP Code		
Name 26d.2. Name Street City nventories Have any inven		State	ZIP Code		
Name 26d.2. Name Street City nventories Have any inven					
26d.2. Name Street City nventories Have any inven					
Name Street City nventories Have any inven	1				
Name Street City nventories Have any inven	1				
City nventories Have any inven ☐ No					
nventories Have any inven No	t				
nventories Have any inven No					
Have any inven ☑ No		State	ZIP Code		
Have any inven ☑ No					
	ntories of the debtor's property bed	en taken within 2 years bef	fore filing this cas	se?	
■ Yes. Give th					
	he details about the two most rece	ent inventories.			
Name of th	he person who supervised the taking	g of the inventory		te of The dollar amount and basis (cost, marke entory other basis) of each inventory	et, or
				\$	
Name and	address of the person who has pos	ssession of inventory records	s		
1. Name					
Street					
City					

Name		Case numbe	(if known)			
Name of the person who su	pervised the taking of the inventory	Date of inventory	other basi	amount and	ventory	ost, market, or
Name and address of the po	erson who has possession of inventory records					
. Name						
Name						
City st the debtor's officers, dire	State ZII ectors, managing members, general partner	Code s, members in contro	ol, controlli	ng shareho	olders, o	r other people
control of the debtor at the	time of the filing of this case.					
Name	Address	Posit	ion and natu	re of any int	erest %	of interest, if a
ne debtor, or shareholders in No	of this case, did the debtor have officers, din control of the debtor who no longer hold the		embers, ge	eneral partr	ners, me	mbers in con
ne debtor, or shareholders ir		hese positions?	embers, ge	·	Period o	during which
ne debtor, or shareholders in No Yes. Identify below.	n control of the debtor who no longer hold t	hese positions?	tion and natu	ire of	Period of position held	during which n or interest was
ne debtor, or shareholders in No Yes. Identify below.	n control of the debtor who no longer hold t	hese positions?	tion and natu	ire of	Period of position held	during which n or interest was To
ne debtor, or shareholders in No Yes. Identify below.	n control of the debtor who no longer hold t	hese positions?	tion and natu	ire of	Period of position held From	during which n or interest was To
ne debtor, or shareholders in No Yes. Identify below.	n control of the debtor who no longer hold t	hese positions?	tion and natu	ire of	Period of position held From From	during which n or interest was To To
ne debtor, or shareholders in No No Yes. Identify below. Name	n control of the debtor who no longer hold t	hese positions?	tion and natu	ire of	Period of position held From From	during which n or interest was To To
ne debtor, or shareholders in No No Yes. Identify below. Name ayments, distributions, or work it in 1 year before filing this or work.	Address Address rithdrawals credited or given to insiders case, did the debtor provide an insider with value.	hese positions? Posi any i	tion and natu	ire of	Period of position held From From From From	during which n or interest was To To To To To
ne debtor, or shareholders in No No Yes. Identify below. Name ayments, distributions, or work in 1 year before filing this concuses, loans, credits on loan in the concuses, loans, credits on loan in the concuses.	Address Address rithdrawals credited or given to insiders	hese positions? Posi any i	tion and natu	ire of	Period of position held From From From From	during which n or interest was To To To To To
ne debtor, or shareholders in No No Yes. Identify below. Name ayments, distributions, or work it in 1 year before filing this or work.	Address Address rithdrawals credited or given to insiders case, did the debtor provide an insider with value.	hese positions? Posi any i	tion and natu	ire of	Period of position held From From From From	during which n or interest was To To To To To
ne debtor, or shareholders in No No Yes. Identify below. Name ayments, distributions, or work it in 1 year before filing this conuses, loans, credits on loan No	Address withdrawals credited or given to insiders case, did the debtor provide an insider with values, stock redemptions, and options exercised?	hese positions? Posi any i	tion and natunterest	ire of	Period of position held From From From From sation, c	during which n or interest was To To To To Iraws,
ne debtor, or shareholders in No No Yes. Identify below. Name ayments, distributions, or we dithin 1 year before filing this conuses, loans, credits on loan No Yes. Identify below.	Address withdrawals credited or given to insiders case, did the debtor provide an insider with values, stock redemptions, and options exercised?	Positions? Positions any i	tion and natunterest	ner compen	Period of position held From From From From sation, c	during which n or interest was To To To To Iraws,
ayments, distributions, or w //ithin 1 year before filing this conuses, loans, credits on loan No Yes. Identify below. Name No No Name and address of reciping the conustry of	Address withdrawals credited or given to insiders case, did the debtor provide an insider with values, stock redemptions, and options exercised?	Positions? Positions any i	tion and natunterest	ner compen	Period of position held From From From From sation, c	during which n or interest was To To To To Iraws,
ayments, distributions, or w //ithin 1 year before filing this conuses, loans, credits on loan No Yes. Identify below. Name No Name and address of recipions	Address withdrawals credited or given to insiders case, did the debtor provide an insider with values, stock redemptions, and options exercised?	Positions? Positions any i	tion and natunterest	ner compen	Period of position held From From From From sation, c	during which n or interest was To To To To
ayments, distributions, or work that I have been been been been been been been be	Address withdrawals credited or given to insiders case, did the debtor provide an insider with values, stock redemptions, and options exercised?	Positions? Positions any i	tion and natunterest	ner compen	Period of position held From From From From sation, c	during which n or interest was To To To To Iraws,

page 13

ent corporation
nian frand
sion fund
ру

April 22-23, 2014 Page 316 of 596

COMMITTEE NOTE

Official Form 207, Statement of Financial Affairs for Non-Individuals Filing for Bankruptcy, replaces Official Form 7, Statement of Financial Affairs, for non-individual debtors. It is renumbered to distinguish it from the forms used by individual debtors and includes formatting and stylistic changes throughout the form.

Official Form 207 is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reducing the need to produce the same information in multiple formats.

The Forms Modernization Project made a preliminary decision that separate forms should be created for individual debtors and for non-individual debtors because separate areas of inquiry apply to each group. The forms for non-individuals do not include questions that pertain only to individuals and use a more open-ended response format. Also, where possible, the forms for non-individuals parallel how businesses commonly keep their financial records.

The form is derived from Official Form 7, *Statement of Financial Affairs*, and has been substantially reorganized. The form is divided into 14 sections grouping similar questions together. Many of the instructions have been shortened, and questions and instructions pertaining to individual debtors have been deleted. The instructions at the beginning of the form have been shortened, and the definitions deleted or moved to other parts of the form.

In Part 1, *Income*, the questions regarding gross revenue from business and non-business revenue have been consolidated, and checkboxes have been added to indicate the source of revenue.

April 22-23, 2014 Page 317 of 596

A definition of gross revenue has been added. Also, the debtor is instructed to include revenue only once.

In Part 2, List Certain Transfers Made Before Filing for Bankruptcy, information that pertains only to individuals has been eliminated, and the questions related to payments made in the 90 days prior to bankruptcy, payments made to insiders within one year prior to bankruptcy, repossessions, and setoffs have been consolidated. Instructions have been added to include expense reimbursements in answer to the questions regarding payments and to exclude regular employee compensation from the question regarding payments within 90 days. A dollar limitation has been added to the instructions for the question regarding payments to insiders. Checkboxes have been added to both questions to provide a reason for the payment, and the explanation that the dollar limitation changes every three years has been moved to the instructions from the footnotes. "Amount still owing" has been removed, and a definition of "insider" has been added along with a statutory citation to the question regarding insiders. Partnerships have been added to examples of "insiders." The question regarding setoffs includes a revised definition and has been revised to require that the debtor provide a description of the creditor's actions and the last four digits of any account number.

In Part 3, *Legal Actions or Assignments*, several questions have been consolidated, instructions pertaining only to individuals have been removed, and additional examples have been added. Checkboxes have been added to indicate the status of the legal action. The requirement to list the terms of any assignment or settlement has been removed.

In Part 4, *Certain Gifts and Charitable Contributions*, instructions pertaining only to individuals have been removed, and the reporting threshold has been changed to \$1,000 per recipient. The look-back period has been increased from one to two years.

Part 5, *Certain Losses*, has been revised to expand the types of payments for losses, and an instruction has been added to list unpaid claims on Official Form 206A/B (*Schedule A/B – Property*). Portions of the instructions that pertain only to

April 22-23, 2014 Page 318 of 596

individuals have been removed. Losses due to gambling have been excluded from this part.

In Part 6, Certain Payments or Transfers, the questions regarding payments related to bankruptcy, payments to self-settled trusts, and other payments or transfers have been consolidated. Instructions and questions that relate only to individuals have been eliminated. An instruction has been added to include payments related to restructuring, and the email or website of the person who received the money or transfer is added as a requirement. In response to the question regarding self-settled trusts and other transfers not already listed, debtors are instructed to include payments or transfers of property made by a person acting on behalf of the debtor. A requirement has been added to the question regarding self-settled trusts to list the name of the trustee. The relationship to the debtor must be included for all transfers not already listed, as well as any debts paid in exchange. There is a reminder added not to include transfers already listed.

Part 7, *Previous Locations*, has been revised in the instructions, and information pertaining only to individuals has been deleted.

Part 8, *Healthcare Bankruptcies*, is new. Part 8 requires additional information if the debtor is primarily engaged in offering services and facilities for diagnosing or treating injury, deformity, or disease or providing any surgical, psychiatric, drug treatment or obstetric care. This part has been added to comply with the special requirements imposed by the Bankruptcy Abuse Prevention and Consumer Protection Act of 2005.

Part 9, *Personally Identifiable Information*, is also new and includes questions about pension and profit sharing plans and adds a question about whether the debtor collects and retains personally identifiable information of customers. Questions are added about whether the debtor is the plan administrator of any pension or profit sharing plan and if any such plan is terminated. Similar to Part 8, this part has been added to comply with the special requirements imposed by the Bankruptcy Abuse Prevention and Consumer Protection Act of 2005.

April 22-23, 2014 Page 319 of 596

In Part 10, Certain Financial Accounts, Safe Deposit Boxes, and Storage Units, money market accounts have been added to the examples provided for the question regarding financial accounts, and checkboxes have been added to indicate the type of account. The requirement of the date of surrender of any safe deposit box has been removed. A question has been added about whether the debtor has property kept in storage units or warehouses within one year of filing, and the debtor must provide the facility name and address, the name and address of anyone with access to the facility, the description of the contents, and whether the debtor still has the storage unit or warehouse. Facilities that are in a part of a building in which the debtor does business are excluded.

In Part 11, *Property the Debtor Holds or Controls That the Debtor Does Not Own*, an instruction has been added to include any property borrowed from, being stored for, or held in trust, and to exclude leased or rented property.

Part 12, *Details About Environmental Information*, has been revised to include new definitions of "Environmental law," "Site," and "Hazardous materials." An instruction to report all notices, releases, and proceedings known, regardless of when they occurred, has been added.

In Part 13, Details About the Debtor's Business or Connections to Any Business, questions regarding various business issues have been consolidated, and instructions that pertain only to individuals have been eliminated. The five-percent ownership limitation has been eliminated. The phrase "kept or supervised the keeping of books or account and records" has been replaced with "maintained the debtor's books and records." The instructions for the question regarding auditing or preparation of financial records have been revised to add compiling and reviewing the debtor's books of account and records. A requirement has been added to explain if the debtor's books of account and records are unavailable. The questions regarding current and former officers, directors, managing members, general partners, members in control, or controlling shareholders have combined the formerly separate corporate and partnership questions. The question regarding former officers and partners has been changed to add the

April 22-23, 2014 Page 320 of 596

requirement of indicating the start and end dates for each listing. The instruction for withdrawals from a partnership or distribution by a corporation has been changed to add salary, other compensation, and draws to the list of examples.

In Part 14, *Signature and Declaration*, the declaration under penalty of perjury has been revised in order to conform to the language of 28 U.S.C. § 1746. See Rule 1008. A statement has been added that the individual signing on behalf of the debtor has reviewed the information in the Statement of Financial Affairs and any attachments and has "a reasonable belief that the information is true and correct." The signature boxes for bankruptcy petition preparers have been eliminated, and checkboxes for the debtor to indicate whether additional pages are attached to the form have been added.

April 22-23, 2014 Page 321 of 596

April 22-23, 2014 Page 322 of 596

Information	to identify the case:				
Debtor 1	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN	
United States E	Bankruptcy Court for the	:	District of(State)	[Date case filed for chapter 7	MM / DD / YYYY OR
Case number:				[Date case filed in chapter Date case converted to chapter 7	MM / DD / YYYY

Official Form 309A (For Individuals or Joint Debtors)

Notice of Chapter 7 Bankruptcy Case — No Proof of Claim Deadline 12/15

For the debtors listed above, a case has been filed under chapter 7 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees. Read it carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtors or the debtors' property. For example, while the stay is in effect, creditors cannot sue, garnish wages, assert a deficiency, repossess property, or otherwise try to collect from the debtors. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees. Under certain circumstances, the stay may be limited to 30 days or not exist at all, although debtors can ask the court to extend or impose a stay.

The debtors are seeking a discharge. Creditors who assert that the debtors are not entitled to a discharge of any debts or who want to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadlines specified in this notice. (See line 9 for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at www.pacer.gov).

The staff of the bankruptcy clerk's office cannot give legal advice.

April 22-23, 2014

To help creditors correctly identify debtors, debtors submit full Social Security or Individual Taxpayer Identification Numbers, which may appear on a version of this notice. However, the full numbers must not appear on any document filed with the court.

Do not file this notice with any proof of claim or other filing in the case. Do not include more than the last four digits of a Social Security or Individual Taxpayer Identification Number in any document, including attachments, that you file with the court.

		About Debtor 1:	About Debtor 2:	
1.	Debtor's full name			
2.	All other names used in the last 8 years			
3.	Address		If Debtor 2 lives at a different address:	
4.	Debtor's attorney Name and address		Contact phone Email	
5.	Bankruptcy trustee Name and address		Contact phone Email	

For more information, see page 2

Official Form B309A (For Individuals or Joint Debtors) Notice of Chapter 7 Bankruptcy Case — No Proof of Claim Deadline

Page 323 of 596

De	btor Name	Case number (if known)			
6.	Bankruptcy clerk's office Documents in this case may be filed at this address. You may inspect all records filed in this case at this office or online at www.pacer.com .		Hours open Contact phone		
7.	Meeting of creditors Debtors must attend the meeting to be questioned under oath. In a joint case, both spouses must attend. Creditors may attend, but are not required to do so.	Date Time The meeting may be continued or adjourned to a later date. If so, the date will be on the court docket.	Location:		
8.	Presumption of abuse If the presumption of abuse arises, you may have the right to file a motion to dismiss the case under 11 U.S.C. § 707(b). Debtors may rebut the presumption by showing special circumstances.	[The presumption of abuse does not arise.] [The presumption of abuse arises.] [Insufficient information has been filed to permit the clerk to determine whether the presumption of abuse arises. If more complete information is filed and shows that the presumption has arisen, the clerk will notify creditors.]			
9.	Deadlines The bankruptcy clerk's office must receive these documents and any required filing fee by the following deadlines.	File by the deadline to object to discharge or to challenge whether certain debts are dischargeable: You must file a complaint: if you assert that the debtor is not entitled to receive a discharge of any debts under any of the subdivisions of 11 U.S.C. § 727(a)(2) through (7), or if you want to have a debt excepted from discharge under 11 U.S.C. § 523(a)(2), (4), or (6). You must file a motion if you assert that the discharge should be denied under § 727(a)(8) or (9).	Filing deadline:		
		Deadline to object to exemptions: The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.	Filing deadline:	30 days after the <i>conclusion</i> of the meeting of creditors	
10.	Proof of claim Please do not file a proof of claim unless you receive a notice to do so.	No property appears to be available to pay creditors. Therefore, please do not file a proof of claim now. If it later appears that assets are available to pay creditors, the clerk will send you another notice telling you that you may file a proof of claim and stating the deadline.			
11.	Creditors with a foreign address	If you are a creditor receiving a notice mailed to a foreign address, you may file a motion asking the court to extend the deadlines in this notice. Consult an attorney familiar with United States bankruptcy law if you have any questions about your rights in this case.			
12.	Exempt property	The law allows debtors to keep certain property as exempt. Fully exempt property will not be sold and distributed to creditors. Debtors must file a list of property claimed as exempt. You may inspect that list at the bankruptcy clerk's office or online at www.pacer.gov. If you believe that the law does not authorize an exemption that the debtors claim, you may file an objection. The bankruptcy clerk's office must receive the objection by the deadline to object to exemptions in line 9.			

Official Form B309A (For Individuals or Joint Debtors) Notice of Chapter 7 Bankruptcy Case — No Proof of Claim Deadline

page 2

April 22-23, 2014 Page 324 of 596

Information	to identify the case:					
Debtor 1	First Name	Middle Name	Last Name		Last 4 digits of Social Security number or ITIN	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name		Last 4 digits of Social Security number or ITIN	
United States B	Bankruptcy Court for the: _		District of	(State)	[Date case filed for chapter 7	
Case number:				-	[Date case filed in chapter	MM / DD / YYYY OR MM / DD / YYYY
					Date case converted to chapter 7	MM / DD / YYYY

Official Form 309B (For Individuals or Joint Debtors)

Notice of Chapter 7 Bankruptcy Case — Proof of Claim Deadline Set 12/15

For the debtors listed above, a case has been filed under chapter 7 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees. Read it carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtors or the debtors' property. For example, while the stay is in effect, creditors cannot sue, garnish wages, assert a deficiency, repossess property, or otherwise try to collect from the debtors. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees. Under certain circumstances, the stay may be limited to 30 days or not exist at all, although debtors can ask the court to extend or impose a stay.

The debtors are seeking a discharge. Creditors who assert that the debtors are not entitled to a discharge of any debts or who want to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadlines specified in this notice. (See line 9 for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at www.pacer.gov).

The staff of the bankruptcy clerk's office cannot give legal advice.

To help creditors correctly identify debtors, debtors submit full Social Security or Individual Taxpayer Identification Numbers, which may appear on a version of this notice. However, the full numbers must not appear on any document filed with the court.

Do not file this notice with any proof of claim or other filing in the case. Do not include more than the last four digits of a Social Security or Individual Taxpayer Identification Number in any document, including attachments, that you file with the court.

		About Debtor 1:	About	Debtor 2:
1.	Debtor's full name			
2.	All other names used in the last 8 years			
3.	Address		If Debt	or 2 lives at a different address:
4.	Debtor's attorney		Contac	t phone
	Name and address		Email	
_	Pankruptov tructoo		Contac	t phone
Э.	Bankruptcy trustee Name and address		Email	
	Name and address			

For more information, see page 2



	Debtor	Case number	SF (if known)
	Name		
6.	Bankruptcy clerk's office		Hours open
	Documents in this case may be		
	filed at this address. You may inspect all records filed in		Contact phone
	this case at this office or online at www.pacer.com .		
7.	Meeting of creditors		
	Debtors must attend the meeting	at Date Time	Location:
	to be questioned under oath. In a joint case, both spouses must attend.	Date Time The meeting may be continued or adjourned to a later date.	
	Creditors may attend, but are not required to do so.	If so, the date will be on the court docket.	
	Presumption of abuse	[The presumption of abuse does not arise.]	
0.	If the presumption of abuse arises,	[The presumption of abuse arises.]	
	you may have the right to file a	[Insufficient information has been filed to permit the clerk	to determine whether the presumption of
	motion to dismiss the case under 11 U.S.C. § 707(b). Debtors may	abuse arises. If more complete information is filed and sl	hows that the presumption has arisen, the
	rebut the presumption by showing special circumstances.	clerk will notify creditors.]	
9.	Deadlines The bankruptcy clerk's office must	File by the deadline to object to discharge or to challenge whether certain debts are dischargeable:	Filing deadline:
	receive these documents and any	You must file a complaint:	
	required filing fee by the following deadlines.	if you assert that the debtor is not entitled to receive a	
		discharge of any debts under any of the subdivisions of 11 U.S.C. § 727(a)(2) through (7), or	
		if you want to have a debt excepted from discharge under 11 U.S.C. § 523(a)(2), (4), or (6).	
		You must file a motion if you assert that	
		the discharge should be denied under § 727(a)(8) or (9).	
		Deadline for all creditors to file a proof of claim (except governmental units):	Filing deadline:
		Deadline for governmental units to file a proof of claim:	Filing deadline:
		Deadlines for filing proof of claim:	
		A proof of claim is a signed statement describing a creditor's clai	·
		notice, obtain one at www.uscourts.gov or any bankruptcy clerk's deadline, you might not be paid on your claim. To be paid, you me the schedules that the debtor filed.	·
		Secured creditors retain rights in their collateral regardless of wh	ether they file a proof of claim. Filing a proof of
		claim submits the creditor to the jurisdiction of the bankruptcy co example, a secured creditor who files a proof of claim may surre- right to a jury trial.	·
		Deadline to object to exemptions:	Filing deadline: 30 days after the conclusion
		The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.	of the meeting of creditors
10	Creditors with a foreign	If you are a creditor receiving a notice mailed to a foreign addres	s, you may file a motion asking the court to extend
10.	address	the deadlines in this notice. Consult an attorney familiar with Unit questions about your rights in this case.	· · · · · · · · · · · · · · · · · · ·
11.	Liquidation of the debtor's	The bankruptcy trustee listed on the front of this notice will collect	
	property and payment of creditors' claims	If the trustee can collect enough money, creditors may be paid so specified by the Bankruptcy Code. To ensure you receive any sh described above.	
40	Evernt property	The law allows debtors to keep certain property as exempt. Fully	
12.	Exempt property	creditors. Debtors must file a list of property claimed as exempt. office or online at www.pacer.gov . If you believe that the law doe claim, you may file an objection. The bankruptcy clerk's office must be a support of the control of the co	s not authorize an exemption that the debtors
		to exemptions in line 9.	•

Information to identify the case:	
DebtorName	EIN
United States Bankruptcy Court for the: District of (s	State) [Date case filed for chapter 7 MM / DD / YYYY OR
Case number:	[Date case filed in chapter MM / DD / YYYY
	Date case converted to chapter 7 MM / DD / YYYYY

Official Form 309C (For Corporations or Partnerships)

Notice of Chapter 7 Bankruptcy Case — No Proof of Claim Deadline 12/15

For the debtor listed above, a case has been filed under chapter 7 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees. Read it carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtor or the debtor's property. For example, while the stay is in effect, creditors cannot sue, assert a deficiency, repossess property, or otherwise try to collect from the debtor. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees.

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at www.pacer.gov).

The staff of the bankruptcy clerk's office cannot give legal advice.

Do not file this notice with any proof of claim or other filing in the case.

1.	Debtor's full name			
2.	All other names used in the last 8 years			
3.	Address			
4.	Debtor's attorney Name and address		Contact phone Email	
5.	Bankruptcy trustee Name and address		Contact phone Email	
6.	Bankruptcy clerk's office Documents in this case may be filed at this address. You may inspect all records filed in this case at this office or online at www.pacer.com .		Hours open Contact phone	
7.	Meeting of creditors The debtor's representative must attend the meeting to be questioned under oath. Creditors may attend, but are not required to do so.	at Date Time The meeting may be continued or adjourned to date. If so, the date will be on the court docket.	Location: a later	
8.	Proof of claim Please do not file a proof of claim unless you receive a notice to do so.	No property appears to be available to pay of If it later appears that assets are available to you may file a proof of claim and stating the	pay creditors, the clerk will send y	·
9.	Creditors with a foreign address	If you are a creditor receiving a notice maile extend the deadlines in this notice. Consult any questions about your rights in this case.	an attorney familiar with United Sta	

April 22-23, 2014 Page 328 of 596

Information to identify the case:			
DebtorName		EIN	_
United States Bankruptcy Court for the:	District of(State)	[Date case filed for chapter 7	MM / DD / YYYY OR
Case number:		[Date case filed in chapter	MM / DD / YYYY
		Date case converted to chapter 7	MM / DD / YYYY

Official Form 309D (For Corporations or Partnerships)

Notice of Chapter 7 Bankruptcy Case — Proof of Claim Deadline Set 12/15

For the debtor listed above, a case has been filed under chapter 7 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees. Read it carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtor or the debtor's property. For example, while the stay is in effect, creditors cannot sue, assert a deficiency, repossess property, or otherwise try to collect from the debtor. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees.

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at www.pacer.gov).

The staff of the bankruptcy clerk's office cannot give legal advice.

Do not file this notice with any proof of claim or other filing in the case.

1.	Debtor's full name			
2.	All other names used in the last 8 years			
3.	Address			
4.	Debtor's attorney Name and address		Contact phone Email	
5.	Bankruptcy trustee Name and address		Contact phone Email	
6.	Bankruptcy clerk's office Documents in this case may be filed at this address. You may inspect all records filed in this case at this office or online at www.pacer.com .		Hours open Contact phone	
7.	Meeting of creditors The debtor's representative must attend the meeting to be questioned under oath. Creditors may attend, but are not required to do so.	at Date The meeting may be cont date. If so, the date will be	Location:	

For more information, see page 2

April 22-23, 2014 Page 329 of 596

Official Form B309D (For Corporations or Partnerships) Notice of Chapter 7 Bankruptcy Case — Proof of Claim Deadline Set

B. Deadlines

The bankruptcy clerk's office must receive proofs of claim by the following deadlines.

Deadline for all creditors to file a proof of claim (except governmental units):

Deadline for governmental units to file a

proof of claim:

Filing deadline:

Filing deadline:

A proof of claim is a signed statement describing a creditor's claim. If a proof of claim form is not included with this notice, obtain one at www.uscourts.gov or any bankruptcy clerk's office. If you do not file a proof of claim by the deadline, you might not be paid on your claim. To be paid, you must file a proof of claim even if your claim is listed in the schedules that the debtor filed.

Secured creditors retain rights in their collateral regardless of whether they file a proof of claim. Filing a proof of claim submits the creditor to the jurisdiction of the bankruptcy court, with consequences a lawyer can explain. For example, a secured creditor who files a proof of claim may surrender important nonmonetary rights, including the right to a jury trial.

Creditors with a foreign address

If you are a creditor receiving a notice mailed to a foreign address, you may file a motion asking the court to extend the deadlines in this notice. Consult an attorney familiar with United States bankruptcy law if you have any questions about your rights in this case.

 Liquidation of the debtor's property and payment of creditors' claims The bankruptcy trustee listed on the front of this notice will collect and sell the debtor's property. If the trustee can collect enough money, creditors may be paid some or all of the debts owed to them, in the order specified by the Bankruptcy Code. To ensure you receive any share of that money, you must file a proof of claim, as described above.

Information	Information to identify the case:					
Debtor 1	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN		
Debtor 2 (Spouse, if filing	First Name	Middle Name	Last Name	Last 4 digits of Social Security number or ITIN		
United States	Bankruptcy Court for the: _		District of (State)		MM / DD / YYYY] OR	
Case number:				[Date case filed in chapter	MM / DD / YYYY	
				Date case converted to chapter 11	MM / DD / YYYY	

Official Form 309E (For Individuals or Joint Debtors)

Notice of Chapter 11 Bankruptcy Case

12/15

For the debtors listed above, a case has been filed under chapter 11 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors and debtors. Read it carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtors or the debtors' property. For example, while the stay is in effect, creditors cannot sue, garnish wages, assert a deficiency, repossess property, or otherwise try to collect from the debtors. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees. Under certain circumstances, the stay may be limited to 30 days or not exist at all, although debtors can ask the court to extend or impose a stay.

Confirmation of a chapter 11 plan may result in a discharge of debt. Creditors who assert that the debtors are not entitled to a discharge of any debts or who want to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadlines specified in this notice. (See line 10 below for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at www.pacer.gov).

The staff of the bankruptcy clerk's office cannot give legal advice.

To help creditors correctly identify debtors, debtors submit full Social Security or Individual Taxpayer Identification Numbers, which may appear on a version of this notice. However, the full numbers must not appear on any document filed with the court.

Do not file this notice with any proof of claim or other filing in the case. Do not include more than the last four digits of a Social Security or Individual Taxpayer Identification Number in any document, including attachments, that you file with the court.

		About Debtor 1:	About Debtor 2:	
1.	Debtor's full name			
2.	All other names used in the last 8 years			
3.	Address		If Debtor 2 lives a	t a different address:
4.	Debtor's attorney		Contact phone	
	Name and address		Email	
5.	Bankruptcy clerk's office		Hours open	
	Documents in this case may be filed at this address.		Contact phone	
	You may inspect all records filed in this case at this office or online at www.pacer.com.		·	

For more information, see page 2

page 1

6.	Meeting of creditors Debtors must attend the meeting to be questioned under oath. In a joint case, both spouses must attend. Creditors may attend, but are not required to do so.	at Date Time The meeting may be continued or adjourned to a later date. If so, the date will be on the court docket.			
7.	Deadlines The bankruptcy clerk's office must receive these documents	File by the deadline to object to discharge or to challenge whether certain debts are dischargeable: You must file a complaint:		aring on confirmation of send you a notice of that	
	and any required filing fee by the following deadlines.	if you assert that the debtor is not entitled to receive a discharge of any debts under 11 U.S.C. § 1141(d)(3) or	Filing deadline f complaints:	or dischargeability	
		if you want to have a debt excepted from discharge under 11 U.S.C. § 523(a)(2), (4), or (6).			
		Deadline for filing proof of claim:	[Not yet set. If a dea send you another no	dline is set, the court will otice.] or	
			[date, if set by the co	ourt)]	
		A proof of claim is a signed statement describing a creditor's claim included with this notice, obtain one at www.uscourts.gov or any b	•		
		Your claim will be allowed in the amount scheduled unless:			
		 your claim is designated as disputed, contingent, or unliquidate you file a proof of claim in a different amount; or you receive another notice. 			
			your claim is designated as disputed, contingent, or unliquidated, you ight not be paid on your claim and you might be unable to vote on a even if your claim is scheduled.		
		You may review the schedules at the bankruptcy clerk's office or online at www.pacer.gov .			
		claim submits a creditor to the jurisdiction of the bankruptcy court,	editors retain rights in their collateral regardless of whether they file a proof of claim. Filing a proof its a creditor to the jurisdiction of the bankruptcy court, with consequences a lawyer can explain. F secured creditor who files a proof of claim may surrender important nonmonetary rights, including ry trial.		
		Deadline to object to exemptions: The law permits debtors to keep certain property as exempt. If you believe that the law does not authorize an exemption claimed, you may file an objection.	Filing deadline:	30 days after the conclusion of the meeting of creditors	
8.	Creditors with a foreign address	If you are a creditor receiving mailed notice at a foreign address, yethe deadlines in this notice. Consult an attorney familiar with Unite questions about your rights in this case.			
9.	Filing a Chapter 11 bankruptcy case	Chapter 11 allows debtors to reorganize or liquidate according to confirms it. You may receive a copy of the plan and a disclosure s may have the opportunity to vote on the plan. You will receive noti you may object to confirmation of the plan and attend the confirmatebtor will remain in possession of the property and may continue	tatement telling you a ce of the date of the ation hearing. Unless	about the plan, and you confirmation hearing, and a trustee is serving, the	
10.	Discharge of debts	Confirmation of a chapter 11 plan may result in a discharge of det 11 U.S.C. § 1141(d). However, unless the court orders otherwise, payments under the plan are made. A discharge means that credi debtors personally except as provided in the plan. If you believe the excepted from the discharge under 11 U.S.C. § 523 (a)(2), (4), or fee in the bankruptcy clerk's office by the deadline. If you believe to fany of their debts under 11 U.S.C. § 1141 (d)(3), you must file a office by the first date set for the hearing on confirmation of the platelling you of that date.	the debts will not be tors may never try to nat a particular debt of (6), you must file a contact the debtors are not complaint and pay the	discharged until all collect the debt from the bwed to you should be complaint and pay the filing to entitled to a discharge the filing fee in the clerk's	
11.	Exempt property	The law allows debtors to keep certain property as exempt. Fully of to creditors, even if the case is converted to chapter 7. Debtors me may inspect that list at the bankruptcy clerk's office or online at wo not authorize an exemption that the debtors claim, you may file an receive the objection by the deadline to object to exemptions in line.	ust file a list of proper <u>vw.pacer.gov</u> . If you b objection. The bank	rty claimed as exempt. You believe that the law does	

Information to identify the case:			
DebtorName	· 	EIN	_
United States Bankruptcy Court for the:	District of (State)	[Date case filed in chapter 11	MM / DD / YYYY OR
Case number:		[Date case filed in chapter	MM / DD / YYYY
		Date case converted to chapter 11	MM / DD / YYYY

Official Form 309F (For Corporations or Partnerships)

Notice of Chapter 11 Bankruptcy Case

12/15

For the debtor listed above, a case has been filed under chapter 11 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees. Read it carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtor or the debtor's property. For example, while the stay is in effect, creditors cannot sue, assert a deficiency, repossess property, or otherwise try to collect from the debtor. Creditors cannot demand repayment from the debtor by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees.

Confirmation of a chapter 11 plan may result in a discharge of debt. A creditor who wants to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadline specified in this notice. (See line 11 below for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at www.pacer.gov).

The staff of the bankruptcy clerk's office cannot give legal advice.

Do not file this notice with any proof of claim or other filing in the case.

1.	Debtor's full name				
2.	All other names used in the last 8 years				
3.	Address				
4	Debtor's attorney			Contact phone	
4.	Name and address			Email	
5.	Bankruptcy clerk's office Documents in this case may be filed at this address. You may inspect all records filed in this case at this office or online at www.pacer.com .			Hours open Contact phone	
6.	Meeting of creditors The debtor's representative must attend the meeting to be questioned under oath. Creditors may attend, but are not required to do so.	at Date The meeting may be cont date. If so, the date will be	Time tinued or adjourned to a latere on the court docket.	Location:	

For more information, see page 2

Official Form B309F (For Corporations or Partnerships) Notice of Chapter 11 Bankruptcy Case

page 1

April 22-23, 2014 Page 333 of 596

7. Proof of claim deadline Deadline for filing proof of claim: [Not yet set. If a deadline is set, the court will send you another notice.] or [date, if set by the court)] A proof of claim is a signed statement describing a creditor's claim. If a proof of claim form is not included with this notice, obtain one at www.uscourts.gov or any bankruptcy clerk's office. Your claim will be allowed in the amount scheduled unless: vour claim is designated as disputed, contingent, or unliquidated; you file a proof of claim in a different amount; or you receive another notice. If your claim is not scheduled or if your claim is designated as disputed, contingent, or unliquidated, you must file a proof of claim or you might not be paid on your claim and you might be unable to vote on a plan. You may file a proof of claim even if your claim is scheduled. You may review the schedules at the bankruptcy clerk's office or online at www.pacer.gov. Secured creditors retain rights in their collateral regardless of whether they file a proof of claim. Filing a proof of claim submits a creditor to the jurisdiction of the bankruptcy court, with consequences a lawyer can explain. For example, a secured creditor who files a proof of claim may surrender important nonmonetary rights, including the right to a jury trial. 8. Exception to discharge You must start a judicial proceeding by filing a complaint if you want to have a debt excepted from discharge under 11 U.S.C. § 1141(d)(6)(A). deadline The bankruptcy clerk's office Deadline for filing the complaint: must receive a complaint and any required filing fee by the following deadline. If you are a creditor receiving notice mailed to a foreign address, you may file a motion asking the court to 9. Creditors with a foreign extend the deadlines in this notice. Consult an attorney familiar with United States bankruptcy law if you have address any questions about your rights in this case. Chapter 11 allows debtors to reorganize or liquidate according to a plan. A plan is not effective unless the court 10. Filing a Chapter 11 confirms it. You may receive a copy of the plan and a disclosure statement telling you about the plan, and you bankruptcy case may have the opportunity to vote on the plan. You will receive notice of the date of the confirmation hearing, and you may object to confirmation of the plan and attend the confirmation hearing. Unless a trustee is serving, the debtor will remain in possession of the property and may continue to operate its business. 11. Discharge of debts Confirmation of a chapter 11 plan may result in a discharge of debts, which may include all or part of your debt. See 11 U.S.C. § 1141(d). A discharge means that creditors may never try to collect the debt from the debtor except as provided in the plan. If you want to have a particular debt owed to you excepted from the discharge under 11 U.S.C. § 1141(d)(6)(A), you must start a judicial proceeding by filing a complaint and paying the filing fee in the bankruptcy clerk's office by the deadline.

April 22-23, 2014 Page 334 of 596

Information	to identify the case:					
Debtor 1	First Name	Middle Name	Last Name		Last 4 digits of Social Security number or ITIN	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name		Last 4 digits of Social Security number or ITIN	
United States E	Bankruptcy Court for the: _		District o	f (State)	[Date case filed for chapter 12	MM / DD / YYYY OR
Case number:				-	[Date case filed in chapter Date case converted to chapter 12	MM / DD / YYYY

Official Form 309G (For Individuals or Joint Debtors)

Notice of Chapter 12 Bankruptcy Case

12/15

For the debtors listed above, a case has been filed under chapter 12 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees. Read it carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtors, from the debtors' property, or from certain codebtors. For example, while the stay is in effect, creditors cannot sue, garnish wages, assert a deficiency, repossess property, or otherwise try to collect from the debtors. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees.

Confirmation of a chapter 12 plan may result in a discharge of debt. Creditors who want to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadline specified in this notice. (See line 13 below for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at www.pacer.gov).

The staff of the bankruptcy clerk's office cannot give legal advice.

To help creditors correctly identify debtors, debtors submit full Social Security or Individual Taxpayer Identification Numbers, which may appear on a version of this notice. However, the full numbers must not appear on any document filed with the court.

Do not file this notice with any proof of claim or other filing in the case. Do not include more than the last four digits of a Social Security or Individual Taxpayer Identification Number in any document, including attachments, that you file with the court.

	About Debtor 1:	About Debtor 2:
1.	Debtor's full name	
2.	All other names used in the last 8 years	
3.	Address	If Debtor 2 lives at a different address:
4.	Debtor's attorney	Contact phone
	Name and address	Email
5.	Bankruptcy trustee	Contact phone
	Name and address	Email
6.	Bankruptcy clerk's office Documents in this case may be	Hours open
	filed at this address. You may inspect all records filed	Contact phone
	in this case at this office or online at www.pacer.com .	

For more information, see page 2

page 1

April 22-23, 2014 Page 335 of 596

Official Form B309G (For Individuals or Joint Debtors) Notice of Chapter 12 Bankruptcy Case

Debtor	Case number (if known)
	, ,

7.	Meeting of creditors		
	Debtors must attend the	at Date Time	Location:
	meeting to be questioned under oath. In a joint case, both	Time	
	spouses must attend.	The meeting may be continued or adjourned to a later date. If so,	
	Creditors may attend, but are not required to do so.	the date will be on the court docket.	
8.	Deadlines	Deadline to file a complaint to challenge	Filing deadline:
	The bankruptcy clerk's office	dischargeability of certain debts:	
	must receive these documents and any required filing fee by the following deadlines.	You must start a judicial proceeding by filing a complaint if you want to have a debt excepted from	
	ronowing deadnines.	discharge under 11 U.S.C. § 523(a)(2), (4), or (6).	
		Deadline for all creditors to file a proof of claim (except governmental units):	Filing deadline:
		Deadline for governmental units to file a proof of claim:	Filing deadline:
		Deadlines for filing proof of claim:	
		A proof of claim is a signed statement describing a creditor's clanotice, obtain one at www.uscourts.gov or any bankruptcy clerk	
		If you do not file a proof of claim by the deadline, you might not proof of claim even if your claim is listed in the schedules that the	
		Secured creditors retain rights in their collateral regardless of w claim submits the creditor to the jurisdiction of the bankruptcy c example, a secured creditor who files a proof of claim may surright to a jury trial.	ourt, with consequences a lawyer can explain. For
		Deadline to object to exemptions:	Filing deadline: 30 days after the conclusion of the meeting
		The law permits debtors to keep certain property as exempt.	of creditors
		If you believe that the law does not authorize an exemption claimed, you may file an objection.	
9.	Filing of plan	[The debtor has filed a plan. The plan or a summary of the plan held on: at Loca	is enclosed. The hearing on confirmation will be ation:
		Date Time]	
		Or [The debtor has filed a plan. The plan or a summary of the p	lan and notice of confirmation hearing will be sent
		separately.]	ian and house of community many will be com-
		Or [The debtor has not filed a plan as of this date. A copy of the	e plan or summary and a notice of the hearing on
		confirmation will be sent separately.]	,
10	. Creditors with a foreign address	If you are a creditor receiving a notice mailed to a foreign address extend the deadlines in this notice. Consult an attorney familiar questions about your rights in this case.	
11	. Filing a Chapter 12 bankruptcy case	Chapter 12 allows family farmers and family fishermen to reorgunless the court confirms it. You may receive a copy of the plan attend the confirmation hearing. The debtor will remain in posse the business unless the court orders otherwise.	. You may object to confirmation of the plan and
12	. Discharge of debts	Confirmation of a chapter 12 plan may result in a discharge of cultiless the court orders otherwise, the discharge will not be effect discharge means that you may never try to collect the debt from want to have a particular debt excepted under 11 U.S.C. § 523 proceeding by filing a complaint and paying the filing fee in the	ective until all payments under the plan are made. A the debtor except as provided in the plan. If you (a)(2), (4), or (6), you must start a judicial
13	. Exempt property	The law allows debtors to keep certain property as exempt. Full to creditors, even if the case is converted to chapter 7. Debtors may inspect that list at the bankruptcy clerk's office. If you belie that the debtors claim, you may file an objection. The bankruptcy deadline to object to exemptions in line 8.	must file a list of property claimed as exempt. You ve that the law does not authorize an exemption

Official Form B309G (For Individuals or Joint Debtors) Notice of Chapter 12 Bankruptcy Case

page 2

April 22-23, 2014 Page 336 of 596

Information to identify the case:			
DebtorName		EIN	_
United States Bankruptcy Court for the:	District of(State)	[Date case filed for chapter 12	MM /DD/YYYY OR
Case number:		[Date case filed in chapter	MM / DD / YYYY
		Date case converted to chapter 12	MM / DD / YYYY

Official Form 309H (For Corporations or Partnerships)

Notice of Chapter 12 Bankruptcy Case

12/15

For the debtor listed above, a case has been filed under chapter 12 of the Bankruptcy Code. An order for relief has been entered.

This notice has important information about the case for creditors, debtors, and trustees. Read it carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtor, the debtor's property, or certain codebtors. For example, while the stay is in effect, creditors cannot sue, assert a deficiency, repossess property, or otherwise try to collect from the debtor. Creditors cannot demand repayment from the debtor by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees.

Confirmation of a chapter 12 plan may result in the discharge of debt. Creditors who want to have a particular debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office within the deadline specified in this notice. (See line 13 below for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at www.pacer.gov).

The staff of the bankruptcy clerk's office cannot give legal advice.

Do not file this notice with any proof of claim or other filing in the case.

Official Form B309H (For Corporations or Partnerships) Notice of Chapter 12 Bankruptcy Case

1.	Debtor's full name		
2.	All other names used in the last 8 years		
3.	Address		
4.	Debtor's attorney Name and address	Contact phone Email	
5.	Bankruptcy clerk's office Documents in this case may be filed at this address. You may inspect all records filed in this case at this office or online at www.pacer.com .	Hours open Contact phone	
6.	Bankruptcy trustee Name and address	Contact phone Email	

For more information, see page 2

Page 337 of 596

7.	Meeting of creditors The debtor's representative must attend the meeting to be questioned under oath. Creditors may attend, but are not required to do so.	Date Time The meeting may be continued or adjourned to a later date. If so, the date will be on the court docket.	Location:
8.	Exception to discharge deadline The bankruptcy clerk's office must receive a complaint and any required filing fee by the following deadline.	You must start a judicial proceeding by filing a complaint if you want to have a debt excepted from discharge under 11 U.S.C. § 523(a)(2), (4), or (6).	Deadline for filing the complaint:
9.	Filing of plan	[The debtor has filed a plan. The plan or a summary of the plan is held on: at Location Date Time] Or [The debtor has filed a plan. The plan or a summary of the plan sent separately.]	on:
		Or [The debtor has not filed a plan as of this date. A copy of the proof confirmation will be sent separately.]	olan or summary and a notice of the hearing on
10	. Deadlines	Deadline for all creditors to file a proof of claim (except governmental units):	Filing deadline:
		Deadline for governmental units to file a proof of claim:	Filing deadline:
		A proof of claim is a signed statement describing a creditor's claim this notice, obtain one at www.uscourts.gov or any bankruptcy claim you do not file a proof of claim by the deadline, you might not be a proof of claim even if your claim is listed in the schedules that the	erk's office. e paid on your claim. To be paid, you must file
		Secured creditors retain rights in their collateral regardless of who claim submits the creditor to the jurisdiction of the bankruptcy confor example, a secured creditor who files a proof of claim may suincluding the right to a jury trial.	urt, with consequences a lawyer can explain.
11	. Creditors with a foreign address	If you are a creditor receiving a notice mailed to a foreign address extend the deadlines in this notice. Consult an attorney familiar vany questions about your rights in this case.	
12	. Filing a chapter 12 bankruptcy case	Chapter 12 allows family farmers and family fishermen to reorgar unless the court confirms it. You may receive a copy of the plan. attend the confirmation hearing. The debtor will remain in posses operate the business.	You may object to confirmation of the plan and
13	. Discharge of debts	Confirmation of a chapter 12 plan may result in a discharge of de Unless the court orders otherwise, the discharge will not be effect A discharge means that you may never try to collect the debt from	tive until all payments under the plan are made.
		If you want to have a particular debt excepted from discharge unstart a judicial proceeding by filing a complaint and paying the filing deadline.	

Page 338 of 596 April 22-23, 2014

Information	to identify the case:					
Debtor 1	First Name	Middle Name	Last Name		Last 4 digits of Social Security number or ITIN	
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name		Last 4 digits of Social Security number or ITIN	
United States	Bankruptcy Court for the: _		District o	(State)	[Date case filed for chapter 13	MM / DD / YYYY OR
Case number:				-	[Date case filed in chapter	MM / DD / YYYY
					Date case converted to chapter 13	MM / DD / YYYY

Official Form 309I

Notice of Chapter 13 Bankruptcy Case

12/15

For the debtors listed above, a case has been filed under chapter 13 of the Bankruptcy Code. An order for relief has been entered

This notice has important information about the case for creditors, debtors, and trustees. Read it carefully.

The filing of the case imposed an automatic stay against most collection activities. This means that creditors generally may not take action to collect debts from the debtors, the debtors' property, and certain codebtors. For example, while the stay is in effect, creditors cannot sue, garnish wages, assert a deficiency, repossess property, or otherwise try to collect from the debtors. Creditors cannot demand repayment from debtors by mail, phone, or otherwise. Creditors who violate the stay can be required to pay actual and punitive damages and attorney's fees. Under certain circumstances, the stay may be limited to 30 days or not exist at all, although debtors can ask the court to extend or impose a stay.

Confirmation of a chapter 13 plan may result in a discharge. Creditors who assert that the debtors are not entitled to a discharge under 11 U.S.C. § 1328(f) must file a motion objecting to discharge in the bankruptcy clerk's office within the deadline specified in this notice. Creditors who want to have their debt excepted from discharge may be required to file a complaint in the bankruptcy clerk's office by the same deadline. (See line 14 below for more information.)

To protect your rights, consult an attorney. All documents filed in the case may be inspected at the bankruptcy clerk's office at the address listed below or through PACER (Public Access to Court Electronic Records at www.pacer.gov).

The staff of the bankruptcy clerk's office cannot give legal advice.

To help creditors correctly identify debtors, debtors submit full Social Security or Individual Taxpayer Identification Numbers, which may appear on a version of this notice. However, the full numbers must not appear on any document filed with the court.

Do not file this notice with any proof of claim or other filing in the case. Do not include more than the last four digits of a Social Security or Individual Taxpayer Identification Number in any document, including attachments, that you file with the court.

	About Debtor 1:	About Debtor 2:
1.	Debtor's full name	
2.	All other names used in the last 8 years	
3.	Address	If Debtor 2 lives at a different address:
4.	Debtor's attorney Name and address	Contact phone Email
5.	Bankruptcy trustee Name and address	Contact phone
6.	Bankruptcy clerk's office Documents in this case may be filed at this address. You may inspect all records filed in this case at this office or online at www.pacer.com .	Hours open Contact phone

For more information, see page 2

page 1

Official Form B309l April 22-23, 2014 exemptions in line 8.

you believe that the debtors are not entitled to a discharge of any of their debts under 11 U.S.C. § 1328(f), you must file a motion. The bankruptcy clerk's office must receive the objection by the deadline to object to

COMMITTEE NOTE

Official Forms 309A-I, collectively the Bankruptcy Case Commencement Notices, have been revised as part of the Forms Modernization Project to make them easier to read and understand. The notices, derived from Official Forms 9A-I are renumbered and stylistic changes have been made.

References to the limitations on the automatic stay imposed by 11 U.S.C. § 362(c)(3) and (4) in some repeat bankruptcy filings by individuals have been deleted from the three versions of the notice for cases filed by corporations and partnerships. Email addresses for the debtor's attorney and the trustee have been added to the form.

The parties are informed that they may review papers filed in the case through the judiciary's PACER system (Public Access to Court Electronic Records) as well as at the bankruptcy clerk's office.

The lettering scheme for the versions of Official Form 309 track the versions of Official Form 9 used in different types of bankruptcy cases with following exceptions. Official Forms 9E(Alt.) and 9F(Alt.) have been eliminated by including alternative language in Official Forms 309E and 309F to be used if the court sets a deadline for filing claims at the start of the chapter 11 case. In addition, the B and C versions have been reversed in order. That is, Official Form 9C has been designated 309B and Official Form 9B as 309C. This groups together the notices for chapter 7 individual debtors and for non-individual debtors. Finally, as a result of the reformatting, Official Form 309C has been reduced to a single page.

The four versions of the form for chapter 7 cases have been renamed to state whether the notice specifies a deadline for filing proofs of claim, rather than whether the case is an "asset" or "no-asset" case.

April 22-23, 2014 Page 341 of 596

April 22-23, 2014 Page 342 of 596

Draft March 24, 2014 Official Form 312 (12/15)

[Caption as in 416A]

Order and Notice for Hearing on Disclosure Statement

To the	the debtor, its creditors, and other parties in interest:						
	A disclosure statement and a plan under chapter 11 [or chapter 9] of the Bankruptcy Code having been filed by on,						
IT IS (IS ORDERED and notice is hereby given, that:						
1.	1. The hearing to consider the approval of the disclosure statement shall be held at:						
-	on, at o'clockm.						
2.	2 is fixed as the last day for filing and serving in accordance with Fed. R. Bankr. P. 3017(a) written objections to the disclosure statement.						
3.	 Within days after entry of this order, the disclosure statement and plan shall be distributed in accordance with Fed. R. Bankr. P. 3017(a). 						
4.	4. Requests for copies of the disclosure statement and plan shall be mailed to the debtor in possession [or trustee or debtor or] at the following mailing address:						
	[].						
	By the court:						
	MM / DD / YYYY United States Bank						

April 22-23, 2014 Page 343 of 596

April 22-23, 2014 Page 344 of 596

COMMITTEE NOTE

Official Form 312, *Order and Notice for Hearing on Disclosure Statement* replaces Official Form 12, *Order and Notice for Hearing on Disclosure Statement*. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

April 22-23, 2014 Page 345 of 596

April 22-23, 2014 Page 346 of 596

Draft March 24, 2014 Official Form 313 (12/15)

[Caption as in 416A]

Order Approving Disclosure Statement and Fixing Time for Filing Acceptances or Rejections of Plan, Combined with Notice Thereof

	losure statement under chapter			•	r. i.
appro,	oriate, and byer 11 of the Code filed by, on		, on , on] [, referring to a plan under <i>if appropriat</i> e, and by
	cation filed on		163p66tiv6ty] [# 6	дрргорнаю, а	3 modified by a
It havi	ng been determined after hearing ation:	g on notice that the	disclosure statemer	nt [or statemer	nts] contain[s] adequate
IT IS (ORDERED, and notice is hereby	given, that:			
A.	The disclosure statement filed and by	•			_[if appropriate,
В.	plan [or plans] referred to abov		ay for filing written a	acceptances c	or rejections of the
C.	Within days after the thereof approved by the court, any, dated, approved [or statements], and a ballot co (Official Form 314) shall be made and shall be transmitted to the	[and [if appropriate] ing the disclosure standard informing to <i>Ballot fo</i> iled to creditors, equ	a summary approvent atement [or statement or Accepting or Reje uity security holders,	ed by the cour ents]], the disc cting Plan of I , and other pa	t of its opinion, if closure statement Reorganization rties in interest,
D.	If acceptances are filed for mor indicated.	e than one plan, pre	ferences among the	e plans so acc	cepted may be
E.	[If appropriate]	is fixed for the hea	aring on confirmatio	on of the plan	[or plans].
F.	[If appropriate] Bankr. P. 3020(b)(1) written ob		, ,	serving pursu	ant to Fed. R.
			By the court	:	
	MM / DD / YYYY			United State	es Bankruptcy Judge

[If the court directs that a copy of the opinion should be transmitted in lieu of or in addition to the summary thereof, the appropriate change should be made in paragraph C of this order.]

April 22-23, 2014 Page 347 of 596

April 22-23, 2014 Page 348 of 596

COMMITTEE NOTE

Official Form 313, Order Approving Disclosure Statement and Fixing Time for Filing Acceptances or Rejections of Plan, Combined with Notice Thereof replaces Official Form 13, Order Approving Disclosure Statement and Fixing Time for Filing Acceptances or Rejections of Plan, Combined with Notice Thereof. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

April 22-23, 2014 Page 349 of 596

April 22-23, 2014 Page 350 of 596

Official Form 314 (12/15)

[Caption as in 416A]

Class [] Ballot for Accepting or Rejecting Plan of Reorganization

[Proponent] filed a plan of reorganization dated [Date] (the *Plan*) for the Debtor in this case. The Court has [conditionally] approved a disclosure statement with respect to the Plan (the *Disclosure Statement*). The Disclosure Statement provides information to assist you in deciding how to vote your ballot. If you do not have a Disclosure Statement, you may obtain a copy from [name, address, telephone number and telecopy number of proponent/proponent's attorney.]

Court approval of the disclosure statement does not indicate approval of the Plan by the Court.

You should review the Disclosure Statement and the Plan before you vote. You may wish to seek legal advice concerning the Plan and your classification and treatment under the Plan. Your [claim] [equity interest] has been placed in class [] under the Plan. If you hold claims or equity interests in more than one class, you will receive a ballot for each class in which you are entitled to vote.

If your ballot is not received by [name and address of proponent's attorney or other appropriate address] on or before [date], and such deadline is not extended, your vote will not count as either an acceptance or rejection of the Plan.

If the Plan is confirmed by the Bankruptcy Court, it will be binding on you whether or not you vote.

Acceptance or Rejection of the Plan

[At this point the ballot should provide for voting by the particular class of creditors or equity holders receiving the ballot using one of the following alternatives;]

[If the voter is the holder of a secured, priority, or unsecured nonpriority claim:]

The undersigned, the holder of a Class [] claim against the Debtor in the unpaid amount of Dollars (\$)

[or, if the voter is the holder of a bond, debenture, or other debt security:]

The undersigned, the holder of a Class [] claim against the Debtor, consisting of Dollars (\$) principal amount of [describe bond, debenture, or other debt security] of the Debtor (For purposes of this Ballot, it is not necessary and you should not adjust the principal amount for any accrued or unmatured interest.)

[or, if the voter is the holder of an equity interest:]

The undersigned, the holder of Class [] equity interest in the Debtor, consisting of _____ shares or other interests of [describe equity interest] in the Debtor Official Form 14 continued (12/03)

April 22-23, 2014 Page 351 of 596

[In each case, the following language should be included:]							
Check one box only Accepts the pl Rejects the pla							
Address: _							
Return this ballot to) :						
[Name and address of proponent's attorney or other appropriate address]							

April 22-23, 2014 Page 352 of 596

COMMITTEE NOTE

Official Form 314, *Ballot for Accepting or Rejecting Plan* replaces Official Form 14, *Ballot for Accepting or Rejecting Plan*. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

April 22-23, 2014 Page 353 of 596

April 22-23, 2014 Page 354 of 596

Official Form 315 (12/15)

[Caption as in 416A]

Order Confirming Plan

The plan under chapter 11 of the Bankruptcy Code filed	by	, on
[if applicable, as modified by a	modification filed on	,] or a
summary thereof, having been transmitted to creditors a	and equity security holders; and	
t having been determined after hearing on notice that th 11 U.S.C. § 1129(a) [or, if appropriate, 11 U.S.C. § 1129	•	th in
T IS ORDERED that:		
The plan filed by	, on	,
If appropriate, include dates and any other pertinent det the plan provides for an injunction against conduct not o information required by Rule 3020.]	· · · · · · · · · · · · · · · · · · ·	-
A copy of the confirmed plan is attached.		

April 22-23, 2014 Page 355 of 596

April 22-23, 2014 Page 356 of 596

COMMITTEE NOTE

Official Form 315, *Order Confirming Plan* replaces Official Form 15, *Order Confirming Plan*. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

April 22-23, 2014 Page 357 of 596

April 22-23, 2014 Page 358 of 596

Draft March 20, 2014

Fill in this information to identify the case:		
United States Bankruptcy Court for the:		
District of(State)		
Case number (If known):	Chapter 15	

☐ Check if this is an amended filing

Official Form 401

Chapter 15 Petition for Recognition of a Foreign Proceeding 12/15

If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write debtor's name and case number (if known).

1.	Debtor's name			
2.	Debtor's unique identifier	For non-individual debtors:		
		Federal Employer Identification Number (EIN) _		
		Other Describ	e identifier	
		For individual debtors:		
		Social Security number: xxx - xx		
		Individual Taxpayer Identification number (ITIN):	9 xx - xx	
		Other Describ	e identifier	
3.	Name of foreign representative(s)			
4.	Foreign proceeding in which appointment of the foreign representative(s) occurred			
5.	Nature of the foreign proceeding	Check one:		
	p.cocoug	oreign main proceeding		
		oreign nonmain proceeding or in the alternative foreign n	onmain proceeding	
6.	Evidence of the foreign proceeding	certified copy, translated into English, of the decision pointing the foreign representative is attached.	commencing the foreign proceeding and	
		certificate, translated into English, from the foreign co oceeding and of the appointment of the foreign repre	urt, affirming the existence of the foreign sentative, is attached.	
		ther evidence of the existence of the foreign proceedi presentative is described below, and relevant documents.		
7.	Is this the only foreign proceeding with respect to	o. (Attach a statement identifying each country in which	ch a foreign proceeding by, regarding, or against the	
	the debtor known to the foreign representative(s)?	es		

Debtor Name	Name Case number (if known)			
8. Others entitled to notice	Attach a list containing the names and addresses of:			
	(i) all persons or bodies authorized to administer foreign proceedings of the debtor,(ii) all parties to litigation pending in the United States in which the debtor is a party at the time of filing of thi petition, and			
	(iii) all entities against whom provisional relief is bein	ng sought under § 1519 of the Bankruptcy Code.		
9. Addresses	Country where the debtor has the center of its main interests:	Debtor's registered office:		
		Number Street		
		P.O. Box		
		City State/Province/Region ZIP/Postal Code		
		Country		
	Individual debtor's habitual residence:	Address of foreign representative(s):		
	Number Street	Number Street		
	P.O. Box	P.O. Box		
	City State/Province/Region ZIP/Postal Code	City State/Province/Region ZIP/Postal Code		
	Country	Country		
10. Debtor's website (URL)				
11. Type of debtor	Check one:			
, po oi wow.oi	☐ Non-individual (<i>check one</i>):			
	Corporation. Attach a corporate ownership described in Fed. R. Bankr. P. 7007.1.	o statement containing the information		
	Partnership			
	Other. Specify:			
	☐ Individual			

Debtor Name	Case n	umber (if known)						
12. Why is venue proper in <i>this</i> district?	 Debtor's principal place of business or principal assets in the United States are in this district. Debtor does not have a place of business or assets in the United States, but the following action or proceeding in a federal or state court is pending against the debtor in this district: 							
	If neither box is checked, venue is consistent with the of the parties, having regard to the relief sought by the	e interests of justice and the convenience he foreign representative, because:						
13. Signature of foreign representative(s)								
	I am the foreign representative of a debtor in a foreign pro sought in this petition, and I am authorized to file this peti							
	I have examined the information in this petition and have and correct.	a reasonable belief that the information is true						
	I declare under penalty of perjury that the foregoing is true	e and correct,						
	x							
	Signature of foreign representative	Printed name						
	Executed onMM / DD / YYYY							
	x							
	Signature of foreign representative	Printed name						
	Executed onMM / DD / YYYY							
14. Signature of attorney	Signature of Attorney for foreign representative	_ Date MM / DD / YYYY						
	Printed name							
	Firm name							
	Number Street							
	City	State ZIP Code						
	Contact phone	Email address						
	Bar number	State						

April 22-23, 2014 Page 362 of 596

COMMITTEE NOTE

Official Form 401 is required for any petition seeking recognition of a foreign proceeding under chapter 15 of the Bankruptcy Code. The form, which applies to foreign proceedings involving individual and non-individual debtors, consolidates information formerly included on Official Form 1 (Voluntary Petition). The petition must be signed by the foreign representative, under penalty of perjury, and by the foreign representative's attorney.

The petition requires disclosure of the foreign proceeding in which the foreign representative has been appointed (Line 4) and whether it is a foreign main proceeding or foreign nonmain proceeding (Line 5). If the foreign representative seeks recognition of the foreign proceeding as a foreign main proceeding or, in the alternative, a foreign nonmain proceeding, that request should be indicated in Line 5. Each country where any additional foreign proceeding known to the foreign representative is pending must be disclosed on Line 7. See Bankruptcy Rule 1004.2. Evidence of the foreign proceeding and of the foreign representative's appointment must accompany the petition. See 11 U.S.C. § 1515(b). These documents must be translated into English in accordance with 11 U.S.C. § 1515(d). The foreign representative must also attach a list of persons or bodies entitled to notice. See Bankruptcy Rule 2002(q).

The petition calls for information about the debtor, including the debtor's name (Line 1), other unique identifying information, if available (Line 2), and center of main interest (Line 9). The type of debtor is also requested (Line 11).

The foreign representative must indicate the basis for venue in the district by selecting an appropriate checkbox and, if necessary, providing additional information, such as a statement explaining why venue in the district is appropriate (Line 12). See 28 U.S.C. § 1410.

April 22-23, 2014 Page 363 of 596

April 22-23, 2014 Page 364 of 596

12/15

Official Form 410

Instructions for Proof of Claim

United States Bankruptcy Court

These instructions and definitions generally explain the law. In certain circumstances, such as bankruptcy cases that debtors do not file voluntarily, exceptions to these general rules may apply. You should consider obtaining the advice of an attorney, especially if you are unfamiliar with the bankruptcy process and privacy regulations.

A person who files a fraudulent claim could be fined up to \$500,000, imprisoned for up to 5 years, or both. 18 U.S.C. §§ 152, 157 and 3571.

How to fill out this form

- Fill in all of the information about the claim as of the date the case was filed.
- Attach any supporting documents to this form. Attach redacted copies of any documents that show that the debt exists, a lien secures the debt, or both. (See the definition of *redaction* on the next page.)

Also attach copies of documents that show perfection of any security interest or any assignments or transfers of the debt. In addition to the documents, a summary may be added. Federal Rule of Bankruptcy Procedure (called "Bankruptcy Rule") 3001(c) and (d).

- Do not attach original documents because attachments may be destroyed after scanning.
- If the claim is based on delivering health care goods or services, do not disclose confidential health care information. Leave out or redact confidential information both in the claim and in the attached documents.
- A Proof of Claim form and any attached documents must show only the last 4 digits of any social security number, individual's tax identification number, or financial account number, and only the year of any person's date of birth. See Bankruptcy Rule 9037.
- For a minor child, fill in only the child's initials and the full name and address of the child's parent or guardian. For example, write A.B., a minor child (John Doe, parent, 123 Main St., City, State). See Bankruptcy Rule 9037.

Confirmation that the claim has been filed

To receive confirmation that the claim has been filed, either enclose a stamped self-addressed envelope and a copy of this form or go to the court's PACER system (www.pacer.psc.uscourts.gov) to view the filed form.

Understand the terms used in this form

Administrative expense: Generally, an expense that arises after a bankruptcy case is filed in connection with operating, liquidating, or distributing the bankruptcy estate. 11 U.S.C. § 503.

Claim: A creditor's right to receive payment for a debt that the debtor owed on the date the debtor filed for bankruptcy. 11 U.S.C. §101 (5). A claim may be secured or unsecured.

Creditor: A person, corporation, or other entity to whom a debtor owes a debt that was incurred on or before the date the debtor filed for bankruptcy. 11 U.S.C. §101 (10).

Debtor: A person, corporation, or other entity who is in bankruptcy. Use the debtor's name and case number as shown in the bankruptcy notice you received. 11 U.S.C. § 101 (13).

Evidence of perfection: Evidence of perfection of a security interest may include a mortgage; lien; certificate of title; financing statement; in some instances, the original security agreement, or other document showing that a security interest has been filed or recorded.

April 22-23, 2014 Page 365 of 596

Information that is entitled to privacy: A *Proof of Claim* form and any attached documents must show only the last 4 digits of any social security number, an individual's tax identification number, or a financial account number, only the initials of a minor's name, and only the year of any person's date of birth. If a claim is based on delivering health care goods or services, limit the disclosure of the goods or services to avoid embarrassment or disclosure of confidential health care information. You may later be required to give more information if the trustee or someone else in interest objects to the claim.

Priority claim: A claim within a category of unsecured claims that is entitled to priority under 11 U.S.C. §507(a). These claims are paid from the available money or property in a bankruptcy case before other unsecured claims are paid. Common priority unsecured claims include alimony, child support, taxes, and certain unpaid wages.

Proof of claim: A form that shows the amount of debt the debtor owed to a creditor on the date of the bankruptcy filing. The form must be filed with the clerk of the same bankruptcy court in which the bankruptcy case was filed.

Redaction of information: Masking, editing out, or deleting certain information to protect privacy. Filers must redact or leave out information entitled to privacy on the *Proof of Claim* form and any attached documents.

Secured claim under 11 U.S.C. §506(a): A claim backed by a lien on particular property of the debtor. A claim is secured to the extent that a creditor has the right to be paid from the property before other creditors are paid. The amount of a secured claim usually cannot be more than the value of the particular property on which the creditor has a lien. Any amount owed to a creditor that is more than the value of the property normally may be an unsecured claim. But exceptions exist; for example, see 11 U.S.C. § 1322(b) and the final sentence of 1325(a).

Examples of liens on property include a mortgage on real estate or a security interest in a car. A lien may be voluntarily

granted by a debtor or may be obtained through a court proceeding. In some states, a court judgment may be a lien.

Setoff: Occurs when a creditor pays itself with money belonging to the debtor that it is holding, or by canceling a debt it owes to the debtor.

Uniform claim identifier: An optional 24-character identifier that certain large creditors use to facilitate electronic payment in chapter 13 cases.

Unsecured claim: A claim that does not meet the requirements of a secured claim. A claim may be unsecured in part if the amount of the claim is more than the value of the property on which a creditor has a lien.

Offers to purchase a claim

Certain entities purchase claims for an amount that is less than the face value of the claims. These entities may contact creditors offering to purchase their claims. Some written communications from these entities may easily be confused with official court documentation or communications from the debtor. These entities do not represent the bankruptcy court, the bankruptcy trustee, or the debtor. A creditor has no obligation to sell its claim. However, if a creditor decides to sell its claim, any transfer of that claim is subject to Bankruptcy Rule 3001(e), any provisions of the Bankruptcy Code (11 U.S.C. § 101 et seq.) that apply, and any orders of the bankruptcy court that apply.

Do not file these instructions with your form.

April 22-23, 2014 Page 366 of 596

Fill in this information to identify the case:	
Debtor 1	
Debtor 2 (Spouse, if filing)	
United States Bankruptcy Court for the:	District of(State)
Case number	. ,

Draft Jan. 17, 2014

Official Form 410

Proof of Claim

Read the instructions before filling out this form. Use this form to make a claim for payment in a bankruptcy case. Do not use this form to make a request for payment of an administrative expense. Make such a request according to 11 U.S.C. § 503.

The law requires that filers **must leave out or redact** information that is entitled to privacy on this form or on any attached documents. Attach redacted copies of any documents that support the claim, such as promissory notes, purchase orders, invoices, itemized statements of running accounts, contracts, judgments, mortgages, and security agreements. **Do not send original documents**; they may be destroyed after scanning. If the documents are not available, explain in an attachment.

A person who files a fraudulent claim could be fined up to \$500,000, imprisoned for up to 5 years, or both. 18 U.S.C. §§ 152, 157, and 3571.

Fill in all the information about the claim as of the date the case was filed. That date is on the notice of bankruptcy (Form 309) that you received.

Part 1: **Identify the Claim** 1. Who is the current creditor? Name of the current creditor (the person or entity to be paid for this claim) Other names the creditor used with the debtor 2. Has this claim been ☐ No acquired from ☐ Yes. From whom? _ someone else? Where should notices to the creditor be sent? Where should notices Where should payments to the creditor be sent? (if and payments to the different) creditor be sent? Federal Rule of Name Name Bankruptcy Procedure (FRBP) 2002(g) Number Street Number Street City State ZIP Code ZIP Code City Contact phone Contact phone Contact email Contact email Uniform claim identifier for electronic payments in chapter 13 (if you use one): Does this claim amend ☐ No one already filed? Yes. Claim number on court claims registry (if known) _____ Filed on / YYYY MM / DD 5. Do vou know if anyone ☐ No else has filed a proof ☐ Yes. Who made the earlier filing? of claim for this claim?

Dο	htor	1

First Name	Middle Name	Last Name

Case number		

Ľ	Give information	n About the Claim as of the Date the Case was Filed
6.	Do you have any number you use to identify the debtor?	No Yes. Last 4 digits of the debtor's account or any number you use to identify the debtor:
7.	How much is the claim?	\$ For leases state only the amount of default.
		Does this amount include interest or other charges? ☐ No
		Yes. Attach statement itemizing interest, fees, expenses, or other charges required by Bankruptcy Rule 3001(c)(2)(A).
8.	What is the basis of the	Examples: Goods sold, money loaned, lease, services performed, personal injury or wrongful death, or credit card.
	claim?	Attach any documents supporting the claim required by Bankruptcy Rule 3001(c).
		Limit disclosing information that is entitled to privacy, such as healthcare information.
9.	Is all or part of the claim secured?	□ No □ Yes. The claim is secured by a lien on property.
		Nature of property:
		Real estate. If the claim is secured by the debtor's principal residence, file a <i>Mortgage Proof of Claim Attachment</i> (Official Form 410-A) with this <i>Proof of Claim</i> .
		☐ Motor vehicle ☐ Other. Describe:
		Basis for perfection:
		Attach redacted copies of documents, if any, that show evidence of perfection of a security interest (for example, a mortgage, lien, certificate of title, financing statement, or other document that shows the lien has been filed or recorded.)
		Value of property: \$
		Amount of the claim that is secured: \$
		Amount of the claim that is unsecured: \$ (The sum of the secured and unsecured amounts should match the amount in line 7.)
		Amount necessary to cure any default as of the date of the petition: \$
		Annual Interest Rate (when case was filed)% □ Fixed □ Variable
10	. Is this claim based on a	□ No
	lease?	Yes. Amount necessary to cure any default as of the date of the petition.
11	. Does this claim involve	□ No
	a right to setoff?	☐ Yes. Explain:

Debtor 1				Case number				
First Name	Middle Name	Last Name						
12. Is all or part of the clain entitled to priority unde 11 U.S.C. § 507(a)?		all that apply:				Amount entitled to priority		
A claim may be partly priority and partly nonpriority. For example,	11 U.S.	cic support obliga C. § 507(a)(1)(A	ations (including alimony a) or (a)(1)(B).	nd child support) under		\$		
in some categories, the law limits the amount entitled to priority.	■ Up to \$1	2,775* of deposital, family, or hous	ts toward purchase, lease, sehold use. 11 U.S.C. § 50	or rental of property o 07(a)(7).	r services for	\$		
	bankrup		nmissions (up to \$12,475*) ed or the debtor's busines			\$		
	☐ Taxes o	or penalties owed	d to governmental units. 1	I U.S.C. § 507(a)(8).		\$		
	☐ Contrib	utions to an emp	loyee benefit plan. 11 U.S	.C. § 507(a)(5).		\$		
	Other. S	Specify subsection	on of 11 U.S.C. § 507(a)(_	_) that applies.		\$		
	* Amounts a	are subject to adjus	stment on 4/1/16 and every 3	ears after that for cases b	pegun on or after	the date of adjustment.		
						·		
Part 3: Sign Below								
The person completing this proof of claim must	Check the appro	ppriate box:						
sign and date it. FRBP 9011(b).	☐ I am the cre							
If you file this claim			or authorized agent.	nt Bankruntov Pula 30	104			
electronically, FRBP 5005(a)(2) authorizes courts to establish local rules	☐ Lama auus	_ · · · · · · · · · · · · · · · · · · ·						
specifying what a signature is.		I understand that an authorized signature on this <i>Proof of Claim</i> serves as an acknowledgment that when calculating the amount of the claim, the creditor gave the debtor credit for any payments received toward the debt.						
A person who files a fraudulent claim could be fined up to \$500,000, imprisoned for up to 5	I have examined the information in this <i>Proof of Claim</i> and have a reasonable belief that the information is true and correct.							
years, or both. 18 U.S.C. §§ 152, 157 and 3571.	I declare under penalty of perjury that the foregoing is true and correct.							
	Executed on date		YYYY					
	Signature				-			
	Print the name	of the person w	who is completing and si	gning this claim:				
	Name	First name	Middle na	me	Last name			
	Title							
	Company	Identify the corpora	ate servicer as the company if the	authorized agent is a servicer.				
	Address	Number	Street					
		City		State	ZIP Code			
	Contact phone			Email				

Mortgage Proof of Claim Attachment

(12/15)

If you file a claim secured by a security interest in the debtor's principal residence, you must use this form as an attachment to your proof of claim. See separate instructions.

Part	1: Mortgage	and Case	Informati	ion	Part 2: Total	l Debt Calcula	tion	Part 3: Arrearage as of Date of the Petition Part 4: Monthly				Monthly M	Mortgage Payment					
Case	e number:				Principal bala	ance:		Pri	Principal & interest due:					Principal & interest:				
Debt	or 1:				Interest due:			Pre	epetition fe	es due:		Monthly escrow:						
Debt	or 2:				Fees, costs of	due:			crow deficion vanced:	ency for fun	ds 		Private mortgage insurance:					
Last	4 digits to ide	ntify:		· 	Escrow defic			Pro	ojected esc	row shortag	je:		Total monthly payment:					
Cred	litor:				Less total fur	nds on hand: -		Le	ss funds or	hand:								
Serv	icer:				Total debt:			To	tal prepetiti	on arrearag	e:]					
	d accrual/daily le interest/oth					L							J					
Part 5	5 : Loan Payn	nent Histo	ory from F	rirst Date of I	Default													
		Account	Activity				How Fun	ds Were	Applied/Am	ount Incurr	ed	Balance	After Amo	unt Receiv	ed or Incu	rred		
Α.	В.	C.	D.	E.	F.	G.	H.	I.	J.	K.	L.	M.	N.	Ο.	P.	Q.		
Date	Contractual payment amount	Funds received	Amount incurred	Description	Contractual due date	Prin, int & esc past due balance		to	Amount to escrow	Amount to fees or charges	Unapplied funds		Accrued interest balance		Fees / Charges balance	Unapplied funds balance		

Mortgage Proof of Claim Attachment (12/15)

Page 1 of

If you file a claim secured by a security interest in individual debtor cases, you must use this form as an attachment to your proof of claim. See separate instructions

Draft 3.26.2014

PART I Mortgage and Case Information			PART II Total Debt Calculation				PART III Arrearage as of Date of the Petition					PART IV Monthly Mortgage Payment				
Case Number		13-1	2335	Principal Due	Principal & Int	erest Due:		\$!	968.30	Principal & Interest \$684.15						
Debtor 1		John	Smith	Interest Due		\$87	72.23	Escrow Deficie	ncy		\$9	978.95	Monthly Escro	w	\$120	.00
Debtor 2		Mary	Smith	Fees, Costs Due		\$70	08.95	Prepetition Fe	es Due:		\$	708.95	PMI		\$44.	55
Last 4 digits used to	Identify	xxxx	7790	Escrow Deficiency \$978.95				Projected Escr	ow Shortage		\$:	300.00				
Servicer:		Wells	Fargo	Less funds on hand		-\$7	4.31	Less funds on	hand		-5	574.31	Total Monthly	Payment	\$848	.70
Creditor:		2010 US E	Bank Trust	Total Debt		\$87,	148.33	Total Prepetiti	on Arrearage		\$2	,881.89				
Fixed accrual/Daily S	Simple Interest	Fix	ed													
					PART V	Loan Pavme	nt History fron	n First Date of I	Default							
		Account	Activity				<u>, , , , , , , , , , , , , , , , , , , </u>		ere applied/Amo	ount Incurred		[Balance after amount received or incurred			
40.5	(B) Contractual			(T)	(F) Contractual	(G) Prin, Int							(R)			
(A) Date	Payment amount	(C) Funds received	(D) Amount Incurred	(E) Description	Due Date	& Esc Past Due Balance	(H) Amount to Principal	(I) Amount to Interest	(J) Amount to Escrow	(K) Amount to Fees or Charges	(L) Unapplied funds	(M) Principal Balance	(O) Accrued interest balance	(P) Escrow Balance	(Q) Fees / Charges Balance	Unapplied Funds Balance
4/16/13			41.05	LATE FEE	Apr-13	804.15				41.05		85,405.01		0.00	41.05	0.00
5/1/13	804.15			CONTRACTUAL PAYMENT DUE	Apr-13	1,608.30						85,405.01		0.00	41.05	0.00
5/20/13			41.05	LATE FEE		1,608.30				41.05		85,405.01		0.00	82.10	0.00
6/1/13	804.15			CONTRACTUAL PAYMENT DUE	Apr-13	2,412.45						85,405.01		0.00	82.10	0.00
6/7/13		1,200.00		PAYMENT RECEIVED	May-13	1,608.30	247.49	436.15	120.00	-82.10	478.46	85,157.52		120.00	-82.10	
6/20/13		1,200.00					247.49	436.13		-82.10	478.40					
7/1/13			300.00	PROPERTY TAXES		1,608.30			-600.00			85,157.52		-180.00	0.00	478.46
7/1/13	804.15	400.00		PAYMENT RECEIVED	Jun-13	2,012.45					400.00	85,157.52		-180.00	0.00	878.46
				PAYMENT FROM SUSPENSE	Juli-13	1,208.30	247.50	436.14	120.00		-804.15	84,910.02		-180.00	0.00	74.31
7/29/13			750.00	ATTORNEY FEES		1,208.30				750.00		84,910.02		-180.00	750.00	74.31
8/1/13	804.15	845.20		PAYMENT RECEIVED	Jul-13	1,208.30	247.51	436.13	120.00	-41.05		84,662.51		-138.95	708.95	74.31
8/16/13			840.00	INSURANCE PAID		1,208.30				-840.00		84,662.51		-978.95	708.95	74.31

April 22-23, 2014 Page 371 of 596

April 22-23, 2014 Page 372 of 596

Official Form 410A

Instructions for Mortgage Proof of Claim Attachment

United States Bankruptcy Court

12/15

Introduction

This form is used only in individual debtor cases. When required to be filed, it must be attached to *Proof of Claim* (Official Form B410) with other documentation required under the Federal Rules of Bankruptcy Procedure.

Applicable Law and Rules

Rule 3001(c)(2)(A) of the Federal Rules of Civil Procedure requires for the bankruptcy case of an individual that any proof of claim be accompanied by a statement itemizing any interest, fees, expenses, and charges that are included in the claim.

Rule 3001(c)(2)(B) requires that a statement of the amount necessary to cure any default be filed with the claim if a security interest is claimed in the debtor's property.

If a security interest is claimed in property that is the debtor's principal residence, Rule 3001(c)(2)(C) requires this form to be filed with the proof of claim. The form implements the requirements of Rule 3001(c)(2)(A) and (B).

If an escrow account has been established in connection with the claim, Rule 3001(c)(2)(C) also requires an escrow statement to be filed with the proof of claim. The statement must be prepared as of the date of the petition and in a form consistent with applicable nonbankruptcy law.

Directions

Definition

This form must list all transactions on the claim from the *first date of default* to the petition date. The *first date of default* is the first date on which the borrower failed to make a payment in accordance with the terms of the note and mortgage, unless the note was subsequently brought current with no principal, interest, fees, escrow payments, or other charges immediately payable.

Information required in Part 1: Mortgage and Case Information

Insert on the appropriate lines:

- the case number;
- the names of Debtor 1 and Debtor 2;
- he last 4 digits used to identify the mortgage loan number (i.e., the last 4 digits of the loan account number or any other information to identify the account);
- the creditor's name;
- the servicer's name, if applicable; and
- he method used to calculate interest on the debt (i.e., fixed accrual, daily simple interest, or other method).

Official Form B410A

Mortgage Proof of Claim Attachment

page 1

Information required in Part 2: Total Debt Calculation

Insert:

- the principal balance on the debt;
- the interest due and owing;
- any fees or costs owed under the note or mortgage and outstanding as of the date of the bankruptcy filing; and
- any Escrow deficiency for funds advanced—
 that is, the amount of any prepetition
 payments for taxes and insurance that the
 servicer or mortgagee made out of its own
 funds and for which it has not been
 reimbursed.

Also disclose the *Total amount of funds on hand*. This amount is the total of the following, if applicable:

- a positive escrow balance,
- unapplied funds, and
- amounts held in suspense accounts.

Total the amounts owed—subtracting total funds on hand—to determine the total debt due.

Insert this amount under Total debt.

Information required in the Part 3: Arrearage as of the Date of Petition

Insert the amount of the principal and interest portion of all prepetition monthly installments that remain outstanding as of the petition date. The escrow portion of prepetition monthly installment payments should not be included in this figure.

Insert the amount of fees and costs outstanding as of the petition date. This amount should equal the *Fees/Charges balance* as shown in the last entry in Part 5, Column P.

Insert any *escrow deficiency for funds advanced*. This amount should be the same as the amount of *escrow deficiency* stated in Part 2.

Insert the *Projected escrow shortage* as of the date the bankruptcy petition was filed. The *projected escrow shortage* is the amount the claimant asserts should exist in the escrow account as of the petition date, less the amount actually held. The amount actually held should equal the amount of a positive escrow account balance as shown in the last entry in Part 5, Column O.

This calculation should result in the amount necessary to cure any prepetition default on the note or mortgage that arises from the failure of the borrower to satisfy the amounts required under the Real Estate Settlement Practices Act (RESPA). The amount necessary to cure should include 1/6 of the anticipated annual charges against the escrow account or 2 months of the monthly pro rata installments due by the borrower as calculated under RESPA guidelines. The amount of the projected escrow shortage should be consistent with the escrow account statement attached to the *Proof of Claim*, as required by Rule 3001(c)(2)(C).

Insert the amount of funds on hand that are unapplied or held in a suspense account as of the petition date.

Total the amounts due listed in Part 3, subtracting the funds on hand, and insert the calculated amount in *Total prepetition arrearage*.

Information required in Part 4: Monthly Mortgage Payment

Insert the principal and interest payment amount of the monthly payment as of the petition date.

Insert the monthly escrow portion of the monthly payment. This amount should take into account the receipt of any amounts claimed in Part 3 as escrow deficiency and projected escrow shortage. Therefore, a claimant should assume that the

Official Form B410A

escrow deficiency and shortage will be paid through a plan of reorganization and provide for a credit of a like amount when calculating postpetition escrow installment payments.

Claimants should also add any monthly private mortgage insurance amount.

Insert the sum of these amounts in *Total monthly payment*.

Information required in Part 5: Loan Payment History from the First Date of Default

Beginning with the First Date of Default, enter:

- the date of the default in Column A;
- amount incurred in Column D;
- description of the charge in Column E;
- principal balance, escrow balance, and unapplied or suspense funds balance as of that date in Columns M, O, and Q, respectively.

For (1) all subsequently accruing installment payments; (2) any subsequent payment received; (3) any fee, charge, or amount incurred; and (4) any escrow charge satisfied since the date of first default, enter the information in date order, showing:

- the amount paid, accrued, or incurred;
- a description of the transaction;
- the contractual due date, if applicable;
- how the amount was applied or assessed;and
- the resulting principal balance, accrued interest balance, escrow balance, outstanding fees or charges balance, and the total unapplied funds held or in suspense.

If more space is needed, fill out and attach as many copies of *Mortgage Proof of Claim Attachment: Additional Page* as necessary.

Official Form B410A

April 22-23, 2014 Page 376 of 596

Fill in this information to identify the case:	Draft March 31, 20
Debtor 1	
Debtor 2 (Spouse, if filing)	
United States Bankruptcy Court for the:	_ District of
Case number	(State)
Official Form 410S1	
Notice of Mortgage Payme	ent Change 12/15
	r's principal residence provided for under the debtor's plan pursuant to anges in the installment payment amount. File this form as a supplement ent amount is due. See Bankruptcy Rule 3002.1.
Name of creditor:	Court claim no. (if known):
Last 4 digits of any number you use to identify the debtor's account:	Date of payment change: Must be at least 21 days after date of this notice
	New total payment: Principal, interest, and escrow, if any
Part 1: Escrow Account Payment Adjustment	
	epared in a form consistent with applicable nonbankruptcy law. Describe ched, explain why:
Current escrow payment: \$	New escrow payment: \$
Part 2: Mortgage Payment Adjustment	
	hange based on an adjustment to the interest rate in the debtor's
variable-rate note?	
lacksquare Yes. Attach a copy of the rate change notice prepared in	n a form consistent with applicable nonbankruptcy law. If a notice is not
Current interest rate:%	New interest rate:%
Current principal and interest payment: \$	New principal and interest payment: \$
Part 3: Other Payment Change	
3. Will there be a change in the debtor's mortgage p	ayment for a reason not listed above?
□ No	sis for the change, such as a repayment plan or loan modification agreement.
Reason for change:	

Official Form 410S1

Current mortgage payment: \$ ___

Notice of Mortgage Payment Change

New mortgage payment: \$_____

page 1

Debtor 1				Case number (if known)
	First Name	Middle Name	Last Name	
Part 4:	Sign Here	1		
-	on completi e number.	ng this Notice n	nust sign it. Sign ar	nd print your name and your title, if any, and state your address and
Check the	appropriate	box.		
□la	m the credit	or.		
□la	m the credit	or's authorized a	gent. (Attach copy o	of power of attorney, if any.)

I declare under penalty of perjury that the information provided in this claim is true and correct to the best of my knowledge, information, and reasonable belief.

x					Date _	
Signature						
Print:	First Name	Middle Name	Last Name		Title _	
Company						
Address	Number	Street				
	City		State	ZIP Code		
Contact phone	()				Email _	

April 22-23, 2014 Page 378 of 596

Fill in this information to identify the case:	Draft N	Jarch 24 24
Debtor 1		/larch 31, 2
Debtor 2 (Spouse, if filing)		
United States Bankruptcy Court for the:	District of	
Case number	(State)	
Official Form 44000		
Official Form 410S2		
lotice of Postpetition Mort	gage Fees, Expenses, and Cha	rges 12/15
	e debtor's principal residence, you must use this form to give ert are recoverable against the debtor or against the debtor's of claim. See Bankruptcy Rule 3002. 1.	
Name of creditor:	Court claim no. (if known):	
Last 4 digits of any number you use to identify the debtor's account:		
•		
Doos this notice supplement a prior notice of p	astractition face	
	ostpetition fees,	
expenses, and charges?	ostpetition fees,	
expenses, and charges?	ostpetition fees,	
expenses, and charges? □ No	ostpetition fees,	
expenses, and charges? No Yes. Date of the last notice://		
expenses, and charges? No Yes. Date of the last notice://		
expenses, and charges? No Yes. Date of the last notice://	, and Charges the debtor's mortgage account after the petition was filed. Do	
Part 1: Itemize Postpetition Fees, Expenses, temize the fees, expenses, and charges incurred on the scrow account disbursements or any amounts previous prev	and Charges the debtor's mortgage account after the petition was filed. Do iously itemized in a notice filed in this case or ruled on by the	bankruptcy court.
Part 1: Itemize Postpetition Fees, Expenses, ltemize the fees, expenses, and charges incurred on the scrow account disbursements or any amounts previous Description	and Charges the debtor's mortgage account after the petition was filed. Do iously itemized in a notice filed in this case or ruled on by the Dates incurred	bankruptcy court. Amount
Part 1: Itemize Postpetition Fees, Expenses, temize the fees, expenses, and charges incurred on the escrow account disbursements or any amounts previous Description 1. Late charges	and Charges the debtor's mortgage account after the petition was filed. Do iously itemized in a notice filed in this case or ruled on by the Dates incurred (1)	Amount \$
Part 1: Itemize Postpetition Fees, Expenses, litemize the fees, expenses, and charges incurred on the lescrow account disbursements or any amounts previous Description 1. Late charges 2. Non-sufficient funds (NSF) fees	and Charges the debtor's mortgage account after the petition was filed. Do iously itemized in a notice filed in this case or ruled on by the Dates incurred (1) (2)	Amount \$ \$
expenses, and charges? No Yes. Date of the last notice:/	the debtor's mortgage account after the petition was filed. Do iously itemized in a notice filed in this case or ruled on by the Dates incurred (1) (2) (3)	Amount \$ \$ \$
Part 1: Itemize Postpetition Fees, Expenses, Itemize the fees, expenses, and charges incurred on tescrow account disbursements or any amounts previous Description 1. Late charges 2. Non-sufficient funds (NSF) fees 3. Attorney fees 4. Filing fees and court costs	and Charges the debtor's mortgage account after the petition was filed. Do iously itemized in a notice filed in this case or ruled on by the Dates incurred (1) (2) (3) (4)	Amount \$ \$ \$ \$
expenses, and charges? No Yes. Date of the last notice:// Part 1: Itemize Postpetition Fees, Expenses, ltemize the fees, expenses, and charges incurred on tescrow account disbursements or any amounts previous Description 1. Late charges 2. Non-sufficient funds (NSF) fees 3. Attorney fees 4. Filling fees and court costs 5. Bankruptcy/Proof of claim fees	the debtor's mortgage account after the petition was filed. Do iously itemized in a notice filed in this case or ruled on by the Dates incurred (1)	SSSS
Part 1: Itemize Postpetition Fees, Expenses, Itemize the fees, expenses, and charges incurred on tescrow account disbursements or any amounts previous Description 1. Late charges 2. Non-sufficient funds (NSF) fees 3. Attorney fees 4. Filing fees and court costs 5. Bankruptcy/Proof of claim fees 6. Appraisal/Broker's price opinion fees	the debtor's mortgage account after the petition was filed. Do iously itemized in a notice filed in this case or ruled on by the Dates incurred	sssss
expenses, and charges? No Yes. Date of the last notice:/	the debtor's mortgage account after the petition was filed. Do iously itemized in a notice filed in this case or ruled on by the Dates incurred (1)	\$ \$ \$ \$ \$ \$ \$ \$

The debtor or trustee may challenge whether the fees, expenses, and charges you listed are required to be paid. See 11 U.S.C. § 1322(b)(5) and Bankruptcy Rule 3002. 1.

Official Form 410S2 Notice of Postpetition Mortgage Fees, Expenses, and Charges page 1

(13) \$ _____

_____ (14) \$ _____

_____ (11) \$_____

______ (12) \$_____

11. Other. Specify:___

12. Other. Specify:___

13. Other. Specify:_

14. Other. Specify:__

Debtor 1			 Case number (if known)
	E1 . 11	 	7

Part 2: Sign Here

The person telephone r		our name and your title, if any, and state your address and	
Check the a	ppropriate box.		
☐ I am the	creditor.		
☐ I am the	creditor's authorized agent. (Attach copy of power of at	torney, if any.)	
I declare under penalty of perjury that the information provided in this claim is true and correct to the best of my knowledge, information, and reasonable belief.			
Signature		Date/	
Print:	First Name Middle Name Last Name	Title	
Company			
Address	Number Street		
	City State ZII	Code	
Contact phone	()	Email	

Official Form 410S2 Notice of Postpetition Mortgage I

Notice of Postpetition Mortgage Fees, Expenses, and Charges

Page 380 of 596

COMMITTEE NOTE

Official Form 410, *Proof of Claim*, applies in all cases. Form 410 replaces Official Form 10, Proof of Claim. It is renumbered to distinguish it from the forms used by debtors for case opening, and includes stylistic changes throughout the form. It is revised as part of the Forms Modernization Project, making it easier to read and, as a result, likely to generate more complete and accurate responses. Because the goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats, many of the open-ended questions and multiple-part instructions have been replaced with more specific questions.

Official Form 410 has been substantially reorganized. A new question has been added at line 10 that solicits information about claims based on leases.

Official Form 410A, *Mortgage Proof of Claim Attachment*, is revised in its content and format. Rather than requiring a home mortgage claimant to fill in blanks with itemized information about the principal, interest, and fees due as of the petition date and the amount necessary to cure a prepetition default, the form now requires the claimant to provide a loan history that reveals when payments were received, how they were applied, when fees and charges were incurred, and when escrow charges were satisfied. Because completion of the revised form can be automated, it will permit claimants to comply with Rule 3001(c)(2)(C) with efficiency and accuracy. Attachment of a loan history with a home mortgage proof of claim will also provide transparency about the basis for the claimant's calculation of the claim and arrearage amount.

The loan history should begin with the first date on which the borrower failed to make a payment in accordance with the terms of the note and mortgage, unless the note was subsequently

April 22-23, 2014 Page 381 of 596

brought current with no principal, interest, fees, escrow payments, or other charges immediately payable.

Official Forms 410S1 and 410S2, *Notice of Mortgage*Payment Change and Notice of Postpetition Mortgage Fees,

Expenses, and Charges, are revised as part of the Forms

Modernization Project. There are formatting changes throughout the forms.

April 22-23, 2014 Page 382 of 596

B416A (Official Form 416A) (12/15) Form 416A. CAPTION (FULL)

Debtor [1]	[Debtor 2]
Other names used by the debtor within the last 8 years	[Other names used by the debtor within the last 8 years]
Address	[Address, if Debtor 2 lives at a different address
Last 4 digits of Social Security number or ITIN	Last 4 digits of Social Security number or ITIN
[Employer Identification Number]	[Employer Identification Number]
United States Bankruptcy Court for the	
[Bankruptcy	district]
Case number	Chapter you are filing under

[Designation of Character of Paper]

April 22-23, 2014 Page 383 of 596

April 22-23, 2014 Page 384 of 596

COMMITTEE NOTE

Official Form 416A, *Caption*, applies on all forms where prescribed. Form 416A replaces Official Form 16A, *Caption*. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

April 22-23, 2014 Page 385 of 596

April 22-23, 2014 Page 386 of 596

B416B (Official Form 416B) (12/15) Form 416B. (Short Title) (May be used if 11 U.S.C. § 342(c) does not apply)

Debtor	
United States Bankruptcy Court for the	
	[Bankruptcy district]
Case number:	Chapter you are filing under:
[Designation of Character of Paper]	

April 22-23, 2014 Page 387 of 596

April 22-23, 2014 Page 388 of 596

COMMITTEE NOTE

Official Form 416B, *Caption*, applies on all forms where prescribed. Form 416B replaces Official Form 16B, *Caption*. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

April 22-23, 2014 Page 389 of 596

April 22-23, 2014 Page 390 of 596

B416D (Official Form 416D) (12/15) Form 416D. Caption to Use in Adversary Proceeding

Debtor 1	Plaintiff	
	Defendant	
United States Bankruptcy Court for the		
	[Bankruptcy district]	
Case number:	Chapter you are filing under:	Adv. Proc. No

Complaint [or other designation]

[If in a *Notice of Appeal* (see Form 417A) or other notice filed and served by a debtor, this caption must be altered to include the the debtor's address and Employer's Tax Identification Numbers (EIN), last four digits of Social Security numbers, or Individual Taxpayer Identification number (ITIN) as in Form 416A.]

April 22-23, 2014 Page 391 of 596

April 22-23, 2014 Page 392 of 596

COMMITTEE NOTE

Official Form 416D, Caption for Use in Adversary Proceeding Other Than for a Complaint Filed by a Debtor, applies on all forms where prescribed. Form 416D replaces Official Form 16D, Caption for Use in Adversary Proceeding Other Than for a Complaint Filed by a Debtor. It is renumbered as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

April 22-23, 2014 Page 393 of 596

April 22-23, 2014 Page 394 of 596

Draft March 24, 2014

Official Form 424 (12/15)

[Caption as described in Fed. R. Bankr. P. 7010 or 9004(b), as applicable]

Certification to Court of Appeals by All Parties

A notice of appeal having been filed in the above-styled matter on[Date],,
, and, and all the
appellees, if any], who are all the appellants [and all the appellees] hereby certify to the court under 28 U.S.C. § 158(d)(2)(A) that a circumstance specified in 28 U.S.C. § 158(d)(2) exists as stated below.
Leave to appeal in this matter:
is required under 28 U.S.C. § 158(a)
is not required under 28 U.S.C. § 158(a).
[If from a final judgment, order, or decree] This certification arises in an appeal from a final judgment, order, or decree of the United States Bankruptcy Court for the District of entered on[Date].
[If from an interlocutory order or decree] This certification arises in an appeal from an interlocutory order or decree, and the parties hereby request leave to appeal as required by 28 U.S.C. § 158(a).
[The certification shall contain one or more of the following statements, as is appropriate to the circumstances.]
The judgment, order, or decree involves a question of law as to which there is no controlling decision of the court of appeals for this circuit or of the Supreme Court of the United States, or involves a matter of public importance.
Or
The judgment, order, or decree involves a question of law requiring resolution of conflicting decisions.
Or
An immediate appeal from the judgment, order, or decree may materially advance the progress of the case or proceeding in which the appeal is taken.

April 22-23, 2014 Page 395 of 596

Official Form 424 continued (12/15)

[The parties may include or attach the information specified in Rule 8001.]

Signed: [If there are more than two signatories, all must sign and provide the information requested below. Attach additional signed sheets if needed.]

Attorney for Appellant (or Appellant, if not represented by an attorney):		
Printed name of signer:		
Address:		
Telephone number:	()	(
Date:	// 	//

April 22-23, 2014 Page 396 of 596

COMMITTEE NOTE

Official Form 424, *Certification to Court of Appeal by All Parties* replaces Official Form 424, *Certification to Court of Appeal by All Parties*. It is revised as part of the Forms Modernization Project, and includes stylistic changes throughout the form.

April 22-23, 2014 Page 397 of 596

April 22-23, 2014 Page 398 of 596

Instructions

For Bankruptcy Forms for Non-Individuals

U.S. Bankruptcy Court

December 2015

April 22-23, 2014 Page 399 of 596

General Instructions2
Overview of the bankruptcy forms and filing bankruptcy
Follow these privacy restrictions
Understand the terms used in the forms
Things to remember when filling out and filing these forms
Filing amended forms
On what date was a debt incurred?3
About the Process for Filing a Bankruptcy Case for Non-Individuals
Instructions for Selected Forms6
Schedule A/B: Real and Personal Property (Official Form 206A/B)
Schedule A/B: Real and Personal Property (Official Form 206A/B)
Schedule A/B: Real and Personal Property (Official Form 206A/B)
Schedule A/B: Real and Personal Property (Official Form 206A/B)

April 22-23, 2014 Page 400 of 596

General Instructions

This document provides instructions for completing selected forms that entities other than individuals and municipalities filing for bankruptcy must submit to the U.S. Bankruptcy Court. All of the required forms can be downloaded without charge from:

http://www.uscourts.gov/FormsAndFees/Forms/BankruptcyForms.aspx.

The instructions are designed to accompany the forms and are intended to help in understanding what information is required to properly file. The representatives of the debtor working on the forms should review each form and any pertinent instructions before supplying the information for each form.

Although the forms often parallel how businesses commonly keep their financial records, it is not always possible to do so because information needed in a bankruptcy case is often different from that prescribed under generally accepted accounting principles. These instructions highlight some of the differences between the bankruptcy documents and accounting records. The debtor must provide all information required.

These instructions are not a substitute for legal advice about bankruptcy and the required forms. Completing the forms is only a part of the bankruptcy process.

Non-individual debtors must have an attorney to file for bankruptcy. Although the attorney may prepare the forms using information supplied by the debtor, representatives of the debtor must ensure that the forms are accurate and complete and must sign the forms under penalty of perjury.

Read This Important Warning

Non-individual debtors must be represented by an attorney.

Bankruptcy can have serious long-term financial and legal consequences, including loss of property. Only an attorney can give legal advice regarding the possible consequences of filing for bankruptcy and the various options that are available.

Entities may not file bankruptcy if they are not eligible to file or do not intend to file the documents necessary to complete the bankruptcy.

Bankruptcy fraud is a serious crime. Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$500,000 or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

General Instructions page 2

April 22-23, 2014 Page 401 of 596

Overview of the bankruptcy forms and filing bankruptcy

Use the forms in the 200 series if the debtor is a non-individual, such as a corporation, partnership, or limited liability company (LLC). Forms in the 100 series are used by individuals or married couples. Sole proprietors must use the forms in the 100 series.

When a bankruptcy petition is filed, the U.S. Bankruptcy Court opens a case. It is important that the answers to the questions on the forms be complete and accurate so that the case proceeds smoothly. A person who gives false information in connection with a bankruptcy case could be charged with a federal crime, and the debtor may lose the benefits of filing for bankruptcy.

Filing a bankruptcy case is not private. Anyone has a right to see a debtor's bankruptcy forms after the debtor files them. In some circumstances, the bankruptcy court may issue a protective order to keep trade secrets or other confidential proprietary information from being disclosed to the public. 11 U.S.C. § 107 and Bankruptcy Rule 9037.

Follow these privacy restrictions

- Do not list a minor child's full name on any form. Instead, fill in only the child's initials and the full name and address of the child's parent or guardian. For example, write A.B., a minor child (*John Doe, parent, 123 Main St., City, State*). 11 U.S.C. § 112; Bankruptcy Rules 1007(m) and 9037.
- Do not list a person's date of birth.
- Do not list anyone's full Social Security number on any form.

Understand the terms used in the forms

To understand terms used in the forms and the instructions, see the *Glossary* at the end of this document.

Things to remember when filling out and filing these forms

- Be as complete and accurate as possible.
- If more space is needed, attach a separate sheet to the form. On the top of any pages added, write the debtor's name and case number, if known. Also identify the form and line number to which the additional information applies.
- Do not file these instructions with the bankruptcy forms that the debtor files with the court.
- For the debtor's records, be sure to keep a copy of the debtor's bankruptcy documents and all attachments that the debtor files.

Filing amended forms

Check the box on the top of the form to show that the debtor is submitting an amendment.

On what date was a debt incurred?

When a debt was incurred on a single date, fill in the actual date that the debt was incurred.

When a debt was incurred on multiple dates, fill in the range of dates. For example, if the debt is from a credit card, fill in the month and year of the first and last transactions, if known.

General Instructions page 3

April 22-23, 2014 Page 402 of 596

About the Process for Filing a Bankruptcy Case for Non-Individuals

To file for bankruptcy, the debtor must give the court several forms and documents. Some must be filed at the time the debtor files the case. Others may be filed up to 14 days later.

When the debtor files its bankruptcy case

The debtor must pay the entire filing fee when the case is filed. The debtor must file the forms listed below on the date the debtor files its bankruptcy case. For copies of the forms listed here, go to http://www.uscourts.gov/FormsAndFees/Forms/BankruptcyForms.aspx.

- □ Voluntary Petition for Non-Individuals Filing for Bankruptcy (Official Form 201). This form opens the case. Directions for completing it are included in the form itself.
- □ A list of names and addresses of all of the debtor's creditors, formatted as a mailing list according to instructions from the bankruptcy court in which the debtor files. (The bankruptcy court may call this a creditor matrix or mailing matrix.)
- ☐ Chapter 11 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims Against Debtor and Are Not Insiders (Official

- Form 204). Fill out this form only if the debtor files under chapter 11.
- □ Attachment to Voluntary Petition for Non-Individuals Filing for Bankruptcy Under Chapter 11 (Official Form 201A). This form is filed only by non-individual debtors who file under chapter 11 and who are required to file periodic reports (for example, Forms 10K and 10Q) with the Securities and Exchange Commission pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934.

When the debtor files its bankruptcy case or within 14 days after filing

The debtor must file the forms listed below with its *Voluntary Petition for Non-Individuals Filing for Bankruptcy* (Official Form 201) or within 14 days, or such additional time as the court may order, after filing. If the debtor does not do so, the case may be dismissed. Although it is possible to open a case by submitting only the documents listed under *When the debtor files its bankruptcy case*, the debtor should file the entire set of forms at one time to help its case proceed smoothly.

The debtor must fill out all of the forms completely even though some forms may ask similar questions.

The list below identifies the documents that all non-individuals must file as well as those that are specific to each chapter. For copies of the official forms, go to http://www.uscourts.gov/FormsAndFees/Forms/BankruptcyForms.aspx.

About the Process for Filing a Bankruptcy Case for Non-Individuals

All non-individuals who file for bankruptcy must file these forms and the forms for the specific chapter:

- □ *Schedules of Assets and Liabilities* (Official Form 206) which includes these forms:
 - □ *Schedule A/B: Real and Personal Property* (Official Form 206A/B)
 - □ Schedule D: Creditors Who Have Claims Secured by Property (Official Form 206D)
 - □ Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F)
 - ☐ Schedule G: Executory Contracts and Unexpired Leases (Official Form 206G)
 - □ *Schedule H: Codebtors* (Official Form 206H)
 - ☐ Summary of Assets and Liabilities for Non-Individuals (Official Form 206Sum). This form gives an overview of the totals on the schedules.
- Declaration Under Penalty of Perjury for Non-Individual Debtors (Official Form 202— Declaration)
- □ Statement of Financial Affairs for Non-Individuals Filing for Bankruptcy (Official Form 207)
- □ Disclosure of Compensation to Debtor's Attorney Unless local rules provide otherwise, Director's Form 2030 may be used.
- □ Statement of current income and current expenditures Unless local rules provide otherwise, debtors may use Schedule I/J:

 Monthly Receipts and Disbursements of Non-Individual Debtors Where Current Accounting Statements Are Unavailable (Form 2060 I/J)

If a small business debtor files under chapter 11, the debtor must also file:

If the debtor files under chapter 11 and meets the criteria and debt limits outlined in 11 U.S.C. § 101(51D), the debtor qualifies as a small business debtor and must file with the petition its most recent

- □ balance sheet,
- □ statement of operations,
- □ cash-flow statement, and
- □ federal income tax return.

If the debtor does not have these documents, the debtor must file a statement made under penalty of perjury that the debtor has not prepared either a balance sheet, statement of operations, or cashflow statement or the debtor has not filed a federal tax return.

April 22-23, 2014 Page 404 of 596

Instructions for Selected Forms

Instructions for Selected Forms page 6

April 22-23, 2014 Page 405 of 596

Schedule A/B: Real and Personal Property

(Official Form 206A/B)

Schedule A/B: Assets – Real and Personal Property (Official Form 206A/B) requires debtors to list most of the property interests that are involved in a bankruptcy case. All debtors filing for bankruptcy must honestly list everything they own or in which they have a legal, equitable, or future interest. Legal, equitable, or future interest are broad terms and include all kinds of property interests in both tangible and intangible property, whether or not anyone else has an interest in that property.

The information in this form is grouped by asset category and, in general, follows the layout and order of liquidity found in a balance sheet. Examples are included for some items and are meant to give debtors an idea of what to include in the categories. The examples are not intended to be complete lists of everything within that category.

An authorized representative of the debtor must verify under penalty of perjury that the information provided is true and correct. Bankruptcy Rule 1008.

If the debtor makes a false statement or conceals property, the debtor may lose the property, be fined up to \$500,000, or be imprisoned for up to 20 years or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Understand the terms used in this form

Current value

In this form, the debtor must report the *current* value of the debtor's interest in any property that it owns in each category. Current value is sometimes called fair market value and, for this form, it is the fair market value as of the date of filing the bankruptcy petition. Current value is how much the property is currently worth, which may be more or less than the amount the debtor paid for the property or the book value of the property.

Valuation method used for current value

In certain asset categories, the debtor must also provide the valuation method used to calculate the current value. Select a reasonable method that provides an accurate estimation of current value.

Examples of valuation methods may include:

- Appraisal (provide the date the appraisal was conducted);
- Comparable sales (for example, blue-book values or comparable sales provided by a broker);
- Revenue-based (for example, present value of revenue streams calculated for a hotel or apartment complex based on rents and available rooms);

Schedule A/B: Real and Personal Property (Official Form 206A/B)

- **Liquidation value** (for example, the price of the property when it is not allowed sufficient time to sell in the open market—this figure is typically provided by a professional);
- **Expert** (for example, an accountant or advisor who has special expertise with regard to the property);
- Replacement value (the cost of replacing the property);
- **Tax records** (for example, the value assessed on the property by the county appraisal);
- Recent cost-based valuations (for example, first-in first-out inventory valuation method).

Net book value of debtor's interest (where available)

If the debtor does not prepare a balance sheet for its financial records or for its tax returns, then it does not need to provide information in this column.

If the debtor prepares a balance sheet for its financial records or for its tax returns, then it must also provide the *net book value of debtor's interest* for certain types of property. For purposes of this form, use the book value reported on the most recent balance sheet prepared before filing this case.

Net book value is the carrying value of an asset on the debtor's books or financial records and is generally calculated by taking the original cost of the property and subtracting depreciation or amortization expenses (if any). Depreciation and amortization expenses are calculated using accounting procedures that allocate the cost of certain property over its useful life. It represents the decline in value over time due to wear and tear, obsolescence, or other factors.

How to list items on this form

- List items only once on this form; do not list an item in more than one category. If an item could fit into more than one category, select the category the debtor thinks is the most suitable and list the item there. For example, a car dealership may report vehicles under *Part 4: Inventory* instead of under *Part 8: Machinery, equipment, and vehicles*.
- List property held for resale in *Part 4: Inventory*. If the debtor separates manufactured items into raw materials, work in progress, and finished goods, report those items in the categories provided as appropriate. If the debtor only purchases items and holds them for resale and does not do any manufacturing, then report the items under finished goods, not as raw materials or work in progress.
- The values reported on this form must match the values reported on *Schedule D:*Creditors Who Have Claims Secured by Property (Official Form 206D).
- In Schedule A/B, list any executory contracts or unexpired lease contracts that have a net value (for example, an unexpired lease for a building, a real estate listing agreement, or leases for machinery or equipment). Also list them on Schedule G: Executory Contracts and Unexpired Leases (Official Form 206G).

Schedule A/B: Real and Personal Property (Official Form 206A/B)

Schedule D: Creditors Who Have Claims Secured by Property (Official Form 206D)

The people or organizations to whom the debtor owes money are called its *creditors*. A *claim* is a creditor's right to payment.

Creditors may have different types of claims:

- Secured claims. Report these on *Schedule D:* Creditors Who Have Claims Secured by Property (Official Form 206D).
- Unsecured claims. Report these on Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F).

Creditors with secured claims may be able to get paid from specific property in which that creditor has a security interest, such as a mortgage or a lien. That property is sometimes called *collateral* for the debt. Creditors with unsecured claims do not have rights against specific property.

Claims may be contingent, unliquidated, or disputed

Many claims have a specific amount which the debtor clearly owes. But some claims are uncertain or become due only after the bankruptcy petition is filed. All claims must be listed in the schedules, even if they are contingent, unliquidated, or disputed.

A claim is *contingent* if the debtor is not obligated to pay it unless a particular event occurs after the bankruptcy petition is filed.

A claim is *unliquidated* if the amount of the debt cannot be readily determined, such as by referring to an agreement or by a simple computation. An unliquidated claim is one for which there may be a definite liability but where the amount of the claim has not been determined.

A claim is *disputed* if the debtor disagrees that it owes all or a portion of the debt.

A single claim can have one, more than one, or none of these characteristics.

Do not omit any secured creditors

In alphabetical order, list all creditors that have judgment liens, garnishments, statutory liens, mortgages, deeds of trust, and purchase money security interests or other consensual liens against property of the debtor. These categories can be used to describe the lien.

The form is divided into parts. List a debt in Part 1 only once and list any other entities that should be notified about that debt in Part 2. For example, if an attorney is trying to collect a debt that the debtor owes to someone else, list the person to whom the debtor owes the debt in Part 1 and list the attorney in Part 2. If the case is a chapter 11 case and the amount of the creditor's unsecured claim in Column C makes it one of the 20 largest unsecured creditors, the creditor must also be included on *Chapter 11 or Chapter 9 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims Who Are Not Insiders* (Official Form B204).

Schedule D: Creditors Who Have Claims Secured by Property (Official Form 206D)

Determine the amount of each secured creditor's claim or claims

To determine the amount of a secured claim, compare the amount of the claim to the value of the debtor's interest in the property that is collateral for the claim. If that value is greater than the amount of the claim, then the entire amount of the claim is *secured*.

If the value of the property that is collateral for the claim is less than the amount of the claim, the difference is *unsecured*.

For example, if the outstanding balance due on an equipment loan is \$100,000 and the equipment is worth \$80,000, the lender has a secured claim of \$80,000 and an unsecured claim of \$20,000. In that situation, list the creditor only once on Schedule D: Creditors Who Have Claims Secured by Property (Official Form 206D). Do not list the creditor again on Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F). In addition, if the case is a chapter 11 case and the creditor's unsecured claim makes it one of the 20 largest unsecured creditors, the creditor must also be included on Chapter 11 or Chapter 9 Cases: List of Creditors Who Have the 20 Largest Unsecured Claims Who Are Not Insiders (Official Form 204).

List a creditor in Schedule D even if it appears that no value exists to support that creditor's secured claim, as long as the creditor has a security interest in some property owned by the debtor. If the claim is secured only by property owned by a non-debtor, list the claim in Schedule E/F.

If there is more than one secured claim against the same property, the amount of the claim that is entitled to be paid first must be subtracted from the property value to determine how much value remains for the next claim.

For example, if a building worth \$300,000 has a first mortgage of \$200,000 and a second mortgage of \$150,000, the first mortgage would be fully secured, and there would be \$100,000 of property value for the second mortgage, and the claim secured by the second mortgage would have an unsecured portion of \$50,000.

	\$300,000	value of a building
_	\$200,000	first mortgage
	\$100,000	remaining property value
	\$150,000	second mortgage
_	\$100,000	remaining property value
	\$ 50,000	unsecured portion of second mortgage claim

Show the amount of any unsecured portion of a secured claim on Schedule D in Column C.

Schedule D: Creditors Who Have Claims Secured by Property (Official Form 206D)

Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F)

The people or organizations to whom the debtor owes money are called its *creditors*. A claim is a creditor's right to payment.

Creditors may have different types of claims:

- Secured claims. Report these on Schedule D: Creditors Who Have Claims Secured by Property (Official Form 206D).
- Unsecured claims. Report these on Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F).

Creditors with unsecured claims typically do not have liens on or other security interests in the debtor's property.

Use Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F) to identify everyone who holds an unsecured claim against the debtor as of the date the bankruptcy petition is filed unless that creditor is already listed on Schedule D: Creditors Who Have Claims Secured by Property (Official Form 206D).

Creditors with secured claims have a right to take property from the debtor if the debtor does not pay them. They should be listed on *Schedule D: Creditors Who Have Claims*Secured by Property (Official Form 206D).

If a secured creditor's full claim exceeds the value of the property securing that claim, the creditor may have a secured claim for the value of the property and an unsecured claim for the deficiency. In that situation, list the creditor only once on *Schedule D: Creditors Who Have Claims Secured by Property* (Official Form 206D). Do not list the creditor again on

Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F).

List a creditor in Schedule D even if it appears that no value exists to support that creditor's secured claim.

Claims may be contingent, unliquidated, or disputed

Many claims have a specific amount which the debtor clearly owes. But some claims are uncertain or become due only after the date the bankruptcy petition is filed. All claims, whether they are certain or uncertain as of the date of the filing, must be listed in the schedules, even if the claims are contingent, unliquidated, or disputed.

A claim is *contingent* if the debtor is not obligated to pay it unless a particular event occurs after the petition is filed.

A claim is *unliquidated* if the amount of the debt cannot be readily determined, such as by referring to an agreement or by a simple computation. An unliquidated claim is one for which there may be a definite liability but where the amount of the claim has not been set.

A claim is *disputed* if the debtor disagrees that it owes all or a portion of the debt.

A single claim can have one, more than one, or none of these characteristics.

Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F)

Unsecured claims may be either priority or nonpriority claims

What are priority unsecured claims?

In bankruptcy cases, *priority unsecured claims* are those debts that the Bankruptcy Code requires to be paid before most other unsecured claims are paid. The most common priority unsecured claims are certain tax debts. Priority unsecured claims include those the debtor owes for:

- Taxes and certain other debts owed to the government—If the debtor owes certain federal, state, or local government taxes, customs duties, or penalties.

 11 U.S.C. § 507(a)(8).
- Wages, salaries, and commissions—If the debtor owes wages, salaries, and commissions, including vacation, severance, and sick leave pay and those amounts were earned within 180 days before the bankruptcy petition was filed or the debtor ceased business. In either instance, only the first \$12,475 per claim is a priority claim.* 11 U.S.C. § 507(a)(4).
- Contributions to employee benefit plans—If the debtor owes contributions to an employee benefit plan for services an employee rendered within 180 days before the bankruptcy petition was filed, or within 180 days before the debtor ceased business. Only the first \$12,475 per employee, less any amounts owed for wages, salaries, and commissions, is a priority claim.*

 11 U.S.C. § 507(a)(5).
- Certain claims of farmers and fishermen— Only the first \$6,150 per farmer or fisherman is a priority claim.* 11 U.S.C. § 507(a)(6).

■ Deposits by individuals — If the debtor obtained from an individual a deposit for the purchase, lease, or rental of property or services for the individual or the individual's family, the deposit may be a priority claim. Unredeemed gift certificates are deposits. The priority is limited to \$2,775.* 11 U.S.C. § 507(a)(7).

Other categories exist.

What are nonpriority unsecured claims?

Nonpriority unsecured claims are those debts that generally will be paid after priority unsecured claims are paid. The most common examples of nonpriority unsecured claims are trade debts, bank loans, contract obligations, and fees for professional services.

In Part 2, list every creditor owed money by the debtor not listed before, regardless of the amount and even if the debtor plans to pay a particular debt.

What if a claim has both priority and nonpriority amounts?

If a claim has both priority and nonpriority amounts, list that claim in Part 1 and show both priority and nonpriority amounts. Do not list it again in Part 2.

On what date was a debt incurred?

When a debt was incurred on a single date, fill in the actual date that the debt was incurred.

When a debt was incurred on multiple dates, fill in the range of dates. For example, if the debtor has a line of credit with multiple draws, fill in the month and year of the first and last transactions, if known.

Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 206E/F)

page **12**

April 22-23, 2014 Page 411 of 596

^{*} Subject to adjustment on 4/1/16, and every 3 years after that for cases begun on or after the date of adjustment.

Schedule G: Executory Contracts and Unexpired Leases (Official Form 206G)

Use Schedule G: Executory Contracts and Unexpired Leases (Official Form 206G) to identify the debtor's ongoing leases and certain contracts. List all of the debtor's executory contracts and unexpired leases.

Executory contracts are contracts between the debtor and another party in which neither party has performed all of the requirements by the time the debtor files for bankruptcy. *Unexpired leases* are leases that are still in effect.

The debtor must list all agreements that may be executory contracts or unexpired leases, even if they are listed on *Schedule A/B: Property* (Official Form 206A/B) or *Schedule E/F: Creditors Who Have Unsecured Claims*, (Official Form 206 E/F) including the following:

- Equipment leases;
- Vehicle leases;
- Leases for business or investment property (for example, office or warehouse space);
- Contracts to sell a building, land, or other real property;

- Service provider agreements (for example, maintenance contracts for office equipment, and contracts for cell phones, personal electronic devices, internet, and cable);
- Sales contracts;
- Supplier or service contracts;
- Leases or timeshare contracts;
- Employment contracts;
- Real estate listing agreements;
- Intellectual property license agreements (such as copyright, patent, trademark, and industrial rights);
- Development contracts; and
- Insurance contracts.

State the contract number of any government contract.

Glossary

Glossary page 14

April 22-23, 2014 Page 413 of 596

Definitions Used in the Forms for Non-Individuals Filing for Bankruptcy

Here are definitions for some of the important terms used in the forms for non-individuals who are filing for bankruptcy. See *Bankruptcy Basics* (http://www.uscourts.gov/FederalCourts) for more information about filing for bankruptcy and other important terms.

Affiliate — As used in the Bankruptcy Code and Rules, an affiliate of the debtor is:

- (a) an entity that directly or indirectly owns, controls, or holds with power to vote at least 20% of the outstanding voting securities of the debtor (excluding entities that hold such securities in a fiduciary or agency capacity without sole discretionary power to vote such securities or solely to secure a debt, if the entity has not in fact exercised such power to vote);
- (b) a corporation 20% or more of whose outstanding voting securities are directly or indirectly owned, controlled, or held with power to vote, by the debtor, or by an entity that directly or indirectly owns, controls, or holds with power to vote, 20% or more of the outstanding voting securities of the debtor (again excluding entities that hold such securities in a fiduciary or agency capacity without sole discretionary power to vote such securities or solely to secure a debt, if the entity has not in fact exercised such power to vote);
- (c) a person whose business is operated under a lease or operating agreement by a debtor, or person substantially all of whose property is operated under an operating agreement with the debtor; or
- (d) an entity that operates the business or substantially all of the property of the debtor under a lease or operating agreement.

Amortization — 1. A non-cash accounting method that allocates the cost of an intangible asset over its useful life. 2. Paying off a liability in regular installments over a period of time.

Amortization schedule — A report that contains a listing of intangible assets and the amount of amortization and accumulated amortization that has been allocated over the life of those assets. These reports are typically maintained for purposes of calculating tax deductions and preparing tax returns.

Annuity — A contract for the periodic payment of money, either for the life of the recipient or for a fixed number of years.

Book value or net book value — The carrying value of an asset on the debtor's books or financial records. This amount is generally calculated by taking the original cost of the property and subtracting depreciation or amortization expenses (if any).

Causes of action — Claims where the debtor is entitled to money or other relief from a third party or where a third party is entitled to money or other relief from the debtor.

Definitions Used in the Forms for Non-Individuals

Claim — A creditor's right to payment, even if contingent, disputed, unliquidated, or unmatured.

Codebtor —A person or entity that may also be responsible for paying a claim against the debtor.

Collateral — Property that secures a debt.

Contingent claim — Debt that is only payable if certain events occur.

Creditor matrix or mailing matrix — A list of names and addresses of all of the debtor's creditors, formatted as a mailing list according to instructions from the bankruptcy court in which the debtor files the case.

Creditor — The person or organization to whom the debtor owes money.

Current value or fair market value — how much the property is worth, which may be more or less than the purchase price or the book value. See the instructions for specific forms regarding whether the value requested is as of the date of the filing of the petition, the date the debtor completes the form, or some other date.

Debt — Liability on a claim.

Depreciation — A non-cash accounting method that allocates the cost of a tangible asset over its useful life.

Depreciation schedule — A report that contains a listing of tangible assets and the amount of depreciation and accumulated depreciation that has been allocated over the life of those assets. These reports are typically maintained for purposes of calculating tax deductions and preparing tax returns.

Discharge — A discharge in bankruptcy relieves a debtor from having to pay certain debts. For non-individuals, it applies only in certain chapter 11 and chapter 12 cases.

Disputed claim —A claim about which there is a disagreement. A claim is disputed if the debtor disagrees about either the validity or amount of the claim.

Doubtful or uncollectible accounts —

Receivables that the debtor has little or no expectation of collecting. This amount is deducted from total receivables to calculate the amount that the debtor reasonably expects will be collected on its receivables.

Executory contract — Contract between the debtor and another party as to which neither the debtor nor the other party has performed all of the requirements by the time the bankruptcy case is filed.

Goodwill — Amount of a purchase price that exceeds the net tangible assets. It can also be the value of an intangible asset that has a quantifiable value in business. Examples include a strong brand or reputation or, in an acquisition, goodwill.

Definitions Used in the Forms for Non-Individuals

Gross income — A company's gross revenue minus cost of goods sold.

Gross revenue — Amount generated by all of a company's operations before deductions for expenses.

Insider — Insiders include officers, directors, and anyone in control of a corporate debtor and their relatives; general partners of a partnership debtor and their relatives; affiliates of a debtor and insiders of such affiliates, and any managing agent of a debtor.

11 U.S.C. § 101.

Intangible assets — Types of property that are not physical in nature and cannot be touched, seen, or held. Examples include intellectual property and name recognition.

Intellectual property — An intangible asset that consists of human knowledge and ideas. Examples include patents, copyrights, trademarks, and software.

Legal or equitable interest — Any interest of the debtor in both tangible and intangible property, whether or not anyone other than the debtor also has an interest in that property.

Lien — A charge against or interest in property to secure a debt.

Nature of claim — The legal type of a claim, not the factual basis for it. Examples include breach of contract, personal injury, malpractice, and fraud.

Negotiable instrument — A written and signed unconditional promise or order to pay a specified sum of money on demand or at a definite time payable to order or bearer. Negotiable instruments include government bonds, corporate bonds, personal checks, cashiers' checks, promissory notes, and money orders.

Net operating loss (NOL) — Occurs when allowable tax deductions exceed taxable income, resulting in negative taxable income. NOLs can generally be used to recover past tax payments (*carry-back*) or reduce future tax payments (*carry-forward*).

Non-individual debtor — A non-individual entity such as a corporation, partnership, or limited liability company (LLC), on whose behalf or against whom a bankruptcy case is filed.

Non-negotiable instrument — Financial instrument of the debtor that cannot be transferred to another party by signing or delivering it.

Nonpriority unsecured claim — Debt that generally will be paid after priority unsecured claims are paid. Examples include amounts due for products purchased, professional services, and utilities.

Priority unsecured claim — Debt that the Bankruptcy Code requires to be paid before most other unsecured claims are paid. Examples include certain income tax debts and certain employee wage claims.

Definitions Used in the Forms for Non-Individuals

Secured claim — A claim that may be satisfied in whole or in part either

- through collateral,
- through a charge against or an interest in the debtor's property, or
- through a right of setoff.

Setoff — Occurs when a creditor pays itself with money belonging to the debtor that it is holding, or by canceling a debt it owes to the debtor. The Bankruptcy Code gives the trustee power to avoid some but not all setoffs that are made pre-petition.

Sole proprietorship — A business that a debtor owns as an individual, rather than a separate legal entity such as a corporation, partnership, or LLC. Sole proprietors must use the bankruptcy forms in the 100 series.

Tangible asset — Types of property that have physical form and can be seen, touched, or held. Examples include cash, machinery, buildings, and land.

Unexpired lease — Lease that is in effect at the time the bankruptcy petition is filed.

Unliquidated claim — A debt for which the amount cannot be readily determined, such as by referring to an agreement or by a simple computation. For instance, an unliquidated claim would arise from the debtor's sale of a defective product if the amount of damage it caused has not been determined.

Definitions Used in the Forms for Non-Individuals

April 22-23, 2014 Page 418 of 596

Appendix C

Comment Summary for Comments on Chapter 13 Plan

April 22-23, 2014 Page 419 of 596

April 22-23, 2014 Page 420 of 596

Comments on the Plan Form.	2
General Comments	2
Part 1: Notice to Interested Parties	15
Part 2: Plan Payments and Length of Plan	18
Section 2.1 (payments to the trustee)	18
Section 2.2 (manner of payments to the trustee from future earnings)	20
Section 2.3 (additional payments to the trustee)	22
Section 2.4 (estimated total amount of plan payments)	26
Section 2.5 (applicable commitment period)	28
Part 3: Treatment of Secured Claims	29
Part 3 (general)	29
Section 3.1 (maintenance and cure)	30
Section 3.2 (request for valuation of security and claim modification)	36
Section 3.3 (secured claims excluded from 11 U.S.C. § 506)	43
Section 3.4 (lien avoidance)	46
Section 3.5 (surrender of collateral)	49
Part 4: Treatment of Trustee's Fees and Administrative and Other Priority Claims	52
Part 4 (general)	52
Section 4.2 (trustee's fees)	53
Section 4.3 (attorney's fees)	55
Section 4.4 (other priority claims)	57
Section 4.5 (domestic support obligations assigned to a governmental unit)	58
Part 5: Treatment of Nonpriority Unsecured Claims	59
Section 5.1 (maintenance and cure)	59
Section 5.2 (separately classified nonpriority unsecured claims)	61
Section 5.3 (nonpriority unsecured claims)	62
Section 5.4 (interest)	65
Part 6: Executory Contracts and Unexpired Leases	65
Part 7: Order of Distribution of Trustee Payments	68
Part 8: Vesting of Property of the Estate	74

Part 9: Nonstandard Plan Provisions	75
Part 10: Signatures	76
Exhibits A and B	77
Comments on the Amended Rules	79
Rule 2002	79
Rule 3002	79
Rule 3007	85
Rule 3012	86
Rules 3015 and 4003	90
Rule 5009	96
Rule 7001	98
Rule 9009	98

Comments on the Plan Form

General Comments

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.): In several places, the plan states that if "none" is checked, the rest of a section need not be completed or reproduced. This is contrary to the standardization that is an advantage of a form plan. Lawyers will leave out paragraphs that should be included. Trustees and the courts should not be required to check each plan to make sure that something has not been eliminated or changed.

The plan does not provide for varying options for paying filing fees.

Comment—BK-2013-0001-0009 Alicia Villines: This looks clean and easily understandable.

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- A nationwide plan is recommended, because plans should be uniform for all debtors and creditors to use and understand. Plan forms have worked well in N.D. Tex. for many years.
- Please do not adopt these rule amendments as proposed and do not require a national plan form.

Comment BK-2013-0001-0014—K. Michael Fitzgerald (Chapter 13 Trustee, W.D. Wash.): The proposed plan will increase the cost for consumers under chapter 13. It is too complicated and unrealistic in its expectations. The average consumer debtor will not be able to

2

complete the form, and consumer bankruptcy attorneys will have more difficulty doing so properly. The costs of filing a chapter 13 case will quite likely double or even triple.

This is not a criticism of the hard work of the Committee. I understand the reasoning behind a national plan and appreciate the Committee's efforts. But the disadvantages far outweigh the benefits.

Comment BK-2013-0001-0015—Judge Jeff Bohm (Bankr. S.D. Tex.), on behalf of the United States Bankruptcy Court for the Southern District of Texas: The proposed Official Form 113 should not be adopted. Although greater uniformity might be helpful for attorneys with multi-jurisdiction practices, we are concerned that the plan form (i) inadequately addresses implementation of chapter 13 plans with conduit payments, (ii) fails to adhere to case law regarding the necessity to make adequate protection payments and to pay debtor's counsel, (iii) lacks a statutory liquidation analysis, thereby making the confirmation process more difficult, and (iv) does not include a summary cover page with an easy-to-read reconciliation of sources and uses of funds.

Comment BK-2013-0001-0027—Shmuel Klein (Attorney): The Chapter 13 plan must include language from §524(i) [regarding the willful failure of a creditor to credit payments received under a confirmed plan].

The plan must also allow separate classification of student loan debt which should not be included in the § 109 limits for unsecured debt.

The plan must state that creditors are receiving more under the plan than under a liquidation scenario, and it should not have a minimum percentage payment of claims.

The plan should provide that the clerk will send notice of the discharge to all three major credit reporting agencies by electronic transmission as is done for major creditors.

Comment BK-2013-0001-0034—Dana Milby (Attorney, Wichita, Kan.): I oppose the national plan form. It does not address specific issues in our jurisdiction, and I suspect that each jurisdiction has specific issues that cannot be addressed properly in a national form.

Comment BK-2013-0001-0036—Kurt Thornbladh (Attorney, Dearborn, Mich.): I oppose the adoption of a national plan form. The E.D. Mich. plan is superior and is better adapted to local conditions. This is true of other local plan forms. The rationale for a national plan form—that mortgage servicing companies cannot train staff to handle different plans in different parts of the country—overlooks the fact that there are 50 different state codes relating to residential mortgages. Requiring a high standard for debtors' attorneys while "dumbing down" the standard for the financial services industry does not fulfill the goal of equal justice under law.

If a national plan form is adopted, those districts that have adopted their own model plans should be allowed to keep them until further study.

Comment BK-2013-0001-0038—William Bonney (Chapter 13 Trustee, E.D. Okla.): I am generally in favor of a national plan form.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): I oppose adoption of the plan form.

April 22-23, 2014 Page 423 of 596

Comment BK-2013-0001-0041—Robert E. Hyman (Chapter 13 Trustee, E.D. Va.): The concept of a national plan form is fine, but some of the content of the form is problematic.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: Whether NACBA supports a national chapter 13 plan form depends on whether the form permits debtors to exercise their right to propose a plan under § 1321, which provides that the debtor shall file the plan, and § 1322(b)(11), which permits the debtor to include in the plan any appropriate provision not inconsistent with the Code.

If the plan form is adopted, the Committee Note should make clear that the terms of the form are not presumptively better or more valid than other plans that do not contravene the Code. A court should not reject non-standard terms on the ground that they are not necessary or that the terms of the official form are better (as a few courts with local plans have done).

Comment BK-2013-0001-0044—Judge Brian D. Lynch (Bankr. W.D. Wash.), on behalf of the bankruptcy judges of the district: We support the proposed rule amendments for the most part. But we strongly oppose adoption of a mandatory national plan form. There is substantial opposition to the proposed form among practitioners, trustees, and judges. There has never been widespread demand for such a form.

We have six general objections to the mandatory plan form:

First, the form increases complexity without a commensurate benefit. It is markedly more complex than any current local chapter 13 plan. There is no empirical support for the claim that a national form will be more efficient. There is a risk that the form will instead add to the administrative burden and expense of chapter 13 at a time when the process is already weighed down by the requirements of BAPCPA.

Second, a single form cannot deal comprehensively with the variety in legal culture and debtor demographics around the country. Some districts, such as the W.D. Tenn., have opted for a short form, while others, such as D. Or., have adopted longer forms. The proposed national form adopts the more comprehensive Oregon approach. Even districts with similar demographics have substantial differences in legal culture. Seattle and Portland have similar debtor profiles but different legal cultures (e.g., Seattle is a conduit district, while Portland is not; plan confirmation is typically not a contested matter in Seattle, while it is a contested matter requiring Rule 7004 service in Portland).

Third, a comprehensive plan form is more costly. Apart from the costs of training practitioners, trustees, and courts in the process, the more comprehensive approach will increase attorney fees. It funnels matters such as lien avoidance toward the confirmation process, which leads to more complicated confirmation proceedings and more attorney disbursements. The Portland division has been in the top five nationally for disbursements to debtor attorneys, with annual average disbursements to attorneys over the last ten years almost double those of the Seattle division.

Fourth, the proposed form will lead to more objections, thereby delaying confirmation and payments to creditors. The order of distribution is one of the most important aspects of a chapter 13 plan. But the form leaves Part 7 open for the debtor to fill out without guidance. This will lead to confusion, mistakes, and mischief, and increase the number of objections.

April 22-23, 2014 Page 424 of 596

Fifth, the proposed form increases the burden on courts and trustees to review whether there has been proper service under Rule 7004. To ensure that the plan provides proper service, debtors, trustees, and courts will need to review whether a plan is a contested matter and, if so, whether proper service has been effected.

Sixth, the proposed form is unlikely to promote uniformity. Courts disagree on issues of lien stripping, claim modifications, and lien avoidance. Even within districts there is disagreement. The alleged benefits of a mandatory plan form are more academic than real, because courts are unlikely to change their existing practices on these issues.

Comment BK-2013-0001-0045—Judge Lamar W. Davis, Jr. (Bankr. S.D. Ga.): The national plan form and related rules should not be adopted. The current system of chapter 13 practice, although variable from district to district, is not broken. The Advisory Committee has suggested six reasons for adopting a national form, but they are not persuasive.

First, the Advisory Committee suggests that the Bankruptcy Clause of the Constitution, Art. I, § 8, favors uniformity. Neither the text of the clause, nor any Supreme Court decision, suggests that it should be read as anything more than a grant of authority to Congress. It does not speak to the judiciary.

Second, there is a suggestion that a national form will lower costs. Perhaps it will lower costs for national or regional lenders and for software developers. But it will lead to higher costs for trustees, consumer debtor attorneys, and courts.

Third, the Advisory Committee suggests that the national form will lead to more effective education on chapter 13 issues. That benefit is outweighed by the costs of upending well-settled practices in each district and reeducating attorneys, trustees, judges, and their staffs.

Fourth, it has been suggested that a national form will lead to more effective appeals. It is more likely that the disruption of local practices will lead to more contested matters, more litigation, and more appeals for years to come. But more appeals does not mean "more effective" appeals.

Fifth, the Advisory Committee believes its proposal will improve procedures in chapter 13 cases. This claim is too vague to support the case for a national plan form.

Sixth, the Advisory Committee has justified its proposal as a response to *Espinosa*. This is a real concern for judges, who take seriously the Court's comments in that case regarding the role of bankruptcy judges. But that language in *Espinosa* was not part of the holding. The decision does not mandate a national plan form.

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.): A national plan form might improve chapter 13 practice, but this form is far from ideal and unacceptable. It is my understanding that this form comes from the local practices of the judges on the Rules Committee. They have missed the mark in putting together a reasonable national plan form.

Comment BK-2013-0001-0048—Mike Walters: This is definitely a step in the right direction.

Comment BK-2013-0001-0050—Hamilton J. Chauvin, Jr. (Attorney, La.): A clear and uniform national plan form is worthwhile. This is even more compelling after *Espinosa*. But I fear that if this form is adopted without serious alterations, numerous courts will opt to

April 22-23, 2014 Page 425 of 596

observe uniformity in the breach. The provisions of §§ 1322 and 1325 regarding plan confirmation include mandatory and permissible plan provisions. A national form should be limited to mandatory plan provisions.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: The bankruptcy judges of the D. Kan. oppose the national plan form. Our plan is better suited to our local conditions and precedent, as shown by the high level of chapter 13 cases that close with a discharge in our district (in fiscal year 2013, 71.3% as opposed to a national average of 44.6%). Our biggest concern is that proposed Official Form 113 will eliminate or greatly jeopardize our well-running mortgage conduit program. For example, § 3.1 allows debtors to choose who disburses post-petition mortgage payments, which is inconsistent with our local standing order.

Form 113 apparently is being adopted to make practice easier for national creditors, including state attorneys general, for software programmers, and to collect data. Thus, allowing judicial discretion will undermine these goals. But reasonable minds differ on the best way to effectual a successful chapter 13 program. The Code does not prohibit these differences. Form 113 as currently written will undermine the chapter 13 system.

We understand that at one point, chief bankruptcy judges were asked whether they favored the concept of a plan form, and the majority indicated approval. But that was not in response to this specific plan form.

Comment BK-2013-0001-0052—O. Max Gardner III (Attorney, Shelby, NC): There should be a detailed § 524(i) provision for mortgage payments and a provision for all mortgage obligations to be made through the plan via the trustee.

Comment BK-2013-0001-0054—David Kestner: The form is too long. Also, giving the debtor the option to pay directly and not to submit to a wage order will lead to an increase in plan defaults and motions to dismiss.

Comment BK-2013-0001-0055—Larry Foyle (Attorney, Tampa, Fla.): The emphasis on the confirmation hearing will require cases to run in lockstep and cause delays. It is better to treat discrete processes, such as claims objections, relief from stay, lien modifications, etc., separately. They can later be implemented through a confirmation order.

Comment BK-2013-0001-0057—Susan Saidian (Attorney, Wichita, Kan.): The debtor should make payments through the trustee.

I agree with the comments by the D. Kansas judges regarding all of the plan form provisions and related rules changes.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): The plan form should not be mandatory.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: The NCBJ takes no position on the advisability of a national form. If a national form is to be adopted, proposed Official Form 113 should be modified.

April 22-23, 2014 Page 426 of 596

Comment BK-2013-0001-0061—Juliane Lore (Attorney, Mont.): This form is too complex and will encourage an exodus from consumer bankruptcy practice. It will increase confusion and cost in chapter 13. I suggest instead a state-by-state approach to chapter 13 plans.

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): We oppose a mandatory plan form.

Comment BK-2013-0001-0064—Steven Wiechman (Attorney, Kansas):

The form should provide for state-by-state variations.

Comment BK-2013-0001-0065—Judge Terrence L. Michael (Bankr. N.D. Okla.): The plan form should not be mandatory. I agree with the comments submitted by Judge Lamar W. Davis, Jr., Judge Jeffrey Bohm, the judges of the United States Bankruptcy Court for the District of Kansas, NACBA, and others.

A mandatory form is a problem in search of a solution. There is no need for uniformity. The form will not result in more consistent decisionmaking, because judges make decisions based on the Code and case law and not based on the provisions of an Official Form. If the plan form is the greatest thing since sliced bread, people will use it voluntarily. If the form is designed to nationalize chapter 13 practice, I have no desire to see debtors in my court represented by lawyers from New York, Chicago, Dallas, or elsewhere.

This is a mistake of epic proportions.

Comment BK-2013-0001-0066—Edward Claxton: The plan is too comprehensive. It is too detailed for nonlawyers to complete.

Repeated use of the phrase "the claim amount listed on the proof of claim controls" is unnecessary. That phrase should be stated once at the beginning of the form ("The claim amount listed on the proof of claim controls unless the court orders otherwise").

Comment BK-2013-0001-0069—Lesley Hoenig: I oppose a mandatory plan form.

Comment BK-2013-0001-0070—Barbara B. Braziel (Attorney):

I oppose a mandatory plan form.

Debtors will not understand this form. Debtors do not know the amounts requested throughout the form. What will happen after confirmation if the designated payment amounts turn out to be wrong?

Comment BK-2013-0001-0071—Judge Dennis Montali (Bankr. N.D. Cal.): I oppose a mandatory plan form.

Comment BK-2013-0001-0073—Ronald Sykstus: I oppose a mandatory plan form.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): Allow local variations in the form.

Provide the option for a limited service plan (e.g., if the plan contains a minor amendment).

April 22-23, 2014 Page 427 of 596

The form should state if it is a first, second, etc., amended plan. It should require debtors to state the reason for the amended plan.

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): The plan form's treatment of secured creditors is unfair. A secured creditor must object, on 28 days' notice, and will be bound by the confirmed plan. But the debtor can avoid or modify a lien months or years later.

Comment BK-2013-0001-0080—Judge John E. Waites (Bankr. D.S.C.): I oppose a mandatory plan form.

If this form is adopted, pro se debtors should be exempted from having to use this form, or an alternative form should be provided for them. The plan form is complicated and intimidating.

Comment BK-2013-0001-0082—Judge Karen K. Brown (Bankr. S.D. Tex.): I agree with the comments of the judges of the S.D. Tex.

The proposed form is overly lengthy and confusing. It will create litigation and uncertainty.

In addition, proposed Rules 3007, 3012, and 3015, together with proposed Official Form 113, violate the Bankruptcy Code and binding precedent. They eliminate the statutory requirement of a claim objection and alter the evidentiary effects and burdens regarding a proof of claim. They also confuse collateral valuation under § 506 with claim allowance and disallowance under § 502. They violate creditors' due process rights.

Comment BK-2013-0001-0085—Judge Roger Efremsky (Bankr. N.D. Cal.), on behalf of six judges of the bankruptcy court: For reasons given by Judge Lamar W. Davis, Jr., we oppose a mandatory form. It is inflexible and will increase costs.

Comment BK-2013-0001-0087—**Anonymous**: The form requests too much detail that no consumer debtor will understand. It will be difficult and time consuming for lawyers to explain it to their clients.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: We oppose a mandatory plan form.

The form is too long and complex and will increase costs of preparation.

It will impose costs for retraining.

It will lead to more contested confirmation hearings.

It will increase the number of arithmetic errors.

Comment BK-2013-0001-0092—Judge Arthur S. Weissbrodt (Bankr. N.D. Cal.): The plan form will improve chapter 13 practice.

The form should expressly permit confirmation even if the debtor has a pending loan modification request. This procedure has been used successfully in the N.D. Cal.

The form explicitly or implicitly takes a position on open issues of law by including certain provisions.

April 22-23, 2014 Page 428 of 596

There should be a national registry for service upon insured depository institutions.

Comment BK-2013-0001-0093—Gary D. Gray (Office of the Chief Counsel, Internal Revenue Service): We support the concept of a standardized chapter 13 plan form.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys:

The references to interest in the form have not been fully thought out. There is no mention of Code § 1322(b)(10).

There should be a special section (perhaps in Part 4) for domestic support orders generally, in addition to current § 4.5. We include detailed suggestions for the information in a new DSO section.

Comment BK-2013-0001-0098—Henry Hildebrand, III (Chapter 13 Trustee, M.D. Tenn.): I strongly support a national plan form. This is a worthy endeavor.

It will be efficient and cost effective by assisting creditors, courts, and trustees in locating and reviewing provisions of chapter 13 plans. It will reduce "hiding the ball," through design or happenstance. It does not usurp the debtor's ability to submit a plan, because it accommodates nonstandard provisions in Part 9. It can help develop a more cohesive body of law interpreting chapter 13 by moving away from parochial plan forms with disparate "boilerplate" provisions.

To ease concerns about the length of the form, the "none" box language could be made more forceful by substituting "should" for "need" so that it reads "If 'none' is checked, the rest of _____ need should not be completed or reproduced."

Comment BK-2013-0001-0099—Chief Judge David S. Kennedy (Bankr. W.D. Tenn.), on behalf of four of five judges of the court: All but one of the judges of our court oppose a mandatory plan form.

The form benefits only national creditors and national law firms.

The form is too long. We have a one-page form.

Our chapter 13 trustees oppose the form. They administer chapter 13 plans and supervise the debtor's estates.

Comment BK-2013-0001-0100—Byron Meredith (Chapter 13 Trustee, S.D. Ga.): We compliment the Advisory Committee for its efforts.

There should be a provision for preconfirmation adequate protection payments.

Comment BK-2013-0001-0101—Chief Judge James K. Coachys (Bankr. S.D. Ind.): I agree with the comments of Judge Lamar W. Davis Jr. and the judges of the bankruptcy court for the District of Kansas. The plan form should not be mandatory.

Over time, a national form will become more unworkable as courts render conflicting decisions about it and the applicable substantive law.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): I support the concept of a plan form but not a nationwide one.

April 22-23, 2014 Page 429 of 596

Comment BK-2013-0001-0107—Lauri Schmid: Minnesota's plan is superior. Each state has its own laws, rules, and cases and is in the best position to administer its own plan.

Comment BK-2013-0001-0108—Mary Kate Kelley-Scheidler (Attorney, Minn.): The form is easy to read. But one size does not fit all.

The form lacks a place to designate a start date and amount for payments based on creditor class.

Comment BK-2013-0001-0109—Devin Derham-Burk (Chapter 13 Trustee, N.D. Cal. (San Jose Division)): The national plan form should not be mandatory.

There is no uniformity among debtors. The plan form will be difficult and costly to administer.

The form is too long and complex. There are too many places where mistakes will be made. It will be harder for trustees to review plans.

Comment BK-2013-0001-0110—Amy Tanner: The national plan form should not be mandatory.

The plan form is too complex and confusing. The details will require many time-consuming plan amendments before confirmation. There is no "one-size-fits-all" plan for the entire nation.

Comment BK-2013-0001-0111—Judge Elizabeth W. Magner (Bankr., E.D. La.): I thank the Rules Committee for its work on these proposals.

Comment BK-2013-0001-0112—John V. LaBarge, Jr. (Chapter 13 Trustee, St. Louis, Mo.): I agree with the comments of the bankruptcy judges for the W.D. Wash.

The confirmation process will not be workable, because creditors will not be able to respond to valuation issues prior to confirmation. This is unfair to creditors. Liens should not be stripped via confirmation order. This should be handled formally.

The form will require more care to complete and will lead to more errors.

Comment BK-2013-0001-0113—Judge Jim D. Pappas (Bankr. D. Idaho): I support a national plan form.

In my rural district, there are few bankruptcy specialists. It is difficult for generalist lawyers to keep current on chapter 13. A national plan form will help these lawyers. It will lead to more attorneys handling chapter 13 cases.

Chapter 13 should be a national, not a local, concept.

Comment BK-2013-0001-0114—Bradley J. Halberstadt (Attorney, Roseville, Minn.): I applaud the Advisory Committee for proposing a national plan form.

The plan form needs a provision for pre-confirmation adequate protection payments. Requiring a separate motion and order for those payments wastes money and time.

There should be a mechanism for calculating interest on over-secured claims prior to confirmation.

April 22-23, 2014 Page 430 of 596

Comment BK-2013-0001-0117—Chief Judge Mary P. Gorman (Bankr. C.D. Ill.): I oppose a mandatory plan form. I agree with the comments of Judge Lamar W. Davis and the bankruptcy judges of the D. Kan.

Comment BK-2013-0001-0119—Kathleen Horne (Attorney, Savannah, Ga.), on behalf of the Bankruptcy Lawyers Advisory Committee of the Bankruptcy Court for the S.D. Ga.: We oppose a mandatory plan form. If it is adopted, it should be tested first in selected districts.

The plan form should provide for adequate protection payments.

Comment BK-2013-0001-0120—Judge Robert E. Grant (Bankr. N.D. Ind.): I oppose a mandatory plan form. It should be offered as a guideline.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: We oppose a mandatory plan form.

The current Maryland model plan is better.

The proposed plan form is complicated and will lead to errors.

If adopted the plan form will increase costs and lead to uncertainty in the short term.

Consider an exhibit for a certificate of service.

Comment BK-2013-0001-0123—Randall Smith: A national plan form will do more harm than good.

The published form is inaccurate in some respects.

The form purports to resolve issues that are subject to local differences in law and practice.

Comment BK-2013-0001-0125—Judge Rebecca B. Connelly (Bankr. W.D. Va.): I support the adoption of an Official Form for chapter 13 plans. I thank the Advisory Committee for its work on this project.

Comment BK-2013-0001-0127—Chief Judge Susan Barrett (Bankr. S.D. Ga.), on behalf of the judges of the court: We oppose adoption of the plan form. A national plan form will damage the professionalism of practice in our court. It will impair the close relationship between debtors and local attorneys.

Regional consolidation of the consumer *creditor* legal work has led to: (i) unprepared counsel; (ii) poor communication between debtors' and creditors' counsel; (iii) reduced accountability of counsel to the court in which they practice; (iv) increased delays and expenses. A national plan form will lead to similar problems with consumer *debtor* practice. Many of our debtors are poor, undereducated, elderly, ill, and ill-suited to fend for themselves. They will become untethered from local attorneys and staffs who are best suited to advise and guide them.

Comment BK-2013-0001-0129—Beverly M. Burden (Chapter 13 Trustee, E.D. Ky.): I support the effort to bring more uniformity to chapter 13 practice. I do not object in principle to a national plan form.

April 22-23, 2014 Page 431 of 596

If there is too much opposition to the plan form, consider instead a form with a summary of the proposed treatment of secured creditors similar to the chapter 7 Statement of Intention.

Comment BK-2013-0001-0130—John P. Gustafson (Chapter 13 Trustee, N.D. Ohio (Western Div.)): I support a mandatory Official Form for chapter 13 plans.

Other bankruptcy Official Forms are mandatory.

Bankruptcy laws should be uniform. See U.S. Const., art. I, § 8. Even without a constitutional mandate, we have long encouraged the adoption of uniform acts.

Trustees receive information from data-enabled forms. But most chapter 13 plans must be handled manually. The ability to extract data from these nonstandard forms is limited. A national plan form will be helpful.

There is no evidence that a plan form will increase attorney's fees. In my division, there is no local chapter 13 plan, and we have one of the lowest chapter 13 attorney fee costs in the country.

The decreased costs for creditors should be considered.

A uniform plan form will help the appellate process. Cases will no longer turn on unique plan language that differs for each local area.

Criticism of the plan form is contradictory. Comments cite the need for flexibility regarding local culture and parochial interests while at the same time complaining that there are too many choices in the plan form.

I urge a change in Proposed Official Form 113. There should be two places for nonstandard provisions. One should be for nonstandard provisions that affect the treatment of creditors and another for nonstandard provisions that do not affect creditors.

Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.): I oppose the adoption of a mandatory plan form. I agree with the comments of Judge Terrence Michael.

The proposed form will add complexity and cost.

There will be arithmetic errors and inconsistent calculations.

Comment BK-2013-0001-0134—Marlene Martel (Attorney), on behalf of Ford Motor Credit Company: The plan form will improve uniformity and consistency in chapter 13 practice. Part 1 and Part 9 will expedite review and administration.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): The form uses the phrase "under the plan" inconsistently. To avoid confusion regarding confirmation, discharge, and administration, the phrase should be "paid by the trustee under the plan" or "paid under the plan by the trustee" if the intent is to have the trustee serve as the disbursing agent.

Add a sentence clarifying that the form does not take positions on issues such as conduit mortgage payments or the payment of post-petition tax refunds.

The form does not mention the date from which interest begins to accrue.

I have attached a revised plan form that addresses my concerns.

April 22-23, 2014 Page 432 of 596

Comment BK-2013-0001-0139—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): I oppose a mandatory plan form. The form is different from my current practice and must reflect the drafter's current practice. Some provisions are contrary to the Code.

The form should state that it does not change the Code, local rules, standing orders, or case law.

Remove references to estimated payments.

Comment BK-2013-0001-0140—Penelope Souhrada: I agree with the comments of Henry Sommer and the NCBJ. What better source for comments than bankruptcy judges?

A mandatory form makes no allowance for local laws and practices. The form will become the de facto law of chapter 13 whether or not intended.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): The plan form should not be mandatory. I agree with Judge Lamar Davis and Judge Terrence Michael.

Include a provision for adequate protection payments.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): I oppose a mandatory plan form. In the alternative, allow local options on the Official Form.

The plan form interferes with courts' discretion in administering chapter 13.

Comment BK-2013-0001-0144—Chontele A. McIntyre (Attorney, Memphis, Tenn.), on behalf of Bankruptcy Liaison Committee, W.D. Tenn. (Western Div.): We oppose a mandatory plan form. The form should serve as a model instead.

The form is more complex than our one-page model plan. It will increase the cost of representation and review.

It will increase the number of objections to confirmation.

Comment BK-2013-0001-0147—Mark Bulovic (Attorney, Savannah, Ga.): I oppose the plan form. The chapter 13 system is not broken. There is no need for uniformity for its own sake. The plan form will encourage the unauthorized practice of law. It will dumb down bankruptcy practice.

There is no provision for pre-confirmation adequate protection payments.

Some courts have held that due process requires more than a plan provision to strip a lien. The plan form and rule amendments overrule these decisions.

This form appears to be part of an effort to encourage pro se filings. This is a bad idea.

Comment BK-2013-0001-0148—Mark J. Giske (Attorney, W.D. Wash.): I support movement toward a national plan form, but I oppose a mandatory plan form. The plan form should be a model, and each judicial district could "opt in" to use it.

Comment BK-2013-0001-0150—Judge Jennie D. Latta (Bankr. W.D. Tenn.): I support a national plan form. The disparity in chapter 13 plans and procedures creates barriers to entry into bankruptcy practice and increases the costs of national creditors, who pass those costs to borrowers.

April 22-23, 2014 Page 433 of 596

The proposed form will promote more attentive and careful representation of consumer debtors. It raises questions that counsel should address in interviews with debtor clients but that are often overlooked.

Nevertheless, it is important to allow some local variations. We require payroll deductions, and we are a mortgage conduit district. Allow these and similar variations.

The plan form is too long. It will be costly to print and serve it.

Comment BK-2013-0001-0152—Cathleen Moran: There is no need for a uniform national plan. There is no "problem" to fix.

The form is too complex. It will increase costs.

It will require multiple amendments.

Comment BK-2013-0001-0153—James M. Davis (Attorney, Nashville, Tenn.): I strongly support the creation of a national chapter 13 plan form. Local differences in chapter 13 practice are a reason to favor—and not oppose—a national plan form.

Comment BK-2013-0001-0156—Daniel Press (Attorney, Va.): I am a consumer bankruptcy attorney representing primarily debtors.

I agree with the comments of Henry Sommer. I also agree with the comments of Judges Connolly, particularly with regard to the use of "serve," and Martin Teel.

I disagree with comments opposing a national plan form. These are parochial views of those saying "our way is best." In many cases, these local practices are contrary to the Code, but it is impossible to challenge them. Increased uniformity will make it easier to ensure that local practices do not contradict the law.

I disagree with comments by Nancy Spencer Grigsby. Procedures for avoiding or modifying liens through a plan do not violate Fourth Circuit law. In one case, the court held only that proper service and notice of the actual plan should be given. In the other, it held that an adversary proceeding was required to invalidate a lien under *nonbankruptcy* law—not to value collateral under § 506—and the decision was based on the Bankruptcy Rules in effect at the time. And prompt cure in order to assume a lease is consistent with § 365.

It is the Maryland plan and local rules that violate the law by mandating choices that the Code leaves to debtors to propose in a plan This is a problem with other local plans, and it highlights why a national plan form is the best approach.

Comment BK-2013-0001-0158—Debra Miller (Chapter 13 Trustee, N.D. Ind.): I support a national plan format.

We began using the new proposed plan form in our district 12 months ago. There has been no outcry from creditors, no litigation about plan terms, and the plan has worked well.

See testimony of January 31, 2014.

Comment BK-2013-0001-0160—Judson C. Hill (Attorney, Savannah, Ga.): The plan form is riddled with ambiguous, extraneous, and unnecessary provisions.

Localized plans benefit the courts and those who appear and practice before them. Banks and other national creditors are behind the push for a national plan form, because they will save money by centralizing their work in-house or with large bankruptcy firms that practice by email and telephone. Professionalism and order in bankruptcy courts will suffer.

April 22-23, 2014 Page 434 of 596

Part 1: Notice to Interested Parties

Comment BK-2013-0001-0007—Judge Jacqueline P. Cox (Bankr. N.D. Ill.): The following sentence should be substituted for the last sentence in Part 1: "In addition, you must file a Proof of Claim before the Claims Bar Date, or one must be filed on your behalf, in order for you to be paid under any plan that may be confirmed."

Comment BK-2013-0001-0008—Jon Waage (Chapter 13 Trustee, M.D. Fla.): Part 1 should state clearly that the inclusion of a provision in the plan form does not mean that the option is available in every district. For example, debtors may argue that they can propose to pay secured creditors directly, even though many jurisdictions require payment through the chapter 13 trustee. If this were to happen, it would effectively shut down many chapter 13 trustee offices.

Comment BK-2013-0001-0014—K. Michael Fitzgerald (Chapter 13 Trustee, W.D. Wash.): The plan fails to state that choices on the form do not mean debtors can choose options inconsistent with local rules, customs, and practices. For example, allowing debtors to make post-petition mortgage payments directly is inconsistent with the local rule in the W.D. Wash. and would require my office to object to any plan proposing such a distribution scheme.

The form should indicate clearly whether debtors are eligible for a discharge.

The form should indicate whether the debtor's income is above or below the median. This will make it easier for the trustee and creditors to determine if the proposed dividend to allowed nonpriority unsecured creditors is appropriate.

Comment BK-2013-0001-0030—Judge Diane Finkle (Bankr. D.R.I.): The following language should be included to deal with implied consent: "Any creditor, including a secured creditor or holder of a judicial lien, who does not timely file an objection to confirmation of the proposed plan or any of the above checked motions shall be deemed to have accepted the treatment of its claim as proposed in the plan."

Comment BK-2013-0001-0032—Judge Jack B. Schmetterer (Bankr. N.D. Ill.): There should be a clearer warning to creditors on the front of the plan. Insert the following language: "The plan seeks to limit the amount of a secured claim *or may void that claim entirely even if that claim has not been filed*, as set out in Part 3, Section 3.2, which may result in a partial payment or no payment at all to the secured creditor."

Comment BK-2013-0001-0035—Robert G. Drummond (Chapter 13 Trustee, D. Mont.): The plan should indicate clearly whether a debtor is eligible for a discharge.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.):

The form does not provide a place to indicate that it is an original, amended, or modified plan.

April 22-23, 2014 Page 435 of 596

The checkboxes in Part 1 will not serve their intended purpose of putting creditors and others on notice that those items are in the plan. Debtors' counsel will always check all of the boxes by default so as to ensure that entire sections of the plan are not made inapplicable.

The language regarding confirmation will require more confirmation hearings, particularly in districts that do not currently hold a confirmation hearing on every plan. For example, in the E.D. Tenn., by local rule a confirmation hearing is necessary only if an interested party objects prior to the meeting of creditors. Also, the 7-day notice period for objections to confirmation is too short for debtors' attorneys to work out confirmation issues without the need for a hearing.

Comment BK-2013-0001-0041—Robert E. Hyman (Chapter 13 Trustee, E.D. Va.): The date of the confirmation hearing should be set forth in Part 1.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: The phrase "may be" in the last paragraph of Part 1 should be changed to "is" so that the concluding sentence reads: "In addition, you must file a proof of claim—or one must be filed on your behalf—in order for you to be paid under any plan that is may be confirmed."

Comment BK-2013-0001-0063—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): The notice in Part 1 should be revised to include the deadline for filing an objection to confirmation. That deadline should be 21 days after the debtor's filing of the notice of the time to object to confirmation.

Comment BK-2013-0001-0066—Edward Claxton: The entire plan affects creditors. Highlighting three sections implies that other sections are not important.

Comment BK-2013-0001-0070—Barbara B. Braziel (Attorney): The form should be identified as a "Plan and Motion."

Debtors will check all boxes to preserve their rights. This will create more litigation.

Comment BK-2013-0001-0071—Judge Dennis Montali (Bankr. N.D. Cal.): The first sentence should state: "The plan seeks to limit or eliminate"

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): This part should include the noticing standards when debtors propose to avoid or reduce a secured creditor's lien.

This part invites debtors to insert provisions that violate the Code.

The plan form requires an objection to be "filed" 7 days before the confirmation hearing. A creditor should have more time to object, including an oral objection at confirmation (e.g., in districts where the § 341 meeting and confirmation are held at the same time).

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys:

The checkbox for nonstandard provisions should have a space to describe what parts of the plan are affected.

April 22-23, 2014 Page 436 of 596

A checkbox should be added if a debtor proposes to pay interest in order to ensure compliance with §§ 1322(b)(10) and 1325(a)(4).

Comment BK-2013-0001-0100—Byron Meredith (Chapter 13 Trustee, S.D. Ga.): It is not clear what obligation the trustee has when information is provided in §§ 3.2, 3.4, or Part 9 but the applicable box in Part 1 is not checked. Must the trustee object? Must the trustee ignore those sections and make no disbursements to the creditors listed in them?

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): The notice provisions are inadequate. The possible consequences of a creditor's failure to object, and of confirmation, should be conspicuous and unequivocal.

Comment BK-2013-0001-0108—Mary Kate Kelley-Scheidler (Attorney, Minn.): The checkbox for nonstandard provisions is a good feature.

Comment BK-2013-0001-0119—Kathleen Horne (Attorney, Savannah, Ga.), on behalf of the Bankruptcy Lawyers Advisory Committee of the Bankruptcy Court for the S.D. Ga.: Debtors will routinely check all three checkboxes.

Comment BK-2013-0001-0125—Judge Rebecca B. Connelly (Bankr. W.D. Va.): I agree with Judge Martin Teel's comments on the language regarding filing an objection to confirmation. Also, the party receiving the plan may not have received the Notice of Bankruptcy and may not know the date, time, or place of the confirmation hearing. It would be better to provide a particular deadline to file an objection (e.g., 21 days from the date of the plan).

Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.): Debtor's counsel will always check all the boxes in this Part, rendering the notice meaningless.

The form does not include an indication whether the plan is an amended or modified plan.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: Add space to indicate what version of an amended plan is being filed.

We recommend stylistic changes regarding the notice and warning language.

Creditors' counsel in our group believe filing an objection 7 days before confirmation is problematic. Debtors' counsel disagree.

Comment BK-2013-0001-0134—Marlene Martel (Attorney), on behalf of Ford Motor Credit Company: Add space to indicate whether it is an amended plan and, if so, what parts have been amended.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Add more spaces to identify the names of each debtor, the case number, counsel, etc.

Add space to identify the plan number.

Add a checkbox for modified plans.

April 22-23, 2014 Page 437 of 596

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Courts that do not allow lien avoidance through chapter 13 plans should be able to remove this provision from the plan form.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): State that the debtor may omit parts of the form that are not used.

Nonstandard provisions should be placed in each part where they logically belong, with that placement indicated in Part 1.

Because a bankruptcy court should be able to direct that no plan shall seek a determination of the amount of a secured claim or request lien avoidance, two of the checkboxes may be unnecessary.

Change references to "these papers" to "this plan."

We set a deadline to object of 28 days after the plan is filed. The plan form interferes with this local variation. Although the plan form states that the court could order otherwise, doing so is unworkable and expensive. It will confuse creditors.

Comment BK-2013-0001-0147—Mark Bulovic (Attorney, Savannah, Ga.): Debtors' attorneys will check all three boxes in Part 1.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Comment BK-2013-0001-0153—James M. Davis (Attorney, Nashville, Tenn.): Draw greater attention to the requirement to file a proof of claim in order for a claim to be paid.

Language throughout the form should be changed to refer to payment of *allowed* claims.

Comment BK-2013-0001-0161¹—Diana D. Herman (Attorney, San Francisco, Cal.), on behalf of Insolvency Law Committee, State Bar of California Business Law Section:

There appears to be an inconsistency between the proposed changes to Rule 3015 and the plan form. Part 1 of the plan form provides that an objection must be filed at least 7 days before the confirmation hearing "unless otherwise ordered by the Bankruptcy Court." This is important for districts that confirm plans on negative notice. But it is unclear whether the proposed changes to Rule 3015(f) relating to the 7-day deadline to object to confirmation may be altered by the court.

Part 2: Plan Payments and Length of Plan

Section 2.1 (payments to the trustee)

Comment BK-2013-0001-0024—Rod Danielson (Chapter 13 Trustee, C.D. Cal.), on behalf of the five chapter 13 trustees of the C.D. Cal.: Section 2.1 should provide for the

April 22-23, 2014 Page 438 of 596

¹ This comment was submitted after the close of the public comment period.

possibility that debtors may propose plans with more than two payment amounts at different times. The addition of another "step" would cover most situations, and this issue may be addressed in any event by chapter 13 software vendors.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): Section 2.1 is too rigid as to how plan payments are made. There is no provision for "split payments" between husband and wife. There is only one place for "step up payments" when disposable income increases. Debtors may have many more payment changes over the life of a plan.

The form requires counting weeks on a calendar as opposed to averaging in order to get an accurate total amount for § 2.4 (estimated total amount of plan payments) and Exhibit B. There is no specified date or even month or year when the payments will begin. By local rule in the E.D. Tenn., the debtor must make a month's worth of payments within 30 days of filing in order to pay administrative expenses in installments and adequate protection payments. These provisions will now end up as non-standard in Part 9.

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.): This should not be limited to only two plan payment steps.

Comment BK-2013-0001-0049—Anne V. Kealing (Attorney, Edina, MN): More payment steps should be included.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: This section does not provide enough detail for the trustee. Also, in districts where debtors are engaged in seasonal work, there should be more options for varying "step" payments.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): Add more payment "steps."

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): Add more payment "steps."

Provide for other plan lengths.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): State specifically the actual plan length.

Comment BK-2013-0001-0080—Judge John E. Waites (Bankr. D.S.C.): Add more payment "steps."

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Add more payment "steps."

Comment BK-2013-0001-0112—John V. LaBarge, Jr. (Chapter 13 Trustee, St. Louis, Mo.): Add more payment "steps."

April 22-23, 2014 Page 439 of 596

Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.): Add information about payments from each joint debtor. Add information on step payments, dates, etc. Alert the debtor of the statutory requirement to commence plan payments within 30 days of filing unless the court orders otherwise.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: Add flexibility to the payment structure.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Include a commencement date for payments.

Comment BK-2013-0001-0138—Robert Morje: Add more payment steps.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): The plan duration and plan amount provisions need substantial changes. Provide different options for "100%" plans and variable rate (pro rata) plans.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Include a start date for payments. By local rule in the E.D. Tenn., payments must begin within 30 days of filing.

Comment BK-2013-0001-0154—David S. Yen (Attorney, Chicago, Ill.): Add more payment steps. In the alternative, add a checkbox referring to Part 9.

Section 2.2 (manner of payments to the trustee from future earnings)

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT: Section 2.2 should not give a choice regarding payroll deductions. The lack of a place to designate the address of an employer for the payroll order will cause delay.

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): The form should not include a provision that makes it appear that a payroll deduction is optional. This is a matter of judicial determination and not merely a choice for the debtor to make. This section should be left out entirely, or else the debtor should be required to give a specific reason why a payroll deduction is not possible.

Comment BK-2013-0001-0030—Judge Diane Finkle (Bankr. D.R.I.): The plan form should include a disclaimer that not all provisions may be available in a debtor's district. This disclaimer is especially applicable to the proposed plan payment options.

Comment BK-2013-0001-0034—Dana Milby (Attorney, Wichita, Kan.): Part 2 does not include a place to designate the address of an employer for the payroll deduction order.

Comment BK-2013-0001-0041—Robert E. Hyman (Chapter 13 Trustee, E.D. Va.): Debtors should not be given the choice with respect to a payroll deduction. If given the choice,

April 22-23, 2014 Page 440 of 596

debtors will not voluntarily select a payroll deduction because most debtors do not want their employers to know of the bankruptcy case. Our court now requires a payroll deduction if the debtor is not self-employed.

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.): A wage order should not be a choice for the debtor to make. It is the trustee's duty to determine the necessity of a wage order. Remove this option.

Comment BK-2013-0001-0049—Anne V. Kealing (Attorney, Edina, MN): Consider a space to specify another method of payment (e.g., ACH).

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: There is no place for details about the wage order (e.g., differing amounts for each joint debtor; the address for the payroll deduction).

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): Provide more detail for division of payments among sources.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): Remove the option for debtors to make payments without a wage order. We require a wage order in our district.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): Remove the option for debtors to make payments without a wage order.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: Remove the debtor's choice with respect to a wage order.

Comment BK-2013-0001-0098—Henry Hildebrand, III (Chapter 13 Trustee, M.D. Tenn.): Expand § 2.2 to identify how much will be contributed by payroll deduction from each debtor in a joint case.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: Make a payroll deduction order the default. Otherwise, the plan failure rate will increase.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: There is too much detail and complexity in this section.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Move the provision requiring turnover of the tax return to immediately above "check all that apply."

Require debtors to turn over both personal and business tax returns, and any state returns. The phrase "plan term" is not defined. Does it mean the applicable commitment period or the number of months listed in § 2.1?

April 22-23, 2014 Page 441 of 596

Comment BK-2013-0001-0139—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): Rephrase the title of § 2.2.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Remove this section or permit courts to alter it according to local usage.

We require payroll deduction in our district.

Include ACH as an option, which we use for self-employed debtors.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): The plan form should discourage debtors from paying directly and should include authorization for a wage order.

Comment BK-2013-0001-0153—James M. Davis (Attorney, Nashville, Tenn.): Add greater flexibility to this section. Permit debtors to select some payments by wage order but others by direct payment (e.g., if one debtor in a joint case is self-employed).

Comment BK-2013-0001-0154—David S. Yen (Attorney, Chicago, Ill.): The first sentence should refer to future income, not future earnings.

Clarify whether selecting the payroll deduction checkbox obviates the need for a separate order signed by the judge.

Provide space for detailed terms of the payroll deduction order.

This section may confuse pro se debtors. They might not realize that direct payment to the trustee is needed until the payroll deduction takes effect.

Section 2.3 (additional payments to the trustee)

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.): Section 2.3 gives the debtor two choices—to turn over any tax refunds received during the plan term, or only refunds in excess of a specified amount. These choices are not statutorily based, are not based on circuit cases, and are not a complete description of all possible options. Tax refunds are not necessarily a part of the estate. In Kansas, for example, state law permits exemption of the refund portion attributable to the earned income tax credit. Also, trustees must consider other factors before requesting refunds, such as the best interest of creditors and the debtor's history in other cases.

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- Section 2.3 should provide an option for the debtor to turn over tax refunds for a period other than the plan term (e.g., the first three years of a five-year plan).
 - The sentence beginning "On or before April 20 . . . " should be removed.
- The turnover of the tax refund and the tax return should not be required to accompany each other. Add an option for the trustee to receive a copy of the return without also receiving the refund.

April 22-23, 2014 Page 442 of 596

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): Section 2.3 calls for the debtor to indicate additional income that will be paid to the trustee. The presumption should run the other way—the form should state that all income is included except as specifically excluded on the form. This section should state: "The plan will include additional payments to the trustee from tax refunds, lawsuits and claims, property sales, and all other sources of post-petition income and assets, except the following:

________."

If this suggestion is not adopted, § 2.3 should nonetheless specify lawsuits and settlements as examples of "other sources of funding."

Section 2.3 should provide for debtors to turn over their tax returns within 7 days after the return is filed, or by April 20 of each year, whichever is earlier.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): If this form is to help pro se debtors file chapter 13, they are unlikely to agree to turn over any of their tax refunds to the trustee. This just creates more work for the trustee or bankruptcy judge, because no pro se will propose turning over a tax refund.

The form takes away the trustee's choice not to receive the debtor's tax return, which will flood trustees with unneeded tax returns. This should be left as an option for the trustee administering the case.

There is very limited space for other sources of funding. It would be better to include a checkbox labeled "see special provisions."

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: The section regarding tax refunds should be deleted. Where appropriate, the debtor could provide for funding from a tax refund under "other sources of funding."

The tax refund language is objectionable for three reasons. First, it does not give the debtor the option of not turning over a tax refund, which is appropriate in many cases. To compute their disposable income in Form 22C, debtors have already accounted for tax refunds if they accurately report their tax liability. Requiring turnover of the refund is double counting. Second, future changes in a debtor's income and expenses may occur, and the debtor should not be required to turn over tax refunds years after the petition without knowing those changes. A plan modification is the appropriate way to deal with such variable numbers. Third, the language about furnishing tax returns to the trustee is ambiguous. It appears to require that the return be submitted to the trustee in every case in conflict with Code § 521(e)(2), which requires the debtor to provide a tax return to the court (not the trustee) only if requested in a specific case by the U.S. Trustee or a party in interest. Also, many debtors seek extensions to file their returns, and thus will have nothing to provide on April 20.

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.): There should be greater flexibility as to the length of time in which refunds are required to be turned over. We require refunds for the first three years, and such an option is not given in this section.

Comment BK-2013-0001-0049—Anne V. Kealing (Attorney, Edina, MN): This provision should make clear whether the tax return, the refund, or both, are due on April 20. Also, many debtors seek extensions and do not file tax returns on April 15.

April 22-23, 2014 Page 443 of 596

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: We commend the language regarding annual submission of a copy of the tax return to the trustee. But other language does not account for districts, such as D. Kan., where portions of the return (e.g., earned income credit) are exempt. The section does not include enough options regarding the length of time for which refunds will be pledged.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: The sentence making submission of the tax return automatic should be deleted. It contradicts the Code. Section 521(f) requires submission of the tax return only "[a]t the request of the court, the United States trustee, or any party in interest" The April 20 deadline ignores the right of debtors to obtain extensions of the time to file tax returns.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): I object to requiring the debtor to turn over a tax refund and annual tax returns. It will increase trustees' costs.

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): We object to the tax refund provision.

Comment BK-2013-0001-0070—Barbara B. Braziel (Attorney): The tax return provision will increase the trustees' costs.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): The tax return language conflicts with Code § 521(f)(4).

There is no deadline for turnover of refunds.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): Specify a date for turning over tax refunds. Require copies of state and federal returns.

Comment BK-2013-0001-0080—Judge John E. Waites (Bankr. D.S.C.): I agree with the NCBJ's comments.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: Specify a date for turning over tax refunds.

Comment BK-2013-0001-0092—Judge Arthur S. Weissbrodt (Bankr. N.D. Cal.): The tax refund provisions should not be part of the plan form.

Comment BK-2013-0001-0096—Nik Geranios: I agree with NACBA's comments.

24

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Require the debtor to include the property tax ID number if real property is referenced.

Comment BK-2013-0001-0098—Henry Hildebrand, III (Chapter 13 Trustee, M.D. Tenn.): Greater flexibility is needed with respect to tax refunds. Debtors may not know the "in excess of" dollar amount if it depends on an earned income tax credit. How the contributed tax refunds are treated (as additional amounts that increase distributions to unsecured creditors or as amounts that accelerate payments to creditors for an earlier plan completion date) should be included.

Comment BK-2013-0001-0100—Byron Meredith (Chapter 13 Trustee, S.D. Ga.): More detail is needed in the tax refund provision. Specify the manner in which tax refunds will be turned over. Specify a deadline.

Comment BK-2013-0001-0111—Judge Elizabeth W. Magner (Bankr., E.D. La.): The plan form excludes after-acquired property by default from distribution to creditors. The form should provide that all property acquired during the case is available for distribution.

Comment BK-2013-0001-0119—Kathleen Horne (Attorney, Savannah, Ga.), on behalf of the Bankruptcy Lawyers Advisory Committee of the Bankruptcy Court for the S.D. Ga.: The debtor should have the option not to provide tax refunds.

The tax provisions will be burdensome for trustees, debtors, and counsel.

Comment BK-2013-0001-0121—Peter D. Schneider (Attorney, Philadelphia, Pa.), on behalf of Community Legal Services, Inc., of Philadelphia: Delete the tax refund provision. It improperly limits the debtor's options.

The language regarding submission of tax returns is inconsistent with the Code. Also, a debtor may receive an extension to file a tax return.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: The tax refund provision should include more options: e.g., "not applicable"; a percentage option, which some trustees require; an amount in excess of the amount pro-rated as income on Schedule I; "other."

The phrase "any tax refunds in excess of . . . received during the plan term" is vague. Does it mean "plan length" or "applicable commitment period"?

Do not put tax refunds and sale plan terms in the same section.

Comment BK-2013-0001-0123—Randall Smith: Debtors should not be required to turn over their tax refunds.

Debtors may seek an extension of time to file their taxes.

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: Clarify that if a debtor agrees to turn over future

April 22-23, 2014 Page 445 of 596

refunds, it is the responsibility of the debtor (and not the payor of the the refund) to notify the trustee of receipt of the refund.

Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.): Clarify the tax refund provision.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: There is too much detail and complexity in this section.

Comment BK-2013-0001-0138—Robert Morje: Debtors may seek an extension to file their tax returns. Use a deadline based on the date of filing instead of a date certain.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): The tax refund provision does not include alternative options. Debtors may seek an extension to file their tax returns.

The "other sources of funding" provision should not limit the trustee if additional sources of funding later become available or are discovered.

It is not clear whether additional payments into the plan increase the base or pay the existing base more quickly.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Directing turnover of tax refunds is administratively unworkable.

Comment BK-2013-0001-0147—Mark Bulovic (Attorney, Savannah, Ga.): Debtors may seek an extension to file their tax returns.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Comment BK-2013-0001-0154—David S. Yen (Attorney, Chicago, Ill.): Include separate provisions for submitting the tax return and the tax refund. That way, a debtor could choose to supply the trustee with a tax return without agreeing to commit to paying any of the refund.

Will selecting the checkbox authorize the trustee to have the IRS send the tax refund directly to the trustee? This is not the practice in the N.D. Ill. The debtor should be informed of this consequence.

Section 2.4 (estimated total amount of plan payments)

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.): The estimated total amount of plan payments is irrelevant. This will generate unnecessary and inaccurate information.

April 22-23, 2014 Page 446 of 596

- Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT: Section 2.4 is too imprecise to administer. An "estimated" amount says nothing to the trustee, creditors, or interested parties.
- Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): What information is sought in this portion of the plan? Is it the sum of payments listed in §§ 2.1 and 2.3? Is § 2.4 simply for informational purposes? If so, the form should say so.
- Comment BK-2013-0001-0014—K. Michael Fitzgerald (Chapter 13 Trustee, W.D. Wash.): Section 2.4 is unnecessary, because the amount of payments is already set out in § 2.1. The additional section creates confusion for parties reviewing the plan.
- Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): The amount in § 2.4 is irrelevant, because it does not include the tax refunds and other sources of funds. Also, in conduit districts, such as the E.D. Tenn., the amount in this section includes ongoing mortgage payments, which will change over the life of the plan.
- Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): Remove this provision. In the alternative, strike "estimated" and add tax refunds.
- Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: Estimates are irrelevant.
- Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: The plan payment amount should not be an estimate.
- Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.):
- What will happen if the estimated plan payment (§ 2.4) conflicts with the plan payment (§ 2.1)?
- Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: Section 2.4 should not be an estimate. Trustees like certainty in plan funding.
- Comment BK-2013-0001-0139—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): Remove this section. Total payments, not estimated, should be under § 2.1.
- Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Clarify that estimates are not binding.
- Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

April 22-23, 2014 Page 447 of 596

Comment BK-2013-0001-0154—David S. Yen (Attorney, Chicago, Ill.): There should be a place for debtors to state a minimum total of plan payments.

Section 2.5 (applicable commitment period)

Comment BK-2013-0001-0008—Jon Waage (Chapter 13 Trustee, M.D. Fla.): Section 2.5 needs a third option for lengths other than 36 or 60 months.

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT: Section 2.5 needs a third option (in addition to 36 months and 60 months)—a blank space for debtors to propose alternative plan lengths.

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): The form should indicate whether the debtor is above or below median income rather than the number of months.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): The form violates the Bankruptcy Code because there is no provision for plans with lengths other than 36 or 60 months. There should be an option for plans longer than 36 months but shorter than 60 months (e.g., 48 months).

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: The statement of the applicable commitment period should be in § 2.1 or § 2.2.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): This section is unnecessary.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): Section 2.5 is unnecessary. It is addressed in § 2.5.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: Remove this section. Form B22C controls the applicable commitment period.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Add a checkbox stating whether the plan dedicates all projected disposable income for the applicable commitment period.

Allow more options for plan length. Include an "other" checkbox.

Comment BK-2013-0001-0119—Kathleen Horne (Attorney, Savannah, Ga.), on behalf of the Bankruptcy Lawyers Advisory Committee of the Bankruptcy Court for the S.D. Ga.: Allow more options for plan length.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.:

April 22-23, 2014 Page 448 of 596

The plan term should be stated precisely. The applicable commitment period is merely informative.

Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.): The plan length or term should be stated clearly. The applicable commitment period is not necessarily the plan term.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: There is no Code requirement that plan must be either 36 or 60 months. Indicate instead whether the Form 22C calculations apply and what those calculations indicate.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Comment BK-2013-0001-0154—David S. Yen (Attorney, Chicago, Ill.): Move § 2.5 to the beginning of Part 2. That way, the reader will quickly see if a reduction in plan payments after 36 months is justified because the applicable commitment period is 36 months.

Part 3: Treatment of Secured Claims

Part 3 (general)

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: The form should provide for other types of treatment for secured creditors (e.g., claims not treated under the plan).

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): Add a separate section for secured claims not treated under the plan.

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): Provide for pre-confirmation adequate protection payments.

Comment BK-2013-0001-0080—Judge John E. Waites (Bankr. D.S.C.): I agree with the NCBJ's comments.

Provide a place for the date of the resumption of direct mortgage payments by the debtor.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Even if the claim controls, the form should require that the plan be revised to reflect the claim amount. This is preferable to requiring a search of the plan and also a court order that may contradict the plan.

Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.): This Part does not adequately address conduit plans, adequate protection payments, allowance of deficiency claims, and cross-collateralization.

April 22-23, 2014 Page 449 of 596

Add a designation of the debtor's principal residence.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: Include a provision for pre-confirmation adequate protection payments.

Comment BK-2013-0001-0139—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): The form appears to take a substantive position by permitting debtors to pay directly and not through the trustee.

Comment BK-2013-0001-0154—David S. Yen (Attorney, Chicago, Ill.): Add a section to provide other options for treating secured claims (e.g., exchange of property). *See In re Pence*, 905 F.2d 1107 (7th Cir. 1990) (upholding plan provision providing for release of lien on debtor's home in exchange for transfer of commercial property).

Section 3.1 (maintenance and cure)

Comment BK-2013-0001-0010—Judge Kay Woods (Bankr. N.D. Ohio): Section 3.1 states that "all payments under this plan as to that collateral will cease and all claims as to that collateral will no longer be treated by the plan." This isn't the case, is it? Although payments for a *secured* claim as to that collateral will cease, shouldn't any deficiency claim be treated as an unsecured claim under the plan? Section 3.1 should reference § 3.5 (surrender of collateral) regarding the unsecured deficiency claim.

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- Section 3.1 is the biggest problem with the plan. The plan should not permit debtors to pay secured creditors directly. This will create problems for trustees who make conduit payments on mortgages and other secured debt. No chapter 13 trustee can operate his or her office without compensation. The drafters of the form need to talk to real persons involved in the administration of chapter 13 cases, rather than the non-professionals who are currently the decisionmakers.
- In keeping with *Espinosa*, do not let the claim control over contrary amounts listed on the plan.
- There are timing differences between when relief from the stay is granted and an order is entered and reflected in the trustee's computer records. Rather than cease payments under the plan as to the collateral, in the C.D. Cal. we address this by providing that the creditor will refund payments received after relief is granted.
- Section 3.1 needs an adequate protection provision for pre- and post-petition payments made through the chapter 13 trustee.
- Section 3.1 states that "[t]he allowed claim for any arrearage amount will be paid under the plan." This provision states that direct payments are not made under the plan. They are, and it is only who makes them that differs.

April 22-23, 2014 Page 450 of 596

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): I like that § 3.1 states that payments will stop if the stay is lifted. But this form, in effect, overrules *Espinosa*. The arrearage amount listed in the plan should control over a contrary claim.

The treatment of the regular monthly payment and the arrearage amount should be separated into two parts—one for maintenance of payments and one for the arrearage—because the arrearage should always be paid by the trustee. Otherwise, there will be many objections when debtors attempt to pay arrearages directly. Remove the choice for debtors to make their arrearage payments directly. Arrearage payments are uniformly made by the trustee.

The form provides no term for arrearage payments. Is the assumption that arrearages will be paid in months 1-60? What if they are paid in months 1-37?

What is meant by "Monthly plan payment on arrearage or other payment arrangement"? What "other payment arrangement" is contemplated? This language should be removed and any special provisions should be treated in Part 9.

Comment BK-2013-0001-0028—Michael D. Clark (Chapter 13 Trustee, C.D. Ill.): The monthly payment on the arrearage should not be set by the plan but by the trustee, after consideration of all claims filed.

Comment BK-2013-0001-0034—Dana Milby (Attorney, Wichita, Kan.): There is no provision for pre-confirmation adequate protection payments, no provision for paying the filing fee through the plan, and no provision for pre-confirmation ongoing mortgage payments. These omissions will delay confirmation and payments to creditors. In Kansas, we have extensive procedures for dealing with real property mortgages. The form does not allow a way to implement our procedures.

Comment BK-2013-0001-0038—William Bonney (Chapter 13 Trustee, E.D. Okla.): Section 3.1 includes the following language: ". . . and all claims as to that collateral will no longer be treated by the plan." This language is not needed and should be removed. Under § 1328, only claims provided for by the plan are discharged.

Comment BK-2013-0001-0038—William Bonney (Chapter 13 Trustee, E.D. Okla.): Section 3.1 does not address the problem of a post-petition arrearage claim that arises before the first plan payment is made.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): There are several problems with § 3.1.

Section 3.1 does not give enough space to list the name of the creditor and collateral. It does not take into account that cross-collateralization language in a loan may mean the collateral secures more than one debt.

Allowing the claim to control will invite motions to dismiss on feasibility grounds when the claim comes in higher than the amount stated in the plan.

Section 3.1 rather naively assumes that a mortgage payment with escrow will not change over the length of the plan. The last column becomes useless in a conduit district and throws off the calculations in Exhibit B, as the mortgage payment will change.

April 22-23, 2014 Page 451 of 596

The portion of § 3.1 regarding relief from the automatic stay ("all payments under this plan paragraph as to that collateral will cease and all claims as to that collateral will no longer be treated by the plan . .") conflicts with the Bankruptcy Code to the extent that the language means there is no unsecured deficiency claim after the stay is lifted and the collateral liquidated.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: There is no need to list the monthly plan payment for each secured claim on which a default is being cured. This amount may not be known due, for example, to litigation about fees and charges. The amount may also change, so that at best an estimated total arrearage is known. It is better to provide that the debtor will propose to pay the arrearage within a reasonable time period and otherwise give sufficient direction to the trustee regarding the order of payment. The amounts are not needed.

The language about stay relief is problematic. The debtor may not wish the payments to that creditor to cease after the stay is lifted (so as to effect a cure before the creditor can foreclose on the collateral). The debtor may also wish to pay other creditors whose claims are secured by the same collateral.

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.): The plan should control over the proof of claim. Otherwise, mortgage creditors will file an estimated claim and amend. There would be no finality to a confirmation order. This is unfair discrimination in favor of home mortgage creditors.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: This section should make clear that any deficiency shall be treated as an unsecured claim to be discharged upon completion of the plan.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): Section 3.1 is too elaborate.

There should be provision for adequate protection payments.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: The phrase "under the plan" in the text following the "None" checkbox is unnecessary and should be deleted.

The column on the far right headed "Estimated total payments by trustee" should be deleted.

The use of the passive voice in this section should be avoided.

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): The wording of this section suggests that interest is appropriate, although it is rarely paid on arrears. We recommend stylistic changes.

Comment BK-2013-0001-0066—Edward Claxton: The last sentence should be rephrased to read: "The final column represents payments disbursed by the trustee."

Comment BK-2013-0001-0071—Judge Dennis Montali (Bankr. N.D. Cal.): This section is confusing. The position of the "disbursed by" boxes should be changed.

April 22-23, 2014 Page 452 of 596

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.):

Add a start date for the current installment payments.

Use "estimated monthly plan payment" instead of "monthly plan payment." This gives debtors and trustees more flexibility.

Remove the option for debtors to pay directly. Conduit payments are more efficient and beneficial to debtors and help to fund trustee operations.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): We do not use conduit payments.

Remove the estimated total payments column. Delete the phrase "or other payment arrangement" in the prior column.

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): This section is confusing in two respects.

First, it provides that "debtor(s) will maintain" the contractual installment payments. But there is an option for the trustee to make disbursements. Clarify that the trustee's disbursements are subject to the terms of the contract.

Second, the language regarding the effects of stay relief ("and all claims as to that collateral will no longer be treated by the plan") should be altered to deal with deficiency claims.

Comment BK-2013-0001-0080—Judge John E. Waites (Bankr. D.S.C.): I agree with the NCBJ's comments.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: We do not use conduit payments.

Comment BK-2013-0001-0092—Judge Arthur S. Weissbrodt (Bankr. N.D. Cal.): Do not require the monthly plan payment on which a default is being cured. The amount may be in dispute or unknown. The amount may change throughout the life of the plan.

Debtors should be allowed to continue paying secured creditors even if relief from stay is granted.

Comment BK-2013-0001-0096—Nik Geranios: I agree with NACBA's comments.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Make conduit payments the default and allow district-by-district nonstandard provisions for direct payments by debtors.

Comment BK-2013-0001-0099—Chief Judge David S. Kennedy (Bankr. W.D. Tenn.), on behalf of four of five judges of the court: Debtors should not have the choice to make payments directly. Our trustees would object if debtors proposed to do so.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): Include a reference to Code § 1322(b)(5).

April 22-23, 2014 Page 453 of 596

Do not give debtors a choice as to whether the trustee or debtor will makes payments.

Include a space to indicate if the claim is secured by a lien on the debtor's principal residence.

Indicate lien priorities when multiple liens exist against the same collateral.

Eliminate the provisions about relief from the automatic stay.

Add provisions for adjustment of plan payment amounts due to interest rate and escrow changes.

Add provisions authorizing payment of postpetition claims under Rule 3002.1(c).

Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: Include a provision for preconfirmation adequate protection payments.

The language regarding relief from the automatic stay should be clarified so that only the secured portion of the claim will no longer be treated under the plan.

Comment BK-2013-0001-0121—Peter D. Schneider (Attorney, Philadelphia, Pa.), on behalf of Community Legal Services, Inc., of Philadelphia: The requirement to list the monthly plan amount applied to each secured claim is complex, unnecessary, and improperly limits a debtor's options. If plan payments are staged (e.g., to allow for payment of the filing fee in installments), the payment will vary and a single amount cannot be set out in the plan.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: This section is confusing.

Does it provide for payment of interest on arrears contrary to *Rake v. Wade*, 508 U.S. 464 (1993)?

Does it mandate conduit payments?

April 22-23, 2014

Does it effectuate relief from stay without motion or merely provide that payments cease upon the entry of an order for relief from stay?

Does it provide valuation of collateral, lien avoidance, or just interest rate adjustments?

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: Clarify the introductory language in § 3.1. State that, upon stay relief, Rule 3002.1 no longer applies. State that the monthly installment payment is dictated by the applicable note, subject to Rule 3002.1. State that any debt provided for in § 3.1 is not "outside the plan," regardless of whether it is paid through the trustee or debtor.

The amount of the current installment payments should be an estimate.

Add a column for the debtor to state whether encumbered real property is a principal residence. A lender could then determine if Rule 3002.1 applies.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: Add a column for pre- and post-confirmation arrearage

Delete the column labeled "monthly plan payment on arrearage." This will mislead creditors into the assumption that they will receive monthly distributions in that amount.

Add more space. There are too many columns with unnecessary information.

Page 454 of 596

Delete or alter the language regarding stay relief. Many debtors are seeking mortgage modifications, and if the modification remains pending after the stay is lifted, debtors benefit from continuing to pay arrears.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): If the trustee is the disbursing agent, then the first sentence should say "The Plan will maintain . . . "

Include the contract maturity date. This is necessary to determine whether Rule 3002.1(a) applies.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Clarify the lift stay language. Does it deny the secured creditor the right to participate with the unsecured class? Some lift stay orders are self executing, and the trustee would not know about the order in time to cease payments to the secured creditor.

The plan does not need to address what happens in the event of a stay lift. Some cases hold that the obligation to pay the balance owed on the allowed secured claim does not cease once the stay lifts.

The current installment payment information is not necessary. It is disclosed in Schedule J.

Remove the estimates. Amounts should be binding.

Do not give debtors a choice regarding whether to disburse arrears or mortgage payments directly.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Revise the treatment of maintenance and cure payments. Remove the estimates of arrearages.

Do not allow debtors the option of choosing the trustee to make current installment payments. We require the debtor to make these payments directly in our district.

Remove the provision for stopping payments when the stay is lifted. If the creditor wants to decline further payments, it can withdraw its claim.

Provide for allowed secured claims that the debtor schedules as unsecured.

Specify the date from which interest, when applicable, runs on the cure of arrears.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Include the option of listing the principal residence.

Include the option of giving the trustee authority to pay a notice of post-petition fees.

Section 3.1 provides for surrender of property but does not provide for allowance of a deficiency claim.

The Committee Note indicates that, upon termination of the stay, the provisions of Rule 3002.1 no longer apply. This language should be in the form.

Comment BK-2013-0001-0152—Cathleen Moran: The debtor often does not know the precise amount of the arrears.

Comment BK-2013-0001-0153—James M. Davis (Attorney, Nashville, Tenn.): Clarify that only allowed claims will be paid.

April 22-23, 2014 Page 455 of 596

Because the proof of claim will control over the amount listed in § 3.1, change the column heading for the installment payment to "estimated current installment payment."

Section 3.2 (request for valuation of security and claim modification)

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.): Why is it necessary to request that the court determine the value of property? Will there be an actual hearing? Shouldn't the plan simply say that the debtor's valuation of the property controls, unless objected to?

The reference to eligibility for discharge is misplaced in § 3.2.

Comment BK-2013-0001-0010—Judge Kay Woods (Bankr. N.D. Ohio): The third line of § 3.2 could be reworded more clearly to state "the value of the secured claim should be allowed as set forth below in the column . . . " (rather than "the value of the secured claim should be as stated below in the column . . . ").

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- This section should not give deference to claims by governmental units. Let governmental units object to confirmation just like other secured creditors are required to do. Using the term "interest" in § 3.2 looks like an attempt to get around the *Till* decision. The term "present value" should be used. *See* § 1325(a).
- In some jurisdictions, the court authorizes the chapter 13 trustee to disburse preconfirmation adequate protection payments to secured or lease creditors pursuant to § 1326(a)(1) if the debtor's plan so provides. Provision should be made for such adequate protection payments through the trustee by including a column titled "Monthly pre-confirmation adequate protection payment" with a subtitle indicating whether the debtor or trustee will make the payment.

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): Section 3.2 is trying to do too much. It should be simplified to provide for the secured portion of the claim and not deal with the ranking of the lien. The section needs information only as to the name of the creditor, the collateral, the amount of the secured claim and value of the collateral, the interest rate, and the estimated total payment. The rest of the information is useless. Is the trustee the police for the debtor and creditors on this section? The section also assumes that all secured creditor payments will be made over the life of the plan.

Section 3.2 should not include the discharge information. Eligibility for discharge should be moved to Part 1.

The columns for "estimated amount of creditor's claim" and "amount of claims senior to creditor's claim" will be erroneous in most cases. Neither is necessary to determine the secured claim or to tell the creditor how the collateral is being valued. These columns should be removed.

Comment BK-2013-0001-0016—Richard Siegel (Attorney, Atlanta, Ga.): The second sentence of § 3.2 refers to non-governmental secured claims "as to which a proof of

April 22-23, 2014 Page 456 of 596

claim has been filed in accordance with Bankruptcy Rule 3002." The verb tense is an issue, because the plan could be filed the day the petition is filed or shortly thereafter. Proofs of claim are filed later, and the present-tense wording limits the debtor to stating values for claims already filed. The language should be changed to include proofs of claim that can or will be filed.

Comment BK-2013-0001-0021—Judge Randall Dunn (Bankr. D. Or.): In the Committee Note to § 3.2, add a sentence like the one in the Committee Note to § 3.4 to the effect that service pursuant to Rule 7004 is required. Section 3.2's Committee Note does refer to Rule 3012, which in turn refers to Rule 7004, but it would be clearer simply to add a reference in the Committee Note to service under Rule 7004.

Perhaps the plan should include a model certification as to service, such as:

In order to assure proper service on all creditors identified in Section 3.2 or 3.4 of the plan, pursuant to FRBPs 3012, 4003(b), 9014, and 7004: (a) I listed on the mailing matrix such creditors, other than insured depository institutions, in care of a person or entity authorized to be served; AND (b) I served by certified mail, on ______ (date), copies of the plan on any insured depository institution(s) affected by Section 3.2 or 3.4 of the plan; AND (c) the following list separately identifies all such creditors served via matrix listing and such creditors served by certified mail, including the names and addresses of all such creditors served (NOTE: With respect to creditors served by matrix listing, the list of names and addresses is identical to that included in the mailing matrix.).

Comment BK-2013-0001-0028—Michael D. Clark (Chapter 13 Trustee, C.D. Ill.): The monthly payment to creditors should be left to the trustee to determine, or should simply be listed as an estimate only.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): Section 3.2 raises several concerns.

This section does not provide enough space to list the name of the creditor and collateral.

The columns labeled "estimated amount of creditor's claim" and "amount of secured claim" are duplicative. The estimated amount of the creditor's claim and the value of the collateral are listed in Schedule D. It would be better to substitute "amount proposed as security" in place of these two columns.

The last column will more than likely change during the confirmation process. Transcribing the amount in Exhibit B will cause arithmetic errors.

This section does not provide for cross-collateralization of claims.

Does the language regarding governmental claims mean that the bankruptcy court cannot rule on an objection to a governmental unit's claim?

This section speaks of "secured claims" instead of "allowed claims" as if to suggest that the claims objection process will go away.

Whether the debtor is eligible for a discharge should not be in this section. If this information must be in the form, it should be at the beginning.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: There are several problems with this section.

There is no reason to exclude governmental creditors from this provision.

April 22-23, 2014 Page 457 of 596

There is no need to list the monthly payment to the creditor, as the payment may vary and be an estimate.

A debtor may provide for a valuation proceeding after confirmation. There is no requirement that valuation occur before confirmation.

There is no place on the form for preconfirmation adequate protection payments to a secured creditor or (as many courts have allowed) postconfirmation adequate protection payments while attorney's fees are being paid.

The title and wording of § 3.2 are misleading and underinclusive, because they suggest that this section is only for secured claims for which the debtor seeks a valuation of collateral. But there is no other provision for payments to secured creditors under § 1325(a)(5) that do not involve valuation of collateral, cure of defaults, or "hanging paragraph" claims. Section 3.2 should be changed to include all other secured claims provided for by the plan (e.g., if the debtor proposes to pay an allowed secured claim according to the contract terms because it carries a favorable interest rate).

Delete the checkboxes regarding the debtor's eligibility for a discharge. That information is already provided in the petition.

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.):

Government creditors should not be excluded.

The term "interest rate" here and in § 3.1 should be changed to "present value rate" in light of the Supreme Court's decision in *Till*.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas:

We agree with Richard Siegel's comment regarding the verb tense.

The language regarding eligibility for a discharge is misplaced in this part. It should be in Part 1.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): Section 3.2 makes use of the plan mandatory in order to avoid or strip a lien and ends the practice in many districts of doing so by motion or adversary proceeding. This should be left to each district.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: We recommend several stylistic changes. The subject and verb do not agree in the phrase "[t]he debtor(s) request" unless it is a joint case.

The word "non-governmental" should be inserted before "secured claims listed below" in the first sentence and the rest of the sentence deleted.

In the second sentence of the first full paragraph, the phrase "in accordance with Bankruptcy Rule 3002" is unnecessary and should be deleted.

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.):

Reword the 8th column ("monthly payment to creditor") to indicate that the payment is estimated and an average that my change.

April 22-23, 2014 Page 458 of 596

Remove the discharge language. We agree with NACBA. Are fully secured non-PMSI claims addressed in this section?

Comment BK-2013-0001-0066—Edward Claxton: This section is confusing to everyone except a law professor. There is not enough space to enter numbers. The phrase "controlling amount of the claim" is confusing. The distinction between "value of collateral" and "amount of secured claim" is confusing. The second paragraph should read: "The portion of any allowed claim that exceeds the value of collateral will be treated as an unsecured claim under Part 5 of this plan."

The statement in (b) regarding discharge is an issue for the court, and not the debtor, to decide.

Comment BK-2013-0001-0071—Judge Dennis Montali (Bankr. N.D. Cal.):

What paragraph is not effective if the checkbox is not selected?

Eliminate the language stating that the provision will not be effective if the checkbox in Part 1 is not marked. It is not a realistic provision.

Add a provision showing the basis for the debtor's valuation.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): This provision is unclear.

It is unclear who serves the valuation request. If the plan is served by the trustee, how will he or she know where to serve?

It is unclear how value is determined.

The exception for governmental units creates confusion.

Add a checkbox for valuation requested separately by motion.

The phrase "estimated monthly payment to creditor" should be used.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): This section conflicts with Fourth Circuit caselaw. *See Cen-Pen v. Hanson*, 58 F.3d 89 (4th Cir. 1993); *In re Snow*, 270 B.R. 38 (D. Md. 2001) ("For a debtor to extinguish or modify a lien during the bankruptcy process, some affirmative step must be taken toward that end."). A free-standing motion should be required instead.

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): What happens if the debtor proposes a \$0 valuation but the secured creditor does not file a proof of claim?

There is a conflict between this section and the Committee Note to Rule 4003(d), which states that "[1]ien avoidance not governed by this rule requires an adversary proceeding."

There is not enough space to describe fully the collateral being valued.

Because the debtor does not sign the plan, the debtor has not authenticated his own opinion of value.

The service provisions under Rule 3012 should be referenced.

Comment BK-2013-0001-0085—Judge Roger Efremsky (Bankr. N.D. Cal.), on behalf of six judges of the bankruptcy court: We agree with Judge Montali's comments regarding valuation. A plan is not evidence.

April 22-23, 2014 Page 459 of 596

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: This section is contrary to Fourth Circuit law. (See comment by Mark D. Sammons.)

Remove the discharge eligibility language.

Comment BK-2013-0001-0092—Judge Arthur S. Weissbrodt (Bankr. N.D. Cal.): I support the option to permit the determination of allowed secured claim values as part of the chapter 13 plan confirmation process. This procedure is accepted in chapter 11 practice. There are no due process issues. The proposal will save significant amounts of time and money by eliminating unnecessary motions. We have used this procedure very successfully in the San Jose Division of the N.D. Cal.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: The term "value of the secured claims" is ambiguous. It could refer to the entire claim value or just the portion that is secured due to the amount of available collateral. The form appears to be using the latter meaning.

This section appears to take a position on whether the debtor can strip liens in a chapter 20 situation.

Comment BK-2013-0001-0098—Henry Hildebrand, III (Chapter 13 Trustee, M.D. Tenn.): Governmental proofs of claim should not be excluded from the valuation provision.

Comment BK-2013-0001-0100—Byron Meredith (Chapter 13 Trustee, S.D. Ga.): Governmental proofs of claim should not be excluded from the valuation provision.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): Government proofs of claim should not be excluded.

Add statutory references to §§ 506 and 1325(a)(5).

Remove the language regarding eligibility for discharge.

Reorient the information and add more space for responses.

Add a disclaimer that amounts are estimates that may fluctuate.

Comment BK-2013-0001-0111—Judge Elizabeth W. Magner (Bankr., E.D. La.): The language allowing a secured creditor's lien to "ride through" should be changed. It could be interpreted to permit enforcement of the entire debt, including the prepetition arrearage and postpetition installments, against the collateral.

Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: The column labeled "Amount of claims senior to creditor's claim" might be used improperly to determine priority of liens and bind creditors.

Revise the language regarding eligibility for discharge to state that debtors "contend" that they are eligible.

April 22-23, 2014 Page 460 of 596

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: This section is confusing. Does it provide for motions under *Till*?

The section's final provision regarding the timing of lien avoidance may contradict Fourth Circuit case law. *See Branigan v. Davis (In re Davis)*, 716 F.3d 331 (4th Cir. 2013).

Include templates for service of process.

Include an additional checkbox for evidence necessary for valuation and debt if lien avoidance will be included in the plan.

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: It is impractical to expect a trial on valuation issues at a chapter 13 confirmation hearing. This will lead to delays. Instead, include language indicating that the amount of the secured claim is an estimate, and that debtor may request valuation of the property and/or limitation of the amount of the secured claim through Rule 3012, 7001, or a claim objection.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: Governmental claims should not be excluded. It is actually less burdensome on governmental creditors if their claims are included, because it eliminates the need to amend a claim or defend a claim objection in every case with a proposed modification.

Too much unnecessary information is included.

Too much instructional language is included. It is too harsh to debtors and takes away the ability to advocate positions on behalf of creditors (e.g., it eliminates the possibility of surrendering collateral in satisfaction of the debt).

Remove the discharge eligibility provision.

Comment BK-2013-0001-0134—Marlene Martel (Attorney), on behalf of Ford Motor Credit Company: Use the phrase "amount of the secured claim" instead of "value of the secured claim." Valuation of the collateral will determine the amount of the secured claim.

Require more detail about the collateral (e.g., VIN, vehicle mileage).

Add a provision to indicate the basis of the valuation.

Alter the lien release and retention language. End sentence (b) after "... § 1328"—the plan cannot unilaterally terminate a lien, which is a state law issue.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): In light of proposed Rule 5009(d), the language regarding lien retention needs another option for debtors who do not pay the debt in full or receive a discharge under § 1328.

If the "none" box is checked for this section, the discharge boxes will not appear anywhere on the form. The discharge language avoids unnecessary litigation over dischargeability.

The discharge language does not account for joint cases in which one debtor is eligible for discharge but the other is not. Use separate lines for each debtor.

Comment BK-2013-0001-0139—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): The form appears to take a substantive position by permitting debtors to pay directly and not through the trustee.

April 22-23, 2014 Page 461 of 596

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Do not exclude government claims. I also agree with the comments of Judge Nugent and Mr. Siegel.

The lien termination and release language is a concern. Some of our courts hold that a lien does not have to be released when there is a non-filing co-debtor. Does the plan form overrule these decisions?

The discharge language belongs in Part 1.

Remove the column "amount of claims senior to creditor's claim." A plan cannot alter the order of priority.

Remove the column "estimated total of monthly payments." Tis will lead to arithmetic errors.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Remove this section. Do not permit valuation of collateral through the plan.

Comment BK-2013-0001-0147—Mark Bulovic (Attorney, Savannah, Ga.): Is the discharge provision subject to res judicata if the debtor is not entitled to a discharge?

Section 3.2 does not provide for secured creditors to retain the liens securing their claims. No plan using this form could be confirmed under § 1325(a)(5)(B)(i)(I) and (II).

Does the confirmation order constitute "otherwise ordered by the court"? If so, what is the significance of this provision?

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Section 3.2 makes no provision for long-term liens and long-term debt to survive discharge.

Include a provision for pre-confirmation adequate protection payments.

Comment BK-2013-0001-0153—James M. Davis (Attorney, Nashville, Tenn.): The verb tenses in the second sentence are confusing. No proofs of claim will have been filed at the time the plan is filed.

The reference to proofs of claim filed "in accordance with Rule 3002" is ambiguous, because (as the Committee Note indicates) proofs of claim may be filed under other rules (e.g., Rule 3004).

The reference to the "controlling amount of the claim" is ambiguous. I suggest revised language.

I suggest revised language for the second paragraph of this section dealing with the unsecured portion of the claim.

The lien retention provision in the final paragraph does not cover claims not secured by any value.

Comment BK-2013-0001-0158—Debra Miller (Chapter 13 Trustee, N.D. Ind.): I agree with NACBA and others about § 3.2.

First, the language appears to permit valuation only if the secured creditor has filed a proof of claim. Although a proof of claim may be required for distribution, the value of the

April 22-23, 2014 Page 462 of 596

secured portion is set at confirmation. Whether or not a proof of claim is filed, if the value of the secured claim is paid during the life of the bankruptcy, the lien is stripped and voided at discharge. The current language gives secured creditors an incentive not to file a proof of claim.

Second, governmental claims should not be excluded. Section 506 of the Code does not distinguish secured claims of governmental units.

See testimony of January 31, 2014.

Section 3.3 (secured claims excluded from 11 U.S.C. § 506)

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.): The boxes in § 3.3 that appear to allow direct payment should not be there. If the debtor proposes a direct payment, that should be a nonstandard provision in Part 9.

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- The final column in § 3.3 includes only payments disbursed by the trustee. This section should include the following language concerning other payments: "Claims paid under the exact terms of the contract with no cure of arrears may be maintained through payment made by debtor or trustee as is customary in the district. Claims that cure arrears or affect the interest rate paid on the claim are impaired and will be paid in accordance with controlling law, or as is customary in the district if no objection is asserted by the Trustee."
- Remove the option of the debtor(s) as disbursing agent, unless they are maintaining or curing default. Also, use "present value" instead of "interest."
 - Section 3.2 should be referenced in § 3.3.

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): Section 3.3 is an attempt to overrule *Espinosa*. A plan that leaves the proof of claim controlling will result in incorrect figures for every secured claim, putting the trustee in the position of doing everyone else's work. The plan should control, and the creditor should object if it disagrees with the amount.

Section 3.3 does not include a space for the term of the payments. A column should be added with the term of the monthly payment (e.g., months 1-24 or 5-56, etc.).

Section 3.3 should not permit direct payment by the debtor. The general rule is provided in Code § 1326(c) that payments are made through the trustee. Plans that propose otherwise are subject to feasibility concerns. The rare exception for direct payment can be included as a nonstandard provision in Part 9.

Comment BK-2013-0001-0028—Michael D. Clark (Chapter 13 Trustee, C.D. Ill.): The monthly plan payment to the creditor(s) should be determined by the trustee, or simply listed as an estimate only.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): There are several problems with § 3.3.

There is not enough space to list the name of the creditor and the collateral.

Allowing the claim to control will invite feasibility issues if the claim is filed after confirmation. For example, if the amount of a 910 claim listed in the plan is incorrect, the creditor should take enough interest in the process to file an objection to confirmation.

April 22-23, 2014 Page 463 of 596

The last column is an invitation to arithmetic errors.

This section refers to "secured claims" instead of "allowed claims" as if to suggest that the claims objection process is going away.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: The form should clarify that a debtor is not required to treat 910 claims in this section. The debtor may choose to deal with those claims in § 3.1. The language in the last paragraph before the listing of creditors should state "except as provided in § 3.1" or words to that effect.

NACBA objects to requiring that monthly payments be listed, because they will be estimates that may vary.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): This section is unclear.

It suggests incorrectly that the interest rate on a 910 day car claim cannot be crammed down.

Because the proof of claim controls, this section will require a claim objection and accomplish little.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: We recommend several stylistic changes.

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): The third column ("amount of claim") and fifth column ("monthly plan payment") should be estimates.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): The "monthly plan payment" heading is inconsistent with the other sections. It should be an estimate.

Do not allow debtors the option of making direct payments.

The total payment information is unnecessary. It will change throughout the case.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): Strike the word "estimated" from the total payments column.

In our district, the trustee and not the debtor makes payments on arrearages.

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): The debtor should not be allowed to modify the interest rate if the debtor (and not the trustee) is making payments.

Comment BK-2013-0001-0080—Judge John E. Waites (Bankr. D.S.C.): I agree with the NCBJ's comments.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: This section is contrary to Fourth Circuit law. (See comments by Mark D. Sammons.)

April 22-23, 2014 Page 464 of 596

Comment BK-2013-0001-0096—Nik Geranios: I agree with NACBA's comments.

Comment BK-2013-0001-0098—Henry Hildebrand, III (Chapter 13 Trustee, M.D. Tenn.): The form should include the option of soliciting a secured creditor's acceptance of the plan under Code § 1325(a)(5)(A). The form includes the two other statutory options for secured claims—surrender (§ 3.5) and payment of the present value in installments (§§ 3.2 and 3.3)—and include the third option of the secured creditor's acceptance of the plan.

Comment BK-2013-0001-0100—Byron Meredith (Chapter 13 Trustee, S.D. Ga.): Make clear that this section might apply to claims other than 910-day car claims (e.g., where the value of the collateral exceeds the value of the secured claim).

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): Do not allow debtors the option of making direct payments.

Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: § 3.3. should also include other secured claims excluded from § 506, such as short-term real property debts or co-signed debts.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: The number of valuation and lien avoidance provisions in the plan is confusing.

Include a template for service.

Comment BK-2013-0001-0123—Randall Smith: This section appears to overrule *Till* by giving the creditor unilateral control over the interest rate on a 910-day car loan. The creditor should have the burden of going forward with evidence that the debtor's proposed rate is inadequate.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: The "amount of claim" column should be an estimate.

The section should not limit claims to the two examples listed. Other claims could include sewer liens, secured tax claims that cannot be modified, homeowner's association liens, etc.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Include language regarding the claim holder's lien retention and release.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Remove the direct payment option. Delete the column "estimated total payments by trustee" or state that estimates are not binding.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): The plan form should address the debtor's option of paying the entire secured claim under the plan, without maintaining any direct payments to the creditor.

April 22-23, 2014

Page 465 of 596

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Section 3.3 makes no provision for payment of claims outside the plan by third parties.

Comment BK-2013-0001-0153—James M. Davis (Attorney, Nashville, Tenn.): Because the claim controls in this section, change the column heading to "estimated amount of claim."

Section 3.4 (lien avoidance)

Comment BK-2013-0001-0010—Judge Kay Woods (Bankr. N.D. Ohio): Section 3.4 should include language (after the bolded statement that the section will be effective only if the box in Part 1 is checked) that Exhibit A must also be completed. There is no basis for lien avoidance without Exhibit A.

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): Exhibit A should state that the amount of the lien avoided will be treated as an unsecured claim.

Comment BK-2013-0001-0019—Gene Killian (Attorney): Motions under Code § 522(f) and generally all other lien avoidance vehicles should be included in the plan form.

Comment BK-2013-0001-0021—Judge Randall Dunn (Bankr. D. Or.): How will lien avoidance work with title companies? The plan does not appear to have a place to include real property descriptions. Are separate orders contemplated?

Comment BK-2013-0001-0028—Michael D. Clark (Chapter 13 Trustee, C.D. Ill.): The first sentence should clarify that (i) either Code § 522(b) or state exemption schemes would apply and (ii) any monthly plan payment stemming from liens not avoided will be determined by the trustee (or listed as an estimate).

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): The form provides for lien avoidance upon confirmation without a motion or adversary proceeding. This provision creates noticing difficulties, because most plans are filed the day the petition is filed. It is difficult to find out who is the proper person for service before claims are filed. This provision will not reduce adversary proceedings or motions to avoid judicial liens. It will encourage pro se filings that will not accomplish removal of judicial liens. It will also lead to more litigation over whether the creditor received proper notice under Rule 4003.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: NACBA supports this provision of the form. But the form should not dictate, as the last sentence of § 3.4 does ["The amount, if any, of the judicial lien or security interest that is not avoided will be paid in full as a secured claim under the plan."], how the portion of the lien that is not avoided should be treated. The debtor may choose not to provide for the remaining allowed secured claim, or to cure and maintain

April 22-23, 2014 Page 466 of 596

payments rather than paying in full. Qualifying language, such as "unless otherwise provided in the plan" should be included.

The Code does not require lien avoidance to occur before confirmation. The form should be flexible and accommodate the possibility of post-confirmation valuation.

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.):

The term "interest rate" is irrelevant if the debtor is paying for an item not subject to valuation under the plan. Only when the debt is paid through the trustee and the interest rate is being modified would a rate other than the contract rate apply.

The debtor's option to make direct payments should be clearly limited to where the contract payment is being maintained.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: We recommend several stylistic changes.

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): The "monthly plan payment" column should be an estimate.

Comment BK-2013-0001-0066—Edward Claxton: The second sentence of § 3.4 ("A judicial lien or security interest securing a claim listed below will be avoided to the extent that it impairs such exemptions upon entry of the order confirming the plan") is not correct. Avoidance is permanent only upon discharge.

Comment BK-2013-0001-0071—Judge Dennis Montali (Bankr. N.D. Cal.): Add a provision showing the basis for the debtor's valuation.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.):

Pursuing lien avoidance through a plan and not by motion could affect court statistics used to determine budgets, staffing, etc.

The monthly plan payment heading is inconsistent with other sections. It should be an estimate.

Add a checkbox for lien avoidance pursued by separate motion.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): A free-standing motion should be required instead.

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): This section should require authenticated evidence in support of a lien avoidance request.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: This section is contrary to Fourth Circuit law. (See comments by Mark D. Sammons.)

April 22-23, 2014 Page 467 of 596

Comment BK-2013-0001-0093—Gary D. Gray (Office of the Chief Counsel, Internal Revenue Service): Clarify that § 3.4 does not apply to statutory liens that cannot be voided under Code § 522(f). Tax liens are specifically protected under § 522(c)(2), but some debtors may try improperly to avoid tax liens.

A similar clarification should be added to the Committee Note of Rule 4003(d).

Comment BK-2013-0001-0096—Nik Geranios: I agree with NACBA's comments.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Clarify the second sentence ("A judicial lien or security interest . . . ") by stating "Such judicial lien or security interest . . . "

Comment BK-2013-0001-0098—Henry Hildebrand, III (Chapter 13 Trustee, M.D. Tenn.): I recommend including a provision that sets a deadline by which a secured creditor must file an amended deficiency claim.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): Eliminate this section. Nothing in the Code or Bankruptcy Rules authorizes challenging the validity of a lien absent a motion in compliance with Rule 4003(d) and 9014. For real property, title companies will not rely on a chapter 13 plan in lieu of a court order.

Comment BK-2013-0001-0121—Peter D. Schneider (Attorney, Philadelphia, Pa.), on behalf of Community Legal Services, Inc., of Philadelphia: We support the provision allowing a debtor to avoid a lien through the plan confirmation process.

The form appears to limit a debtor's ability to avoid a lien by motion.

The debtor may choose not to provide for the unavoided portion of the lien, or to cure and maintain payments. The form does not include these options.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: If § 3.4 is for lien avoidance under Code § 522(f), it should say so.

The requirement for Exhibit A in this section should be retained.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Clarify whether the plan or the proof of claim controls with respect to interest rates.

Add language regarding the claim holder's lien retention and release.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Local courts should be able to remove the lien avoidance provisions from the plan form.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Remove this section. Lien avoidance should not be done through a plan.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

April 22-23, 2014 Page 468 of 596

Section 3.4 should provide that lien avoidance occurs at discharge so that the creditor does not have to refile its lien if the case is dismissed after confirmation.

Comment BK-2013-0001-0153—James M. Davis (Attorney, Nashville, Tenn.): The final column is confusing. Revise it to read "estimated total amount to be paid by the trustee of secured claim."

Comment BK-2013-0001-0154—David S. Yen (Attorney, Chicago, Ill.): The last column heading should be "estimated total of monthly payments" and not "estimated total amount of secured claim."

Section 3.5 (surrender of collateral)

Comment BK-2013-0001-0010—Judge Kay Woods (Bankr. N.D. Ohio): In our court, the plan and confirmation order require the creditor with surrendered collateral to file a proof of claim within 90 days after the confirmation order is entered. If the collateral has not been liquidated by that time, the creditor is required to file a good-faith estimate of the deficiency, if any, and then amend when the exact amount is known. An untimely unsecured claim is barred. Add a similar 90-day period requirement or a blank for when the proof of claim is required to be filed.

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- Section 3.5 is well stated and addresses all the issues with surrender.
- Can't debtors surrender collateral in full satisfaction of debt? Section 3.5 should include a paragraph relating to the value of the property surrendered and a checkbox indicating surrender "in full satisfaction of debt."

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): Section 3.5 does not state a surrender value. Does this mean the plan assumes that surrender is in full satisfaction of the claim? Language should be added to the effect that surrender is in full satisfaction of the secured claim.

Comment BK-2013-0001-0016—Richard Siegel (Attorney, Atlanta, Ga.): Section 3.5's second sentence states: "The debtor(s) consent to termination of the stay under 11 U.S.C. § 362(a) and § 1301 with respect to the collateral upon confirmation of the plan." Section 1301 involves the co-debtor stay. While the debtor may consent to relief from the co-debtor stay, his interest and that of the co-debtor are often not the same. If their interests differ, the debtor cannot dispose of both interests in a plan provision. Creditors may read this language to permit them to proceed, wrongfully, to collection or foreclosure on the ground that the co-debtor stay is terminated along with the automatic stay. Clarification is needed. The second sentence should read: "The debtor(s) consent to termination of the stay under 11 U.S.C. Section 362(a) and Section 1301 with respect to the collateral upon confirmation of the plan, but nothing herein shall be deemed to terminate the 11 U.S.C. Section 1301 stay upon confirmation." The creditor would then be required to seek co-debtor relief by motion if it desires. It may be more clear simply to

April 22-23, 2014 Page 469 of 596

eliminate the reference to § 1301 in the clause referring to the debtor's consent to termination of the stay.

A new provision should be added to the plan so that parties are relieved from compliance with Rule 3002.1 if relief from the automatic stay is granted.

Comment BK-2013-0001-0066—Edward Claxton: Section 3.5 does not provide for surrender of collateral in full satisfaction of the debt. Also, how can the stay be lifted upon confirmation without a required motion from a party in interest under § 362(d)? In addition, the debtor cannot consent to co-debtor stay relief under § 1301.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: A debtor cannot waive a co-debtor's stay rights under § 1301. We agree with Richard Siegel's comment on this point.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges:

The last sentence ("Any allowed unsecured claim resulting from the disposition of the collateral will be treated in Part 5 below") should be deleted, because it takes a position on a disputed question of substantive law. Courts differ on when and how a secured creditor obtains an unsecured deficiency claim after disposing of collateral surrendered in a plan. See In re Sneijder, 407 B.R. 46 (S.D.N.Y. 2009).

We recommend several stylistic changes.

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): The debtor should be allowed to surrender in full satisfaction or provide other treatment when surrendering.

Comment BK-2013-0001-0068—Nancy Whaley (Chapter 13 Trustee, N.D. Ga.): Permit the debtor to pay secured claims other than 910-day car claims in full (e.g., co-signed debts).

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.):

Elimination of stay-relief motions will reduce revenues from filing fees for those motions. It will also affect statistics used to determine court budgets, staffing, etc.

State courts in our district often require an order from the court granting relief from stay.

Comment BK-2013-0001-0081—James Draper: The second sentence of § 3.5 ("The debtor(s) consent to termination of the stay . . . with respect to the collateral upon confirmation . . .") is ambiguous. It could be read to mean that the consent (and not the termination of the stay) is effective upon confirmation, thereby implying that some further action by the creditor is necessary for stay termination. This should be clarified.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): Provide space for the debtor to indicate the alleged value of collateral being surrendered.

April 22-23, 2014 Page 470 of 596

Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: The language of this section is ambiguous. It should state that the stay and co-debtor stay are lifted as to the collateral without further notice or court order upon confirmation.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: Provide that "the stay shall terminate" instead of "[t]he debtor(s) consent to termination"

Comment BK-2013-0001-0123—Randall Smith: Section 3.5 implies that the debtor has an affirmative obligation to deliver to the creditor surrendered property.

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: Section 3.5 should state directly that upon confirmation, the debtor and co-debtor stays are lifted as to a lender's *in rem* rights. Language to this effect is included in plans in some districts (e.g., N.D. Tex., S.D. Ohio (Cincinnati); S.D. Fla.).

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: Provide detail about when a deficiency claim must be filed.

Add clearer language (e.g., "without further order of the court") in the stay termination provision.

Comment BK-2013-0001-0134—Marlene Martel (Attorney), on behalf of Ford Motor Credit Company:

Add a provision with a deadline by which the debtor must surrender the collateral. Require termination of the stay, and not merely "consent" to termination, upon surrender.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Address how an allowed unsecured claim from disposition of collateral will be calculated and treated.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Add a column for the value of collateral being surrendered.

Surrender at confirmation is at odds with our local rule, which provides for surrender after a specified number of days after the plan is filed. Our rule cuts costs and the creditor does not need to await a potentially drawn out confirmation process.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Clarify the term "surrender." It could mean surrender of possession or surrender to the processes of nonbankruptcy law.

The consent to lifting the stay is ineffective without a court order. The plan should effect a termination of the stays under §§ 362(a) and 1301.

April 22-23, 2014 Page 471 of 596

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: The plan form does not address the timing of an amended deficiency claim for surrendered collateral.

Section 3.5 does not address stay relief as to the trustee.

How does the debtor consent to the termination of the co-debtor stay when the co-debtor is not participating in the case?

Part 4: Treatment of Trustee's Fees and Administrative and Other Priority Claims
Part 4 (general)

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): There is no provision for the court fee to be paid in installments. This will cause the arithmetic to be wrong in Exhibit B, which should not be in the plan.

Comment BK-2013-0001-0080—Judge John E. Waites (Bankr. D.S.C.): Allow the debtor to pay postpetition priority claims directly.

Comment BK-2013-0001-0093—Gary D. Gray (Office of the Chief Counsel, Internal Revenue Service): Section 4.1 refers to payment of allowed priority claims "without interest." Clarify that this refers to postpetition interest and not prepetition interest.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: The combination of §§ 4.1, 4.4, and 4.5 does not make sense. The provision for interest appears to be incorrect. Under Code § 1322(b)(10), the debtor should not pay interest unless there is enough disposable income to pay all allowed claims in full.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: The sections in Part 4 should be combined.

The form should state that, with each disbursement, trustee's fees and attorney's fees will be paid in accordance with local rules

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Retitle this Part "Treatment of Trustee's Fees, Administrative Claims, and Other Priority Claims."

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Section 4.1 should read "All allowed priority claims <u>provided for under this plan</u> other than those treated in § 4.5 will be paid in full <u>by the Trustee</u> without interest, unless otherwise stated."

Section 4.1 should require the plan to provide for a priority claim in order for the claim to be paid. This is necessary for that claim to be discharged under § 1328. Otherwise, debtors may emerge from bankruptcy owing post-petition interest and penalties that were not discharged.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Combine and shorten §§ 4.1 through 4.4.

April 22-23, 2014 Page 472 of 596

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Section 4.2 (trustee's fees)

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.): Section 4.2 should simply disclose the 10% maximum trustee's fee. Also, some trustees are paid on disbursements and some on receipts.

Comment BK-2013-0001-0008—Jon Waage (Chapter 13 Trustee, M.D. Fla.): Section 4.2 should state that the maximum fee is 10% and that, in the event the trustee does not retain the full 10%, any portion not retained will be paid to unsecured creditors pro rata. In the alternative, the fee calculation should be eliminated and language included to state instead that the fee percentage is fixed periodically by the U.S. Trustee.

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

Section 4.2 should not require debtors to estimate the trustee's fee. These fees are set by the Department of Justice and fluctuate. This section will increase trustees' objections to confirmation. Eliminate the section or, in the alternative, include a calculation set at the maximum possible fee (10%) under 28 U.S.C. § 586(e)(1)(B)(i).

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): The 10% maximum fee should be disclosed instead.

Comment BK-2013-0001-0014—K. Michael Fitzgerald (Chapter 13 Trustee, W.D. Wash.): Section 4.2 is problematic and unnecessary. It will generate objections by trustees to avoid being bound by the fee calculated in the plan.

Comment BK-2013-0001-0019—Gene Killian (Attorney): Some chapter 13 trustees require an allocation of 11% for the trustee's fees on the ground that this provides a margin for future increases in the fee during the plan term. This is a hardship for debtors and leaves less money for payment toward other plan items. Marginal cases can be rendered unfeasible by the trustees' demands. Limit trustee payments to what is actually paid as of the time of filing.

Comment BK-2013-0001-0035—Robert G. Drummond (Chapter 13 Trustee, D. Mont.): This provision is unnecessary. The amount of the trustee's fee changes periodically throughout the year. This is especially true in a state like Montana with a low volume of chapter 13 cases.

Comment BK-2013-0001-0038—William Bonney (Chapter 13 Trustee, E.D. Okla.): Trustee's fees are not an administrative expense in a chapter 13 case, because compensation for the trustee is not allowed under § 330, and therefore not allowed under § 503(b), and therefore not a priority claim under § 507. If the Advisory Committee believes the plan must contain a statement about trustee's fees, then the statement should read: "These fees shall be paid in

April 22-23, 2014 Page 473 of 596

accordance with 28 U.S.C. § 586" or "These fees shall be paid in accordance with 28 U.S.C. § 586 and are estimated to be \$."

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): The amount of the trustee's fee fluctuates. This provision would not bind the trustee in any event. It should be removed from the form.

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.): Eliminate the estimated total trustee fees.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: This section serves no purpose.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: We recommend stylistic changes.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): This information is unnecessary.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): This section serves no purpose.

Comment BK-2013-0001-0080—Judge John E. Waites (Bankr. D.S.C.): Trustee fees may vary over the course of a case.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: The trustee's fee may vary.

Comment BK-2013-0001-0108—Mary Kate Kelley-Scheidler (Attorney, Minn.): Add more detail about trustee's fees. Are they paid at once or does the trustee get a portion of each payment?

Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: The trustee's fee varies.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: Omit the arithmetic from this section. It is unnecessary.

The trustee's fee varies.

April 22-23, 2014

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): The term "plan payments" is vague. Fees should be stated as a percentage of "all payments received under the plan." That is the language used in the trustee fee statute. 28 U.S.C. § 586(e)(2).

The phrase "plan term" is not defined.

Page 474 of 596

Comment BK-2013-0001-0139—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): Remove the total trustee's fee. It is unnecessary.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): The trustee's fee should be set at 10%. That way, fluctuations in the actual fee will not require modification.

Remove the estimate of total fees.

Add a provision for the trustee to charge and collect noticing fees. This is allowed in a number of our courts.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Remove the estimate of trustee's fees.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Comment BK-2013-0001-0152—Cathleen Moran: Remove the estimate of trustee's fees. The amount varies.

Section 4.3 (attorney's fees)

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.): Section 4.3 does not tell the trustee the exact amount to pay the debtor's attorney.

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): Section 4.3 should require the debtor's attorney to recap all fees to assure they do not exceed the allowable amount. This section should include information on the total fees to be paid, the amount paid separately by the debtor, and the amount to be paid in the plan.

Comment BK-2013-0001-0024—Rod Danielson (Chapter 13 Trustee, C.D. Cal.), on behalf of the five chapter 13 trustees of the C.D. Cal.: Section 4.3 does not give direction about the treatment of outstanding attorney's fees. Jurisdictions differ on this issue. It would be helpful if the plan informed debtors, creditors, and trustees about how the attorney is to be paid. The "order of distribution" in Part 7 is not sufficiently specific.

Comment BK-2013-0001-0034—Dana Milby (Attorney, Wichita, Kan.): Section 4.3 does not allow the trustee to know what exact amount is to be paid to the attorney.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): This section takes into account only "no look" fees. It does not disclose what the attorney was paid prior to filing and assumes that the attorney will be paid the fee listed in the plan based on the bankruptcy judge's final determination of the fee.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: This section must clearly indicate who is

April 22-23, 2014 Page 475 of 596

paying what fee. All parties in interest are entitled to know the entire amount of the fees to be paid, and the total should be disclosed along with the amount already paid, if any.

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): Add more detail regarding fee arrangements.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): Add more detail regarding fee arrangements. An attorney may elect to submit a fee application instead of a flat fee.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: Add more detail regarding attorney's fees. This provision conflicts with our local rules.

Comment BK-2013-0001-0098—Henry Hildebrand, III (Chapter 13 Trustee, M.D. Tenn.): Add more detail regarding attorney's fees. Section 1326(b) recognizes that attorney's fees can be paid "before or at the time of each payment to creditors under the plan."

Comment BK-2013-0001-0108—Mary Kate Kelley-Scheidler (Attorney, Minn.): Add more detail regarding attorney's fees. The form does not indicate when they will be paid.

Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: Add more detail regarding attorney's fees. Include the amounts and increments.

Comment BK-2013-0001-0119—Kathleen Horne (Attorney, Savannah, Ga.), on behalf of the Bankruptcy Lawyers Advisory Committee of the Bankruptcy Court for the S.D. Ga.: Add more detail regarding attorney's fees.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: The attorney's fee should be a precise number unless otherwise ordered by the court.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Add more detail regarding attorney's fees.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Add more detail regarding attorney's fees. Pre-petition fees should also be disclosed.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Remove the estimate of attorney's fees.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

April 22-23, 2014 Page 476 of 596

Section 4.4 (other priority claims)

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.):

There is no section addressing non-assigned domestic support obligations. Include a specific reference to those obligations in § 4.4 (on going; arrearage; payment through the trustee; payment through an existing state court order).

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- What does "Other Priority Claims" mean? Is the term limited to domestic support obligation claims assigned to a governmental unit and paid less than the full amount? Is this section meant to enumerate claims referenced in § 4.1? Maybe the term "other" is misleading. Combine this section with § 4.1.
 - Section 4.4 should reference § 4.2.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): Section 4.4 requires the debtor to "estimate" the amount of priority claims and the interest. This is meaningless. If an allowed claim is a priority claim, it will be paid. The only reason for this section is to permit the amount to be transcribed to Exhibit B, which is going to be inaccurate in any event because of the number of "estimates" in that exhibit.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: As with secured claims, the monthly payments on priority claims should not be required to be listed. They often cannot be known at confirmation. There is no need to list priority creditors that will be paid in full, because they are already listed in the debtor's schedules.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: For §§ 4.4 and 4.5, debtors should be required to state affirmatively whether they do or do not have a DSO and whether it is ongoing or only an arrearage.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: We recommend stylistic changes.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): Add the following to the sentence regarding priority claims: "... however a filed proof of claim prevails over the estimate, unless a timely objection is upheld."

Comment BK-2013-0001-0092—Judge Arthur S. Weissbrodt (Bankr. N.D. Cal.): Do not require the monthly plan payment on priority claims. The amount may be in dispute or unknown. The amount may change throughout the life of the plan.

Comment BK-2013-0001-0096—Nik Geranios: I agree with NACBA's comments.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Add language as to whether the plan or claim controls.

April 22-23, 2014 Page 477 of 596

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): Add space for the debtor to explain the statutory basis for the priority specified.

Comment BK-2013-0001-0121—Peter D. Schneider (Attorney, Philadelphia, Pa.), on behalf of Community Legal Services, Inc., of Philadelphia: See comments on § 3.1. The form should not list the monthly amount applied to each priority claim.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: Omit the requirement to state the basis for priority treatment. This is the creditor's burden.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Clarify whether the plan or proof of claim controls with respect to interest rates.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Do not include the names of holders of other priority claims, the basis for priority treatment, and the estimates of amounts. The schedules and proofs of claim already indicate this information.

The interest rate does not need to be repeated multiple times. Clarify that this is postconfirmation interest.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Section 4.5 (domestic support obligations assigned to a governmental unit)

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

Part 4 does not contain any provision for existing domestic support orders to continue. This is not in compliance with the domestic support provisions of the Code.

Comment BK-2013-0001-0034—Dana Milby (Attorney, Wichita, Kan.): Part 4 has no provision for existing support orders to continue.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: The title of § 4.5 is underinclusive. The words "or owed" should be inserted after "assigned," because Code § 507(a)(1)(B) covers domestic support obligations owed to a governmental unit, whether or not they have been assigned.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): This section is unclear. The amount of claim to be paid should be an estimate. The column for estimated total amount of payments is unnecessary.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): Strike "estimated" from the total payments column.

April 22-23, 2014 Page 478 of 596

Comment BK-2013-0001-0096—Nik Geranios: I agree with NACBA's comments.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: The instructions should state that this provision cannot be used unless the debtor dedicates all disposable income for the applicable commitment period and the plan length of five years was selected in § 2.5.

Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: This section should address all domestic support obligations, not only those assigned to a governmental unit and paid less than the full amount.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Clarify whether the plan or proof of claim controls with respect to interest rates.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Simplify this section.

Part 5: Treatment of Nonpriority Unsecured Claims

Section 5.1 (maintenance and cure)

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- Section 5.1 is confusing in conjunction with § 5.2. If a nonpriority unsecured claim is listed to be paid separately, then the claim is being classified and should be entered in § 5.2. The two sections should be combined.
 - Section 5.1 should be removed.
- The only claims that could possibly be included in § 5.1 are student loans and cosigner debts. Both should be treated in § 5.2.

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): Section 5.1 starts with the wrong presumption by allowing a class of unsecured claims whose last payment is beyond the term of the plan to be paid in full, plus arrearages. Reorganize and renumber the order of the treatment of nonpriority, not separately classified, unsecured creditors paid on a pro rata basis. Current § 5.1 should be § 5.3.

Comment BK-2013-0001-0014—K. Michael Fitzgerald (Chapter 13 Trustee, W.D. Wash.): Part 5 provides for cure and maintenance of payments on unsecured claims where the last payment is due after the final plan payment. This is an unusual provision with unknown effects. If this is intended to address student loans, it would be better to do so as a special class of nonpriority unsecured debt.

Comment BK-2013-0001-0038—William Bonney (Chapter 13 Trustee, E.D. Okla.): Section 5.1 does not address the problem of a post-petition arrearage claim that arises before the first plan payment is made.

April 22-23, 2014 Page 479 of 596

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.): Remove this section. This should be a non-standard provision.

Comment BK-2013-0001-0057—Susan Saidian (Attorney, Wichita, Kan.): Section 5.1 seems to permit the debtor to prefer unsecured creditors without justification for different treatment. Does it address something other than special class creditors or executory contracts in default?

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: The heading of the second column from the right should be changed to insert "estimated" at the beginning, so that the heading reads: "Estimated Aamount of arrearage to be paid."

Comment BK-2013-0001-0066—Edward Claxton: Aren't "amount of arrearage to be paid" and "estimated total payments by trustee" the same amount?

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): It is not clear what debts are included in this section.

The current installment payment and amount of arrearage to be paid should be estimates.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Make conduit payments the default and allow district-by-district nonstandard provisions for direct payments by debtors.

Add language as to whether the plan or claim controls.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): More detail is needed. Add a column for the total amount of the debt as well as the amount to be paid. If this section is intended for student loan debts, include a disclosure regarding the payment of interest and the basis for discriminating against other unsecured claims.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: This entire Part is too complicated and may be unnecessary.

If this section is for student loans, then say so.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: The first sentence should read "The Plan will maintain "

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Clarify whether the plan or proof of claim controls with respect to interest rates.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Combine §§ 5.1 and 5.2. Do not give debtors the option of paying directly.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Remove the estimates of claims.

60

Do not give debtors the option of having the trustee make current installment payments. Include a checkbox for debtors to make payments directly to cure arrears on a § 1322(b)(5) claim.

Include a column for the date the payments under the claim are due to expire.

Comment BK-2013-0001-0153—James M. Davis (Attorney, Nashville, Tenn.): To be consistent with other parts of the plan, this section should provide for "estimated" installment payments and arrearages, with the proof of claim controlling.

Section 5.2 (separately classified nonpriority unsecured claims)

Comment BK-2013-0001-0038—William Bonney (Chapter 13 Trustee, E.D. Okla.): Most separately classified unsecured claims are for co-signed consumer debt. There should be a monthly payment amount for this paragraph.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): This is the one section of a plan where the claim should control, but the form does not do so. This section will generate guesses or estimates that will need to be changed once the claim is filed.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: The word "allowed" should be shifted in the text below the "None" checkbox. The revised text should read: "The <u>allowed</u> nonpriority unsecured allowed claims listed below are separately classified and will be treated as follows:"

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): There should be an election for payment by debtor, co-debtor, or trustee.

The amount of claim to be paid should be an estimate.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys:

Clarify the label for this section.

This section should not suggest that the payment of interest on nondischargeable claims is appropriate, except when Code § 1322(b)(10) has been satisfied.

The form should not take a position suggesting that separate classification of unsecured claims is permitted. It is a minority position.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): Add a statutory reference to the basis for separate classification.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: This section requires a level of legal knowledge that inexperienced practitioners or pro se debtors are unlikely to have.

61

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: The "amount of claim to be paid" should be an estimate.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Clarify whether the plan or proof of claim controls with respect to interest rates.

Comment BK-2013-0001-0139—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): Remove this section. It should be a nonstandard provision in Part 9.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Eliminate this section. Separate classification is rare and should be a nonstandard provision.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Section 5.3 (nonpriority unsecured claims)

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.): The third box under § 5.3 suggest that unsecured creditors cannot be paid until all other claims are paid. There is an argument that unsecured creditors could be paid pro rata with other claims, particular when it is a best-interest-of-creditors-test issue. The asset may be dissipated or the debtor may dismiss the case if one waits until the end to pay.

It is helpful to know how the debtor arrived at the best-interest-of-creditors number. Our local plan form provides a place for that calculation.

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- Is the first option under § 5.3 intended to represent projected disposable income? If it is, then clearly say so.
- The third option will encourage debtors to propose the acceleration of secured debt payments and thereby consume funds that would otherwise be disbursed to unsecured claims.
- The form does not make clear that the debtor must pay the greater of the estate's liquidation value or the projected disposable income on a means test. The language concerning liquidation value suggests that liquidation value will be disbursed exclusively to nonpriority unsecured claims, which is incorrect. The concept includes payment to priority creditors. The third option should be eliminated and the form clarified to state that priority claims will share in the disbursement of the estate's liquidation.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): There are a number of problems with § 5.3.

This section will generate arithmetic errors.

Many debtors' attorneys will propose a low amount or percentage, because after confirmation the debtor will be required to pay back that exact amount or percentage.

This section will generate more modified plans after the claims bar date passes, because plans will not pay the amounts or percentages stated as a result of the number of estimates called for by the form. If the goal is to give unsecured creditors a quick look at how much they will be

April 22-23, 2014 Page 482 of 596

paid on their claims, the form should include instead a range of dividend percentages. We use this type of scale in the E.D. Tenn., where there are checkboxes for 0%, 1-5%, 6-20%, etc.

Comment BK-2013-0001-0043—Dan Melchi (Attorney, Alpharetta, Ga.): The third checkbox should be removed. An unsecured creditor should be told in unambiguous terms what that creditor's claim will receive under the plan. The language of the third checkbox will be chosen by every debtor, because (unlike the first two checkboxes) it does not require the debtor to commit to an amount certain. For example, a debtor who selects the third checkbox could later decide to cure a post-petition arrearage on a mortgage with funds that would otherwise go to unsecured creditors. If the first or second checkbox had been selected, the debtor would have been required to modify the plan.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: While we approve the inclusion of a "best interest of creditors" number, we agree with the comment submitted by the NACTT that the concept of liquidation includes payment to priority creditors.

The debtor should explain how the best interest number was derived.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges:

Delete the phrase "up to the full amount of the claims" from the sentence immediately below the heading "Nonpriority unsecured claims." The phrase is unnecessary. The revised sentence should read: "Allowed nonpriority unsecured claims that are not separately classified will be paid, pro rata, up to the full amount of the claims, as follows:"

The phrase "under chapter 7" in the last paragraph should be changed to "in a chapter 7 case" so that the sentence reads: "If the estate of the debtor(s) were liquidated <u>under in a chapter 7 case</u>, nonpriority unsecured claims would be paid"

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): Only one choice should be selected.

Comment BK-2013-0001-0066—Edward Claxton: Is there a need for the phrase "up to the full amount of the claim"? Could more be paid?

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): Only one choice should be selected.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): This section conflicts with our local practice. The debtor should pay a sum certain. That way, if a creditor does not file a timely proof of claim, other creditors (and not the debtor) will benefit.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: These options are confusing and misleading. This section is contrary to the "pot plan" practice in D. Md.

The liquidation test will be difficult to determine when the plan is filed. The value of assets and exemptions may not be determined until confirmation.

April 22-23, 2014 Page 483 of 596

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys:

Does the form contemplate that debtors will be able to object to proofs of claim after confirmation? Is a claim res judicata after confirmation?

What is the process for challenging timely government claims filed after confirmation?

Comment BK-2013-0001-0098—Henry Hildebrand, III (Chapter 13 Trustee, M.D. Tenn.): Modify § 5.3 to include a "base" amount in the last option—that is, the monthly plan payment multiplied by the applicable commitment period or plan length. This number is in § 2.4.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): This section does not make sense. Debtors will not enter correct amounts under these options. The calculations are estimates.

Comment BK-2013-0001-0117—Chief Judge Mary P. Gorman (Bankr. C.D. Ill.): I am troubled by § 5.3. Offering the option to specify a percentage distribution raises concerns under Seventh Circuit case law.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: If this section is intended to state what creditors would get in a chapter 7 liquidation scenario, it should be labeled more clearly.

The section requires calculations that are too complicated.

Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.): A dollar amount should not be listed if another overriding option is being selected.

There will be arithmetic errors and inconsistent calculations. What results when, e.g., the estimated plan payment (§ 2.4) conflicts with the plan payment (§ 2.1).

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: Delete "nonpriority" from the liquidation analysis sentence. Liquidation includes priority unsecured claims.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Move the liquidation value paragraph to the beginning of this section. Include a sentence identifying the property from which the liquidation value is derived.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Simplify this section. A liquidation analysis is unnecessary. Unlike chapter 11, there is no requirement that the debtor provide a disclosure statement regarding the plan.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

April 22-23, 2014 Page 484 of 596

Comment BK-2013-0001-0153—James M. Davis (Attorney, Nashville, Tenn.): Delete the language in the first checkbox that states "unless a greater amount is required under another checked option." Otherwise, there is an implication that other options would operate as caps when multiple boxes are checked.

Section 5.4 (interest)

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- Section 5.4 is useful.
- The contents of this section are not required. Why include this but not other optional items?
- Be more specific in requesting the interest on allowed unsecured claims election by reference to solvent estates.

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): Is this section about paying interest on unsecured claims or about paying present value of the claims? Change this section to a present value calculation and change the word "interest" to "annual discount rate."

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.): This section is not required by the Code. No debtor proposes interest even when the liquidation test would require 100%. Remove this section.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): Interest is not generally paid on unsecured claims. This section will create confusion.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys:

This section suggests that interest will be paid in some circumstances but does not discuss § 1322(b)(10).

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: Include another option for "other."

It is not necessary to estimate the total interest.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): This section should be combined with § 5.3. See comments on § 5.3.

Part 6: Executory Contracts and Unexpired Leases

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): I like the presumption that a contract is rejected unless specifically assumed. This keeps the debtor and debtor's attorney from falling into a trap early in the case.

65

Comment BK-2013-0001-0024—Rod Danielson (Chapter 13 Trustee, C.D. Cal.), on behalf of the five chapter 13 trustees of the C.D. Cal.: This provision would automatically void all telephone, home rent, and auto rental agreements unless specifically assumed and articulated. Debtors—particularly pro se debtors—may not intend this outcome. Consider reversing the presumption, so that all executory contracts and unexpired leases are assumed unless specifically rejected.

Comment BK-2013-0001-0038—William Bonney (Chapter 13 Trustee, E.D. Okla.): Part 6 does not address the problem of a post-petition arrearage claim that arises before the first plan payment is made.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): Allowing the claim to control in this section will invite feasibility issues after confirmation.

This section also allows the trustee to pay "lease to own" payments through the plan. Pro se debtors may propose to have the trustee cure their rent-to-own furniture through the plan as opposed to getting legal advice to reject the contract for overpriced furniture.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: The default should not be rejection of leases and executory contracts. The form should allow the debtor neither to assume nor reject a lease, as many courts have recognized "ride through" in that situation.

If the debtor does assume, the plan must provide for prompt cure of defaults in addition to current installments.

Part 6 should simply list (i) any executory contracts or leases to be rejected and (ii) any to be assumed, along with the treatment to be given to the assumed contract or lease.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: We prefer our form plan, which opts for the opposite presumption (i.e., executory contracts and unexpired leases are assumed unless rejected). We feared that debtors would unwittingly reject their apartment lease, cell phone lease, automobile lease, gym memberships, etc.

Comment BK-2013-0001-0053—Judge Meredith Jury (Bankr. C.D. Cal.): Use "cure" or "prompt cure"—the words of § 365(b)—rather than "arrearage." Unlike an arrearage, a cure cannot be spread over an entire plan term in most jurisdictions.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): Reverse the presumption regarding rejection. In the alternative, leave this section blank.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: We recommend stylistic changes.

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): We agree with NACBA's comments.

66

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): The column for estimated total payments by trustee is unnecessary.

The current installment payment and amount of arrearage to be paid should be estimates.

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): Clarify that the debtor (and not the trustee) assumes the lease.

The automatic stay of §§ 362(a) and 1301 should terminate upon confirmation if the debtor rejects.

For assumed leases, the automatic stay should expire upon maturity to allow a secured creditor to take possession.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: Permitting cure over time on executory contracts, unless consensual, is contrary to the Code.

Comment BK-2013-0001-0096—Nik Geranios: I agree with NACBA's comments.

Comment BK-2013-0001-0098—Henry Hildebrand, III (Chapter 13 Trustee, M.D. Tenn.): Because Code § 365 requires that, upon assumption, cure of any default must be made "promptly," this Part should require the debtor to identify how the cure will be effected (e.g., pro rata with priority creditors, monthly payments in a cure class, etc.).

Comment BK-2013-0001-0119—Kathleen Horne (Attorney, Savannah, Ga.), on behalf of the Bankruptcy Lawyers Advisory Committee of the Bankruptcy Court for the S.D. Ga.: We are troubled by the choice of rejection as the default.

Comment BK-2013-0001-0121—Peter D. Schneider (Attorney, Philadelphia, Pa.), on behalf of Community Legal Services, Inc., of Philadelphia: The form does not give debtors the choice of neither assuming nor rejecting unexpired leases or executory contracts.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: As drafted, this Part suggests that lease arrearages may be cured over time, which is contrary to the Code.

Comment BK-2013-0001-0123—Randall Smith: Some courts permit a third option for leases and executory contracts: "ride through," in which debtors keep possession and continue making payments.

Comment BK-2013-0001-0134—Marlene Martel (Attorney), on behalf of Ford Motor Credit Company: Add language addressing post-assumption liabilities (e.g., lease-end charges).

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Debtors should not have the option of paying the arrearage cure directly. If only the current payment will be paid directly, then say so.

April 22-23, 2014

Remove the estimates.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Remove this Part. There is a third option in addition to assumption and rejection—"ride through."

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Include a provision for the monthly payment on the arrearage claim.

Part 7: Order of Distribution of Trustee Payments

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.): This part is big trouble. Debtors should not be allowed to choose the order of payments willy nilly, and that is what will occur. This part will nearly always draw an objection.

Comment BK-2013-0001-0008—Jon Waage (Chapter 13 Trustee, M.D. Fla.): Part 7 and Exhibit B will create cases that are impossible to administer. It will create chaos if each debtor controls the order of distributions.

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- This part is a good idea because it lets everyone know where to look for the order of payment.
- This part should be left blank and allow local procedures to determine the order of payment. The Code does not dictate the order of payment in a chapter 13 case, only which creditors must be paid.
- The Code provision requiring secured creditors to be paid in equal monthly payments should not result in those creditors being paid before attorneys get paid. If attorneys are not paid within the first year it would have a negative impact on case filings. Leave this part blank and delete Exhibit B.
- Add the following language: "Local Jurisdictions may have a default order of distributions that controls. If you are not sure, check with your jurisdiction."
- Allow each district to determine a uniform distribution order and permit the debtor to propose a different order in Part 9.
- My current trustee software program is customized for local procedures. This plan would require a separate database for new cases filed under the plan. My software is not capable of performing monthly disbursements under different distribution schemes. It would require running two different databases over 5 years until my current cases are closed.
- Exhibit B will cause confusion and discrepancies. I anticipate the need to object because it will be wrong.

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): Debtors' attorneys will not practice if they have to wait for all secured creditors to be paid. They will also have an incentive to seek direct payment to secured creditors so that attorney's fees can be paid earlier. This plan form is extremely trustee averse and the treatment of attorney's fees will cause the debtor bar to construct plans to serve their own interests and avoid paying trustee

April 22-23, 2014 Page 488 of 596

fees in the process. Administrative priority claims should be paid after the trustee's fees but before secured creditors.

If the distribution is monthly, Part 7 should be labeled "Monthly Order of Distribution of Trustee Payments." If not, using the word "monthly" to describe secured claim payments is confusing. Either remove the word monthly from § 7.1(b) or add the word monthly to the title of this part.

Comment BK-2013-0001-0014—K. Michael Fitzgerald (Chapter 13 Trustee, W.D. Wash.): This part is troublesome. It will lead to objections, impede the trustee's ability to administer plans efficiently, and delay confirmation.

Exhibit B is not helpful to a trustee's analysis of a plan. The summary will create problems. Because previous parts of the plan call for estimates, there is no value in debtors' further summarizing those estimates. Perhaps this exhibit is intended to serve as a feasibility calculation, but it will create more arithmetic mistakes.

Comment BK-2013-0001-0024—Rod Danielson (Chapter 13 Trustee, C.D. Cal.), on behalf of the five chapter 13 trustees of the C.D. Cal.: Part 7 could be subject to misuse or unlawful ordering of distributions in violation of the Code. This part is also unclear. Is the trustee to pay the first claim listed in full before any funds are disbursed to the second? This part will draw objections from creditors and trustees. Delete this part or replace it with a statement that distributions will be made in accordance with bankruptcy law.

Comment BK-2013-0001-0025—Robert Martinez (Attorney): Part 7 lists secured creditors as being paid behind the trustee's administrative claim. Does this payment priority apply to both adequate protection payments and the equal payments required by § 1325(a)(5)(B)(ii)? My concern arises from *In re DeSardi*, 340 B.R. 790 (Bankr. S.D. Tex. 2006), which holds that "equal payments" does not mean equal from the time the plan commences but only once equal payments start to be made. The plan should make clear that § 1325(a)(5)(B)(ii) applies to all payments on secured debt.

Comment BK-2013-0001-0028—Michael D. Clark (Chapter 13 Trustee, C.D. Ill.): The order of payments should be determined by the Code and not the debtor.

Comment BK-2013-0001-0034—Dana Milby (Attorney, Wichita, Kan.): Part 7 is overcomplicated and does not comply with the priority required by the Code.

Comment BK-2013-0001-0035—Robert G. Drummond (Chapter 13 Trustee, D. Mont.): The order of distributions suggests that secured creditors will be paid first. If the debtor's attorney is not paid up front, many attorneys will discontinue their chapter 13 practice.

Comment BK-2013-0001-0038—William Bonney (Chapter 13 Trustee, E.D. Okla.): Part 7 should be left blank. In the alternative, the order should be listed as follows:

- a. Trustee's fees.
- b. Monthly post-petition ongoing payments on claims provided for under § 3.1.
- c. Monthly payments on claims provided for under §§ 3.2, 3.3, 3.4, and 6.1.
- d. Monthly post- petition ongoing payments on claims provided for under § 5.1.

69

- e. Monthly payments on claims provided for under § 5.2.
- f. Monthly payments on the pre-petition arrearage portion of claims provided for under §§ 3.1 and 6.1.
- g. Monthly payments on the post-petition arrearage portion of claims provided for under §§ 3.1, 5.1 and 6.1.
 - h. Pro rata payments on the arrearage portion of claims provided for under § 5.1.
 - i. Pro rata payments on claims provided for under § 4.3.
 - j. Pro rata payments on claims provided for under § 4.4.
 - k. Pro rata payments on claims provided for under § 4.5.
- 1. Pro rata payments on claims provided for under § 5.3 and any interest thereon under § 5.4.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): The trustee distributes funds in accordance with Code § 507(a). Debtors' attorneys or pro se debtors should not be able to tell the trustee the order in which to distribute funds.

If the debtor were to propose paying a credit card debt first, or if the debtor's attorney were to propose paying attorney's fees first, Part 7 will conflict with § 5.3, which provides that unsecured creditors will be paid pro rata.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: Part 7 should not dictate that secured claims are to be paid second. The debtor should be allowed to propose any order of payments permitted by the Code, including, among many other possibilities: (i) payment of administrative expenses, such as attorney's fees, before other claims, as provided by Code § 1326(b)(1); (ii) payment of domestic support obligations before secured claims; (iii) payment of adequate protection on secured claims; and (iv) payments to cure a default on an assumed unexpired lease.

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.): Leave this section blank.

Comment BK-2013-0001-0049—Anne V. Kealing (Attorney, Edina, MN): Leave this section blank.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas:

Delete this section.

The form should include some provision discussing how filing fees are paid (e.g., directly by the debtor in installments, in a lump sum, through the plan, etc.).

Comment BK-2013-0001-0054—David Kestner: Part 7's order of distribution should not delay payments on counsel fees and DSO payments to benefit secured creditors.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): Leave this section blank.

April 22-23, 2014

Page 490 of 596

Comment BK-2013-0001-0061—Juliane Lore (Attorney, Mont.): Remove this section.

Comment BK-2013-0001-0062—Keith Welch (Attorney, Shreveport, La.): We agree with NACBA's comments.

Comment BK-2013-0001-0066—Edward Claxton: The trustee, and not the debtor, should determine these payments.

Comment BK-2013-0001-0068—Nancy Whaley (Chapter 13 Trustee, N.D. Ga.): Leave this section blank.

Comment BK-2013-0001-0071—Judge Dennis Montali (Bankr. N.D. Cal.):

Remove this section.

It appears to make a conduit plan mandatory.

Leaving the rest of the order of distribution blank invites mischief.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): Remove this section.

Payment of all secured claims prior to attorney's fees will harm chapter 13 practice. Code § 1322(b)(4) allows debtors to make payments on unsecured claims such as attorney's fees concurrently with payments on secured claims. The order of distribution should be governed by local practice.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): This section conflicts with the Code.

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): The payment order should be more flexible. Claims are often paid concurrently on a pro rata basis.

The section should specify whether the trustee's fees include solely the trustee's fees (and not administrative and priority claims).

Comment BK-2013-0001-0080—Judge John E. Waites (Bankr. D.S.C.): Prohibiting distribution of attorney's fees in initial payments by the trustee will discourage representation.

Domestic support obligation arrearage payments should be first in the priority of distribution.

Comment BK-2013-0001-0085—Judge Roger Efremsky (Bankr. N.D. Cal.), on behalf of six judges of the bankruptcy court: This part should not be left for debtors to fill out.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: This part should not be left to debtors to complete.

April 22-23, 2014 Page 491 of 596

Comment BK-2013-0001-0092—Judge Arthur S. Weissbrodt (Bankr. N.D. Cal.): Debtors should be permitted to propose the order of payments. The form should not take a position on the order of payment.

Comment BK-2013-0001-0096—Nik Geranios: I agree with NACBA's comments.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys:

This part should be more specific. Do not leave it to debtors. Include a default order of distribution that controls absent nonstandard provisions.

Comment BK-2013-0001-0098—Henry Hildebrand, III (Chapter 13 Trustee, M.D. Tenn.): Remove this Part. Instruct debtors that distributions will be made by the trustee.

Comment BK-2013-0001-0099—Chief Judge David S. Kennedy (Bankr. W.D. Tenn.), on behalf of four of five judges of the court: Debtors should not be able to decide the order of distribution without reference to the Code.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): Debtors should not be able to decide the order of distribution. This is governed by the Code.

Comment BK-2013-0001-0109—Devin Derham-Burk (Chapter 13 Trustee, N.D. Cal. (San Jose Division)): Debtors should not be able to decide the order of distribution.

Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: The debtor should propose the order of distribution. Do not designate the trustee's fee as a payment, because it is not a claim.

Comment BK-2013-0001-0117—Chief Judge Mary P. Gorman (Bankr. C.D. Ill.): This Part does not account for Code § 1322(b)(4), which permits concurrent payments on secured and unsecured claims.

Comment BK-2013-0001-0119—Kathleen Horne (Attorney, Savannah, Ga.), on behalf of the Bankruptcy Lawyers Advisory Committee of the Bankruptcy Court for the S.D. Ga.: This Part will lead to litigation.

Comment BK-2013-0001-0121—Peter D. Schneider (Attorney, Philadelphia, Pa.), on behalf of Community Legal Services, Inc., of Philadelphia: This Part improperly limits the debtor's ability to direct the application of payments. An aggrieved party can always object.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: This Part is unnecessary and will require multiple plan amendments.

Comment BK-2013-0001-0123—Randall Smith: This order of payments is not required by the Code and contravenes local practice in my district.

April 22-23, 2014 Page 492 of 596

Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.): The Code, and not the debtor, controls the order of distributions.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: The language "monthly payments on secured claims" is confusing. Does it mean payment on arrearages before distributions for administrative expenses? Or does it mean adequate protection payments?

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): This Part should be titled "Order of Distribution of Payments Made by Trustee."

The form should propose a default order of distributions.

Comment BK-2013-0001-0137—Phillip Oliver (Attorney, Mont.): The form appears to provide that attorneys will no longer be paid ahead of other claims. That will be a problem if the debtor fails to complete the plan.

Comment BK-2013-0001-0139—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): Leave this Part blank for the debtor to complete.

Comment BK-2013-0001-0140—Penelope Souhrada: Debtors and trustees should determine the order in which claims are paid.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Remove Part 7. Debtors should not be able to determine their own order of distribution.

The phrase "monthly payments on secured claims" is unclear. Does it include mortgage arrears that are not entitled to priority under § 1325(a)(5)?

Part 7 appears to overrule *In re DeSardi*, 340 B.R. 790 (Bankr. S.D. Tex. 2006), on the question of attorney's fee payments.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Remove this part. Distributions should be left to the trustee.

Comment BK-2013-0001-0146—Greg Arnove: The order of payment listed in this Part is not required by Code § 1324(b)(4). The debtor should be able to propose concurrent payments of different categories of claims.

Comment BK-2013-0001-0147—Mark Bulovic (Attorney, Savannah, Ga.): The order of distribution effectively subordinates attorney's fees.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Comment BK-2013-0001-0150—Judge Jennie D. Latta (Bankr. W.D. Tenn.): The trustee and not the debtor should set the order of distribution.

73

Part 8: Vesting of Property of the Estate

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: Our plan provides for vesting at discharge but gives the debtor the choice to vest at confirmation or some other time. Putting "at confirmation" as the first option will lead debtors to believe it is the best option, even though for most debtors it is not.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): Debtors should not be able to choose when property of the estate revests. It will create confusion. If property does not revest at confirmation, the trustee bears the risk of loss if the property is not insured.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: Debtors should not be able to choose when property of the state revests. This is contrary to our local rules.

If property revests at confirmation, debtors may liquidate it without court approval. Upon a conversion to chapter 7, this would prevent a chapter 7 trustee from liquidating assets.

Comment BK-2013-0001-0093—Gary D. Gray (Office of the Chief Counsel, Internal Revenue Service): Do not give debtors the option to retain property in the bankruptcy estate by merely checking a box. Under Code § 1327(b), revesting at confirmation is the default rule, and a court should determine if the debtor has a valid reason for changing the default. We are concerned that debtors will elect to keep property in the bankruptcy estate so that the automatic stay immunizes them from postpetition creditors, including the IRS.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: This part (or the Bankruptcy Rules) should include a default provision for revesting of property of the estate that clarifies what revesting means. Does revesting mean all new wages are the debtor's money?

Comment BK-2013-0001-0099—Chief Judge David S. Kennedy (Bankr. W.D. Tenn.), on behalf of four of five judges of the court: Debtors should not have discretion as to when property of the estate will revest. In our district, property revests only upon discharge, and our trustees would object to a plan that proposed otherwise.

Comment BK-2013-0001-0121—Peter D. Schneider (Attorney, Philadelphia, Pa.), on behalf of Community Legal Services, Inc., of Philadelphia: The debtor should be able to check more than one box. The debtor may wish to revest certain property upon confirmation and other property upon the close of the case.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: The vesting provisions are a trap for the unwary. It raises questions about the applicability of the automatic stay.

April 22-23, 2014 Page 494 of 596

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: Change the revesting language from "closing of the case" to "discharge." Case closing is an administrative act by the clerk that has nothing to do with vesting.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): The time at which property revests should be left to local control.

The word "applicable" is misspelled.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Part 8 should be eliminated or made optional. We revest when plan payments are completed. The plan form is contrary to that.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: There should be a preferred, uniform provision for vesting of property of the estate.

Comment BK-2013-0001-0154—David S. Yen (Attorney, Chicago, Ill.): Allow for vesting of property of the estate in an entity other than the debtor, as permitted in Code § 1322(b)(9).

Part 9: Nonstandard Plan Provisions

Comment BK-2013-0001-0006—Jan Hamilton (Chapter 13 Trustee, D. Kan.): In Kansas, we have extensive procedures for dealing with real property mortgages. Would we be able to plug all of our existing form plan language into the non-standard provisions part?

One place for a hodgepodge of non-standard provisions seems counter to the apparent goals of a national plan.

Comment BK-2013-0001-0049—Anne V. Kealing (Attorney, Edina, MN): There should be space for non-standard provisions at the end of each part of the plan, as with the Kansas model plan. This approach will avoid ambiguities that occur when all provisions are thrown together in one place.

Comment BK-2013-0001-0051—Chief Judge Robert E. Nugent (K. Kan.), on behalf of the bankruptcy judges of the District of Kansas: In our plan, non-standard provisions are included at the end of each discrete part. This is more clear for creditors and other parties in interest. At a minimum, the debtor should indicate which section of the form is being altered.

Also, does Part 9 permit a district to require that all debtors include language regarding conduit mortgage payments?

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: We recommend stylistic changes.

Comment BK-2013-0001-0071—Judge Dennis Montali (Bankr. N.D. Cal.): I doubt that failure to check the box in Part 1 will invalidate this section.

April 22-23, 2014

Page 495 of 596

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: This Part is necessary for counsel to make adjustments and revisions consistent with local practice.

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: Neither Part 9 nor proposed Rule 3015 contains any limitation on what may be included as a nonstandard provision. We are concerned that courts or trustees may encourage the use of provisions that undermine the uniformity sought by adoption of an Official Form.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: How many provisions can be added or modified in this section?

Part 10: Signatures

Comment BK-2013-0001-0012—Annette Crawford (Chapter 13 Trustee, M.D. La.): Debtors should have to sign the plan whether or not they have counsel. They can plead ignorance if they do not have to sign. The debtor's signature also protects the debtor's attorney.

Comment BK-2013-0001-0024—Rod Danielson (Chapter 13 Trustee, C.D. Cal.), on behalf of the five chapter 13 trustees of the C.D. Cal.: The debtor's signature should be mandatory. Otherwise, the plan will have no evidentiary value. Signing the plan shows that the debtor has knowledge of the treatment of creditors' claims. It is also protects the debtor's counsel.

Where the plan is being used to impair or avoid a lien or to value collateral, there will be no evidence to support the valuation of collateral. Having the debtor sign under penalty of perjury will ensure that the court has the evidence it needs to confirm the plan.

Comment BK-2013-0001-0035—Robert G. Drummond (Chapter 13 Trustee, D. Mont.): In addition to the signature lines, it would be helpful to require an email address as well.

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): The debtors should be required to sign the plan.

Comment BK-2013-0001-0041—Robert E. Hyman (Chapter 13 Trustee, E.D. Va.): All debtors should be required to sign the plan. When debtors question the amount of their plan payments, I can refer them to their signature on the plan. Most attorneys will want their clients to sign the plan as protection from a debtor's later denial of knowledge of the plan's terms.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: We recommend using the phrase "attorney for the debtor(s)" rather than "debtor's attorney" and "debtors' attorney." The attorney's certification should be rephrased to reflect stylistic changes.

April 22-23, 2014 Page 496 of 596

Comment BK-2013-0001-0071—Judge Dennis Montali (Bankr. N.D. Cal.): The debtors should be required to sign the plan.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): The debtors should be required to sign the plan. The signatures should be under penalty of perjury.

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): The debtors should be required to sign the plan. The signatures should be under penalty of perjury.

The debtor's attorney (or the debtor, if pro se) should certify that any nonstandard provisions do not violate the Code or Bankruptcy Rules.

Comment BK-2013-0001-0087—Anonymous: The debtors should be required to sign the plan.

Comment BK-2013-0001-0090—Nancy Spencer Grigsby (Chapter 13 Trustee, Bankr. D. Md.) on behalf of the standing trustees for the D. Md.: The debtors should be required to sign the plan. This is required by our local rules.

Comment BK-2013-0001-0100—Byron Meredith (Chapter 13 Trustee, S.D. Ga.): The debtors should be required to sign the plan.

Add a certification in the signature box that the amounts and values listed in the plan are grounded in fact and accurate to the best of the debtor's knowledge.

Comment BK-2013-0001-0132—Tracy Updike (Attorney, Pa.), on behalf of the M.D. Pa. Bankruptcy Bar Ass'n: The debtors should be required to sign the plan. They should verify its correctness and their intent to comply with it. Not having their signature deprives the plan of evidentiary value and exposes the attorney to liability.

Comment BK-2013-0001-0134—Marlene Martel (Attorney), on behalf of Ford Motor Credit Company: The debtors should be required to sign the plan. The debtors should certify under penalty of perjury that the contents are true and correct.

Comment BK-2013-0001-0136—Rick A. Yarnall (Chapter 13 Trustee, D. Nev.): Include space for the signer's address and telephone number as required by Rule 9011(a).

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

Exhibits A and B

Comment BK-2013-0001-0039—Cynthia T. Lawson (Attorney, Knoxville, Tenn.): Exhibit B serves no apparent purpose other than a quick review by an interested party who does not want to review the debtor's entire filing. The numbers and sums in Exhibit B will be estimates, and there are bound to be arithmetic errors. Also, plan payments may be a lot more than the bottom number in Exhibit B (as is the case in the E.D. Tenn.).

April 22-23, 2014 Page 497 of 596

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: Lines (c) and (e) of Exhibit A should more closely track the statutory language.

Line (c) should read: "Amount of exemption debtor could claim if there were no liens on the property."

Line (e) should read: "Value the debtor's interest in the property would have in the absence of any liens."

Comment BK-2013-0001-0047—Keith A. Rodriguez (Chapter 13 Trustee, W.D. La.): No party in interest needs this information to be laid out in an exhibit. Remove Exhibit B.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges:

For Exhibit A, it should be made clear that additional pages should be submitted if the debtor seeks to avoid multiple liens. Also, the form does not provide for a single lien on multiple properties. We recommend stylistic changes.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): Exhibit B is unnecessary and burdensome. Ongoing payments and arrearage payments on secured claims should not be combined. Allowed secured claims and secured claims not subject to § 506 should be paid at the same level.

Comment BK-2013-0001-0080—Judge John E. Waites (Bankr. D.S.C.): Provide for avoidance of multiple liens on the same collateral.

Consider circumstances of nondebtor co-owners' interests in property subject to judicial lien avoidance.

Comment BK-2013-0001-0122—Lisa Tancredi (Attorney, Md.), on behalf of Md. State Bar Ass'n, Consumer Bankruptcy Section, and Bankruptcy Bar Ass'n for the D. Md.: In Exhibit A, part c of the calculation should be replaced with "Value of available or applicable exemptions." Exemptions do not need to be claimed.

Omit Exhibit B. It is too complicated and unlikely to be completed properly.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): Exhibit B is not necessary or helpful.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Remove both exhibits. They add complexity to the preparation of a plan.

Comment BK-2013-0001-0149—Cynthia R. Wyrick (Attorney, Tenn.), on behalf of the Tennessee Bar Ass'n: See comments of Cynthia T. Lawson.

April 22-23, 2014

Page 498 of 596

Comments on the Amended Rules

Rule 2002

Comment BK-2013-0001-0027—Shmuel Klein (Attorney): Rule 2002 and Rule 3015(f) should require the same 28-day notice to debtor and debtor's counsel.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: NACBA supports setting a deadline in Rule 2002 for objections to confirmation of a chapter 13 plan. In some districts, objections can be raised for the first time at the confirmation hearing, which causes delay. The proposed deadline allows sufficient time for the debtor to respond to or attempt to resolve objections.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): The judges of the D. Md. believe that 21 days' notice is sufficient for a hearing on confirmation.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Stylistic suggestion.

Comment BK-2013-0001-0129—Beverly M. Burden (Chapter 13 Trustee, E.D. Ky.): See comment under Rule 3015.

Rule 3002

Comment BK-2013-0001-0004—David Peterson: The amendment to Rule 3002(a) adds the sentence: "A lien that secures a claim against the debtor is not void due only to the failure of any entity to file a proof of claim." This is substantive and not procedural. It is an incomplete statement of the law and may create confusion in chapter 11 cases. Under § 1141, a lien can be modified by a plan in chapter 11, the secured debt can be discharged, and the collateral revested in the debtor free and clear of any liens or claims except as stated in the plan or confirmation order. Perhaps the amendment means that the mere failure to file the claim (as opposed to plan confirmation and § 1141) does not void the lien, but the verbiage may create confusion without further elaboration, even though it apparently seeks to restate the substance of § 506(d).

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

- Shortening the claims filing period will be helpful, but mortgage servicers will ask for, and probably be granted, additional time to file their claims and attachments.
- The amended rule would give mortgage creditors longer than the filing period under the current rule to file a proof of claim with supporting documents. The rule does not provide for sanctions for non-compliance by mortgage creditors. Delay with respect to supporting documents will mean delay in confirmation and

April 22-23, 2014 Page 499 of 596

delay in commencement of payment to creditors. In addition, the trustee will need to check for claims twice as often—once at the bar date, and then again for supporting documents.

Comment BK-2013-0001-0014—K. Michael Fitzgerald (Chapter 13 Trustee, W.D. Wash.): It is important to require secured creditors to file a proof of claim. The 60-day deadline, however, is too optimistic and may be burdensome. A 90-day deadline is more realistic.

Comment BK-2013-0001-0018—Richard Thompson (Attorney): Secured creditors should not have to file a proof of claim in order to have an allowed claim as proposed in amended Rule 3002(a). This goes against hundreds of years of bankruptcy practice. It also raises a due process issue, because the secured creditor is placed in an untenable position—either file a claim and submit to the court's jurisdiction, or risk having the lien stripped or avoided without the opportunity to dispute it.

Comment BK-2013-0001-0024—Rod Danielson (Chapter 13 Trustee, C.D. Cal.), on behalf of the five chapter 13 trustees of the C.D. Cal.: The proposed 60-day filing deadline in amended Rule 3002(c) for nongovernmental claims may complicate the scheduling of § 341 meetings in some jurisdictions. The proposed amendment is intended to permit the confirmation hearing to occur after the claims filing period. In jurisdictions where the § 341 meeting is held on the same day as the confirmation hearing, however, that will be difficult to arrange, because Rule 2003(a) requires the § 341 meeting in chapter 13 cases to be held earlier than the proposed bar date (less than 21 and not more than 50 days after the order for relief). Rule 2003(a) should be amended either to allow a later date for the § 341 meeting (e.g., within 75 days of the date the petition is filed) or to retain the current timing for the meeting but include "unless otherwise ordered by the court" or similar words.

Comment BK-2013-0001-0027—Shmuel Klein (Attorney): Rule 3002 must identify whether the secured claim will be paid inside or outside the plan. Tax creditors must identify which portion of the claim is secured, general unsecured, or priority, or else the claim will be treated as unsecured. All claims must include actual true copies of supporting documentation and not summaries. There must be some safe-guarding against "robo-signed" proofs of claims.

To allow Rules 3012, 3015, 4003, and 5009 to have real effect, the treatment of the filed claim would shift the presumption of the claim stated in 3002 and allow treatment under the plan different from the amount stated in the claim.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: NACBA supports requiring a secured creditor to file a proof of claim. We also support shortening the deadline for filing proofs of claim.

We question why mortgage creditors are given additional time to file supporting documents for the claim. The proof of claim should not be filed unless those documents have been reviewed beforehand.

The last sentence of the Committee Note should be clarified. It could be read to raise a negative implication that objections to proofs of claim other than those covered by subdivision

April 22-23, 2014 Page 500 of 596

(c)(7)(B) [the provision for mortgage claims] cannot be filed after confirmation. The Bankruptcy Rules do not contain a deadline for objecting to proofs of claim, and in many cases confirmation does not preclude objections to a proof of claim after confirmation.

Comment BK-2013-0001-0055—Larry Foyle (Attorney, Tampa, Fla.): I am concerned about the sea change with respect to secured creditors. Making the filing of a claim mandatory and subject to a 60-day bar date is a new concept. Whatever happened to liens passing through bankruptcy?

Comment BK-2013-0001-0056—Scott Ford, on behalf of the Bankruptcy Clerks Advisory Group: The proposed language in Rule 3002(a) regarding liens ("A lien that secures a claim against the debtor is not void due only to the failure of any entity to file a proof of claim") should be reworded to state: "The failure of any entity to file a proof of claim does not void a lien that secures a claim against the debtor."

In Rule 3002(c), the language for measuring the time for filing the proof of claim should be consistent. The first part of Rule 3002(c) measures that time from the "date the petition is filed or the date of the order of conversion," while Rule 3002(c)(7)(A)-(B) looks to when "the order for relief is entered." While the language has the same meaning, it should be uniform for clarity.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): Requiring secured creditors to file and shortening the deadline to file claims are improvements. But the language of Rule 3002(c)(7)(B) is a mistake. There is no reason to delay producing the documents supporting the creditors' claims. All supporting documents should be required within 60 days, or the rule should be kept as it is.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges:

The NCBJ supports requiring secured creditors to file proofs of claim in Rule 3002(a). But we suggest the deletion of the last sentence ("A lien that secures a claim against the debtor is not void due only to the failure of any entity to file a proof of claim."). It is a statement of substantive law and has no place in the rules. It also duplicates § 506(d)(2) of the Code and is unnecessary.

We also support the 60-day bar date in amended Rule 3002(c). Some judges have expressed reservations about the shortened bar date, but we do not believe it will be the cause of serious concerns. We note, however, that amended Rule 3002(c) uses inconsistent terminology—referring to the "order for relief" at some places and the "petition" in others. These should be made consistent by using "order for relief."

Subparagraphs (c)(6)(A) and (B) do not make clear whether they provide the exclusive bases for granting extensions. If so, the word "only" should be added after "granted." Historically, the grounds for permitting extensions have been limited.

We question why a special exception has been created for mortgage creditors in subparagraph (c)(7). It would be better to set a single, longer filing period for these creditors rather than separate periods for different parts of a single proof of claim.

April 22-23, 2014 Page 501 of 596

Comment BK-2013-0001-0060—Chief Judge Eric Frank (Bankr. E.D. Pa.): I endorse the NCBJ's comments, except with respect to Rule 3002. Proposed Rule 3002(c) goes too far. If notice of the § 341 meeting is not sent until the 39th day, the creditor would have 19 days (assuming 2 days for mail) to file a proof of claim. This is a draconian and unnecessary reduction. The 90-day deadline should be retained, or the deadline could be shortened to 60 days or 45 days but continue to run from the § 341 meeting.

Comment BK-2013-0001-0074—Chief Judge Laura K. Grandy (Bankr. S.D. Ill.): Rule 3002(c)(7) will be inefficient and confusing, and it will delay confirmation.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): The judges of the D. Md. support this amendment, but would condition it on the debtor's timely filing of a list of creditors.

Comment BK-2013-0001-0077—Han Lee (Attorney): Would the new deadline for filing proofs of claim under Rule 3002(c) apply to all pending cases on the effective date or only cases filed thereafter?

Comment BK-2013-0001-0078—Sheryl Ith (Attorney, Costa Mesa, Cal.): Rule 3002(a) refers to the "lien that secures a claim against the debtor." It should be "against the property of the debtor."

Comment BK-2013-0001-0093—Gary D. Gray (Office of the Chief Counsel, Internal Revenue Service): Delete the word "only" in the new sentence in Rule 3002(a). It is unnecessary and implies that a plan can void the lien of a secured creditor that does not file a claim for reasons other than for not filing a claim.

Rule 3002(a) should clarify that a creditor's right of setoff will also not be void due to the failure to file a claim.

Rule 3002(a) should not be limited to creditors with claims against the debtor. A creditor need only have a claim against the debtor's property in order to have a claim.

We suggest three changes to Rule 3002(c). First, the "except as follows" clause as written would apply only to involuntary cases and should therefore be corrected. Second, instead of requiring a motion to file an untimely claim, the time to file should be extended automatically when a debtor fails to file the creditor list as required. Third, a new exception (subparagraph (9)) should be added to deal with the problem of untimely notice to creditors in chapter 12 and chapter 13 cases.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Rather than separately listing secured and unsecured creditors, the rule should say a "creditor" must file a claim. The new sentence of Rule 3002(a) ("A lien that secures . . .") should include the following language at the end: "or to file such proof of claim in the time prescribed under these Rules."

In Rule 3002(c), we are concerned about the shorter bar date. Also, the provision in (c)(6) for extending the date to file claims by 60 days should cover all scenarios—(i) where the debtor fails to file the list; (ii) omits the creditor(s); (iii) lists the creditors(s) with an incorrect address. In addition, how is a court to determine a "reasonable time to file a proof of claim"? A

April 22-23, 2014 Page 502 of 596

60-day period should be provided instead, or an excusable neglect standard should be included. In the alternative, chapter 13 could be included in Rule 3003. In any event, the rule should address amending a claim after the bar date.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): The proposed bar date is too short. Subparagraph (c)(6) should be broadened to give courts authority to extend the bar date at the request of any party in interest, after notice a hearing, upon reasonable grounds.

Comment BK-2013-0001-0111—Judge Elizabeth W. Magner (Bankr., E.D. La.): I thank the Rules Committee for its work on these proposals.

The shorter bar date is not practical for secured lenders and will result in significant numbers of late-filed claims. These creditors will "ride out" the case rather than attempt to file an untimely claim that will be disallowed.

Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: Rule 3002(c)(6) does not address the claim of creditors who are not timely scheduled by the debtor. We suggest that the rule provide a bar date that is 60 days from the date the creditor is scheduled. The proposed rule appears to require the court to make factual findings, which will deter parties from agreeing to an extension of time.

Comment BK-2013-0001-0118—Alberta E. Hultman, on behalf of USFN: We are an association of law firms and trustee companies specializing in mortgage servicing matters.

The shortened bar date increases burdens on mortgage creditors, who already must comply with requirements imposed by the CFPB and national mortgage settlements.

The bifurcated bar date under Rule 3002(c) creates a conflict. Executing and signing a proof of claim, as the amended rule would permit, without reviewing supporting documentation raises concerns under Rule 9011. The amended rule would also tax trustee staff and debtors' attorneys by requiring review of claims at two points.

We recommend a bar date of 90-days after the petition date, with no bifurcation.

The proposed rule could be read to require proofs of claim in no-asset chapter 7 cases. This should be clarified.

Rule 3002(a) should be clarified so that, where a secured claim rides through the bankruptcy, the amount set forth in the plan is binding only as to plan distribution and not as to the actual amount owing for lien enforcement purposes at the time of discharge.

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: We are the nation's four largest mortgage servicers and are signatories to the national mortgage settlement. We support the goals of the Advisory Committee.

We do not oppose amended Rule 3002(a). But the rule should be clarified in two respects. First, it is ambiguous as to the effect of a secured creditor's failure to file a proof of claim on that creditor's ability to credit bid. Because § 363(k) protects the right to credit bid of a creditor with an "allowed claim," the amended rule should be clarified so that the right to credit bid is preserved. Second, the rule or the Committee Note should make clear that the obligation

April 22-23, 2014 Page 503 of 596

of a secured creditor to file a proof of claim applies even in cases where the debtor intends to surrender the property.

We have three concerns about Rule 3002(c). First, completing Attachment A to the proof of claim form can be time consuming and frequently requires more than 60 days. Bifurcation of the process could give rise to new disputes with respect to proofs of claim, such as the sufficiency of a secured creditor's due diligence with respect to the initial filing. Rule 3001(c)(2)(C) should be amended. Second, the proposed amendment might be read to limit secured creditors' rights to otherwise amend or supplement proofs of claim outside the additional 60 day period. The rule should be clarified to state that it is a safe harbor but does not alter any rights of creditors to amend or supplement thereafter. Third, the shorter bar date applies to chapter 7 cases as well, and we would be required to file proofs of claim promptly even in no asset chapter 7 cases when we do not do so now.

Comment BK-2013-0001-0133—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): The bifurcated bar date will require parties to check for claims on two dates. Keep one bar date at 90 days.

Section 9 [sic] of Rule 3002 will require secured creditors to object to confirmation. The proof of claim should control.

Comment BK-2013-0001-0140—Penelope Souhrada: I agree with the comments of NACBA and the NCBJ.

The new sentence in Rule 3002(a) ("A lien that secures a claim against the debtor is not void . . . ") is awkward and should be reworded.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Proposed Rule 3002(c)'s bar date is too short. The new 60-day deadline is also inconsistent with the 90-day notice under Rule 3002(c)(5). Leave the deadline unchanged.

Proposed Rule 3002(c)(6) does not address the creditor whose name and address were not included on the list of creditors or were erroneously listed. It also does not address a creditor who received notice only a short time before the bar date.

Comment BK-2013-0001-0145—John McMickle, on behalf of eCast: We are a subsidiary of JPMorgan Chase & Co. that is in the market for purchasing and selling chapter 13 accounts.

The shorter bar date in Rule 3002(c) will make it difficult for some creditors to conduct quality assessments before filing proofs of claim. This will interfere with the market for the transfer of claims.

Instead of a shorter bar date, the Bankruptcy Rules should specify that § 1324(b) permits a court to hold a hearing on confirmation and issue a preliminary and conditional order of confirmation that gives creditors additional time to submit proofs of claim. This would resemble the bifurcated deadline for mortgage claims.

Comment BK-2013-0001-0150—Judge Jennie D. Latta (Bankr. W.D. Tenn.): I support the amendments to Rule 3002(a) and Rule 3002(c). There could be mention of the time period for debtors to file proofs of claim for secured creditors who fail to do so as provided by Code § 501(c).

April 22-23, 2014 Page 504 of 596

Comment BK-2013-0001-0151—Raymond J. Obuchowski, on behalf of the National Association of Bankruptcy Trustees: NABT supports the proposed change to Rule 3002(a). We also support shortening the bar date. But we are concerned that the shorter bar date will conflict with Rule 2002(e) and the provision that proofs of claim in chapter 7 cases need not be filed until such time as the trustee files a notice of assets.

We agree with the comments of the Bankruptcy Clerks Advisory Group regarding the consistency between paragraphs (c) and (c)(7)(A)-(B).

Comment BK-2013-0001-0153—James M. Davis (Attorney, Nashville, Tenn.): I support proposed Rule 3002(a), which ratifies existing law. But I suggest that the rule include an authorization for courts to impose sanctions like those under Rules 3001(c)(2)(D) and 3002.1.

Rule 3002(c)(6)(A) is ambiguous when the debtor timely files the list of creditors but omits a creditor. The rule should be reworded to address the debtor's failure to file a timely list of creditors "that complies with Rule 1007(a)."

Rule 3007

Comment BK-2013-0001-0056—Scott Ford, on behalf of the Bankruptcy Clerks Advisory Group: The language of the Committee Note ("Subdivision (a) is amended to provide that an objection to a claim is unnecessary if the determination of the amount of the claim is made through a chapter 12 or chapter 13 plan in accordance with Rule 3012") is more clear than the proposed text of the rule itself. The Committee Note's language should be used instead.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): This rule change conflicts with Fourth Circuit law.

Comment BK-2013-0001-0082—Judge Karen K. Brown (Bankr. S.D. Tex.): I agree with the comments of the judges of the S.D. Tex. In addition, proposed Rules 3007, 3012, and 3015, together with proposed Official Form 113, violate the Bankruptcy Code and binding precedent. They eliminate the statutory requirement of a claim objection and alter the evidentiary effects and burdens regarding a proof of claim. They also confuse collateral valuation under § 506 with claim allowance and disallowance under § 502.

Comment BK-2013-0001-0093—Gary D. Gray (Office of the Chief Counsel, Internal Revenue Service): The amended rule should settle the conflict over proper service upon the Federal Government by requiring broader service upon officials consistent with Rule 7004.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: The language of Rule 3007(a) should be clarified to say that the determination under Rule 3012 only goes to valuation of the secured portion of a claim. The rule is unclear as to the connection between the objection process and plan confirmation. If the plan does not mention an objection to the claim, would that plan not be binding, and if so, how could a rule override it?

April 22-23, 2014 Page 505 of 596

Comment BK-2013-0001-0121—Peter D. Schneider (Attorney, Philadelphia, Pa.), on behalf of Community Legal Services, Inc., of Philadelphia: We support the changes to Rule 3007 and Rule 3012. They will simplify chapter 13 practice.

But we see no reason for an exception for governmental claims.

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: See comment under Rule 3012.

Comment BK-2013-0001-0129—Beverly M. Burden (Chapter 13 Trustee, E.D. Ky.): See comment under Rule 3012.

If a request for valuation requires Rule 7004 service, then a claim objection should also require Rule 7004 service.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): See comments under Rule 3012.

Comment BK-2013-0001-0151—Raymond J. Obuchowski, on behalf of the National Association of Bankruptcy Trustees: We are concerned about the interplay between Rules 3007 and 3012. In the event of conversion from chapter 12 or 13 to chapter 7, will the trustee be precluded from challenging lien validity?

 ${\bf Comment~BK-2013-0001-0161^2-Diana~D.~Herman~(Attorney,~San~Francisco,~Cal.),} on behalf of Insolvency~Law~Committee,~State~Bar~of~California~Business~I.aw~Section:$

See comments under Rule 3012 and 3015.

Rule 3012

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: NACBA strongly supports the option to include determination of allowed secured claim values as part of the chapter 13 plan confirmation process.

Nevertheless, there is no reason to carve out an exception for governmental units. The determination of property value does not require a proof of claim to be filed, so the extended deadline for government proofs of claim should be irrelevant. The exception for governmental units may delay confirmation (and payments to creditors) in many cases where a governmental unit's allowed secured claim is relevant to feasibility or other plan confirmation standards.

Comment BK-2013-0001-0043—Dan Melchi (Attorney, Alpharetta, Ga.): The proposed amendments to Rules 3012 and 3015 are unconstitutional. In combination with § 3.2

April 22-23, 2014 Page 506 of 596

² This comment was submitted after the close of the public comment period.

of the plan form, they violate the Fifth Amendment by depriving creditors of due process and by taking their property without compensation. When a creditor files a secured proof of claim, that creditor is presumed to be a secured creditor until proven otherwise by the debtor or another party in interest. See Bankruptcy Rule 3001(f). A creditor has the right to know before a confirmation hearing whether it is secured or unsecured—the arguments the creditor may wish to make in the case depend on knowing that status. The Advisory Committee's proposed changes mean that a secured creditor will not know whether it is secured or unsecured before confirmation. If a debtor wishes to strip a lien, then notice and a separate valuation hearing should be required so that a creditor receives a ruling from the court prior to confirmation.

Comment BK-2013-0001-0056—Scott Ford, on behalf of the Bankruptcy Clerks Advisory Group: The proposed amendment to Rule 3012(a) separates the phrase "after notice and a hearing" in a manner that is not typical. Because this phrase is defined in Code § 102, we suggest that the proposed amendment be modified so that the phrase is kept together.

Regarding Rule 3012(b), we concur with the comments [about Rule 3015] of Matthew Loughney on behalf of the Bankruptcy Noticing Working Group. The rules should clarify that the debtor is responsible for determining whether Rule 3012 is applicable and whether service is required.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): More clarification is needed as to whether a wholly unsecured second mortgage lien could be stripped through the plan, because there is no reference to § 1322.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges:

We support a change in the rule that permits valuation of secured claims to be combine with objections to the claims themselves. We take no position on the proposed amendment, which permits secured claims to be valued as part of the plan confirmation process.

The proposed rule does not adequately address valuation of secured claims in chapter 11 cases. It addressed chapter 12 and chapter 13 cases but is silent as to chapter 11. It should state the proper procedure for valuing secured claims in chapter 11 cases.

We question the portion of subpart (b) authorizing the amount of a priority claim to be determined by motion or claim objection. That sentence should be clarified or deleted.

We support the requirement in subpart (b) that requests for valuation must be served pursuant to Rule 7004. But the Advisory Committee should consider clarifying the requirements for service in Rule 7004, particularly for circumstances when a creditor has participated in the case in some fashion before a request for valuation is made.

To clarify subpart (c), the word "only" should be inserted so that the amended rule describes the exclusive means and time for requesting a determination as to the amount of a secured claim of a governmental unit.

Comment BK-2013-0001-0071—Judge Dennis Montali (Bankr. N.D. Cal.): I oppose the proposed changes to Rules 3012(b) and 4003(d). In order for a court to make a valuation decision, evidence is needed. But a chapter 13 plan is not evidence.

Also, the proposed rules incorporate Rule 7004 for service. That is burdensome and expensive.

April 22-23, 2014 Page 507 of 596

Comment BK-2013-0001-0082—Judge Karen K. Brown (Bankr. S.D. Tex.): See comments under Rule 3007.

Comment BK-2013-0001-0089—Scott W. Ford (Clerk, Bankr. N.D. Ala.): this will place additional burdens on the clerk's office to review a plan for a request to determine the amount of secured and priority claims. The clerk's office may also have to send separate notices for the request in addition to the notice for the confirmation hearing.

Comment BK-2013-0001-0092—Judge Arthur S. Weissbrodt (Bankr. N.D. Cal.): I support the option to permit the determination of allowed secured claim values as part of the chapter 13 plan confirmation process. This procedure is accepted in chapter 11 practice. There are no due process issues. The proposal will save significant amounts of time and money by eliminating unnecessary motions. We have used this procedure very successfully in the San Jose Division of the N.D. Cal.

Comment BK-2013-0001-0093—Gary D. Gray (Office of the Chief Counsel, Internal Revenue Service): We commend the provision of Rule 3012 with respect to governmental claims. But the rule does not prohibit a debtor from objecting to priority status for a governmental claim before the claim is filed. The proposed rule should be altered to do so.

Rule 3012 should address setoff rights by making clear that determinations of secured status in a plan cannot expunge setoff rights in light of § 553.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: When are determinations under Rule 3012 to be heard—before, after, or at confirmation? Greater clarification is needed regarding governmental priority claims.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): The exception for governmental units alters current practice and will delay distribution of funds to all creditors.

Comment BK-2013-0001-0121—Peter D. Schneider (Attorney, Philadelphia, Pa.), on behalf of Community Legal Services, Inc., of Philadelphia: See comment under Rule 3007.

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: The revisions to Rules 3012, 3015(g), 4003, and 7001 are substantial changes from current law and limit the procedural protections now available in actions to strip off a lien. It is impractical to expect chapter 13 confirmation hearings to include a trial on matters of valuation without delaying the plan confirmation process. We oppose these changes.

If the Advisory Committee proceeds with these changes, however, the amendments to Rules 3007 and 3015 should be modified. They could be read to suggest that a determination under Rule 3012 eliminates the need to file an objection to the total amount of a claim filed by a creditor.

April 22-23, 2014 Page 508 of 596

Comment BK-2013-0001-0125—Judge Rebecca B. Connelly (Bankr. W.D. Va.): The request to determine the amount of a secured claim under Rule 3012(b) should be served on either the holder of the claim or "the creditor subject to the request." This change will reduce delays and costs associated with determining the holder of the claim.

Comment BK-2013-0001-0129—Beverly M. Burden (Chapter 13 Trustee, E.D. Ky.): I do not object to the national plan form and generally support the rules.

Rule 3012 should be altered to provide that a request for determination of the amount of a secured claim may be made in an objection to confirmation.

Service of the entire plan pursuant to Rule 7004 is not necessary. Rule 2002(g) service is sufficient for most valuation requests. Perhaps a \$0 valuation should require Rule 7004 service.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): I oppose the changes to Rule 3012. A governmental unit should not be treated differently in the confirmation process.

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Rule 3012 should be left unchanged, except for an amendment clarifying that valuation may be pursued by objection to a claim.

If Rule 3012 is not left unchanged, Rules 3007 and 3012(b) need to be changed to permit service of an objection via mail to the address on the proof of claim.

Comment BK-2013-0001-0150—Judge Jennie D. Latta (Bankr. W.D. Tenn.): I do not object to requiring Rule 7004 service, but this change will require explanation and perhaps a proof of service form.

Comment BK-2013-0001-0151—Raymond J. Obuchowski, on behalf of the National Association of Bankruptcy Trustees: See comments under Rule 3007.

Comment BK-2013-0001-0154—David S. Yen (Attorney, Chicago, Ill.): I oppose amended Rule 3012(b). The proposed change will harm priority creditors, especially those who are owed domestic support obligations.

Comment BK-2013-0001-0161³—Diana D. Herman (Attorney, San Francisco, Cal.), on behalf of Insolvency Law Committee, State Bar of California Business I.aw Section:

There are due process concerns with pursuing valuation and exemption-impairing-lien avoidance through a chapter 13 plan. The rules governing confirmation hearings in chapter 13 do not address service and notice as well as the rules governing motions and adversary proceedings.

April 22-23, 2014

Page 509 of 596

³ This comment was submitted after the close of the public comment period.

Rules 3015 and 4003

Comment BK-2013-0001-0010—Judge Kay Woods (Bankr. N.D. Ohio): Does Rule 4003(d), when taken together with amended Rule 7001, mean that only judicial liens in chapter 7 and chapter 11 cases can be avoided by motion? If so, then the first part of the first sentence of proposed Rule $4003(d)^4$ is overly broad. What is the purpose of requiring Rule 7004 service to avoid a judicial lien with a chapter 12 or chapter 13 plan when a motion governed by Rule 9014 can be used to avoid a judicial lien in a chapter 7 or chapter 11 case?

Comment BK-2013-0001-0011—Jon Waage (Chapter 13 Trustee, M.D. Fla.), on behalf of the NACTT:

The requirement [in Rule 3015(f)] that objections must be filed 7 days before the confirmation hearing presents a problem in jurisdictions where a plan is confirmed without a hearing unless an objection is filed. There would be no real deadline to file an objection to the plan in those jurisdictions. The rule should include an alternative deadline to file an objection if a plan is confirmed, absent objection, without a hearing.

Comment BK-2013-0001-0013—Matthew T. Loughney (Clerk, M.D. Tenn.), on behalf of the Bankruptcy Noticing Working Group: We support elimination of the requirement in Rule 3015(d) that the plan "shall be included with each notice of the hearing on confirmation mailed pursuant to Rule 2002." But amended Rules 3015 and 4003 should be clarified regarding service of notice.

Rule 3015 should state that the responsibility for noticing a plan rests on the debtor, and that the responsibility for noticing a plan modification rests on the proponent of the modification. The proposed amendment to Rule 3015(d) ("the debtor shall serve the plan on the trustee and all creditors when it is filed with the court") does not make clear if the plan would need to be served under Rule 7004 or otherwise. Proposed amendments to Rule 3012 and 4003 suggest that Rule 7004 service is necessary only when the debtor is seeking to determine the amount of a secured claim or seeking to avoid a lien or other transfer or exempt property under Code § 522(f). Rule 3015(d) should require that the debtor "give notice" of the plan instead of "serve" it, and the debtor should have the burden of determining whether Rules 3012 or Rule 4003 apply. The clerk should not be expected to review plan language to make that determination. Similarly, Rule 3015(h) states that a modification is to be "served . . . in the manner provided for service of the plan" in Rule 3015(d). But Rule 3015(d) does not set out a manner of service. Rule 3015(h)

April 22-23, 2014 Page 510 of 596

⁴ The first sentence of proposed Rule 4003(d) provides: "A proceeding by the debtor to avoid a lien or other transfer of property exempt under § 522(f) of the Code shall be commenced by motion in the manner provided for by Rule 9014, or by a chapter 12 or 13 plan served in the manner provided by Rule 7004 for service of a summons and complaint."

should be clarified to state: "The proponent of the modification shall give notice of the proposed modification, or a summary thereof, to the trustee and all creditors when it is filed with the court."

The option to send a summary of the plan should be retained in Rule 3015(d). It is much less expensive to mail a summary than a multi-page plan. Perhaps a plan summary form could be developed in conjunction with the full plan form.

Rule 4003 should specify upon whom the plan is to be served, pursuant to Rule 7004, when the debtor is seeking to avoid a lien or other transfer of exempt property under Code § 522(f). Language such as the proposed amendment to Rule 3012 (which provides, "The request shall be served on the holder of the claim and any other entity the court designates in the manner provided for service of a summons and complaint by Rule 7004") should be included in Rule 4003. This would clarify whether the plan must be served like a summons and complaint on all recipients or only on specified recipients.

Comment BK-2013-0001-0014—K. Michael Fitzgerald (Chapter 13 Trustee, W.D. Wash.): The plan allows debtors to fix the value of a secured creditor's collateral and to avoid liens. The proposed rules require that a plan attempting to do so must be served in accordance with Rule 7004. Allowing the plan to trump a proof of claim causes more problems than the potential benefits. It will generate creditor objections. I question whether plans will be properly served and who will be responsible for monitoring service. That responsibility should not be delegated to the trustee, and judges and clerks do not have the time and resources to ensure service compliance.

Comment BK-2013-0001-0030—Judge Diane Finkle (Bankr. D.R.I.): Proper service on secured creditors is important when plans propose to modify a secured claim or avoid a judicial lien. Service on a creditor's payment address may be ineffective, and Rule 7004(h) service on federal depository institutions may be inadequate because that rule was designed for service of a summons and complaint for litigation. Perhaps a new federal rule could be added to address these issues, or perhaps an additional service requirement can be added by way of an acknowledgment by the debtor in the plan as to the particular service method. [A local rule from D.R.I. that sets forth service requirements for chapter 13 plans is attached to the comment.]

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys:

There are serious problems with the amendment to Rule 3015(d). The changes to this subdivision would increase costs for chapter 13 debtors. First, the word "serve" should not be used, because it could lead courts to require service under Rule 7004. Second, requiring even the mailing of the entire chapter 13 plan to every creditor would impose an additional cost that is not currently required in most districts. Most large institutional creditors receive electronic notice though the Bankruptcy Noticing Center, which should be responsible for sending the plan electronically to all creditors that receive electronic notice. Third, insured depository institutions should be required to provide to the courts the names and addresses to be used for service under Rule 7004(h) in order to end the "gotcha" game of finding the right officer and address for service under that rule.

91

NACBA supports the changes to subdivisions (f) and (g).

April 22-23, 2014

Subdivision (h), which requires that the proposed modifications be served on all creditors if notice of modification does not include the modification, raises the same problems as requiring service of the plan in subdivision (d). The language of the amendment to subdivision (h) should be altered to require that the modification be "sent" (not "served") with the notice in all cases, and sent electronically by the Bankruptcy Noticing Center when possible. Also, there is no reason to require notice to the trustee and creditors only when the debtor (as opposed to the trustee or an unsecured creditor) seeks a modification.

NACBA supports amending Rule 4003 to permit the option of effectuating lien avoidance under Code § 522(f) as part of the confirmation of a chapter 13 plan. But the rule and the Committee Note should be reworded to make clear that service under Rule 7004 is needed only with respect to the creditors whose liens are being avoided.

Comment BK-2013-0001-0043—Dan Melchi (Attorney, Alpharetta, Ga.): The proposed amendments to Rules 3012 and 3015 are unconstitutional. [This comment is described under Rule 3012.]

Comment BK-2013-0001-0044—Judge Brian D. Lynch (Bankr. W.D. Wash.), on behalf of the bankruptcy judges of the district: The proposed amendment to Rule 3015(c) (Form of Chapter 13 Plan) should not be adopted. It appears to bar changes to the form, and is too restrictive.

Comment BK-2013-0001-0056—Scott Ford, on behalf of the Bankruptcy Clerks Advisory Group: We concur with the comments of Matthew Loughney on behalf of the Bankruptcy Noticing Working Group. Rule 3015(d) should be clarified to state that it is the debtor's responsibility to give notice of the plan. We also support the recommendation that the option to provide a summary of the plan be returned to the rule.

We also concur that Rule 3015(h) should make clear that the proponent of a modification is responsible for providing notice and for determining whether Rule 3012 or Rule 4003 apply.

We reiterate these concerns with respect to the proposed amendment to Rule 4003(d). Also, Rule 4003(d) does not specify who should be served. We agree with the Noticing Group's comment that this should be clarified.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): The plan should not be mandatory.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges:

The second sentence of proposed Rule 3015(c) should be deleted. A declaration as to which plan provisions are "effective" is a declaration of substantive law, not procedure. This amendment is inconsistent with Code § 1327(a) and *Espinosa*. As an alternative, the amended rule should be re-written to avoid use of the word "effective."

Subpart (d) should be altered to discuss explicitly service of an amended plan prior to confirmation.

Subpart (g) should be deleted. It prescribes the "effect" of confirmation and is therefore substantive and not procedural.

April 22-23, 2014 Page 512 of 596

The proposed new language in subpart (h) should be deleted. It is unnecessary to amend a rule to impose a service requirement if the current service requirement is disobeyed.

Comment BK-2013-0001-0063—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): Proposed Rule 3015(f) interferes with a bankruptcy judge's ability under Code § 102 to issue promptly an order confirming the plan without a hearing when there are no objections filed within a fixed time after filing of the plan. The deadline for objecting to confirmation should not be based on the date of the confirmation hearing. If the hearing is set for a late date, this will cause delay in confirmation. The deadline should be 21 days after the filing of the notice of the time to object unless the court sets a longer deadline. [Alternative language is provided in the comment.]

Comment BK-2013-0001-0065—Judge Terrence L. Michael (N.D. Okla.): The proposed revisions to Rule 3015(c) exist solely to make the plan form mandatory and should be discarded in their entirety. The plan form should not be mandatory. I agree with the comments submitted by Judge Lamar W. Davis, Jr., Judge Jeffrey Bohm, the judges of the United States Bankruptcy Court for the District of Kansas, NACBA, and others.

Comment BK-2013-0001-0071—Judge Dennis Montali (Bankr. N.D. Cal.): I oppose the proposed changes to Rules 3012(b) and 4003(d). In order for a court to make a valuation decision, evidence is needed. But a chapter 13 plan is not evidence.

Also, the proposed rules incorporate Rule 7004 for service. That is burdensome and expensive.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): The judges of the D. Md. oppose proposed Rule 3015(c). The plan should not be mandatory. Also, Rule 3015(g) conflicts with 4th Circuit law.

With respect to proposed changes to Rule 4003, lien avoidance is better achieved by motion served under Rule 7004.

Comment BK-2013-0001-0080—Judge John Waites (Bankr. D.S.C.): I agree with the NCBJ's comments. The language of proposed Rule 3015 regarding provisions deviating from the Official Form should be deleted. Amended plans that do not adversely affect creditors should require service only upon the trustee.

Comment BK-2013-0001-0082—Judge Karen K. Brown (Bankr. S.D. Tex.): See comments under Rule 3007.

Proposed Rule 3015(g) violates the Rules Enabling Act. It modifies substantive law because it is contrary to §§ 502(a) and 506.

Comment BK-2013-0001-0085—Judge Roger Efremsky (Bankr. N.D. Cal.), on behalf of six judges of the bankruptcy court: For reasons given by Judge Lamar W. Davis, Jr., we oppose Rules 3015(c) and 9009. The form should not be mandatory.

April 22-23, 2014 Page 513 of 596

Comment BK-2013-0001-0089—Scott W. Ford (Clerk, Bankr. N.D. Ala.): We believe the current one-page plan in our district works well, and we are reluctant to support a national plan.

Rule 3015 should require the debtor and not the court to serve the plan.

Rule 4003 will place more work on the clerk's office to review the plan for a request to avoid a lien or other transfer of exempt property.

Comment BK-2013-0001-0093—Gary D. Gray (Office of the Chief Counsel, Internal Revenue Service): Rule 3015(g) should be clarified so that a plan may not determine the amount of a governmental unit's secured claim in violation of Rule 3012(c). The phrase "Except as provided in Rule 3012(d)" should be added at the start of Rule 3012(g).

Rule 4003, together with § 3.4 of Official Form 113, does not specify what kinds of nonpurchase-money security interests are excluded. We are concerned that debtors may attempt improperly to avoid statutory liens, including tax liens. Clarification should be added to the Committee Note.

Comment BK-2013-0001-0095—Nik Geranios: I agree with comments submitted by NACBA regarding Rule 3015.

Comment BK-2013-0001-0097—James Jacobsen, on behalf of the States' Association of Bankruptcy Attorneys: Stylistic changes suggested for Rule 3015(a) and (b).

Clarifying language suggested for Rule 3015(c) regarding the effect of a nonstandard provision.

Stylistic changes suggested for Rule 3015(g).

Comment BK-2013-0001-0100—Byron Meredith (Chapter 13 Trustee, S.D. Ga.): Eliminating the option to serve a summary of the plan will increase costs. As an alternative, a single-page notice could be provided to parties in interest with a link to the location of the plan on the court's ECF system.

Comment BK-2013-0001-0102—William Heitkamp (Chapter 13 Trustee, S.D. Tex.): Proposed Rule 3015 will increase the cost of notice by eliminating the option of serving a summary of the plan.

The national plan form should not be mandatory.

Comment BK-2013-0001-0120—Chief Judge Robert E. Grant (Bankr. N.D. Ind.): The rules mandating the chapter 13 plan form cross the line from procedure into substance. The language of § 1325(a) ("the court shall confirm a plan if . . .") raises doubts as to whether rules of procedure can mandate use of a particular form for a chapter 13 plan. See Petro v. Mishler, 276 F.3d 375, 378 (7th Cir. 2002). Unlike chapter 11, where Congress has authorized promulgation of a form plan for some aspects of practice, no such authorization is given to promulgate a chapter 13 plan form. No other form is mandatory.

Comment BK-2013-0001-0121—Peter D. Schneider (Attorney, Philadelphia, Pa.), on behalf of Community Legal Services, Inc., of Philadelphia: We are concerned about Rule

April 22-23, 2014 Page 514 of 596

3015(c), which would mandate use of the chapter 13 plan form. Courts may interpret this rule to prohibit debtors from using non-standard provisions that are fully consistent with the Code.

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: See comments under Rule 3012.

Rule 3015(c) should limit what may be included in a plan as a nonstandard provision.

Comment BK-2013-0001-0125—Judge Rebecca B. Connelly (Bankr. W.D. Va.): I support the proposed amendments and form. I applaud the efforts of the Advisory Committee.

I agree with the comments of the Bankruptcy Noticing Working Group and the Bankruptcy Clerks Advisory Group regarding this rule (use of the word "serve").

Add language to Rule 3015(b) to accommodate those creditors who wish to designate an address for notice in bankruptcy cases.

Comment BK-2013-0001-0127—Chief Judge Susan Barrett (Bankr. S.D. Ga.), on behalf of the judges of the court: We oppose a mandatory national plan form. It will damage the professionalism of practice in our court and impair the close relationship between consumer debtors and local attorneys.

Comment BK-2013-0001-0129—Beverly M. Burden (Chapter 13 Trustee, E.D. Ky.): Rule 3015 should set forth procedures for pre-confirmation modifications. Rule 2002(a)(5) should be amended to exclude modifications of chapter 13 plans. Rule 3015 should permit changes to a plan prior to confirmation without notice to all creditors if those changes do not affect all creditors.

I support Rule 4003. For consistency, the rule should provide that a "request" to "avoid a lien may be made by . . . " rather than stating that "a proceeding . . . shall be commenced."

Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.): I agree with the comments of Bankruptcy Judge Terrence L. Michael of the N.D. Okla. A mandatory form is a solution in search of a problem.

Comment BK-2013-0001-0133—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): I oppose the changes to Rules 3015 and 9009.

Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): I oppose the national plan form and oppose Rule 3015(c).

But I support Rule 3015(d).

Comment BK-2013-0001-0142—Judge S. Martin Teel, Jr. (Bankr. D.D.C.): The proposed official form should not be mandatory. It should be a director's form.

If Official Form 113 is adopted as a mandatory form, the form or Rule 3015(c) should be modified to permit use of a plan that omits parts of the form that are not used.

Comment BK-2013-0001-0145—John McMickle, on behalf of eCast: The proposed amendment to Rule 3015 should make clear that nothing in Form B10 or the proposed rules overrides § 502 and the debtor's duty to object to proofs of claim.

April 22-23, 2014 Page 515 of 596

In the alternative, the plan form should require debtors to list in Part 9 how each unsecured claim is modified in the plan. Proposed Rule 3015(g) should be altered to permit creditors to object to their treatment post-confirmation if an objection to the proof of claim was not filed.

Comment BK-2013-0001-0147—Mark Bulovic (Attorney, Savannah, Ga.): I oppose a national plan form. The system is not broken, and this proposal will lead to the further "dumbing down" of bankruptcy practice.

What is the consequence of a failure to comply with the national plan form? Would it be a bad faith plan? Would it be considered forbidden?

Comment BK-2013-0001-0150—Judge Jennie D. Latta (Bankr. W.D. Tenn.): I support a national plan form. The disparity in plans and procedures creates barriers to entry into bankruptcy practice by young lawyers and increases costs to national creditors. The proposed form will promote more attentive and careful representation of consumer debtors by raising a number of issues that are often overlooked by counsel.

Nevertheless, the form and rules should permit greater local modification of the form. For example, we are a conduit district and require payroll deductions. I would like to be able to retain those practices.

I support the changes to Rule 4003.

Comment BK-2013-0001-0161⁵—Diana D. Herman (Attorney, San Francisco, Cal.), on behalf of Insolvency Law Committee, State Bar of California Business I.aw Section:

The changes to the Bankruptcy Rules will make plan confirmation hearings longer and more elaborate. At the least, courts will have to adjust their calendaring practices.

There appears to be an inconsistency between the proposed changes to Rule 3015 and the plan form. Part 1 of the plan form provides that an objection must be filed at least 7 days before the confirmation hearing "unless otherwise ordered by the Bankruptcy Court." This is important for districts that confirm plans on negative notice. But it is unclear whether the proposed changes to Rule 3015(f) relating to the 7-day deadline to object to confirmation may be altered by the court.

Rule 5009

Comment—BK-2013-0001-0033 Ed Shaw: I oppose the amendment to Rule 5009. The current rules concerning case closing are adequate, and the new rule would compromise the substantive rights of parties.

96

Page 516 of 596

_

April 22-23, 2014

⁵ This comment was submitted after the close of the public comment period.

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: NACBA supports amending Rule 5009 to allow the debtor to obtain an order that a lien has been satisfied. The amendment to the rule should be further expanded, however, to provide for an order that a lien has been avoided. If a separate motion for lien avoidance will not be required, then there will be no resulting order by which a debtor may show that a lien has been avoided. The amended rule should make clear that a separate order avoiding a lien can be entered if a plan including a lien avoidance provision is confirmed.

Comment BK-2013-0001-0053—Judge Meredith Jury (Bankr. C.D. Cal.): The word "satisfied" should not be used in Rule 5009(d). Satisfaction generally requires payment. A better word is "avoided," "ineffective," or some other term that makes clear the lien is gone.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): This rule would effectively mandate another motion, prior to case closing, to hold open the case to allow time to confirm satisfaction. This would require notice and a hearing on the motion to hold open and then notice and a hearing on the motion to deem the lien "satisfied." This is not a simplification.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: We recommend two changes.

The initial subject of subpart (d) as well as the first sentence should reflect that a "claim secured by a lien" is satisfied. Debtors satisfy claims, not liens.

The last sentence of subpart (d) should be deleted. Whether an order is "effective as a release of the lien" is substantive.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): We support this amendment.

Comment BK-2013-0001-0093—Gary D. Gray (Office of the Chief Counsel, Internal Revenue Service): We object to proposed Rule 5009(d). An adversary proceeding should be required to determine that a lien was satisfied by plan payments.

Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: References to lien "satisfaction" should be changed to "release." Satisfaction requires that the underlying debt has been paid in full, while release does not.

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: We oppose Rule 5009(d). No substantive provision of the Bankruptcy Code provides for a bankruptcy court to make such a determination, which would be a matter of private right that cannot be heard and determined by the bankruptcy court under *Stern v. Marshall*.

Comment BK-2013-0001-0150—Judge Jennie D. Latta (Bankr. W.D. Tenn.): I support the changes to Rule 5009.

April 22-23, 2014 Page 517 of 596

Rule 7001

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): This amendment would bar a district from requiring a motion or adversary proceeding to strip a lien. This raises the difficulties I expressed regarding Rules 3012 and 5009.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: This amendment reflects two mistaken legal assumptions.

First, the determination under Rule 3012 does not involve the determination of the "validity, priority, or extent of a lien." It involves valuation.

Second, relief under Rule 3012 is sought through motion and is therefore a contested matter under Rule 9014. It is not a "proceeding" as the proposed amendment incorrectly says.

Furthermore, amended Rule 7001 is unnecessary in light of amended Rule 3012(b).

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): We support this amendment.

Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: See comments under Rule 3012.

Comment BK-2013-0001-0150—Judge Jennie D. Latta (Bankr. W.D. Tenn.): I support the proposed amendment to Rule 7001.

Comment BK-2013-0001-0151—Raymond J. Obuchowski, on behalf of the National Association of Bankruptcy Trustees: We agree with the NCBJ's comments regarding the interplay between Rules 7001 and 3012.

Rule 9009

Comment—BK-2013-0001-0031 Walter Oney (Attorney, Fitchburg, Mass.): Proposed Rule 9009(a) is too restrictive and will cause great hardship to software developers. The rule should allow "minor changes in layout and formatting not affecting substance or the order of presenting information." An alternative would be to allow "changes in the width and orientation of columnar data."

Comment BK-2013-0001-0042—Henry Sommer, on behalf of the National Association of Consumer Bankruptcy Attorneys: NACBA is concerned that amended Rule 9009 would be unduly restrictive. For example, by requiring "substantially similar" font faces, it does not give enough flexibility for a debtor to save printing and mailing costs by reducing the typeface size and using double-sided printing. The current rule language permitting appropriate alterations and rearrangements to permit economies should be maintained. There have been few, if any, situations where courts have found inappropriate alterations under Rule 9009's current language.

98

Comment BK-2013-0001-0044—Judge Brian D. Lynch (Bankr. W.D. Wash.), on behalf of the bankruptcy judges of the district: The existing language of Rule 9009 is adequate. It allows for flexibility. The proposed amendments to Rule 9009 are an attempt to preempt suggestions by certain chapter 13 trustees that local rules or forms could be incorporated by reference in lieu of the national form plan. The proposed official form may be a worthy model for courts to consider and to adopt as they see fit. If it fulfills all the promises claimed by proponents, the form will become the prevailing approach to processing chapter 13 cases.

Comment BK-2013-0001-0056—Scott Ford, on behalf of the Bankruptcy Clerks Advisory Group: The proposed amendments to Rule 9009 are a departure from the current rule. The clerk's office will be tasked with checking that submitted forms comply with the revised rule, and this will generate more work for clerk's offices.

We also endorse the recommendation by the E.D. Missouri bankruptcy court that the Committee Note or the rule text itself should clarify that a party need not submit a proposed order that is part of an Official Form if the court directs that it is not required.

Comment BK-2013-0001-0058—Michael Gallagher (Attorney, Plymouth Meeting, Pa.): I oppose this change. The current rule should be retained.

Comment BK-2013-0001-0059—Judge A. Benjamin Goldgar (Bankr. N.D. Ill.), on behalf of the National Conference of Bankruptcy Judges: The proposed restriction on the alteration of forms is too drastic. It would be better to require debtors to indicate on each form whether it has been modified and, if so, what portion has been modified.

We also recommend rewriting or deleting subparagraph (a)(5). It is poorly worded and its purpose is not clear. Does it restrict the court's ability to order changes to forms, or does it prohibit the court from using orders in a way that might conflict with the official form or applicable rule? The subparagraph is not necessary, as it does not address a real problem. To the extent it is meant to prevent courts from entering orders that contradict forms or rules, parties have remedies in the form of a motion under Rules 9023 or 9024 or an appeal.

Comment BK-2013-0001-0065—Judge Terrence L. Michael (N.D. Okla.): The proposed revisions to Rule 9009 exist solely to make the plan form mandatory and should be discarded in their entirety. There is no sound reason to wrest discretion from each and every bankruptcy judge in the country. I agree with the comments of the Bankruptcy Clerks Advisory Group regarding Rule 9009.

Comment BK-2013-0001-0067—Dana C. McWay (Clerk, Bankr. E.D. Mo.): Under the proposed amendment, the clerk's office will receive documents that courts do not want. For example, we do not expect counsel to submit Official Form B3A (filing fee installment). Instead, the CM/ECF system creates an order concerning payment. I suggest clarifying in the Committee Note that such forms are exempted from Rule 9009.

Comment BK-2013-0001-0075—Mark D. Sammons (Clerk, Bankr. D. Md.): Other than our opposition to the plan form, we support this rule.

April 22-23, 2014

Page 519 of 596

- Comment BK-2013-0001-0080—Judge John Waites (Bankr. D.S.C.): I agree with the NCBJ's comments. Subsection (a)(5) should be deleted.
- Comment BK-2013-0001-0085—Judge Roger Efremsky (Bankr. N.D. Cal.), on behalf of six judges of the bankruptcy court: For reasons given by Judge Lamar W. Davis, Jr., we oppose Rules 3015(c) and 9009. The form should not be mandatory.

We agree with Judge Montali's comments regarding valuation issues.

- **Comment BK-2013-0001-0094—Traci E. Abrams (Clerk, Bankr. N.D. Fla.):** Amended Rule 9009 will interfere with how courts have programmed forms into CM/ECF systems and Autodocketing Interface (ADI). For example, Form B113 is the "Claims Register." The claims register is now a part of CM/ECF, which uses a very different format.
- Comment BK-2013-0001-0101—Chief Judge James K. Coachys (Bankr. S.D. Ind.): I agree with the comments of Judge Lamar W. Davis Jr. and the judges of the bankruptcy court for the District of Kansas. The plan form should not be mandatory.
- Comment BK-2013-0001-0109—Devin Derham-Burk (Chapter 13 Trustee, N.D. Cal. (San Jose Division)): The national plan form should not be mandatory.
- Comment BK-2013-0001-0115—M. Regina Thomas (Clerk, Bankr. N.D. Ga.), on behalf of Bankr. N.D. Ga. Bench and Bar Committee: The proposed rule is too restrictive.
- Comment BK-2013-0001-0124—Craig Goldblatt (Attorney, Washington, D.C.), on behalf of Bank of America, N.A., et al.: See comments on Rule 3015.
- Comment BK-2013-0001-0131—Gwendolyn M. Kerney (Chapter 13 Trustee, E.D. Tenn.): See comments under Rule 3015.
- Comment BK-2013-0001-0133—Ray Hendren (Chapter 13 Trustee, W.D. Tex.): See comments under 3015.
- **Comment BK-2013-0001-0140—Penelope Souhrada:** I oppose a mandatory form. This rule makes no allowance for local laws, rules, practices, and even provisions of the Code. The national plan form does not include provisions such as those required under § 524(i).
- Comment BK-2013-0001-0141—Pam Bassel (Chapter 13 Trustee, N.D. Tex.): See comments under Rule 3015.
 - Proposed Rule 9009 is too restrictive. I agree with Judge Terrence Michael's comments.
- Comment BK-2013-0001-0150—Judge Jennie D. Latta (Bankr. W.D. Tenn.): See comments under Rule 3015.
- Comment BK-2013-0001-0151—Raymond J. Obuchowski, on behalf of the National Association of Bankruptcy Trustees: We support the concept of Rule 9009. But we also support the need for greater flexibility. We agree with the comments of the NCBJ.

April 22-23, 2014 Page 520 of 596

Appendix D

Comment Summary for Comments on Published Forms

April 22-23, 2014 Page 521 of 596

April 22-23, 2014 Page 522 of 596

General Comments on Published Individual Forms

<u>USC-RULES-BK-2013-0001-0029 – David Kingsbury</u> – The proposed revamped forms will encourage additional pro se filings. As we all know bankruptcy is not about filling in forms - it is about understanding the rules, statutes and case law that dictate and interpret how forms are completed. The proposed forms may be more user friendly but the certain resulting surge in pro se filings will not benefit the system or debtors.

<u>USC-RULES-BK-2013-0001-0035 – Chapter 13 Trustee Robert G. Drummond</u> – I want to compliment the committee for its work. Your task is huge.

<u>Bankruptcy Attorneys (NACBA)</u> – Although not contained in the proposed rules or forms themselves, The Advisory Committee's Report to the Standing Committee refers to plans for "the ability to store information on the forms as data so that authorized users can produce customized reports suitable for their needs." NACBA strongly opposes any form of data-enabled forms or capability to mine the information that bankruptcy debtors report to the courts beyond that which is statutorily required. In this era of government snooping on citizens, further expansion of data gathering is totally unwarranted. It goes far beyond the purpose for which the bankruptcy forms are designed — to provide information in individual cases. There is no reason that bankruptcy debtors, in court solely because of economic hardships, should be subjected to such intrusive scrutiny.

NACBA also fears the uses to which such data could be put. We have seen politicized government agencies use data for political ends, sometimes contrary to the interests of consumer debtors. We also fear that any system that enables collection of data could be accessed, either with the consent of the courts or illicitly, for illegitimate purposes such as marketing or targeting consumer debtors in various other ways.

<u>USC-RULES-BK-2013-0001-0046</u> – <u>Rommel Jairam</u> – Thank you for heading in this direction. I would encourage the IRS and many other agencies to purpose similar changes in their publications and forms. Many issues with forms and taxpayer delinquencies are due to the lack of understanding from the wording of these documents. I believe that this regulation will benefit many taxpayers and even the employees who must interpret them.

<u>USC-RULES-BK-2013-0001-0048 – Mike Walters</u> – This is definitely a step in the right direction.

<u>USC-RULES-BK-2013-0001-0059</u> – <u>National Conference of Bankruptcy Judges (NCBJ)</u> – The NCBJ applauds and endorses the project to revise the bankruptcy forms. As the NCBJ noted in its comments last year, the revised forms are more readable, easier to fill out, and easier to understand than their predecessors.

April 22-23, 2014 Page 523 of 596

The current set of proposed revised forms requires further careful review to ensure the prose is clear and grammatical, the format is consistent from form to form, and any statements of law are accurate.

The NCBJ also has serious reservations about the degree to which the revised forms – and particularly the instructions that accompany them – make statements about the law. Presumably, these statements are meant to guide pro se filers. But forms and instructions should not attempt to substitute for actual legal representation because they can never do so adequately. This is especially true in the bankruptcy context because bankruptcy law is riddled with exceptions (and sometimes exceptions to exceptions), making general assertions at best incomplete and at worst misleading. Requesting information from a debtor is one thing, giving legal advice about the information to be provided another. In the NCBJ's view, assertions in the forms about the law should be eliminated except when they are absolutely necessary to instruct the debtor how to complete the form.

<u>USC-RULES-BK-2013-0001-0100 – O. Byron Meredith III, Chapter 13 Trustee</u> – Debtors should be allowed to elect to receive notices from the Court electronically. It would reduce costs, assure notices are received timely, and avoid the volume of returned mail when debtors relocate without notifying the Court.

The Rules Committee should consider recommending that 11 U.S.C. § 342 be amended to require all creditors that receive a certain volume of notices from the Court to designate an electronic address for service.

<u>USC-RULES-BK-2013-0001-0140</u> – <u>Penelope Souhrada</u> – I heartily concur with most of the comments made by Henry Sommer of the National Association of Consumer Bankruptcy Attorneys and the very thorough and thoughtful analysis of the revised forms submitted by the National Conference of Bankruptcy Judges. (What better source for comments than the bankruptcy judges, who deal with these issues every day?)

The new forms in general have way too many spaces for additional entries in the same category. We print out hard copies of the petition for clients and unless the forms can shrink to fit the individual case, there will be a terrible waste of paper.

<u>USC-RULES-BK-2013-0001-0147 – Mark Bulovic</u> – There seems to be a push to encourage pro se filings by providing rudimentary bankruptcy advice to would-be pro se debtors. The explanation of bankruptcy terms is so dumbed-down as to be grossly misleading. For example, "Exempt Property" is certainly not property which a debtor may necessarily keep. Explaining the significance of exemptions, especially where the debtor has both exempt and non-exempt equity in property, is tough enough now. When the forms start telling clients that they may keep "exempt property" they will think their lawyer is crazy when she tells them the trustee will sell their home and send them a little check.

Before the Rules Committee makes a policy decision that has the effect of encouraging pro se filings, you need to decide whether debtors are generally better-off without lawyers. If so, can bankruptcy judges award fees to debtors' counsel? Do lawyers have an ethical obligation to

April 22-23, 2014 Page 524 of 596

decline debtor representation because the debtors would be better-off filing pro se? The answers to these questions seem obvious to me: My unvarying advice to people who ask is that every debtor needs not just a lawyer, but a lawyer who has solid bankruptcy experience and knowledge. If pro se is a bad idea, please don't encourage it.

Also, I believe that turning schedule preparation into a "turbo tax" process is an invitation to bankruptcy fraud, or at least non-disclosure. It is incredible how many debtors will provide a different answer about their debts and assets depending on who asks the question, how many times it's asked, and how the question is phrased. If you haven't interviewed a would-be debtor recently, you may have forgotten how poorly—or selectively—they recall the details of their financial affairs.

April 22-23, 2014 Page 525 of 596

General Comments about Schedules and Instructions

<u>USC-RULES-BK-2013-0001-0058 – Michael W. Gallagher</u> – Proposed amended Schedules A/B, C, D, E/F, G, H, and the proposed amended Statement of Financial Affairs: To be blunt, this is an attempt at a cure where there was no disease. In fact, as an attorney whose practice includes a large percentage of individual debtors' cases, I will say that the new proposed forms are a complete mistake. The current forms are comprehensible.

They are a bit intimidating to the pro se filer—but doubling their length and including voluminous instructions is NOT going to increase accuracy or compliance from pro se filers. Many such persons already avoid such things—and you propose to give them more to read, fill out, and follow. This will likely increase the number of pro se filers—but it will also increase the number of grossly inaccurate filings. The additional burden on panel trustees will be significant, as well as the burden on the courts. The new forms are a mistake, period.

I did a rough calculation of the number of pages a simple Chapter 7 filing would take, assuming that the Debtor was above-median income, but had sufficient secured debt deductions to qualify under Chapter 7. The minimum number of pages this would produce, by my count, is 67: the current average is a little over 30. Wonderful for an HP or Lexmark, seeking to sell more laser toner and ink; or for a Weyerhauser, seeking to sell more copy paper. For anyone else, however, this is grossly wasteful.

<u>USC-RULES-BK-2013-0001-0155</u> – <u>David S. Yen</u> – The forms are misleading because they limit the property that can be claimed as exempt to property listed on Schedule A/B, even though leases and executory contracts can be claimed as exempt.

Form 106 A/B tells the debtor "On this form, do not list any interests you may have in executory contracts".

Form 106C, Part 1, section 2, says: "For any property you list on Schedule A/B that you claim as exempt, fill in the information below."

In the booklet of instructions, the instructions for Schedule C, under the heading "Claiming Exemptions", say: "Using the property and values that you listed on Schedule A/B: Property (Official Form 106 A/B) as your source, list on this form the property that you claim as exempt."

A debtor who followed the instructions on Form 106C and in the booklet of instructions would not claim any exemption in an executory contract or unexpired lease. However, many exemption regimes allow the debtor to claim unexpired leases as exempt, sometimes under the jurisdiction's homestead exemption. See, e.g, In re Casserino, 379 F.3d (9th Cir. 2004). Failure to understand that an unexpired lease is an asset that has value which therefore can be sold or assigned by a trustee, which may or may not be exemptible, can have drastic consequences that a pro se debtor may be unaware of. See "Widow's Bankruptcy Case Poses Risk to Rent-Stabilized Tenants", New York Times, October 13, 2013, http://www.nytimes.com/2013/10/21/nyregion/widows-bankruptcy-case-poses-risk-to-millions-with-rent-stabilized-leases.html? r=0

April 22-23, 2014

Page 526 of 596

Recommendation: All references in Form 106 C that refer to "Schedule A/B" should refer to "Schedule A/B or G." Schedule G should have a column for the debtor to list the value, if any, of a lease or executory contract.

April 22-23, 2014 Page 527 of 596

Official Form 22A-1

<u>USC-RULES-BK-2013-0001-0042 – Henry Sommer, NACBA</u> – The language in the box at the upper right hand corner stating that "The presumption of abuse is determined by Form 22A-2" is inconsistent with that in Form 22A-2 and the current form, which recognize that the debtor may disagree with the way the means test is calculated on the forms. The language "according to the calculations required by Form 22A-2" should be added.

The form is also simply inaccurate in stating that it is a calculation of current monthly income. As defined by section 101(10A), current monthly income does not include the income of a nondebtor spouse that is not used for the household expenses of the debtor or the debtor's dependents.

The form repeats the error in the current form of calculating the section 707(b)(7) exemption from the means test based on an amount that is not current monthly income. As a number of courts have held in the chapter 13 context under section 1325(a)(4), the current monthly income of the "debtor and the debtor's spouse combined" does not include the income of a nondebtor spouse that is not used for household expenses of the debtor or the debtor's dependents. *See* Collier on Bankruptcy 1325.11[4][d] and cases cited therein. To the extent the United States trustee or a creditor would want to challenge the reasoning of those decisions, they can do so, but the form should not take a position on an issue of substantive law that has been rejected by virtually every court to consider it. At a minimum, the form should allow the debtor the option of taking the "marital deduction" prior to calculating whether current monthly income is above or below the applicable median income.

As to the Form 22A-1 supplement, if a debtor completes this form, at least if a verification is added to it, why is Form 22A-1 necessary at all?

In addition, NACBA does not believe an exemption from the means test for reservists or National Guard members is only temporary. The means test applies to the date of the petition. Moreover, in general, the law applicable to a case is the law in effect on the petition date. Congress intended those exempted from the means test to be permanently exempted. The current form and the new form take a substantive legal position that this was not the intent of Congress.

<u>USC-RULES-BK-2013-0001-0056</u> – <u>Scott Ford, Bankruptcy Clerks Advisory Group</u> (<u>BCAG</u>) – The BCAG recommends that, in all instances on these forms, the references to "ask for help at the clerk's office of the bankruptcy court" should be replaced with "consult your attorney." Employees of the clerk's office are prohibited from giving legal advice to the public. Suggesting that debtors ask for help from the clerks' office staff in completing forms may violate 28 U.S.C. § 955, which prohibits clerks, deputies and assistants from practicing law in any court of the United States. Providing legal advice by the clerk's office may also violate state restrictions on the practice of law.

April 22-23, 2014 Page 528 of 596

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Proposed Forms 22A-1 through 22C-2 are the Chapter 7 Statement of Current Monthly Income, Statement of Exemption from Presumption of Abuse, Chapter 7 Means Test Calculation, Chapter 11 Statement of Current Monthly Income, Chapter 13 Statement of Current Monthly Income and Calculation of Commitment Period, and Chapter 13 Calculation of Disposable Income. The NCBJ believes these proposed forms represent a substantial improvement over the existing forms. They are considerably easier to read and understand, making them easier for debtors to complete and more transparent to creditors and trustees.

April 22-23, 2014 Page 529 of 596

Official Form 22A-2

<u>USC-RULES-BK-2013-0001-0042 – Henry Sommer, NACBA</u> – Form 22A-2 continues the error in the current forms of providing no space for a debtor to deduct other expenses permitted by the Internal Revenue Service Expense standards (IRS Manual § 5.15.1.10) that are necessary expenses "to provide for a taxpayer's and his or her family's health and welfare." These could include such expenses as a security system or household help for an elderly or disabled person unable to perform household chores. At a minimum, the form should allow a debtor to claim "other expenses that I believe I am entitled to deduct", with an explanation of the statutory provision relied upon. Such a line would flag these expenses for anyone who wished to challenge them.

What is meant by "Do not deduct mortgage payments previously deducted as an operating expense in Line 9"? This language does not appear in the current form. We do not understand what this language means or its purpose. If we don't understand it, no pro se debtor will understand it.

<u>USC-RULES-BK-2013-0001-0056</u> – <u>Scott Ford, BCAG</u> – The BCAG recommends that, in all instances on these forms, the references to "ask for help at the clerk's office of the bankruptcy court" should be replaced with "consult your attorney." Employees of the clerk's office are prohibited from giving legal advice to the public. Suggesting that debtors ask for help from the clerks' office staff in completing forms may violate 28 U.S.C. § 955, which prohibits clerks, deputies and assistants from practicing law in any court of the United States. Providing legal advice by the clerk's office may also violate state restrictions on the practice of law.

April 22-23, 2014 Page 530 of 596

Official Form 22B

<u>USC-RULES-BK-2013-0001-0042 – Henry Sommer, NACBA</u> – This form contains the same error as Official Form 22A-1. Although labeled as a calculation of current monthly income, it does not take into account the "marital deduction" necessary to calculate the debtor's current monthly income when there is a nonfiling spouse whose income is listed in Column B. Without that deduction, the calculation simply does not conform to the definition of "current monthly income."

April 22-23, 2014 Page 531 of 596

Official Form 22C-1

<u>USC-RULES-BK-2013-0001-0024 – Chapter 13 Trustees, Central District, CA</u> – The revised form 22C perpetuates an error in the current form by permitting debtors to deduct business expenses before the determination of the applicable commitment period (see #5 of part 1 of the revised form), in clear contradiction of the Bankruptcy Code. This conflicts with current law in a large number of jurisdictions, and is an invitation for the unwary (particularly debtors filing in pro se) to make significant errors in the calculation of the plan term. See In re Weigand, 386 B.R. 238 (9th Cir. B.A.P. 2008).

<u>USC-RULES-BK-2013-0001-0035 – Chapter 13 Trustee Robert G. Drummond</u> – Line 5 of official Form 22C-1 calculates net income from operating a business. That line requires the debtor to list gross receipts and then deduct ordinary and necessary operating expenses. In Drummond v. Wiegand (In re Wiegand), 386 B.R. 238 (9th Cir. BAP. 2008), the Ninth Circuit Bankruptcy Appellate Panel correctly determined that calculation of current monthly income does not allow a deduction for business expenses. Business expenses are appropriately deducted after the calculation of current monthly income when calculating projected disposable income. II U.S.C. § 1325(b)(2)(B) specifically references "disposable income" as being current monthly income less expenditures "necessary for the continuation, preservation, and operation of such business." The proposed form does not follow the statute.

<u>USC-RULES-BK-2013-0001-0042</u> – <u>Henry Sommer, NACBA</u> – The language added in Part 3 of the form in response to the Lanning decision covers a time period that is far too long. It is neither possible nor appropriate to predict the debtor's financial situation several years into the case based simply on a change in 1 or 2 items of income or expense. By the time the listed change occurs, other elements of the means test calculation will also have changed. For example, it is virtually certain that the Internal Revenue Service expense standards will change every year. Thus, simply changing the debtor's calculated disposable income based on a reduction in one or two items of expense would be comparing apples and oranges. In times of high inflation, it would assume the debtor's other expenses have held constant, when in fact the debtor may be less able to make payments because other expenses have increased. As the Supreme Court held in *Ransom*, the appropriate mechanism for dealing with an anticipated change, such as the debtor's car payments ending, is plan modification, rather than prognostication at the time of confirmation. "If car payments cease during the life of the plan, just as if other financial circumstances change, an unsecured creditor may move to modify the plan to increase the amount the debtor must repay. See 11 U.S.C. § 1329(a)(1)." Ransom v. FIA Card Servs., N.A., 131 S. Ct. 716, 729 (2011).

If language is added to reflect known or virtually certain changes in income and expenses, it should be limited to the 1-year period after the order for relief. That is the most that could possibly be predicted with any accuracy, and after 1 year it is already virtually certain that the Internal Revenue Service expense standards will have changed, making unfair and inaccurate a simple adjustment to 1 or 2 items without considering other elements of the calculation.

<u>USC-RULES-BK-2013-0001-0056 – Scott Ford, BCAG</u> – The BCAG recommends that, in all instances on these forms, the references to "ask for help at the clerk's office of the bankruptcy

April 22-23, 2014 Page 532 of 596

court" should be replaced with "consult your attorney." Employees of the clerk's office are prohibited from giving legal advice to the public. Suggesting that debtors ask for help from the clerks' office staff in completing forms may violate 28 U.S.C. § 955, which prohibits clerks, deputies and assistants from practicing law in any court of the United States. Providing legal advice by the clerk's office may also violate state restrictions on the practice of law.

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – The first sentence in the first full paragraph of text at the end of Para 1, part 1 is the same as the comparable sentence in Form 22A-1 and should be revised the same way. The revision would make identical the sentences in Forms 22A-1, 22B, and 22C-1.

The first sentence in the first full paragraph of text at the end of Part 1, Section 1 has an extra word, an extra comma, and one other apparent typographical error. The sentence instructs a debtor to "[f]ill in the average monthly income that you received from all sources, derived during the 6 full months before you file this bankruptcy case." The comma after "sources" should be deleted, as should the word "derived" which is redundant of "during."

The word "file" should be "filed." These changes would make the sentence the same as the comparable sentence in Form 22B. Rewritten, the sentence would read: Fill in the average monthly income you received from all sources, derived during the 6 months before you filed this bankruptcy case."

<u>USC-RULES-BK-2013-0001-0153 – James M. Davis, attorney</u> – Line 5 (and possibly line 6). Like the current Form 22C, proposed Form 22C-1 instructs debtors to include only net business income in the calculation of current monthly income. As the committee is no doubt aware, a number of courts have found that this approach is contrary to the Code. In the chapter 13 context, in fact, most reported decisions seem to hold that debtors should use gross business income in calculating current monthly income. See, e.g., In re Harkins, 491 B.R. 518, 539 n.20 (Bankr. S.D. Ohio 2013) (compiling cases).

Because courts are split on this question, it may be appropriate to leave an option for a debtor to complete the form using net business income. But, especially for the Form 22C-1, I would suggest revisions to the instructions to provide some notice of the split in authority and to permit debtors to fill out the form under the majority approach. For example, line 5 might be revised to instruct the debtor to deduct ordinary and necessary operating expenses only "if you contend these expenses reduce current monthly income under 11 U.S.C. § 101(10A)." And a new line might be added to Form 22C-2 for ordinary and necessary business operating expenses with an instruction not to include any amounts entered on line 5 of Form 22C-1.

April 22-23, 2014 Page 533 of 596

Official Form 22-C2

<u>USC-RULES-BK-2013-0001-0024 – Chapter 13 Trustees, Central District, CA</u> – At #13 "vehicle ownership expense" the form continues to tie the vehicle ownership expense to the IRS Local Standards, in violation of 11 U.S.C. 707(b)(2)(A)(ii)(I) which provides: "... Notwithstanding any other provision of this clause, the monthly expenses of the debtor [to be determined by reference to the IRS Standards] shall not include any payments for debts." (emphasis added).

Vehicle payments are "debts"; these and other secured debts are required to be addressed solely under 11 U.S.C. 707(b)(2)(A)(iii)(I): "The debtor's average monthly payments on account of secured debts shall be calculated as the sum of all amounts scheduled as contractually due to secured creditors in each month of the 60 months following the date of the petition; . . ." There is no authority for linking vehicle payments to the IRS standards. The new form's treatment of vehicle payments creates a phantom exemption that is not permitted under the Code. This analysis follows the Supreme Court's reasoning and holdings in In re Ransom, 131 S.Ct. 716 (U.S. 2011) and In re Lanning, 130 S. Ct. 2464 (U.S. 2010). This provision creates a trap for unwary debtors, particularly those filing in pro se.

<u>USC-RULES-BK-2013-0001-0035 – Chapter 13 Trustee Robert G. Drummond</u> – Part 2, Line 41 of Form 22C-2 allows the debtor to deduct ongoing retirement plan contributions "as specified in 11 U.S.C. § 541(b)(7)." The Ninth Circuit Bankruptcy Appellate Panel in Parks v. Drummond (In re Parks), 475 B.R. 703 (9th Cir. B.A.P. 2012) held that deductions for a debtor's post-petition voluntary 401(k) contributions are not allowed by Section 541(b)(7) when calculating projected disposable income. The proposed Form 22C-1 incorrectly instructs debtors to deduct voluntary contributions to retirement plans when calculating projected disposable income. See also, Burden v. Sea/art (In re Sea/art), 669 F.3d 662 (6th Cir. 2012).

<u>USC-RULES-BK-2013-0001-0056</u> – <u>Scott Ford, BCAG</u> – The BCAG recommends that, in all instances on these forms, the references to "ask for help at the clerk's office of the bankruptcy court" should be replaced with "consult your attorney." Employees of the clerk's office are prohibited from giving legal advice to the public. Suggesting that debtors ask for help from the clerks' office staff in completing forms may violate 28 U.S.C. § 955, which prohibits clerks, deputies and assistants from practicing law in any court of the United States. Providing legal advice by the clerk's office may also violate state restrictions on the practice of law.

<u>USC-RULES-BK-2013-0001-0153</u> – <u>James M. Davis, attorney</u> – The proposed forms' distinction between "household size" and the number of people used in determining deductions from income is an excellent revision. I also agree that the number of tax exemptions is a reasonable reference point for establishing the number of people used in determining deductions from income. I would suggest, however, revising the instructions as follows to remove the implications about the proper number, an issue on which courts disagree:

April 22-23, 2014 Page 534 of 596

"Fill in the number of people you will use below in determining the deductions from income. If this number differs from the number of exemptions on your most recent tax return, please explain why below:

This number may be different from the number of people in your household on Line 16b of Form 22C-1."

Blackline: "Fill in the number of people who could be claimed as exemptions on your federal income tax return, plus the number of any additional dependents whom you support. you will use below in determining the deductions from income. If this number differs from the number of exemptions on your most recent tax return, please explain why below:

This number may be different from the number of people in your household on Line 16b of Form 22C-1."

I would suggest adding check boxes to indicate whether the payments listed on Line 9b includes taxes and insurance. Lines 33d-33f include this information (as does line 47 of the current Form 22C), but payments on Line 9b seem to be the payments for which the information is most relevant.

I think the language in the instruction at line 10 is imprecise. It refers to the possibility that a debtor might claim that the "division of the IRS Local Standard for housing does not accurately compute the amount that applies" to the debtor. This wording implies that the division of the standard "computes" an amount. I think it is more accurate to say that the division affects the computation of the monthly expenses. A suggested revision would be: "If you claim that the U.S. Trustee Program's division of the IRS Local Standard for housing is incorrect and affects the calculation of your monthly expenses, fill in any additional amount you claim."

Blackline: "If you claim that the U.S. Trustee Program's division of the IRS Local Standard for housing does not accurately compute the amount that applies to you, is incorrect and affects the calculation of your monthly expenses, fill in any additional amount you claim."

It would be helpful if the form required debtors to show the calculation when estimating taxes using withholding and estimated tax refunds. A suggested revision would be:

Blackline: "The total monthly amount that you will actually owe for federal, state and local taxes, such as income taxes, self-employment taxes, social security taxes, and Medicare taxes. You may include the monthly amount withheld from your pay for these taxes. However, if you expect to receive a tax refund, you must divide the expected refund by 12 and subtract that number from the total monthly amount that is withheld to pay for taxes. You may estimate this amount using the following calculation:

16a. The monthly amount withheld from your pay for these taxes \$______16b. Your anticipated tax refund, divided by 12: -\$_____

April 22-23, 2014 Page 535 of 596

16c. Estimated tax expense.

Subtract line 16b from line 16a. \$______

If your monthly tax withholding has changed or is virtually certain to change from the amount stated on line 16a, indicate the amount of the change on Line 46.

Note: This instruction does not specify what a debtor should do if the monthly refund exceeds the monthly withholding (whether the debtor should report [sic]

Do not include real estate, sales, or use taxes."

The instructions seem to contain a typo—the final sentence is cut short and may be missing a word. I think it is supposed to say ". . . or any other form of life insurance."

At line 23, I would suggest mirroring the language of the IRS Collection Financial Standards in the 1st sentence of the 2nd paragraph (in particular, adding cable television to the list of telecommunications services that should not be included on line 23). I would also suggest changing "and" to "or." My suggested revision would read: "Do not include payments for residential telephone service, cell phone service, cable television, or internet service."

Blackline: "Do not include payments for basic home residential telephone service, cell phone service, cable television, or internet service and cell phone service."

The numbering for Lines 33a-33g appears to be incorrect. The numbers show up as Lines 3a-3g. On the lines that should be numbered 33d-33f, the check boxes are missing next to the yes and no options (for whether the payments include taxes or insurance).

I would suggest including an itemization for line 41. It would be helpful, at a minimum, to break out the amount claimed as retirement contributions from the amount claimed as retirement loan payments. The statutory basis for these deductions is different, and some courts have reached different conclusions about the permitted deductions. See, e.g., Seafort v. Burden (In re Seafort), 669 F.3d 662, 674 n.7 (6th Cir. 2012).

For the retirement loans, moreover, debtors may need to adjust the deduction for the expected completion of loan repayment during the plan. See id. at 667-74.

It would be helpful to provide instructions for the proper method of doing so. The easiest method would be to instruct debtors to report average monthly retirement loan payments using the same formula as the average monthly secured debt payments (divide the total payments over 60 months by 60). But the basis for this calculation with respect to retirement loans is probably that the loan payoff is known or virtually certain, which suggests that Line 46 is the proper place to adjust for the anticipated repayment completion. It is harder to provide detailed instructions for the adjustment if it should appear on Line 46, but the instructions for Line 41 could at least refer to Line 46.

The breakout on Line 41 would also be helpful for anyone trying to compare the claimed deductions to the actual deductions on a debtor's pay checks. For that purpose, it would be even more helpful to instruct debtors to list deductions for each retirement account and each retirement loan.

April 22-23, 2014 Page 536 of 596

14

I would suggest revising Line 46 so that it provides a final calculation of monthly projected disposable income. That could be accomplished by simply adding a line for the number.

As Line 46 is currently drafted, however, the instructions for that calculation might be complicated. Because Line 46 combines changes to the amounts on Form 22C-1 and Form 22C-2, determining whether a given change on Line 46 represents an increase or decrease to the bottom line requires an algorithm with logic, not just addition and subtraction. That makes the manual calculation slower and makes an automated calculation more complex.

I would suggest providing separate sets of lines for reporting changes to Form 22C-1 and Form 22C-2 and instructing debtors to simply report decreases by filling in negative numbers (removing the check boxes indicating whether the change is an increase or decrease). With that format, the calculation of the projected monthly disposable income would be much more straightforward. The changes to Form 22C-1 would be added to Line 45 of Form 22C-2 and the changes to Form 22C-2 would be subtracted.

April 22-23, 2014 Page 537 of 596

Comments on Official Form 101

<u>USC-RULES-BK-2013-0001-#### – Professor Gibson</u> – Add instruction for attaching small business documents: If you indicate that the debtor is a small business as defined in 11 U.S.C. § 101(51D), you must append the attachments required under 11 U.S.C. § 1116(a)(1).

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – The NCBJ is concerned that the explanatory regarding waiver of the filing fee may create unrealistic expectations on the part of debtors about the probability that the fee will be waived. The NCBJ suggests adding additional text so that the explanation will read:

By law, a judge may, but is not required to, waive your fee, and may do so only if your income is

The explanation of what constitutes a "sole proprietorship" would be more accurate if it referred to a sole proprietorship as a business "operated" by the debtor rather than one "owned" by him. Since a sole proprietorship is indistinguishable from the debtor himself, it is not correct to say the debtor "owns" it. The distinction to be drawn is instead between the debtor doing business under an assumed name and doing business through an artificial business entity. The suggested change is:

A sole proprietorship is a business you own operate as an individual, rather than and is not a separate legal entity such as a corporation, partnership, or LLC.

The examples under the initial question should be expanded to include an example more common than perishable goods or hungry livestock: a building that urgently needs repairs. The suggested change is:

For example, do you own perishable goods, or livestock that must be fed, or a building that needs urgent repairs?

On page 5, the third paragraph under the third checkbox says, "Your case may be dismissed if the court is dissatisfied with your reasons for not receiving a briefing before you file *this bankruptcy filing package*." (Emphasis added). The NCBJ objects strongly to the phrase "bankruptcy filing package," used not only here but elsewhere in the proposed forms. Whether an attempt to be colloquial or informal, the phrase appears nowhere in the Code or the Rules, is inaccurate (debtors file cases, petitions, schedules, statements, and so on but not "packages"), and diminishes the importance of the documents a debtor files with the court. The phrase should be replaced by the words "this bankruptcy case" or perhaps "this bankruptcy petition." The same change is necessary in the identical paragraph for "Debtor 2."

In Part 7, the debtor is required to declare that he understand the criminal penalties for making false statements. The phraseology used here appears in other forms where it is sometimes worded as it is in Part 7 and sometimes not. In the interest of both style and consistency, the

April 22-23, 2014 Page 538 of 596

language should be modified – and should be modified in every form where it appears – to read as follows:

I understand that if I make a false statement, I could be fined up to \$250,000, or imprisoned up to 5 years, or both.

In the next paragraph, a debtor filing under chapter 7 is obliged to state that he is "aware that I may proceed under Chapter 7, 11, 12, or 13 of title 11, United States Code, understand the relief available under each such chapter, and choose to proceed under Chapter 7."

Although this text appears in the current form, the NCBJ suggests that it be narrowed or deleted. For many debtors, the statement will be untrue: a debtor without regular income, for example, is eligible to file a chapter 7 case but not a case under chapter 13. Pro se debtors, moreover, are unlikely to understand the relief available under other chapters and cannot truthfully make the statement. No purpose is served in forcing them to make it. Also, the statement is redundant. The debtor is required to state that he has obtained and read the section 342(b) notice (which the clerk of the court is obligated to provide). That notice contains a description of the other chapters.

Untitled Page for Pro Se Debtors: Following the signature page, finally, there is an additional, untitled form that pro se debtors must sign. The form is obviously designed to encourage debtors to consider obtaining counsel. The NCBJ supports the use of the form but has a few suggestions:

- (1) The form should be titled.
- (2) The instruction again refers to a "bankruptcy filing package." The phrase should be changed to "this bankruptcy case" or "this bankruptcy petition."
- (3) In the 2nd and 3rd paragraphs, the passive voice should be replaced with the active voice as follows:

For example, your case may be dismissed the court may dismiss your case because you did not file a required document, pay a fee on time, attend a meeting or hearing, or cooperate with the court, case trustee, U.S. trustee, bankruptcy administrator, or audit firm if your case is selected for audit. If that happens, you could lose your right to file another case, or you may lose protections, including the benefit of the automatic stay.

If you do not list a debt, the court may deny your discharge. The debt may not be discharged.

Proposed Forms 101A and 101B address the debtor's opportunity under section 362(l) of the Code, 11 U.S.C. § 362(l), to remain in possession of a leased residence when the landlord has obtained a prepetition judgment for possession, notwithstanding section 362(b)(22). Form 101A is intended for a debtor who wishes to remain in the property for 30 days post-petition. Form 101B is intended for a debtor who wishes to remain there beyond the 30 days. Since the forms will typically be used together, they are best discussed together.

The NCBJ believes both proposed forms need changes.

April 22-23, 2014 Page 539 of 596

Form 101A should be redesigned to permit the form to perform 2 required functions: (1) provide a place for a debtor to indicate that he has a pre-petition judgment against him, and (2) provide a way for the debtor comply with the requirements for remaining in the property during the first 30 days post-petition. As now drafted, Form 101A only performs the second function. A debtor is instructed to fill out the form "only if" 3 conditions are met. The 3rd condition is that "you want to stay in your rented residence after you file your case for bankruptcy." But section 362(1)(5)(A) of the Code requires a debtor to indicate "on the bankruptcy petition" whether "there was a judgment for possession of the residential real property in which the debtor resides" regardless of whether the debtor wants to remain in the property. Form 101A does not offer the debtor a place to give the required statutory indication.

The NCBJ recommends redesigning Form 101A to divide it into 2 clearly separate portions, one where a debtor can indicate the existence of the judgment and a second where a debtor can make the certifications necessary to remain in the property.

Forms 101A and 101B should also be designated to indicate more plainly – in large, bold type – the different functions of each form.

Second, the wording of the certifications in both Forms 101A and 101B should be changed. The second sentence in each form requires the debtor to certify that he has the right to stay in his residence by paying the landlord "the entire amount I owe." The phrase is inconsistent with section 362(l) which uses the phrase "entire monetary default." See 11 U.S.C. § 362(l)(1)(A), (2). The "entire amount a debtor owes" might be misinterpreted to mean the entire amount to be paid under the lease itself, including future amounts not yet due. To avoid misinterpretation and make the provision consistent with the statutory language, the NCBJ suggests replacing the words "I owe" with "in default" as follows:

"Under the state or other nonbankruptcy law that applies to the judgment for possession (eviction judgment), I have the right to stay in my residence by paying my landlord the entire amount I owe in default.

Third, the first line in the text immediately below the signature lines is missing a word. The sentence should be written as follows:

If you checked both boxes above, signed the form to certify that both apply, and served your landlord with a copy of this statement...."

Fourth, the NCBJ believes the instructional note in bold below the line at the bottom of Form 101A should be turned into an expression of the debtor's awareness of what he must do to take advantage of the statute. The note should be reworded to say:

"I am aware that if I wish to stay in my residence ...I must pay the entire amount in default ...and file Official Form 101Bwithin the 30 days after the petition." That statement should be relocated to the certification portion, perhaps above the name of the landlord, and put in its own box with a check box next to it. The debtor would then have to check a box indicating his

April 22-23, 2014 Page 540 of 596

awareness of the law's requirements. Even if this change is not made and the instructional note is left where it is, the note has the same faulty phrase identified above: "entire amount I owe." That phrase should be changed to "entire amount in default."

April 22-23, 2014 Page 541 of 596

Comments on Official Form 106Sum

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Proposed Form 106Sum is the debtor's summary of assets and liabilities and certain statistical information. The NCBJ has a single stylistic suggestion. In Part 2, the verb in sections 2 and 3 should be the same. Section 2 currently requires an amount for "Creditors Who Hold Claims Secured By Property." Section 3 requires an amount for "Creditors Who Have Unsecured Claims." The verb should either be "hold" in both or "have" in both, whichever is preferred.

April 22-23, 2014 Page 542 of 596

Comments on Official Form 106A/B: Property

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 106A/B requires the debtor to list property. The proposed form consolidates and amends Forms 6A and 6B. The NCBJ believes the proposed form's instructions are confusing. The confusion stems from the different terms the instructions use for what a debtor must list. The form is entitled: "Property." The first sentence in the instructions requires a list of "items" (a term that arguably excludes collections). The next sentence tells a debtor to list "an asset" only once. In the space of a few lines, in other words, a debtor is told to list "property," "items," and "assets." An effort should be made to arrive at a uniform term. Of the possibilities, "property" is preferable since that is the term that appears in section 541(a) of the Code, 11 U.S.C. § 541(a).

The first sentence of the instructions is also inaccurate. A debtor is told to list and describes items "worth more than \$500." The origin of the \$500 threshold is not explained and is inconsistent with section 541(a), which contains no dollar amount in defining the "property" that makes up the bankruptcy estate, as well as with Rules 1007(b)(1)(A) and 1007(h), which also contain no dollar limit. The NCBJ recommends eliminating the \$500 threshold and instructing a debtor simply to list "all legal and equitable interests in property."

<u>USC-RULES-BK-2013-0001-0089 – William Doyle</u> – I am writing to comment on proposed Official Bankruptcy Form 106A/B. I am a lawyer in Mobile Alabama and a majority of my practice consists of representing credit unions in chapter 13 proceedings. My issue with Form 106A/B is with Part 2 pertaining to motor vehicles. Typically, debtors tend to undervalue vehicles if there is a chance to strip the lien. From my standpoint, there are two important considerations:

- (1) what is the probable value of the vehicle as of the filing date and,
- (2) what is the probable rate of depreciation of the vehicle (important for adequate protection payment amount computation).

Knowing the vehicle's actual mileage at the time of filing is important for both determinations. If you know the mileage on the vehicle at the time of filing, you can run an accurate NADA estimate of the value for that date of a vehicle in normal condition. You can also subtract the mileage on the vehicle at the time of the issuance of the certificate of title and determine the average miles per month the vehicle has been driven. You can then utilize that information to estimate the mileage on the vehicle at any time in the past - say 12 months prior to filing. You can run a reasonably accurate NADA estimate 12 months prior to filing, subtract it from the date of filing NADA value and divide the result by 12 which gives you a fairly accurate determination of the monthly loss in value over the 12 month period ending with the filing date. Eureka! You now have the information you need to make a reasonable determination of the two values you absolutely must have to properly represent your client.

Form 106A/B, however, only requires the debtor to check a mileage box which is divided in 25,000 mile increments. The problem is that there is a vast difference in value between a vehicle

April 22-23, 2014 Page 543 of 596

that has, say 0 miles and one that has 24,999 miles and the same can be said for any of the other 25,000 mile gaps. Therefore, the information is quite useless.

Why not have the form require the entry of the actual mileage? It would require that only one line of data be entered. It would probably result in more accurate value listings by debtors. It would make it easier for creditors' counsel and debtors' counsel to reach agreement on confirmation objections in chapter 13 proceedings and reaffirmation details in chapter 7 proceedings. It would reduce the number of times the Court had to get involved in the issue. It would also allow chapter 7 trustees to know of the probable existence of equity from the date the schedules are filed.

<u>USC-RULES-BK-2013-0001-0116 – Stanley Kartchner</u> – I particularly concur with the NCBJ's discussion and recommendations concerning the \$500 asset disclosure threshold stated in the instructions to Form 106A/B (see page 15 of the NCBJ's comment) and the 100% FMV exemption election in Form 106C (see pages 15-17 of the comment).

Additionally, I note that for trustees who are uniquely tasked with reviewing every page of each set of schedules and statements of financial affairs filed in assigned cases, the ease in which that information is accessed is of utmost importance. Therefore, for instance, being able to readily see which assets are affected by liens or exemptions is helpful to the trustee in determining whether an asset can or should be administered for the benefit of creditors.

For this reason, I suggest that for each listed asset there be added two data fields that tie into the information contained in proposed Forms 106C and 106D, either by dollar amount if such is available (which is especially helpful with real property and personal property assets such as vehicles), or at least a check box to indicate that there is a lien or exemption affecting the particular asset.

<u>USC-RULES-BK-2013-0001-0135 – William Pierce</u> – I concur with the comments, concerns and recommendations of NCBJ especially the one regarding the \$500 asset disclosure.

It is very important to keep the asset showing a lien on Schedule D tied to Schedule A and B assets that have liens as it is presently. That is a real time saver when reviewing Petitions.

USC-RULES-BK-2013-0001-0151 – National Association of Bankruptcy Trustees (NABT)

– In the scheduling of real property interests the form has eliminated information regarding both the Nature of the Debtor's Interest, and the Amount of Secured Claim as currently provided in Official Form B6A. The elimination of the nature the property interest and lien information diminishes the expediency with which Chapter 7 Trustees can assess whether there is an asset to administer without reference to other Schedules or independent documents.

NABT also joins the comments by NCBJ relative to the direction to list items "worth more than \$500," and simply list "all legal and equitable interests in property."

April 22-23, 2014 Page 544 of 596

NATURE OF DEBTOR'S INTEREST IN PROPERTY For instance, in a case of a married individual filer residing in a state that recognizes tenancy by the entirety ownership, if all of the obligations are solely the Debtor's, this information may be dispositive of the case administration of entirety assets. The designation under "Who is the owner of the property?" fails to address the legal nature in which the interest is owned.

AMOUNT OF SECURED CLAIM The failure to list secured indebtedness against scheduled real property will require the Trustee, Creditors or other users to refer to proposed Official Form 106D, whereas existing Official Form B6A provides that immediate snapshot of whether there is equity in the property potentially available for administration.

OTHER INFORMATION YOU WISH TO ADD ABOUT THIS ITEM, SUCH AS LOCAL PROPERTY IDENTIFICATION NUMBER: The form indicates that a property record card number is not a mandatory item and makes this information only a suggestion. In many areas of the country, an actual deed book and page is necessary in order to determine where the property is located. NABT would suggest that this be made mandatory.

VALUATION While Current value is defined in the Instructions as fair market value, it may be helpful to have available space to set out how they determined the current value; namely a block for appraisal, comparative market analysis, personal opinion, etc.

VALUATION FOR VEHICLES it may be helpful to have available space to set out how they determined the current value; namely a block for appraisal, NADA, Kelly Blue Book, personal opinion, etc.

In the scheduling of non-vehicle personal property the form has eliminated information regarding who is the owner of the property, which is provided in Part 1 and 2. The failure to define or identify the owner diminishes both the efficiency and expediency which Chapter 7 Trustees can assess whether there is an asset to administer without reference to independent information, and can directly affect exemption analysis. The inclusion of ownership is necessary in a jointly administered case.

CASH AND DEPOSITS OF MONEY 16 & 17, start with inquiry of the Debtor(s)' ownership of cash and deposit of money. In this continually changing age of electronic life, NABT would suggest that the proposed Form include under Cash additional language to address electronic or other forms of cryptocurrency or digital currency; namely Bitcoins, Ripple, Litecoin, Zerocoin, etc.

As to the Deposits of Money, in addition to the institutional name it is necessary to have the identifying account number (the last 4 digits) as to insure all bank statements that are provided to the Trustee as part of the Rule 4002, or the corresponding Local Rule requirements are accounted for.

Once again Part 4 fails to provide for declaring ownership as between Debtor 1 and Debtor 2, which needs to be provided.

April 22-23, 2014 Page 545 of 596

BONDS, MUTUAL FUNDS, OR PUBLICLY TRADED STOCKS; RETIREMENT OR PENSION ACCOUNTS; ANNUITIES & EDUCATIONAL IRA Questions inquires of the Debtor(s) ownership of Bonds, etc. and Retirement Accounts. Again, in addition to the institutional name it is necessary to have the an identifying account number (the last 4 digits) as to insure all accounts, and statements that are provided to the Trustee as part of the Rule 4002, or the corresponding Local Rule requirements are scheduled.

Further, the number of shares held of publically held stocks would be helpful, when not part of a brokerage account.

INTERESTS IN INSURANCE POLICIES form inquires of the Debtor(s) ownership of Interests in insurance policies. Again, in addition to the institutional name it is necessary to have the an identifying account number (the last 4 digits) to initiate verification or inquiry.

CLAIMS AGAINST THIRD PARTIES, WHETHER OR NOT YOU HAVE FILED A LAWSUIT OR MADE A DEMAND FOR PAYMENT OTHER CONTINGENT AND UNLIQUIDATED CLAIMS OF EVERY NATURE, INCLUDING COUNTERCLAIMS OF THE DEBTOR AND RIGHTS TO SET OFF CLAIMS Questions inquire of the Debtor(s) ownership of claims. The Debtor should be required to provide the name of the counsel handling the claim and that counsel's contact information including address and telephone number. While this can be perceived as solely assisting the Trustee, by requiring this information, Debtor(s)' counsel will have additional impetus to include this in the due diligence inquiry prior to filing, which ultimately will improve the level of both representation and disclosure.

FARM AND FISHING EQUIPMENT AND IMPLEMENTS Question inquires of the Debtor(s) ownership of Farm and Fishing Equipment. While Part 5, Question 40, in the description of business-related property inquires of "machinery, fixtures, equipment, supplied you use in business, and tools of your trade." Question 49 is limited only to "equipment." As to consistency, a similar inquiry should be made of machinery, fixtures, etc. in Question 49.

<u>USC-RULES-BK-2013-0001-0153 – James M. Davis, attorney</u> – The committee notes indicate that the committee elected to avoid terms such as "life estate" and "joint tenancy" because debtors often do not fully understand the nature of their ownership interest in real property. This information, however, may be important. I would suggest at least revising the "other information" field to add a prompt for this type of information if known. For example: "Please describe the nature of your ownership interest (such as tenancy by the entirety or life estate), if known."

I would prefer to retain the secured claim information that is required on the current Schedule A. In fact, I think it would also be helpful to carry over exemption information onto Schedule A/B as well, so that the Schedule would provide a straightforward statement of the non-exempt equity the debtor holds in property. Under the current forms (and the proposed revised forms), one must flip between schedules A through D to compile this information.

April 22-23, 2014 Page 546 of 596

Comments on Official Form 106C

<u>USC-RULES-BK-2013-0001-0042 – Henry Sommer, NACBA</u> – The new language in this Schedule attempting to resolve the issues raised by the Supreme Court's *Schwab v. Reilly* decision fails to accomplish its goal. The purpose of Rule 4003 is to provide a procedure for exemptions to be finalized at an early date so the debtor can know what property can be dissipated, sold or otherwise used free from any claim of the bankruptcy trustee. In many parts of the country, prior to *Schwab*, it had always been assumed that if the trustee believed the value of the debtor's interest in an asset was more than the debtor could exempt, the trustee had a duty to object to the claim the asset was exempt. In *Schwab*, the Supreme Court held that there was no duty to object if the claim of exemption was ambiguous, and suggested that a debtor could make clear an intent to exempt the debtor's entire interest in an asset by adding language stating the debtor exempted 100% of fair market value. There could then be no doubt that the trustee should object if the trustee believed the value of the debtor's interest exceeded the exemption the debtor could claim. If the trustee did not object, the debtor would have finality on whether the debtor's interest was exempt.

The new language does not accomplish the purpose intended by the Court or by Rule 4003. By limiting the claim of exemption to "100% of fair market value, up to the statutory limit" the form does exactly the opposite. For every single asset subject to a monetary exemption limit, there is at least a theoretical, and often a real, possibility that a trustee could later claim that a particular asset, or collection of assets, like household goods, exceeded the statutory limit. In asset cases, this possibility would exist until the case was closed, often a year or even several years. During that time, a debtor disposing of or otherwise dissipating an asset would be subject to possible sanctions or even a revocation of discharge if those assets were later claimed by the trustee. There would be no finality on exemptions, thwarting the purpose of an objection deadline. The only way for a debtor to avoid this outcome would be to file a motion for abandonment of the property, completely reversing the burdens of going forward and proof set forth in Rule 4003.

We are aware that trustees do not like having a deadline to object to valuation. They have argued that they will have to object to every exemption, an absurd argument that in no way reflects the experience in districts where there was a long history of a firm deadline for objections on valuation issues. Trustees objected only when there was a reasonable argument that the value of an asset exceeded the exemption limits.

They have also argued that the time limits for objections would cause them to lose assets, but they have control over those time limits. The deadline runs from the conclusion of the meeting of creditors. If a trustee wishes to investigate the value of an asset, the trustee can adjourn the meeting for that purpose, thus delaying the running of the deadline. The trustee also can seek an extension of the deadline from the court, which would routinely be granted for any reasonable cause.

And trustees have argued that debtors would just claim 100% of every asset as exempt, regardless of its value. This argument has no basis in past experience, makes assumptions about debtor honesty that are inconsistent with experience in the vast majority of cases, and ignores the

April 22-23, 2014 Page 547 of 596

fact that virtually every debtor with significant assets is represented by attorney bound by Rule 9011. Moreover, if a debtor did fraudulently claim an exemption, Rule 4003(b)(2) already gives the trustee a lengthy extended period to object.

NACBA suggests that the schedule list:

- 1) the value of the debtor's interest in the property on the petition date (see § 522(a)(2))1
- 2) the amount of that interest the debtor claims exempt and
- 3) whether the debtor asserts that the value claimed exempt is 100% of the value of the debtor's interest in the asset.

All three of these are necessary because when several items are claimed within a dollar-limited category, such as household goods or a wild card, the trustee should know how that dollar limit has been allocated among the items claimed exempt, and not be required to object to the total of all items in a category if only one item's valuation is being challenged. We also suggest that Rule 4003(b) be amended to specifically provide that objections to exemptions include objections based on valuation of the asset claimed exempt.

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Proposed Form 106C amends Form 6C and allows the debtor to claim certain property as exempt.

The most significant changes are designed to respond Schwab v Reilly, ___U.S.___, 130 S. Ct. 2652 (2010), in particular the advice the Court offered if a debtor wants to claim 100% of the property's value as exempt.

Part 1, line 2 has a 3rd column in which the debtor must state the "[a]mount of the exemption you claim." There are two check boxes giving the debtor alternatives. Either the debtor can check the first box and provide a dollar amount, or he can check the second box next to which it says: "100% of fair market value, up to any applicable statutory limit." Other aspects of the proposed form and related instructions are built around those choices.

The NCBJ believes that the proposed form and its instructions conflict. In the NCBJ's view, the instructions better implement the spirit of Schwab. Schwab adopted a relatively straightforward, "what-you-see-is-what-you-get" approach to a debtor's claimed exemptions. Under that approach, the debtor must express what he wants, leaving no room for interpretation. When a debtor wants to exempt the entire value of an asset, Schwab tells the debtor to say so directly with an indication like "'full fair market value (FMV)" or '100% of FMV." See Schwab, ____

April 22-23, 2014 Page 548 of 596

26

¹ The proposed form deviates from the current form by requesting the value set forth in Schedule A/B, which does not take into account liens on the property. The value on the exemption schedule should be the value of the debtor's interest in the property after deduction of liens, so it can more easily be compared to the allowable exemptions. The proposed form would confuse pro se debtors who might believe that a house subject to a mortgage could not be exempted because its total unencumbered value exceeded the homestead exemption.

U.S. at _____, 130 S. Ct. at 2668. Schwab's advice accords well with the short deadline for objections to exemptions, the significant consequences of not objecting, and the need to make expeditious and economical decisions about the administration of the bankruptcy estate. If the debtor follows Schwab, the reader can quickly determine whether a claimed exemption is proper without any additional information.

The applicable nonbankruptcy law authorizing the exemption will either contain a dollar limit or not. If it does, the claimed exemption must fit within that limit or be objectionable. An exemption claimed as "100% of FMV" will be readily identifiable as potentially objectionable because it could exceed the dollar limit. If an exemption has no dollar limit, a "100% of FMV" option is useful. If the exemption does have a dollar limit, however, the debtor must give a dollar figure for the claimed exemption so that the exemption claimed can be compared to that limit.

The instructions for Proposed Form 106C (page 16 of the "Instructions for Selected Forms"[p. 214, 2nd column, 4th para]) are consistent with Schwab. In describing how a debtor should list the amount of each exemption, the instructions state: "Usually a specific dollar amount is claimed as exempt, but in some circumstances, the amount of the exemption claimed might be indicated as 100% of fair market value. For example, a debtor might claim 100% of fair market value for an exemption that is unlimited"

The instructions thus suggest that an exemption will usually be claimed in a dollar amount, but if the exemption is unlimited, 100% of FMV may be used.

The instructions in the proposed form (in bold type, preceding Part 1) similarly state that "[s]ome exemptions ...may be unlimited in dollar amounts," implying that "100% of fair market value" is an option only with exemptions of that kind.

The substance of the proposed form, on the other hand, contradicts this concept. The 2nd check box in Part 1, line 2, allows a debtor to claim as the amount of the exemption "100% of fair market value, *up to any applicable statutory limit.*" (Emphasis added). The proviso appears to contemplate some type of hybrid limited/unlimited exemption.

The instructions in the form seem to contemplate the same thing, telling a debtor that if he claims 100% of fair market value under an exemption with a dollar limit and the property turns out to be worth more, the exemption "would be limited to the applicable statutory amount." The NCBJ believes this scheme is needlessly confusing and will lead to increased litigation over exemptions. Taken together, the instructions and the 2nd check box encourage debtors always to claim the "100% of FMV, up to any applicable statutory limit" option because the only negative consequence will be to scale back an excessive claim. Litigation is particularly likely in jurisdictions that allow people to exempt a broad category of property (such as tangible personal property) and place no dollar limit on specific types of property (such as motor vehicles) within that category.

Take for example a debtor who lives in a state that offers an \$8,000 combined exemption for tangible personal property and non-residential real estate. The debtor owns a car worth \$8,000, furniture and household goods worth \$8,000, and an \$8,000 time share, all of which he wants to

April 22-23, 2014 Page 549 of 596

claim as exempt. If he checks the second box for each item – "100% of fair market value, up to any applicable statutory limit" – no individual exemption will be objectionable because none exceeds the statutory cap although taken together they do. When the exemptions are separately proper but collectively improper, it is impossible for a trustee or creditor to say which is objectionable or know which will be limited without obtaining an appraisal of each item claimed as exempt. That will be prohibitively expensive in many cases and will also be difficult to do in the short time for objecting to exemptions.

To avoid these problems, the NCBJ recommends requiring a debtor to state a particular dollar amount for an exemption that has a dollar limit and restricting the 100% of fair market value option to exemptions that are unlimited. After the 2nd check box, the phrase "up to any applicable statutory limit" should be deleted and a parenthetical added, as follows:

"100% of fair market value, up to any applicable statutory limit (for exemptions unlimited in dollar amount)

The instructions (in bold type, preceding part 1) should also be changed to delete the last sentence:

Some exemptions – such as those for health aids, rights to receive certain benefits, and tax-exempt retirement funds – may be unlimited in dollar amount. However, if you claim an exemption of 100% of fair market value under a law that limits the exemption to a particular dollar amount and the value of the property is determined to exceed that amount, your exemption would be limited to the applicable statutory amount.

<u>USC-RULES-BK-2013-0001-0106</u> – <u>Annette Crawford, M.D. Louisiana, National</u>
<u>Association of Chapter 13 Trustees (NACTT)</u> – The proposed Schedule C does not include "totals" information needed in order to complete the required Final Account (UST Form 101-13 FR-s(4/1/2009) which requires the following information: "total assets exempted". Staff time involved in having to total assets claimed exempt when they currently don't have to because the existing Schedule C includes the required totals and dovetails into the required Final Account.

<u>USC-RULES-BK-2013-0001-0135 – William Pierce</u> – I totally disagree to the using 100% FMV on Form 106C. Trustees would be objecting to most if not all exemptions as Debtors and Debtors attorneys do not put the real market value on the schedules most of the time. A lot of assets exceed FMV so Trustees will be in the Court with objections constantly.

<u>USC-RULES-BK-2013-0001-0140 – Penelope Souhrada</u> – Part 3 on page 1 asks: Are you claiming a homestead exemption of more than \$155,675? This applies only in jurisdictions which use the federal exemptions. Opt-out states each have different homestead exemptions and using this number in this form will be extremely confusing for debtors.

<u>USC-RULES-BK-2013-0001-0151 – NABT</u> – form addresses the Debtor(s)' claim of exempt property. In 2011, NABT strongly opposed the inclusion of the 100% Fair Market Value (FMV) exemption checkbox with the Proposed Schedule C amendment in 2011. (Comment 11-BK-11).

April 22-23, 2014 Page 550 of 596

While NABT appreciates the efforts of the Advisory Committee on Bankruptcy Rules and its Subcommittee on Forms and the Forms Modernization to continue to address its concerns arising from the United States Supreme Court's decision in Schwab v. Reilly, 130 S. Ct. 2652, 2668 (2010), NABT believes that many of the same deficiencies with the use of a 100% FMV exemption checkbox as commented upon by NABT in 2011 still exist.

OWNERSHIP AS TO DEBTOR 1 & DEBTOR 2 Proposed Form 106C in the scheduling of exemptions has eliminated information regarding who is the owner of the property, which is provided in Part 1 and 2 of Form 106A/B. The failure to define or identify the owner is diminishes both the efficiency and expediency which Chapter 7 Trustees can assess whether there is a proper exemption without reference to independent information, which directly affects exemption analysis. The inclusion of ownership is necessary in a jointly administered case.

AMOUNT OF THE EXEMPTION YOU CLAIM Form provides two alternatives for the amount claimed as exempt. The 1st alternative is a dollar amount (\$_____) while the 2nd is for "100% of fair market value, up to any applicable statutory limit."

The form of the 100% FMV checkbox for dollar limited exemptions will only create confusion and increased litigation over exemptions as described in the NCBJ Comment BK-2013-0001-0059 at pages 15-17.

The 30–day time limit in Bankruptcy Rule 4003 only applies to objections based on "three, and only three" entries on a debtor's claim of exemptions in Schedule C: (1) the description of the property claimed as exempt; (2) the Bankruptcy Code provisions governing the exemption claimed; and (3) the amount listed in the column titled "value of claimed exemption." Schwab v. Reilly, 130 S. Ct. 2652, 2663 (2010) Where the basis of the objection is on any other entry, such as the fair market value assigned to the exempt property by the debtor, the 30–day time limit does not apply. Id. at 2665 n. 15.

As reviewed by Bankruptcy Judge J. Rich Leonard in the case of In re Gregory, 487 B.R. 444, 451 (Bkrtcy.E.D.N.C. March 7, 2013):

The validity of utilizing a designation such as "100% of the FMV," "100% Equity" or comparable language to value exemptions (the value of which are expressly limited to a maximum value by the Bankruptcy Code or applicable state law), has been addressed by several courts following Schwab. See, e.g., In re Orton, 687 F.3d 612, 618 n. 1 (3d Cir. 2012); Mwangi v. Wells Fargo Bank, N.A., 473 B.R. 802, 810 (D. Nev. 2012); Hefel v. Schnittjer (In re Hefel), No. 11–CV–1010–LRR, 2011 WL 3292929, at *4–5 (N.D. Iowa July 29, 2011); Messer v. Maney (In re Messer), Nos. AZ–11–1505, 11–3007, 2012 WL762828, at *3–4 (B.A.P. 9th Cir. Mar. 9, 2012); Massey v. Pappalardo (In re Massey),465 B.R. 720, 730 (B.A.P. 1st Cir. 2012) (concluding "that there is no legitimate reason for the Debtors to ignore the burdens imposed by the [Bankruptcy] Code, and reiterated by Schwab, to state their exemptions accurately and in conformance with statutory limits, by identifying the value of the claimed exemption up to or in a specific dollar amount.");***

April 22-23, 2014 Page 551 of 596

Unless applicable federal or state law allows debtors to exempt their entire interest in an item itself regardless of value, they have no right to utilize language claiming an exemption equal to 100% of the fair market value. See, e.g., Orton, 687 F.3d at 618 ("[W]here the statutory basis for a debtor's claim of exemption provides only for an exemption of an interest in certain property up to a specific dollar amount, the 'value of the claimed exemption' must be identified as a monetary value." (citations omitted)); Hefel, 2011 WL 3292929, at *4 (affirming the bankruptcy court's order sustaining the objections of the trustee to the debtor's exemption of certain business interests as "FMV"); Massey, 465 B.R. at 726; Messner, 2012 WL 762828, at *4; Winchel, 2010 WL 5338054, at *1–2 (sustaining the trustee's objection to the debtors' valuation of the claimed exemption in their residence as "Full FMV" because the value of the residence (\$200,000.00) exceeded the statutory limit set forth in § 522(d)(1) of the Bankruptcy Code).

In 2012, shortly following the testimony provided by NABT to the Advisory Committee in support of its Comment 11-BK-11, the First Circuit bankruptcy Appellate Panel issued its opinion in the case of In re Massey, 465 B.R. 720 (B.A.P. 1st Cir. 2012). In Massey, the Court in considering the use of "100% of FMV" claim of exemption, stated:

The foregoing interpretations of Schwab are supported by the Supreme Court's own admonition that "the only burdens [its] conclusion imposes are burdens the Code itself prescribes, specifically, the burdens the Code places on debtors to state their claimed exemptions accurately and to conform such claims to statutory limits." 130 S. Ct. at 2667 n.17. Even Moore, supra, which the Masseys describe in their Amended Brief as having reached a "sensible result," declined to interpret Schwab as broad license to exempt an asset in-kind where the authorizing statute caps the exemption. Instead, in ruling on an objection to a "100% of FMV" exemption, the court concluded that the trustee was entitled to an evidentiary hearing regarding the value of the debtor's exemptions. In re Moore, 442 B.R. at 868. At such a hearing, the debtor would have the burden of showing a "plausible basis for the claim that '100% of FMV' of an asset falls within the statutory limit on the amount that may be exempted." Id.

NABT requests that, in the event Proposed Official Form 106C, Part 1 and 2 in Column 3 is to provide a 100% FMV checkbox exemption, the Form should require "a debtor to state a particular dollar amount for an exemption that has a dollar limit and restricting the 100% of fair market value option to exemptions that are unlimited. After the second check box, the phrase "up to any applicable statutory limit" should be deleted" as suggested by the NCBJ in Comment BK-2013-0001-0059 at page 17.

Lastly, as noted in NABT's Comment 11-BK-11, the current Schedule C – Property claimed as exempt form (Official Form 6C) and Proposed Official Form 106C provides only a minimal amount of information. While each version set forth a description of the property, the statutory basis and the amount of the claimed exemption, each fails to provide critical information as to the net value available for administration, and each fails to provide without cross-referencing both liens against such property and property not claimed as exempt. It is NABT's position that this could be easily remedied by the inclusion of a few additional columns, which would clearly set forth what properties and values are subject to administration.

An example of such modification might be as follows:

April 22-23, 2014 Page 552 of 596

Property Description from Schedule A/B	Full market value of the property	Current value of the portion you own	Liens against the full market value	Amount you claim as exempt	Specific laws that allow exemption	Net value to the Estate
2008 Toyota Tacoma	\$ 20,000.00	\$ 20,000.00	\$ 12,000.00	\$	11 USC	\$4,550.00
1995 Chevy Truck	\$ 3,000.00	\$ 1,500.00	\$ -	3,450.00 \$ -	522(d)(2) N/A	\$1,500.00

A modified form as suggested above would allow for quick verification, in summary fashion of the property, the exemption and if there is an administrable asset, which would benefit Debtors, their counsel, as well as the Judiciary and Chapter 7 Trustees.

April 22-23, 2014 Page 553 of 596

Comments on Official Form 106D

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Proposed Form 106D requires debtors to list creditors who hold claims secured by property. The NCBJ recommends both substantive and stylistic changes to the proposed form.

Part 1, line 2 requires a debtor to list secured claims in alphabetical order of "the major creditor" who holds each claim. The term "major" requires debtors to determine which creditors are "major" or "minor" and then apparently not disclose the "minor" ones. The word "major" should be deleted so that all creditors must be listed.

The NCBJ notes that in line 2, there is a checkbox on the lower left side that requires a debtor to "check if this is a community claim." The meaning of the term "community claim" is unclear. The term is not defined in the proposed form, in the Instructions, in the Bankruptcy Code, or in the Bankruptcy Rules. The check box and accompanying text should either be clarified or deleted.

The NCBJ suggests changing the phrase "check all that apply," a phrase that appears several times, to "check any that apply" and deleting the check box and phrase "none of the above apply."

The NCBJ recommends adding a check box for "None" at the beginning of Part 2 of the proposed form and revising the instructions at the beginning of Part 2 as follows:

Use this page only if you have others to be notified about your bankruptcy for a debt that you already listed in Part 1. For example, if If a collection agency or an attorney is trying to collect from you on a debt from you that you owe to someone else, list the creditor in Part 1, and then list the collection agency or attorney here. Similarly, if you have more than one creditor for any of the debts that you listed in Part 1, list the additional creditors here. If there are no you do not have additional persons to be notified for any debts listed in Part 1, do not fill out or submit this page check the "None" box at the top of this page.

<u>USC-RULES-BK-2013-0001-0155 – David S. Yen</u> – Schedules D and E/F Should Not Require That Creditors Be Listed Alphabetically

The revised forms would instruct debtors to list creditors alphabetically. This will not be a problem for attorneys who use commercial software programs, but it will impose an unjustified burden on pro se debtors. These debtors have a hard enough time preparing their schedules without adding this requirement.

At the pro se help desk we will frequently see debtors who think that they have listed all of their creditors, but who have omitted creditor(s). They may have forgotten that a debt was owed, or mistakenly thought that a creditor did not have to be listed because the debtor does not believe that the debt is owed at all. Other times the debtor has included a debt on Schedule D because it was secured when the debt was incurred, but it is no longer secured because the collateral has

April 22-23, 2014 Page 554 of 596

been repossessed. To make the debtor redo the schedules just so that the creditors would appear in alphabetical order is a burden that is not justified by any possible benefit.

Schedules D and E/F Should Not Require That Creditors Be Listed Repeatedly – Account Numbers Should be Optional

Schedule E/F part 2, question 4 tells the debtor: "If a creditor has more than one priority unsecured claim, list the creditor separately for each claim. For each claim listed, identify what type of claim it is." Many debtors receive separate bills from hospitals or other medical providers for every visit. Some creditors assign a different account number for each visit, even though they also have a unique patient identifier number. The burden of listing the creditor separately for each account number, and of sending separate notices for each account, is not justified by the benefit to creditors. Many credit reports list partial account numbers by omitting numbers at the end rather than at the beginning. Account number have a limited value for creditors and listing them should be optional.

April 22-23, 2014 Page 555 of 596

Comments on Official Form 106E/F

<u>USC-RULES-BK-2013-0001-0042 – Henry Sommer, NACBA</u> – The instruction paragraph at the top of this schedule uses the word "priority" twice where the word "nonpriority" should be used.

In addition, there is no need to list separate claims of the same creditor separately. The purpose of the schedule is to identify creditors and the amount owed to them. If a debtor has purchased several items from the same creditor, the identity of the creditor and the total debt should be sufficient for all parties. Similarly, if the debtor has three Citibank credit card accounts that have not been assigned, listing the total owed is sufficient for the needs of all parties in the case.

Likewise, the approach of the current forms, which makes account number information optional should be retained. We are unaware of problems caused by lack of an account number for creditors, who can use the debtor's social security number, which they insisted on receiving in the notice of the creditors meeting, to identify the debtor.

<u>USC-RULES-BK-2013-0001-0056 – Scott Ford, BCAG</u> – The BCAG suggests removing the language addressing what is needed for "statistical purposes" because this is an internal requirement for the judiciary, and it might create confusion on the part of the debtor.

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 106E/F requires debtors to list creditors with unsecured claims. The proposed form consolidates and amends Forms 6E and 6F. The NCBJ's comments on and proposed changes to Parts 1 and 3 of the proposed form are identical to the NCBJ's comments on and proposed changes to Parts 1 and 2 of Proposed Form 106D.

April 22-23, 2014 Page 556 of 596

Comments on Official Form 106G

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 106G revises Form 6G requiring the debtor to list executory contracts and unexpired leases. In the NCBJ's view, the parenthetical examples of possible contracts or leases are confusing, since no one contracts for or leases "rent" or a "vehicle lease." The examples provided should be parallel, describing an item or premises contracted for or leased. The parenthetical phrase should be changed to: "an apartment, a car or truck, a cell phone."

April 22-23, 2014 Page 557 of 596

Comments on Official Form 106H

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 106H requires a debtor to list his codebtors. The NCBJ believes the proposed form would be clearer if the definition of "co-debtor" were moved from line 3 to become the 1st sentence in the form (before "Be as complete and accurate as possible").

The first sentence of the form should be rewritten as follows: "Co-debtors are people or entities who are also liable for any debts you may have." The 1st sentence in line 3 would then be rewritten simply to say: "In Column 1, list all of your codebtors."

April 22-23, 2014 Page 558 of 596

Comments on Official Form 106Dec

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 106Dec is an individual debtor's declaration that the information in the schedules is true and correct. The 3rd sentence in the proposed form misstates the criminal penalties for a false statement, giving the maximum fine as "\$500,000."

Under 18 U.S.C. § 3571(b)(3), the maximum fine for an individual convicted of a federal criminal offense is \$250,000, not \$500,000. In the interest of consistency, the 3rd sentence should also have the same structure as comparable sentences in other forms. The sentence should be rewritten as follows:

If you make a false statement, you could be fined up to \$500,000 \$250,000, or imprisoned for up to five years, or both."

April 22-23, 2014 Page 559 of 596

Comments on Official Form 107

<u>USC-RULES-BK-2013-0001-Oral comment to C. Kohn, made by Assistant U.S. Attorney</u> – Is "were you a party" to proceedings sufficient, or would it be better to say "were you a plaintiff, defendant, or otherwise a party" to proceedings. She had a bankruptcy fraud referral in which the debtor failed to report that she was the beneficiary of an employment discrimination action prosecuted against her employer by the EEOC. Her suggested language might have provided a better signal to list that action. It might also address class members who are not named plaintiffs in class actions.

It would be clearer to say: "Are you now or, within 1 year before you filed for bankruptcy, were you a party in any lawsuit . . ." rather than "Within 1 year before you filed for bankruptcy, were you a party in any lawsuit . . ."? I realize that we have a check box for "Pending" status which indicates that the question includes current cases, but use of "were" alone seems a little confusing.

<u>USC-RULES-BK-2013-0001-0035</u> – <u>Chapter 13 Trustee Robert G. Drummond</u> – Line 18 requests information regarding transfers to self-settled trusts within the ten years prior to the filing of the bankruptcy case. The Official Form does not request information of property transfers within the last 10 years converting non-exempt property to increase the homestead exemption. The 2005 changes to the Code added 11 U.S.C. § 522(0) which reduces the value of the homestead exemption to the extent that the value is attributable to property disposed by the debtor in the prior 10 years with the intent to hinder, delay, or defraud. An additional question about dispositions that increased equity in the homestead within the prior ten years would be helpful.

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 107 is the debtor's statement of financial affairs and revises Form 7. The NCBJ recommends several changes, most of them stylistic.

Line 3 asks about the debtor's sources of income. The initial question should be changed to read:

Did you have any income from employment being employed or from operating a business ...? The 2nd sentence of line 3 instructs debtors awkwardly to "Fill in a total amount for the income you received" The sentence should be changed to read: Fill in the a total amount of for the income you received

Form 107 contains 2 check boxes labeled "Yes," one under the heading "My debts are not primarily consumer debts," the other under the heading "My debts are primarily consumer debts." In the text next to each "Yes" check box, the debtor is told to list creditors who have received certain payments. The last sentence instructs: "Also do not include payments to an attorney for this bankruptcy case." The phrase "for this bankruptcy case" should be deleted and the following substituted: "if the payments are listed in your answer to question 15 in Part 7." Rewritten, the reworded sentence would read:

April 22-23, 2014 Page 560 of 596

Also, do not include payments to an attorney for this bankruptey case if the payments are listed in your answer to question 15 in Part 7.

10 poses a question about set-offs by creditors but limits the information a debtor must provide to instances where a set-off has occurred "without your permission." In the interest of fuller disclosure, this limitation should be removed so that interested parties can evaluate for themselves whether the set off was improper or otherwise avoidable. Rewritten, the question would read:

Within 90 days before you filed for bankruptcy, did any creditor, including a bank or financial institution, set off or otherwise take anything from your accounts without your permission or refuse to make a payment because you owed a debt?

11 asks about property in possession of various parties. The indefinite article "a" should be added before "custodian" and "other" should be changed to "another," partly to remove the implication that the custodian or official must be "court appointed." Rewritten, the question should read:

Within 1 year before you filed for bankruptcy, was any of your property in the possession of an assignee for the benefit of creditors, a court-appointed receiver, a custodian, or another official?

17 asks whether within 2 years of the bankruptcy filing a debtor transferred property outside the ordinary course of business. The sentence following the question instructs: "Include both outright transfers and transfers made as security." The NCBJ is concerned that the phrase "made as security" will be unclear to lay people. A parenthetical example would lessen the likelihood of confusion. The sentence should be rewritten as follows:

Include both outright transfers and transfers made as security (such as the granting of a security interest or mortgage on your property).

21 requires a debtor to list the location of storage units in which the debtor has stored property within 1 year of the bankruptcy filing. The sentence following the question instructs the debtor:

"Do not include storage units that are part of the building in which you live."

The reason for the instruction is not clear. Often apartments and condo units have storage rooms located in other parts of the building that will not be obvious during a trustee or appraiser's visit. In the interest of fuller disclosure, the NCBJ recommends removing the instruction:

Do you store property in a storage unit, or have you stored property in a storage unit within 1 year before you filed for bankruptcy? Do not include storage units that are part of the building in which you live.

26 asks a debtor to give details about his business or connections with any business. The NCBJ believes a different structure would make the provision clearer.

April 22-23, 2014 Page 561 of 596

The "Yes" check box should be moved to the top and the instruction changed to say "Check any below that apply." The phrase "None of the above applies" after the "No" check box should deleted. The indefinite articles "a" or "an" should also be used at the beginning of each example to make the structure uniform. Rewritten, the portion of the paragraph would read as follows:

- "Yes. Check any below that apply and fill in the details for each business."
- "A sole proprietor or self-employed in a trade,
- "A Member member of a limited liability company
- "An Owner owner of at least 5%
- "No. Go to Part 12.
- "Yes. Check all that apply and fill in the details below for each business.

Part 12 requires the debtor's signature. Other forms calling for the debtor's signature contain an explicit warning about the consequences of making a false statement. The NCBJ recommends placing the same warning here and making it identical to the warnings in the other forms:

"If you make a false statement, you could be fined up to \$250,000, imprisoned up to 5 years, or both. 18 U.S.C. §§ 152 and 3571."

<u>USC-RULES-BK-2013-0001-0105</u> – <u>Annette Crawford, M.D. Louisiana, NACTT</u> – New Schedule I left off some very valuable information, marital status. Now proposed SOFA also removes the requirement to give the name of a current or ex-spouse.

It is very important to have the marital status and current or ex spouse's name when investigating the debtor's financial situation. It assists in looking for property settlements, and transfers of assets. It also allows at trustee to discover income that is not being reported and assists in reviewing tax returns which we are required to collect and examine, for erroneous head of household claims.

Expand Question 2 on the Proposed Statement of Financial Affairs to require the debtor to pick a marital status from

- a) married.
- b) married but separated,
- c) divorced and
- d) single (never married),

and require the name of any current or ex-spouse within the past 8 years.

<u>USC-RULES-BK-2013-0001-0151 – NABT</u> – ARE EITHER DEBTOR 1'S OR DEBTOR 2'S DEBTS PRIMARILY CONSUMER DEBTS? line 5 fails to distinguish as between Debtor 1 or Debtor 2 in a jointly administered case, in which the responses could be different. As previously noted, inclusion of ownership or designation as between Debtors is necessary in a jointly administered case.

April 22-23, 2014 Page 562 of 596

Within 1 year before you filed for bankruptcy, did you make a payment on a debt you owed anyone who was an insider? 7. Within 1 year before you filed for bankruptcy, did you make any payments or transfer any property on account of a debt that benefited an insider? lines 6 & 7 as to Insider payments, in addition to the description, should also include the category which the Debtor determines the transferee is an "insider," by example "Parent," "Partner," "Officer," etc.

April 22-23, 2014 Page 563 of 596

Comments on Official Form 112

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 112 is the statement of intention for individuals filing chapter 7 cases. The NCBJ recommends adding a comma and the word "if" in the first sentence of the instructions at the top of the form. Rewritten, the sentence would read:

If you are an individual filing under Chapter 7 and creditors have claims secured by your property, or if you have leased personal property and the lease has not expired, . . .

The NCBJ also recommends changes to the check boxes and text under: "What do you intend to do with the property that secures a debt?" The 1st checkbox says "give the property to the creditor."

The word "give" should be changed to "surrender." "Surrender" is the term used in the Code, see 11 U.S.C. § 521(a)(2)(A), and courts disagree on what constitutes the "surrender" of property, see, e.g., In re Pratt, 462 F.3d 14, 18 (1st Cir. 2006) (noting that "surrender" does not necessarily contemplate physical delivery).

To make the text with the check boxes parallel, the phrase "I will" next to the 3rd and 4th check boxes should be deleted (so that the text would simply say "Redeem the property" and "Sign a Reaffirmation Agreement).

<u>USC-RULES-BK-2013-0001-0078 – Sheryl K. Ith of Cooksey, Toolen, Gage, Duffy & Woog</u> – The Committee Note to Official Form 112, Statement of Intention for Individuals Filing Under Chapter 7, states that the form was revised to "make it easier to read" and "uses language likely to be understandable to non-lawyers."

The Note goes on to say that Part 1 "refers to signing a 'Reaffirmation Agreement' rather than asking whether the debtor intends to 'reaffirm the debt.'" CTGDW asserts that if a debtor does not understand what it means to reaffirm a debt, then that same debtor is not likely to understand what it means to sign a "Reaffirmation Agreement."

Further, the change in the form allowing a debtor to state his intent to "sign a Reaffirmation Agreement" infers that in merely signing the agreement the debtor will have complied with the duties set forth in 11 USC §§521 and 524. However, this is not the case as 521(a)(2)(A) states "that the debtor intends to reaffirm debts secured by such property" and 521(a)(6) provides that debtor shall not retain possession of personal property unless the debtor enters into an agreement which the creditor pursuant to 524(c) with respect to the claim secured by such property.... Sections 524(c) and (d) require the debtor to do much more than simply "sign a Reaffirmation Agreement" to reaffirm a debt.

This committee does a great disservice to the debtor who relies on the forms and allows the debtor to believe that simply signing a Reaffirmation Agreement results in the debt being reaffirmed.

April 22-23, 2014 Page 564 of 596

<u>USC-RULES-BK-2013-0001-0155 – David S. Yen</u> – The form should use the word "surrender," not the word "give".

This form is required by section 521(a)(2) of the Code. That section requires that a debtor who has debts secured by property of the estate state his or her intention with respect to "retention or surrender" of such property.

Surrender of property is not equivalent to physical transfer of property. For tangible personal property, use of the word "give" implies transfer of physical possession of the thing itself, or sometimes a key or other device. Such actions are not required when a debtor is surrendering property. Indeed, if more than one creditor has a lien on a piece of property, the debtor cannot give the property to both. What the debtor can do is surrender the property, and let secured creditors or other interested parties resolve issues of ownership and possession between themselves.

Use of the word "give" may also be misconstrued as requiring the debtor to take affirmative steps and incur expenses in doing so.

April 22-23, 2014 Page 565 of 596

Comments on Official Form 119

<u>USC-RULES-BK-2013-0001-0042 – Henry Sommer, NACBA</u> – The forms omit the sections in each form that are to be signed by a nonattorney bankruptcy petition preparer. Section 110(b)(1) requires each document prepared by a petition preparer to be signed by the preparer, with the preparer's name, address and corporate information, if any. The statute does not permit these separate signatures to be combined into one form as the proposed forms appear to do.

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – 119 is the bankruptcy petition preparer's notice, declaration, and signature. The NCBJ suggests that the introductory paragraph's final sentence concerning potential penalties for non-compliance be rewritten (a) to make its structure consistent with the structure of similar sentences in other forms and (b) to be accurate.

The inaccuracy lies in the warning that petition preparers who do not "comply with the provisions" of title 11 or the Bankruptcy Rules may be "fined and imprisoned." Under 18 U.S.C. § 156(b), however, the offense is committed if a case or related proceeding is "dismissed" because of a petition preparer's "knowing attempt" to disregard the Bankruptcy Code or Rules, and the penalty is a fine, imprisonment, or both. The sentence should be rewritten as follows:

A bankruptcy petition preparer who does not comply with the provisions of title 11 of the United States Code whose knowing attempt to disregard the Bankruptcy Code or the Federal Rules of Bankruptcy Procedure causes a bankruptcy case to be dismissed may be fined, and imprisoned, or both. 11 U.S.C. § 110; 18 U.S.C. § 156.

April 22-23, 2014 Page 566 of 596

Comments on Official Form 121

<u>USC-RULES-BK-2013-0001-0056</u> – <u>Scott Ford, BCAG</u> – The introduction to this form suggests that the debtor's SSN may appear on multiple notices in a case, raising additional concerns regarding exposure. The BCAG suggests deleting the following sentence: "To help creditors correctly identify a case, full Social Security Numbers may appear on an electronic version of some notices." The only notice that contains the full SSN is the 341 notice. There are two versions of the 341 notice in each case. One version contains the full SSN, which is mailed or transmitted electronically to the specified recipients pursuant to Rule 2002(a)(1). The other version of the 341 notice that appears on the case docket is redacted.

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 121 is the statement about the debtor's social security number. The NCBJ suggests rewriting the final sentence in the instruction to make its wording and structure consistent with the wording and structure of similar sentences in other forms and to add the applicable statutory citations. The sentence should be rewritten as follows:

If you do not tell the truth make a false statement on this form, you may be fined up to \$250,000, you may be imprisoned for up to 5 years, or both. 18 U.S.C. §§ 152 and 3571."

<u>USC-RULES-BK-2013-0001-0089</u> – <u>Scott W. Ford, Clerk of Court</u> – The introductory language should be clarified to ensure that this form is filed not in paper but submitted in CM/ECF as a private entry available to internal court users only.

In 2012, the B21 form was revised to include cautionary language not to file the B21 as part of the public case file. Generally, clerks' offices feared this language would increase the numbers of B21 s filed conventionally. This clerk's office has not observed a spike in paper B21 s; nonetheless, we suggest clarifying the introductory language to ensure that this document is submitted electronically and privately and not in paper.

<u>USC-RULES-BK-2013-0001-0140 – Penelope Souhrada</u> – The instructions certainly imply that it is normal and legal to use more than one social security number.

April 22-23, 2014 Page 567 of 596

Comments on B318: Order of Discharge

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 318 is the discharge order (currently Form 18). Unlike the other forms, it lacks a heading with the form number. The NCBJ suggests placing the form number at the top so that the format is the same as the format of other forms.

The NCBJ also objects to the revised format for the notices accompanying the discharge order in proposed Form 381. The distinction between information "to the creditors" and "to the debtor" is artificial. The revised format misleadingly suggests that debtors need only read the "Notice to the debtor" and creditors should only read the "Notice to the creditors." Yet, each constituency would benefit from reading the information contained under the heading intended for the other. The NCBJ recommends retaining the existing format with a single notice.

<u>USC-RULES-BK-2013-0001-0151 – NABT</u> – NABT believes that Proposed Form 318 has serious deficiencies as it relates to case administration. The Order of Discharge should explain that the trustee's administration of the case can extend beyond the discharge date.

Proposed Form 318 should indicate that the issuance of the discharge is not the closing of the case if the trustee is still administering the case.

Debtor's non-exempt property remains subject to administration of the trustee until the case is closed and debtors are not at liberty to sell or transfer property without a court order while the case remains pending.

Further, the Proposed Form provides "a creditor with a lien may enforce a claim against the debtors' property subject to that lien." The Form should, as noted above, warn that property, even with such a lien, remains property of the estate until the case is closed or the property abandoned.

April 22-23, 2014 Page 568 of 596

Comments on Form 423: Certification about a Financial Management Course

<u>USC-RULES-BK-2013-0001-0056 – Scott Ford, BCAG</u> – The second paragraph to the introductory information to this form includes only a "does not" scenario.

[Original: "After you finish the course, the provider will give you a certificate. The provider may notify the court that you have completed the course. If the provider does not do so, then Debtor 1 and Debtor 2 must each file this form with the certificate number before your debts will be discharged."]

The BCAG suggests adding a "does" scenario as follows: "The provider may notify the court that you have completed the course. If the provider does, Debtor 1 and Debtor 2 need not file this form. If the provider does not do so, then Debtor 1 and Debtor 2 must each file this form with the certificate number before your debts will be discharged."

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 423 is the debtor's certification regarding the financial management course requirement. The NCBJ has stylistic suggestions. The first sentence should be modified to read as follows:

If you are an individual, and if you filed for bankruptcy under chapter 7 or 13, or filed bankruptcy under chapter 11 and § 1141(d)(3) does not apply, applies ..."

April 22-23, 2014 Page 569 of 596

Comments on Form 427: Cover Sheet for Reaffirmation Agreement

<u>USC-RULES-BK-2013-0001-0056</u> – <u>Scott Ford, BCAG</u> – The BCAG recommends that the instructions before Part 1 of this form be modified to remove "attach it to the reaffirmation agreement," and instead include "file it as one document with the reaffirmation agreement." The BCAG recommends placing the emphasis on filing the agreement instead of the cover sheet. The BCAG also recommends clarifying which forms (240A and/or 240A/B ALT) require the submission of this cover sheet. Much of the information included on Form 427 duplicates the information required by Form 240A.

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Form 427 is the cover sheet for reaffirmation agreements and revises Form 27. The NCBJ recommends changes to line 11.

The question in bold—"Did counsel represent the debtor in negotiating the reaffirmation agreement?" – takes a position on a legal issue about which courts disagree. The relevant Code section requires the debtor's attorney to sign the certification if the debtor "was represented by an attorney during the negotiation of [the] reaffirmation agreement," 11 U.S.C. § 524(k)(3)(J)(i). Courts differ on what "during the negotiation" means. Some interpret that phrase to require the attorney's actual participation in the negotiation; others find it enough merely that the debtor was represented by an attorney in the bankruptcy case at the time. Rather than take a position, line 11 should be revised to use the statutory language. Also, sections 524(k)(3)(J)(i) and (k)(5) require a "declaration" from debtor's attorney, not a "declaration or an affidavit," as the proposed form suggests. Again, the statutory language should be used.

NCBJ recommends splitting the inquiry in line 11 into 2 separate questions, one concerning the debtor's representation by an attorney, the other concerning the attorney's execution of the declaration. Having 2 questions will simplify the current structure of the paragraph and emphasize the equal importance of each question. Rewritten, the text would read:

- 11. Counsel's representation of the debtor
- a. Did counsel represent the debtor during the negotiation of in negotiating the reaffirmation agreement?
- "Yes
- "No
- b. Has counsel executed the declaration required by section 524 (k)(5) of the Bankruptcy Code?
- "Yes
- "No

April 22-23, 2014 Page 570 of 596

Comments on Instructions – General

<u>USC-RULES-BK-2013-0001-United States Trustee Program (USTP)</u> – The United States Trustee Program appreciates the continued opportunity to review and offer comments on the bankruptcy forms drafted by the Forms Modernization Group. Overall, we believe the instructions foster greater understanding, accessibility and accuracy. We like the layout and spacing of the instructions. We appreciate the additional information and detail that is provided. Also, we appreciate having one set of instructions for all individual forms.

Consider adding a warning to debtors to exercise caution with respect to personally identifiable information, and perhaps a brief description of the scope and requirements of Bankruptcy Rule 9037. For example, the following language could be used or adapted: "Bankruptcy Rule 9037 prohibits the disclosure of certain personal information in documents that are filed with the bankruptcy court. The types of information protected are individuals' social-security numbers and taxpayer-identification numbers, birth dates, the names of minor children, and financial-account numbers. An individual can waive the protections of Rule 9037 by filing a document containing his or her own personal information that is not redacted and is not filed under seal."

Consider amending the following admonition for accuracy, completeness and consistency with the forms: "If you deliberately make a false statement, you could be fined up to \$250,000 or imprisoned for up to 5 years, or both. 18 U.S.C. §§ 152 and 3571. Making a false statement, concealing property, or obtaining money or property by means of fraud in connection with a bankruptcy case is subject to a fine of up to \$250,000 and imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571."

On p. 5 for the second bullet, consider amending the following language for accuracy and completeness: "It is important that you file in the correct district within your state. 28 U.S.C. § 1408. To find out. . ."

On p. 6 for the fourth bullet, consider amending the following language for consistency with the forms: "Use these forms if a bankruptcy petition preparer helped you paid or agreed to pay someone who is not an attorney to help you fill out your forms." See Proposed Official Forms 101 and 106Dec.

On p. 7 for the last bullet, consider adding language following the checkbox for payment advices (pay stubs) so that debtors are informed that only the last four digits of a Social Security number should be made known to the general public. See Bankruptcy Rule 9037; General Comments, supra. For example, the following language could be used or adapted: "Bankruptcy Rule 9037 prohibits the disclosure of certain personal information in documents that are filed with the bankruptcy court. The types of information protected include individuals' social-security numbers and taxpayer-identification numbers. An individual can waive the protections of Rule 9037 by filing a document containing his or her own personal information that is not redacted and is not filed under seal."

April 22-23, 2014 Page 571 of 596

On p. 10 at the top of the second column, consider amending the following language for accuracy and completeness: "the U.S. trustee, bankruptcy administrator, or creditors can file a motion to dismiss your case under § 707(b)(2) of the Bankruptcy Code. Additionally, these parties can file a motion to dismiss your case under § 707(b)(3) of the Bankruptcy Code if your bankruptcy petition was filed in bad faith or the totality of the circumstances of your financial situation demonstrates abuse. If a motion is filed....."

On p. 12 in the box, consider amending the following language for accuracy: "Section 521(a)(1) of the Bankruptcy Code Bankruptcy Rule 1007 requires that you promptly file detailed information about your creditors. . . ."

In the same box, consider amending the following admonition for accuracy, completeness and consistency with the forms: "If you make a false statement, you could be fined up to \$250,000 or imprisoned for up to 5 years, or both. 18 U.S.C. §§ 152 and 3571. Making a false statement, concealing property, or obtaining money or property by means of fraud in connection with a bankruptcy case is subject to a fine of up to \$250,000 and imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571." See Proposed Official Forms 101, 106Dec, 202, and 205.

On p. 12, consider amending the following language for clarity and internal consistency: "For debtors in districts outside of Alabama and North Carolina, a list of approved agencies can be found at: http://www.justice.gov/ust/eo/bapcpa/ccde/cc_approved.htm. For debtors in districts in Alabama and North Carolina, a list of approved agencies can be found at: http://www.uscourts.gov. The clerk of the bankruptcy court also has a list of approved agencies." See Proposed Instructions – Bankruptcy Forms for Individuals, at 5.

<u>USC-RULES-BK-2013-0001-0042 – Henry Sommer, NACBA</u> – NACBA supports placing the instructions for the forms in a separate packet so the length of the forms can be reduced. There is also some merit in attempting to provide more simple instructions for pro se debtors. But it must be recognized that any attempt to simplify and generalize must sacrifice the precision of the statute, rules, and case law. Therefore, it is imperative that the Committee Notes or the form instructions themselves make clear that the instructions are not a statement of the law and should not be relied upon by courts in interpreting the law.

<u>USC-RULES-BK-2013-0001-0056 – Scott Ford, BCAG</u> – On p. 8 under the chapter 7 box, the references to Forms 108, 109, and 110 should be changed to reference the correct form number. (Numbers 22A-C).

On p. 12, This notice states that "The clerk of the bankruptcy court has a list of approved agencies." The BCAG recommends changing this sentence to accurately reflect that the U.S. Trustee Program and Bankruptcy Administrators keep a list of approved agencies. The notice should be amended as follows:

A list of approved agencies may be found at: http://justice.gov/ust/eo/bapcpa/ccde/cc_approved.html. In Alabama and North Carolina, go to:

April 22-23, 2014 Page 572 of 596

http://www.uscourts.govlFederaICourts/Bankruptcy/BankruptcyResources/ ApprovedCreditAndDebtCounselors.aspx

The BCAG notes that bankruptcy courts neither have nor maintain these lists; the courts merely provide links from their websites.

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – The NCBJ recommends a careful examination and substantial revision of the Instructions accompanying the forms before they are promulgated. Close examination of the Instructions reveals several problem, among them:

- The Instructions make statements of the law that are not always accurate.
- Most important, the Instructions purport to give legal advice. Apparently, the Instructions are intended to serve as a kind of a lay person's guide to bankruptcy law and practice. The NCBJ believes strongly that this effort is misguided, even when the guidance is accompanied by warnings that debtors should always consult an attorney rather than go it alone.

The Instructions should be limited as much as possible to guidance on how to fill out the forms. They should not attempt to distill bankruptcy law into something a lay person can understand, nor should they characterize bankruptcy practice or offer predictions on what is likely or unlikely to happen in a debtor's case.

The NCBJ offers on the Instructions with the caveat at the beginning that the NCBJ has chosen to focus its attention on the Proposed Rules and Forms. The comments below therefore should be taken only as examples of proposed revisions to the Instructions, not as an exhaustive list.

(beginning "The instructions"), the following sentence should be added:

"These instructions are not intended to provide, and should not be understood to provide, legal advice."

The warning box should be retitled: "Read These This Important Warnings."

The paragraphs in the box should be numbered to make them less daunting and so more likely to be read.

The third paragraph should be deleted, since it is inaccurate and unnecessary. If the third paragraph remains, the word "may" should be changed to "should," and the phrase "complete the bankruptcy" should be deleted. Rewritten, the third paragraph would read as follows:

You should may not file bankruptcy if you are not eligible to file or if you do not intend to file the necessary documents necessary to complete the bankruptcy.

On p. 2, the last paragraph should be revised as follows to be more specific about what petition preparers can and cannot do and to make the structure of the sentence parallel:

April 22-23, 2014 Page 573 of 596

Although bankruptcy petition preparers can help you type the bankruptcy forms, they cannot tell you how to complete the forms, they cannot file the forms documents for you, and they cannot give you legal advice.

On p. 3, The 3rd paragraph offers legal advice about the ability of a debtor to keep certain information confidential. The advice, which appears in the third sentence, is not only needless but inaccurate, since the implication is that entry of a protective order is common in the right "circumstances." In fact, protective orders in consumer bankruptcy cases are so rare as to be virtually unheard of. Rather than give this misimpression, the instructions should stress the point in the opening sentence:

"[A] bankruptcy case is not private." The NCBJ recommends deleting the 3rd sentence altogether and adding to the 2nd sentence the phrase "unless the court orders otherwise." A stylistic change to the 1st sentence is also recommended. The revised paragraph would read:

You should understand that filing a bankruptcy case is not private. Anyone has a right to see your bankruptcy forms after you file them, unless the court orders otherwise. However, in some circumstances, if a court issues a protective order to keep your address, telephone number, or other information from being disclosed to the public, it may be possible to protect your information under 11 U.S.C. § 107 and Bankruptcy Rule 9037.

The NCBJ recommends several changes to page 5.

(the paragraph beginning "Receive a briefing"), the sentence describing the consequences of failing to receive prepetition credit counseling needlessly predicts what is likely to happen in a bankruptcy case. What will happen depends on the practice in the district where the case is pending. If some warning is necessary, the NCBJ believes it is better to take no position on the likelihood of dismissal and replace the phrase "will almost certainly" with the word "may." Rewritten, the sentence would read:

Failure to do so may will almost certainly result in the dismissal of your case.

Asserting that it is "particularly difficult" to succeed in certain bankruptcy cases "without an attorney," should be deleted. The warning is out of place on this page. The same warning also appears in the right-hand column on page 2 and need not be repeated on page 5.

Several changes to page 6 are necessary.

•a debtor who has received credit counseling but not the certificate is instructed to file the certificate "when you receive it."

The instruction fails to tell the debtor that the certificate must be filed within 14 days. See Fed. R. Bankr. P. 1007(c). The 14-day deadline should be added.

April 22-23, 2014 Page 574 of 596

The last sentence of the first paragraph in the right-hand column should be deleted. The sentence tells debtors: "Waivers [of credit counseling] are rare and if you do not qualify for a waiver, your case will be dismissed."

No basis is given for this legal advice, and its characterization of bankruptcy practice is not only unnecessary but inaccurate. The categorical declaration that a case "will be dismissed" if the credit counseling requirement has not been waived does not comport with practice in many districts.

The paragraph describing Forms 101A and 101B should be rewritten to distinguish between the forms. The paragraph should make clear that Form 101A is used when a debtor wishes to remain in his leased residence for 30 days post-petition, and Form 101B should be used when a debtor wishes to remain in his leased residence beyond the first 30 days post-petition.

If, as the NCBJ urged earlier, Form 101A is redesigned, the description of the forms on page 6 should be rewritten to match the redesigned forms.

Concerning petition preparers, the instruction that a debtor must use certain forms if a petition preparer "helped you fill out your forms" is incorrect. The phrase "helped you fill out" should be replaced with the word "typed," as follows:

Use these forms if a bankruptcy petition preparer typed helped you fill out your forms.

The NCBJ recommends 2 changes on page 8.

First, the last paragraph in the left-hand column should be deleted. The paragraph consists of a single sentence that says: "If you file under chapter 11, you must also file additional documents." The sentence is too vague to be helpful and serves no purpose.

Second, the last bullet tells prospective chapter 13 debtors that they must file a "Chapter 13 Plan." It then informs debtors parenthetically that "[m]any bankruptcy courts require you to use a local form plan" and suggests that debtors consult their court's local web site. If Proposed Form 113 is approved and made mandatory, it will be used in all districts nationwide. The NCBJ takes no position on the advisability of the adoption of such a form. If Form 113 is adopted, however, there will no longer be "local form plans." The content in parentheses would then need to be deleted and replaced with "Official Form 113."

On p. 9, the first sentence of the second column should be revised. The sentence says that chapter 7 "is for individuals who have financial difficulty and cannot pay their debts." That broad description, however, is equally true of cases under chapters 11, 12, and 13. The sentence also fails to disclose a critical feature of a chapter 7 case: that a debtor's non-exempt assets are subject to sale to pay creditors. The NCBJ recommends revising the sentence as follows:

Chapter 7 is for individuals who have financial difficulty, and cannot pay their debts, and wish to retain their non-exempt property and allow their non-exempt property to be sold to pay their creditors.

April 22-23, 2014 Page 575 of 596

The second sentence is awkward, since it says the primary purpose of a chapter 7 case is "for a debtor... to have your debts discharged." The phrase "a debtor" should be changed to "you."

The last bullet of the second column describes certain nondischargeable debts. The last such debt is: "certain debts that are not properly listed in your bankruptcy papers." The word "properly" is unnecessary, is unhelpful, and should be deleted.

The list of potentially nondischargeable debts continues on page 10. The second such debt mentioned is "breach of fiduciary duty." Section 523(a)(4) of the Code, however, makes nondischargeable only debts arising from "fraud or defalcation while acting in a fiduciary capacity," 11 U.S.C. § 532(a)(4), which is narrower. To be accurate, the statutory phrase should be substituted for the phrase "breach of fiduciary duty."

(beginning "If you are an individual"), the NCBJ recommends adding this sentence: "The trustee may sell your property to pay your debts."

At the top of the second column on p. 11, the description of potential court action on a proposed chapter 13 plan is misleading, since the phrase "[t]he court must approve your plan" incorrectly implies that the court has no choice but to confirm a plan and cannot deny confirmation. The NCBJ suggests revising the sentence to make it conditional. Rewritten, the sentence would read: If the court must approves your plan, the court will and may allow you to repay your debts within 3 years or 5 years

In the same column, the word "all" should be added before "the payments" and the words "are generally" deleted before the word "discharged. Another sentence should also be added as a transition. Rewritten, the sentences would read:

After you make all the payments under your plan, your nondischargeable debts are generally will be discharged. However, not all debts are dischargeable. You may still be responsible to pay

The paragraph goes on to list some debts that are not discharged. As with the comparable phrase on page 9, the word "properly" should be deleted in the 5th example ("certain debts . . . not properly listed in your bankruptcy papers").

The NCBJ also suggests adding to the list debts nondischargeable under section 523(a)(2) and (4) of the Code, 11 U.S.C. §§ 532(a)(2), (4), since those debts are no longer discharged in chapter 13 cases, see 11 U.S.C. § 1328(a)(2), and the exceptions are among those most commonly invoked.

The box on p. 12 contains essentially the same warnings as the box on page 2. The NCBJ recommends the same revisions to the warnings on page 12. The additional warning in the box on page 12 should also be revised so that is has the same structure as the other times it appears:

If you make a false statement, you could be fined up to \$250,000, or imprisoned for up to 5 years, or both.

April 22-23, 2014 Page 576 of 596

<u>USC-RULES-BK-2013-0001-0151 – NABT</u> – Since NABT has elected to focus its attention on the Proposed Rules and Forms, it decided to forego comments on the Instructions as the National Conference of Bankruptcy Judges ("NCBJ") Comments address many of the same concerns of NABT.

April 22-23, 2014 Page 577 of 596

<u>Comments on Instructions – For Voluntary Petition, Official Form 106A/B</u>

<u>USC-RULES-BK-2013-0001- USTP</u> – Consider amending the following language for accuracy, completeness and consistency with the forms: "If you fail to list any property, you may lose the property, lose your bankruptcy discharge, be fined up to \$250,000, and be imprisoned for up to 5 **20** years. 11 U.S.C. §§ 554, 727; 18 U.S.C. §§ 152, 157, **1519**, 3559, 3571, and 3581." See Proposed Official Forms 101, 106Dec, 202, and 205.

Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

<u>USC-RULES-BK-2013-0001-0042 – Henry Sommer, NACBA</u> – The new form omits the information currently required about the nature of the debtor's interest in real property. Such information as whether the debtor is a tenant by the entireties or owns only a life estate is critical to how the property is treated in the case and should be included.

Health insurance, homeowner's insurance, and renter's insurance should not be listed as assets in this schedule. These types of insurance, like auto insurance, are more in the nature of ongoing expenses and are never liquidated by trustees. Moreover, the debtor already provides information about the existence of such insurance on Schedule J.

The definition of "current value" is unhelpful. The 2nd sentence gives "fair market value" as a synonym but fails to define "fair market value."

The 3rd sentence then defines "current value" as "how much the property is worth," a tautology. If "current value" means "fair market value," the NCBJ recommends adding the common definition of "fair market value" since it is one non-lawyers should understand.

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – The word "honestly" in the second sentence should be deleted. The adverb "honestly" adds nothing to the requirement that debtors "list everything they own," and the word does not appear elsewhere.

April 22-23, 2014 Page 578 of 596

<u>Comments on Instructions – For Voluntary Petition, Official Form 106C</u>

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Here again, the instructions offer legal advice, twice telling debtors (first in the separate box in the left-hand column and again at the top of the right-hand column) that they may lose property if they do not claim proper exemptions. The NCBJ recommends deleting both statements, leaving the 2nd sentence in the box as the only advice: "You are strongly encouraged to hire a qualified attorney to advise you."

April 22-23, 2014 Page 579 of 596

Comments on Instructions – For Voluntary Petition, Official Form 106D

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – The NCBJ has 2 stylistic comments on the text of page 17.

(beginning "If your debts") should be rewritten. The sentence says that if debts are not paid, "creditors" with secured claims may be able to get paid from specific property in which "that creditor" has an interest. Either the initial word "creditors" should be replaced with "a creditor," or the later word "creditor" should be replaced with "the creditors" (and the verb "has" changed to "have").

The two possible revisions, either of which is acceptable, would read as follows:

If your debts are not paid, a creditor with a secured claim may be able to get paid from specific property in which that creditor has an interest \dots

If your debts are not paid, creditors with secured claims may be able to get paid from specific property in which that the creditors have has an interest . . .

The last paragraph of column 1 introduces the concepts of contingent, unliquidated, and disputed claims. The 1st sentence, stating that "many claims" have a specific amount and "you clearly owe them," should be changed. Debtors owe debts, not "claims."

The paragraph also properly goes under the heading at the top of the right-hand column, since the introduction belongs with the discussion it introduces. If the paragraph is moved under the heading, the bullet points currently under the heading become redundant and should be deleted.

Last sentence of left-hand column: You must list the claims of all creditors all claims in your schedules, even if the claims are contingent, unliquidated, or disputed.

replace text, so sentence reads:

You must list the claims of all creditors in your schedules, even if the claims are contingent, unliquidated, or disputed.

This helps clarify that it's a creditor's claim against the debtor, and that claims of all creditors need to be included

April 22-23, 2014 Page 580 of 596

Comments on Instructions – For Voluntary Petition, Official Form 106E/F

<u>USC-RULES-BK-2013-0001- USTP</u> – Consider amending the following language for consistency with the forms: "If you owe wages, salaries, and commissions, including vacation, severance, and sick leave pay and those amounts were earned within 180 days before you open **filed** your bankruptcy case **petition** or ceased business." See Proposed Form 106E/F.

<u>USC-RULES-BK-2013-0001-0056 – Scott Ford, BCAG</u> – In the last sentence of the first paragraph of instructions to Schedule J, it states: "On your initial filing in Part 2, select 'Initial estimate at the beginning of the case." Upon review of Schedule J, Part 2, there does not appear to be an option to select "Initial estimate at the beginning of the case." The BCAG suggests addressing this issue to avoid confusion.

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – The NCBJ has 3 suggestions.

Last paragraph is the same as the 4th full paragraph of the left-hand column on page 17 [p.215] and should be rewritten the same way.

The 1st full paragraph in the right-hand column on page 20 introduces the concepts of contingent, unliquidated, and disputed claims, just as the last paragraph in the left-hand column on page 17 does. The text on page 20 should be rewritten the same way.

The left-hand column continues from the preceding page a list of examples of priority unsecured debts. The first definition, "deposits by individuals," should be rewritten to replace the word "took" before the word "money," since the connotation – theft – is unintended and unfortunate.

The NCBJ suggests rewriting the sentence this way:

If you received took money from someone for the who planned to purchase, lease, or rental of your property or use the use of your services but you never delivered or performed.

The 4th example is not parallel to the others. All of the other examples both on pages 21 and 22 are phrased in the conditional, beginning with "if."

If the last example is retained, it should have the same structure. Since the example is limited to debtors who own or operate grain storage, fish produce storage, or fish processing facilities, a very small number of people, the example might simply be deleted.

April 22-23, 2014 Page 581 of 596

<u>Comments on Instructions – For Voluntary Petition, Official Form 112</u>

<u>USC-RULES-BK-2013-0001-0042</u> – <u>Henry Sommer, NACBA</u> – The instructions to the Chapter 7 Statement of intention are substantively inaccurate. They describe "surrender" as "You may give the property to the creditor." Numerous courts have held that this is not required by the statute. *See In re* Pratt, 462 F.3d 14, 18-19 (1st Cir. 2006). To *surrender property* means to make the property available to the creditor. *See* Collier on Bankruptcy ¶ 521.14[3] and cases cited therein.

The instructions also diverge substantively from the language of section 365(p) regarding personal property leases. It is not at all clear that a lessor has the option of refusing assumption of the lease, as opposed to setting certain conditions for assumption.

It also not clear that the conditions can include cure of a default "before" the lease is assumed. The instructions should track the statute and say something like: "The lessor may respond that it agrees you may assume the lease and may require you to pay any past due amounts. If the lessor does respond to your request to assume the lease, you must write to the lessor within 30 days stating that you assume the lease."

April 22-23, 2014 Page 582 of 596

Comments on Instructions – For Voluntary Petition, Official Form 106H

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – Column 2 below the box lists examples of codebtors. The last example has a phrase set off with dashes – "even if not the spouse a cosigner" – that is unintelligible. If the NCBJ interprets the underlying intent correctly, the phrase should be rewritten as follows:

Nonfiling spouse – even if not the spouse is not a cosigner – where the debt is for necessities ...

April 22-23, 2014 Page 583 of 596

Comments on Instructions – For Voluntary Petition, Official Form 106I

<u>USC-RULES-BK-2013-0001- USTP</u> – Consider deleting the following language as unnecessary given prior language on this page discussing marriage and living situations: "In Part 1, line 1, fill in employment information for you and, if appropriate, for a non-filing spouse."

April 22-23, 2014 Page 584 of 596

Comments on Instructions – For Voluntary Petition, Official Form 106J

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – The 1st sentence of the form is missing the word "of": Schedule J: Your Expenses (Official Form 106J) provides an estimate of the monthly expenses

April 22-23, 2014 Page 585 of 596

Comments on Instructions – Official Forms 108, 109, and 110

<u>USC-RULES-BK-2013-0001 – USTP</u> – Consider deleting the following language as unnecessary: "A presumption of abuse does not mean you are actually trying to abuse the bankruptcy system. Rather, the simply means that you are presumed to have enough income that you should not be granted relief under chapter 7."

<u>USC-RULES-BK-2013-0001-0056 – Scott Ford, BCAG</u> – These instructions reference Forms 108, 109, and 110. The BCAG recommends changing these references to the correct form numbers (22A-C).

April 22-23, 2014 Page 586 of 596

Comments on Instructions – Official Forms 110-1 and 110-2

<u>USC-RULES-BK-2013-0001 – USTP</u> – Consider deleting the following language as unnecessary: "You must file 110—1, the *Chapter 13 Statement of Your Current Monthly Income and Calculation of Commitment Period* (Official Form 110—1) if you are an individual and you are filing under chapter 13." See 11 U.S.C. § 109(e).

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – At the beginning of the 2nd full paragraph in the left-hand column, and again at the beginning of the first full paragraph in the right-hand column, the text refers simply to "110-1" and "110-2" without the word "Form." The word "Form" should be inserted before the form number in both places.

In the 3rd paragraph of the right-hand column, the word "on" should be substituted for the word "for" in the second sentence:

In those cases, the form has instructions on for where to find the information you need.

April 22-23, 2014 Page 587 of 596

Comments on Instructions – Official Form 112

<u>USC-RULES-BK-2013-0001 – USTP</u> – On p. 35 in the fourth paragraph, consider amending the following language for consistency: "11 U.S.C. § 522(f)".

On p. 36, consider deleting the following language as redundant: "The creditor may, at its option, notify you that it is willing to have you assume the lease and may condition the assumption on cure of any outstanding default, respond by telling you whether it agrees that you may assume the lease and may require you to pay any past-due amounts before you can do so."

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – The NCBJ has several stylistic suggestions.

The initial sentence in the first full paragraph of the left-hand column on page 34 should be rewritten as follows:

You must fill out the *Statement of Intention for Individuals Filing Under Chapter* 7 (Official Form 112) if you are an individual filing under chapter 7, or if your case has been converted to chapter 7 and creditors have claims secured by your property, or if you have any unexpired leases of personal property.

The initial sentence of the second paragraph of the second column should be rewritten to avoid the split infinitive, replace the word "give" with the word "surrender," and remove the needless comma:

You may choose to either to give surrender the property to the creditor, or keep the property.

The third paragraph of the second column should be revised in several respects. The word "surrender" should replace the word "give" in the initial sentence in bold type. The italicized parenthetical about surrender in the middle of the second sentence is unnecessary and should be deleted. The word "give" in the undeleted portion of the sentence should be replaced with the word "surrender." The remainder of the 2nd sentence does not mention a sale of the property and so does not explain how a "deficiency" might arise. The sentence should be rewritten as follows:

You may give surrender the property to the creditor. If you give surrender the property to the creditor (you surrender the property), your bankruptcy discharge will protect you from any claim for the difference between what you owe the creditor and what the creditor receives from a sale of the property a deficiency if the property is worth less than what you owe the creditor, unless the court determines that the debt is nondischargeable.

A comma should be inserted between the word "creditor" and the word "and" in the last sentence beginning in the right-hand column on page 34 and continuing on page 35:

You may seek to reaffirm the debt if you sign a Reaffirmation Agreement, which is a contract between you and a creditor, and you follow the proper procedure

66

The 1st sentence of the last paragraph on p. 35 should be revised as follows:

To obtain court authorization to redeem your property, you must file a motion with the court to redeem.

April 22-23, 2014 Page 589 of 596

Comments on Instructions – Official Form 103A

<u>USC-RULES-BK-2013-0001 – USTP</u> – Consider amending the following language for consistency with the forms: "If a bankruptcy petition preparer helped you paid or agreed to pay someone who is not an attorney to help you complete this form" <u>See</u> Proposed Official Forms 101 and 106Dec.

April 22-23, 2014 Page 590 of 596

Comments on Instructions – Official Form 104

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – The first sentence in the third full paragraph in the left hand column of page 38 (below the bullet points, beginning "If your debts are not paid") has the same problems as the last paragraph in the left-hand column on page 20 and the 4th full paragraph of the left-hand column on page 17 and should be rewritten the same way. The last 2 sentences of the paragraph are confusing and appear to be missing words. They should be rewritten as follows:

If a creditor has a security interest in your property, but the value of the property security available to pay the creditor is less than the amount you owe the creditor, the creditor has both a secured and an unsecured claim against you. The amount of the unsecured claim is the total claim minus the value of the property security that is available to pay the creditor.

April 22-23, 2014 Page 591 of 596

<u>Comments on Instructions – Glossary</u>

<u>USC-RULES-BK-2013-0001 – USTP</u> – Consider adding a definition for the term "meeting of creditors" in the Glossary as it constitutes an important term used in the forms. <u>See</u> Proposed Official Forms 112 and 423. For example, the following language could be used or adapted: "Meeting of creditors — A meeting that the debtor is required to attend where creditors, any trustee or examiner in the case, and the United States Trustee may examine the debtor under oath."

Consider rewording the definition of "exempt property." Debtors do not always get to keep the property. Sometimes, for example in the case of real property, the property is sold, and the debtor merely retains a right to a portion of the proceeds of that sale, not necessarily the property itself. For example, the following language could be used: "Property, or proceeds from the sale of property, that the law permits you to keep."

<u>USC-RULES-BK-2013-0001-0056 – Scott Ford, BCAG</u> – The Committee has added a Glossary with definitions of terms that appear throughout the forms. The BCAG suggests that the Committee consider including the following terms:

```
"publicly traded stock" (106A/B)
```

<u>USC-RULES-BK-2013-0001-0059 – NCBJ</u> – The Advisory Committee may want to consider whether it is helpful to provide a glossary. The glossary is presumably intended to assist pro se filers, the same intention behind many of the statements in the proposed forms about which the NCBJ has commented above. As the glossary currently stands, however, its definitions tend to be more intimidating than helpful.

And intimidating as they can be, the definitions often actually oversimplify complex concepts, leading to potential confusion. Many of what purport to be definitions also cross the line and shade into legal advice. In the NCBJ's view, only so much can be done to make bankruptcy, a

April 22-23, 2014 Page 592 of 596

[&]quot;accounts receivable" (106NB)

[&]quot;tangible property" and "intangible property" (106A/B Instructions)

[&]quot;exempt property" (better define) (106C)

[&]quot;homestead exemption" (106C)

[&]quot;community debt" (1060)

[&]quot;judgment lien" (1060)

[&]quot;statutory lien" (1060)

[&]quot;garnishment" (1060)

[&]quot;deed of trust" (1060)

[&]quot;community claim" (106EIF)

[&]quot;cosigner" (106H)

[&]quot;guarantor" (106H)

[&]quot;former spouse" (106H)

[&]quot;unmarried partner" (I 06H)

[&]quot;joint contractor" (106H Instructions)

[&]quot;fair market value" (106C) - also defined in instructions for Schedule AlB – currently grouped under "current value" but perhaps should have a separate entry in the Glossary.

complex area of law, understandable to people who have chosen to proceed without lawyers. At a certain point, the effort does more harm than good.

If the Advisory Committee decides to retain the glossary, the NCBJ recommends that it be subjected to a thorough examination – so thorough that the Advisory Committee might consider whether the glossary can be made ready in time to be promulgated with the rest of the Instructions. The definitions should be reviewed carefully

- (1) to ensure their structure and format are consistent (each definition currently appears to have had a different author or been drawn from a different source);
- (2) to ensure that the information they contain is accurate (some of the definitions are incorrect); and
- (3) to eliminate any legal advice.

The following [items listed separately] are examples of problems with the glossary. These examples are offered with the NCBJ's earlier caveat. NCBJ has focused its attention on the Proposed Rules and Forms. The examples therefore are not intended as an exhaustive list of proposed revisions.

The title at the top of the page should be changed to add the phrase "of Terms" after the word "Definitions" so that the title makes clear the glossary is providing "Definitions of Terms" used in the forms. The terms are used in the forms, not the definitions themselves.

In the first sentence of the introductory paragraph, the word "of" should also replace the word "for":

Here are definitions of for some of the important terms used in the forms

In addition, the NCBJ recommends adding the following sentences, perhaps in bold type, at the end of the introductory paragraph:

"These definitions are intended only to provide guidance. They are not a substitute for legal advice."

The indefinite article "a" should replace the definite article "the" wherever it appears before a defined noun and should be added wherever it is absent. So, for example, the definition of "Creditor" should begin, "A person or organization," not "The person or organization." The definition of "Creditor with secured claims" similarly should begin "A creditor" (and "creditor" should be singular, not plural, since the term defined is singular).

The definition of "Dependent" does not require the introductory phrase, "The term dependent generally means," and since the term defined is singular, the definition should also be phrased as singular. The first sentence should be rewritten to read: "Generally, someone who is

April 22-23, 2014 Page 593 of 596

economically dependent on the debtor regardless of whether the person can be claimed as a dependent...." The subject and verb in the next sentence disagree; the error should be fixed.

The definition of "Disputed claim" should be changed from the conditional "If you disagree about whether you owe a debt," a structure most of the other definitions lack, to: "A debt you do not agree you owe."

The definition of "Eviction judgment" should be changed to make its structure consistent with the structure of the other definitions: "A judgment for possession your landlord has obtained"

The definition of "Executory contract" is incorrect as a matter of law, since it fails to acknowledge that a contract is not executory if payment is the only contractual obligation left unfulfilled. See, e.g., Johnson v. Smith (In re Johnson), 501 F.3d 1163, 1174 (10th Cir.2007). The definition of "Exempt property" is also incorrect. "Exempt property" is not simply property that "the law permits you to keep," a definition that, all other things being equal, applies to any property belonging to a person. The definition also fails to mention that a debtor in bankruptcy has certain obligations he must satisfy before property can be "exempt."

The definition of "Legal equivalent of a spouse" contains a grammatical error. Because "person" is the subject of the verb "having" and the phrase "applicable nonfederal law recognizes" merely modifies "person," the definition should begin "A person who," not "A person whom." To avoid any grammatical questions, as well as to make the style more graceful, the word between "person" and "applicable" might be eliminated altogether. The definition would begin: "A person applicable nonfederal law recognizes as having"

The definition of "Non-individual debtor" – "You are filing for bankruptcy on behalf of a non-individual" – has a structure inconsistent with the structure of the other definitions and is not legally correct. The definition also encourages the unauthorized practice of law: only a lawyer can file bankruptcy (not "for bankruptcy," the phrase that appears many times in the Proposed Forms and Instructions) "on behalf of" an artificial entity. A better definition of "non-individual debtor," one lacking these problems, would be: "A debtor that is not a human being – for example, an artificial entity such as a corporation, partnership, or limited liability company."

The definition of "Sole proprietorship" is confusing. Rewritten to be clearer (and to avoid addressing the reader as "you"), the definition would read:

A business that a person you own operates as an individual and is not rather than a separate legal entity such as a corporation, partnership, or LLC.

The definition of "Unliquidated claim" begins with a conditional phrase, the same problem as the definition of "Disputed claim" discussed earlier. Rewritten to make the structure consistent with the structure of other definitions, the definition would begin:

If the A debt with an amount of a debt that cannot be readily determined...

April 22-23, 2014 Page 594 of 596

Other Miscellaneous Comments

Schedules I+J

<u>USC-RULES-BK-2013-0001-0031 – Walter Oney</u> – Commenter attached samples of Schedule I and J that he redesigned.

My sch I has less vertical space in part 1 than the official form; my sch I uses grid lines rather than underlines to delineate the dollar amounts and spreads out the table to use more of the horizontal width of the page. Given this, my sch I doesn't put a box around the subtotals on lines 4, 9, and 10; my sch I centers the dollar amounts on lines 10-12 below the two column display for lines 2-9. This is why I could spread out the two-column display -- I did not need to waste a wide margin just so lines 10-12 could stand alone at the right-hand side of the columnar display; my schedules use an initial capital on the word "Page" appearing in the page footers; my sch J uses grid lines throughout rather than boxes around individual dollar amounts.

Form 201 – Voluntary Petition

<u>USC-RULES-BK-2013-0001 – Professor Gibson</u> – Add instruction at line 8 for attaching small business documents statement:

If you indicate that the debtor is a small business as defined in 11 U.S.C. § 101(51D), you must append the attachments required under 11 U.S.C. § 1116(a)(1).

Rule 9009 and Form 103A Application and Order to Pay Filing Fees in Installments

<u>USC-RULES-BK-2013-0001-0067 – Dana C. McWay, Clerk of Court</u> – Proposed Rule 9009 requires parties to use the Official Forms prescribed by the Judicial Conference without alteration, with limited exceptions. It is my view that members of the practicing bar who abide by this provision will submit to bankruptcy courts documents that the courts do not want to receive and will cause additional work for Bankruptcy Clerk's Offices. Courts do not want to receive and will cause additional work for Bankruptcy Clerk's Offices.

Official Form B3A (to be renumbered B103A) Application and Order to Pay Filing Fee in Installments illustrates this unintended consequence clearly. Official Form B3A consists of 2 pages: 1 for applying to pay the filing fee in installments and 1 to submit as a proposed order to the court. Our court, and many others across the nation, no longer expects counsel or parties to submit the second page as a proposed order. Instead, the court leverages the CM/ECF system to create the order concerning payment of the filing fee in installments.

The content of the court's order complies with the content of the 2nd page of Official Form B3A and is created and entered at the direction of the judge to whom the case has been assigned in our court. No proposed order is submitted by counsel or a party that the court must route and

73

subsequently docket. In fact, the court specifically instructs counsel and parties not to submit proposed orders with installment payment applications.

If counsel or parties abide by proposed Rule 9009, I believe our court, and other courts, will see submission of these proposed orders for installment payments. This will cause deputy clerks to spend time identifying which of the proposed orders received by the court are the ones that should be submitted and which ones should not be submitted and determining what to do with those proposed orders that should not be submitted. The time of deputy clerks would be better spent not parsing through stacks of proposed orders that should not be received by the court in the first place, but attending to the remainder of their work.

April 22-23, 2014 Page 596 of 596